

Progress in Neurobiology Vol. 57, pp. 1 to 164, 1999 © 1999 Elsevier Science Ltd. All rights reserved Printed in Great Britain 0301-0082/98/\$19.00

PII: S0301-0082(98)00048-3

THE INDUCTION OF PAIN: AN INTEGRATIVE REVIEW

MARK J. MILLAN*

Institut de Recherches Servier, Centre de Recherches de Croissy, Psychopharmacology Department, 125, Chemin de Ronde, 78290 Croissy-sur-Seine, Paris, France

(Received 30 March 1998)

Abstract—The highly disagreeable sensation of pain results from an extraordinarily complex and interactive series of mechanisms integrated at all levels of the neuroaxis, from the periphery, via the dorsal horn to higher cerebral structures. Pain is usually elicited by the activation of specific nociceptors ('nociceptive pain'). However, it may also result from injury to sensory fibres, or from damage to the CNS itself ('neuropathic pain'). Although acute and subchronic, nociceptive pain fulfils a warning role, chronic and/or severe nociceptive and neuropathic pain is maladaptive. Recent years have seen a progressive unravelling of the neuroanatomical circuits and cellular mechanisms underlying the induction of pain. In addition to familiar inflammatory mediators, such as prostaglandins and bradykinin, potentially-important, pronociceptive roles have been proposed for a variety of 'exotic' species, including protons, ATP, cytokines, neurotrophins (growth factors) and nitric oxide. Further, both in the periphery and in the CNS, non-neuronal glial and immunecompetent cells have been shown to play a modulatory role in the response to inflammation and injury, and in processes modifying nociception. In the dorsal horn of the spinal cord, wherein the primary processing of nociceptive information occurs, N-methyl-Daspartate receptors are activated by glutamate released from nocisponsive afferent fibres. Their activation plays a key role in the induction of neuronal sensitization, a process underlying prolonged painful states. In addition, upon peripheral nerve injury, a reduction of inhibitory interneurone tone in the dorsal horn exacerbates sensitized states and further enhance nociception. As concerns the transfer of nociceptive information to the brain, several pathways other than the classical spinothalamic tract are of importance: for example, the postsynaptic dorsal column pathway. In discussing the roles of supraspinal structures in pain sensation, differences between its 'discriminative-sensory' and 'affective-cognitive' dimensions should be emphasized. The purpose of the present article is to provide a global account of mechanisms involved in the induction of pain. Particular attention is focused on cellular aspects and on the consequences of peripheral nerve injury. In the first part of the review, neuronal pathways for the transmission of nociceptive information from peripheral nerve terminals to the dorsal horn, and therefrom to higher centres, are outlined. This neuronal framework is then exploited for a consideration of peripheral, spinal and supraspinal mechanisms involved in the induction of pain by stimulation of peripheral nociceptors, by peripheral nerve injury and by damage to the CNS itself. Finally, a hypothesis is forwarded that neurotrophins may play an important role in central, adaptive mechanisms modulating nociception. An improved understanding of the origins of pain should facilitate the development of novel strategies for its more effective treatment. © 1998 Elsevier Science Ltd. All rights reserved

CONTENTS

1.	Introduction: aims and scope of review	6
2.	Acute and chronic pain: origins and pathophysiological significance from an evolutionary perspective	7
	2.1. The adaptive significance of acute vs chronic pain	7
	2.2. Lack of endogenous mechanisms for the effective control of chronic pain	9
3.	Receipt, encoding and transmission to the DH of primary afferent, nociceptive information	9
	3.1. Sensory properties of PAFs: detection of noxious stimuli	9
	3.1.1. Cutaneous C, A δ and A β fibres encoding noxious and innocuous information	9
	3.1.2. Nocisponsive fibres in other organs	12
	3.1.3. 'Silent' nociceptors	13
	3.1.4. 'Ectopic' functions of damaged PAFs and 'central' pain	13
	3.2. Role of neurotransmitters and neuromodulators in the transmission of nociceptive information by PAFs	
	to the DH of the spinal cord	13
	3.2.1. Multiple mediators of nociceptive information in PAFs: interaction with intracellular transduction	
	mechanisms	13
	3.2.2. Role of colocalized EAAs, tachykinins and CGRP in small calibre PAFs projecting to the DH	14
	3.2.3. Multiple isoforms of NMDA, AMPA and mGlu receptor	15
	3.2.4. Actions of EAAs and SP at PAF terminals in the DH	16
	3.2.5. Cooperative, excitatory actions of EAAs, tachykinins and CGRP on DH neurones	16
	3.2.6. Other PAF-derived neuropeptides potentially modulating nociception in the DH	16
	3.2.7. Putative neurotransmitters in A β fibres	18

^{*} Tel.: 33.1.41.18.24.25; Fax: 33.1.41.18.24.70.

Mark J. Millan

CONTENTS (continued)

3.2.8. Pronociceptive mediators in the DH derived from PAF terminals and intrinsic DH sources: focus	
on ATP, NO and PGs	18
3.2.8.1. Unconventional mediators of nociception in the DH	18
3.2.8.2. A1P	18
3.2.8.3. NO 2.2.9.4 DCa	18
3.2.6.4. FOS	20
4 Spinal targets of primary afferent nociceptive information: intrinsic elements in the DH	20
4.1. Laminae organization of the DH	21
4.2. Response characteristics of DH neurones	21
4.3. Receptor fields of DH neurones	21
4.4. Output targets of DH neurones	23
4.5. PNs and EXINs in the DH	23
4.6. INTING IN THE DH	24
4.6.1. Actions and neuronennear classes of nume 4.6.2. A <i>β</i> fibre activation of ININs: the absence of pain upon selective A <i>β</i> fibre stimulation	24
4.6.3. Consequences of a loss of ININ tone in the DH	25
4.7. Novel modes of neuronal communication	25
4.8. Glial and immunecompetent cells: focus on cytokines	26
4.8.1. Glial cells	26
4.8.2. Microglial cells	27
4.8.3. Actions of cytokines; relevance to nociception	27
4.8.4. Tissue-protective and instale-destructive actions of cytokines	27
5. Modulation of nociceptive processing in the DH by descending inhibitory and <i>facilitatory</i> pathways	28
5.1. Descending inhibition	28
5.2. Descending facilitation	29
6. Transfer of nociceptive information to supraspinal centres: discriminative-sensory and affective-cognitive	
dimensions of pain	29
6.1. Ascending nociceptive pathways: organization	29
6.1.1. Ascending pathways, neurotransmitters and uner roles	29
6.1.2. Ascending pathways: supraspinal tagets of nedeptive mornatoria	32
6.1.4. Ascending pathways: sensory-discriminative and affective-conitive dimensions of pain	32
6.1.5. Visceral nociceptive input to the thalamus	33
6.2. Cerebral targets of ascending nociceptive pathways: functional roles	34
6.2.1. The key role of the thalamus	34
6.2.2. Contrasting roles of thalamic nuclei with respect to the sensory-discriminative and affective-	24
6.2.3. Neurotransmitters involved in the processing of nocicentive transmission in the thalamus: focus on	54
EAAs and GABA	35
6.3. Thalamic input to the cortex: differential integration of sensory-discriminative as compared to affective-	
cognitive aspects of pain	36
6.3.1. Neurotransmitters involved in processing nociceptive information in the cortex	36
6.3.2. Cortical structures activated by noxious stimulation: patterns of responsiveness	36
6.5.5. Patterns of inaliamic input to the cortex: relationship to discriminative-sensory and cognitive-	37
6.3.4 Contrasting roles of specific cortical regions: encoding properties of nocisponsive neurones	38
6.3.5. Relationship of cortical structures to limbic regions involved in nociceptive processing	39
6.3.6. Pain and mood: focus on monoamines and opioids	39
6.3.7. Sensory-discriminative and emotional-cognitive dimensions of pain: a summary of complementary	
roles	40
6.4. Modulation of descending pathways by ascending, nociceptive information	41
7. Nocceptive, initiality pain, peripheral picesses	41
7.2 Primary hyperalgesia: thermal and mechanical hypersensitivity	41
7.3. Multiple, interactive, peripheral mechanisms of hyperalgesia	42
7.4. Substances modulating the activity of nocisponsive PAF terminals: tissue sources, receptors implicated	
and excitatory actions	42
7.4.1. Orthodromic and antidromic activation of PAFs	42
7.4.2. BN 7.4.3. Prostatlandins and leukotrienes	42
7.4.4. ATP	45
7.4.5. Protons and vanilloids	46
7.4.5.1. Excitation of PAF terminals by acidosis (a reduction in pH)	46
7.4.5.2. Relationship of proton-gated ion channels to vanilloid-gated ion channels	47
7.4.5.4. Influence of motions (figure still still) we are the structure of the figure of motion of motions (figure still still) we are the structure of the structure still still structure of the structure still structure still structure still structure str	47
7.4.5.4. Influence of protons (ussue acidosis) upon other receptor types	48 19
7.4.7. NAD derived from the sympathetic system	+0 49
7.4.8. Histamine	49
7.4.9. EAAs, tachykinins and CGRP	49

2

		The Induction of Pain: an Integrative Review	3
		CONTENTS (continued)	
	7.5.	Mechanisms underlying sensitization of PAF terminals	50
		7.5.1. Synergistic sensitization of PAFs	50
		7.5.2. Ionic and metabotropic mechanisms involved in sensitization	51
		7.5.2.1. Multiple mechanisms of sensitization	51
		7.5.2.2. Increases in levels of CAMP $7.5.2.3$ Activation of a tetrodotoxin-resistant Na ⁺ -current	51
		7.5.2.4 Activation of PKC via PLC	51
		7.5.2.5. Inhibition of K^+ -currents	52
		7.5.2.6. Activation of NO synthase	52
		7.5.2.7. A potential role of NGF	52
		7.5.2.8. General significance of an induction of PKs and of an increase in $[Ca^{2+}]_i$	53
		7.5.2.9. Recruitment of silent nociceptors	53
	/.6.	Altered phenotype of PAF neurones	53
		7.6.1. Focus on pronoceceptive actions of NGF at TKK A receptor-bearing, small canore C notes	55
	7.7.	Modulatory effects: altered activity and phenotype of PAFs mediated <i>indirectly</i> by other elements of the	55
		inflammatory response	55
	7.8.	Peripheral antinociceptive mechanisms	58
	7.9.	Antidromic activation of PAFs: NI, vasorelaxation and migraine headache	58
		7.9.1. The efferent role of PAFs	58
		7.9.2. NI, vascular mechanisms and migraine	59
0	Nou	/.9.3. Vasorelaxation as a trigger of migraine headaches: a role for NO and 5-H1?	60
ð.	Neu 8 1	Experimental models of neuropathic pain	61
	8.2	Perinheral models of neuropatine pain Perinheral mechanisms underlying neuropathic pain	62
		8.2.1. Inflammatory events: focus on cytokines	62
		8.2.2. Alterations in the density and function of ion channels in sensory neurones	63
		8.2.2.1. Ectopic activity of damaged PAFs	63
		8.2.2.2. Ca^{2+} -channels	63
		8.2.2.3. Na ⁺ -channels	63
		8.2.2.4. K ² -channels β_{12} contraction solution in the contract (C_{12}^{2+1}) and DAE reconcentric	64
		6.2.2.3. Spontaneous electrical activity, for currents, [Ca]; and FAF regeneration	65
		8.2.4 Sympathetic excitation of damaged sensory neurones	65
		8.2.4.1. Increased sympathetic innervation and excitation of the DRG upon PAF injury	65
		8.2.4.2. Mechanisms underlying sympathetic excitation of damaged nerves: a role for α -ARs on	
		PAFs or sympathetic terminals	66
		8.2.4.3. Other sympathetic mechanisms potentially modulating the activity of PAFs	67
		8.2.4.4. Sympathectomy-induced pain	67
		8.2.6. The significance of a reduction in DEG access to NGE and other traphic factors in modifying the	08
		nhenotyne of damaged PAEs	70
		8.2.6.1. Phenotypic consequences of a loss of DRG access to NGF for C fibres	70
		8.2.6.2. A potential role of other trophic factors: focus on LIF	70
		8.2.6.3. Gene transcription factors: immediate early genes	71
		8.2.6.4. Factors involved in phenotype changes in large calibre $A\beta$ fibres	71
		8.2.6.5. Functional considerations	71
0	Nac	8.2.7. Phenotype changes in damaged PAFs: neuromodulators other than neuropeptides	72
9.	9.1	Evidence for central mechanisms of sensitization	72
	<i>J</i> .1.	9.1.1. The significance of central events	72
		9.1.2. Mechanical allodynia: evidence for mediation by A β fibres	73
		9.1.3. Secondary hyperalgesia: peripheral maintenance of central sensitization	73
		9.1.4. Referred pain	75
		9.1.4.1. Characteristics of referred pain	75
	• •	9.1.4.2. Potential mechanisms underlying referred pain	75
	9.2.	C hores and allodynic states	// 77
		9.2.1 Punctate: hyperalogical hyperalogical and the states	78
		7.2.2 An AB _C C fibre link in the DH underlying allodynia?	78
		9.2.4. Cold allodynia	79
10	. No	priceptive and neuropathic pain: processes integrated in the DH	79
	10.1	. Multiple mechanisms integrated in the DH	79
	10.2	2. Sensitization and increased excitability of DH neurones	80
		10.2.1. LTP and neuronal sensitization	80
	10.2	10.2.2. Induction of sensitization in DH neurones: characteristics of sensitized neurones	80 80
	10.3	10.3.1. Molecular mechanisms underlying neuronal sensitization: the key role of NMDA recentors	82 82
		10.3.2. The roles of AMPA, $NK_{1/2}$, mGlu and CGRP recentors in neuronal sensitization	83
		10.3.2.1. AMPA receptors	83
		10.3.2.2. $NK_{1/2}$ receptors	83
		10.3.2.3. mGlu receptors	84
		10.3.2.4. CGRP receptors	85

Mark J	. Millan
--------	----------

CONTENTS (continued)	
10.4. Cellular processes underlying sensitization of DH neurones: focus on Ca^{2+} and PKs	85
10.4.1 Increases in Ca^{2+} -influx and in $[Ca^{2+}]$.	86
10.4.2 Euclidean effects of increases in $[Ca^{2+1}]$	86
10.4.2.1 Enhancement of the activity of PKA PKC and other PKs	86
10.4.2.2. Induction of NO synthesis	86
10.4.2.3 Generation of PGs	87
10.42.4. Counter regulatory processors induction of phosphataces	87
10.4.2.4. Contributing of DC supervision and the supervision in DL supervision	0/
10.4.2.5. Modulation of IEG expression and gene transcription in DH neurones	8/
10.4.3. The roles of specific PKs	88
10.4.3.1. Actions of PKC	88
10.4.3.2. Actions of cAMP and PKA	89
10.4.3.3. Actions of cGMP and PKG	90
10.4.3.4. Key roles of $[Ca^2 +]_i$	90
10.5. A β fibre-mediated sensitization of DH neurones in inflammatory states	90
10.6. Reduction in the functional activity of ININs	91
10.6.1. Inhibition of ININ activity mimics and enhances sensitization in the DH	91
10.6.2. Mechanisms underlying a reduction of ININ activity	92
10.6.2.1. Evidence for a reduction in GABAergic ININ activity upon PAF injury	92
10.6.2.2. Temporal evolution of changes in ININ activity:	93
10.6.3. Alterations in ININ activity upon PAF injury: a summary	94
10.7. Reduction of stimulatory A β fibre input to ININs upon PAF injury	94
10.8. Modulatory role of descending pathways	95
10.8.1. A decrease in descending inhibition and an increase in descending facilitation	95
10.8.2. Adrenergic mechanisms	95
10.8.3. Serotonergic mechanisms	95
10.8.4. Dopaminergic mechanisms	96
10.8.5. Role of cerebral mechanisms in triggering descending facilitation	96
10.8.6 Physiological Activation of mechanisms of descending facilitation	96
11 Neuronathic pain: reorganization of central and peripheral projections of PAEs following injury	97
11.1 Peripheral-central interface in the DH	97
11.1. Control degeneration and rearrowth of injured PAEs: abnormal patterns of AB fibre connectivity)/
11.2. Contain degeneration and regionario in the control and participation of injured PAF	08
11.5. The fole of GAP 45 and other factors in the central and periphetal regulation of injured rAPs	90
11.5.1. The induction of GAP 45 in injured C holes, short-term and long-term fore in neuronal	0.0
sensuization	98
11.3.2. The role of GAP 43 and other intrinsic factors in the central regeneration of damaged PAFs	99
11.3.5. Mechanisms underlying $A\beta$ fibre migration into foreign laminae	99
11.3.4. Peripheral regeneration of damaged PAFs	100
11.3.4.1. The importance of Schwann cells	100
11.3.4.2. NGF and processes of regeneration in GAP 43-positive C fibres	101
11.3.4.3. Roles of GDNF and other trophic factors	101
11.3.4.4. Cellular adhesion and extracellular matrix molecules	101
11.3.4.5. NO and neuropeptides	102
11.3.4.6. The interactive nature of regenerative mechanisms	102
11.4. Reorganization of PAF input to the DH following injury: overview of functional consequences for	
nociceptive transmission	102
11.5. Peripheral and central sprouting of intact PAFs	103
11.5.1. Sprouting of intact collaterals into denervated, peripheral tissue: a role for NGF	103
11.5.2. Collateral sprouting of intact PAFs to the DH	103
11.6. Surgical and chemical destruction of PAF input into the DH	104
11.6.1. Surgical interruption of PAF input to the DH via lesions proximal to the DRG	104
11.6.2. Chemical destruction of small calibre C fibres by capsaicin	105
12. Nociceptive and neuropathic pain: processes occurring in supraspinal centres	105
12.1 Sensitization of neurones in the S L and thalamus	105
12.2 Adaptive changes in higher centres	105
12.2.1 Distinctive changes in supraspinal structures	105
12.2.2 Adaptive changes at the cortical level (S I)	105
12.2.3. Adaptive changes at the thalamic level	106
12.2.4. Other structures	100
12.3. Poles of neurotransmitters in adaptive changes in higher centres	107
12.3. Constant of and EAs	107
12.2.2. CADA allu LAAS	107
12.3.2. CADACIENT methanisms	107
12.3.4. Other potential mediators	107
12. Dis due to CNS demogr control pair	108
12.1. The particular is a function of a set of the set	108
13.1. The pathological origins of central pain	108
13.2. A model of pain due to spinal damage: loss of GABAergic inhibition in the DH	108
13.3. Reduced GABAergic transmission in the thalamus and cortex: burst firing and reverbatory circuits	109
13.4. Disinhibition of thalamocortical pathways by thalamic damage: cold allodynia	110
13.5. Mechanisms involved in the response of the thalamus and other CNS tissues to injury	110
14. Central actions of neurotrophins and their relevance to nociceptive processing	111
14.1. A role for neurotrophins in the modulation of nociception in the CNS?	111
14.2. Neuronal sources of neurotrophins: physiological modulation	112

CONTENTS (continued)	
14.3. Induction and release of neuronal pools of neurotrophins: NMDA receptor-mediated stimulation	112
14.4. Actions of neurotrophins: modulation of synaptic transmission	112
14.4.1. Anterograde transport of BDNF	112
14.4.2. Modulation of glutaminergic transmission: presynaptic and postsynaptic actions	112
14.4.3. Modulation of GABAergic transmission	113
14.4.4. Overview of short- and long-term actions of BDNF	113
14.4.5. Summary	114
14.5. Sites at which neurotrophins may modulate nociceptive processing	114
14.5.1. The S I of the cortex	114
14.5.2. The thalamus and other cerebral structures	115
14.5.3. The DH of the spinal cord	115
14.6. Summary	116
15. General discussion and conclusions: a summary of key issues	116
15.1. The complexity of mechanisms involved in the induction of pain	116
15.2. Mechanisms of antinociception, nociception and the development of novel analgesics	118
15.3. Concluding comments	119
Acknowledgements	119
References	119

The Induction of Pain: an Integrative Review

5

ABBREVIATIONS

А	Adenosine receptor	ININ	Inhibitory interneurone
$[Ca^{2+}]_{i}$	Intracellular concentration of Ca ²⁺	LIF	Leukaemia inhibitory factor
5-HT	Serotonin	LTD	Long-term depression
AC	Adenyl cyclase	LTP	Long-term potentiation
cAMP	Cyclic adenosine monophosphate	MDVC	Mediodorsal ventrocaudal thalamus
AMPA	DL-a-NH2-2,3-dihydro-5-methyl-3-oxo-4-	MGLU	Metabotropic
	isoxazolepropanoic acid	NAD	Noradrenaline
AP	Action potential	NGF	Nerve growth factor
AR	Adrenoceptor	NI	Neurogenic inflammation
ATP	Adenosine triphosphate	NK	Neurokinin
В	Bradykinin receptor	NMDA	<i>N</i> -methyl- <i>D</i> -aspartate
BDNF	Brain derived neurotrophic factor	NO	Nitric oxide
BK	Bradykinin	NON-N	Non-nociceptive
CAM	Cell adhesion molecule	NPY	Neuropeptide Y
CBV	Cerebral blood vessel	NS	Nociceptive-specific
CCK	Cholecystokinin	Р	Purinoceptor
CGMP	Cyclic guanosine monophosphate	PACAP	Pituitary adenyl cyclase activating protein
CGRP	Calcitonin-gene related peptide	PAD	Primary afferent depolarization
CNS	Central nervous system	PAF	Peripheral afferent fibre
CREB	$Ca^{2+}/cAMP$ response element binding protein	PAG	Periaqueductal grey
COX	Cyclooxygenase	PBN	Parabrachial nucleus
CRF	Corticotropin releasing factor	PG	Prostaglandin
CSF	Cerebrospinal fluid	РК	Protein kinase
DA	Dopamine	PLA	Phospholipase A
DCN	Dorsal column nuclei	PLC	Phospholipase C
DAG	Diaminoglycine	PN	Projection neurone
DH	Dorsal horn	PSDC	Postsynaptic dorsal column
DLF	Dorsolateral funiculus	RF	Receptor field
DRG	Dorsal root ganglia	S I/II	Somatosensory area I/II
DRP	Dorsal root reflex	SP	Substance P
DRP	Dorsal root potential	STT	Spinothalamic tract
DYN	Dynorphin	TNF	Tissue necrosis factor
EAA	Excitatory amino acid	TRK	Tyrosine kinase
EPSP	Excitatory postsynaptic potential	TTX	Tetrodotoxin
EP	PGE ₂ receptor	VDCC	Voltage-dependent calcium channel
EXIN	Excitatory interneurone	VH	Ventral horn
GABA	Gamma-amino-butyric acid	VIP	Vasoactive intestinal peptide
GAL	Galanin	VLF	Ventrolateral funiculus
GAP	Growth associated protein	VMPO	Ventromedial posterior
GDNF	Glial cell-line derived neurotrophic factor	VPI	Ventroposterior inferior
GLU	Glutamate	VPL	Ventroposterior medial
IEG	Immediate early gene	VPM	Ventroposterior lateral
IL	Interleukin	WDR	Wide-dynamic range
IN	Interneurone	Y	Neuropeptide Y receptor

1. INTRODUCTION: AIMS AND SCOPE OF REVIEW

The exposure of the skin and other organs to damaging, or potentially-damaging, 'noxious' stimuli elicits the intensely unpleasant sensation of pain, an experience which is ultimately integrated in corticolimbic centres of the brain. Intervening between this initial exposure to a noxious stimulus and the conscious appreciation of pain is an extraordinarily complex and interactive series of mechanisms whereby the noxious stimulus is encoded as a nociceptive message and is progressively transmitted to, and processed in, higher nervous centres. Such modulatory events occur in the periphery, in the dorsal horn (DH) of the spinal cord ('primary processing'), supraspinal relay centres such as the thalamus ('secondary processing'), as well as in corticolimbic structures themselves. Further, under certain pathological conditions, such as peripheral nerve injury or damage to the central nervous system (CNS), pain can be engendered downstream, and independently, of sensory nerve endings.

The purpose of the present article is to provide an integrative account of:

- 1. neuronal pathways involved in the receipt and transmission of nociceptive information from the periphery to higher CNS centres; and
- 2. the cellular and molecular bases of mechanisms underlying the induction of pain.

Special emphasis is afforded to processes underlying the prolonged pain provoked by peripheral nerve injury, the unravelling of which has progressed considerably in recent years.

Endogenous mechanisms for the attenuation of nociception, which are engaged at all levels of transfer and processing, have been extensively reviewed elsewhere and do not comprise the focus of the present article (Basbaum and Fields, 1984; Besson and Chaouch, 1987; Fields and Basbaum, 1994; Hammond, 1997; Hayashi and Maze, 1993; Le Bars, 1988; Millan, 1986, 1997; Ossipov et al., 1997; Picard et al., 1997; Stamford, 1995).

Since the present review is directed not only towards those active in pain research, but also-and primarily-towards those less familiar with this field, no expert knowledge is assumed and several explanatory diagrams have been incorporated. In addition, Table 1 outlines several technical terms frequently encountered in the experimental and clinical literature. The reference list (completed early in 1998), though comprehensive, has no pretensions to exhaustivity. It is compiled from a selection of key papers, reviews and, in particular, recent articles as yet to be covered by reviews. The latter assume particular importance in view of the astonishing rapidity with which this field is developing. Details on many of the specific subjects discussed herein may be found in several books and authoritative review articles, of which the following are particularly useful sources of information (Belemonte and Cervero, 1996; Besson and Chaouch, 1987; Besson et al., 1994; Carli and Zimmerman, 1996; Carstens, 1995; Cervero, 1994, 1995b; Coderre et al., 1993; Coggeshall and Carlton, 1997; Coyle, 1996; Dickenson and Besson, 1997; Dray and Urban, 1996; Dubner and Bennett, 1983; Gebhart, 1995; Handwerker and Kobal, 1993; Hökfelt et al., 1994; Koltzenburg, 1995; Melzack and Wall, 1965; Mense, 1993; Pairet and Engelhardt, 1996; Rang et al., 1991; Ruda et al., 1986; Schaible and Grubb, 1993; Tracey and Walker, 1995; Treede et al., 1992b; Wall and Melzack, 1994; Willis, 1985; Willis and Coggeshall, 1991; Wilson and Kitchener, 1996; Woolf, 1994; Woolf and Doubell, 1994).

The article is organized as follows. Section 2 considers the origins and pathophysiological significance of pain from an evolutionary perspective. Sections 3–6 outline pathways for the peripheral receipt, central transfer and supraspinal passage of nociceptive information: the modulatory role of pathways descending from the brain to the DH is also outlined. These sections provide a neuronal

Table 1. Explanation of technical terms [see also Merskey (1986); Merskey and Bogduk (1994); Millan (1993)]

- 1. Pain: an unpleasant sensory and emotional experience associated with actual or potential tissue damage
- 2. Nociceptive pain: pain due to (excessive) stimulation of nociceptors localized in the skin, viscera and other organs
- 3. Neurogenic pain: pain reflecting damage to neuronal tissue in the periphery or CNS ('central' pain). Neuralgia refers to the pain localized to a region innervated by a specific nerve, or group of nerves

- 6. Noxious stimulus: a damaging, or potentially damaging, stimulus
- 7. Nociceptor: a peripherally-localized receptor sensitive and responding to noxious stimuli
- 8. *Analgesia (antinociception)*: a reduction in 'spontaneous' pain, or the pain elicited by a noxious stimulus. Operationally-defined in terms of a reduced response to, or an increase in the threshold to respond to, a noxious
- Hyperalgesia: an increase in the pain elicited by a noxious stimulus. Operationally-defined in terms of an increased
- response to, or a decrease in the threshold to respond to, a noxious stimulus. 'Spontaneous' pain occurs in the apparent absence of extraneous stimulation
- 10. *Primarily and secondary hyperalgesia*: pain in the region of tissue damage (primary), or in the region surrounding this area (secondary). Mediated by different mechanisms
- 11. *Allodynia*: pain evoked by a normally innocuous stimulus. Operationally-defined in terms of a response to an innocuous stimulus which is normally only evoked by a noxious stimulus. (Some authorities consider allodynia an 'extension' of hyperalgesia at the left of the stimulus-response function.)
- 12. Paresthesia: an abnormal sensation—if unpleasant, referred to as dysaesthesia—which is either spontaneous or evoked

^{4.} *Neuropathic pain*: pain due to a dysfunction of, or damage to, a nerve or group of nerves. Primarily peripheral nerves, although pain due to CNS damage ('central' pain) may share these characteristics

^{5.} *Psychogenic pain*: pain not due to an identifiable, somatic origin and which *may* reflect psychological factors

framework for an understanding of peripheral and central mechanisms underlying 'nociceptive' and 'neuropathic' pain (Table 1). Thus, Sections 7 and 8 deal with peripheral processes triggered by inflammatory tissue damage/nociceptor stimulation and sensory nerve injury, respectively. Section 9 discusses the roles of various classes of sensory nerve in mediating and maintaining painful states, and considers evidence that central mechanisms are involved in their induction. Section 10 discusses spinal mechanisms of sensitization elicited both by tissue inflammation and by peripheral nerve injury. Section 11 outlines processes involved in the regeneration of injured peripheral nerves and in the reorganization of their input to the DH, together with the consequences for nociception. Section 12 summarizes adaptive processes occurring in supraspinal structures. Section 13 is specifically concerned with the 'central' pain provoked by damage to the CNS itself, including several processes resembling these initiated by peripheral nerve injury. In Section 14, a hypothesis is developed according to which nerve growth factors (neurotrophins) may play a role in modulating central, synaptic events underlying prolonged, painful states. Finally, Section 15 concludes with a summary of several major themes and a brief consideration of how an improved understanding of mechanisms underlying the induction of pain may contribute to the development of novel analgesic agents.

The present account concentrates on our knowledge of *cutaneous* sensory input inasmuch as the skin has been the most intensely-studied of organs as concerns pain elicited both by nociceptor stimulation and by peripheral nerve damage. In general, the mechanisms discussed are pertinent to other tissue types, though several major differences to other organs, in particular the viscera, are mentioned. Specific aspects concerning mechanisms of nociception in non-cutaneous tissue have been reviewed in detail elsewhere (Cervero, 1994, 1995b; Gebhart, 1995; Guilbaud, 1988; Meyer *et al.*, 1994; McMahon *et al.*, 1994; Mense, 1993; Meller and Gebhart, 1992; Ness and Gebhart, 1990; Schaible and Grubb, 1993; Willis and Coggeshall, 1991).

Throughout the text, it has been attempted to keep duplication of material to a minimum. To this end, in view of the interdependent nature of many issues discussed, extensive reference is made to other Sections. Nevertheless, in order to avoid the need for incessant consultation of other parts of the review, individual sections are intended to be selfsufficient in information.

2. ACUTE AND CHRONIC PAIN: ORIGINS AND PATHOPHYSIOLOGICAL SIGNIFICANCE FROM AN EVOLUTIONARY PERSPECTIVE

2.1. The Adaptive Significance of Acute Vs Chronic Pain

The persistent pain which accompanies—and can outlast—inflammatory tissue damage and/or nerve injury is generally characterized by its spontaneous

nature (not elicited by extrinsic stimuli) and by the presence of hyperalgesia (an increase in the pain elicited by a noxious stimulus) and/or allodynia (pain elicited by normally innocuous stimuli) (Table 1). Prolonged, chronic pain is regarded as fulfilling no physiological purpose (Table 2). This lack of adaptive value may be contrasted to the protective function of the motor withdrawal (integrated via a reflex arc in the ventral horn) elicited by phasic, acute exposure to noxious stimuli, and the recuperative, protective behaviour associated with subchronic, painful states (Table 2). Thus, it is reasonable to speak of 'pathological' vs 'physiological' for chronic vs acute/subchronic pain, respectively. The alerting function of acute pain reflects the phasic activation of sensors (nociceptors) by potentially dangerous stimuli exceeding the physiological range. This warning purpose of pain is most evident as concerns the skin, which is exposed to external dangers. Acute, cutaneous pain evokes, thus, motor withdrawal and/or a 'flight' reaction, protective responses intended to discontinue exposure to the noxious stimulus and, thereby, to terminate the pain. On the other hand, under certain conditions, actual pain, or the threat of pain, may elicit a generalized behavioural arousal, endocrine responses (such as corticosterone secretion) and sympathetic activation (leading to elevations in blood pressure and heart rate, etc) which, together with a transient antinociception, improve the performance of behavioural repertoires permitting successful disengagement from aggressive encounters or other situations potentially leading to serious tissue damage (Millan, 1986; Wiertelak et al., 1994c; Traub, 1997).

Under most conditions, an adaptive association between dangerous, tissue-damaging (internal) stimuli and pain is likewise apparent for skeletal muscle and, probably, the joints (Mense, 1993; Schaible and Grubb, 1993) (Section 3.1.2). This relationship is, however, less evident for the viscera (Section 3.1.2). For example, pain is seldom, if ever, experienced from the liver or lung, even when they are severely diseased or damaged, possibly since a network of sensory nerves for the transmission of nociceptive information to the CNS is absent in these organs (Cervero, 1994). Further, whereas global distension of the hollow viscera (gut and urinary tract) elicits pain, their *local* perforation may not necessarily be painful, despite its potentially more dramatic consequences (Cervero, 1994, 1995b; Mailiani et al., 1984; McMahon, 1994; Meller and Gebhart, 1992). Although ischaemic tissue damage, inflammation and distension clearly can elicit visceral pain and hypersensitivity (for example, cystitis), one striking illustration of a dissociation between damage to internal organs and pain (a 'failure to warn') is provided by the phenomenon of 'silent ischaemia'. That is, painless heart attacks which, despite their apparent innocuousness, are predictive of subsequent heart failure. Contrariwise, angina-like thoracic pain in the apparent *absence* of a somatic cause may also be observed (Arnim and Maseri, 1987; Casey, 1996; Casey et al., 1996b; Cervero, 1994; Meller and Gebhart, 1992). These differences to cutaneous pain possibly reflect the fact that visceral and deep pain is often 'inescapable': protective behaviour can be

1	Ì		Mark J. M	illan	ī
	Examples	Contact with hot surface	Inflamed wound	Arthritis PAF or CNS injury Metastatic disease	
	Adaptive response	Withdrawal Escape	Quiescence Avoidance of contact with injured tissue	Psychological and cognitive	
of pain	Adaptive value	High Preventative	Protective Recovery	None Maladaptive	
n of some major features	Source of pain	Transient nociceptor activation	See Tables 5 and 6	See Tables 5 and 6	
siological classificatio	class	Nociceptive	Principally nociceptive. But also neuropathic.	Neuropathic Nociceptive	
. A general, pathophys	Major characteristics	'Proportional to cause'	1° and 2° Hyperalgesia Allodynia Spontaneous pain	1° and 2° Hyperalgesia Allodynia Spontaneous pain Parasthesias/ dysthesias Pronounced affective component	
Table 2.	Temporal features relationship to cause	Instantaneous Simultaneous	Resolves upon recovery	Persistant Long-term disease May exceed resolution of tissue damage	respectively.
	Duration	Seconds	Hours to days	Months to years	rimary and secondary,
	Type	Phasic Acute.	Prolonged Subchronic	Chronic Clinical	1° and 2°, P

featu
major
some
<u>6</u>
classification
ā
. <u></u> .
9
ysi.
ų.
atho
d,
general
≺
le 2.

less easily engaged in response to internal as compared to external threats. Evolutionary pressure will, thus, have been less intense for the development of an alerting system for internal pain. Notwithstanding these observations, deep visceraland somatic-pain, when severe, tends to evoke a reaction of quiescence, disinterest, hyporeactivity, hypothermia and bradycardia. Such temporary responses may be adaptive inasmuch as they reduce discomfort, accelerate recovery and limit the risk of deleterious encounters. Further, visceral pain, in particular from the peritoneal region, via a series of reflex pathways, can lead to muscular contractions or spasms which rigidify the abdominal wall. This may protect ('guard') the viscera from further trauma, thereby encouraging recuperation. Further, sensations of overdistension and gastric irritation from the gut may also elicit appropriate behavioural responses.

In addition to their protective-sensory, afferent (orthodromic) role, the antidromic stimulation of fine calibre, nocisponsive fibres is associated with 'efferent' functions in the periphery. That is, the peripheral release of neuropeptides and other substances elicits a constellation of local responses involving actions at neighbouring nocisponsive fibres, the vasculature and immunecompetent cells etc. This phenomenon, which is often termed 'neurogenic inflammation' (NI) (Holzer, 1988; Kilo et al., 1994, 1997; Maggi, 1991a,b) (Section 7.9), may fulfil a beneficial role in, for example, combatting tissue infection and accelerating would healing. Further, the release of neuropeptides and other molecules from damaged primary afferent fibres (PAFs) may facilitate mechanisms allowing for their regeneration and functional recovery (Hökfelt et al., 1994; White and Maasfield, 1996) (Sections 8.2.5 and 11.3.4.5). However, the antidromic activation of sensory terminals, via a feedback mechanism, further increases their activity and potentiates nociception (Section 7.4.9). Further, where unrestrained and/or prolonged, the efferent excitation of nocisponsive fibres may have other deleterious consequences: for example, an enhancement of those neurogenic and vascular events in dural blood vessels which underlie migraine headaches (Section 7.9), and an exacerbation of pathological processes of arthritic inflammation and joint degeneration (Schaible and Grubb, 1993; Murphy, 1993).

2.2. Lack of Endogenous Mechanisms for the Effective Control of Chronic Pain

As mentioned above, temporary inactivity and protective behaviour in response to subchronic pain may be adaptive. However, *persistent* pain may lead to a *long-term* state of withdrawal, anhedonia and environmental indifference. This response resembles depressive-like states ('defeat', 'helplessness' and/or 'despair') triggered by inescapable stress, and cannot, as such, be considered adaptive (Bloom and Kupfer, 1995; Craig, 1994; Millan and Gobert, 1997). Indeed, although the evolution of acute (and subchronic) pain may be attributed to its protective (adaptive) function, the question arises as to why pathological, chronic pain has 'evolved'? In fact, the question should probably be inversed. Why have specific mechanisms *not* evolved allowing for the counteracting of chronic, painful states, inasmuch as endogenous processes involved in the moderation of shorter-term pain are manifestly inadequate for its suppression?

This may reflect an absence of sufficient selective pressure for their evolution under the circumstances in which chronic pain is generally encountered. Thus, afflicted individuals are frequently old and/or infirm. For example, the pain associated with degenerative tissue diseases (such as rheumatoid arthritis), metabolic disorders (such as late-stage, insulindependent diabetes), severe injury (such as limb fracture) or terminal disease (such as cancer). Correspondingly, the probability that affected individuals will survive is meagre, and they probably do not comprise a part of reproductive populations. Genes favouring a moderation of chronic pain under these conditions are unlikely to possess major adaptive value and likely show limited 'penetration' (transmission to succeeding generations). Indeed, even the alleviation of pain would not necessarily be adaptive, since the pathological state would persist and perhaps even be aggravated.

In addition, several processes contributing to chronic pain may be secondary consequences of the adaptive, efferent functions of sensory fibres involved in palliating the effects of tissue damage and/or nerve injury (Sections 2.1 and 8.2.5). Thus, stimulation of, or damage to, PAFs is associated with the induction and *peripheral* release of several trophic factors, in particular neuropeptides. These exert a diversity of actions on local tissues, the vasculature and PAFs themselves which collectively encourage wound healing and *facilitate* the recovery of injured nerves (Section 11.3.4.5). However, upon their peripheral and/or central release, certain neuropeptides potentiate the transmission of nociceptive information (Section 8.2.5). Similarly, although immunecompetent and glial cells exert beneficial actions following injury, they also secrete cytokines and other mediators which aggravate nociception associated with inflammation or PAF injury (Sections 7.7 and 8.2.1). Under some conditionsprobably of limited duration-the adaptive value of these restorative actions may offset the negative consequences of the pain thereby engendered. However, ultimately, chronic pain becomes deleterious to the organism.

3. RECEIPT, ENCODING AND TRANSMISSION TO THE DH OF PRIMARY AFFERENT, NOCICEPTIVE INFORMATION

3.1. Sensory Properties of PAFs: Detection of Noxious Stimuli

3.1.1. Cutaneous C, Aδ and Aβ Fibres Encoding Noxious and Innocuous Information

The nature and organization of primary afferent input to the DH has been comprehensively and expertly reviewed elsewhere (Belemonte and Cervero, 1996; Besson and Chaouch, 1987; Cervero, 1994; Laem *et al.*, 1993; Lawson, 1992; Mense,

Mark J. Millan



Fig. 1. Organization of cutaneous, primary afferent input to the dorsal horn of the spinal cord. Abbreviations are as indicated in the general list. This schema does *not* quantitatively differentiate between various laminae as concerns PAF input. For example, the density of $A\delta$ fibres innervating lamina I is considerably greater than that projecting to lamina II, a principal target of C fibre input. Unmyelinated fibres from muscle, viscera and joints appear to preferentially innervate laminae I and IV/V/IV rather than II. Cell types (NS, WDR or NON-N) are also qualitatively—rather than quantitatively—indicated in terms of the major class encountered in various laminae. Laminae VI, which functionally complements lamina V, is clearly identifiable only at the level of the cervical and thoracic cord. Lamina X corresponds to the grey matter surrounding the central canal (Sections 3 and 4).

1993; Meyer *et al.*, 1994; Raja *et al.*, 1988; Schaible and Grubb, 1993; Treede and Magerl, 1995; Treede *et al.*, 1992b; Willis and Coggeshall, 1991). Cutaneous PAFs can be classified into essentially three types on the basis of their diameter, structure and conduction velocity (Fig. 1, Table 3).

- 1. C: thin $(0.4-1.2 \ \mu \text{m}$ in diameter), unmyelinated and slowly-conducting $(0.5-2.0 \ \text{m sec}^{-1})$.
- 2. A δ : medium (2–6 μ M), myelinated and of intermediate velocity (12–30 m sec⁻¹).
- 3. A β : large (>10 μ M), myelinated and fast (30–100 m sec⁻¹).

In skin, they are typically present in proportions of ca 70, 10 and 20%, respectively, though their ratios may vary. Each of these classes of PAF encodes sensory information, but they are differentially sensitive to noxious and innocuous stimuli. Thus, although *all* three classes of cutaneous fibres can transmit *non*-nociceptive information, under normal circumstances, only *C* and $A\delta$, but *not* $A\beta$, fibres transmit *nociceptive* information (Table 3) (Section 9.1.2).

						Sensation	mediated
Fibre class	Threshold for activation	Principal transmitters*	Receptors engaged*	DH Laminae innervated†	Neurone types targeted	Physiological	Pathological
С	High	SP/NKA CGRP EAA	NK _{1/2} CGRP _{1/2} NMDA/ AMPA mGlu	I/II ₀ IV/V, X	NS WDR	Noxious (pain)	Highly noxious (hyperalgesia) Cold allodynia (pain)
Αβ	Low	EAAs	AMPA	III–VI	NON-N WDR	Innocuous (no pain)	Mechanical allodynia (pain)

Table 3. Comparative properties of primary afferent C and $A\beta$ fibres

*Partial list. Further, under conditions of nerve injury, PAF phenotype may change, for example, VIP is induced in C fibres and substance P is induced in A β fibres (Section 8.2).

†Under conditions of nerve injury, the organization of PAF projections in the DH may change, for example, $A\beta$ fibres sprout to more superficial (II₀) laminae wherein they may contact NS neurones (Section 11).

Generally-speaking, upon exposure of the skin to a noxious stimulus, myelinated $A\delta$ fibres elicit a rapid, first phase of pain, which is 'sharp' in nature, whereas unmyelinated C fibres evoke a second wave of 'dull' pain (Belemonte and Cervero, 1996; Handwerker and Kobal, 1993; Meyer et al., 1994; Torebjörk and Hallin, 1973; Treede et al., 1992b). A further difference, recently quantified by Yeomans and Proudfit (1996) and Zachariou et al. (1997), is that, for cutaneous thermosponsive units, low and high noxious skin heating rates preferentially engage nocisponsive C and A δ fibres, respectively. One further distinction is that the C fibres responsive to intense and painful, cold stimuli are located along vein walls, whereas thermosensory A δ fibres responsive to cooling are localized cutaneously (Chen et al., 1996; Craig et al., 1996; Klement and Arndt, 1992; LaMotte and Thalhammer, 1982).

Multiple classes of both C and $A\delta$ fibre exist. However, their characterization and description is complicated by several factors including terminological inconsistencies, species differences, contrasting properties of nocisponsive fibres in various tissues for example, hairy vs non-hairy ('glabrous') skin and the methods employed for their detection (Belemonte and Cervero, 1996; Handwerker and Kobal, 1993; Koltzenburg *et al.*, 1997; Meyer *et al.*, 1994; Treede *et al.*, 1990a, 1992b; Willis and Coggeshall, 1991). The following paragraphs provide a simplified overview.

As concerns C fibres, chemospecific nociceptors, thermoreceptors responsive to warming or cooling, as well as low threshold mechanoceptors responsive to pressure, have all been described (Meyer et al., 1994). The role of cold-sensitive C fibres in mediating cold allodynia is discussed in Sections 9.2.4 and 13.4. Of particular pertinence in the present context are those high threshold receptors localized on C fibres which encode nociceptive information pertaining to noxious thermal and/or mechanical stimuli. Of these, there is an abundance which respond to both mechanical and heat stimuli and the term 'polymodal' or 'CMH' is generally reserved for the population which also responds to irritant, chemical stimuli. The crucial role of polymodal, nocisponsive C fibres in triggering the sensitization of spinal neurones, a process underlying chronic painful states, is described in Sections 10.2, 10.3 and 10.4. A specific class of C fibre nociceptor which exclusively (or primarily) responds to noxious, heat stimuli (and which is coupled to a cation-permeable ion channel) has been detected in primates (including man), but is rare in rodents. Together with polymodal C fibres, this C fibre class is likely involved in mediating the 'axonal flare response' (a component of NI) following tissue damage (Belemonte and Cervero, 1996; Lynn et al., 1996; Meyer et al., 1994; Reichling et al., 1997; Schmidt et al., 1995) (Section 7.9). Collision studies have revealed a minor degree of bidirectional coupling between polymodal C fibres terminals, a process which may be involved in expansion of the flare response. Recently, a microneurography approach in human skin permitted the identification of a very thin and slowly-conducting, histamine-responsive class of mechano-insensitive C fibres which showed extensive terminal arborization: a role of these PAFs in the mediation of itch was suggested (Meyer *et al.*, 1994; Schmelz *et al.*, 1997b).

As regards $A\delta$ fibres, one class is activated by high intensity, mechanical stimuli in the noxious range (for example, pinch). These so-called 'Type I', high threshold, rapidly-conducting mechanoceptors, are weakly responsive to high intensity heat, cold (and chemical) stimuli (Handwerker and Kobal, 1993; Meyer et al., 1994; Simone and Kajander, 1997; Treede and Magerl, 1995). However, repetitive thermal stimulation may sensitize them to further stimulation and, following tissue damage, they become heat-responsive and show sustained responses to thermal stimuli of long duration and slow (sec) latency. This increase in responsiveness has been contrasted to the decrease in responsiveness displayed by polymodal C fibres in glabrous skin upon repetitive heat stimulation or injury (Meyer et al., 1994). A further, less common and less rapidly-conducting class of A δ fibre ('Type II') has principally been examined in the hairy skin of primates. Type II A δ fibres display a lower threshold to noxious heat stimuli than their Type I counterparts, and respond thereto more rapidly (Beydoun et al., 1996; Treede et al., 1990a,b). A δ fibres more responsive to cooling than mechanical stimuli, and capable of encoding stimulus intensity, also exist (Craig et al., 1996; Simone and Kajander, 1997).

For cutaneous populations, receptive fields (RFs) may be quite small (2 mm diameter) and homogenous for nocisponsive C fibres: for example, as concerns mechanical, noxious stimuli in the rat (Belemonte and Cervero, 1996; Lynn and Carpenter, 1982). Others may be large (≈ 10 mm) and heterogenous (containing islands of responsive areas): for example, for mechanical and thermal stimulation of primate and human, hairy skin. Similarly, nocisponsive, high threshold, mechanosensitive A δ fibres display complex RFs with multiple sites of sensitivity (Belemonte and Cervero, 1996; Treede *et al.*, 1992b).

In the *absence* of tissue or nerve injury, cutaneous $A\beta$ fibres are responsive *only* to touch, vibration, pressure and other modes of *non*-noxious, low intensity mechanical stimuli—although it has been suggested that a sub-population of rapidly-conducting A fibres, of ambiguous $A\delta$ - or $A\beta$ -like characteristics, may contribute to nociceptive transmission from muscle (Mense, 1993; Woolf *et al.*, 1994; Willis and Coggeshall, 1991; Wilson *et al.*, 1996b).

Microneurographic recordings of the activity of cutaneous sensory nerves in man, in parallel with psychophysical assessments of pain intensity, suggest that nociceptors can encode the intensity of noxious thermal (and mechanical) stimuli (Bromm *et al.*, 1984; Brown and Culberson, 1981; Dubner *et al.*, 1989; Gybels *et al.*, 1979; Handwerker and Kobal, 1993; Meyer *et al.*, 1994; Treede *et al.*, 1992b). They also encode information concerning the spatial localization of noxious, cutaneous stimuli and, for the performance of this function, overlapping territories of individual nociceptive C fibres in skin may be a prerequisite (Schmidt *et al.*, 1997). Generally, a single discharge of an individual nocisponsive fibre is *not* perceived as noxious and many nocisponsive

units need to be recruited over a period of time for 'pain' to be experienced. Further, actual pain thresholds are higher in man than the thresholds for activation of individual nociceptors. These observations suggest the existence of central mechanisms for both spatial and temporal summation for pain signalling (Vierck et al., 1997) (Sections 10 and 12). Nevertheless, the 'instantaneous' function of nociceptors in signalling exposure to acute, noxious stimuli corresponds to the 'alerting' role of pain (Section 2) and likely occurs with little intervention from modulatory (inhibitory) peripheral or central mechanisms (Handwerker and Kobal, 1993). As mentioned above, in the absence of tissue damage, repetitive noxious stimulation (like other modes of sensory input) may be associated with a decreased response of some polymodal C-fibres. However, under conditions of tissue damage, this phenomenon is greatly outweighed by both peripheral and central processes of sensitization, such that the sensation of pain will be intensified and prolonged. These mechanisms of sensitization comprise a major part of this review and are described in detail in Sections 7, 10 and 12.

3.1.2. Nocisponsive Fibres in Other Organs

There is an extensive literature, authoritatively reviewed elsewhere, which exemplifies several similarities, and certain differences, between mechanisms involved in the receipt and processing of nociceptive information from non-cutaneous tissues as compared to the skin (Cervero, 1994, 1995a; Gebhart, 1995; Marchettini *et al.*, 1996; McMahon, 1994; Mense, 1993; Meyer *et al.*, 1994; Schaible and Grubb, 1993; Svensson *et al.*, 1997; Willis and Coggeshall, 1991). Certain of these aspects were evoked above and the following paragraphs summarize several, other distinctive features of nociceptive input from non-cutaneous tissues.

The above-mentioned differentiation between temporally-distinct components of 'sharp' and 'dull' cutaneous pain is not easily recognised for other organs. For example, both 'A δ ' fibre and 'C' fibremediated muscular pain (see below), irrespective of the time-scale, tends to present with a dull, aching and/or cramp-like quality (Marchettini *et al.*, 1996; Mense, 1993). Further, a distinguishing feature of deep somatic and visceral—as compared to cutaneous-pain is the pronounced and intenselyunpleasant, autonomic component (hypotension, nausea, perspiration, etc): this reflects the excitation of sympathetic and parasympathetic pathways. In addition, deep and visceral pain is often diffuse (poorly-localized) and frequently referred to other intact tissues (Section 9.1.4).

The majority of the uncapsulated terminals which innervate skeletal muscle correspond to small 'group III' myelinated fibres (equivalent to $A\delta$ fibres) and to 'group IV' unmyelinated fibres (equivalent to C fibres). Group III fibres, which are engaged by mechanical stimuli, are responsive to muscle stretch, contractions and innocuous pressure. However, they can be sensitized by thermal and chemical stimuli and about one third may be classified as nocisponsive, typically being activated by ischaemia/hypoxia

and localized, noxious increases in pressure (Abrahams et al., 1984; Marchettini et al., 1996; Mense, 1986, 1993; Mense and Meyer, 1985). Group IV fibres share several characteristics of group III fibres and they may also respond to thermal stimuli. Indeed, almost 50% of group IV fibres are nociceptive and, like group III fibres, they respond to ischaemia/hypoxia and localized, noxious increases in pressure. This population also mimics group III fibres in their polymodal-like response to, and sensitization by, chemical stimuli (Mense and Meyer, 1988; Mense, 1993). In line with the above observations, it has been suggested that 'polymodal-like' A δ (group III) fibres are more prominent in muscle than in skin and that their stimulation, in analogy to polymodal, cutaneous C fibres, plays an important role in initiating processes of sensitization in the spinal cord (Handwerker and Kobal, 1993; Marchettini et al., 1996; Mense, 1993). The intense discharge of nocisponsive, group III and IV skeletal fibres in response to excessive increases in pressure and local ischaemia (Mense and Stahnke, 1983; Mense, 1993) may be considered as fulfilling a 'warning' function analogous to the protective role of cutaneous nociceptors inasmuch as the onset of muscular pain leads to the discontinuation of excessive muscle contraction. Although, clinical, muscular pain is usually diffuse, it can be reasonably welllocalized under certain conditions (Marchettini et al., 1996).

Many group III and IV fibres (free nerve endings) innervating joint capsules—as well as a small proportion of large, *corpuscular*, myelinated $A\delta$ -like fibres—discharge only upon extremes of movement beyond the normal range (Grigg *et al.*, 1986; Schaible and Grubb, 1993). Thus, in analogy to muscle group III and IV afferents, one may consider their triggering of pain under these circumstances as fulfilling a 'warning' function.

Pain is an all-too-familiar and prominent sensation from the teeth. It likely reflects the activation of polymodal C and A δ fibres preferentially transmitting nociceptive information from the tooth pulp and the dentin, respectively (Meyer *et al.*, 1994).

Cerebral blood vessels (CBVs) are surrounded by a dense plexus of sensory nerves, many of which branch extensively. A recent description of nocisponsive PAFs innervating dural vessels suggested that this network constitutes an homogeneous population of polymodal (C) nociceptors (Bove and Moskowitz, 1997). These observations are of evident pertinence to the pathology of migraine and other forms of intracranial pain (Section 7.9).

Although less well-characterized than nociceptors in the skin, there is evidence for the existence of (polymodal) C and A δ fibres in internal organs such as the heart and gut (Cervero, 1982, 1994, 1995a; Gebhart, 1995; McMahon, 1994; Meller and Gebhart, 1992; Willis and Coggeshall, 1991). Nociceptive information from the viscera reaches the CNS via the sympathetic chains (for example, from the thorax and upper abdomen) and pelvic, parasympathetic chains (notably from the bladder and lower gut) (Cervero, 1994; Meyer *et al.*, 1994). Interestingly, Meller and Gebhart (1992) have suggested that both sympathetic *and* vagal (parasympathetic) afferents contribute to myocardial pain. The generally diffuse nature of deep and visceral pain, notably from the intestinal tract, and its frequent referral to other tissues, may be distinguished from the focal quality of cutaneous pain (Section 9.1.4). Indeed, the density of visceral afferents, for example in the gut, is low as compared to the skin. This sparse innervation, together with the large, weakly-defined, multiple and complex RFs of visceral nocisponsive units, accounts for the poor localization of noxious stimuli in the gut and other visceral tissues: it also explains why extensive stimulation of large tissue areas may be required to elicit pain, presumably via spatial summation (Cervero, 1995b; Handwerker and Kobal, 1993; Mense, 1993; Schaible and Grubb, 1993). Further, the absence of genuine $A\beta$ fibres in the viscera suggests that nonnoxious information from the viscera must be transmitted by A δ and C fibres, leading to a controversy as to whether visceral pain is either:

- mediated by specific nociceptors (the 'specificity theory'); or
- more probably, intensity-encoded by 'non-specific' fibres likewise responsive to innocuous stimulation ('pattern theory') (Cervero, 1994, 1995b; Handwerker and Kobal, 1993; Mailiani *et al.*, 1984; McMahon *et al.*, 1994).

3.1.3. 'Silent' Nociceptors

There is one further class of nociceptor which has attracted much recent interest within the context of prolonged pain: so-called 'silent' or, perhaps more appropriately, 'sleeping', nociceptors (Cervero, 1994; Koltzenburg, 1995; McMahon and Koltzenburg, 1990a,b; Schaible and Grubb, 1993; Treede et al., 1992b). Although no class of nociceptor actually shows spontaneous activity, the term 'silent' refers to a subset (perhaps 10-20%) of unmyelinated C fibres in the skin, joints and viscera-though see Cervero (1994)-which are normally irresponsive to acute noxious stimuli. Under conditions of inflammation and tissue injury, these 'silent' nociceptors are sensitized and activated by a variety of chemical mediators (Dmitrieva and McMahon, 1996: Koltzenburg, 1995; Reeh et al., 1987; Sengupta and Gebhart, 1994). Silent receptors have also been identified in human skin (Schmidt et al., 1995). This recruitment of otherwise-silent, chemosensitive PAFs under pathological states may contribute to temporal and spatial summation and considerably enhances the C fibre afferent barrage to the DH. For example, PAFs in arthritic joints develop both spontaneous activity and emit signals in response to previously-ineffective joint movements, and inflammation of the viscera may enormously enhance their sensitivity to otherwise, innocuous mechanical stimulation (Cervero, 1995a; Coutinho et al., 1996; Clément et al., 1996; Gebhart, 1995; Grigg et al., 1986; Schaible and Grubb, 1993; Schaible and Schmidt, 1988). Treede and Magerl (1995) have pointed out that some types of PAF may be regarded as 'partially silent' as concerns effective stimulus qualities. Thus, some high threshold (Type I), mechanosensitive A δ fibres only respond to thermal stimuli following their sensitization (Section 3.1.1). Further, an expansion of the RFs of PAFs following tissue injury may reflect the engagement of previously-silent branches. The activation of silent nociceptors underlines the distinctive nature of prolonged, inflammatory painful states as compared to the acute exposure to phasic noxious stimuli. Indeed, the recruitment of silent nociceptors likely makes an important contribution to the sensitization and central adaptive changes which underlie primary hyperalgesia due to tissue inflammation (Sections 7, 9 and 12).

3.1.4. 'Ectopic' Functions of Damaged PAFs and 'Central' Pain

As described in Section 8.2, injured peripheral nerves may form structures termed neuromas. These neuromas and/or their dorsal root ganglia (DRG) develop spontaneous electrical activity and become sensitive to inflammatory mediators, noxious stimuli and sympathetic input. Thus, orthodromic impulses conducted to the spinal cord are initiated independently and downstream of terminal-localized nociceptors. The induction of pain by damage to, or a dysfunction of, specific regions of the CNS itself is considered in Section 13.

3.2. Role of Neurotransmitters and Neuromodulators in the Transmission of Nociceptive Information by PAFs to the DH of the Spinal Cord

3.2.1. Multiple Mediators of Nociceptive Information in PAFs: Interaction with Intracellular Transduction Mechanisms

In Section 3.1.1, various types of PAF were classified on functional grounds according to the mode of sensory information which they encode and transfer to the DH. In this regard, several subtypes of C and A δ fibre were recognized. A complementary, neurochemical system of PAF categorization may be based upon the nature of the molecules which they contain and release. Indeed, nocisponsive PAFs synthesize a diversity of substances potentially involved in the central transmission and modulation of nociceptive information. These include glutamate (GLU) and other excitatory amino acids (EAAs); neuropeptides, such as substance P (SP) and calcitonin gene related peptide (CGRP); the universal, cellular energy source, adenosine triphosphate (ATP); the diffusible gas, nitric oxide (NO); the phospholipid metabolites, prostaglandins (PGs) and neurotrophins (growth factors). These potential transmitters, a diversity of other neuropeptides, various enzymes, several lectins and a miscellany of other molecules display a complex pattern of colocalization, comodulation and corelease in PAFs (He et al., 1990; Hunt and Rossi, 1985; Kashiba et al., 1996; Lawson, 1992; Lawson et al., 1997; Levine et al., 1993; Lynn, 1997; Mense, 1993; Merighi et al., 1991; Salt and Hill, 1983; Schaffar et al., 1997; Schaible and Grubb, 1993; Willis and Coggeshall, 1991; Yaksh and Malmberg, 1994). In principle, each specific, functional subclass of PAF might possess a characteristic complement of markers, but this remains to be demonstrated. In any case, the neurochemical composition of PAFs varies both qualititatively and quantitatively as a function of several factors.

Thus, differences are apparent:

- amongst various tissue types—skin, muscle, joints and the viscera (De Groat, 1986; McMahon, 1994; Gebhart, 1995; Lawson, 1992; Lawson et al., 1997; Marchettini et al., 1996; Mense, 1993; Salt and Hill, 1983; Schaffar et al., 1997; Schaible and Grubb, 1993);
- 2. between the intact state vs peripheral tissue inflammation and PAF injury (Goff *et al.*, 1998; Hökfelt *et al.*, 1994) (Sections 7.6 and 8.2.5); and
- 3. amongst various classes of PAF, for example, C fibres as compared to $A\beta$ fibres (Hökfelt *et al.*, 1994; Lawson, 1992; Salt and Hill, 1983; Willis and Coggeshall, 1991; Xu and Wiesenfeld-Hallin, 1997).

In addition, differences have been documented within a specific class of PAF, and 'peptidergic' and 'non-peptidergic' subpopulations of small calibre PAFs have been recognized for some years (Hunt and Rossi, 1985). A substantial percentage of C fibres are sensitive to the pungent ingredient of hot peppers and vanilloid, capsaicin (Section 7.4.5) and, of these, at least two major subpopulations may be identified (Averill et al., 1995; McMahon and Bennett, 1997; Molliver et al., 1997; Vulchanova et al., 1996; Willis and Coggeshall, 1991). The first contains CGRP and SP and is developmentally dependent upon the neurotrophin, nerve growth factor (NGF) (Section 7.6.1). The second, which is defined by presence of the lectin, IB-4, possesses a subclass of excitatory receptor for ATP on their peripheral terminals and is dependent upon glial cell derived nerve growth factor (GDNF) (Sections 7.4.4 and 11.3.4.3). Each of these classes of C fibre fulfils a key role in nociceptive transmission in the DH. Peptidergic populations of C fibres can be further sub-divided into additional subclasses based on supplementary, neurochemical criteria (Hökfelt et al., 1994; Kashiba et al., 1996; Lawson, 1992; Willis and Coggeshall, 1991).

Of particular interest, a recent study found that SP was more prominent in C fibres emanating from the muscle and other deeper, subcutaneous (s.c.) tissues than in C fibres innervating the skin, while cutaneous A δ fibres contained little of no SP (Lawson *et al.*, 1997; Lynn, 1997) (Section 4.7). Thus, although SP may be the most familiar and comprehensively-characterized, pronociceptive transmitter in PAFs, these observations draw attention to the fact that nocisponsive PAFs do *not* necessarily contain this tachykinin. Correspondingly, the presence of SP in PAFs is *not* indispensable for the central transmission of nociceptive information (Hunt and Rossi, 1985; Lawson, 1992; Lawson *et al.*, 1997; Lynn, 1997; Willis and Coggeshall, 1991).

Evidently, the neurochemical characterization of specific classes of PAF in relationship to their functional properties is far from complete and remains the topic of intensive study (Coggeshall and Carlton, 1997; Hökfelt *et al.*, 1994; Hunt and Rossi, 1985; Liedtke and Edelmann, 1996; Lynn, 1997; Todd and Spike, 1993; Yajiri et al., 1997; Yaksh and Malmberg, 1994).

In analysing the roles of PAF mediators in the central transmission of nociceptive information, it is essential to take account of their coupling to intracellular transduction mechanisms inasmuch as the nature of their influence upon ionic fluxes and soluble second messengers determines their impact upon cellular excitability. The opening of cation-ion permeable channels depolarizes neurones. Further, via a cascade of intracellular signals detailed in Section 10.4, activation of phospholipase C (PLC) and adenyl cyclase (AC) also triggers mechanisms leading to an increase in neuronal excitability and sensitivity. Correspondingly, for a diversity of transmitters released from PAF terminals onto intrinsic, nocisponsive DH neurones, where their receptors are positively or negatively coupled to cation ion flux, PLC and/or AC, they exert pronociceptive or antinociceptive properties, respectively (Sections 3.2.4 and 3.2.6).

The following paragraphs summarize the role of PAF-localized mediators in the transmission of nociceptive information to the DH in relation to their interaction with intracellular transduction mechanisms. Notwithstanding the above comments concerning multiple, neurochemical classes of nocisponsive fibre, the ensuing discussion focuses on the synergistic roles of colocalized EAAs, SP and CGRP in view of the abundance of information available concerning their cooperative actions and influence upon intracellular transduction mechanisms involved in the sensitization of DH neurones, a process playing a key role in the induction of longterm painful states. Thereafter, the potential significance of other neuropeptides contained in PAFs is outlined (Section 3.2.6). The functional roles of ATP, NO and PGs in the DH are discussed in Section 3.2.8.

Molecular and intracellular mechanisms underlying the excitation and sensitization of DH neurones by nociceptive input from PAFs are discussed in detail in Sections 10.2 and 10.3.

3.2.2. Role of Colocalized EAAs, Tachykinins and CGRP in Small Calibre PAFs Projecting to the DH

As mentioned earlier, EAAs, tachykinins and CGRP are colocalized in a subset of capsaicin-sensitive, nocisponsive, small calibre fibres projecting to the DH from the skin and other tissues (Battaglia and Rustioni, 1988; Kashiba *et al.*, 1996, 1997; Lawson, 1992; Merighi *et al.*, 1991; Willis and Coggeshall, 1991).

The pronociceptive tachykinins, SP and neurokinin (NK) A, act at NK₁ and NK₂ receptors, respectively—although NKA may also exert actions at NK₁ sites, and SP at NK₂ sites. Both NK₁ and NK₂ receptors are positively coupled to PLC (Bentley and Gent, 1995; Chapman *et al.*, 1996; Hoheisel *et al.*, 1997; Maggi and Schwartz, 1997; Mantyh *et al.*, 1997; Nagy *et al.*, 1994; Radhakrishnan and Henry, 1997; Rupniak *et al.*, 1996; Seguin *et al.*, 1995; Sluka *et al.*, 1997a,b; Tao *et al.*, 1997; Traub, 1996).

A further, pronociceptive neuropeptide, CGRP, exists as two (α and β) isoforms. CGRP exerts its

actions via (at least) two receptor types, CGRP₁ and CGRP₂, both of which are positively coupled to AC, although their respective roles in mediating the actions of CGRP on intrinsic neurones in the DH remain to be characterized (Brain and Cambridge, 1996; Edvinsson *et al.*, 1997; Wimalawansa and El Kholy, 1993; Olesen and Janssen-Olesen, 1997; Woolf and Wiesenfeld-Hallin, 1986; Yu *et al.*, 1996).

EAAs, such as GLU and aspartate, act at both metabotropic (mGlu) receptors (coupled via G-proteins to soluble second messengers) and ionotropic receptors (coupled directly to cation-permeable ion channels): these receptors show a complex pattern of localization on various neuronal classes in the DH (Chaplan et al., 1997; Coggeshall and Carlton, 1997; Conn and Pin, 1997; Dickenson, 1997; Dougherty et al., 1992b,c; Fletcher and Lodge, 1996; Grubb et al., 1996; Haley et al., 1990; Lerea, 1997; Thompson et al., 1995; Valerio et al., 1997; Yoshimura and Nishi, 1992). As outlined in Section 3.2.3, based on the nature of their coupling to intracellular transduction mechanisms and pharmacological properties, three major classes of mGlu receptor-and at least eight subtypes-have been recognized (Conn and Pin, 1997; Valerio et al., 1997). The major types of ionotropic receptor involved in transmitting nociceptive messages in the DH are termed DL-a-NH2-2,3-dihydro-5-methyl-3oxo-4-isoxazolepropanoic acid (AMPA) and Nmethyl-D-aspartate (NMDA). The significance of NMDA and AMPA receptor isoforms is summarized in Section 3.2.3. Ionotropic, kainate receptors are present in the DH, but the lack of selective, pharmacological tools has hitherto hampered analysis of their putative role in the spinal transmission of nociceptive information (Fletcher and Lodge, 1996). Nevertheless, a selective, high affinity ligand at kainate receptors, SYM 2081, was recently described. SYM 2081 rapidly desensitizes kainate receptors and attenuates the mechanical allodynia and thermal hyperalgesia provoked by PAF injury. However, it is unclear whether the kainate receptors involved are engaged by small and/or large calibre PAFs (Sutton et al., 1997; Zhou et al., 1997a). Further, it is uncertain whether they are localized on intrinsic DH neurones or on the central terminals of small calibre PAFs (Bettler and Mulle, 1995; Coggeshall and Carlton, 1997; Dickenson, 1997; Fletcher and Lodge, 1996; Huettner, 1990) (Sections 3.2.4 and 9.1.4.2).

3.2.3. Multiple Isoforms of NMDA, AMPA and MGlu Receptor

The cloning of NMDA and AMPA receptors has identified a multiplicity of subunits allowing for the assembly of a variety of heteromultimeric isoforms (subtypes) in each case (Dickenson, 1997; Hollmann and Heinemann, 1994). The various isoforms may display contrasting properties: for example, differences in their affinity for GLU and in their threshold for activation. The putative roles of individual isoforms in the transmission of nociception in the DH (and elsewhere) is the focus of much current interest. Multiple classes of mGlu subunits have also been cloned. The functional properties of multiple NMDA, AMPA and mGlu receptor subtypes may be briefly summarized as follows.

NMDA receptors are composed of heteromultimeric subunits and comprise several, distinct classes. These are differentially distributed in the CNS, with 'NMDA NR₁' component predominant the throughout the DH. Although NMDA receptor isoforms may show contrasting pharmacological properties, they all display slow channel kinetics, a certain degree of voltage-dependent Mg⁺-block and marked permeability to Ca^{2+} (Conti *et al.*, 1997; Dickenson, 1997; Hollmann and Heinemann, 1994; Liu et al., 1997d; Momiyama et al., 1996; Sucher et al., 1996) (Section 10.3.1). In addition to the recognition site for NMDA itself, NMDA receptors bear a modulatory (glycine B) site for glycine, the occupation of which is essential for functioning of the coupled, ion channel (Dickenson, 1997; Heppenstall and Fleetwood-Walker, 1997a,b; Hollmann and Heinemann, 1994; Leeson and Iversen, 1994; Seguin et al., 1995). In the DH, glycine is apparently present in a concentration sufficient to saturate these sites.

Like NMDA receptors, AMPA receptor isoforms are composed of several subunits: of these, the 'Glu R-B' (or 'Glu R2') moiety plays a major role in determining Ca^{2+} -permeability, conductance and certain other properties. Several post-transcriptional variants of AMPA receptors may also exist (Bettler and Mulle, 1995; Pellegrini-Giampietro et al., 1997; Washburn et al., 1997). AMPA receptor isoforms are differentially distributed within the spinal cord (Dickenson, 1997; Fletcher and Lodge, 1996; Furuyama et al., 1993; Pellegrini-Giampietro et al., 1997; Tachibana et al., 1994; Tomiyama et al., 1996). In contrast to NMDA receptors, the conductance of AMPA receptors localized on nocisponsive neurones in the DH is predominantly to Na^+ rather than to Ca2+. Further, in distinction to NMDA receptors, AMPA receptors display lower affinity for GLU, low voltage-dependence and rapid kinetics (Furuyama et al., 1993; Tachibana et al., 1994; Tomiyama et al., 1996). Moreover, AMPA receptors can rapidly desensitize upon their selective stimulation (Fletcher and Lodge, 1996). Like NMDA receptors, AMPA receptors possess several allosteric site(s). Actions at these sites modulate the activity and sensitivity of AMPA receptors, in particular the kinetics of desensitization. However, in contrast to the glycine_B site of NMDA receptors, 'co-agonist' occupation of allosteric sites on AMPA receptors does not appear to be obligatory for activation of the coupled ion channel (Bettler and Mulle, 1995; Fletcher and Lodge, 1996). The ionic and functional properties of kainate receptors are similar to those of their AMPA counterparts (Fletcher and Lodge, 1996).

Numerous (8) types of mGlu receptor have been identified, of which splice variants may also exist. They are divided into three groups. Group I (mGlu₁ and mGlu₅) receptors, which are concentrated in the superficial DH, are positively coupled to PLC—and possibly, NO synthase. Group II (mGlu₂ and mGlu₃) receptors and group III mGlu receptors (mGlu₁ and mGlu₆₋₈), in contrast, are both negatively coupled to AC, but show differences in their

pharmacological profiles (Conn and Pin, 1997; Valerio et al., 1997). Thus, specific mGlu receptor types differentially modify intracellular calcium concentrations ($[Ca^{2+}]_i$) and the activity of several transduction mechanisms, including various protein kinases (PKs). As described in Section 10.4. $[Ca^{2+}]_i$, and PKs play key roles in the control of neuronal excitability via the functional regulation (phosphorylation) of several receptor types/ion channels and the modulation of gene transcription. Certain types of mGlu receptor may also modify K⁺- and Ca^{2+} -ion currents, but such actions have not, as yet, been characterized at DH neurones (Conn and Pin, 1997). The implication of various mGlu receptor subtypes in the modulation of nociceptive transmission is currently being explored as a function of their differential levels of expression in the DH and of their contrasting influence upon intracellular transduction mechanisms. In this regard, it is likely that group I (mGlu₁ and mGlu₅) receptors, which predominate in the superficial DH, play an important role in the mediation of nociceptive input from small calibre PAFs (Budai and Larson, 1998; Fisher and Coderre, 1996; Pin and Duvoisin, 1995; Toms et al., 1996; Valerio et al., 1997; Young et al., 1997) (Section 10.3.2.3).

3.2.4. Actions of EAAs and SP at PAF Terminals in the DH

The modulation of nociception by EAAs in the DH reflects not only effects on intrinsic DH neurones, but also actions on the terminals of PAFs from which they are released (Ferreira and Lorenzetti, 1994). Indeed, NMDA and various mGlu receptors have been documented both postsynaptically to PAFs on DH neurones, as well as presynaptically on the central terminals of PAFs. The DRG of small calibre PAFs also express kainate and AMPA receptors, but their presence on central PAF terminals in the DH remains to be definitively demonstrated (Coggeshall and Carlton, 1997; Dickenson, 1997; Fletcher and Lodge, 1996) (Section 7.4.8). The activation of presynaptic populations of NMDA receptors on PAF terminals in the DH, via a positive feedback action, further increases the release of EAAs and SP, thereby amplifying synaptic transmission (Li et al., 1997a; Liu et al., 1994b; Marvizon et al., 1997; Sato et al., 1993a; Ye and Westlund, 1996). On the other hand, the activation of specific mGlu receptor subtypes localized on PAF terminals might either inhibit or enhance presynaptic release as a function of their coupling to intracellular transduction mechanisms, and the precise roles of individual mGlu receptor types on PAF terminals in the DH remain to be clarified. Currently, most evidence favours a predominantly inhibitory role of group II and III mGlu receptors via the inhibition of Ca^{2+} currents and/or an enhancement of K⁺-currents (Fisher and Coderre, 1996; Herrero et al., 1992; Li et al., 1997a; Valerio et al., 1997). In addition to NMDA and mGlu receptors, the presence of excitatory receptors for SP (presumably of the NK1 type) on PAF terminals in the DH has been proposed (Coggeshall and

Carlton, 1997; Hu et al., 1997). Like EAAs, then, SP in the DH may exert not only direct, postsynaptic actions on intrinsic DH neurones, but also positive feedback actions to enhance its own release from PAF terminals. Further, NMDA and AMPA receptors may also be present on the terminals of *large* calibre, $A\beta$ fibres in the DH (Coggeshall and Carlton, 1997; Wood and Docherty, 1997) raising the intriguing possibility that their activation may play a role in the induction of mechanical allodytnia (Section 9.1.2). In addition to these direct feedback actions at PAF terminals, EAA and SP may indirectly enhance their own release by triggering the liberation of retrograde messengers (such as NO), from their target neurones in the DH (Section 3.2.8).

The above comments concern actions of EAAs and SP at sites *other* than intrinsic DH neurones. In addition, apart from PAF terminals, EAAs and SP in the DH may be derived from *other* sources: that is, intrinsic neurones, the terminals of descending pathways and/or glial cells (Coggeshall and Carlton, 1997; Todd and Spike, 1993; Willis and Coggeshall, 1991). Evidently, then, the role of EAAs/SP in the DH is not restricted to their release from small calibre PAF terminals and the subsequent, postsynaptic activation of intrinsic neurones in the DH. Nevertheless, those actions remain the principal focus of the present article in view of their key importance in eliciting central sensitization (Section 10.3).

3.2.5. Cooperative, Excitatory Actions of EAAs, Tachykinins and CGRP on DH Neurones

EAAs, tachykinins and CGRP all cooperatively and synergistically elicit excitatory postsynaptic potentials (EPSP)s in DH neurones with contrasting temporal patterns, and their respective roles depend upon the nature and duration of noxious stimulation (Section 10.3). The initial trigger is probably provided by activation of AMPA receptors, the principal mode of rapid, excitatory transmission throughout the CNS. AMPA receptors display extremely rapid kinetics and their stimulation mediates a rapid depolarization (inward current) over a few msec. The subsequent engagement of NMDA, group I mGlu, NK₁, NK₂ and CGRP receptors is associated with slower and more sustained EPSPs of up to tens of seconds. Consequently, repetitive or persistent noxious stimulation can lead to temporal summation and amplification of responses in the DH. The precise sequence of events, as well as their significance for the induction of prolonged painful states, is explained in detail in Section 10.3. Presynaptic inhibition of the release of EAAs, tachykinins and/or CGRP, or blockade of their postsynaptic actions, offers, then, presynaptic and postsynaptic mechanisms, respectively, for the interruption of nociceptive transmission in the DH.

3.2.6. Other PAF-Derived Neuropeptides Potentially Modulating Nociception in the DH

In addition to tachykinins and CGRP, many other neuropeptides, displaying varying patterns of colocalization, are present in normal and/or injured, fine and/or large calibre PAFs innervating the DH (Coggeshall and Carlton, 1997; Hökfelt *et al.*, 1994; Kashiba *et al.*, 1997; Lawson, 1992; Mense, 1993; Salt and Hill, 1983; Schaible and Grubb, 1993; Willis and Coggeshall, 1991). Several of these reappear later in this review as playing a role—either peripherally and/or centrally—in the modulation of nociception under pathological conditions (Sections 7.6 and 8.2.5).

In discussing the putative, functional properties of neuropeptides, it is important to note several, general characteristics common to EAAs, tachykinins and CGRP as follows.

- 1. They may be derived from several neuronal sources: PAFs, terminals of descending pathways and intrinsic DH neurones (Sections 4.6 and 5).
- 2. Individual neuropeptides may exert a complex pattern of pro and/or antinociceptive actions at multiple loci: intrinsic neurones in the DH, PAF terminals, preganglionic sympathetic neurones and non-neuronal (glial and immunecompetent cells). Similarly, in the periphery, they may act at PAF terminals, sympathetic fibre terminals, glial and immunecompetent cells and the vasculature, etc.
- 3. For most neuropeptides, they exert their actions via a multiplicity of receptor types.
- 4. In considering their influence upon the functional activity of target cells and nociception, it is *imperative* to take account of their coupling to intracellular transduction mechanisms, the nature of which may differ for specific receptor types.

Under normal conditions, levels of neuropeptide Y (NPY) in the DRG are very low. However, following PAF injury, NPY is up-regulated in small and, to a greater extent, large calibre PAFs (Hökfelt et al., 1994) (Section 8.2.5). The spinal and peripheral role(s) of NPY in the modulation of nociception are highly complex (Munglani et al., 1996). However, NPY appears to predominantly decrease nociception in the DH, yet primarily to increase nociception in the periphery. NPY exerts these actions via a multiplicity of 'Y' receptor types: all are negatively coupled to AC but they differentially enhance K⁺-currents and reduce Ca²⁺-currents and $[Ca^{2+}]_i$ levels. These receptor subtypes are also differentially localized on intrinsic neurones in the DH, and on the central and peripheral terminals of PAFs and sympathetic nerves (Bromqvist and Herzog, 1997; Colmers and Bleackman, 1994; Munglani et al., 1996; Sun et al., 1998; Zhang et al., 1994c, 1997b).

A further neuropeptide which exerts antinociceptive actions in the DH, somatostatin, is likewise negatively coupled via multiple receptor subtypes to AC and it is found principally in small calibre PAFs, independently of CGRP/SP (Carlton and Coggeshall, 1997; Kashiba *et al.*, 1996, 1997; Mollenholt *et al.*, 1990).

Corticotropin releasing factor (CRF) has been detected in capsaicin-sensitive, fine calibre PAFs in the DH, wherein it may play a role in the modulation of nociception via several different types of receptor positively coupled to AC (Coggeshall and Carlton, 1997; Schäfer et al., 1997; Skofitsch et al., 1985).

For technical reasons, the demonstration of cholecystokinin (CCK) in PAFs has proven problematic and, under resting conditions, it is only weakly expressed. However, it is markedly induced by PAF injury (Hökfelt et al., 1994) (Section 8.2.5). CCK interferes with antinociceptive mechanisms in the DH, possibly via an inhibition in the release of opioids and via the induction of PLC which leads to an increase in $[Ca^{2+}]_i$ levels. In rodents, CCK_B receptors, which are involved in these actions, have been detected both on intrinsic DH neurones and on the terminals of small and large calibre PAFs (Hökfelt et al., 1994; Ossipov et al., 1997; Stanfa et al., 1994; Vanderah et al., 1996a,b; Wang et al., 1992; Wiertelak et al., 1992a, 1994d). In primates and man, however, CCKA receptors predominate in the DH and may likewise contribute to the pronociceptive actions of CCK (Benedetti, 1997; section 9.5).

The role of galanin (GAL) is of particular interest inasmuch as it acts predominantly pronociceptively under normal conditions, yet antinociceptively following damage to PAFs (Section 8.2.5). It is found in both C and A β fibres and exerts its effects via several receptor types. Of these, GAL₁ sites are coupled negatively to AC and they enhance and reduce Ca^{2+} - and K⁺-currents, respectively. GAL₁ sites presumably mediate the antinociceptive actions of GAL on DH neurones-and, possibly, PAF terminals. In contrast, GAL₂ receptors are positively coupled to PLC. Small calibre DRG cells express both GAL₁ and GAL₂ receptors (Kask et al., 1997; Parker et al., 1995; Selve et al., 1996; Shi et al., 1997; Xu et al., 1990; Xu et al., 1996b, 1997d; Xu and Wiesenfeld-Hallin, 1997).

Vasoactive intestinal peptide (VIP) is more highly expressed in visceral than in cutaneous PAFs. Further, it is present in both small and large calibre cutaneous afferents. VIP acts pronociceptively in the DH, and this role may be accentuated subsequent to nerve injury (De Groat, 1986; Hökfelt et al., 1994; Kask et al., 1997; Vertongen et al., 1997; Xu et al., 1990) (Section 8.2.5). Pituitary adenylate cyclase activating peptide (PACAP), which is structurally related to VIP, possesses higher affinity than VIP at PACAP_I receptors, and shares actions with VIP at $PACAP_2/VIP_1$ and $PACAP_3/VIP_2$ receptors. Various splice variants of PACAP₁ receptor differentially activate AC and PLC, while PACAP₂/VIP₁ and PACAP₃/VIP₂ receptors are positively coupled to AC (Rawlings and Hezareh, 1996). Like VIP, PACAP is detected in small calibre, capsaicin-sensitive fibres and it is an important trophic factor (Moller et al., 1997a,b). PACAP can probably enhance nociception at the spinal level, though antinociceptive actions have also been reported (Xu and Wiesenfeld-Hallin, 1996; Yamamoto and Tatsuno, 1995; Zhang et al., 1993c, 1996b, 1997g).

To summarize, many neuropeptides are present in small and/or large calibre PAFs innervating the DH. As discussed in Sections 7.6 and 8.2.5, respectively, their expression is subject to profound changes upon either stimulation and/or damage to PAFs. Various neuropeptides are differentially coupled via multiple receptor types to contrasting transduction mechanisms and, correspondingly, fulfil divergent roles in the modulation of nociception. It is important to emphasize, thus, that actions of neuropeptides in PAFs do not exclusively enhance nociception. Further, the relationship between pronociceptive and antinociceptive species is modified by manipulations of PAFs which trigger prolonged pain. As discussed throughout this review, a 'balance' between pronociceptive and antinociceptive mechanisms may be a general feature of the processing and transfer of nociceptive information at all hierarchical levels, from the periphery via the DH to supraspinal centres. A disruption of this equilibrium may be reflected in the induction or aggravation of painful states (Section 15.1).

3.2.7. Putative Neurotransmitters in $A\beta$ Fibres

Activation by EAAs, such as GLU and aspartate, of AMPA receptors is likely the principal mechanism involved in the influence of $A\beta$ fibres upon intrinsic neurones in the DH (Coggeshall and Carlton, 1997; Cumberbatch *et al.*, 1994; Dickenson, 1997; Dougherty *et al.*, 1992b; Neugebauer *et al.*, 1993; Sherman and Loomis, 1996; Sherman *et al.*, 1997a,b; Xu *et al.*, 1993b). However, as mentioned above, following PAF stimulation and/or injury, the expression by $A\beta$ fibres of several neuropeptides is augmented, including NPY, GAL, CCK and, notably, SP (Hökfelt *et al.*, 1994; Willis and Coggeshall, 1991) Section 8.2.5).

3.2.8. Pronociceptive Mediators in the DH Derived from PAF Terminals and Intrinsic DH Sources: Focus on ATP, NO and PGs

3.2.8.1. Unconventional mediators of nociception in the DH

Recently, a role has emerged for ATP, NO and PG as potential transmitters of nociceptive information in the DH. Each of these mediators may be derived from small or large calibre PAF terminals, and the potential excitation of nocisponsive DH neurones by ATP and PG released from C fibres is a particularly intriguing possibility (Sections 3.2.8.2 and 3.2.8.4). In addition to PAF terminals, ATP, NO and PGs may also be derived from intrinsic neuronal and non-neuronal sources in the DH—in particular neurones targeted by PAFs themselves (Valtschanoff *et al.*, 1992, 1995; Stanfa *et al.*, 1996; Yaksh and Malmberg, 1994). Further, ATP, NO and PGs may all exert actions at intrinsic DH neurones, at PAF terminals and/or at non-neuronal cells.

3.2.8.2. ATP

A putative role of ATP in the transmission of nociceptive input from PAF terminals onto DH neurones has attracted much recent attention. A diversity of ionotropic, metabotropic—and even cytolytic-receptor types exist for ATP (Burnstock, 1996; North and Barnard, 1997). Some years ago, ATP was reported to excite a sub-population of neurones in the DH, and to potentiate their responsiveness to NMDA (Jahr and Jessel, 1983). Such

actions likely involve excitatory, ionotropic and Ca^{2+} -permeable receptors of the P_{2X} family. Thus, P_{2x2} , P_{2x3} , P_{2x4} and, possibly, other P_{2x} receptor subtypes may be localized both on intrinsic neurones and on the central terminals of fine calibre, nocisponsive fibres in the DH (Collo et al., 1996; Le et al., 1998; Vulchanova et al., 1996). A central, pronociceptive action of ATP would be analogous to its pronociceptive actions exerted via $P_{2X2/X3}$ receptors on the peripheral terminals of C fibres (King et al., 1997b) (Section 7.4.4). Indeed, the activation of presynaptic P_{2X2} receptors on the terminals of sensory neurones in the DH was recently shown to elicit the release of GLU (Gu and MacDermott, 1997). In addition, an action of ATP at G-protein coupled, metabotropic P2Y receptors activating PLC was previously indicated to enhance EAA-mediated transmission in the superficial DH (Li and Perl, 1995).

ATP is rapidly metabolized to adenosine which, like ATP, acts via a multiplicity of receptor subtypes. The role of adenosine in the modulation of nociception is complex and expressed in interaction with both opioids and monoamines. One consistent action is its ability to alleviate the mechanical allodynia associated with PAF- or CNS-injury (Section 13.2). This effect is likely mediated via A_1 receptors, which are negatively coupled to AC and which inhibit and enhance Ca²⁺- and K⁺-currents, respectively (Cui et al., 1997b; Lee and Yaksh, 1996; Reeve and Dickenson, 1995; Salter and Henry, 1987; Sjölund et al., 1996, 1998; Sollevi et al., 1993; Sawynok and Sweeney, 1989; Sawynok and Reid, 1996; Yang et al., 1994). The A₁ receptors involved in this action are probably localized on intrinsic DH neurones rather than on PAF terminals (Dickenson, 1997; Hu and Li, 1997; Sánchez-Prieto et al., 1996; Sawynok and Sweeney, 1989). Indeed, A1 receptor agonists inhibit C fibre/NMDA receptor-mediated sensitization of DH neurones (Dickenson, 1997; Reeve and Dickenson, 1995). The source of adenosine exerting these actions is unclear, but an intriguing possibility is that it may be derived from $A\beta$ fibres which themselves mediate mechanical allodynia (Salter and Henry, 1987) (Section 9.1.1). Based on studies in the hippocampus, A_{2A} receptors, which are *positively* coupled to AC and PKA, may act pronociceptively in the DH via a postsynaptic enhancement of AMPA currents, though this remains to be directly demonstrated (Ben-Ari et al., 1992; Kessey and Mogul, 1997; Salter et al., 1993; Sawynok and Sweeney, 1989).

3.2.8.3. NO

The intra and intercellular messenger, NO, is produced in a wide variety of cell types and exert actions both at its site of formation as well as, following extracellular diffusion and membrane penetration, at neighbouring sites. NO is generated upon the catabolism of L-arginine to L-citrulline by the enzyme, NO synthase. Several (3) isoforms of NO synthase exist, including 'neuronal' (NOS-I) and 'endothelial' forms, both of which are found in neurones and of which splice variants may exist (Dawson and Snyder, 1994; Garthwaite and Boulton, 1995; Zhang and Snyder, 1995). Neuronal NO synthase is the major source of neuronal pools of NO. It shows constitutive (ongoing) activity, although its activity can also be enhanced by an increase in $[Ca^{2+}]_i$, triggered by, for example, stimulation of NMDA receptors. Indeed, NO may play a broad role throughout the CNS in alterations (facilitation) of synaptic efficacy mediated by NMDA receptors and other mechanisms (Brenman and Bredt, 1997; Dawson and Snyder, 1994; Hölscher, 1997; Lerea, 1997; Malen and Chapman, 1997; Stanfa et al., 1996; Zhang and Snyder, 1995) (Section 10.4.2.2). Under certain conditions, such as tissue damage, non-neuronal glial and immunecompetent microglial cells may become important sources of NO and these cell types possess an induci*ble* NO synthase likewise activated by $[Ca^{2+}]_i$, and responsive to, for example, cytokines, neurotrophins and PGs (Section 4.8). NO is coupled to several intracellular transduction mechanisms, including the generation of cyclic guanidine monophosphate (cGMP) which subsequently activates a specific PKG (Clementi and Meldolesi, 1997; Garthwaite and Boulton, 1995; Wang and Robinson, 1997) (Section 10.4.3.3).

In line with a pronociceptive role of NO in the DH, NO synthase is up-regulated in the DH- and DRG-under conditions of cutaneous or visceral inflammation and PAF injury, while drugs which either induce ('donors') or inhibit NO synthase, respectively, potentiate and inhibit the accompanying nociception (Herdegen et al., 1993; Lam et al., 1996; Meller and Gebhart, 1993; Meller et al., 1992a,c; Rice, 1995b; Pandita et al., 1997; Seguin et al., 1995; Vergé et al., 1992b; Xu and Wiesenfeld-Hallin, 1997; Wiesenfeld-Hallin et al., 1993). Further, there is evidence that a C fibre-mediated induction of neuronal NO synthase is involved in the sensitization of DH neurones which underlies painful states due to inflammatory tissue damage and PAF injury (Meller and Gebhart, 1992; Seguin et al., 1995; Stanfa et al., 1996; Taylor et al., 1995; Xu and Wiesenfeld-Hallin, 1997) (Section 10.4.2.2).

NO can exert actions intracellularly at its site of origin. It can also act as a vehicle of intercellular communication (Dawson and Snyder, 1994; Garthwaite and Boulton, 1995). Correspondingly, there are several potential mechanisms whereby NO may facilitate nociception.

- NO may exert actions in intrinsic DH neurones transmitting nociceptive information to cerebral centres. These pools of NO are likely generated in the neurones themselves following C fibre stimulation/NMDA receptor activation and an increase in [Ca²⁺]_i. In these DH neurones, NO plays a role in their sensitization by the generation of cGMP and the subsequent PKGmediated phosphorylation of specific membraneassociated proteins: it also, in the longer-term, modulates gene expression (Clementi and Meldolesi, 1997; Garthwaite and Boulton, 1995; Wang and Robinson, 1997) (Section 10.4.2.5).
- NO may increase nociception by retrogradely diffusing from sensitized DH neurones into presynaptic, fine calibre PAF terminals wherein it enhances the release GLU, SP and CGRP, poss-

ibly via a mechanism involving induction of cGMP, though this remains to be clarified (Coderre *et al.*, 1993; Coderre and Yashpal, 1994; Garry *et al.*, 1994c; Kitto *et al.*, 1992; Meller and Gebhart, 1993; Niedbala *et al.*, 1995; Sorkin, 1993). Intrinsic PAF and other pools of NO could also be involved in these actions.

3. Intrinsic and extrinsic pools of NO may modify the activity of non-neuronal glial and immunecompetent cells: for example, enhancing their synthesis of PGs and cytokines (Section 4.8).

The relative importance of these mechanisms, and of PAF and other tissue sources of NO, in the modulation of nociception remains to be further clarified (Sections 4.8 and 10.4.2.2). Further, the precise role(s) of NO in the modulation of nociception are still under discussion inasmuch as NO likely exerts a multiplicity of actions upon many cell types, some of which may even lead to a decrease in the excitability of DH neurones transmitting nociceptive information (Clementi and Meldolesi, 1997; Dickenson, 1997; Haley et al., 1992; Hoheisel et al., 1995; Inoue et al., 1997; Manzoni et al., 1992; Miller et al., 1997; Pehl and Schmid, 1997; Valtschanoff et al., 1992, 1995; Xu et al., 1997d; Xu and Wiesenfeld-Hallin, 1997; Wiesenfeld-Hallin et al., 1993).

3.2.8.4. PGs

The enzyme, cyclooxygenase (COX), catalyses the synthesis of PGs from arachidonic acid, a ubiquitous component of cells which is generated from phospholipids by an action of phospholipase A2 (PLA₂). In analogy to NO, PGs are derived from both non-neuronal and neuronal pools and there exist several isoforms of COX. Generally-speaking, a constitutively-active COX 1 generates the physiological pools of PG required for cellular function while upon appropriate stimulation, an inducible COX 2 generates potentially huge and pathological quantities of PG. Although the COX 2 isoform identified in neurones shows some constitutive activity, its functional status can be markedly modulated by synaptic events. Notably, neuronal synthesis of PGs is accelerated via the activation of NMDA receptors and an increase in [Ca²⁺]_i (Beiche et al., 1996; Breder et al., 1995; Willingale et al., 1997; Yamagata et al., 1993) (Section 10.4.2.3). This COX 2 isoform predominates in the spinal cord, wherein it is concentrated in several regions receiving nociceptive input: laminae I, II, and X (Section 4.1). COX 1 is also found in the DRG of PAFs (Beiche et al., 1996; Willingale et al., 1997).

The peripheral administration of capsaicin or exposure to inflammatory stimuli is accompanied by a rise in extracellular levels of PG in the spinal cord and by an increase in the gene expression of COX 2. However, the relative contribution of neuronal (PAF terminals and intrinsic neurones) and non-neuronal sources to increases in spinal levels of PG remains to be elucidated (Hay and Belleroche, 1997; Hingtgen and Vasko, 1994; Hingtgen *et al.*, 1995; Hua *et al.*, 1997; Ichitani *et al.*, 1997; Scheuren *et al.*, 1997; Willingale *et al.*, 1997; Yang *et al.*, 1996). In

line with an increase in PG activity, spinal administration of anti-inflammatory (COX-inhibiting) drugs suppresses C fibre reflexes, inhibits DH neuronal sensitization and attenuates prolonged, inflammatory pain (Beiche et al., 1996; Bianchi and Panerai, 1996; Bustamante et al., 1997; Herrero et al., 1997; Malmberg and Yaksh, 1992; McCormack, 1994; Willingale and Grubb, 1996; Willingale et al., 1997; Yamamoto and Nozaki-Taguchi, 1996, 1997). Under long-term, painful conditions, an antinociceptive action of selective COX 2 inhibitors has been observed, but COX 1 may also be involved in generating PGs when noxious stimulation is of a limited duration (Dirig et al., 1997; Hammond and Gregory, 1996; Ichitani et al., 1997; Yamamoto and Nozaki-Taguchi, 1996). Consistent with these findings, the spinal administration of PGs induces hyperalgesia to noxious stimuli and allodynia to innocuous, tactile stimuli (Minami et al., 1997b; Taiwo and Levine, 1986).

The mechanisms underlying PG-induced nociception are unclear. One possibility is that PGs exert actions at excitatory receptors localized on intrinsic DH neurones targetted by PAF terminals (see next paragraph and Sections 7.4.3 and 7.5.2.3). A retrograde function might also be evoked whereby, following their release from PAFs or postsynaptic sites, PGs may enhance the release of GLU and SP from PAF terminals, likely via a mechanism involving an increase in levels of cAMP and an increase in Ca²⁺- and/or Na⁺-conductance (Ferreira and Lorenzetti, 1996; Hingtgen et al., 1995; Minami et al., 1997a,b; Nishihara et al., 1995; White, 1996). In addition, arachidonic acid and leukotrienes, via an influence upon PKC activity in PAF terminals, may likewise retrogradely enhance GLU/SP release (Collins and Davies, 1998; Sánchez-Prieto et al., 1996) (Section 10.4.2.3).

The role of multiple (G-protein coupled) prostanoid receptors in mediating the pronociceptive actions of PGs also remains to be further clarified (Coleman et al., 1994; Boie et al., 1997). Notably, PGD₂ and PGI₂ are coupled to receptors increasing AC activity, while PGF_2 is coupled to receptors increasing levels of phosphoinositides and [Ca²⁺]_ipresumably via PLC (Coleman et al., 1994; Minami et al., 1997b; Mnich et al., 1995; Taiwo and Levine, 1986). Of particular interest is PGE₂, which is probably the principal form of pronociceptive PG in the periphery (Sections 7.4.3 and 7.5.2). PGE₂ acts via 4 receptor types coupled to the following transduction mechanisms: EP_1 (elevation of $[Ca^{2+}]$); EP_2 and EP_4 (activation of AC) and EP_3 (inhibition of AC and, possibly, generation of phosphoinositides and $[Ca^{2+}]_i$ (Coleman *et al.*, 1994; Boie *et al.*, 1997). MRNA encoding EP₂ receptors is concentrated in laminae I and II, while mRNA for EP1 and EP3 receptors has been found in sensory neurones (Sugimoto et al., 1994; Kawamura et al., 1997a). Although their relationship to multiple EP receptors is unclear, binding sites for PGE₂ have been found both in the trigeminal nucleus and in superficial DH laminae (Matsumura et al., 1992). This pattern of localization is consistent with putative actions of PGs at PAF terminals and/or intrinsic DH neurones in promoting nociception. While a role for cerebral

 EP_3 receptors in enhancing mechanical nociception has been claimed (Oka *et al.*, 1997), the functional roles of individual EP receptor types at the level of the DH remains unclear. However, it has been suggested that PGE_2 elicits allodynia via EP_1 receptors and hyperalgesia via EP_2/EP_3 receptors (Minami *et al.*, 1994).

As concerns the actions of various PGs, it has been suggested that PGE₂ and PGD₂ increase nociception via complementary GLU- and SP-mediated mechanisms, respectively. Thus, PGE2- but not PGD₂-induced hyperalgesia is abolished in NMDA receptor knock-out mice, while PGD2- but not PGE₂-induced hyperalgesia is blocked by the NK₁ antagonist, CP 96345 (Minami et al., 1997a,b). These findings would be consistent with a role of PGs in facilitating the release of EAAs or SP from PAF terminals but require further mechanistic exploration (vide supra). Interestingly, PGD₂ may play a more complex role in the modulation of nociception inasmuch as, apart from its intrinsic pronociceptive actions, it has been shown to attenuate tactile allodynia under certain conditions, possibly via an interaction with inhibitory glycinergic neurones in the DH (Minami et al., 1997a) (Sections 4.6 and 10.6).

Over the longer term, for example following their induction by NMDA receptor activation, PGs likely act by modifying gene transcription in the DH. In this respect, in line with their more prominent role in the mediation of prolonged nociception (*vide supra*), it has been proposed that COX 2-derived pools of PG are principally involved (Lerea, 1997; Lerea *et al.*, 1997; Smith and De Witt, 1996).

3.2.8.5. Similarities in the actions of NO and PGs

To summarize, there are several intriguing similarities between NO and PGs as concerns their potential mechanisms of action in facilitating nociception in the DH. That is:

- 1. neuronal and non-neuronal sources;
- multiple constitutive and inducible enzyme isoforms;
- 3. generation in DH neurones following stimulation of C fibres, leading to the activation of NMDA receptors and an increase in [Ca²⁺], and
- 4. pronociceptive effects mediated both via intrinsic DH neurones and via PAF terminals.

Further, both NO and PGs modulate gene transcription in sensitized DH neurones. It is possible that the key action of NO is an intracellular role in the sensitization of DH neurones, whereas a receptor-mediated excitation of DH neurones—and PAF terminals—may be of particular importance for PGs. Thus, although the precise significance of spinal pools of NO and PGs, and their mechanisms of action, remain to be elucidated, they likely fulfil important (and interactive) roles in DH-integrated processes of sensitization underlying painful states. Further, as discussed in Sections 10.6.2 and 13.5, both NO and PGs fulfil important—but conflicting—roles in modulating degenerative processes triggered by CNS injury.

4. SPINAL TARGETS OF PRIMARY AFFERENT NOCICEPTIVE INFORMATION: INTRINSIC ELEMENTS IN THE DH

4.1. Laminae Organization of the DH

The grey matter of the spinal cord may be divided on a cytoarchitectonic basis into 10 laminae (Besson and Chaouch, 1987; Dubner and Bennett, 1983; Rexed, 1952; Willis and Coggeshall, 1991) (Fig. 1). Of these, laminae I (marginal layer), II (substantia gelatinosa), III and IV (nucleus propius) and V and VI (deep layers) comprise the DH. Lamina VII corresponds to the intermediate grey matter, laminae VIII and IX comprise the medial and lateral ventral horn (VH), respectively, while lamina X is the region surrounding the canal. (It should be noted that lamina VI is clearly defined only in the lumbosacral and cervical enlargements.) The two most superficial lamina, I and II₀ (the outer part of lamina II), together with deeper laminae, V and VI, and lamina X, constitute those regions predominantly implicated in the reception, processing and rostral transmission of nociceptive information. Each DH lamina is organized in the horizontal plane to topographically encode the body surface. Individual cutaneous PAFs have, thus, clearly-defined, central targets and PAF input is also organized dorso-ventrally (Molander and Grant, 1986; Swett and Woolf, 1985).

Cutaneous nocisponsive C fibres project heavily to II₀ and, less intensely, to lamina I: they also provide a comparatively weak input to lamina V and, probably, lamina X. On the other hand, high threshold, cutaneous, nocisponsive A δ fibres terminate predominantly in lamina I, and to a sparse extent in II₀: they also provide an input to laminae X. Unmyelinated, nocisponsive afferents from the viscera, joints and muscle project primarily to laminae I and V/VI. Unmyelinated, visceral afferents also innervate lamina X. There is a significant degree of convergence between input from various tissues such that individual DH neurones may be targetted by PAFs from different sources: for example, muscles and skin, or viscera and skin. As detailed in Section 9.1.4, this somatovisceral convergence onto individual DH neurones provides an anatomical basis for the phenomenon of 'referred' pain. In contrast to small calibre, nocisponsive fibres, larger, low threshold $A\beta$ fibres mediating non-nociceptive information arborize profusely in laminae III-IV, less markedly in laminae V/VI, possibly to a limited extent in laminae II_i I, and do not innervate lamina II₀.

At the cerebral level, the medullary caudalis nucleus of the trigeminal system ('trigeminal nucleus') fulfils a role similar to that of the spinal DH. It receives nociceptive information via the trigeminal nerve, of which the Vth cranial nerve transmits nociceptive input from the cerebral vasculature. The dilation (and inflammation) of subdural, Cerebral blood vessels (CBV)s activates the trigeminal nerve, thereby triggering the pain of migraine headache (Section 7.9).

4.2. Response Characteristics of DH Neurones

Within the DH, three basic types of neurone may be identified as concerns the nature of their response to nociceptive input (Cervero, 1995b; Dubner and Bennett, 1983: Mense, 1993: Schaible and Grubb, 1993; Willis and Coggeshall, 1991) (Figs 1 and 2). First, typically-silent, nociceptive-specific (NS) neurones which are activated exclusively by high intensity, noxious stimuli mediated by C and A δ fibres. They are most concentrated in laminae I (and II_0), although they are also found in deeper laminae (V and VI). The ability of these neurones to encode stimulus intensity is limited. Secondly, multireceptorial or wide-dynamic range (WDR) neurones which manifest considerable convergence from cutaneous, muscle and visceral input (Gebhart, 1995; Mense, 1986, 1993; Ness and Gebhart, 1990). They are found predominantly in laminae V, as well as in laminae IV and VI, though they are also encountered in laminae I and II₀, as well as in lamina X and the VH. 'WDR' refers to the fact that they produce a dynamic response over a broad stimulus range: that is, they manifest an incremental, stimulus-response relationship from innocuous to noxious stimulus intensities. Correspondingly, WDR neurones are the major type of DH neurone encoding stimulus intensity (Dubner et al., 1989). WDR neurones are excited by thermal, mechanical and chemical stimuli mediated via both C and A δ - as well as $A\beta$ -fibres. WDR neurones in deeper laminae are most clearly implicated in C fibre-mediated processes of sensitization and amplification contributing to prolonged pain (Section 10.3). The third class of neurone is non-nociceptive (NON-N). They are found primarily in laminae II, III and IV, but a few may also occur in lamina I.

As concerns the marginal layer (I), recent studies have detected neurones responding specifically to cooling (cold), as well as polymodal units influenced by heat, pinch and cold (Dostrovsky and Craig, 1996; Zhang and Craig, 1997). These findings indicate that, in parallel to the functional diversity of various classes of PAF described in the previous paragraph (Section 3.1.1), the encoding properties of neurones in the DH may be more complex than accounted for by a simplistic NS, WDR and NON-N classification (Morgan, 1998).

4.3. Receptor Fields of DH Neurones

Within the DH, all neurones possess a RF (Willis and Coggeshall, 1991). For each neurone, this corresponds to the region of skin from which stimulation increases its activity above resting levels. That is, the collective stimulation of nocisponsive PAFs within the RF region centrally summates to elicit action potentials (AP). Surrounding the RF region is a subliminal fringe from which stimulation may affect the neurone, for example, evoking EPSPs, but not sufficiently to provoke firing since a lesser, overall PAF input is recruited. For deep WDR neurones, even mild, mechanical stimuli at the epicentre of their RFs may yield APs whereas, at the periphery, only noxious stimuli trigger a response (Price *et al.*, 1978). In accordance with defined patterns of PAF







Physiological: Aß (Touch), C (Pain)

	5 ↓	nit respons	se	 Col 	↑
Fibre	SN	N-noN	WDR	Touch	Pain
U	‡	0	+	0	‡ ‡
Aß	0	+	0	+	0

	5	aciindeai 1	ţ		
ore	NS	N-noN	WDR	Touch	Pain
	‡ ‡	0	ŧ	0	*****
	‡	+	‡	*(+)	‡ ‡ ‡

* Stimulus-response overshadowing likely submerges touch in a global, painful response (allodynia).

- Descending inhibition and A descending facilitation
- 2. Activation of injured PAFs by inflammatory substances

Aß stimulation of ININs

Fig. 2. Transmission and integration of nociceptive C ($A\delta$), as compared to innocuous $A\beta$, fibre input to the DH, and its modification following peripheral nerve injury. C and $A\delta$ fibres project predominantly to laminae Π_0 and I, respectively. Abbreviations are as indicated in the general list. In the upper panel, the 'normal', resting situation is indicated and, in the lower panel, changes triggered by PAF injury at sensitized, nociceptive DH neurones, are illustrated. Following injury, both the DRG and the neuroma become the origin of spontaneous and stimulated discharges (Sections 8 and 11). The 'dark' ININs do *not* necessarily degenerate, but their functional inhibition of PNs is reduced by mechanisms described in the text (Section 10.6). There are three, additional changes which may also contribute to hyperalgesia and allodynia, but which are not illustrated for reasons of clarity: 1, a shift in the functional balance of descending controls from the inhibitory towards the excitatory; 2, an excitation of the neuroma and DRG of damaged nerves by cytokines, PGs and other inflammatory, pronociceptive substances; and 3, a loss of A β fibre stimulation of ININs. input to specific DH laminae, neuronal RFs in the DH are organized into a somatotropic map whereby specific, but overlapping, populations of neurones represent specific areas of the skin (Schmidt et al., 1997; Wang et al., 1997c; Willis and Coggeshall, 1991). Tissue injury may be associated with an (at least transient) perturbation and expansion of the RFs of DH neurones which reflects not only an increase in peripheral input but also their sensitization by processes described in Section 10. Similarly, peripheral nerve injury can lead to the disruption, expansion and partial reorganization of the somatotropic RFs of neurones in the DH-as well as, the thalamus, cortex and other supraspinal centres (Darian-Smith and Gilbert, 1995; Koerber, 1996; Pettit and Schwark, 1996; Wilson and Kitchener, 1996; Zhang and Rowe, 1997) (Sections 10 and 12). Although the underlying mechanisms are unclear, such changes probably involve multiple processes including abnormal modes of PAF input, sensitization of DH (and supraspinal) neurones, an interference with local, inhibitory circuits in the DH and cerebral tissue and inappropriate patterns of DH innervation by regenerating, injured PAFs (Brewer and Ray, 1995; Lewin et al., 1994a) (Sections 8 and 12). Somatotropic shifts in RFs may be pertinent to the phenomenon of referred pain (Section 9.1.4) and emphasize the importance of central processes in the response to peripheral tissue and nerve damage, a theme reiterated throughout this article (Wilson and Kitchener, 1996).

Although PAF input to the DH is predominantly ipsilateral, a small population of PAFs travel to the contralateral DH (Culberson *et al.*, 1979; Dumoulin *et al.*, 1996; Light and Perl, 1979; Sugimoto *et al.*, 1987; Sugiura *et al.*, 1986; Willis and Coggeshall, 1991). This contralateral input may also be pertinent to the phenomena of 'referred' pain and to certain *bilateral* neurochemical, morphological and functional changes provoked by *unilateral* manipulation of peripheral tissue and nerves (Section 9.1.4).

4.4. Output Targets of DH Neurones

The above-discussed categories of WDS, NS and NON-N units provides one system for the classification of intrinsic DH neurones responsive to primary afferent, sensory information (Section 4.2). They may also be classed in terms of their output destination (Willis and Coggeshall, 1991). Supraspinal, that is, projection neurones (PNs); propriospinal intersegmental (and contralateral) neurones; interlaminar, intrasegmental interneurones (INs) and intralaminar, intrasegmental (local) INs (Biella et al., 1997; Willis and Coggeshall, 1991). Propriospinal neurones likely play an integrative role within the spinal cord in communicating between various segments and between the ipsilateral and contralateral DH. In addition, they may be involved in initiating (and mediating) descending mechanisms of inhibition in the DH following activation by heterosegmental, noxious stimuli (Sandkühler, 1996). They may also participate in reverbatory, excited circuits implicated in DH processes of neuronal sensitization and referred pain (Sections 9.1.4.2 and 10.6.2). Nevertheless, the present discussion focuses on the differential roles of PNs and INs.

In each case, these may comprise WDR, NS or NON-N neuronal types. Further, both PNs and INs are activated by nociceptive PAF input (Coggeshall and Carlton, 1997; Willis and Coggeshall, 1991). PNs transmitting nociceptive information supraspinally are predominantly found in laminae I and V/VI, although a few may be encountered in lamina II and X. By definition, only PNs directly transfer nociceptive information to supraspinal centres. In contrast, INs (particularly the population in lamina II) are involved in the inter- and intralaminar transfer, integration and modulation of PAF information. In this regard, INs may be sub-divided into excitatory (EXIN) and inhibitory (ININ) types. Although this subdivision is useful, it should not be forgotten that certain classes of IN interacting with intrinsic DH neurones and/or PAF terminals may be able to exert both inhibitory and/or excitatory actions, likely mediated by different receptor types.

4.5. PNs and EXINs in the DH

Although PNs may be directly (monosynaptically) activated by C, A δ and A β fibres, EXINs play a role in their indirect (polysynaptic) activation by PAF input (Fig. 2). EXINs may also mediate feedback, excitatory actions on PAF terminals. As concerns NS (and WDR) PNs in superficial laminae, many of these are *directly* targetted by $A\delta$ (I) and C fibres (II₀). EXINs in lamina II₀, which target NS PNs in lamina I, are also involved in their activation (Coggeshall and Carlton, 1997; Dubner and Bennett, 1983; Maxwell and Rethelyi, 1987; Price et al., 1978; Ruda et al., 1986; Willis and Coggeshall, 1991). Since A β fibres do not markedly innervate laminae I and II₀, it is reasonable to suppose that EXINs contribute to the influence of $A\beta$ fibres on those few WDR and NON-N neurones located therein. As regards deeper laminae, some WDR and NS neurones in lamina V receive a direct input from nocisponsive C fibres attaining this region (De Koninck et al., 1992). Further, some of these deeper PNs possess dendrites which project up to superficial laminae allowing for their monosynaptic activation. However, it is widely assumed that many PNs in deep laminae are indirectly (polysynaptically) activated by C fibres via intervening EXINs in lamina II₀ (Coggeshall and Carlton, 1997; Dubner and Bennett, 1983; Light and Kavookjian, 1988; Todd et al., 1994b) or via processes of 'volume transmission' (Section 4.7). On the other hand, WDR and NON-N neurones of deeper laminae are directly innervated by $A\beta$ fibres. Thus, although C and $A\beta$ fibres directly target certain populations of PNs, a role of EXINs is also of importance in relaying their influence thereon.

While the transmitters involved in mediating the actions of EXINs have not been fully characterized, EAAs are likely involved. Further, many neuropeptides, such as neurotensin, VIP and SP, which have been detected in DH-localized INs, could be implicated (Coggeshall and Carlton, 1997; Dubner and Ruda, 1992; Paleckova *et al.*, 1992; Ruda *et al.*, 1986; Todd and Spike, 1993; Todd *et al.*, 1994b;

Willis and Coggeshall, 1991; Yajiri *et al.*, 1997; Yoshimura and Nishi, 1992).

Although they should not necessarily be classified as EXINs, DH-localized INs contain the pronociceptive peptide, neuropeptide FF, which counters the antinociceptive properties of opioids via actions expressed on PNs and, possibly, PAF terminals (Carlton and Coggeshall, 1997; Gouardères et al., 1996, 1997). Although no changes have been detected in the levels of neuropeptide FF in the cerebrospinal fluid (CSF) of chronic pain patients (Sundblom et al., 1997), DH concentrations of neuropeptide FF are increased under conditions of peripheral inflammation in rats (Gouardères et al., 1996, 1997; Kontinen et al., 1997). Similarly, a population of INs containing the pronociceptive peptide, CCK-which is likewise contained in descending pathways and, under certain conditions, PAFs (Sections 3.2.2, 8.2.5 and 10.8)-should be mentioned (Stanfa et al., 1994; Willis and Coggeshall, 1991). In rats, an action of CCK at CCK_B receptors on PAFs and/or PNs likewise counteracts opioidergic- and adrenergic-mechanisms of antinociception although, in primates and man, CCK_A receptors may be involved in these actions of CCK (Benedetti, 1997). CCK may act by an interference with the ability of μ -opioids and adrenergic agonists to decrease Ca^{2+} -currents and $[Ca^{2+}]_i$, thereby preventing their inhibitory influence upon neuronal excitability (Vanderah et al., 1996a,b; Wang et al., 1992). A decrease and increase in the activity of spinal pools of CCK may account for the enhancement and attenuation of μ -opioidergic antinociception under conditions of inflammatory and neuropathic pain, respectively (Benedetti, 1997; Stanfa et al., 1994; Vanderah et al., 1996a; Xu et al., 1993a,b, 1994; Xu and Wiesenfeld-Hallin, 1997) (Section 8.2.5). The pathophysiological significance of neuropeptides FF, CCK and other DH-localized, pronociceptive mediators would be of interest to explore further (Watkins et al., 1997b). Further, the relative importance of PAF input and/or descending modulatory mechanisms (Section 5) in their engagement would be of interest to establish.

EXINs also represent a potential source of ATP, NO and PGs, the excitatory, pronociceptive actions of which are described in Section 3.2.8.

4.6. ININs in the DH

4.6.1. Actions and Neurochemical Classes of ININ

As discussed above, EXINs and PNs form a network (in series and activated by $A\delta$ and C fibres) underlying transfer of nociceptive information from PAFs to PNs and, subsequently, to supraspinal centres. ININs, by contrast, play an important modulatory role in limiting the flow of nociceptive information (Fig. 2). Although certain classes of ININ may attenuate the activity of EXINs, little direct information is available in this regard (Dubner and Ruda, 1992). Indeed, most ININs likely regulate nociceptive transmission by directly targeting WDR and NS PNs and/or the terminals of nocisponsive (and non-nocisponsive) PAFs (Hammond, 1997; Malcangio and Bowery, 1996b; Rudomin, 1990; Schmidt, 1971; Todd and Spike, 1993; Todd *et al.*, 1996; Willis and Coggeshall, 1991). This organization offers, thus, *pre-* and *post*synaptic mechanisms for the inhibition of nociceptive input from PAFs.

Many different, neurochemical classes of ININ have been identified and these display various patterns of neurotransmitter/neuropeptide co-localization (Coggeshall and Carlton, 1997; Dray and Urban, 1996; Hammond, 1997; Ruda *et al.*, 1986; Todd and Spike, 1993; Willis and Coggeshall, 1991). The following are of particular note inasmuch as they can exert an antinociceptive role.

- 1. Cholinergic ININs, which act via multiple muscarinic and—probably—nicotinic receptors localized on PAF terminals and intrinsic DH neurones (Coggeshall and Carlton, 1997; Damaj *et al.*, 1994; Höglund and Baghdoyan, 1997; Ribeiro-Da-Sylva and Cuello, 1990; Sheardown *et al.*, 1997; Travagli, 1996).
- 2. Opioidergic ININs containing enkephalins and/ or dynorphin (DYN), which exert their actions via μ -, δ - and/or κ -opioid receptors (Coggeshall and Carlton, 1997; Fields and Basbaum, 1994; Millan, 1986, 1990; Ossipov *et al.*, 1997; Todd and Spike, 1993).
- 3 Gamma-amino-butyric acid (GABA)ergic ININS, which act via GABAA and GABAB receptors. GABAA receptors are composed of four different sub-units organized as a pentamer to form the wall of a chloride-permeable ion channel, the activation of which usually hyperpolarizes cells. Many (>10) heteromeric isoforms of GABA_A receptor occur in the CNS. GABA_B receptors are G protein-coupled. Their stimulation inhibits AC and decreases and increases Ca^{2+} - and K^+ -currents, respectively, thereby hyperpolarizing neurones and decreasing transmitter release. Multiple GABA_B subtypes may also exist (Ben-Ari et al., 1997; Dirig and Yaksh, 1995; Diversé-Pierluissi et al., 1997; Hammond, 1997; Kaneko and Hammond, 1997; Kaupmann et al., 1997; Nadeson et al., 1996; Teoh et al., 1996; Valeyev et al., 1996). GABAergic ININs display a varied pattern of co-localization with several other neurotransmitters involved in the modulation of nociception. Thus, one subpopulation is co-localized with acetylcholine, another with enkephalin and an additional subpopulation is co-localized with a further, major class of inhibitory neurotransmitter, glycine, which exerts its actions via strychnine-sensitize glycinergic (glycine_A) receptors. Glycinergic receptors are composed of two units arranged to form the pentameric wall of a chloride-permeable ion channel, the activation of which hyperpolarizes neurones. They are predominantly postsynaptic to PAFs and several isoforms/splice variants may exist (Hammond, 1997; Kemp et al., 1996; Sorkin and Puig, 1996; Todd and Spike, 1993; Todd et al., 1996). The co-localization of GABA with NO synthase in a sub-population of neurons in lamina II is of note in light of the implication of NO in the modulation (primarily facilitation) of spinal nociception (Valtschanoff et al., 1992) (Sections 3.2.8.3 and 10.4.2.2). This observation

illustrates the difficulty of ascribing an exclusively 'pro'- or 'anti'-nociceptive role to specific classes of DH neurone. Further, various classes of (nonglycinergic) GABAergic ININ may contain a diversity of neuropeptides: these include GAL/ NPY and GAL/enkephalin subpopulations (Munglani *et al.*, 1996; Simmons *et al.*, 1995; Todd and Spike, 1993). Like GABA, GAL and NPY fulfil principally antinociceptive roles in the DH (Munglani *et al.*, 1996; Xu and Wiesenfeld-Hallin, 1997) (Section 3.2.6).

ININS are themselves targetted by nocisponsive C and A δ fibres, as well as by A β fibres (Bernardi *et al.*, 1995; Coggeshall and Carlton, 1997; Hayes and Carlton, 1992; Todd and Spike, 1993; Todd *et al.*, 1994a; Yoshimura and Nishi, 1992, 1995; Willis and Coggeshall, 1991). This suggests that C and A δ fibres exert a form of 'counter-regulatory', inhibitory feedback control upon their parallel excitation of WDR and NS PNs. Consequently, the pain elicited upon direct PN stimulation by C and A δ fibres may be limited by simultaneous activation of ININs, at least when noxious stimulation is of *short* duration and sub-maximal intensity (Fig. 2).

4.6.2. $A\beta$ Fibre Activation of ININs: the Absence of Pain Upon Selective $A\beta$ Fibre Stimulation

The existence of ININs also helps provide an explanation as to why $A\beta$ fibre stimulation is *not* normally perceived as noxious.

- 1. Like C fibres, $A\beta$ fibres also activate a sub-population of ININs targeting WDR PNs, thereby indirectly interfering with their direct excitation of the latter. However, in *contrast* to C fibres, the activation of $A\beta$ fibres does *not* excite *NS* neurones in parallel, but rather *NON-N* neurones. Thus, their global influence upon DH output is qualitatively different to that of C fibres (Fig. 2).
- 2. The influence of selective $A\beta$ fibre stimulation upon PNs differs to that of C fibre activation. Fast synaptic actions triggered by $A\beta$ fibres are probably mediated by AMPA receptors which primarily gate a Na⁺-conductance, and rapidly desensitize. Selective activation of $A\beta$ fibres will not, thus, normally lead to a protracted excitation of WDR neurones via Ca2+-permeable NMDA receptors, and NK/CGRP/group I mGlu receptors positively coupled to $AC/PLC/[Ca^{2+}]_i$ are quiescent (Fletcher and Lodge, 1996) (Sections 3.2.2 and 10.3). Thus, even though AMPA receptor activation engages voltagedependent calcium channels (VDCCs), increases in $[Ca^{2+}]$, remain modest and sensitization mechanisms are not initiated.
- 3. NMDA and AMPA receptors are, to a certain extent, expressed on different classes of neurone in the DH. Further, even where colocalized, there is increasing evidence for compartmentalization of changes in the intracellular concentrations of [Ca²⁺]_i and other intracellular signals (Bito *et al.*, 1997; Deisseroth *et al.*, 1998; Kharazia and Weinberg, 1997; Lerea, 1997; Morris, 1997; Ottersen and Landsend, 1997; Simpson *et al.*, 1995). Thus, any limited increase in [Ca²⁺]_i levels provoked via the activation of AMPA receptors

(and VDCCs) may be topographically separated from neuronal regions targetted by C fibre terminals and bearing NMDA receptors (Section 10.3).

To summarize, the intensity and duration of the excitation of WDR PNs neurones by $A\beta$ fibres is less pronounced than that provoked by C fibres, $A\beta$ fibres activate *NON-N* rather than NS PNs in parallel and the discrete engagement of AMPA receptors by $A\beta$ fibres does not engage intracellular processes leading to PN sensitization. Consequently, $A\beta$ and C fibre-mediated sensation is perceived as non-noxious and noxious, respectively, and the *selective* activation of $A\beta$ fibres does not sensitize PNs in the DH (Table 3) (Section 10.5).

4.6.3. Consequences of a Loss of ININ Tone in the DH

When the crucial, inhibitory role of ININs in the DH fails—under conditions of nerve injury or pharmacological blockade, for example—the consequences for both $A\delta/C$ and $A\beta$ fibre-mediated transmission are serious: hyperalgesia and allodynia, respectively, due to 'escape' of sensitized WDR and NS PNs from ININ control (Fig. 2) (Dickenson, 1997; Hammond, 1997; Traub, 1997) (Section 10.6). Further, when WDR neurones in the DH are preexcited by a persistent C fibre input, and/or a decrease in ININ tone, the superimposition of $A\beta$ fibre stimulation may be able to elicit their sensitization and aggravate hyperalgesia and allodynia (Section 10.3).

4.7. Novel Modes of Neuronal Communication

The above-described, differential localization in various DH laminae of the majority of C fibre terminals (II₀ and I) as compared to the majority of deeper laminae WDRs has led to the suggestion that, apart from the intervention of EXINs, volume transmission may play a role in the activation of PNs in deeper laminae by fine calibre PAFs conveying nociceptive input to the DH (Coggeshall and Carlton, 1997; Fuxe and Agnati, 1991; Zoli and Agnati, 1996). A role of volume transmission would also be consistent with the overall mismatch between the DH localization of SP in PAF terminals (primarily in lamina II) and NK1 receptors (primarily in laminae I and, less densely, in deeper laminae IV/V/VI) (Abbadie et al., 1996; Bleazard et al., 1994; Coggeshall and Carlton, 1997; De Koninck et al., 1992; Lawson et al., 1997; Littlewood et al., 1995; Liu et al., 1994a; Marshall et al., 1996; Sugimoto et al., 1997; Valtschanoff et al., 1995). Indeed, NKA and SP may spread and persist following their localized release in the DH (Duggan et al., 1990). Volume transmission may, thus, be of significance in the neuropeptide-mediated transfer of nociceptive information from PAF terminals to intrinsic neurones in the DH (Coggeshall and Carlton, 1997) and to other regions of the spinal cord, such as preganglionic neurones in the intermediolateral cell column (Pollock et al., 1997). Nevertheless, recent studies have indicated a certain degree of correspondence between DH patterns of SP input on the one hand, and the dendritic arbors and nociceptive responsiveness of NMDA receptorbearing NS and WDR cells in laminae I and IV/V, on the other. This suggests that, at least for certain neurones, SP may act close to its site of release (King *et al.*, 1997a; Mantyh *et al.*, 1997; Naim *et al.*, 1997; Weiya *et al.*, 1996).

Volume transmission should, thus, be conceived of as complementary to conventional modes of transmission for SP and other neuropeptides released from PAFs (Coggeshall and Carlton, 1997). It is *not* likely to play a major role in fast synaptic transmission mediated by EAAs at ionotropic receptors, although *extrasynaptic* actions of GLU ('spillover') at NMDA and AMPA receptors in other structures have recently attracted attention (Barbour and Häusser, 1997; Ben-Ari *et al.*, 1997; Coggeshall and Carlton, 1997; Kullmann and Asztely, 1998). Indeed, a potential role of non-synaptic, volume transmission should not be regarded as restricted to PAF input inasmuch as:

- GABAergic ININs have been shown to nonsynaptically modify the activity of certain ascending pathways (Sakatani *et al.*, 1993);
- volume transmission may be of significance in mediating the effects of descending monoaminergic fibres (Section 5) upon DH neurones (Millan, 1997; Ridet *et al.*, 1993); and
- 3. opioids may also exert actions in the DH via volume transmission (Coggeshall and Carlton, 1997) (Section 10.6.2.2).

As concerns other non-conventional, putative modes of neuronal communication, a recent morphometric analysis of rat spinal cord suggested the existence of a substantial number of 'mixed' (chemical-electrical) synapses on somata and dendrites: they even appeared to predominate as concerns excitatory synapses (Rash et al., 1996). This obsersuggests *bi*-directional transfer of vation information and, if functionally-active, such 'mixed' synapses would raise fundamental questions as regards the interrelationships amongst various DH elements involving in the modulation of nociception: in particular, as regards the synchronized (bursting) activity of certain populations of sensitized DH neurones underlying painful states (Eblen-Zajjur and Sandkühler, 1997) (Sections 9.1.4.2 and 10.6). Such putative, 'bidirectional' forms of neuronal transmission might, further, be considered as analogous to 'retrograde' modes of synaptic transmission displayed by postsynaptic, DH neuronal pools of EAAs, NO and PGs at presynaptic terminals controlling their release (Sections 3.2.4 and 3.2.8).

'GAP' junctions (membrane channels allowing for direct, interneuronal passage of molecules and ions) may also play a role in the rhythmic neuronal firing engendered by a loss of ININ tone in the DH (Section 10.6). In addition, supraspinally, they may be involved in the transmission of 'cortical spreading depression' which is implicated in the genesis of migraine headaches, in particular the accompanying 'aura'. This process involves both neurones and glial cells in which waves of successive depolarization and hyperpolarization—triggered by 'stress' and ischaemic or other forms of tissue damage—descend the cortex and, subsequently, deeper structures (Nedergaard *et al.*, 1995; Olesen and Janssen-Olesen, 1997; Peinado *et al.*, 1993; Verkhratsky and Kettenmann, 1996) (Section 7.9).

Unconventional modes of communication between damaged PAFs are discussed in Section 8.2.3.

4.8. Glial and Immunecompetent Cells: Focus on Cytokines

4.8.1. Glial Cells

Resident glial cells (astrocytes and oligodendrocytes) have traditionally been considered to perform an essentially homeostatic role in the CNS in offering physical support for neurones, improving synaptic efficacy, preserving tissue integrity upon CNS injury and guaranteeing an appropriate ionic and physical environment for neuronal activity (Murphy, 1993; Pfrieger and Barres, 1997; Ridet *et al.*, 1997; Vernadakis, 1996).

However, glial cells are now known to bilaterally communicate with neurones via both chemical messengers and electrotonic junctions, and they may play a more direct role in modulating neuronal transmission than previously imagined. Changes in glial and neuronal levels of $[Ca^{2+}]_i$ appear to play a key role in this regard (Kaplan *et al.*, 1997; Pasti *et al.*, 1997; Pfrieger and Barres, 1997; Verkhratsky and Kettenmann, 1996; Nedergaard, 1994). Indeed, glial cells:

1. possess cation-permeable ion channels;

- display multiple types of neurotransmitter receptor, including several which modulate nociceptive transmission-adrenergic, serotoninergic, purinergic, GABAergic, AMPA, mGlu and glycinergic;
- 3. play a major role in transmitter uptake—for example, of GABA and GLU; and
- 4. liberate several substances which may modify nociception (Chessel et al., 1997a,b; Coggeshall and Carlton, 1997; Conn and Pin, 1997; Conti et al., 1997; Jalonen et al., 1997; Kimelberg, 1995; Martin, 1990; Mayer et al., 1997; Miller et al., 1997; Smith, 1994; Sontheimer et al., 1996; Steinhäuser and Gallo, 1996; Vernadakis, 1996; Vulchanova et al., 1997; Winder and Conn, 1996) (see below).

Notably, GLU controls gene expression and differentiation in glial cells, while AMPA and mGlu receptors on astrocytes provoke a Ca²⁺-dependent release of GLU and NO (Bezzi et al., 1998; Steinhäuser and Gallo, 1996). Glial cells also respond to SP and additional signals, such as NO itself, by an increase in the activity of COX and the release of PGs, which provokes a further feedback increase in NO production (Bezzi et al., 1998; McCormack, 1994). Both NO and PGs exert pronociceptive actions in the DH, and glial cells are an abundant source of a further pronociceptive mediator, ATP (Section 3.2.8). Consistent with the liberation of NO, PG, ATP and other potential, pronociceptive agents by glial cells, their recruitment in the DH has been implicated in mechanisms underlying inflammatory nociception (Meller *et al.*, 1994; Watkins *et al.*, 1997a).

4.8.2. Microglial Cells

The CNS contains resident, immunecompetent microglial cells, which enter tissue via blood vessels. In addition to microglial cells, following tissue damage or inflammation, immigrant macrophages, T cells and mast cells, also provide a potential source of glutamate NO, PGs, ATP and other potential mediators (Chao et al., 1996; Ebadi et al., 1997; Minghetti et al., 1997; Minghetti and Levi, 1998; Piani et al., 1991). Further, microglial cells also possess several receptor types. Although these are less well characterized than those on glial cells, the presence of multiple receptors for ATP has been reported, including both metabotropic P_{2Y} receptors as well as ionotropic P_{2X} sites (Chessel *et al.*, 1997a,b; Collo *et al.*, 1997; Heese *et al.*, 1997; Mayer *et al.*, 1997; Nörenberg *et al.*, 1997; Vulchanova *et al.*, 1997). Activation of P_{2X7} receptors promotes the formation of large, cytolytic pores which may provoke the release of cell contents (Chessel et al., 1997b).

4.8.3. Actions of Cytokines: Relevance to Nociception

Glial cells, microglial cells and other immunecompetent cells are all abundant sources of cytokines. Via interactions with both neurones and non-neuronal cells, cytokines can potentially modify synaptic transmission and nociception (Connor and Leonard, 1998; Ebadi *et al.*, 1997; Eder, 1997; Elmquist *et al.*, 1997; Gadient and Otten, 1997; Merrill and Benveniste, 1996; Pan *et al.*, 1997; Watkins *et al.*, 1995, 1997a).

For example, interleukin (IL)-6 induces allodynia and hyperalgesia via actions in the DH (De Leo et al., 1996) while IL-1 β enhances the release of SP in the spinal cord (Malcangio et al., 1996) and induces COX 2 expression (Newton et al., 1997). Further, tissue necrosis factor α (TNF_{α}) may facilitate postsynaptic, EAA-dependent ion currents (Grassi et al., 1994). A potential role of cytokines in modulating nociceptive transmission and synaptic plasticity in the DH is supported by studies of other CNS structures. Thus, there is evidence that ILs contribute to cortical mechanisms of sensitization. Indeed, IL-6 has been shown to modify sensory responses in the cortex and to elicit nociception upon intracerebroventricular administration to rats via a mechanism involving PGs (Oka et al., 1995; Shin et al., 1997). On the other hand, central application of IL-1 β elicits antinociception and certain other actions of cytokines may also potentially reduce nociception (Yabuuchi et al., 1997). Thus, IL-1 β enhances GABAergic transmission and suppresses, via a postsynaptic mechanism, the induction of NMDA receptor-mediated processes of amplification in the hippocampus: this phenomenon, termed 'long-term potentiation' (LTP), provides a synaptic substrate for memory formation and presents many parallels to NMDA-mediated, sensitization of DH neurones, a process underlying prolonged painful states (Bliss and Collingridge, 1993; Coogan and O'Connor, 1997; D'Arcangelo *et al.*, 1997; Heyser *et al.*, 1997; Malek-Ahmadi, 1996; Miller *et al.*, 1991; Neumann *et al.*, 1996; Tancredi *et al.*, 1990) (Sections 10.2.1 and 10.3). Levels of IL-1 β and TNF α are increased in the DH following peripheral nerve damage (De Leo *et al.*, 1997). Further, it was recently shown that a further cytokine, leukaemia inhibitory factor (LIF), which plays an important role in the periphery following PAF injury (Sections 8.2.4.1 and 8.2.6.2), is upregulated in superficial DH laminae by inflammation and damage to PAFs (Thompson *et al.*, 1996, 1997).

These observations suggest that cytokines may affect nociception via the modification of NMDA receptor-mediated, and GABA receptor-modulated, processes of sensitization integrated in the DH.

4.8.4. Tissue-Protective and Tissue-Destructive Actions of Cytokines

Cytokines such as LIF, TNF_{α} and $IL-1\beta$, as well as glutamate NO, PGs, reactive oxygen species and other substances released from glia and microglial cells, modulate processes of neuronal survival, recuperation and degeneration. Their complex patterns of cytotoxic and mitogenic/regenerative actions likely involve contrasting, intracellular transduction mechanisms (Chao et al., 1996; Kawamura et al., 1997a,b; Minghetti and Levi, 1998; Nishio and Watanabe, 1998; Pan et al., 1997; Piani et al., 1991). Such mechanisms may be involved in the loss of intrinsic ININs (and other neuronal types) which occurs upon lesions to PAFs or to the CNS itself (Sections 10.6.2 and 13.4). These actions of cytokines are effected in interaction with neurotrophins and other trophic factors, the secretion and binding of which is similarly modified in glial and immunecompetent cells following injury to the CNS or PAFs (Kotzbauer et al., 1996; Liedtke and Edelmann, 1996; Rubio, 1997; Sontheimer et al., 1996). Indeed, an induction of growth factors (neurotrophins) in non-neuronal and neuronal cells may assist in the recovery from CNS injury (Section 13.5), and like cytokines, via the modification of synaptic transmission, neurotrophins may more directly and rapidly participate in the processing of nociceptive informations (Kotzbauer et al., 1996; Liedtke and Edelmann, 1996; Rubio, 1997; Sontheimer et al., 1996) (Section 14).

4.8.5. Activation of Glial and Immunecompetent Cells: a Summary

To summarize, glial and immunecompetent cells in the DH, via the secretion of diverse substances, including GLU, cytokines, neurotrophins, NO, PGs and ATP, and by virtue of their interactions with both neuronal and other non-neuronal elements, likely play an important role in the functional, adaptive and organizational changes underlying pain caused by damage to the CNS itself or to events in the periphery (Vernadakis, 1996). Emphasizing the potential role of such mechanisms in the induction of neuropathic pain, peripheral nerve damage may provoke both glial and microglial hypertrophy in the DH (Kane *et al.*, 1997; Molander *et al.*, 1997). The complex and multifarious roles of glial and immunecompetent cells in the DH (and supraspinal structures) in modulating nociception are still at an early stage of characterization, but are likely to attract increasing attention.

5. MODULATION OF NOCICEPTIVE PROCESSING IN THE DH BY DESCENDING INHIBITORY AND *FACILITATORY* PATHWAYS

5.1. Descending Inhibition

Descending pathways originating in the brainstem and other cerebral structures play an important role in the modulation and integration of nociceptive messages in the DH. Serotoninergic, noradrenergic and, to a lesser extent, dopaminergic networks comprise major components of these descending mechanisms (Fields and Basbaum, 1994; Holstege *et al.*, 1996; Millan, 1995, 1997; Sandkühler, 1996; Willis, 1988; Yezierski *et al.*, 1983; Zhang *et al.*, 1997f). The following general points should be emphasized.

- 1. In line with their assumed antinociceptive role, descending pathways modulate (generally reduce) the release of neurotransmitters from the terminals of nocisponsive PAFs (Fields and Basbaum, 1994; Millan, 1997; Travagli and Williams, 1996). Further, the activation of descending pathways inhibits nocisponsive PNs both directly, and indirectly, via the inhibition of EXINs and the excitation of ININs. These actions of descending pathways at intrinsic DH neurones-'post-synaptic' with respect to PAF terminals—are probably of the greater importance in their modulation of nociception (Fields and Basbaum, 1994; Lin et al., 1994, 1996c,d; Lopez-Garcia and King, 1996; Grudt et al., 1995; Millan, 1997; Peng et al., 1996; Sandkühler, 1996; Willis and Coggeshall, 1991; Yang et al., 1998). Certain studies have indicated that descending pathways preferentially inhibit the excitation of WDR neurones by noxious as compared to innocuous stimuli (Fields and Basbaum, 1994; Millan, 1997; Willis, 1994). There are several possible (and non-exclusive) mechanisms which might allow for such a selective interference with noxious A δ and C fibre vs innocuous $A\beta$ fibre mediated activation of WDR neurones:
 - 1. as indicated above, descending pathways may presynaptically inhibit the release of nociceptive transmitters from $A\delta$ and C, but not $A\beta$, fibres targeting WDR neurones;
 - 2. descending pathways may also indirectly inhibit the activity of $A\delta$ and C vs $A\beta$ fibre terminals via the engagement of an intervening population of specific ININs (for example, GABAergic);
 - 3. descending pathways may inhibit the activity of EXINs selectively targetted by $A\delta$ and C, but not by $A\beta$, fibres; and
 - 4. descending pathways may, at the level of PNs themselves, exert effects upon ionic and soluble second messenger transduction which *selectively* interfere with actions mediated by NMDA and other receptor types coupled to C

fibres. For example, via alterations in intracellular levels of $[Ca^{2+}]_i$, NO or the activity of various PKs (Section 10.4). Further, inasmuch as modifications in $[Ca^{2+}]_i$ and other intracellular signals in DH neurones may be spatially restricted (Kharazia and Weinberg, 1997; Lerea, 1997; Morris, 1997; Ottersen and Landsend, 1997; Simpson *et al.*, 1995), a selective influence of descending fibres upon the excitation of WDR neurones by C (and $A\delta$) vs $A\beta$ fibres would be possible should terminals of C fibres and descending pathways be *contiguous*, yet those of $A\beta$ fibres topographically *separated* (Sections 4.6.2 and 10.5).

- 2. Several electrophysiological studies have suggested that, under certain conditions, the inhibitory influence of descending systems on WDR neurones in deeper laminae can be non-selectively expressed against both noxious (A δ and C fibre) input and innocuous $(A\beta)$ input, in line with a direct, monosynaptic inhibition of PN excitability (Fields and Basbaum, 1994; Lin et al., 1996c,d; Willis, 1994; Willis and Coggeshall, 1991). Such observations have received comparatively scant attention but appear of importance in light of the phenomenon of mechanical, $A\beta$ fibre-mediated allodynia (Section 9.1.2). Indeed, a 'lack of selectivity' might translate to an antinociceptive action of descending pathways against $A\beta$ -fibre mediated allodynia under pathological conditions. Such an inhibition of $A\beta$ fibre-mediated activity could be expressed both presynaptically at their terminals, and postsynaptically at PNs which they target. In this regard, it is of interest that the DRG of both fine calibre $(A\delta/C)$ and *large* calibre $(A\beta)$ PAFs express mRNA encoding α_{2A} -adrenoreceptors (ARs): that is, the predominant α_2 -AR subtype mediating the antinociceptive actions of descending adrenergic pathways in the DH (Gold et al., 1997; Hunter et al., 1997a; Millan, 1997, 1998; Millan et al., 1994).
- 3. Much behavioural evidence for descending inhibition (and facilitation) is based upon reflexive, algesiometric tests involving a motor response. The underlying reflex, motor arc is *not* necessarily modulated by DH mechanisms in an identical fashion to nociceptive information ascending to cerebral centres. For example, serotonin (5-HT) may suppress reflexes triggered by noxious stimuli *without* modifying passage of this nociception information to the supraspinal level (Borszcz *et al.*, 1996; Millan, 1997; Zemlan *et al.*, 1983).
- 4. Although monoamines are generally considered to be the major neurotransmitters released from descending pathways, a key role of 5-HT in mediating descending inhibition has been challenged (Gao et al., 1997; Millan, 1995, 1997) and centrifugal networks contain a diversity of other neurotransmitters potentially involved in the modulation of nociception. Several of these may be colocalized with serotoninergic and noradrenergic pathways. In this light, acetylcholine, enkephalin and GABA, all of which are found in intrinsic ININs in the DH, deserve special mention (Section 4.6) Interestingly, evidence for

cholinergic and, possibly, GABAergic mechanisms of descending inhibition has been presented (Antal et al., 1996; Arvidsson et al., 1995; Jones et al., 1991b; Iljima et al., 1992; Maxwell et al., 1996; Millan, 1997; Peng et al., 1996) (Section 4.6). These observations suggest that a component of the antinociception mediated by descending pathways may be exerted via mechanisms common to ININ-mediated antinociception in the DH (Antal et al., 1996; Blomqvist et al., 1994; Bouaziz et al., 1996; Lin et al., 1994, 1996c Lin et al., 1996d; Millan, 1997; Sorkin et al., 1993). Further, many neuropeptides found in PAF terminals, such as CCK and SP, are also contained in descending pathways-although primarily in the VH (Fang and Proudfit, 1996; Millan, 1997; Nicholas et al., 1992; Urban et al., 1996a,b; Wu and Wessendorf, 1992; Zhuo and Gebhart, 1990; Xu and Wiesenfeld-Hallin, 1997) (Sections 3.2.2 and 3.2.6).

The existence of multiple DH pools of neurotransmitters and neuropeptides derived from different neuronal sources complicates studies of their modulation and—likely diverse-functional roles. Moreover, the above observations underpin the assertion that descending mechanisms for the modulation of nociception in the DH are by no means restricted to a 'textbook' role, whereby 5-HT and noradrenaline (NAD) reduce the release of SP or other pronociceptive transmitters from small calibre, PAF terminals.

5.2. Descending Facilitation

Descending pathways do not exclusively exert inhibitory actions in the DH. Indeed, individual transmitters may exert multiple actions in the DH as a function of the type of neurone targetted (ININ vs PN, for example) and the receptor activated (inhibitory vs excitatory serotoninergic receptors, for example) (see below). Moreover, descending inhibitory and *facilitory* pathways to the DH may even be derived from the same structure. For example, descending inhibitory and facilitatory pathways run from the medullary nucleus reticularis gigantocellularis pars alpha to WDR laminae IV/V and NS laminae I neurones, respectively (Monhemius et al., 1997). Interestingly, spinal channels conveying descending facilitatory as compared to inhibitory influences to the DH from the rostroventromedial medulla may be topographically separated. (Zhuo and Gebhart, 1997). (The differential roles of various spinal channels in transmitting ascending sensory information to higher centres is outlined in Section 6.1.3) In addition, other cerebral regions, including the cortex, may be the origin of excitatory (and inhibitory) influences to the DH (Yezierski et al., 1983; Zhang et al., 1996a). There is evidence that descending facilitatory mechanisms exert excitatory actions both on the terminals of nociponsive PAFs, as well as on intrinsic DH neurones (Almeida et al., 1996; Bardin et al., 1997; Grudt et al., 1995; Jones, 1992; Jordan et al., 1979; Millan, 1995, 1997; Urban et al., 1996a,b; Urban and Gebhart, 1997; Zhuo and Gebhart, 1997). Some of these intrinsic neurones may be ININs, but others are likely EXINs and PNs themselves. In the latter case, their activation will result in an *enhancement* of nociceptive transmission.

The roles of various DH-localized transmitters in mediating descending facilitation is still under exploration, although an involvement of DH pools of EAAs, NO and CCK is probable (Nicholas et al., 1992; Sorkin et al., 1993; Urban et al., 1996a,b; Wiertelak et al., 1994b) (Section 10.8). As concerns monoamines, there is evidence that they exert inhibitory or excitatory actions on individual DH neurones via distinct receptor subtypes differentially coupled to intracellular transduction mechanisms. For example, 5-HT_{1A} receptors hyperpolarize cells by opening K⁺-channels, whereas 5-HT_2 and 5-HT_3 receptors depolarize neurones by closing K⁺-channels and/or opening cation-permeable ion channels/ VDCCs. Further, 5-HT_{1A} receptors are coupled negatively to AC, whereas 5-HT_{2A} and 5-HT₃ receptors activate PLC (Boess and Martin, 1994; Millan, 1995) (Section 10.8). Correspondingly, where colocalized on individual neurones, 5-HT_{1A} as compared to 5-HT₂/5-HT₃ receptors will oppositely modulate nociception in the DH. Further, the activation of 5-HT1A and 5-HT2/5-HT3 receptors on ININs and PNs, respectively, may result in a facilitation of nociception (Millan, 1995, 1997). Dopaminergic D_1 and D_2 receptors, which stimulate and inhibit AC activity, respectively, may also enhance and inhibit nociception, respectively, by actions on PNs (Millan, 1997). Finally, PN-localized excitatory α_1 -ARs (positively coupled to PLC) and inhibitory α_2 -ARs (negatively coupled to AC) may also, respectively, increase and decrease nociception (Millan, 1997). Such differential roles of multiple monoaminergic receptor types would clarify many contradictatory and confusing data in the literature concerning the role of 5-HT, NAD and dopamine (DA) in the DH (Millan, 1997) (Section 10.8).

A potentiation and attenuation of descending facilitation and inhibition, respectively, may contribute to the enhancement in the sensitivity of DH neurones triggered by PAF inflammation or injury (Section 10.8). Correspondingly, the selective *inacti*vation of descending facilitatory pathways might offer a strategy for analgesia complementary to the traditional approach of *mimicking* the activity of descending inhibitors pathways (Section 15.3).

6. TRANSFER OF NOCICEPTIVE INFORMATION TO SUPRASPINAL CENTRES: DISCRIMINATIVE-SENSORY AND AFFECTIVE-COGNITIVE DIMENSIONS OF PAIN

6.1. Ascending Nociceptive Pathways: Organization

6.1.1. Ascending Pathways: Neurotransmitters and their Roles

Following integration in the DH, nociceptive information is conducted via PNs to higher centres in the brain (Almeida and Lima, 1997; Berkley and Hubscher, 1995; Bernard *et al.*, 1995; Besson and Chaouch, 1987; Besson *et al.*, 1995; Burstein *et al.*, 1990; Giesler et al., 1994; Guilbaud et al., 1994; Willis, 1985, 1989; Willis and Coggeshall, 1991) (Table 4). Many neurotransmitters and modulators involved in the rostral transmission of nociceptive information likely remain to be discovered. However, electrophysiological studies suggest that GLU and other EAAs, via actions at both ionotropic and mGlu receptors, are involved in the transfer of nociceptive information from the spinothalamic tract (STT) to the thalamus, and from the spinomesencephalic tract to the periaqueductal grey (PAG) (Azkue et al., 1997; Eaton and Salt, 1995; Ericsson et al., 1995; Jensen and Yaksh, 1992; Salt and Eaton, 1995, 1996). Further, AMPA receptors have been specifially implicated in the input of postsynaptic dorsal column (PSDC) pathways to the gracile and cuneate nuclei (Popratiloff et al., 1997) (Section 6.2.3).

The presence of several excitatory neuropeptides, including SP and CCK, in supraspinal projections suggests that they may also play a role. Further, the discovery of DYN in ascending pathways is intriguing since it acts principally as an inhibitory transmitter via activation of k-opioid receptors (Li et al., 1996; Li et al., 1997b; Millan, 1986, 1990, 1993; Ossipov et al., 1997; Willis and Coggeshall, 1991). It is conceivable that DYN inhibits ININs in supraspinal centres such as the thalamus, thereby disinhibiting nocisponsive neurones and contributing to the onward flow of nociceptive information. Further, as discussed in Section 10.6.2.2, there is evidence that an allosteric interaction of DYN at (the glycinergic site of) NMDA receptors enhances their activity, offering a further mechanism whereby DYN might facilitate the passage of nociceptive information in ascending pathways. However, the possibility that DYN (or other inhibitory transmitters in ascending pathways) may negatively modulate the passage of nociceptive information to higher centres might be raised. In this light, it is of interest that evidence for a spino-supraspinal 'antinociceptive' pathway has been presented: this was suggested to tonically stimulate a cerebral, opioidergic mechanism facilitating descending inhibition (Gear and Levine, 1995).

Thus, in analogy to descending inhibition and facilitation, there may exist a component of ascending antinociception as well as nociception. This provides a parallel to the existence of both excitatory (pronociceptive) as well as inhibitory (antinociceptive) transmitters in PAFs innervating the DH (Section 3.2.6). Nevertheless, the overwhelming function of ascending pathways is the transmission of nociceptive information from the skin, viscera and other organs to the brain. Therein, specific relay neurones further process nociceptive information and undertake its transfer to cortical structures which, in interaction with limbic circuits, are responsible for the conscious appreciation of pain.

6.1.2. Ascending Pathways: Supraspinal Targets of Nociceptive Information

The neuroanatomy and organisation of ascending pain projection pathways is highly complex (Almeida and Lima, 1997; Berkley and Hubscher, 1995; Bernard *et al.*, 1995; Besson and Chaouch, 1987; Besson *et al.*, 1995; Bullitt, 1990; Burstein *et al.*, 1990; Giesler *et al.*, 1994; Guilbaud *et al.*, 1994; Lanteri-Minet *et al.*, 1994; Willis, 1985, 1989; Willis and Coggeshall, 1991). Table 4 summarizes several of their key characteristics. Neuroanatomicallyspeaking, two types of ascending pathways may be recognized. These may conveniently be termed 'monosynaptic' and 'polysynaptic', respectively.

'Monosynaptic' pathways project directly to higher, cerebral structures and include the STT, the spinomesencephalic tract, the spinoparabrachial tracts, the spinohypothalamic tract and the spinoreticular tract. One component of the spinoreticular tract-of limited relevance herein-projects to the lateral reticular formation, a precerebellar nucleus involved in motor control. The other component projects to the medial, pontomedullary reticular formation and, therefrom, to thalamocortical circuits. The spinomesencephalic pathway is considered to possess a 'spinoannular tract' running to the PAG and a 'spinotectal' tract projecting mainly to the deep superior colliculus (Keay et al., 1997; Willis and Coggeshall, 1991). A major target of the spinomesencephalic tract is the parabrachial nucleus (PBN) of the pons, a region which plays an important role in integrating the cardiovascular, autonomic and motivational response to pain (Allen and Pronych, 1997; Bernard et al., 1995) (Section 6.3.5). Two specific channels running to the PBN are termed the spinoparabrachiohypothalamic and spinoparabrachioamygdaloid pathways, respectively, in accordance with the subsequent projections of two subpopulations of PBN neurones (Bernard and Besson, 1990; Bernard et al., 1990, 1993, 1994, 1995: Bester et al., 1997; Burstein and Potrebic, 1993; Jasmin et al., 1997; Menendez et al., 1996). The PBN can, thus, be accessed by several ascending pathways and the direct and/or indirect innervation of specific cerebral nuclei by multiple ascending pathways is a general rule exemplified in Table 4. Further, single neurones in the PBN may provide axons to both the thalamus and spinal cord emphasizing its integrative role in sensory transmission (Li and Mizuno, 1997; Yoshida et al., 1997). The amygdala, globus pallidus/putamen (striatum of rats and lenticular nucleus of man), nucleus accumbens and septum, as well as the frontal, orbital (cingulate) and infralimbic cortex may also be directly accessed by spinal nociceptive neurones (Burstein and Potrebic, 1993; Giesler et al., 1994; Jasmin et al., 1997; Newman et al., 1996; Willis and Coggeshall, 1991) (Section 6.3.5). Certain limbic targets of ascending nociceptive information are closely interlinked with the hypothalamus and have been referred to as a 'spinotelencephalic' projection: this appears to represent a rostral and bilateral extension of the spinohypothalamic tract, the course of which is outlined below (Section 6.1.3). Further, other collaterals from the spinohypothalamic pathway project to the thalamus (including its ventroposterior and posterior aspects) and also innervate the medulla and pons of the brainstem, sites of origin of descending, modulatory pathways (Fields and Basbaum, 1994; Giesler et al., 1994; Kostarczyk et al., 1997; Millan, 1997) (Section 5).

	Louinoo					Comotio		Dhulo constis	
Tract	of origin	Cell types	Tissue input	Ascending pathways	Principal sub-cortical targets	Organization	Axon types	distribution	Possible roles
Spinothalamic tract	I II (few) IV V/VI VII/VIII LSN	NS WDR Non-N (few)	Skin Viscera Joints/muscle	Mainly VLF DLF (I, LSN) Mainly contralateral	Thalamus: VLF → VPL/VPM DLF → VMP₀/VPI/MDvc Also PAG and collaterals →Reticular structures	Yes (I–IV)	Unmyelinated Small and large myelinated	All mammals Prominent in primates	Discriminative-sensory (VLF) Motivational-affective (DLF) Descending inhibition
Spinoreticular tract	I V/VI VII/VIII X (few)	NS (most) WDR Non-N (few)	Skin Viscera Muscle	Mainly VLF Mainly contralateral But ipsilateral (I–V) via dorsal columns to DRN	RF of brainstem \rightarrow LRN (NPGC/NGC), medial thalamus and DRN (few)	No	Small and large myelinated	All vertebrates	Motivational-affective? Descending inhibition
Spinomesence-phalic tract	NSN VII LSN LSN	NS (I) WDR Non-N	Skin Viscera Joints/muscle	Mainly VLF DLF (I, LSN) Mainly contralateral	Midbrain and PAG Deep SCL, NCF and PBN Thalamus (few)	Weak	Unmyelinated Small and large myelinated	All vertebrates	Motivational-affective. Autonomic, motor
Spinoparabrachio- amygdaloid tract	I II (few)	NS	Skin Viscera Joints/muscle	DLF-LF Mainly contralateral	$PBN \rightarrow amygdala$ and stria terminalis	Yes	Unmyelinated Small, myelinated	Mammals	Motivational-affective. Autonomic
Spinoparabrachio- hypothalamic tract	I II (few)	NS	Skin Viscera Joints/muscle	DLF-LF Mainly contralateral	$PBN \rightarrow hypothalamus$ (VMH).	No	Unmyelinated Small, myelinated	Mammals	Motivational-affective. Endocrine
Spinohypothalamic (spinotelencephalic) tract	I X LSN	NS WDR Non-N (few)	Skin Viscera	VLF Mainly contralateral	Hypothalamus and thalamus. Also pons, amygdala, striatum (bilateral)	ć	Unmyelinated Small, myelinated	Mammals	Sleep, autonomic and endocrine function Thermoregulation
Spinocervical tract	I (few) III/IV (most) V	WDR Non-N (most)	Skin Joints/muscle	DLF Ipsilateral—then contralateral (from LCN)	Relay LCN (CI–C3 level) →Contralateral thalamus (VPL/VMPo) and midbrain (PAG and SCL) Some LCN cells → spinal cord	Cats and monkeys Not rats	Small and large myelinated	All vertebrates Prominent in carnivores and primates	Discriminative-sensory Motivational affective Autonomic?
Postsynaptic dorsal column (ML) pathway	III-V (most) VI VII	NS WDR Non-N	Skin Viscera Joints/muscle	DF (and DLF) Ispsilateral—then contralateral (from DCN)	Relay DCN of caudate medulla: via ML \rightarrow contralateral thalamus (VPL/VMPo) Also SCL and spinal cord (few)	Yes (VPL)	Small and medium myelinated	Not fish Prominent in mammals	Discriminative-sensory (VPL) Motivational-affective (VMPo)
WDR, wide-dyn sal raphe nuclei; L ventral aspect; NP perior colliculus; V and VMH, ventror	amic rang CN, latera GC/NGC, 'LF, ventr nedial hyp	e; NS, nocicepti Il cervical nucleu nucleus (paro) olateral funicult othalamus. —Sy	ve-specific; Nor us; LRN, latera gigantocellulari us; VMPo, vent ymbolizes subse	n-N, non-nociceptive; I reticular nucleus; L is; PAG, periaquaduc romedial posterior th equent, second order	; CL, centrolateral; DCN, dorsal colum SN, lateral spinal nucleus; ML, mediar ital grey; PBN, parabrachial nucleus; P aalamus; VPL/VPM, ventroposterolate projections.	in nuclei; DF, 1 lemniscus; N O: posterior g ral/ventropos	dorsal funiculus; VCF, nucleus cume group of thalamic teromedial thalam	, DLF, dorsolat eiformus; MDv nuclei; RF, ret aus; VPI, ventr	eral funiculus; DRN, dor- c, medial dorsal thalamus, icular formation; SCL, su- oposterioinferior thalamus

Table 4. Overview of ascending pathways transmitting nociceptive information to higher centres

The Induction of Pain: an Integrative Review

31

'Polysynaptic' pathways possess, in contrast to their monosynaptic counterparts, a relay station of second order neurones en route to higher centres. The two, principal polysynaptic routes are the spinocervical pathway, which projects to the lateral cervical nucleus at the C1-C3 level, and the PSDC pathway (or 'lemniscal system'), which projects to the cuneate (from thoracic and cervical segments) and gracile (from lumbosacral segments) dorsal column nuclei (DCN) of the caudal medulla. The DCN receives, in addition, a direct ascending innervation from PAFs carrying proprioceptive and normally innocuous rather than nociceptive information. It is possible that this channel plays a role in the mediation of mechanical allodynia (Al-Chaer et al., 1996a; Berkley and Hubscher, 1995; Miki et al., 1998). (Some spinoreticular tract neurones may also ascend in the dorsal columns.) From the lateral cervical nucleus, information runs via the cervicothalamic tract to several thalamic nuclei, including ventroposterior and posterior groups of nuclei, and via a cervicomesencephalic pathway to the midbrain, including the PAG and superior colliculus. As concerns the DCN, output neurones project via the medial lemniscus (ML) to the ventroposterior and posterior groups of nuclei of the thalamus and to the superior colliculus, while a few fibres descend the spinal cord.

Thus, there is a complex—and likely interactive pattern of direct and indirect, multiple innervation of the thalamus, midbrain, limbic system, cortex, reticular formation and many other cerebral structures via *multiple* ascending pathways. These supraspinal regions are extensively interlinked and they also interact with mechanisms of descending modulation running to the DH (Sections 6.3.5 and 6.4).

6.1.3. Ascending Pathways: Contralateral and Ipsilateral Channels in the Spinal Cord

The PSDC pathway occupies the dorsomedial region of spinal cord white matter and this ascending channel is referred to as the dorsal (posterior) funiculus (Hirshberg et al., 1996). Some PSDC neurones access the DCN via the dorsolateral funiculus (DLF) which corresponds to the dorsolateral quadrant of the white matter. The DLF is also the major route for ascending spinocervical and, probably, spinoparabrachial pathways. Specific populations of PNs contributing to the STT and the spinomesencephalic tract may also use the DLF: in particular, STT neurones in lamina I and the lateral spinal nucleus (located within the VH). This dorsally-segregated, lamina I-derived STT pathway is sometimes referred to as the 'dorsal' STT in contrast to the 'ventral' STT which runs, together with the spinomesencephalic and spinoreticular tracts, in the ventrolateral funiculus (VLF): that is, the anterolateral quadrant of the white matter (Apkarian, 1995b) (see Section 6.2.2 for differences in cerebral targets of the dorsal/ventral STT). Within the VLF, there is a lateral organization of ascending pathways such that PNs originating in cervical and thoracic segments occupy the most medial, and PNs originating in lumbar and sacral segments, the most lateral, positions—a pattern opposite to that seen in the PSDC. In parallel to this segregation of ascending pathways, it is of interest that descending inhibitory and facilitatory pathways from the ventral rostro-medial medulla are localized in the DLF and VLF, respectively (Zhuo and Gebhart, 1997).

As concerns the three 'classical', monosynaptic pathways, the STT, the spinoreticular tract and the spinomesencephalic tract, their projections are predominantly contralateral. That is, their PNs of origin traverse the midline within the same segment or, possibly, at the level of an adjacent segment, before ascending to their cerebral targets. The spinoparabrachial pathways in the DLF are also largely contralateral. In distinction, the *polysynaptic*, spinocervical and PSDC pathways are ipsilateral at the spinal level. However, subsequent to synaptic contact with their second order neurones in the lateral cervical nucleus and the DCN, respectively, the output from these relay nuclei decussates to the contralateral brain. The spinohypothalamic pathway possibly constitutes an exception. The majoritythough not all-fibres attain the hypothalamus via the contralateral VLF. However, following their hypothalamic sites of termination in diverse nuclei, collaterals pursue their course to the supraoptic decussation, at which point many make a 'U-turn' to the contralateral brain and continue to run caudally to innervate the hypothalamus itself, the thalamus, pons and limbic structures, such as the amygdala, septum and striatum (Giesler et al., 1994; Giesler, 1995a,b; Li et al., 1997b; Newman et al., 1996). Thus, some axons appears to provide a bilateral innervation of the hypothalamus and thalamus while their innervation of limbic structures is ultimately ipsilateral, since they decussate both in the spinal cord before entering the VLF and again at the supraoptic level (Giesler et al., 1994). Detailed quantitative descriptions of uni- and bi-, ipsi- and contralateral projections of this pathway have recently appeared (Burstein et al., 1996; Kostarczyk et al., 1997)

Thus, although the supraspinal receipt of nociceptive input is predominantly contralateral, there *does* exist a significant component of ipsilateral nociceptive input to cerebral structures. This is of note with respect to studies of the activation of specific cerebral regions by pain in man (Section 6.3.2).

6.1.4. Ascending Pathways: Sensory-Discriminative and Affective-Cognitive Dimensions of Pain

For an understanding of the role of ascending pathways and supraspinal mechanisms in the modulation and experience of pain, it is imperative to distinguish between its two fundamental components, processed by distinctive, yet interconnected and interactive, circuits.

- 1. The sensory-discriminative: that is, the perception and detection of noxious stimuli *per se* as concerns their intensity, location, duration, temporal pattern and quality.
- 2. The affective-cognitive: that is, the relationship between pain and mood, the attention to and memory of pain, the capacity to cope with and tolerate pain and its rationalization (Besson *et*

al., 1995; Craig, 1994, 1996; Derbyshire *et al.*, 1997; Jones and Derbyshire, 1996; Millan, 1990).

In this respect, it is possible to differentiate between the roles of particular ascending pathways-and even specific components of individual tracts. Accordingly, ascending pathways possess differing networks of connectivity to various supraspinal centres, while their PNs of origin in the DH and their target neurones in higher structures display contrasting patterns of response to noxious stimuli. In this light, then, it is necessary to take account of the ability of neurones to localize and intensity-code noxious stimuli. For example, as described in detail below (Section 6.2.2), deep laminae WDR neurones with small RFs encoding the intensity and location of cutaneous, noxious stimuli provide a major con-tribution to the ventral STT. This pathway densely innervates thalamic nuclei involved in the sensorydiscriminative aspects of pain and possessing neurones with similar response properties to their DH counterparts. In distinction, NS-neurones of lamina I project-via the dorsal STT-predominantly to other thalamic nuclei concerned with the emotionalcognitive aspects of pain and similarly possessing nocisponsive neurones with less refined stimuluscoding properties (Besson et al., 1995; Craig et al., 1994; Dostrovsky and Guilbaud, 1990; Lima and Coimbra, 1988; Ralston and Ralston, 1992). In analogy to the ventral STT, the spinocervical tract and a component of the PSDC pathway indirectly project, via the lateral cervical nucleus and DCN, respectively, to thalamic regions involved in the intensity-coding of noxious stimuli, suggesting that they may also play a role in the discriminative-sensory dimension of pain. On the other hand, those components of the spinocervical and PSDC pathways which project to (posterior) thalamic nuclei targeted by the dorsal STT, the spinomesencephalic tract (in particular via the PAG), the spinoreticular pathway (via thalamocortical circuits), the spinoparabrachialamygdaloid pathway and the spinohypothalamic pathway all play an important role in the affective dimension of pain (Apkarian, 1995b; Besson et al., 1995; Jones and Derbyshire, 1996; Willis and Coggeshall, 1991) (Table 4).

Although it would be naive to recognise an absolute dichotomy between sensory and affective roles of individual ascending nociceptive systems, which are collectively responsible for the overall experience of pain (Jones and Derbyshire, 1995, 1996) (Section 6.3.7), this distinction has evident implications for the induction and treatment of clinical pain. An example is provided by the apparently multiple roles of the STT, which is still widely considered the principal route via which nociceptive information accesses the brain. Thus, a 'breakthrough' of clinical pain may be observed in patients who have undergone an apparently total anterolateral cordotomy of the VLF to interrupt the STT (Gybels and Sweet, 1989; White and Sweet, 1969). This supports the notion that the ventral STT-as well as the spinomesencephalic, the spinoreticular and the spinohypothalamic tracts, which also use the VLF-are not the only sources of nociceptive information underlying clinical pain. Indeed, it has been proposed, on

the basis of these and other observations (including imaging studies in man), that the classical ventral VLF/STT pathway may *not* be a major avenue for the induction of intractable, visceral pain (Apkarian, 1995b). This may be an overgeneralization but serves to emphasize the point that other pathways are involved in the induction of *clinical* pain. Notably, the *dorsal* component of the STT which is spared by VLF cordotomy (Craig et al., 1994). In this regard, the potential significance of the spinoparabrachial tracts, the spinocervical tract and, in particular as concerns abdominal and pelvic visceral pain, the PSDC, should be accentuated (Al-Chaer et al., 1996a, 1997a,b; Berkley and Hubscher, 1995; Bernard et al., 1993, 1994, 1995; Hirshberg et al., 1996; Nauta et al., 1997; Willis and Coggeshall, 1991) (Table 4). Each of these ascending pathways utilizes channels *other* than the VLF and each plays an important role in the affective component of clinical pain.

Finally, a more general role has been attributed to a spinothalamocortical projection originating from both NS and other types of 'polymodal' neurone in lamina I whereby this pathway is assumed to permit 'a generalized assessment of the physiological status of the entire organism' (Craig, 1996; Zhang and Craig, 1997) (Section 4.2).

To summarize, in particular as concerns visceral (and deep somatic) clinical pain, it is arguable that *no* single pathway for the ascending passage of nociceptive information should, in principle, be considered as dominant. Dependent on the precise locus of pain, the duration and type of noxious stimulus and many other factors, multiple channels of nociceptive (and other modes of sensory) information likely converge and interact at the supraspinal level to yield the global sensation of pain (Apkarian *et al.*, 1995; Berkley, 1997; Berkley and Hubscher, 1995; Besson *et al.*, 1995; Gebhart, 1995; Hirshberg *et al.*, 1996; Watkins *et al.*, 1995; Wiertelak *et al.*, 1994c) (Section 6.3.7). This contention is further underpinned by imaging studies of patterns of cerebral activation elicited by pain (Section 6.3.2).

6.1.5. Visceral Nociceptive Input to the Thalamus

When discussing the emotional and cognitive components of acute and prolonged *clinical* pain, the viscera and other deep tissues are of special pertinence. Thalamic encoding of cutaneous as compared to visceral input differs since, in accordance with the large, poorly-defined RFs of PAFs innervating the viscera and the limited ability to localize visceral pain, input from visceral tissues to the thalamus is, in general, not topographically organized (Cervero, 1994, 1995b; Gebhart, 1995; McMahon et al., 1994). As mentioned above, the STT is not the only ascending pathway which conveys nociceptive information from the viscera to higher centres (Table 4). For example, the marked responsiveness of spinohypothalamic tract neurones in the DH to noxious, visceral stimulation has been underlined by several authors (Bullitt, 1990; Burstein et al., 1991, 1996; Giesler, 1995a,b; Giesler et al., 1994). Via projections to limbic regions, this pathway may also play an important role in the induction of pain from the thoracic (chest) region (Katter et al., 1996a,b; Newman et al., 1996). The high degree of thalamic convergence of cutaneous, deep somatic and visceral information from multiple ascending pathways is in line with the presence of a substantial population of neurones responding to noxious stimulation of the viscera and the skin at both the DH and thalamic level, in particular, its ventroposterior aspect (Apkarian, 1995b; Brüggemann et al., 1994; McMahon et al., 1994). The interrelationship between nociceptive information arriving from various tissue regions onto a common population of thalamic neurones is an issue of obvious relevance to the phenomenon of 'referred' pain (Apkarian et al., 1995) (Section 9.1.4). Although little quantitative information concerning the relative influence of somatic vs visceral input upon supraspinal neurones is available, it is of note that the induction of visceral inflammation modifies the balance of efficacy in exciting convergent neurones from the skin towards the viscera (Al-Chaer et al., 1996b). Indeed, it more markedly sensitizes cells of the PSDC pathway to noxious visceral, mechanical as compared to cutaneous input (Al-Chaer et al., 1997a,b).

As indicated above, recent studies suggest that the PSDC/ML system may fulfil a hitherto unsuspectedly important role in visceral pain transmission, notably from the pelvic region and abdomen. In accordance with this view, surgical myelectomy to interrupt this pathway has been used for the alleviation of pelvic pain (Berkley, 1997; Hirshberg *et al.*, 1996; Houghton *et al.*, 1997; Nauta *et al.*, 1997). Although clinical observations are, as yet, *limited*, such findings should be considered in the light of the above-mentioned reports that surgical interruption of the VLF *fails* to invariably and irreversibly eliminate clinical, visceral pain (Berkley, 1997).

Thus, the thalamus clearly receives multiple sources of visceral input: notably, via the VLF from the ventral STT, and via the dorsal funiculus/DLF and ML from the PSDC (Al-Chaer et al., 1997a,b; Berkley and Hubscher, 1995; Gybels and Sweet, 1989; Hirshberg et al., 1996; Katter et al., 1996a,b; Vierck et al., 1986; Newman et al., 1996) (Table 4). It would be of interest to directly examine the comparative influence, and possibly synergistic interactions, of diverse pathways mediating visceral pain upon the activity of specific, nocisponsive units in the thalamus. A pathological disequilibrium in the flow of nociceptive and other forms of sensory information to the thalamus, PBN, solitary tract nucleus (via the parasympathetic, vagal nerve) and other supraspinal structures involved in the integration of sensory, emotional and autonomic processes may result in the overall, clinical phenomenon of visceral pain (Aylwin et al., 1997; Berkley, 1997; Hubscher and Berkley, 1995; Wiertelak et al., 1994c; Watkins et al., 1995).

6.2. Cerebral Targets of Ascending Nociceptive Pathways: Functional Roles

6.2.1. The Key Role of the Thalamus

Table 4 indicates several of the principal cerebral sites of termination of ascending pathways for the

transmission of nociceptive information. It has long been considered that the most important pathway for the experience of 'pain' is the VLF-channelled component of the STT which innervates the thalamus and, subsequently, the postcentral gyrus of the cortex. However, as discussed herein and elsewhere (Apkarian et al., 1992, 1995; Apkarian et al., 1995; Berkley and Hubscher, 1995; Berkley, 1997; Besson et al., 1995), this appears to be an oversimplification, in particular as concerns visceral pain and the affective-emotional dimension of prolonged, clinical pain. In line with this ostensibly predominant role of the STT, the thalamus has long been regarded as the key relay structure for the supraspinal receipt, integration and onward transfer of nociceptive information (Willis and Coggeshall, 1991; Guilbaud et al., 1994). As indicated in Table 4, for motivational, cognitive, motor and autonomic aspects of pain, other direct and indirect targets of ascending nociceptive information are also of major importance. Nevertheless, the thalamus is not only targetted by the ventral SST, but also, directly and indirectly, by other tracts carrying nociceptive information (Table 4). Thus, the thalamus must still be considered as the crucial relay for the reception and processing of nociceptive (and other forms of sensory) information en route to the cortex (Bushnell, 1995). Indeed, the thalamus encodes information concerning the type, temporal pattern, intensity and-for cutaneous input-topographic localization of pain. Further, it interlinks with cortical and limbic structures responsible for both the sensory-discriminative and emotional dimension of pain (Section 6.3.5). The thalamus also plays an active, adaptive role in processing nociceptive information (Section 12.2.3). Recent experimental and clinical studies suggest that contrasting functional roles of individual thalamic nuclei should be carefully distinguished.

6.2.2. Contrasting Roles of Thalamic Nuclei with Respect to the Sensory-Discriminative and Affective-Cognitive Components of Pain

A discussion of the roles of various thalamic nuclei in the integration and relaying of nociceptive information is complicated by confusing terminological and species-dependent inconsistencies. Nevertheless, electrophysiological, clinical and anatomical observations, as well as metabolic studies of energy utilization, have identified several major clusters of thalamic nuclei (described herein primarily according to primate nomenclature) involved in the receipt, processing and onward transfer of nociceptive information, all of which receive an intense nociceptive input from DH neurones, in particular those in laminae I and V/VI (Apkarian, 1995a,b; Besson et al., 1995; Bullitt, 1989, 1990; Bushnell, 1995; Chapman and Besson, 1997; Clément et al., 1996; Coghill et al., 1994; Craig, 1995; Dong and Chudler, 1995; Guilbaud et al., 1994; Jones, 1985; Jones et al., 1991a; Jones and Derbyshire, 1995; Price et al., 1995).

1. Much of the literature refers to the 'ventrobasal complex' of the rat (e.g. Eaton and Salt, 1995; Guilbaud *et al.*, 1994), and the term 'principal

sensory nucleus' has been employed in the primate to define a key, thalamic region for the arrival and processing of nociceptive input. For many authors, the current preference is to distinguish the ventroposteriolateral (VPL) and ventroposteriomedial (VPM) components of this region in primates, as well as a contiguous (though functionally distinct) area of the lateral thalamus, termed the ventroposterioinferior nucleus (VPI) (Apkarian, 1995b; Besson *et al.*, 1995; Bushnell, 1995). Whereas the ventral STT terminates primarily in the VPL/VPM, the dorsal STT appears to preferentially innervate the VPI.

- 2. The posterior complex of thalamic nuclei incorporates the pulvinar oralis, the posterior nucleus and the posterior division of the ventromedial nucleus (VMpo)-or posteriolateral thalamus. This region is a major site for the integration of nociceptive and thermosensory input, and its functional disregulation has been implicated in thalamic pain (Craig et al., 1994, 1996; Dostrovsky and Craig, 1996) (Section 13.4). Like the VPI, it is primarily accessed by the dorsal rather than ventral STT (Apkarian, 1995b; Craig et al., 1996). Cortical inactivation can inhibit the response of certain thalamic neurones to noxious stimulation and it has been suggested that the posterior thalamus forms part of a reverberating, cortico-thalamo-cortical loop whereby the cortex reinforces the excitation of thalamic neurones by nociceptive input and vice versa (Eaton and Salt, 1995; Nothias et al., 1988) (Sections 12.3 and 13.3). Further underpinning a role of the posterior thalamic region in nociceptive processing, most nocisponsive STT neurones of the sacral spinal cord terminate herein, whereas those preferentially responsive to innocuous, mechanical stimuli project predominantly to other thalamic nuclei (Katter et al., 1996a).
- 3. The medial thalamus of the rat includes the nucleus submedius and the nucleus parafascicularis. The primate homologue of the rat nucleus submedius may be the ventrocaudal portion of the medial thalamus which is termed its medial dorsal (MDvc) aspect. This region, like the VMpo, receives a pronounced input from lamina I neurones of the dorsal STT (Bushnell, 1995; Dostrovsky et al., 1995; Guilbaud et al., 1994). The anatomically-related nucleus centralis lateralis of the intralaminar nucleus also receives a proinput from the STT and is nounced interconnected with the striatum and other regions controlling attention and motor function, including the cerebellum. These links are consistent with the notion that this division of the thalamus is involved in the escape reaction to acute pain (Apkarian, 1995b; Casey et al., 1996a,b; Molinari et al., 1986). Interestingly, imaging studies of the cerebellum have revealed a pronounced response to noxious stimuli in man (Allen et al., 1997b).
- 4. Lenz (1995) identified a region in the human and primate thalamus which, by virtue of its interconnections with corticolimbic structures, may play a key role in the emotional-cognitive aspects of pain. This area was—somewhat cryptically-

characterized as 'posterior-inferior to the Vc'. The 'Vc' (ventralis caudalis) corresponds, apparently, to the core region of the 'principal sensory nucleus' (VPL/VPM). This suggests that the term 'posterio-inferior to the VC' may be functionally equivalent to the VPI and posterior thalamic nuclei. Notwithstanding its nebulous localization, Lenz *et al.*, 1997 have made the interesting proposition that this region plays a key role in memorization of the affective dimension of pain.

Many neurones in the above-mentioned thalamic nuclei, which include both NS as well as WDR types, can encode and discriminate the intensity of noxious stimuli (Bushnell, 1995; Guilbaud et al., 1994). However, parallel electrophysiological and psychophysical studies in primates suggest that the VPM and VPL may be the most finely-tuned in this regard and, in behavioural tests, their reversible inactivation by lidocaine markedly diminishes the ability to detect minor changes in noxious temperatures applied to the skin (Bushnell, 1995). Moreover, RFs of neurones in this region are consistently small and contralateral, as compared to the larger, less welldefined-and often bilateral-RFs for neurones in posterior thalamic nuclei. The VPM/VPL region is particularly suited, thus, to the encoding of both the intensity and location of noxious stimuli, consistent with an important sensory-discriminatory role. In distinction, neurones in the VPI, posterior group of nuclei (VMpo etc) and other thalamic regions, such as the nucleus submedius/MDvc, may be more sensitive to attentional, cognitive and emotional states and, correspondingly, involved in the affective aspects of pain (Bushnell, 1995; Dostrovsky and Guilbaud, 1990; Lenz, 1995). This contention is underpinned by anatomical studies of the differential patterns of connectivity of these thalamic regions to specific, cortical structures (Section 6.3.3).

To summarize, then, specific thalamic nuclei play distinctive and complementary functions in integrating the sensory-discriminative (e.g. VPL/VPM) as compared to affective-cognitive (e.g. VMpo/VPI) components of pain. These divergent roles of individual regions reflect, in particular:

- the contrasting electrical properties of their neurones (stimulus intensity-encoding and RF size);
- differential sources of direct and indirect nociceptive input from the SST and other ascending pathways; and
- 3. contrasting patterns of connectivity to cortical and other supraspinal structures (Section 6.3.3).

6.2.3. Neurotransmitters Involved in the Processing of Nociceptive Transmission in the Thalamus: Focus on EAAs and GABA

As mentioned above (Section 6.1.1), the precise identity of neurotransmitters involved in conveying nociceptive information via ascending pathways to the thalamus and other supraspinal structures remains unclear. Nevertheless, EAAs are implicated and they have been shown to exert a pronounced, excitatory influence upon the activity of nocisponsive neurones in the thalamus and other regions receiving nociceptive information (Aylwin *et al.*, 1997; Azkue et al., 1997; Di Biasi et al., 1994; Eaton and Salt, 1995; Ericsson et al., 1995; Jensen and Yaksh, 1992; Salt and Eaton, 1996). In this regard, in analogy to the DH (Sections 3.2.8.3 and 10.4.2.2), NO may be involved in the activation by NMDA receptors of nocisponsive neurones in the thalamus (Do et al., 1994; Kolhekar et al., 1997). Studies of the VPL/VPM and posterior thalamic nuclei have suggested a predominant, and possibly cooperative, role of NMDA and mGlu receptors in mediating the influence of noxious stimuli upon thalamic neurones (Dougherty et al., 1996; Eaton and Salt, 1995, 1996; Kolhekar et al., 1997). Interestingly, in analogy to thalamic projections to the cortex (Section 6.3.3), reciprocal, cortical pathways to thalamic nuclei utilize EAAs acting at NMDA (and mGlu) receptors. This is consistent with the notion that a feedback action of the cortex may increase the 'gain' (amplify the response) of some nocisponsive, thalamic neurones (Eaton and Salt, 1995; Isaac et al., 1997: Nothias et al., 1988). A role of NMDA receptors in mediating the excitation of thalamic neurones by ascending, nociceptive input is of particular interest since, in analogy to the DH (Section 10.3), NMDA receptor-mediated processes of synaptic sensitization may contribute to the increased activity of thalamic neurones seen upon prolonged, noxious stimulation (Guilbaud et al., 1994; Kolhekar et al., 1997) (Section 12.3.3). Interestingly, in contrast to noxious input, non-noxious, sensory activation of thalamic neurones may primarily involve activation of NMDA and AMPA other than mGlu receptors (Eaton and Salt, 1995, 1996).

Providing a further analogy to neuronal mechanisms in the DH of the spinal cord (Sections 4.6 and 10.6), the thalamus is equipped with inhibitory, modulatory mechanisms in which GABAergic ININs play a prominent role. These local-circuit, GABAergic ININs constitute a substantial proportion (10-25%) of the total population of neurones in the thalamus (Ralston et al., 1995; Ulrich and Huguenard, 1997) and they play a major role in setting the output properties of thalamic relay neurones. The possibility that a perturbation in the activity of GABAergic ININs contributes to thalamic pain and other painful conditions is outlined in Sections 12.3 and 13.3. (Alloway et al., 1989; Kim et al., 1997a; Oliveras and Montagne-Clavel, 1994; Paré et al., 1991; Roberts et al., 1992; Ralston et al., 1995; Ulrich and Huguenard, 1997).

In analogy to actions characterized in the DH (Section 4.5), neuropeptide FF has been shown to act pronociceptively in the thalamus (Dupouy and Zajac, 1997). Indeed, a variety of other neuropeptides are also likely to be involved in the modulation and transfer of nociceptive information at the supraspinal level (Altier and Stewart, 1997a,b; Heinricher *et al.*, 1997; Morgan *et al.*, 1997; Saleh, 1997; Yasui *et al.*, 1989).

6.3. Thalamic Input to the Cortex: Differential Integration of Sensory-Discriminative as Compared to Affective-Cognitive Aspects of Pain

6.3.1. Neurotransmitters Involved in Processing Nociceptive Information in the Cortex

There is little precise information available concerning the neurochemistry of neurones transmitting nociceptive information to the cortex and, similarly, little is known regarding the transmitters involved in the exchange of nociceptive information amongst various cortical regions. Nevertheless, the following general points should be noted.

Studies of sensory input to the somatosensory cortex, in particular the visual cortex, suggest that EAAs acting at NMDA, AMPA and group I mGlu receptors likely play a key role in the cortical transfer of nociceptive information by thalamic output neurones, which comprise the majority of neurones in the thalamus (Armstrong-James et al., 1993; Castro-Alamancos and Connors, 1997; Conti and Hicks, 1996; Conti et al., 1997; Gil and Amitai, 1996a,b; Isaac et al., 1997; Itazawa et al., 1997; Ralston et al., 1995; Salt et al., 1995; Vickery et al., 1997; Zhou and Hablitz, 1997) (Section 14.4). The detection of COX 2 synthase in excitatory, cortical cells is of particular interest since NMDA receptors have been shown to enhance its activity in the cortex: a putative role of PGs in the cortical modulation of nociceptive transmission would, thus, be of interest to examine (Kaufmann et al., 1996; Miettinen et al., 1997; Vanegas et al., 1997). The presence of a substantial population of GABAergic ININs in the sensory cortex, engaged by thalamocortical afferents and acting via GABAA and GABA_B receptors, should be underlined (Castro-Alamancos and Connors, 1997; Conti and Hicks, 1996). Cytokines and neurotrophins may also, via interactions with EAAs/NMDA receptors and GABAergic transmission, be involved in the modulation of nociceptive processing in the cortex (Sections 4.8 and 14.4). The potential significance of neuropeptides in cortical transmission of nociceptive information remains to be elucidated.

Sections 12.3.3 and 13.5 discuss the potential role of cortical NMDA receptors and GABAergic transmission in adaptive processes underlying pain due to peripheral noxious input and CNS damage.

6.3.2. Cortical Structures Activated by Noxious Stimulation: Patterns of Responsiveness

In has generally been considered that nociceptive information emanating from the thalamus is directed to the first, somatosensory area (S I) of the postcentral, cortical gyrus. However, more recent studies employing:

- 1. parallel electrophysiological and anatomical observations;
- 2. metabolic and cerebral blood flow imaging techniques for the identification of tissues activated by noxious stimuli in man; and
- an examination of the effects of damage to various cortical structures upon nociceptive processing,

have collectively refined our understanding of the
relationship between specific thalamic nuclei and circumscribed regions of the cortex and other sub-cortical structures.

Such studies reinforce the concept of a *differential* role of particular thalamocortical networks in the sensory-discriminative (e.g. S I) as compared to the affective-cognitive (e.g. anterior cingulate) aspects of pain (Apkarian *et al.*, 1992; Casey *et al.*, 1994; Casey and Minoshima, 1995; Coghill *et al.*, 1994; Derbyshire *et al.*, 1997; Di Piero *et al.*, 1997; Dong *et al.*, 1996; Jones *et al.*, 1991a; Jones and Derbyshire, 1995, 1996; Kenshalo and Willis, 1991; Rainville *et al.*, 1997).

The following cortical regions (defined primarily according to primate anatomy) possess nocisponsive NS and/or WDR neurones and, in imaging studies, they are consistently activated by noxious stimulation to cutaneous and other tissue regions:

- 1. the S I of the postcentral gyrus;
- 2. the somatosensory area II (S II) which occupies the (parietal) operculum in the sylvian sulcus of the lateral parietal cortex;
- several regions of the inferior and anterior parietal cortex;
- 4. the insular cortex;
- 5. the anterior cingulate cortex; and
- the medial prefrontal cortex (Casey and Minoshima, 1995; Casey et al., 1996b; Davis et al., 1997b; Derbyshire et al., 1997; Jones and Derbyshire, 1995; May et al., 1998; Neal et al., 1990; Svensson et al., 1997).

These cortical structures display a complex pattern of connections amongst themselves and they may also be interlinked via the thalamus or various limbic structures (Sherman and Guillery, 1996). For example, the S II is interconnected with both the S I and the insular cortex (Cavala and Goldman-Rakic, 1989; Neal et al., 1990). EAAs acting at NMDA (and non-NMDA) sites are probably involved in these intracortical networks (Conti et al., 1997; Gil and Amitai, 1996b; Nothias et al., 1988; Salt et al., 1995; Sherman and Guillery, 1996). It is possible thus, that in addition to direct thalamocortical projections, certain cortical structures may be indirectly activated via such pathways (Pons et al., 1991). In this light, temporal features of the activation of specific, cortical regions are of importance. For example, the S I and cingulate cortex appear to be rapidly activated and may participate in the sequential activation of the prefrontal cortex, the S II and the insula cortex (Casey and Minoshima, 1995; Casey et al., 1996b; Davis et al., 1997b; Jones and Derbyshire, 1995; Neal et al., 1990; Svensson et al., 1997; Zhang et al., 1996a).

In imaging studies, cortical activation is generally detected *contralaterally* to noxious stimulation—and almost exclusively so in the case of the S I region, in line with a pain-localizing, discriminative-sensory function of this region. However, some studies have reported an additional ipsilateral activation, for example, in prefrontal cortex, anterior cingulate cortex—as well as the cerebellum and thalamus (Casey and Minoshima, 1995; Casey *et al.*, 1996b; Derbyshire *et al.*, 1997; Di Piero *et al.*, 1997; Jones

and Derbyshire, 1995, 1996; May et al., 1998; Svensson et al., 1997; Telford et al., 1996). The reasons for this are unclear, although some cortical cells show bilateral RFs and a contribution of ipsilateral pain projection pathways provides one obvious explanation (Dong et al., 1994; Lenz, 1995) (Section 6.1.3., Table 4). Alternatively, the interhemispheric cortical transfer of nociceptive information might be possible via either commissural connections or the thalamus (Bramham et al., 1996; Calford and Tweedale, 1990; Jones and Powell, 1969; Sherman and Guillery, 1996; Shin et al., 1997; Zhang et al., 1996a). Further, pain intensity is an important factor in activation patterns (Di Piero et al., 1997) and it appears that intense and prolonged, repetitive stimulation is preferentially accompanied by a *bilateral* activation of the cortex. This observation implicates mechanisms of sensitization (and alterations in RFs) in response spread: indeed, a transfer of synaptic plasticity (LTP) has been documented from ipsilateral to contralateral cortex (Bramham et al., 1996). In addition, emotional-cognitive factors may underlie a secondary, bilateral pattern of activation in regions, such as the cingulate and prefrontal cortex-plus interlinked thalamic nuclei, implicated in the affective aspects of pain (Jones and Derbyshire, 1996; Masterman and Cummings, 1997; Rainville et al., 1997: Soares and Mann, 1997; Vogt et al., 1996). Various types of pathological condition may be accompanied by ipsilateral and/or bilateral patterns for activation. For example, neuropathic pain has been associated with a bilateral activation of cortical structures (Casey et al., 1996a; Hsieh et al., 1995; Jones et al., 1995; Vogt et al., 1996). On the other hand, patients with unilateral cluster headaches showed a lateralized perturbation of nociceptive processing (Di Piero et al., 1997).

In analogy to the thalamus, then, specific cortical regions may fulfil *contrasting* roles in the sensory–discriminative and emotional-cognitive components of pain, a contention further supported by several lines of evidence:

- 1. differential innervation by various thalamic nuclei;
- 2. the contrasting ability of neurones in various cortical regions to localize and intensity-code noxious stimuli; and
- 3. contrasting pattern of connectivity to subcortical limbic structures (Besson *et al.*, 1995).

These observations are considered in the following paragraphs.

6.3.3. Patterns of Thalamic Input to the Cortex: Relationship to Discriminative-Sensory and Cognitive-Affective Dimensions of Pain

Anatomical relationships between individual thalamic and cortical structures involved in the processing of nociception generally correlate with their mutual roles in the perception as compared to the conscious appreciation of pain (Apkarian, 1995b; Bushnell, 1995; Craig *et al.*, 1982, 1996; Coffield *et al.*, 1992; Dostrovsky *et al.*, 1995; Hsu and Shyu, 1997; Lenz, 1995; Vogt *et al.*, 1993; Rausell *et al.*, 1992a; Schmahmann and Pandya, 1990; Zhang *et* *al.*, 1997f). As concerns thalamic sources of nociceptive input to the cortex, the following points should be emphasized:

- 1. the VPL/VPM provides an intense input to SS I;
- the nucleus submedius of the medial thalamus innervates the ventrolateral orbital cortex in rats which subsequently projects to the PAG and SS I. This may be homologous to a projection from the MDvc (and nucleus parafasciculus) to the cingulate cortex in primates;
- 3. the VPI and intralaminar nuclei project to S II.

Further, in man, a structure 'posterior-inferior to the Vc' region of the thalamus, which likely incorporates the posterior thalamic nuclei, projects to the S II and the insula cortex (Lenz *et al.*, 1997) (Section 6.2.2). Apkarian (1995b) has commented on the quantitatively different input to S I vs S II regions of the cortex. For the S I, the principal source is, as indicated above, the VPL (64%) with the VPI (16%) and the posterior nuclei (VMpo) (3%) fulfilling comparatively minor roles. In contrast, the equivalent values are 18, 36 and 20%, respectively, for the S II.

The above observations imply that the distinction between the ventral STT-VPL/VPM-S I and dorsal STT-VPI/VMpo-S II divisions of the STT may, at least partially, be maintained up to the level of the cortex. The former and latter networks are, to reiterate, concerned primarily with the discriminative as compared to affective dimensions of pain. This separation, then, together with an important role of other ascending pathways in the emotional response to pain (Table 4), explains the viewpoint of certain authors that the prototypical STT/VLF pathway to the VPL/VPM of the thalamus, and therefrom to the S I, may not be the most important tract involved in mediating clinical pain (Apkarian, 1995b) (Section 6.1.4). As further indicated below, the above-described, differential innervation of various, cortical regions by specific thalamic nuclei has clear functional implications.

6.3.4. Contrasting Roles of Specific Cortical Regions: Encoding Properties of Nocisponsive Neurones

Both NS and WDR neurones responsive to noxious stimuli are found in those cortical regions which respond to noxious stimulation (Section 6.3.2) and, in analogy to the thalamus and DH, a substantial degree of convergence for cutaneous and visceral noxious stimuli has been established: some neurones are even affected by other modes of sensory stimuli (Apkarian, 1995b; Dong and Chudler, 1995; Dong et al., 1989, 1996; Lenz, 1995). In the thalamus, PBN, amygdala and other cerebral targets of ascending nociceptive information, the ability of nocisponsive neurones to intensity-encode nociceptive stimuli is variable (Section 6.2.2) and similar differences are found amongst nocisponsive neurones in various cortical regions (Bernard et al., 1990; Bushnell, 1995; Dong et al., 1989; Menendez et al., 1996).

Consistent with an involvement in the sensory-discriminative aspects of pain, the S I cortex contains many WDR and NS neurones with small (contralateral) RFs encoding both the location and intensity

of noxious stimuli (Dong and Chudler, 1995; Dong et al., 1989; Kenshalo and Willis, 1991). These observations correspond well with the preferential innervation of S I by the ventral STT and the VPL/ VPM, the principal thalamic nuclei involved in sensory-discriminative function. Indeed, in analogy to anaesthesia of the VPL/VPM nuclei of the thalamus (Bushnell, 1995), lesions to the S I in man reduce the ability to discriminate noxious stimulus intensity but do not markedly alleviate clinical pain (Devinsky et al., 1995; Kenshalo and Willis, 1991). Further, in analogy to the VPL region of the thalamus, many S I neurones respond to both visceral and cutaneous noxious stimuli indicating the occurrence of convergence at the cortical level, an observation of relevance to referred pain (Section 9.1.4). Interestingly, the response of WDR cells in the S I increases upon repeated, noxious, thermal stimulation of the skin, a finding indicating that sensitization mechanisms are apparent at cortical levels (Kenshalo and Isensee, 1983). This issue is discussed further in Section 12.

Other cortical regions involved in nociceptive processing do not necessarily display such a high fidelity of intensity-coding. For example, neurones in the inferior parietal cortex reveal large, bilateral, cutaneous RFs and less precisely encode stimulus location and intensity than those in S I (Chatrian et al., 1975; Dong and Chudler, 1995; Dong et al., 1994). Further, neurones in the ventrolateral orbital cortex of the rat also display large RFs incompatible with a precise localizing function (Dostrovsky et al., 1995; Zhang et al., 1997f). Indeed, homologous neurones in the anterior cingulate cortex of the primate may show half or whole body RFs in line with their often bilateral patterns of activation by pain (Jones and Derbyshire, 1996; Vogt et al., 1993, 1996) (Section 6.3.2). This comparative inability to intensity-code and locate noxious stimuli is consistent with observations that these cortical regions receive nociceptive information from thalamic nuclei likewise less efficient in registering the sensory-discriminative aspects of pain (Section 6.2.2). Thus, to reiterate, the S II, inferior parietal, anterior cingulate, prefrontal and insular cortices likely play key roles in the affective rather than discriminative components of pain (Craig, 1995; Dong and Chudler, 1995; Coghill et al., 1994; Kenshalo and Willis, 1991; Masterman and Cummings, 1997; Soares and Mann, 1997; Rainville et al., 1997). Further, within the cingulate cortex itself, further subdivisions have been attributed specific functions concerning the affective, attentional and other components of the emotional-cognitive dimension of pain (Davis et al., 1997b; Jones and Derbyshire, 1996).

To summarize, in parallel with the specific thalamic nuclei via which they are innervated, discrete cortical regions appear to be differentially involved in the sensory-discriminative (S I cortex) vs affective-cognitive (S II, cingulate, inferior parietal, prefrontal and insular cortex) aspects of pain.

6.3.5. Relationship of Cortical Structures to Limbic Regions Involved in Nociceptive Processing

The differential involvement of particular cortical (and thalamic) regions in the sensory as compared to emotional components of pain may also reflect their contrasting patterns of connectivity to diverse, limbic regions. Indeed, cortical structures possessing nocisponsive neurones display an extensive pattern of reciprocal connections not only with the thalamus but also with limbic (and motor) regions such as the amygdala, striatum (lenticular nucleus of man), hypothalamus, accumbens, PBN, PAG and hippocampus (Casey and Minoshima, 1995; Casey et al., 1997; Coghill et al., 1994; Derbyshire et al., 1997; Lenz, 1995; Jones and Derbyshire, 1995; Vogt et al., 1993; Zhang et al., 1997f). These structures are both involved in nociceptive processing (Table 4) and also fulfil keys role in the control of cognition and mood (Bloom and Kupfer, 1995; Craig, 1994). By virtue of their particularly extensive limbic connections, the insular cortex and the anterior cingulate cortex justify special attention (Devinsky et al., 1995; Dong and Chudler, 1995; Kenshalo and Willis, 1991; Lenz, 1995; Tommerdahl et al., 1996). Notably, a component of the analgesic properties of μ -opioids may be mediated by an action in insular cortex (Burkey et al., 1996). In addition, the primate cingulate cortex, together with the adjacent and interconnected prefrontal cortex region, plays a key role in the control of cognition, attention and mood (Masterman and Cummings, 1997; Millan and Gobert, 1997; Soares and Mann, 1997). Correspondingly, manipulations of the cingulate region modify nociception in rodents and in man and, of particular note, its lesioning can alleviate chronic, clinical pain (Devinsky and Luciano, 1993; Gybels and Sweet, 1989; Jones and Derbyshire, 1995: Pastoriza et al., 1996; Vaccarino and Melzack, 1989; Zhang et al., 1997f).

The above-mentioned limbic regions receive, both directly (e.g. the PAG and PBN) and indirectly, via the thalamus and cortex, etc, nociceptive input from the spinal cord (Table 4). Further, their lesioning and/or pharmacological modulation can modify nociception, effects which likely involve an influence upon the activity of opioidergic, monoaminergic and/or other networks therein (Altier and Stewart, 1996, 1997a,b; Dong and Chudler, 1995; McKenna and Melzack, 1994; Vaccarino and Melzack, 1992). Although the lenticular nucleus (striatum) is not generally associated with nociception and pain, is of particular interest since:

- 1. it is contralaterally activated by noxious stimulation in man (Casey and Minoshima, 1995);
- it receives direct input from spinal nocisponsive—including visceral—PNs (Cliffer *et al.*, 1991; Giesler *et al.*, 1994; Giesler, 1995a Giesler, 1995b; Newman *et al.*, 1996) and
- 3. a perturbation of its function may be associated with pain in Parkinson's disease (Boivie and Östeberg, 1995; Dong and Chudler, 1995).

The PAG is also of especial interest. This region is:

 excited by noxious stimulation from visceral and other organs;

- receives a pronounced input from the anterior cingulate in primates (and ventro-orbital cortex in the rat);
- 3. plays a pivotal role in integrating behavioural and autonomic responses to pain; and
- is of key importance in the modulation of mood, especially the response to aversive stimulation (Clément *et al.*, 1996; Dostrovsky *et al.*, 1995; Graeff *et al.*, 1993; Jones and Derbyshire, 1996; Zhang *et al.*, 1997f).

Further, the PAG comprises a major site for the polysynaptic activation of descending, inhibitory pathways to the DH (Basbaum and Fields, 1984; Fields and Basbaum, 1994; Le Bars, 1988; Millan, 1986, 1990, 1993, 1997).

Thus, although the precise, functional interrelationships between the limbic system, the thalamus and the cortex require further clarification, processes integrated in cortical and limbic structures are likely of crucial significance as concerns the cognitive and affective aspects of pain and its modulation. In this regard, monoaminergic and opioidergic systems are of particular relevance and demand a brief digression.

6.3.6. Pain and Mood: Focus on Monoamines and Opioids

As indicated above, several limbic and cortical regions targetted by ascending nociceptive information play key roles in the regulation of mood, and specifically in the induction of anxious (amygdala, hippocampus and PAG) and depressive (nucleus accumbens, septum, cingulate cortex and frontal cortex) states (Bloom and Kupfer, 1995; Graeff et al., 1993; Masterman and Cummings, 1997; Millan and Gobert, 1997; Noel and Gratton, 1995; Soares and Mann, 1997). In this light, the reciprocal relationship between mood and pain deserves accentuation. For example, anxiety often accompanies, and may intensify, the sensation of pain in response to an acute, noxious stimuli. On the other hand, in experimental models of 'conditioned fear', the anticipation of noxious stimulation may be sufficient to engage opioidergic mechanisms of antinociception: these fulfil an adaptive role in preparing and protecting the organism under conflict situations (Millan, 1986; Watkins et al., 1994, 1997b; Wiertelak et al., 1992b) (Section 2). Chronic pain states are frequently comorbid with, and exacerbated by, depression. However, the question of whether the efficacy of antidepressant agents in treating chronic pain reflects their intrinsic antidepressant properties, or the activation of antinociceptive mechanisms, is still under discussion (Abrams, 1996; Craig, 1994; Dellemijn and Fields, 1994; Eschalier et al., 1994; Millan, 1997; Romano and Turner, 1985; Ruoff, 1996). The distortion and erroneous processing of noxious and other forms of sensory information in psychiatric and geriatric patients is also well-known, while the rewarding properties of drugs may be modified under painful conditions (Suzuki et al., 1996).

Both the thalamus, as well as diverse regions of the cortex and limbic system involved in the integration of nociceptive information, possess an intense monoaminergic input and a crucial role of monoaminergic mechanisms in the modulation of mood and cognition is well-established (Bloom and Kupfer, 1995; Millan and Gobert, 1997; Oke et al., 1997). For example, the anxiolytic properties of α_{2} -AR agonists—which reduce adrenergic transmission via actions at α_{2A} -autoreceptors—are of importance to their clinical utility in the induction of anaesthesia (Havashi and Maze, 1993; Millan, 1997). An impressive illustration of the importance of mood in the experience and relief of pain is provided by μ opioid agonists, such as morphine. It has been suggested that the euphorigenic, mood-improving properties of morphine are fundamental to the unique quality of pain relief which it affords (Franklin, 1989; Millan, 1990). Indeed, a component of the analgesic actions of μ -opioids may be directly expressed in limbic and cortical structures (Burkey et al., 1996; Franklin, 1989). Contrariwise, κ-opioid agonists, which exert a negative impact upon mood (dysphoria), may not, despite their spinal antinociceptive actions, be able to engender equivalent pain relief (Millan, 1990) (Section 15.2). Such differential, affective actions of μ - vs κ -opioid agonists involve a contrasting facilitation and diminution, respectively, of the activity of mesolimbic dopaminergic pathways: this projection plays a pivotal, prohedonic role in processes of positive reinforcement (Altier and Stewart, 1997a; Di Chiara and Imperato, 1988; Millan, 1990; Noel and Gratton, 1995; Xi et al., 1998). Moreover, mesolimbic dopaminergic pathways originating in the ventrotegmental area are themselves activated by noxious stimulation and have been speculated to comprise an endogenous 'pain-suppression' system (Ma et al., 1993). Interestingly, in addition to μ -opioids, this projection is activated by an action of SP at NK₁ receptors-despite their opposite, pronociceptive role in the periphery and DH (Altier and Stewart, 1997b) (Sections 7.4.9 and 10.3). Other structures, such as the PBN, may also be involved in nociceptive processing, the control of mood and opioidergic mechanisms of reward (Bernard et al., 1994, 1995; Moufid-Bellancourt et al., 1996).

A positive and negative impact upon mood may, thus, respectively, attenuate and intensify the global sensation of pain. A further exploration of the interrelationships between nociceptive processing in corticolimbic regions and the roles of monoaminergic, opioidergic and other systems in the control of mood and cognition should yield important insights into mechanisms underlying the induction and management of clinical pain.

6.3.7. Sensory-Discriminative and Emotional-Cognitive Dimensions of Pain: a Summary of Complementary Roles

To summarize, then, it is reasonable to distinguish several circuits fulfilling contrasting and complementary roles in the detection and appreciation of pain (Berkley, 1997; Craig, 1994; Vogt and Gabriel, 1993) (Table 4). Certain of these, including the VPL/VPM-S I link, which receives a major input from the ventral STT via the VLF, play a decisive role in gauging the intensity, and determining the lo-

cation, of noxious stimuli (Kenshalo and Willis, 1991). This sensory-discriminatory component of nociception/pain is likely of adaptive value in triggering evasive action to a threatening noxious stimulus in the comparative absence of modulation by emotional and cognitive factors (Section 2.1). On the other hand, regions involved in the emotionalcognitive aspects of pain, such as the cingulate, prefrontal and insular cortices, receive a major input from the dorsal STT (and PSDC) via the VPI/VMpo and medial thalamus: they possess an extensive network of connections with other cortical and limbic regions involved in the control of pain, mood, cognition and attention. Certain of these structures, such as the amygdala, the PAG, the PBN and the lenticular nucleus, also receive nociceptive input from ascending pathways other than the ventral STT (Table 4). These circuits implicated in the affective dimension of pain will both influence, and be influenced by, emotional and cognitive factors. Rather than the detection of noxious stimuli per se, such circuits are of particular relevance to the ability to rationalize, cope with and tolerate pain, in particular long-term, clinical pain of diverse origins (Table 3). These structures are thus intimately associated with mood changes reflecting and modifying pain. The ability of μ -opioids to relieve pain (as echoed in the familiar anecdote, 'I can still feel it but it doesn't bother me any more') may be principally related to a modulation of their activity (Franklin, 1989; Millan, 1990).

Nevertheless, it would be simplistic to impose an absolute dissociation between the sensory-discriminative and emotional-cognitive dimensions of pain. These aspects should be regarded as complementary and as operating reciprocally and interactively rather than independently. Further, a 'pain centre' or 'pain centres' may not, as such, exist. Rather, a matrix of cerebral structures and multiple, parallel thalamocorticolimbic networks synergistically contributes to the global experience of pain (Apkarian, 1995b; Berkley, 1997; Coghill et al., 1994; Crick and Koch, 1998; Derbyshire et al., 1997; Dong et al., 1994; Parker et al., 1998; Xu et al., 1997b). Illustrative of this point, removal of the S I does not abolish clinical pain, an observation analogous to the recurrence of clinical pain despite surgical elimination of the VLF (Devinsky et al., 1995; Kenshalo and Willis, 1991) (Section 6.1.5). Precisely which supraspinal elements require manipulation for the selective alleviation of clinical pain, and how to achieve this, will require further evaluation. It is unlikely that the inactivation of any single component would suffice. Indeed, in line with the above assertion, recent imaging studies in man (Xu et al., 1997b) have indicated that pain involves complex neural networks (a 'neuromatrix') with a limited somatotopic organisation, implying that neurosurgical interventions in specific regions are unlikely to achieve complete and broad-based pain relief (Sections 6.1.5, 6.3.7 and 15.1).

6.4. Modulation of Descending Pathways by Ascending, Nociceptive Information

The roles of descending pathways to the DH are not the major subject of this review (Sections 5 and 10.8). Nevertheless, it should be noted that nociceptive input to the brain exerts a pronounced influence upon the activity of centrifugal pathways projecting to the DH of the spinal cord. This influence is expressed both directly and indirectly via complex circuits and mechanisms outlined above. For example, via a supraspinal loop, localized noxious, cutaneous stimulation can activate heterotropic mechanisms of descending inhibition which elicit a generalized antinociception in tissues innervated by other spinal segments. Such observations, which have been assimilated into the notion of 'diffuse noxious inhibitory controls' and various other conceptual models, are of relevance to 'referred' changes in nociception seen in tissue regions distinct from the site of noxious stimulation (Basbaum and Fields, 1984; Fields and Basbaum, 1994; Le Bars, 1988; Millan, 1997) (Sections 9.1.4.2 and 10.8). In the present context, it is of pertinence that the inactivation of descending inhibition, or the engagement of descending facilitation, may enhance the onward flow of nociceptive information to the brain (Section 10.8). Further, the existence of supraspinal loops triggered by noxious input underpins the potential importance of above-described, cerebrally-integrated events in modulating nociception at the level of the DH. Inasmuch as DH mechanisms can modulate activity in PAFs (Section 9.1.4.2), further, the activation of descending pathways by supraspinal nociceptive information completes a complex and extensive circuit whereby, at least theoretically, activity in nocisponsive PAFs might, via the DH, ascending pathways, supraspinal processes of integration, descending pathways and secondary DH mechanisms ultimately lead to an alteration in their own activity and/or that of other PAFs.

Thus, via modulation of mechanisms of descending inhibition and facilitation, nociceptive messages arriving in the brain may negatively or positively modify the subsequent access of nociceptive information to ascending pathways and cerebral structures (Section 10.8).

7. NOCICEPTIVE, INFLAMMATORY PAIN: PERIPHERAL PROCESSES

7.1. Nociceptive Pain Associated with Inflammation and Tissue Damage

Cutaneous tissue damage provoked by chemical irritants (such as capsaicin), heat injury, local electrical stimulation or, in some cases, dermatological disease, is associated with two principal zones of pain (Aloe *et al.*, 1997; Bessou and Perl, 1969; Davis *et al.*, 1993b; Hardy *et al.*, 1950; Lewis, 1942; Raja *et al.*, 1984; Simone *et al.*, 1989a,b, 1991; Treede *et al.*, 1992b). The first, termed the zone of 'primary' hyperalgesia, comprises the region of tissue damage itself and is characterized by spontaneous pain and an increased sensitivity to heat, mechanical (and chemical) stimuli. Surrounding this area is the unda-

maged zone of 'secondary' hyperalgesia which displays an increase in sensitivity to mechanical but not heat stimuli, and which may also manifest supersensitivity to cold stimuli. An increase in sensitivity implies a decrease in the stimulus threshold required to provoke a response, and an exaggerated response to a suprathreshold stimulus. Such changes are accompanied-though not invariably-by spontaneous pain (Bessou and Perl. 1969; Kilo et al., 1994). Primary hyperalgesia may involve a contribution of processes integrated in the CNS, but can predominantly be explained by events occurring at the level of peripheral nociceptors themselves. On the other hand, secondary hyperalgesia (and the phenomenon of mechanical allodynia in general) can predominantly be attributed to central mechanisms and is discussed in Section 9.

7.2. Primary Hyperalgesia: Thermal and Mechanical Hypersensitivity

As mentioned above, the primary hyperalgesia which accompanies tissue damage is characterized by a displacement of thermal and mechanical stimulus-response curves to the left (decreased threshold), an increased maximal effect of suprathreshold stimuli and tonic, stimulus-independent pain. This increased responsivity to heat stimuli appears to involve an enhanced sensitivity of individual, peripheral nociceptors and can also be seen in human skin (Dray and Urban, 1996; Reeh et al., 1986, 1987; Torebjörk et al., 1984; Treede and Magerl, 1995; Treede et al., 1992b). The class of nociceptor implicated appears to depend upon stimulus intensity and/or the type of skin, whether non-hairy (glabrous) or hairy. In the latter case, selective blockade of A fibres does not modify the time-course of sensitization suggesting that polymodal C fibres are primarily responsible (LaMotte et al., 1982; Treede et al., 1992b). In contrast, in non-hairy skin, the responsiveness of polymodal C fibres may even be diminished, whereas high threshold, 'Type I', mechanosensitive A δ fibres may develop an increased sensitivity to heat. This suggests that the latter are predominantly involved. In line with this argument, their blockade can block hyperalgesia to heat in non-hairy skin (Meyer and Campbell, 1981; Meyer *et al.*, 1994; Treede *et al.*, 1992b). The respective roles of C and A δ fibres require, thus, further characterization (Section 3.1.1). At the molecular level, there may exist multiple classes of nociceptor responsive to heat, including a recentlycloned vanilloid receptor (Caterina et al., 1997) (Section 7.4.5). In addition, studies of DRG neurones have revealed a potential multiplicity of heatactivated currents which only partially adapt upon repetitive heat stimulation, and which are potentiated by PGs and other algogenic agents (Belemonte and Cervero, 1996; Cesare and McNaughton, 1996; Lynn et al., 1996; Reichling and Levine, 1997).

Notwithstanding behavioural evidence for mechanical supersensitivity in primary hyperalgesia, neurophysiological evidence that a peripheral sensitization of *individual* mechanoceptors underlies *mechanical* hyperalgesia and allodynia is unconvincing, and only rare instances of chemical sensitization of some branches of mechanosensitive $(A\delta)$ fibres have been documented (Cline et al., 1989; Fitzgerald and Lynn, 1977; Raja et al., 1984; Reeh et al., 1987; Treede et al., 1992b). The question arises, thus, as to the underlying mechanisms. In some tissues, an increase in the sensitivity of polymodal C fibres might contribute to mechanical hyperalgesia. Further, is possible that the RFs of individual, mechanosensitive fibres may be expanded (spatial summation): that is, the number of mechanosensitive nociceptors stimulated per fibre (and, correspondingly, per DH neurone) is enhanced (Reeh et al., 1987; Thalhammer and LaMotte, 1982). In this regard, assuming that they become sensitive to pressure, the engagement of otherwise 'silent' nociceptors under tissue injury and inflammation might amplify the collective, afferent barrage to the DH in response to mechanical stimulation (Cohen and Perl, 1988; Davis et al., 1991; Koltzenburg et al., 1995; Kress et al., 1992; Schaible and Schmidt, 1988) (Section 3.1.3). That is, for the induction of mechanical hyperalgesia, the recruitment of silent nociceptors is equivalent to the adding of new units rather than to an increase in the sensitivity of existing units per se.

Nevertheless, it is difficult to extrapolate an involvement of the above-specified mechanisms, including silent nociceptors, to the mechanical allodynia which occurs *outside* the area of cutaneous tissue damage (secondary hyperalgesia), and which accompanies nerve injury. As discussed below (Sections 10 and 12), processes of *central* sensitization in the DH (and higher centres) offer a complementary theory for the induction of mechanical allodynia, a major component of which is mediated via $A\beta$ -fibres (Section 9.1.2).

7.3. Multiple, Interactive, Peripheral Mechanisms of Hyperalgesia

The primary hyperalgesia which ensues upon tissue injury can, then, at least partially, be accounted for by changes in the transduction sensitivity, responsiveness and activity of peripheral nociceptors and by the recruitment of 'silent' nociceptors. Processes underlying the activation and sensitization of nocisponsive PAF terminals are highly complex. They involve synergistic and supra-additive actions of a diversity of substances derived from nocisponsive PAFs themselves, damaged tissue, immunecompetent cells, the vasculature and sympathetic terminals (Dray and Perkins, 1993; Dray et al., 1994; Handwerker, 1991; Handwerker and Kobal, 1993; Keele and Armstrong, 1964; Levine and Taiwo, 1994; Reeh and Kress, 1995; Wilcox, 1993a; Wood and Docherty, 1997) (Figs 3 and 4, Table 5). Bradykinin (BK), 5-HT and PGs have long been implicated and, more recently, attention has been directed towards several novel, pro-nociceptive mediators, such as protons, cytokines and the neurotrophin, NGF. It is possible to distinguish several mechanisms whereby the function of nocisponsive PAFs is affected, and all may ultimately be mediated by alterations in the properties of ion channels controlling their electrical activity. Herein,

these mechanisms are termed 'excitation', 'sensitization', 'phenotype alteration' and 'indirect modulation'. The first two of these mechanisms concern actions expressed directly on sensory neurones and the last concerns actions mediated, by definition, via *other elements* of the inflammatory response. Phenotype alteration may be mediated either directly and/or indirectly (Cohen and Perl, 1988; Dray and Urban, 1996; Reeh and Kress, 1995; Levine and Taiwo, 1994; Taiwo and Levine, 1988, 1992).

The distinction between mechanisms of excitation as compared to mechanisms of sensitization is exploited herein as a descriptive framework. Generally-speaking, excitation of PAFs involves their rapid and direct depolarization following by spike activity and the induction of APs. In contrast, sensitization refers to a prolonged increase in the likelihood of firing in response to additional stimulation. Nevertheless, processes of excitation and sensitization are not necessarily separated either temporally or spatially, and may even be simultaneously triggered via common transduction mechanisms. On the other hand, specific pronociceptive agents may initiate both excitation or sensitization via interactions with multiple receptor types coupled to contrasting, intracellular transduction mechanisms.

The following account is based largely upon observations made in cutaneous tissue, but the basic events are of broader relevance to other organs (Cervero, 1995a; Gebhart, 1995; Mense, 1993; Schaible and Grubb, 1993).

7.4. Substances Modulating the Activity of Nocisponsive PAF Terminals: Tissue Sources, Receptors Implicated and Excitatory Actions

7.4.1. Orthodromic and Antidromic Activation of PAFs

The pronociceptive actions of algogenic substances on nocisponsive PAFs ultimately reflect their depolarization and the orthodromic transmission of impulses to the DH. Concurrently—corresponding to their efferent function—antidromic impulses may be triggered in collateral fibres ('axon reflex'). This provokes the peripheral release of EAAs, SP and other mediators which enhance nociception both by feedback actions on PAF terminals themselves, and by actions on other tissues involved in pronociceptive and pro-inflammatory processes (Section 7.9).

The actions of algogenic substances on nocisponsive PAFs, together with their tissue sources and the receptors involved in their actions, may be summarized as follows (Figs 3 and 4, Table 5).

7.4.2. BK

Tissue damage, inflammation and an acidic environment result in the activation of plasma and tissue, proteolytic kallikreins (Marceau, 1995). These enzymes generate the kinins, BK (the major product in plasma) and kallidin (the major product in tissue), from circulating polypeptide (kininogen) precursors (Marceau, 1995). BK and kallidin are rapidly metabolized to des-arg⁹-BK and des-arg¹⁰-

kallidin (Lys-des-arg9-BK), respectively. These, and other, metabolites, may also be involved in modulating nociception (Marceau, 1995). Intradermal infusion of BK, application of BK to skin blister bases or intravenous (i.v.) injection of BK evokes an immediate and intense pain in parallel with an excitation of polymodal C and mechanosensitive, high threshold $A\delta$ fibres in cutaneous and other tissues (Khan et al., 1992; Lang et al., 1990; Manning et al., 1991; Whalley et al., 1987). Two G-protein coupled receptor types exist: B_1 receptors, for which the des-arg derivatives have higher affinity than BK and kallidin themselves, and BK2 receptors, at which BK and kallidin are preferentially active (Hall, 1997; Marceau, 1995). Although B₁ receptors are not normally present in tissue (except in blood vessels), they play a pronociceptive role under conditions of prolonged inflammation following their induction by, for example, NGF and the cytokines, TNF α and IL-I β (Dray and Perkins, 1993; Dray and Urban, 1996; Petersen et al., 1998b; Raymond et al., 1996; Rueff et al., 1996; Seguin et al., 1995) (Section 7.7). Interestingly, in a model of persistent joint inflammation, B_1 and B_2 receptors were recently suggested to exert predominantly excitatory and sensitizing actions, respectively (Tonussi and Ferreira, 1997). Nevertheless, pharmacological approaches, as well as studies of transgenic mice, have indicated that B_2 rather than B_1 receptors normally mediate the phasic, algesic actions of BK on sensory neurones. Correspondingly, selective B₂ antagonists exert more marked antinociceptive properties than B₁ antagonists against pain of limited duration (Buritova et al., 1997; Haley et al., 1990; Heapy et al., 1993; Rupniak et al., 1997; Seabrook et al., 1997; Seguin et al., 1995; Whalley et al., 1987).

Studies of sensory neurones and cell lines suggest that the excitation (depolarization) of PAF terminals by B₂ receptors reflects a rapid increase in cation membrane conductance, primarily to Na (Burgess et al., 1989; Dray and Perkins, 1993; Dray and Urban, 1996). This action is likely initiated via stimulation of PLC leading to the generation of $[Ca^{2+}]_i$ and DAG which, via the synergistic activation of PKC, result in the phosphorylation (increased activity) of a Na⁺-permeable ion channel (Burgess et al., 1989; Cesare and McNaughton, 1996; Dray and Perkins, 1993; Dray et al., 1994 Higashida et al., 1986; McGuirk and Dolphin, 1992). The rise in $[Ca^{2+}]_i$ levels can potentially initiate a broad array of intracellular mechanisms modifying PAF activity, including neuropeptide release, PG synthesis, stimulation of NO synthase and enhancement of AC activity (Burgess et al., 1989; Dray and Urban, 1996; Gammon et al., 1989) (Section 7.5.2).

Although neuronal depolarization activates VDCCs, there are reports that, at least in heterologous expression systems, stimulation of B_2 receptors *reduces* the activity of N- and L-type VDCCs (Ewald *et al.*, 1988; Connor and Henderson, 1997). A role of a mitogen-activated PK has been implicated in this action (Wilk-Blaszczak *et al.*, 1998). Further, the increase in $[Ca^{2+}]_i$ elicited via cloned B_2 receptors has been shown to activate hyperpolarizing chloride channels (McEachern *et al.*, 1991). It is unclear whether such effects are of physiological importance in PAF terminals. However, they would counterbalance the stimulatory actions of BK and a

Nocisponsive C fibre terminal



Fig. 3. Influence of pronociceptive, inflammatory mediators upon the activity of nocisponsive, polymodal C fibres. Abbreviations are as indicated in the general list. In addition, BV, blood vessel. There is a reciprocal, facilitatory relationship between nocisponsive PAF terminals and the various components of the inflammatory response. Positive feedback actions of EAAs and SP at PAF terminals should be noted. The schema is non-exhaustive. Details of the relationships between various mediators and intracellular transduction mechanisms are given in Section 7. See also Fig. 4.

Nocisponsive C fibre terminal



Fig. 4. Roles of diverse receptors and intracellular signals in mediating the actions of pronociceptive mediators at nocisponsive, polymodal C fibre terminals. Abbreviations are as indicated in the general list. In addition, ASIC, acid-sensing-ion-channel. The mechanisms indicated relate to common modes of action, and not necessarily to common loci of action. For example, Ca^{2+} -permeable ion channels coupled to vanilloid receptors are distinct from Ca^{2+} -permeable ion channels coupled to NMDA receptors. Dotted lines are given where interactions may be indirect and/or remain uncertain. Actions of ATP at metabotropic P_{2Y2} receptors, potentially involved in mediating mechanical allodynia, are exerted at $A\beta$ rather than C fibres, but are included for comparative purposes. Roles of multiple EP receptors for PGs are indicated, but their respective contributions remains to be elucidated. A modulation of gene transcription probably plays a role in the longer-term actions of several mechanisms indicated, as well as those of NGF/TRK A receptors (not illustrated) (Section 7).

Table 5. Summary of principal mechanisms underlying the induction of prolonged, painful states due to noxious stimulation of nocisponsive PAFs (see Section 10)

- Sensitization and increased excitability of DH neurones (PNs and EXINs) transmitting nociceptive information to higher centres. The engagement of NMDA receptors by activation of small calibre, C fibre input plays a particularly important role in triggering intracellular mechanisms (increases in [Ca²⁺]_i, PK activity, NO synthesis, IEG expression, etc. underlying these changes
- 2. Diminution and accentuation of descending mechanisms of nociceptive inhibition and facilitation, respectively
- 3. Adaptive changes in the thalamus, cortex and other higher centres responsible for the discriminative-sensory and affective-cognitive dimension of pain

A. Peripheral mechanisms

^{1.} Direct excitation of nocisponsive PAFs. Mediated via, for example, 5-HT₃, P_{2X3} and vanilloid receptors coupled to excitatory, cation-permeable ion channels

Sensitization of nocisponsive PAFs by the engagement of intracellular transduction systems: includes the activation
of AC and PLC, as well as increases in [Ca²⁺]_i. Thereby, the likelihood, intensity and duration of further discharges
is increased. Mediated via, for example, PG and BK receptors

Modulatory events involving a complex pattern of reciprocal interactions amongst PAF terminals, glial cells, immunocompetent cells, sympathetic terminals and other elements of the inflammatory response. Mediated via, for example, cytokines, NO and NGF

^{4.} Altered phenotype of PAFs. For example, an increase in the synthesis of SP and of specific, excitatory Na⁺-channel subtypes

^{5. &#}x27;Antidromic' activation of nocisponsive PAFs. This further enhances their activity by a positive feedback mechanism, and elicits vascular effects and other processes underlying NI. Mediated by, for example, SP (especially NI), CGRP (especially vasorelaxation) and EAAs (especially feedback actions). This mechanism is of relevance to events in cerebral blood vessels underlying migraine headaches

B. Central mechanisms

further consequence of the increase in $[Ca^{2+}]_i$ levels elicited by B₂ receptors may be the initiation of Ca²⁺-dependent processes for their desensitization by dephosphorylation (Kress and Reeh, 1996). Together with the putative activation by $[Ca^{2+}]_{i/}$ PKC of an ATPase-dependent pump for Ca^{2+} extrusion (Young et al., 1998), such processes would further counteract the stimulatory actions of BK at PAF terminals, Nevertheless, it has been questioned whether, under physiological conditions, desensitization to the actions of BK is pronounced in PAF terminals (Kress and Reeh, 1996). Thus, even though BK is rapidly metabolized, the generation of active metabolites (vide supra), its ongoing synthesis and the progressive development of B_1 receptormediated mechanisms of nociception (Section 7.7) suggest that BK plays a major role in direct and indirect mechanisms underlying PAF activation and primary hyperalgesia (Rueff et al., 1996).

It should be mentioned that both kinins and BK receptors exist in the CNS. In the DH, BK receptors (probably of the B_2 type) are found on PAF terminals, intrinsic neurones and terminals of descending pathways. A possible role of these central pools of BK in the modulation of nociception is currently under exploration (Coggeshall and Carlton, 1997; Walker *et al.*, 1995).

7.4.3. Prostaglandins and Leukotrienes

As summarized in Section 3.2.8.4, arachidonic acid is metabolized by several constitutive and inducible isoforms of COX to generate PGs, on the one hand, and via lipoxygenase enzymes, to generate leukotrienes, thromboxanes and hydroxyeicosotetranoic acid derivatives, on the other.

Although PGs may be derived from virtually all tissue types, under conditions of tissue inflammation and nerve injury, immunecompetent cells and sympathetic terminals may be particularly important sources (Levine and Taiwo, 1994; Levine et al., 1986b; Miao et al., 1996; Taiwo and Levine, 1988). A contribution of the constitutive COX 1 enzyme to pro-nociceptive pools of PGs should not be discounted. However, activation of the (predominantly) inducible COX 2 synthase is likely the principal pathological source of PGs interacting with nocisponsive PAFs upon inflammatory tissue damage (Buritova et al., 1996; Dirig et al., 1997; Dray and Urban, 1996). Factors potentially responsible for the induction of COX 2 include NO (Section 7.7) and immune cell-derived cytokines, such as IL-1 and TNF α , although certain other cytokines, such as IL-4 and IL-10, may actually inhibit COX 2 activity and reduce nociception/inflammation (Appleton, 1997; Bakhle and Botting, 1996; Pairet and Engelhardt, 1996; Perkins and Dray, 1996. Wagner et al., 1998).

Of the many species of PG known, the major contributors to hyperalgesia, PGE_2 and PGI_2 (prostacyclin)—which may retrogradely modify COX 2 activity—act via (multiple) EP receptors and IP receptors, respectively (Appleton, 1997; Bakhle and Botting, 1996; Boie *et al.*, 1997; Coleman *et al.*, 1994; Khasar *et al.*, 1995; Matsumura *et al.*, 1995; Schaible and Grubb, 1993) (Section 3.2.8.4).

Although application of PGs to skin is not usually painful, high levels of PGE2 and PGI2 may contribute to excitatory mechanisms through the suppression of a K^+ -conductance and the increase in a Na⁺- (and Ca²⁺)-conductance. These actions may lead to an increase in neuropeptide release from C fibre terminals (Birrell *et al.*, 1991; Collins and Davies, 1998; Nicol *et al.*, 1992, 1997; Puttick, 1992; Sánchez-Prieto et al., 1996; Schaible and Schmidt, 1988; Schepelmann et al., 1992; Wang et al., 1996a) (Sections 7.5.2.3 and 7.5.2.4). Further, as discussed below (Section 8.2.1), following nerve damage, PGs can excite C fibres in neuromas. The receptor subtypes mediating such effects remain unclear, but actions at EP/IP receptors positively coupled to activation of $[Ca^{2+}]_i$ and AC and/or PLC are, presumably, involved (Coleman *et al.*, 1994). Interestingly, PGI₂-deficient, transgenic mice were recently shown to display a reduction in peripheral inflammation and nociception, consistent with a role of PGI_2 in mediating nociception (Murata et al., 1997).

Comparatively little is known about the putative, pronociceptive actions of leukotriene (and thromboxane) products of the lipoxygenase (non-COX) pathway of arachidonic acid metabolism. However, leukotriene B_4 (derived from the 5-lipoxygenase arm), which accumulates in inflamed tissue, appears to trigger the release of 8R, 15S, -diHETE (derived from the 15-lipoxygenase arm), from polymorphonuclear leukocytes. This leukotriene subsequently interacts with nocisponsive cutaneous C fibre terminals to elicit hyperalgesia via an, as yet, unclear mechanism (Amann *et al.*, 1996; Bisgaard and Kristensen, 1985; Levine and Taiwo, 1994; Levine *et al.*, 1984, 1986a; Martin, 1990; Martin *et al.*, 1988; White *et al.*, 1990) (Section 7.7).

7.4.4. ATP

ATP can elicit marked pain upon infusion into the skin, apparently due to the direct excitation of a subset of small calibre PAFs (Bland-Ward and Humphrey, 1997; Burnstock, 1996; Burnstock and Wood, 1996; Rang et al., 1991). These actions of ATP involve the stimulation of rapidly-desensitizing, cation-permeable, *ionotropic* receptors of the P_{2X} family. Their engagement results in an immediate depolarization (reflecting an inward cation flux) and an increase in $[Ca^{2+}]_i$ (Bouvier *et al.*, 1991; Robertson et al., 1996; Sawynok and Reid, 1997; Thorne and Housely, 1996). MRNA encoding at least six of the seven members of the P_{2X} family is found in the DRG and trigeminal ganglia, to a variable degree colocalized with SP and CGRP. However, only the P_{2X3} receptor subtype is expressed exclusively in small calibre, sensory neurones (Burnstock, 1996; North and Barnard, 1997; Vulchanova et al., 1997). Indeed, P_{2X3} receptors are localized on a capsaicin-sensitive population of fibres projecting specifically to lamina II which contain little or no CGRP, SP or receptors for NGF, and which are defined by presence of the lectin, IB-4 (Vulchanova et al., 1996) (Section 3.2.1). Although excitatory actions of ATP at P2X2 receptors and other classes of P_{2X} receptor should not be discounted, (King et al., 1997b; Simon et al., 1997),

 P_{2X3} receptors may be of particular importance in the rapid, excitatory actions of ATP on nocisponsive PAF terminals (Chen *et al.*, 1996; Cook *et al.*, 1997; Guzman *et al.*, 1997; Lewis *et al.*, 1995; Collo *et al.*, 1996; North, 1996).

 $P_{2{\bf X}3}$ receptors rapidly desensitize, possibly due to a $Ca^{2+}\mbox{-}$ (and/or calmodulin)-dependent dephosphorylation mediated by the phosphatase calcineurin (King et al., 1997b; North, 1996; Vulchanova et al., 1996). P_{2X2} receptors do not, in contrast, rapidly desensitize and the co-expression of P_{2X2} and P_{2X3} receptor subtypes in oocytes produces heteromultimeric channels with slower deactivation kinetics than P_{2X3} receptors alone. Inasmuch as P_{2X2} and P_{2X3} receptors are co-localized in some DRG cells, and such heteromultimeric channels might be assembled in vivo, this observation may be of relevance to longer-term, sensitizing actions of ATP (Bland-Ward and Humphrey, 1997; Lewis et al., 1995; North and Barnard, 1997; Vulchanova et al., 1997; Werner et al., 1996). Indeed, there is preliminary evidence for similar, slowly-desensitizing actions of ATP at endogenous receptors on rat DRG neurones (Sansum et al., 1997; Vulchanova et al., 1997).

An additional aspect of potential relevance to pain and allodynia is the presence of ATP-excitable, *metabotropic* P_{2Y} receptors (coupled to PLC and an increase in $[Ca^{2+}]_i$) on *large*—but not small-calibre PAFs (Collo *et al.*, 1996; Khakh *et al.*, 1997; Mayer *et al.*, 1997; Svichar *et al.*, 1997a). Indeed, mRNA encoding a G-protein-coupled, metabotropic P_{2Y} receptor was recently extracted from sensory neurones and shown to render oocytes mechanosensitive to light touch (Nakamura and Strittmatter, 1996; Svichar *et al.*, 1997b). A putative, pathophysiological role of P_{2Y} sites in mediating mechanical allodynia would, thus, be of interest to evaluate.

Although ATP is universally present in cells, and may be derived from multiple tissue sources, several, specific pools may generate those high levels of ATP which pathologically activate nocisponsive terminals (Burnstock, 1996). For example:

- tumour cells contain high levels of ATP and may be implicated in the activation of PAFs in cancerous tissue;
- vascular, endothelial cells and/or platelets may be a source of pronociceptive ATP in migraine, angina and ischaemic muscle pain (Crea *et al.*, 1990); and
- 3. sympathetic nerve endings, which release ATP as a co-transmitter to NAD, might be involved in a variety of conditions including cutaneous, joint and muscle inflammation, as well as PAF damage (Guieu *et al.*, 1996).

Indeed, sympathetically-maintained pain is sensitive to manipulations, such as guanethidine treatment and surgical sympatheticomy, which block release from sympathetic terminals, although other sympathetic mediators, such as NAD and PGs, are also involved in the modulation of nociception by sympathetic fibres (Section 8.2.4).

ATP is rapidly metabolized to adenosine which modulates nociception via several receptor types, and which may interact with the pronociceptive effects of ATP at P_{2X2}/P_{2X3} receptors (Aley and Levine, 1997; Dray and Urban, 1996: Doak and Sawynok, 1995). Activation of A_{2A} receptors, which are positively coupled to AC, appears to be pronociceptive. In contrast, activation of peripheral A_1 receptors, which inhibit cAMP formation, and which increase and decrease K⁺- and Ca²⁺-currents, respectively, appears to be antinociceptive, in analogy to the role of spinal A_1 sites (Aley and Levine, 1997; Dray and Urban, 1996; Doak and Sawynok, 1995; Taiwo and Levine, 1991) (Sections 3.2 and 7.5.2). In line with a pronociceptive role of (peripheral) A_{2A} receptors, knock-out mice lacking these sites display hypoalgesia (Ledent *et al.*, 1997).

The importance of Schwann cells as a source of inflammatory mediators and factors controlling the phenotype and regeneration of damaged PAFs is underscored in Sections 7.7, 8.2.6 and 11.3.4.1 (Bolin *et al.*, 1995). It is, thus, intriguing to note that Schwann cells bear metabotropic P_{2Y2} receptors as well as ionotropic P_{2X} receptors. These findings may be compared to reports of P_{2X2} and P_{2X7} receptors on CNS-localized microglial cells (Chessel *et al.*, 1997a,b; Collo *et al.*, 1997; Mayer *et al.*, 1997; Nörenberg *et al.*, 1997) (Section 4.8). A potential, functional role of Schwann cell-localized P_{2Y} and P_{2X} sites in the modulation of nociception would be of interest to evaluate.

7.4.5. Protons and Vanilloids

7.4.5.1. Excitation of PAF terminals by acidosis (a reduction in pH)

Although an increase in the extracellular concentration of K^+ ions has long been considered an excitatory component of inflammatory exudates, the actions of K^+ may be non-specific. More recently, attention has been directed towards the potential importance of a further ion, H^+ (protons), as an algogenic agent for nocisponsive PAF terminals (Bevan and Geppetti, 1994; Waldmann *et al.*, 1997a,b; Wood and Docherty, 1997). Under certain conditions, nociceptors may be exposed to an environment sufficiently acidic to trigger their excitation. For example:

- inflammatory exudates (in which immigrating leucocytes provide an important source of protons);
- 2. ischaemic muscle, in which low pHs may be attained under intensive exercise;
- 3. cardiac tissue following an infarction;
- 4. synovial fluid of arthritic joints;
- 5. haematomas provoked by fractures; and
- 6. tissue surrounding tumours.

Further, disruption of the mucosal lining of the gut could expose intestinal tissue to pHs as low as 3.0, while damage to the urothelium might expose renal tissue to acidic urine (Bevan and Geppetti, 1994; Steen *et al.*, 1995). Application of acidic solutions (pH \leq 5.0) to the skin is painful and the activation of PAFs by protons may, in accordance with the above observations, contribute to inflammatory pain, muscle cramps, gastric ulcers and cystitis, etc. (Steen *et al.*, 1992; Steen and Reeh, 1993).

Studies of isolated sensory neurones suggest that protons depolarize PAFs by at least two mechanisms (Bevan and Yeats, 1991).

- 1. In a majority of neurones—though these are *not* all necessarily nocisponsive—comparatively mild, physiological changes in pH (down to 7.0) may trigger a transient, rapidly inactivating Na⁺-permeable ion channel.
- In a smaller population of neurones involved in pain transmission, more pronounced changes in pH directly lead to the opening of poorly-selective ion channels permeable to Na⁺, K⁺ and Ca²⁺ (Bevan and Geppetti, 1994; Docherty *et al.*, 1991; Liu *et al.*, 1997c; Wegner, 1996; Wood and Docherty, 1997; Zeilhofer *et al.*, 1996, 1997). This is associated with an inward current and a *sustained* depolarization over minutes (slow inactivation).

7.4.5.2. Relationship of proton-gated ion channels to vanilloid-Gated ion channels

Although some neurones apparently respond with excitation to either protons or capsaicin (and other so-called vanilloids), there is very considerable overlap, if not identity in these populations (Bevan and Geppetti, 1994; Steen et al., 1992; Wood and Docherty, 1997). Further, like protons, capsaicin provokes intense pain by opening cation-permeable ion channels and there exists several similarities in the actions of protons and capsaicin on nocisponsive neurones (Bevan and Geppetti, 1994). Thus, it has been suggested that protons and vanilloids may, via a common binding domain, co-activate a mutual ion channel localized on cutaneous, visceral and other types of nocisponsive PAF (Bevan and Geppetti, 1994; Szallasi, 1997; Winter et al., 1995). Protons might, then, be conceived of as 'mimics' or ' ligands' of vanilloid receptors, for which endogenous agonists remain unknown (Wood and Docherty, 1997). The possibility of a common site of action is supported by observations that a low pH inhibits binding of [³H]resiniferatoxin (a capsaicin analogue) to vanilloid receptors (Szallasi and Blumberg, 1996). Further, the capsaicin antagonist, capsazepine, prevents protons from opening a population of cationsensitive ion channels (Liu and Simon, 1994). However, the following counter-arguments may be advanced:

- capsazepine does *not* block all responses to protons (Bevan and Geppetti, 1994; Bevan *et al.*, 1993; Liu and Simon, 1994; Steen *et al.*, 1992);
 capsaicin-gated channels are preferentially Ca²⁺-
- capsaicin-gated channels are preferentially Ca²⁺permeable in contrast to their proton-gated counterparts (Zeilhofer *et al.*, 1997);
- in cultured DRG neurones, vanilloid-gated channels are *not* activated by protons (Oh *et al.*, 1996); and
- 4. Ca^{2+'}-influx lead to the rapid *desensitization* of vanilloid—but not proton-gated receptors, probably via a calcineurin-dependent dephosphorylation triggered by an increase in [Ca²⁺]_i (Docherty *et al.*, 1996; King *et al.*, 1997b; Koplas *et al.*,

1997; Wood and Docherty, 1997; Zeilhofer *et al.*, 1997).

In addition, it has been demonstrated that capsazepine behaves as an antagonist at VDCCs suggesting caution in the interpretation of its actions (Docherty *et al.*, 1997). A possible role of an intervening chemical transmitter in the excitatory actions of protons cannot, further, be entirely excluded (Dray, 1996; Lou and Lundberg, 1992).

7.4.5.3. Properties of cloned, vanilloid-gated, as compared to cloned, proton-gated, ion channels

The above observations tend to question the concept of a single and common site mediating the actions of protons and vanilloids. In fact, an important insight into the relationship between proton and vanilloid receptors was recently afforded by the cloning of an acutely-desensitizing, vanilloid receptor from sensory neurones, the mRNA encoding which is found almost exclusively in the DRG and trigeminal ganglia (Caterina et al., 1997). This vanilloid receptor couples to an ion channel preferentially permeable to Ca^{2+} -ions, and prolonged exposure to capsaicin kills cells expressing these receptors. (This suggests, in analogy to in vivo studies, that a progressive loss of responsiveness to vanilloids may reflect not only acute events but also, and ultimately, the destruction of neurones by excessive Ca²⁺-influx). A further important feature of cloned vanilloid receptors is their activation by increases in temperature. This indicates that the vanilloid channel may transduce noxious thermal stimuli, consistent with the burning sensation provoked by cutaneous application of capsaicin, and with the co-expression of heat and capsaicinevoked inward currents in cultures of DRG neurones (Kirschstein et al., 1997; Caterina et al., 1997)-although other types of (BK and/or PG facilitated) heat-sensitive channel may also exist on small calibre PAFs (Cesare and McNaughton, 1996; Reichling and Levine, 1997) (Section 3.1.1). Notably, the response of cloned vanilloid receptors both to capsaicin itself and to heat is facilitated by protons. This latter observation is consistent with an action of protons at vanilloid receptors. However, these findings do not exclude the putative existence of other vanilloid receptor types either directly activated by protons alone, or completely insensitive thereto. Indeed, it has been suggested that subtypes of vanilloid receptor may exist in several tissue types. For example, in muscle, at peripheral as compared to central terminals of PAFs and, most pertinently, on the peripheral terminals of PAFs themselves (Acs et al., 1997; Griffiths et al., 1996; Wood and Docherty, 1997). In addition, the lack of direct activation of cloned vanilloid receptors by protons suggests that other, proton-sensitive channels may exist.

Indeed, a novel, proton-gated Na^+ -channel specific to sensory neurones has been cloned and termed 'DRASIC' (Waldmann *et al.*, 1997a,b). This channel responds to reductions in extracellular pH with both a rapidly-inactivating current

and a further, sustained Na+-mediated current which clearly resembles the prolonged kinetics of currents on PAF proton-gated terminals. However, native, proton-gated ion channels in the DRG are *not* selective for Na⁺. This difference raises the possibility of a post-transcriptional modification of 'DRASIC' ion channels in situ, the existence of other acid-sensitive channels or a heteromultimeric association of several different receptor subunits. Indeed, several types of acidsensitive channel have been cloned and detected in nocisponsive PAFs (Bassilanat et al., 1997; Waldmann et al., 1997b). Further, the assembly of functional channels from a combination of subunits of the DRASIC channel together with those from a further 'MDEG2' splice variant channel which co-exists in DRG yielded sustained currents discriminating poorly between Na⁺ and K⁺ (Lingueglia et al., 1997). That is, two characteristic features of native, proton-gated channels mediating the tonic sensation of pain (Bevan and Geppetti, 1994).

It remains to be determined whether, apart from protons, other *endogenous* ligands exist for vanilloid—or proton-gated ion channels, of which additional subtypes will likely be identified.

Interestingly, NGF may modulate (up-regulate) the functional status of both proton-coupled and vanilloid-coupled ion channels on small calibre, cutaneous PAFs (Bevan and Winter, 1995; Winter *et al.*, 1993). Further, BDNF (but not NGF) modulates the capsaicin-sensitivity of sensory vagal neurones (Winter, 1998).

7.4.5.4. Influence of protons (tissue acidosis) upon other receptor types

Protons may modify activity at other sites controlling PAF excitability. Thus, tissue acidosis considerably enhances nociceptor sensitivity to inflammatory mediators such as BK and 5-HT (Kress et al., 1997). This synergism between protons, BK and 5-HT may be mutual inasmuch as both BK and 5-HT themselves reinforce actions at proton-(and/or vanilloid-) gated ion channels, possibly via intracellular mechanisms involving PLC (Damas et al., 1997; Kress et al., 1997; Steen et al., 1995). An influence of tissue acidosis upon the actions of ATP has also been demonstrated inasmuch as protonation enhances the activity of agonists at recombinant P_{2X2} receptors, a subtype found on nociceptive C fibres (King et al., 1997b) (Section 7.4.4). However, protons appear to reduce rather than increase activity at P_{2X3} sites (King et al., 1997b). Further, a low pH may exert other effects inhibiting PAF excitability inasmuch as NMDA- and VDCC-mediated currents are reduced under some circumstances (Philippi et al., 1995).

These data suggest that acidosis exerts a complex pattern of influence upon the actions of diverse agents modulating inflammatory states, PAF excitability and nociception. Nevertheless, the predominant effect of protons/low pH at nocisponsive PAF terminals is a slowly-adapting increase in excitability mediated via proton-sensitive receptors coupled to cation-permeable ion channels and provoked in synergy with thermal stimuli and other inflamma-tory mediators.

It should be mentioned that vanilloid binding sites are likewise localized on the central terminals of fine calibre PAFs in the DH. They also exist in the thalamus and other cerebral structures involved in nociceptive processing (Acs et al., 1996; Szallasi et al., 1995). Further, tissue acidosis may be provoked in the CNS under conditions which are associated with pain: for example, ischaemic tissue damage and seizure-like neuronal activity. These observations suggest that localized alterations in pH may also influence nociception via the modification of synaptic transmission in the DH and suprapinal structures. An interesting potential target of protons in the CNS is provided by GABAA receptors, the activity of which way be inhibited or enhanced by protons depending upon their subunit composition (Krishek et al., 1996).

7.4.6.5-HT

DRG express mRNA encoding several types of 5-HT receptor. Correspondingly, nocisponsive PAFs may be excited by peripheral pools of 5-HT derived from platelets, mast cells and endothelial cells as well as, in the case of the cerebral vasculature, the perivascular terminals of central serotoninergic fibres originating in the raphe nucleus (Millan, 1995; Pierce et al., 1997) (Section 7.9). For example, 5-HT released from aggregating platelets in the diseased heart may be a source of cardiac pain (Meller and Gebhart, 1992). Sympathetic terminals may also provide a source of 5-HT inasmuch as they can take up and release 5-HT. Administration of 5-HT onto the skin (blister base) elicits pain in man. Further, the pain and pseudoaffective responses elicited by i.v. administration of 5-HT reflect pronociceptive actions in the heart and other internal organs (Lang et al., 1990).

The excitation of nocisponsive, polymodal C and A δ PAFs by 5-HT appears to involve the activation of 5-HT₃ receptors directly gating ion channels preferentially permeable to Na^+ (and K^+). The ensuing neuronal depolarization triggers, in turn, the opening of (possibly L-type) VDCCs (Abbott et al., 1996, 1997; Barann et al., 1993; Ebersberger et al., 1995; Grubb et al., 1988; Guilbaud et al., 1989a; Millan, 1995; Smith et al., 1997b; Todorovic et al., 1997). In analogy to B_2 receptors, 5-HT₃ receptors are positively coupled to PLC and they may also initiate changes in PAF activity via intracellular mechanisms involving DAG-induced activation of PKC and IP₃-induced increases in [Ca²⁺]_i. The influx of Ca²⁺ via VDCCs triggers, moreover, a further increase of $[Ca^{2+}]_i$ levels via activation of ryanodine receptors localized on the endoplasmic reticulum: this results in a further mobilization of intracellular Ca²⁺-stores (Rondé and Nichols, 1997) (Sections 7.5.2.8 and 10.4.1). 5-HT₃ antagonists possess antinociceptive properties in several models of inflammatory pain although, in clinical studies of inflammatory bowel disease and migraine, at best equivocal results have been obtained (Giordano and Sacks, 1997; Greenshaw and Silverstone, 1997; Talley, 1992). The recent discovery of mRNA

encoding many other 5-HT receptor types in DRG suggests the need for further study of the excitatory actions of 5-HT on nocisponsive PAFs (Millan, 1997; Pierce *et al.*, 1996, 1997).

7.4.7. NAD Derived from the Sympathetic System

Under conditions of inflammation and nerve injury (Section 8.2.4), the release of NAD (and adrenaline) by the sympathetic system plays a role in the modulation of nociception. The release of NAD controlled by a multiplicity of mechanisms including GLU, NO, BK, PGs, 5-HT, NPY, cytokines and NAD itself (Carlton et al., 1998a; Jonakait, 1993; Miao et al., 1996; Munglani et al., 1996; Seabrook et al., 1997). Even high concentrations of NAD do not generally excite PAF neurones (Lang et al., 1990), but there is functional evidence that NAD directly and indirectly enhances the activity of cutaneous, muscle and joint PAFs under inflammatory conditions (Hu and Zhu, 1989; Mense, 1993; Sato and Perl, 1991; Sato et al., 1993a, 1994; Schaible and Grubb, 1993).

Several studies have implicated α_1 -ARs in mediating inflammatory hyperalgesia, for example, that provoked by capsaicin, and a role of the α_{1A} -AR subtype was proposed to mediate formalin-induced 1996: nociception in the rat (Drummond, Drummond et al., 1996; Hong and Abbott, 1996; Hunter et al., 1996; Kinnman and Levine, 1995; Kinnman et al., 1997; Millan, 1997; Ouseph and Levine, 1995). A pronociceptive role of α_1 -ARs would be consistent with their positive coupling to PLC and increases in $[Ca^{2+}]_i$. Further, although a *direct* action of NAD at α_1 -ARs localized on PAFs is inconsistent with anatomical evidence suggesting that PAFs express few α_1 -ARs, they may be upregulated by tissue inflammation (Pieribone et al., 1994). On the other hand, vascular α_1 -ARs might be involved in an (indirect) influence of NAD upon PAF terminals (Section 8.2.4). As concerns α_2 -ARs, DRG express mRNA encoding $\alpha_{2A}\text{-},\,\alpha_{2B}\text{-}$ and $\alpha_{2C}\text{-}$ ARs (Gold et al., 1997), suggesting that they might modulate the activity of PAF terminals (Dray et al., 1994). However, experimental evidence for a direct, pronociceptive role of α_2 -ARs at PAF terminals under inflammatory conditions is poor, consistent with *their* negative coupling to cAMP and Ca^2 channels. Indeed, α_{2B} -ARs at PAF terminals may exert antinociceptive actions (Aley and Levine, 1997; Hunter et al., 1996; Khasar et al., 1995)-and a peripheral, antinociceptive action of α_2 -AR agonists via a local release of opioids has been reported by Nakamura and Ferreira (1988). Nevertheless, α_2 -ARs were recently reported to reduce K⁺-currents in the terminals of damaged PAFs, providing a potential mechanism of excitation (Abdulla and Smith, 1997). Additional pharmacological characterization of the nature of *α*-AR sites enhancing PAF activity under conditions of inflammation is needed (Section 8.2.4.2).

Recent clinical studies have provided support for a pronociceptive, peripheral role of adrenergic mechanisms in showing that the ongoing pain and mechanical allodynia associated with capsaicin application to the skin was inhibited by phentolamine (Kinnman *et al.*, 1997; Liu *et al.*, 1996). The underlying mechanism is unclear. Some $A\beta$ fibres possessing α_2 -ARs and mediating mechanical allodynia may be excited by NAD (Chen *et al.*, 1996; Gold *et al.*, 1997; Liu *et al.*, 1996; Pieribone *et al.*, 1994). However, NAD may act at sensitized $\alpha_{1/2}$ -ARs on fine calibre PAFs. This would enhance the peripheral drive to sensitized DH neurones mediating the allodynic effects of $A\beta$ -fibre stimulation (Liu *et al.*, 1996) (Section 9.1.2). In addition, C fibres themselves mediate certain forms of mechanical allodynia (Section 9.2).

Apart from direct actions at PAF terminals, the possible, indirect modulation of nociception via an influence of NAD upon inflammatory events *per se*, or via cardiovascular actions (vasoconstriction), is further discussed in Section 8.2.4. Potential, peripheral actions of NAD/ α -ARs are importance to clarify in the light of:

- 1. the potential role of sympathetic mechanisms in the pathology of inflammatory states (Miao *et al.*, 1996; Takahashi *et al.*, 1993); and
- 2. the involvement of 'sympathetic' mechanisms in changes in nociception accompanying PAF damage (Section 8.2.4).

Nevertheless, the *clinical* significance of sympathetic mechanisms in inflammatory nociception still requires further evaluation and, in a recent study, the pain and hyperalgesia elicited by a heat injury of skin in man was not modified by a sympathetic block (Pedersen *et al.*, 1997).

7.4.8. Histamine

Although histamine, which is derived from mast cells, is a familiar component of inflammatory media, its actions are poorly understood. Generally, histamine elicits the sensation of itch rather than pain. However, higher concentrations may be painful (Carstens, 1997; LaMotte et al., 1987; Levine and Taiwo, 1994; Rang et al., 1991). Further, exposure to histamine may release SP and CGRP from PAF terminals (Bileviciute et al., 1997b) and activate nocisponsive neurones in the DH in a manner similar to that observed with cutaneous noxious stimuli (Chapman and Besson, 1997; Yao et al., 1992). The direct, excitatory (and sensitizing) actions of histamine on PAFs are likely mediated by H₁ receptors positively coupled to PLC and to an increase in $[Ca^{2+}]_i$: this may secondarily result in the engagement of VDCCs and an induction of NO synthesis (Hill et al., 1997; Tani et al., 1990). A (possibly synergistic) role of H₂ receptors, which are positively coupled to AC and PKA, should also be mentioned (Hall, 1997) (Section 7.9).

7.4.9. EAAs, Tachykinins and CGRP

An increase in GLU levels—derived from PAF terminals themselves and, perhaps, from immunocompetent cells—has been observed in inflamed tissue (Jeftinija *et al.*, 1991; Nordlind *et al.*, 1993; Piani *et al.*, 1991; Westlund *et al.*, 1992; Wood and Docherty, 1997). There are several reports that GLU elicits hyperalgesia via the direct excitation of PAF terminals (Carlton et al., 1995; Davidson et al., 1997; Jackson et al., 1995; Wang et al., 1997b). Indeed, studies of mRNA expression in DRG, together with pharmacological analyses of responses to EAAs, have indicated that the peripheral terminals of small calibre PAFs possess NMDA, AMPA and kainate receptors (Ault and Hildebrand, 1993; Carlton et al., 1995; Coggeshall and Carlton, 1997; Huettner, 1990; Lawand et al., 1997b; Liu et al., 1994b, 1997b; Lovinger and Weight, 1988; Sato et al., 1993b; Wood and Docherty, 1997: Zhou et al., 1996a). Warncke et al. (1997a,b) have, further, shown that local administration of the NMDA channel blocker, ketamine, inhibits primary and secondary hyperalgesia of cutaneous tissue in human subjects suggesting that peripheral NMDA receptors may participate in pronociceptive processes in man. A direct, excitatory, positive-feedback action of GLU on PAF terminals would be consistent with evidence that activation of NMDA receptors triggers SP release from their central endings (Liu et al., 1994b, 1997b) (Section 3.2.4). One component of this central action of NMDA may be mediated via NO from PAF terminals (Sorkin, 1993) and NO may also intervene in the peripheral, PAF-mediated actions of GLU (Jackson et al., 1995; Lawand et al., 1997b; Wang et al., 1997b). Further, an action of GLU at NMDA receptors vasodilates cerebral microvessels via an indirect, NO-mediated mechanism (Fergus and Lee, 1997) Thus, it would be of interest to establish whether a GLU-mediated release of NO may be involved in the vasodilation of subdural CBVs which is implicated in migraine headaches (Section 7.9.3). The discovery of NMDA receptors on postganglionic sympathetic terminals suggests that the release therefrom of NAD, ATP, PG or other mediators might also contribute to the peripheral, pronociceptive effects of GLU (Carlton et al., 1998a) (Section 8.2.4).

A release of SP by GLU from PAFs (or sympathetic terminals) may also be involved in mediating its peripheral, pronociceptive actions. Indeed, in analogy to GLU, SP may activate PAFs in several tissues, including muscles, skin and CBVs—though not blood vessels—by direct actions at NK₁ receptors coupled to PLC/PKC and increases in $[Ca^{2+}]_i$ (Andoh *et al.*, 1996; Carlton *et al.*, 1996; Heppelmann and Pawlak, 1997; Hu and Li, 1996; Holthusen *et al.*, 1997; Hu *et al.*, 1997; Hu and Li, 1996; Inoue *et al.*, 1995; Jonakait, 1993; Pedersen-Bjergaard *et al.*, 1989). NK₂ receptors and CGRP₁ receptors (positively coupled to AC) may also activate nocisponsive, sensory nerve terminals (Edvinsson *et al.*, 1997; Weinreich *et al.*, 1997).

It is intriguing to note evidence that $A\beta$ fibres also possess NMDA and AMPA receptors (Coggeshall and Carlton, 1997; Wood and Docherty, 1997). This raises the interesting possibility that their peripheral (or central) terminals might also be affected by EAAs. Whether such actions contribute to mechanical allodynia and/or the regenerative changes seen in PAFs following their injury would be of interest to determine (Komuro and Rakic, 1993). (Section 11.3). Inhibitory group II—and probably other subtypes of—mGlu receptor are localized on the central terminals of fine calibre PAFs in the DH (Sections 3.2.3 and 10.3.2.3). It would be of interest to establish whether they are likewise localized on peripheral PAF terminals and corresponding play a role in the modulation of nociception at these sites (Carlton and Coggeshall, 1997).

7.5. Mechanisms Underlying Sensitization of PAF Terminals

7.5.1. Synergistic Sensitization of PAFs

The sensitization of nociceptors by tissue damage and prolonged, noxious stimulation contributes to the familiar pain, tenderness and hyperalgesia of injury and inflammation (Dray and Bevan, 1993; Dray and Urban, 1996). Sensitization refers to an increase in the responsiveness of nocisponsive fibres to pronociceptive substances and noxious stimuli. Processes of sensitization are broadly displayed by nocisponsive PAFs and are involved in the recruitment of silent nociceptors normally irresponsive to noxious stimuli (Section 3.1.3). Sensitization of PAFs may be elicited by several of the above-mentioned substances which excite sensory terminals: notably, PGs and BK (Figs 3 and 4, Table 5). Sensitizing agents (and noxious stimuli) display marked reciprocal interactions. Although certain interactions may be inhibitory, this is likely the exception and the effects of inflammatory mediators are generally expressed in a highly synergistic fashion (Dray and Bevan, 1993; Dray and Urban, 1996; Lopshire and Nicol, 1997). For example, inflammatory mediators generally potentiate the effects of tissue acidosis (protons) (Kress et al., 1997; Steen et al., 1996).

BK sensitizes polymodal C and A δ fibres not only to protons, but also to 5-HT and PGs as well as to thermal and, less markedly, mechanical stimuli (Kumazawa *et al.*, 1991; Khan *et al.*, 1992; Lang *et al.*, 1990; Manning *et al.*, 1991; Neugebauer *et al.*, 1989) via actions at BK₂ and, under long-term inflammatory conditions, BK₁ receptors (Davis and Perkins, 1996; Dray and Perkins, 1993; Rueff *et al.*, 1996; Rupniak *et al.*, 1997; Seguin *et al.*, 1995). BK may also increase the production of PGs via the activation of PLA₂ which generates arachidonic acid (the precursor of PGs) from phospholipids (Prado *et al.*, 1997) (Section 3.2.8.4).

In a reciprocal fashion, PGs can sensitize PAFs to the actions of BK, as well as several other stimuli. Thus, PGE₂ and PGI₂, acting via (multiple) EP and IP receptors, respectively (Coleman *et al.*, 1994) (Section 3.2.8.4), have been shown to sensitize neurones to thermal and mechanical stimuli as well as to BK and capsaicin (Chahl and Iggo, 1977; Damas *et al.*, 1997; Handwerker, 1976; Khasar *et al.*, 1995; Martin *et al.*, 1988; Schepelmann *et al.*, 1992; Taiwo and Levine, 1988; Wang *et al.*, 1996a). In the light of these pronounced sensitizing actions of PGs, it is not surprising that aspirin and other anti-inflammatory drugs inhibiting COX 1 and COX 2 activity, as well as selective COX 2 inhibitors, suppress the peripheral sensitization of PAFs in inflamed tissue (Appleton, 1997; Buritova et al., 1996; Pairet and Engelhardt, 1996). However, a key-and, as yet, incompletely resolved-question is whether selective COX 2 inhibitors elicit robust, clinical analgesia without the gastric, renal and other side-effects of traditional anti-inflammatory agents which have been attributed to COX 1 inhibition (Langenbach et al., 1995; Mitchell et al., 1997; Pairet and Engelhardt, 1996; Smith et al., 1996). Initial clinical data are promising in this respect. However, the clinical demonstration of pronounced and adequate pain relief by genuinely 'selective' COX 2 inhibition in the absence of an influence upon COX 1 or unwanted side-effects is awaited (Bakhle and Botting, 1996; Frölich, 1997; Lichtenberger et al., 1995; Pairet and Engelhardt, 1996; Parnham, 1996, 1997; Seibert et al., 1994). Indeed, PGs derived from COX 1 and COX 2 might both, and via different mechanisms, be involved in the phasic and tonic modulation of nociception (Dirig et al., 1997; Smith and De Witt, 1996).

7.5.2. Ionic and Metabotropic Mechanisms Involved in Sensitization

7.5.2.1. Multiple mechanisms of sensitization

As mentioned above, the distinction between mechanisms underlying excitation and sensitization should not be regarded as absolute. Nevertheless, excitation generally involves direct activation of ion channels and the induction of APs, whereas sensitization generally reflects an increase in the likelihood of firing in response to stimulation. This increased probability of discharge is effected via the activation of intracellular transduction systems which either directly modulate activity at specific ion channels or exert long-term modifications in their activity via alterations in gene transcription (Section 7.6). Several inter-related and interactive mechanisms may be exemplified (Figs 3 and 4, Table 5).

7.5.2.2. Increases in levels of cAMP

A decrease in the afterpolarization which normally follows APs increases the probability that PAFs will discharge repetitively in response to further stimuli. Such an alteration in excitability can be mediated via the activation of AC and the generation of cAMP. CAMP triggers a PKA-mediated phosphorylation of K⁺-channels, thereby decreasing K⁺-conductance, as well as several other changes in terminal function. Indeed, increases in PAF terminal levels of cAMP may play a generalized role in mediating the sensitization of PAFs in response to heat (and other qualities of noxious stimulus), PGs, adenosine (via A_{2A} receptors), CGRP, histamine and 5-HT (Kress et al., 1996; Levine and Taiwo, 1994; Malmberg et al., 1997a; Taiwo et al., 1989; Wang et al., 1996a, 1997h; Weinreich and Wonderlin, 1987) (Section 7.5.2.3).

As concerns 5-HT, the original proposition that 5-HT_{1A} receptors mediate pronociceptive actions at PAF terminals is questionable since they couple *negatively* to AC: further, their activation increases and decreases K^+ - and Ca^{2+} -currents, respectively (Boess and Martin, 1994). More recent data suggests that activation of 5-HT₇ receptors may be involved

inasmuch as they are *positively* coupled to AC and expressed in DRG cells (Abbott *et al.*, 1997; Cardenas *et al.*, 1997; Doak and Sawynok, 1997; Kress and Reeh, 1996; Kress *et al.*, 1996; Pierce *et al.*, 1996, 1997; Rueff and Dray, 1992; Taiwo and Levine, 1992; Taiwo *et al.*, 1989; Rueff and Dray, 1992). 5-HT₄ and 5-HT₆ receptors also stimulate AC. Although mRNA encoding the latter was *not* found in rat DRG, mRNA *was* present for 5-HT₄ receptors (Pierce *et al.*, 1996). Indeed, evidence that 5-HT₄ sites may contribute to peripheral nociception and oedema has recently been presented (Cardenas *et al.*, 1997; Doak and Sawynok, 1997).

The potential importance of cAMP/PKA in mechanisms of sensitization is underscored by the diminution in inflammatory pain and PG-induced nociception observed in mice with a targetted mutation of a regulatory subunit of PKA which is colocalized with TRK A receptors in small-calibre PAFs. Nevertheless PKA localized in the central terminals of PAFs, or in intrinsic DH neurones, might be involved in the deficits shown by these mice (Malmberg *et al.*, 1997a) (Section 10.4.3.2).

7.5.2.3. Activation of a tetrodotoxin-resistant Na^+ -Current

It has been suggested that several substances, including PGE₂, 5-HT (possibly, via 5-HT₄ receptors) and adenosine, facilitate a tetrodotoxin (TTX)resistant, voltage-gated Na⁺-current (termed 'PN 3') which is specific to small calibre, nocisponsive neurones. The activation threshold is decreased, while both the magnitude and the rate of activation are enhanced (Gold et al., 1996a). As indicated above, the importance of an increase in cAMP/PKA activity in PAF terminals may not be restricted to an alteration in K^+ -conductance and it is possible that cAMP, another soluble second messenger, or NGF may be involved in the potentiation of this Na⁺-current, although this remains to be clarified (Akopian et al., 1996a,b; Gardenas et al., 1997; England et al., 1996; Friedel et al., 1997; Gold et al., 1996a,b; Sangameswaran et al., 1996; Wang et al., 1996a).

7.5.2.4. Activation of PKC via PLC

The induction of PLC by B_2 or NK_1 receptors, for example, triggers several mechanisms leading to an increased activity of PAF terminals. Thus, stimulation of PLC, via an increase in $[Ca^{2+}]_i$ levels, may indirectly enhance the activity of AC and elevate cAMP levels. Further, stimulation of PLC increases levels of DAG which, together with $[Ca^{2+}]_i$ activates PKC, leading to the phosphorylation and increased activity of certain, cation-permeable ion channels (including VDCCs) or receptors involved in the control of PAF activity and release (Dray and Urban, 1996).

For example, in a recent study of isolated, primary sensory neurones, BK sensitized their response to thermal stimuli by activation of PLC/PKC (Cesare and McNaughton, 1996). An activation of PLC/PKC by BK, leading to phosphorylation of synaptic, terminal-localized proteins, has also been implicated in the ability of BK to increase CGRP and SP release from sensory terminals—although terminal depolarization and an increase in $[Ca^{2+}]_i$ levels are also likely involved in this effect (Barber and Vasko, 1996; Geppetti *et al.*, 1991). Arachidonic acid, PGs and other metabolites, have also been suggested to enhance release from neuronal terminals via a mechanism involving PKC and, possibly, mGlu receptors. Whether such processes occur in nocisponsive PAF terminals would be of interest to evaluate (Collins and Davies, 1998; Sánchez-Prieto *et al.*, 1996).

SP may itself enhance the sensitivity of PAF terminals by a facilitation in ATP-mediated currents: this action involves a PKC-mediated acceleration of the recovery of P_{2x3} receptors from desensitization (Gammon et al., 1989; Hu and Li, 1996). Consistent with this possibility, the potentiation by SP (and CGRP) of responses at cloned, heterologouslyexpressed P_{2X2} receptors involves the activation of PLC (Hu and Li, 1996; Wildman et al., 1997). Further, an induction of PLC may be involved in the inhibition of K⁺-currents in PAFs by SP: this action-together with an increase in the cation-permeability of VDCCs-contributes to the sensitizing actions of SP via an increase in post-AP afterhyperpolarizations (Hu et al., 1997; Inoue et al., 1995; Quartara and Maggi, 1997).

7.5.2.5. Inhibition of K^+ -Currents

PAFs possess a substantial population of 5-HT_{2A} receptors (Carlton and Coggeshall, 1997; Pierce et al., 1996, 1997), the principal 5-HT receptor type involved in the sensitization of PAFs. Activation of 5-HT_{2A} receptors results in a cAMP-independent reduction of \overline{K}^+ -currents via the intervention of a Gprotein, although it is unknown whether an action of PKC is involved in PAF terminals (Abbott et al., 1997; Dray et al., 1994; Fan et al., 1994; Philippi et al., 1995; Taiwo and Levine, 1992; Taiwo et al., 1989; Todorovic et al., 1997). This attenuation of K⁺-currents reduces afterpolarization and is the basis for the potent sensitizing actions of 5-HT_{2A} receptor agonists on the human blister base and other inflamed tissues. Correspondingly, 5-HT_{2A} antagonists inhibit nociception associated with inflammatory states of the skin, nasal musoca, viscera and other tissues in animals and in man (Abbott et al., 1997; Damas et al., 1997; Dray et al., 1994; Fan et al., 1994; Greenshaw and Silverstone, 1997; Langlois et al., 1996; Nicol et al., 1997; Philippi et al., 1995; Taiwo and Levine, 1992; Taiwo et al., 1989). An indirect cAMP/PKA-mediated suppression of an outward K⁺-conductance may also be involved in the actions of PGs (PGE₂) and other AC-activating transmitters at PAF terminals (Malmberg et al., 1997a; Nicol et al., 1997) (Section 7.5.2.1). Further, it is possible that a PLC-mediated decrease in K⁺-currents is involved in the actions of SP (Quartara and Maggi, 1997) (Section 7.5.2.4).

7.5.2.6. Activation of NO synthase

NO synthase is upregulated in DRG and immune-competent cells upon inflammation (Dickenson, 1997; Xu and Wiesenfeld-Hallin, 1997). NO released from PAFs themselves, macrophages,

vascular (endothelial) cells and other sources may sensitize the terminals of nocisponsive PAFs and enhance neuropeptide release, possibly via mechanisms involving the activation of a cGMP-dependent PKG (Fozard, 1995; Laszlo et al., 1995; Lawand et al., 1997a; Miyasaka and Hirata, 1997; Qian et al., 1996b) Correspondingly, an interruption of (peripheral) NO synthesis inhibits inflammatory hyperalgesia (Herdegen et al., 1993; Kress et al., 1994; Lawand et al., 1997a; Meller and Gebhart, 1993; Seguin et al., 1995) and s.c. injection of NO acts algesically in man (Holthusen and Arndt, 1994). An induction of NO synthase may also be involved in the sensitizing actions of BK (Holthusen and Ding, 1997), although a more complicated pattern of interactions between NO and BK at PAF terminals has also been proposed (Miyamoto et al., 1997).

Notwithstanding indications for pronociceptive actions of NO at PAF terminals, in analogy to multiple actions of NO in the DH (Section 3.2.8.3), its role is likely to be more complex. Thus, several studies have suggested that the induction of NO may even participate in antinociceptive processes, for example, those initiated by the peripheral actions of opioids (Duarte et al., 1990, 1992; Kawabata et al., 1994) (Section 7.8). Indeed, although a key intracellular effector mechanism for NO is the generation of cGMP, this second messenger does not, as compared to cAMP, appear to be of major importance in sensitizing PAF terminals (Holthusen and Ding, 1997; Kress et al., 1996). Further, via generation of cGMP, NO may even accelerate tachyphylaxis of PAF terminals to BK (MacGhee et al., 1992). Quite apart from direct actions at nocisponsive PAF terminals, the peripheral influence of NO upon PAF activity may also be largely indirect, reflecting alterations in PG synthesis, an influence upon blood flow and the modulation of inflammatory processes per se (Fletcher et al., 1998) (Sections 4.8 and 7.9).

The multiple actions of NO mediated directly at PAFs, and indirectly via the vasculature etc, are likely of special relevance to vasoactive and neurogenic mechanisms precipitating migraine headaches (Fozard, 1995) (Section 7.9.3).

7.5.2.7. A potential role of NGF

Under conditions of tissue inflammation, NGF plays a major, long-term role in modifying the activity of sensory neurones via a modulation of gene expression and a corresponding change in their phenotype (McMahon, 1996; McMahon and Bennett, 1997) [see Lexin and Barde (1996) and Section 7.6.1 for an overview of the neurobiology of NGF]. However, there is evidence from studies of central neurones that NGF-and other neurotrophins-can rapidly modify neurotransmitter release, synaptic transmission and cellular excitability via diverse processes of phosphorylation, alterations in ion flux and changes in the levels of $[Ca^{2+}]_i$ and PKs (Berninger et al., 1993; Berninger and Poo, 1996; Bevan and Winter, 1995; Greene and Kaplan, 1995; Jiang et al., 1997; Kang and Schuman, 1995; Sherwood et al., 1997; Toledo-Aral et al., 1995; Winter, 1998) (Section 14). Should such effects occur in the terminals of C fibre, they could

contribute to the rapid induction of hyperalgesia by NGF (Dmitrieva and McMahon, 1996; Dmitrieva *et al.*, 1997; Lewin *et al.*, 1994a,b; McMahon and Bennett, 1997) (Section 7.7).

7.5.2.8. General significance of an induction of PKs and of an increase in $[Ca^{2+}]_i$

To summarize, the excitation and sensitization of PAF terminals is effected via multiple transduction mechanisms, including the induction of PKA and PKC via actions of cAMP and DAG/[Ca²⁺]_i, respectively. The activation of PKA and PKC by various inflammatory mediators may be of broad importance in modulating the activity of PAF terminals via the phosphorylation of specific proteins, such as ion channels and receptors, which control their electrical activity (vide supra). In addition, the production of NO in PAF terminals leads to the generation of cGMP, which can activate PKG. Although the significance of the latter intracellular cascade in nocisponsive PAFs is currently unclear, together with PKA and PKC, PKG plays a key role in events underlying the sensitization of DH neurones in response to sustained, nociceptive input (Fig. 4) (Lin et al., 1996a, 1997; Wang and Robinson, 1997) (Section 10.4). A related and crucial intracellular event underlying changes in the excitability of PAF terminals (and DH neurones) is an increase in Ca^{2+} -influx and an elevation in $[Ca^{2+}]_i$ levels. Increases in $[Ca^{2+}]_i$ may be elicited via several mechanisms. Notably, via VDCCs, via Ca²⁺permeable ion channels coupled to vanilloid receptors and via the PLC-mediated mobilization of Co^{2+1} for the PLC-mediated mobilization of $[Ca^{2+}]$ from intracellular stores following activation of B₂ (and NK₁) receptors. The influx of Ca²⁺ may provoke an additional increase in [Ca²⁺]_i via activation of Ca²⁺-dependent VDCCs and the 'feedforward' activation of intracellular ryanodine receptors which further mobilize cellular stores of $[Ca^2]$ (Dettbarn and Palade, 1997; Rondé and Nichols, 1997; Usachev and Thayer, 1997). Increases in the levels of $[Ca^{2+}]_i$ and PKs play a major role in the control of gene transcription, an alteration of which underlies longer-term changes in phenotype.

The crucial importance of increases in $[Ca^{2+}]_i$ and PK levels in the sensitization of *DH neurones* by C fibre input is discussed in Section 10.4.

7.5.2.9. Recruitment of silent nociceptors

The engagement of silent nociceptors provides a special case of sensitization. It is thought that silent nociceptors are chemosensitive and, once sensitized, they begin to respond to (low and high) intensity thermal and mechanical stimuli. That is, sensitization reflects the addition of *new* units rather than an amplification of the unit response. The precise mechanisms involved in the awakening of sleeping nociceptors are unclear. However, interactions between PGs and BK, key sensitizing elements, were outlined above and Stucky *et al.* (1996) have recently shown that PGE₂ *increases* the *number* of SP-containing PAFs which respond to BK. This action of PGE₂ is possibly mediated as EP₂ sites, and may involve the formation of cAMP and an increase in $[Ca^{2+}]_i$, leading to an increase in SP

release and/or an augmentation in the density of B_2 receptors: this remains to be clarified. Such findings are consistent with observations that, after exposure to PGs, other pro-inflammatory substances or tissue injury, up to half of mechanically-insensitive fibres begin to discharge in response to thermal or mechanical stimuli (Davis *et al.*, 1991; Handwerker *et al.*, 1991; Kress *et al.*, 1992; Meyer *et al.*, 1991).

7.6. Altered Phenotype of PAF Neurones

7.6.1. Focus on Pronociceptive Actions of NGF at TRK A Receptor-Bearing, Small Calibre C Fibres

Changes in the activity and properties of nocisponsive C (and A δ) fibres innervating inflamed tissue may reflect an altered phenotype: in particular, a modification of gene expression. Certain changes in phenotype might counter nociception. However, the most prominent components of the complex and time-dependent pattern of changes displayed by PAF neurones upon peripheral inflammation lead to a facilitation of nociception. Notably, increases in PAF levels of SP, CGRP (as well as of a structurally-related 'islet amyloid polypeptide'), NO and GLU (Cho et al., 1998; Dickenson, 1997; Donnerer et al., 1992, 1993; Donaldson et al., 1992; Goff et al., 1998; Mulder et al., 1997; Sluka et al., 1994b; Sluka and Westlund, 1993). Interestingly, under certain conditions, a parallel increase in the DH density of NK₁ receptors may further enhance the excitatory actions of SP released from PAF terminals onto neurones therein (Abbadie et al., 1996; Coggeshall and Carlton, 1997; Goff et al., 1998).

Little information is available concerning the density of specific receptor types on PAFs themselves under conditions of inflammation, although it has been shown that GAL1 and GAL2 receptors on small calibre PAFs are respectively up- and downregulated by inflammation (Puttick et al., 1994; Xu et al., 1996b, 1997d). Further, an increase in the density of mRNA encoding Y₁ receptors in the DRG (and DH) under inflammatory conditions has been documented. This may be associated with an alteration in the central actions of ININ-derived NPY at Y₁ receptors on small calibre PAF terminals or PNs in the DH (Ji et al., 1994; Munglani et al., 1996). The roles of multiple GAL and NPY receptors in the modulation of nociception is complex and the functional consequences of these changes unclear, although an increased activity of DH pools of NPY may, overall, counter nociception (Section 8.2.5) (Ji et al., 1994; Kask et al., 1997; Munglani et al., 1996; Xu and Wiesenfeld-Hallin, 1997).

In addition to rapid functional actions (such as phosphorylation), substances involved in the sensitization of PAFs may eventually influence gene transcription under conditions of prolonged, inflammatory pain. However, little concrete information is available in this regard. Thus, the present discussion focuses on evidence that the neurotrophin, NGF, which is indispensable for the initial survival and phenotypic development of a subpopulation of fine calibre, CGRP-containing sensory neurones (Lewin, 1995; Lewin and Barde, 1996; Liebl *et al.*, 1997; Lindsay *et al.*, 1994), plays a key

role in mediating pain-related alterations in the phenotype of C fibres and, correspondingly, in the enhancement of inflammatory hyperalgesia (Lewin and Mendell, 1993; McMahon and Bennett, 1997; Woolf, 1996).

In both man and rats, the acute- or chronic administration of NGF elicits an increase in the sensitivity of cutaneous and visceral small calibre PAFs to noxious mechanical and thermal stimuli (though not spontaneous pain). Further, transgenic mice lacking NGF show a perturbation of nociceptive transmission (Akopian et al., 1996b; Davis et al., 1993a; Dmitrieva and McMahon, 1996; Dmitrieva et al., 1997; Lewin et al., 1993; McMahon and Bennett, 1997; Petty et al., 1994; Smeyne et al., 1994). Consistent with a pro-nociceptive role of NGF in inflammatory pain, its levels are markedly increased in inflamed cutaneous, joint and visceral tissues. These pools of NGF are derived from various sources including Schwann cells, mast cells, macrophages, fibroblasts and, in skin, keratinocytes. Multiple factors, in particular cytokines, have been implicated in the induction of NGF synthesis and secretion (Aloe et al., 1992; Donnerer et al., 1992; McMahon and Bennett, 1997; Safieh-Garabedian et al., 1995, 1996, 1997) (Section 7.7). The influence of NGF on the activity of PAF terminals is complex and involves both indirect actions, mediated via sympathetic afferents and mast cells (Section 7.7), as well as direct, phenotype-related actions described in this section (Lewin, 1995; Lewin and Barde, 1996; Lewin and Mendell, 1993).

About 90% of CGRP-containing, capsaicin-sensitive, cutaneous small calibre C fibres bear specific TRK A receptors: that is, almost half of DRG-localized PAFs. Visceral nocisponsive afferents are also rich in TRK A receptors (McMahon and Bennett, 1997). These, and other types of PAF, also possess a low affinity, p75^{NTR} receptor responsive to NGF (and other neurotrophins). The engagement of $p75^{NTR}$ sites exerts a complex, modulatory (generally facilitatory) influence upon the actions of NGF at TRK A receptors and enhances retrograde flow of NGF from PAF terminals to the DRG (Averill et al., 1995; Bergmann et al., 1997; Bothwell, 1995; Davies, 1997; Delcroix et al., 1997; Gardano et al., 1997; Hantzopoulos et al., 1994; Jiang et al., 1997; Kashiba et al., 1995, 1996; Levi-Montalcini et al., 1996; MacPhee and Barker, 1997). In the course of development, an action of NGF at p75^{NTR} receptors on sensory neurones and certain other cell types may promote constitutive processes of apoptotic cell death, an action diametrically opposed to its prevention of programmed cell death via actions at TRK A receptors. However, p75^{NTR} sites may be capable of antonomous signalling, and the interrelationship between p75^{NTR} receptors on neuronal and nonneuronal cells, NGF and cellular degeneration is still under intensive discussion (Bredesen and Rabizadeh, 1997; Dechant and Barde, 1997; Ladiwala et al., 1998). TRK A receptors are dimerized and activated by NGF. Although NGF is retrogradely transported to cell bodies, wherein it initiates nuclear events modifying gene transcription, this may not be the exclusive mechanisms involved in its modulation of PAF phenotype. Thus, both the

endosome-assisted dispatch of activated TRK A receptors to the DRG, as well as the transport of a NGF–TRK A complex, may also be involved in the expression of its actions (Ehlers *et al.*, 1995; Grimes *et al.*, 1996; Riccio *et al.*, 1997).

Via such mechanisms, NGF can enhance the production of many molecules occurring in the terminals of TRK A-bearing C fibres, including several involved in the induction of nociception: vanilloid receptors, certain classes of Na⁺-channel, protonactivated ion channels, SP, CGRP and, possibly, B1 receptors (Aguayo and White, 1992; Donnerer et al., 1992; Hökfelt et al., 1994; Kuraischi et al., 1989; Leslie et al., 1995; Malcangio et al., 1997a Malcangio et al., 1997b; Petersen et al., 1998b; Rueff and Mendell, 1996; Sherwood et al., 1997; Toledo-Aral et al., 1995, 1997; Winter et al., 1993; Woolf et al., 1994) (Section 7.7). As concerns Na⁺channels, it is of particular interest that NGF increases the expression of a 'PN 1 class'-at least in transfected cell lines (Toledo-Aral et al., 1997). This PN 1 channel differs in two major respects to the 'PN 3' Na⁺-channel involved in the sensitization of C fibre terminals (Section 7.5.2.3):

- 1. it is TTX-sensitive; and
- 2. it occurs in both small *and* large calibre, DRG neurones—as well as in sympathetic neurones.

The PN 1 channel is potentially involved, thus, in the induction of hyperalgesia and mechanical allodynia—although, presumably, only Na⁺-channels in TRK A-bearing C fibres and sympathetic terminals can be up-regulated by NGF (Toledo-Aral *et al.*, 1995, 1997). The modulation by NGF and other factors of the expression of various Na⁺-channels in injured PAFs is discussed in Section 8.2.2.3.

In line with such phenotype changes, there is evidence that the delayed-onset (several hours) mechanical hyperalgesia (Lewin et al., 1994b) provoked by systemic or spinal administration of NGF to rats (both of which routes can access the DRG) involves an increase in the production, transport and spinal (or peripheral) release of SP, CGRP and NKA. Thus, antibodies against NGF, or TRK A fusion products, inhibit both inflammatory hyperalgesia and the accompanying increase in PAF levels of these neuropeptides (Bowles et al., 1994; Donnerer et al., 1993; Lewin and Mendell, 1993; Lewin et al., 1993, 1994b; Malcangio et al., 1997a,b; McMahon et al., 1995; Woolf et al., 1994: though see Rueff et al., 1996). In addition, NGF-induced hyperalgesia and wind-up of DH neurones is inhibited by NK1 antagonists, consistent with a role of sensitized, SPresponsive DH neurones in the longer-term, pronociceptive actions of NGF (Lewis et al., 1995; McMahon and Bennett, 1997; Thompson et al., 1995). Recently, it was suggested that the ability of NGF to elicit hyperalgesia via SP release may be auto-limited by the accumulation of N-terminal SP metabolites which inhibit its pronociceptive actions (Larson and Kitto, 1997).

Of particular interest, NGF induces the expression of a further neurotrophin for *non*-C PAFs, brain-derived neurotrophic factor (BDNF) (Section 7.6.2), in TRK A-positive, cutaneous, nocisponsive C fibres. BDNF may elicit actions in the periphery,

and it is also transported anterogradely to the DH wherein it may modulate nociception (Cho *et al.*, 1997a,b; Michael *et al.*, 1997a; Apfel *et al.*, 1996; Ho *et al.*, 1997) (Section 14). A NGF-mediated increase in BDNF expression has, notably, been observed, under conditions of peripheral inflammation (Apfel *et al.*, 1996; Cho *et al.*, 1997b,c).

To summarize, NGF can enhance nociception by the sensitization of TRK A receptor-bearing, small calibre PAFs innervating skin, viscera and, probably, other tissues. These peripheral actions ultimately contribute to processes of neuronal sensitization in the DH (Section 10). The actions of NGF may involve rapid, TRK A receptor-mediated interactions with intracellular signals, such as $[Ca^{2+}]_i$ and PKs, thereby altering the activity of other excitatory/sensitizing receptors ion channels on nocisponsive PAF terminals. In addition, NGF modulates the expression of several pronociceptive receptors, ion channels and transmitters in nocisponsive PAF terminals, resulting in a reinforcement of nociceptive transmission to the DH.

7.6.2. A Potential Role of Growth Factors Other Than NGF

The above-described pattern of phenotype alterations in C fibres may be paralleled by changes in the phenotype of $A\beta$ fibres which, under inflammatory conditions, and via a so-called 'phenotypic switch', begin to synthesize and release SP (Neumann et al., 1996; Woolf et al., 1992). The release of SP by $A\beta$ fibres may directly (or via volume effects) stimulate DH-localized $NK_{1/2}$ receptors and contribute to processes of sensitization underlying the mechanical allodynia and hyperalgesia of inflammation (Neumann et al., 1996) (Sections 11.2 and 11.4). Since large calibre PAFs do not bear TRK A receptors, NGF cannot be directly involved in this change of phenotype. Thus, other neurotrophins are likely implicated in mediating phenotypic changes in non-C PAFs: notably BDNF and NT-4/5, both of which express their actions via TRK B receptors, and NT-3, which expresses its actions via TRK C receptors (Acheson et al., 1995; Davies, 1997; Di Stefano et al., 1992; Friedel et al., 1997; Lewin and Barde, 1996; Lexin and Barde, 1996; Liebl et al., 1997; Yan et al., 1997).

TRK C receptors (and actions of NT 3) are, in fact, primarily localized to large calibre, myelinated, propriosensitive and mechanoceptive fibres and a role in nociception remains to be established. Nevertheless, a recent study suggested that retrograde transport of NT 3 to the DH may actually *reduce* SP release and nociception by an, as yet, unclear mechanism (Airaksinen *et al.*, 1996; Helgren *et al.*, 1997; Malcangio *et al.*, 1997; Wilkinson *et al.*, 1996; Wright and Snider, 1995) (Section 14.5.3).

On the other hand, TRK B receptors for BDNF are found on neurones of a broad range of diameters, including a population involved in mechanical nociception (Koltzenburg *et al.*, 1995; Lewin and Barde, 1996; Wright and Snider, 1995). Thus, it would be of interest to determine whether BDNF might be involved in the phenotype changes shown by large $A\beta$ fibres under conditions of inflammation,

and to establish whether BDNF plays a peripheral role in the modulation of sensory transmission and the induction of A β fibre-mediated mechanical allodynia (Koltzenburg et al., 1995). In this light, it is of note that BDNF, like NGF, modifies $[Ca^{2+}]_i$ levels and ionic currents in central neurones (Berninger et al., 1993; Sherwood et al., 1997) (Section 14.4). Further, as mentioned above, BDNF is itself synthetized in a sub-population of PAFs of various diameter, and its expression is enhanced by NGF in nocisponsive C fibres under inflammatory conditions (Apfel et al., 1996; Cho et al., 1997a,b; Ernfors et al., 1990; Ho et al., 1997; Michael et al., 1997a; Zhou and Rush, 1996). In contrast to cutaneous PAFs, the majority (~90%) of visceral afferents coexpress TRK A and TRK B receptors and only a few (<5%) express TRK C receptors: this provides a clear difference to cutaneous PAFs (McMahon, 1994). Inasmuch as inflammation up-regulates BDNF in visceral PAFs, a potential role of BDNF in the modulation of visceral phenotype and pain would be of interest to explore (McMahon et al., 1994; Lewin, 1995: Lewin and Barde, 1996). In line with such a potential role, BDNF has been shown to enhance the sensitivity of vagal, sensory, SP-containing PAFs to capsaicin (Winter, 1998).

The above observations concerning neurotrophins and nociception clearly would justify further functional exploration and an extension to other growth factors (Acheson et al., 1995; Kim et al., 1994; Zhou and Rush, 1996). For example, the subpopulation of P_{2x_3} -receptor-bearing nocisponsive, small calibre PAFs which does not express CGRP/TRK A receptors is dependent upon GDNF. A potential, acute role of GDNF in the modulation of phenotype and nociception would be of interest to evaluate (Leclere et al., 1997; Molliver et al., 1997; Naveilhan et al., 1997) (Sections 8.2.6.2 and 11.3.4.3). Whether other types of modulatory and trophic factor are involved in the control of phenotype and nociception under conditions of inflammation also remains to be clarified. This is not unlikely inasmuch as many other molecules potentially capable of modulating neuronal phenotype, growth and activity are secreted from glia and immunecompetent cells in the vicinity of PAF terminals and inflamed tissue (Berninger and Poo, 1996; Chalazonitis et al., 1992; Lo, 1995; Patterson and Nawa, 1993) (Section 4.8) (Figs 3 and 5).

7.7. Modulatory Effects: Altered Activity and Phenotype of PAFs Mediated *indirectly* by Other Elements of the Inflammatory Response

The final mechanism for an enhancement in the activity of nocisponsive PAF terminals, *modulatory effects*, is a general term. It refers to a complex mesh of reciprocal interactions amongst the broad diversity of elements which directly and indirectly control the activity of nociponsive PAFs (Figs 3 and 4) (Dray and Urban, 1996; Dray *et al.*, 1994; Reeh and Kress, 1995; Treede *et al.*, 1992b). Certain components of the inflammatory response, such as cytokines, exert principally modulatory actions and may *not* directly affect PAF terminals. Further, modulatory effects are a virtually universal mechanism for



Fig. 5. Receptorial and intracellular mechanisms underlying the sensitization of WDR neurones in the DH by C fibre-mediated, nociceptive input. Abbreviations are as indicated in the general list. In addition: HETE, hydroxyeicotetraenoic acid; and LT, leukotriene. For reasons of clarity, additional interactions are not shown (Sections 10.3 and 10.4). Ultimately, membrane-localized ion channels represent the final common pathway via which the electrical activity of DH neurones (and nociceptive output) is modified. This may be expressed either rapidly, via the transduction mechanisms indicated, and/or, in the long-term, by alterations in gene transcription (neuronal phenotype). Some of the actions indicated are based upon observations made in other tissues displaying NMDA receptor-mediated processes of synaptic amplification: in particular, hippocampal mechanisms of LTP. The putative role of neurotrophins is based upon a line of reasoning discussed in detail in Section 14 but not, as yet, proven to occur in DH neurones. In addition, the feedback action of SP on PAF terminals may be expressed both directly and indirectly. The following (non-illustrated), potential mechanisms should also be noted. Both PKA and a calmodulin-dependent PK may trigger the phosphorylation (activation) of AMPA receptors (Section 10.4). The anterograde transport of BDNF to PAF terminals may allow it to influence DH neurones (Section 14.5.3). NO also may be derived from PAF terminals and other neuronal and nonneuronal cell types (Section 3.2.8.3). ATP (derived from PAF terminals and other sources) may excite DH neurones via ionotropic P_{2X} receptors and metabotropic P_{2Y} receptors (Section 3.2.8.2). PGs (derived from PAF terminals and other sources) may activate DH neurones via excitatory EP receptors coupled to a variety of intracellular signals (Section 3.2.8.4).

the modification of PAF activity by algogenic substances in inflamed tissue, including SP, BK, NGF and PGs etc. Specific tissue types (such as mast cells) may release several modulatory substances, while an individual modulatory agent (such as NGF) may be derived from multiple sources. Several examples may be cited (Fig. 3): platelets (5-HT); mast cells (5-HT, histamine and NGF); polymorphic leukocytes (protons, PGs and the leukotriene, di-HETE); lymphocytes and macrophages (cytokines); Schwann cells (NGF and PGs); fibroblasts and keratinocytes (NGF); blood vessel endothelial walls (NO, cytokines and PGs) and sympathetic terminals (NAD, NPY, ATP, NO and PGs) (Fig. 3). Illustrative of such modulatory mechanisms are the following interactions.

1. Several pro-inflammatory and pronociceptive cytokines, including IL-1 β and TNF α , are secreted from macrophages, fibroblasts and other immune cell types in response to infection, inflammatory stimuli and algogenic factors such as NO, SP and NGF. Cytokines and, in particular, TNFa, sensitize nocisponsive neurones by multiple, indirect mechanisms leading to, for example, an induction of B1 receptors, possibly on PAF terminals themselves, on immune cells and on sympathetic terminals (Aloe et al., 1997; Cunha et al., 1991, 1992; Davis and Perkins, 1994, 1997; Hua et al., 1996; Maier et al., 1993; Perkins and Kelly, 1994; Petersen et al., 1998b; Rueff et al., 1996; Seabrook et al., 1997; Wagner and Myers, 1996a; Wagner et al., 1998; Wang et al., 1997d; Watkins et al., 1994) (see below and

Section 8.2.1). Further, $TNF\alpha$ and IL-1B enhance the synthesis and release of other inflammatory mediators, including PGs (COX 2 induction) NO and, via a positive feed-back loop, NGF from fibroblasts and Schwann cells (Arias-Negrette et al., 1995; Davis and Perkins, 1994; Kelly et al., 1997; McMahon and Bennett, 1997; Miyasaka and Hirata, 1997; Newton et al., 1997; Perkins and Kniss, 1997; Safieh-Garabedian et al., 1995; Thomazzi et al., 1997; Watkins et al., 1995; Woolf et al., 1997) (see below). Interestingly, PGs may fulfil an inhibitory, feed-back role since they decrease cytokine secretion from lymphocytes and macrophages. Paradoxically, therefore, despite relief of pain, anti-inflammatory drugs inhibiting COX activity may exacerbate joint disintegration in arthritis by indirectly increasing cytokine levels (Bakhle and Botting, 1996; Huskisson et al., 1995; Renz et al., 1995).

- In addition to (*delayed*) hyperalgesic actions of NGF expressed *directly* upon PAFs via a change in their phenotype (Section 7.6.1), NGF can *indirectly* (and *rapidly*) sensitize nocisponsive PAFs (Lewin *et al.*, 1993). This action of NGF (seen over a period of 5–30 min) reflects both:
 - 1. the degranulation of mast cells and other types of immunocompetent cell; and
 - 2. an activation of sympathetic neurones, for which NGF is a trophic factor (Rueff and Mendell, 1996; Tal and Liberman, 1997).

Indeed, like small calibre PAF and sympathetic terminals, mast cells possess TRK A receptors for NGF-and, possibly, also TRK B receptors for BDNF, the potential significance of which remains to be evaluated (Davis et al., 1996; Horigome et al., 1993; Lewin and Barde, 1996). Interestingly, further, mast cells themselves synthesize and release NGF (Leon et al., 1994). Mast cell inactivation, or pharmacological blockade of serotoninergic transmission, can prevent the immediate (5 min) hyperalgesic response to NGF: this suggests that NGF stimulates mast cells to release 5-HT. Sympathectomy abolishes a somewhat later (30 min) component of NGF-hyperalgesia suggesting that sympathetic terminals may be sequentially activated (Andreev et al., 1995; Gentry et al., 1997; Lewin et al., 1994b). However, the precise temporal pattern of events remains to be further elucidated. Indeed, a recent study of inflammatory hyperalgesia over a longer observation period of several days (Woolf et al., 1996) suggested that the contribution of sympathetic terminals to NGF-induced hyperalgesia may be relatively *short*-lived whereas mast cells and direct transcription-dependent actions on sensory neurones (Section 7.6.1) may later re-assume a more preponderant role. A further, recent study also suggested that mast-cell activation may be the principal mechanism involved in the long-term (several days) thermal hyperalgesia provoked by NGF in rats (Rueff et al., 1996). The proximate mechanisms were suggested to be:

1. an increase in BK levels triggered by a NGFinduced degranulation of mast cells to liberate kallikrein enzymes; and a consequent enhancement in cytokine (TNFα/ IL-1β) release from macrophages mediated cojointly by NGF and BK and leading to the above-mentioned upregulation of B₁ receptors on PAF and/or sympathetic terminals (Petersen *et al.*, 1998b; Rueff *et al.*, 1996).

That is, cytokines mediate NGF-hyperalgesia by up-regulating B_1 receptors on sympathetic and/or PAF terminals. On the other hand, there is evidence that, in an *opposite* fashion, NGF mediates the pronociceptive actions of TNF α , and that NGF directly induces B_1 receptors on PAF terminals via p75^{NTR} sites (Petersen *et al.*, 1998b; Watkins *et al.*, 1995) (*vide supra*). Thus, the interrelationship between NGF and cytokines in pronociceptive mechanisms (and B_1 receptor induction) remains unclear. Indeed, the roles of NGF and cytokines are emerging to be increasingly complex and their multiple and reciprocal interactions require further mechanistic elucidation. The influence of sympathetic mechanisms is further discussed in Section 8.2.4.

- 3. As mentioned already (Section 7.4.3), leukotriene B_4 (a lipoxygenase product of arachidonic acid metabolism) indirectly increases the thermal and/or mechanical sensitivity of C fibres by releasing a further leukotriene, 8R,15S-diHETE, from polymorphonuclear leukocytes (Amann *et al.*, 1996; Dray *et al.*, 1994; Levine *et al.*, 1986a; Martin, 1990; White *et al.*, 1990). A role of leukotrienes in the hyperalgesic actions of NGF has also been proposed (Amann *et al.*, 1996).
- 4. Principally via B₂ receptors, BK induces a panoply of pronociceptive/proinflammatory effects including: degranulation of mast cells to release cytokines (vide supra), PGs, 5-HT and histamine; venous extravasation and vasodilation; chemotaxis of lymphocytes, and an enhancement of the release of NAD and other mediators from sympathetic neurons-via stimulation of B2 and/or B1 receptors on their terminals (Berg and Koteng, 1997; Dray and Perkins, 1993; Dray and Urban, 1996; Green et al., 1997; Holthusen, 1997; Marceau, 1995; Miao et al., 1996; Reeh and Kress, 1995; Seabrook et al., 1997; Sorkin et al., 1997; Taiwo and Levine, 1988; Wang et al., 1997a; Yuan et al., 1997) (vide supra). These multiple actions of BK add to its direct effects at PAFs in collectively enhancing nociception and inflammation. SP, a principal mediator of NI, exerts a similar, multifarious pattern of pro-inflammatory actions (Section 7.9). The actions of SP include mast cell degranulation, and it has been suggested that the activation of A_3 receptors may likewise increase nociception and inflammation via the release of histamine and 5-HT from mast cells (Sawynok et al., 1997; Maggi, 1991b).
- 5. In addition to direct actions at PAFs, NO may participate in the induction of COX 2 in other cells types, thereby increasing the production of PGs and exacerbating inflammation and nociception. However, the generality of this action has been challenged and reciprocal, tissue-dependent interactions between NO synthase and COX

synthase require further elucidation (Busija and Thore, 1997; Henrion et al., 1997; Payá et al., 1997; Swierkosz et al., 1995; Salvemini et al., 1996). An induction by NO of TNFa production in immunecompetent cells may also be of significance in view of the pronociceptive actions of this cytokine (Wagner and Myers, 1996a,b; Wang et al., 1997e) (vide supra and Section 8.2.1). Indirect actions of NO also likely reflect its vascular effects. These vascular actions of NO are mediated via a complex interplay between cGMP, PKG and $[Ca^{2+}]_{i}$, the sequestration of which modifies the phosphorylation of myosin and other smooth muscle components in the vascular wall, resulting in an increase in permeability (extravasation) and contractility (vasodilation) (Lee et al., 1997b; Payá et al., 1997; Yuan et al., 1997)

6. Finally, 5-HT may influence the activity of sympathetic nerve terminals, the ganglia of which express mRNA encoding multiple 5-HT receptor types: 5-HT_{1A}, 5-HT_{1B}, 5-HT_{2A}, 5-HT_{2B}, 5-HT₃ and 5-HT7 (Pierce et al., 1996, 1997; Newberry et al., 1996). Excitatory actions of 5-HT_{2B} and 5-HT₃ receptors on sympathetic terminals might, for example, contribute to the pronociceptive and proinflammatory actions of 5-HT whereas a 5-HT_{1B} receptor-mediated inhibition of release from sympathetic terminals might decrease nociception and play a role in the vasoconstrictor/ anti-inflammatory actions of 5-HT at CBVs (Jones et al., 1995; Molderings et al., 1996; Newberry and Gilbert, 1989) (Section 7.9). In addition, receptors for 5-HT (e.g. 5-HT_{2A} sites) have been described on Schwann cells suggesting that they may indirectly mediate the influence of 5-HT upon the activity of PAF terminals (Yoder et al., 1997).

7.8. Peripheral Antinociceptive Mechanisms

Although the previous paragraphs focused on the *pronociceptive* properties of inflammatory mediators following tissue damage, certain actions within inflammatory tissue may counter pathology and hyperalgesia. The existence of such mechanisms again illustrates the principle of counter-regulation whereby noxious stimulation is associated with the simultaneous activation of (preponderantly) pronociceptive, yet also counter-regulatory, antinoceptive mechanisms.

The cytokine, IL-10, can reduce inflammation and nociception via an inhibition of COX 2-mediated PG production, and by an inhibition of the production and actions of pronociceptive cytokines, such as TNF α (Mertz *et al.*, 1994; Poole *et al.*, 1995; Wagner *et al.*, 1998). In addition, a further endogenous mediator, the inducible polypeptide, lipocortin-1, has been proposed to mediate the peripheral, antihyperalgesic actions of glucocorticoids (Ferreira *et al.*, 1997). Histamine generally acts as a pro-inflammatory agent via H₁ and H₂ sites (Sections 7.4.8 and 7.9.3). However, its release from mast cells in CBVs may *reduce* nociception via an action at inhibitory H_3 sites localized on the terminals of trigeminal PAFs innervating dural vessels: a reduction in Ca²⁺-dependent release may be involved in this action (Dimitriadou *et al.*, 1997; Matsubara *et al.*, 1992). It has also recently been suggested that somatostatin exerts peripheral, antinoceptive properties in reducing the activity and mechanosensitivity of articular PAFs in inflamed joints (Heppelmann and Matthias, 1997).

Nevertheless, most recent attention has been directed towards reports that locally-generated opioid peptides (including β -endorphin derived from immune cells) may decrease nociception via actions at μ -, δ - and/or κ -receptors localized on (and upregulated in) sensory neurones, sympathetic terminals and/or other elements in inflammatory tissue (Catheline et al., 1996; Eriksson-Mjöberg et al., 1997; Kalso et al., 1997; Khasar et al., 1995; Lyons and Blalock, 1997; Stein and Yassouridis, 1997; Stein et al., 1989; Stein, 1993; Taiwo and Levine, 1992; Wilson et al., 1996a; see also Picard et al., 1997). A disruption of the perineural blood-nerve barrier in inflamed tissue might also lead to an improved access of circulating pools of opioids to PAF terminals, a change of broader relevance to the peripheral actions of other agents (Antonijevic et al., 1995; Coggeshall et al., 1997). IL-1 β has also been claimed to elicit antinociception via the release of opioids from immune cells in inflamed tissue and, IL-1 β -pretreated joints, des-Arg⁹-BK in was reported to elicit antinociception via the local release of opioids (Davis and Perkins, 1997). However, as described earlier (Section 7.7), IL-1 β elicits predominantly pronociceptive and proinflammatory actions and it has been suggested that its antinociceptive effects reflect a weak affinity for opioid receptors (Wang et al., 1997e). CRF, derived from immune cells, sympathetic or PAF terminals, has also been proposed to elicit peripheral antinociception via a local, opioidergic mechanism (Bileviciute et al., 1997a; Schäfer et al., 1994, 1997). Further, CRF exerts a complex pattern of anti-inflammatory and pro-inflammatory actions, for example, via mast cell degranulation (Kiang and Wei, 1985; Schäfer et al., 1997; Theoharides et al., 1998). Interestingly, in analogy to mechanisms well-documented in the DH, CCK interferes with peripheral opioidergic mechanisms of antinociception, possibly via the induction of PKC and an increase in $[Ca^{2+}]_i$ levels (Schäfer *et* al., 1998) (Sections 3.2.6 and 8.2.5). The novel heptadecapeptide, nociceptin, has high affinity for a recently-cloned, opioid-like-receptor (Meunier, 1997). Its levels are increased in the DRG by inflammation (Andoh et al., 1997). Whether nociceptin plays an anti- or pronociceptive role in peripheral inflammation would be of interest to clarify (Meunier, 1997).

7.9. Antidromic Activation of PAFs: NI, Vasorelaxation and Migraine Headache

7.9.1. The Efferent Role of PAFs

The above paragraphs described the direct, excitatory and sensitizing actions, as well as the indirect, modulatory effects, of algogenic agents at sensory, nocisponsive neurones under conditions of local tissue damage. Together with changes in PAF phenotype, such actions collectively increase the orthodromic input from nocisponsive PAFs to the DH. However, PAFs also transmit antidromic impulses, propagated via the invasion of collateral fibres and a limited degree of C fibre terminal coupling. This leads to an enhancement in the peripheral release of EAAs, CGRP SP, etc. The liberation of these substances, which is further amplified by their local, positive feedback actions on PAF terminals (Section 7.4.9), underlies processes of NI. In this respect, SP is probably the principal mediator. NI occurs in a broad variety of tissues (skin, muscle, joints and viscera) and refers to a constellation of effects including: an increase in blood flow; vasodilation; plasma extravasation; mast cell degranulation; hyperaemia: and an enhancement of leukocyte adhesion to vascular endothelium, actions expressed in interaction with immune cells, sympathetic endings, blood vessel walls and other tissues (Boolell and Tooke, 1990; Donnerer and Amann, 1993; He et al., 1990; Holzer, 1988; Mantyh et al., 1989; McMahon, 1994; Maggi, 1991a,b; Marchettini et al., 1996; Sann et al., 1996; Schaible and Grubb, 1993; Smith et al., 1993).

In analogy to their afferent, 'warning' role in the DH, the *efferent*, peripheral role of PAFs is adaptive in promoting wound healing and counteracting local infection, actions effected via an enhancement of tissue irrigation and infiltration by immune cells, etc. (Brain and Cambridge, 1996; Holzer, 1988; Maggi, 1991a,b) (Section 2). As a more specific example of a beneficial consequence, the (proton-induced) release of CGRP in muscle during intense exercise enhances local blood flow and oxygen supply and stimulates electrogenic Na⁺-K⁺ muscle pumps: thereby, the efficiency of muscle fibre excitability and contractibility is enhanced, and performance is sustained under intense effort (Andersen and Clausen, 1993; Maggi, 1991a,b). Under certain pathological conditions, however, such as arthritis, processes of NI exacerbate tissue damage and pain by indirectly facilitating the interplay of the immune, vascular and sympathetic components of pro-inflammatory and pronociceptive events detailed above. In this light, the antidromic activation of (and release of neuropeptides from) sensory terminals may be considered a special case of those above-discussed 'Modulatory Actions' (Section 7.7) which indirectly potentiate the activity of nocisponsive PAFs.

Interestingly, not all classes of C fibre mediate NI, and the population(s) of C fibres which orthodromically convey nociceptive information to the DH are not identical to those antidromically eliciting NI. Indeed, in certain (non-rodent) species, a distinctive and specialised class of heat-responsive PAF may mediate cutaneous vasodilation and the axonal flare component of NI (Lynn *et al.*, 1996) (Sections 3.1.1 and 3.2.1).

7.9.2. NI, Vascular Mechanisms and Migraine

Although NI is a general feature of tissue damage, processes of NI in CBVs have been specifically

implicated in the induction of migraine headaches. Thus, the theory of 'sterile NI' advocates that central mechanisms result in the antidromic invasion of trigeminal PAF terminals (Donnerer and Amann, 1993; Moskowitz, 1992), thereby provoking the release of proinflammatory (vasodilatory) and pronociceptive substances, such as SP, CGRP and EAAs. Their actions result in the vasorelaxation and inflammation of subdural CBVs (including arteries) (Dimitriadou et al., 1997; Edvinsson et al., 1997; Schmelz et al., 1997a). An appealing feature of this neurogenic theory is that it can accommodate an involvement of 'stress' or other central 'mood' changes in the precipitation of migraine attacks. For example, the 'stress'-induced phenonemon of spreading depression underlies the aura of migraine headaches and may be involved in triggering the antidromic activation of trigeminal fibres (Olesen and Janssen-Olesen, 1997). As outlined in Section 4.8, spreading depression refers to a process whereby waves of successive hyperpolarization and depolarization in neuronal and non-neuronal cells sweep across the cortex and ultimately attain deeper structures (Olesen and Janssen-Olesen, 1997).

However, the question of the mechanism(s) underlying the induction of migraine headache is still intensely debated (Lance, 1993; Moskowitz, 1992; Olesen and Janssen-Olesen, 1997; Ferrari and Saxena, 1993). Indeed, although SP is probably the major mediator of NI, clinical results with NK1 antagonists in migraine have, to date, been disappointing. This suggests that selective blockade of-at least, the NK1 receptor-mediated component of-NI is not sufficient to interrupt migraine attacks (Clayton et al., 1997; Cutrer et al., 1995; Williamson et al., 1997). Further, although the serotoninergic agent, sumatriptan, and other drugs effective in terminating migraine attacks, inhibit NI via actions on the terminals of trigeminal PAFs in CBVs, the rapidity of their actions in relieving migraine attacks has been incorporated as evidence into an alternative, vascular hypothesis of migraine (Ferrari and Saxena, 1993; Williamson et al., 1997). This vascular theory posits that the *primary* event in migraine is a vasodilation of CBVs (arteries) which secondarily activates trigeminal PAFs. A vascular trigger would, thus, be analogous to the role of injury to the skin in initiating excitatory and sensitizing events in cutaneous PAFs. Via actions at CGRP1 receptors coupled to AC, CGRP may play a role in the vasodilation of CBV arteries inasmuch as the levels of this powerful vasorelaxant increase during migraine attacks. Further, sumatriptan, via an action at 5-HT_{1B} receptors, suppresses the release and synthesis of CGRP in trigeminal pathways (Buzzi et al., 1991; Durham et al., 1997; Merhi et al., 1998; Olesen and Janssen-Olesen, 1997).

It is arguable that both neurogenic *and* vascular mechanisms are involved in the induction of migraine headaches and that they reciprocally reinforce each other. In any case, via direct, neuronal (decreased PAF activity) or indirect, vascular (vasoconstriction of CBVs) mechanisms, the inhibition of trigeminal input to the brain by sumatriptan and other serotoninergic agents can interrupt migraine attacks. Their actions are mediated by 5-HT_{1B}, 5-HT_{1D} and/or 5-HT_{1F} receptors on trigeminal terminals in the CBV (Bouchelet et al., 1996; Bruinvels et al., 1994; Gupta et al., 1995; Johnson et al., 1997; Lucaites et al., 1996; Phebus et al., 1996; Rebeck et al., 1994; Yu et al., 1997), and by 5-HT_{1B} and/or other 5-HT receptor types on CBVs themselves (Bouchelet et al., 1996; De Vries et al., 1996; Ferrari and Saxena, 1993; Gross, 1995; Hamel et al., 1993; Moskowitz, 1992; Razzaque et al., 1997; Saxena and Ferrari, 1989; Ferreira et al., 1997). Central actions at 5-HT_{1B} receptors localized on intrinsic PNs and/ or on the central terminals of trigeminal afferents in the trigeminal nucleus may also be involved (Castro et al., 1997; Cumberbatch et al., 1997; Goadsby and Hoskin, 1996; Hoskin et al., 1996). In addition, 5-HT_{1B} agonists may counter vascoconstriction of CBVs via their inhibitory influence upon the activity of sympathetic terminals innervating dural vessels (Jones et al., 1995; Molderings et al., 1996). Apart from 5-HT_{1B} and the other, above-specified classes of 5-HT receptor, other, as yet unidentified, 5-HT receptor types may also be involved in attenuating NI and/or vasodilation: in this regard, the possibility of species differences should be borne in mind (Gupta et al., 1995; Shepherd et al., 1997).

Despite the importance of actions at CBV-localized 5-HT_{1B} receptors in alleviating migraine attacks, it must be mentioned that stimulation of 5-HT_{1B} receptors on coronary vessels may lead to their vasoconstriction and cardiac side-effects (Terron, 1996). Further, action at 5-HT_{1D} receptors on the terminals of sympathetic fibre innervating the heart may likewise modify cardiac function (Molderings *et al.*, 1996).

7.9.3. Vasorelaxation as a Trigger of Migraine Headaches: a Role for NO and 5-HT?

As mentioned above, the vascular theory of migraine proposes that the vasodilation of CBVs (arteries) triggers migraine attacks. A multiplicity of factors may be involved in modulating the tone of CBVs and in precipitating their vasorelaxation, including CGRP derived from trigeminal afferents (Section 7.9.2), adrenergic mechanisms, endothelin, EAAs, thromboxanes and histamine (Bockman *et al.*, 1996; Edvinsson, 1982; Kalaria *et al.*, 1989; Matsubara *et al.*, 1992; Merhi *et al.*, 1998; Paul and Page, 1997; Raffa *et al.*, 1996). In addition, serotoninergic mechanisms may be of special importance in provoking the vasodilation of CBVs.

Serotonin in the CBVs is derived from central, raphe-localized serotoninergic neurones, circulating platelets, endothelial cells, mast cells and, following 5-HT reuptake, perivascular sympathetic elements (De Keyser *et al.*, 1993; Maruki *et al.*, 1984; Mathiau *et al.*, 1994; Stanley *et al.*, 1993). The 5-HT receptor type(s) involved in eliciting migraine attacks remain unclear. Certain authors have proposed a principal role for 5-HT_{2C} or, more likely, 5-HT_{2B} receptors, though based on a rather cavalier interpretation of the evidence (Fozard and Kalkman, 1994; Kalkman, 1994). As mentioned in Section 7.9.2 5-HT_{1F}, 5-HT₇ and/or other, as yet unidentified, 5-HT receptor types may also be of significance in modulating the functional status of CBVs (Cohen and Hamel, 1996; Connor and Beattie, 1996).

It has been suggested that an activation of central sources of 5-HT by 'stress', and/or the mobilization of peripheral pools of 5-HT, may trigger migraine attacks by an (endothelium-dependent) release of NO (Olesen *et al.*, 1993: Olesen and Janssen-Olesen, 1997). NO may also be involved in the NMDA receptor-mediated vasodilation of cerebral microvessels by GLU, a phenonemon of potential relevance to vascular processes in CBVs underlying headache (Fergus and Lee, 1997). CBVs contain mRNA encoding both H_1 and H_2 receptors and a role of NO in the vasodilatory actions of histamine at CBVs has also been proposed. Thus histamine relaxes CBVs by activating both:

- 1. endothelium-dependent H_1 receptors triggering (via PLC/[Ca²⁺]_i) the synthesis and release of NO which subsequently elicits vasodilation and NI; and
- endothelium/NO-independent H₂ receptors coupled to increases in cAMP in smooth muscle cells (Gimeno *et al.*, 1998; Janssen-Olesen *et al.*, 1997; Olesen and Janssen-Olesen, 1997; Olesen *et al.*, 1993; Yuan *et al.*, 1997).

In support of the hypothesis that NO is involved in the induction of migraine headache, it can trigger smooth muscle relaxation (CBV arterial vasodilation) and extravasation via the generation of cGMP and, ultimately, a decrease in free $[Ca^{2+}]_i$ (Lee *et* al., 1997b; Payá et al., 1997; Yuan et al., 1997) (Section 7.7). In addition, several other actions of NO might synergistically aggravate vasorelaxation and pain. For example, NO may interact with trigeminal terminals to faciliate neuropeptide release (Section 7.5.2.5), it increases PG levels via the induction of COX 2 and it enhances $TNF\alpha$ production from immunocompetent cells (Cohen et al., 1996; Fozard, 1995; Salvemini et al., 1996; Taiwo and Levine, 1988; Valentin et al., 1996; Verheggen et al., 1996; Wang et al., 1997e) (Section 7.7). The supersensitivity of migraine patients to the vasodilatory effects of NO on CBVs has been advanced in support of the argument that NO plays a key role in provoking the vasodilation of CBVs and migraine headaches (Olesen et al., 1993: Olesen and Janssen-Olesen, 1997).

Interestingly, a vasoconstriction of extracranial, non-CBV vessels (such as the temporal artery) also contributes to migraine pain in *ca* 30% of patients. 5-HT_{2A} and 5-HT_{1B} receptors mediate vasoconstriction of these extracranial CBVs—an effect *counteracted* by NO. These vessels also contain several other 5-HT types, including 5-HT₄ and 5-HT₇ receptors. However, it is unclear whether a perturbation of the function of 5-HT receptors is involved in provoking the vasoconstriction of extracranial vessels (Salamone *et al.*, 1997; Verheggen *et al.*, 1996, 1997).

Notwithstanding the role of neurogenic and central processes, as well as muscular mechanisms (pericranial muscle tenderness), migraine headaches provide a distinctive illustration of the involvement of vascular mechanisms in the generation of pain (Lipchick *et al.*, 1996; Procacci *et al.*, 1994).

8. NEUROPATHIC PAIN: PERIPHERAL PROCESSES

8.1. Experimental Models of Neuropathic Pain

Neuropathic pain refers to the pain caused by damage to nervous tissue (Tables 1, 3 and 6, Fig. 2). Although pain elicited by peripheral nerve injury is sometimes used synonymously with 'neuropathic' pain, this latter term also incorporates the 'central' pain associated with damage to the CNS: for example, the thalamic syndrome which (albeit rarely) may be a sequel of stroke (Boivie et al., 1989) (Section 13). Paradoxically, in addition to a disturbance of sensory and motor function and a disruption of RFs, one prominent feature of peripheral nerve damage may actually be an enhancement in pain 'sensation'. Indeed, partial damage to-and even destruction of-peripheral nerves may be associated with a variety of syndromes including an increase in the sensitivity to noxious stimuli (hyperalgesia), a painful response to normally innocuous mechanical and cold stimuli (allodynia), parasthesic and dysthesic sensations and spontaneous pain (Table 1). Of these changes, mechanical allodynia comprises the most striking perturbation of sensation. Thresholds to respond to innocuous heat stimuli are not invariably modified-although a thermal hyperalgesia is seen in several models of PAF damage (Hunter et al., 1996, 1997b; Shir and Seltzer, 1990). On the other hand, a hyperalgesia/ allodynia to cold may be pronounced, in particular sympathetically-maintained in painful states (Cervero and Laird, 1996b; Frost et al., 1988; Lindblom, 1985) (Section 8.2.4). This pattern of altered sensitivity in neuropathic pain clearly differs to the primary (heat) hyperalgesia of nociceptive pain, in which cooling of the site of injury may even reduce pain, and more closely resembles the secondary hyperalgesia which accompanies tissue damage, in particular as concerns the occurrence of mechanical allodynia (Section 9.1.2).

As explained in Section 7, peripheral mechanisms integrated in the terminals of nocisponsive PAFs appear to be the principal cause of primary hyperalgesia. In contrast, neuropathic pain—and secondary hyperalgesia—may largely be attributed to changes occurring *downstream* of PAF terminals. Extensive experimental studies of neuropathic pain, involving axotomy, or various models of sub-total nerve damage ('mononeuropathies'), have allowed for a detailed exploration of mechanisms underlying neuropathic pain due to PAF injury (Coyle, 1996).

An important question concerning the induction of prolonged, clinical pain is how damaged nerves trigger changes in the DH and higher structures. A related issue is whether input from injured PAFs needs to be maintained for pain to persist (Section 9.1.3) (Fig. 2 and Table 6). In this regard, it is of note that PAF injury is associated with the rapid, intense, central discharge of both $A\beta$ and C fibres for a period of minutes-and even several days for some individual fibres (Seltzer, 1995). This pronounced discharge of PAFs upon their injury mimics their intense physiological stimulation (via nociceptor activation) and may elicit central processes of sensitization (Sections 10 and 12). Indeed, for several animal models, an initial discharge of damaged PAFs post-injury is a critical factor in the development of allodynia and other pain-related behaviours. In other models, it may play a minor role and subsequent events are of greater importance (Dougherty et al., 1992a; Seltzer, 1995). In any case, injury to PAFs may be followed by their 'ectopic' firing (independently of nociceptor stimulation) both spontaneously and in response to sympathetic, mechanical and other modes of stimulation (Babbedge et al., 1996; Devor et al., 1994; Kajander and Bennett, 1992; Sheen and Chung, 1993) (Sections 8.2.2 and 8.2.4).

Table 6. Summary of principal mechanisms underlying the induction of prolonged, painful states due to PAF injury (see Sections 8 and 11)

- 2. Increase in the spontaneous excitability and responsiveness of damaged PAFs. Mediated by changes in the expression levels and operating characteristics of multiple ion channel subtypes localized in the DRG and/or neuroma of PAFs: various classes of Ca²⁺-, K⁺- and, in particular, Na⁺- channel are involved
- 3. Abnormal patterns of inter-neuronal communication in the DRG and/or neuroma: 'Crossed-after-discharge' (possibly mediated via a diffusable mediator) and 'ephaptic transmission' (reflecting direct, interneuronal current flow)
- 4. Increased sympathetic innervation and excitation of the DRG and/or neuroma of PAFs. Mediated via actions of NAD at α_{2^-} (and α_{1^-}) adrenergic receptors, as well by the release of other mediators, such as PGs
- 5. Altered phenotype of damaged, small calibre C fibres: for example, an induction of pronociceptive mediators such as VIP and NO
- 6. Altered phenotype of damaged, large calibre $A\beta$ fibres: notably, an induction of SP
- 7. Aberrant patterns of peripheral regeneration of damaged PAFs, and alterations in their functional properties. In addition, sprouting of intact, collateral PAFs into areas of denervated, peripheral tissue
- B. Central mechanisms
- 1. Reorientation of the DH terminals of $A\beta$ fibres-'primed' by PAF injury—into alien territories: in particular, migration from *non*-nocisponsive laminae III/IV into lamine II₀, a region involved in the reception and transmission of nociceptive information. C fibres and descending pathways may also invade inappropriate regions of the DH
- 2. Functional reduction in the activity (or physical degeneration) of those ININs which normally suppress the supraspinal flow of nociceptive information via inhibitory actions at PNs, EXINs and PAF terminals
- 3. Other central events specified in Table 5

A. Peripheral mechanisms

^{1.} Activation of damaged (and adjacent, intact) PAFs by inflammatory mediators (see Table 5, Peripheral Events)

In a representative model of damage to the sciatic nerve, such questions were addressed quantitively and it was shown that the mechanical sensitivity of certain, neurectomized A β fibres developed rapidly over hours and peaked within weeks, thereafter declining to a low level (Babbedge et al., 1996; Devor, 1994, 1995). On the other hand, the onset of their spontaneous firing took some days to occur, peaked (30% of neurones) within a week and returned to a level of 3-4% after a month (Babbedge et al., 1996). C fibres revealed a different pattern, with spontaneous firing emerging later and totalling 2-3% of neurones. Apparently, then, under some conditions, activity in a relatively small population of affected PAFs may suffice to maintain changes in nociception which can be blocked by destruction of the injured nerve or application of anaesthetics thereto (Kajander and Bennett, 1992; Seltzer, 1995) (Section 9.1.3).

In fact, such quantitative and temporal aspects of the sensory disruption provoked by PAF damage are complex, model-dependent and difficult to relate to the clinical situation. Indeed, the characteristics of sensory deficits resulting from PAF injury, the delay to their appearance, the time required for resolution of symptomatology and the underlying mechanisms all may depend on the type of lesion and the precise experimental paradigm employed (Coyle, 1996). Further, the relative importance of input from injured and intact PAFs in mediating alterations in nociception may be variable. Nevertheless, it is possible to identify several important, peripheral and central events common to various models of PAF injury and which likely contribute to the clinical features of neuropathic pain due to peripheral nerve damage (Bennett and Xie, 1988; Coderre, 1992; Coderre et al., 1993; Coyle, 1996; Guilbaud and Benoist, 1995; Kim and Chung, 1992; Koltzenburg, 1995; Na et al., 1996; Woolf and Doubell, 1994; Yoon et al., 1996) (Section 11).

8.2. Peripheral Mechanisms Underlying Neuropathic Pain

8.2.1. Inflammatory Events: Focus on Cytokines

Certain of those mechanisms described in Section 7 which underlie inflammatory pain may contribute to rapid changes in nociception resulting from peripheral nerve injury (Michaelis et al., 1997; Tracey and Walker, 1995). Chronic constriction-induced nerve injury is accompanied by a pronounced, local inflammatory response and chromic, nerve ligatures (which release chemicals) are more effective in eliciting hyperalgesia than non-chromic sutures. Further, hyperalgesia can be attenuated by blocking the inflammatory response to ligation (Clatworthy et al., 1995; Maves et al., 1993). In line with these observations, degenerating, nocisponsive neurones release pro-inflammatory mediators (SP, CGRP, etc). Further, at the site of nerve damage and in neighbouring injured tissue, there is a local, inflammatory response which involves acidosis and infiltration by mast cells, macrophages and other immunecomponent cells releasing 5-HT, ATP, NO, leukotrienes,

PGs, NGF and cytokines etc (Perry *et al.*, 1993; Silver *et al.*, 1996; Yoder *et al.*, 1997; Wagner and Myers, 1996b). Proliferating Schwann cells in the vicinity of damaged nerves provide an important source of PGs, NGF, other neurotrophins, NO and the cytokine, LIF (Bolin *et al.*, 1995; Fu and Gordon, 1997; Wagner and Myers, 1996b; Zochodne *et al.*, 1997). LIF may modify nociception and plays an important role in modifying the phenotype and growth of injured PAFs and (intact) sympathetic fibres (Sections 8.2.6.2 and 11.3.4). The sympathetic innervation of damaged nerves is also intensified, and sympathetic terminals constitute a further source of PGs, ATP, NAD and other potential pro-inflammatory and pronociceptive mediators (Section 8.2.4).

Evidently, where the appropriate receptors and transduction mechanisms are available on damaged-and intact-PAF terminals, inflammatory mediators may exert direct and indirect actions contributing to an increase in nociception (Frisén et al., 1993; Michaelis et al., 1997; Tracey and Walker, 1995). Indeed, the prostaglandin, PGI₂, and the leukotriene, di-HETE, has been shown to enhance ectopic (non-terminal) discharges of damaged C (but not A) afferents in neuromas and to excite otherwise-silent C fibres, suggesting that they increase the afferent barrage to the DH (Devor et al., 1992b; Wagner and Myers, 1996a). Recently, PAF injury was shown to up-regulate B₂ receptors and to induce the 'de novo' expression of B₁ receptors in the DRG, suggesting a role for BK in the excitation of damaged PAFs (Petersen et al., 1998a).

Macrophages are of particular interest since they are the principal effector cells involved in processes of 'Wallerian' degeneration following injury to myelinated nerves (Fu and Gordon, 1997) (Section 11.3.4.1). By removing axonal myelin, this mechanism plays an important role in generating conditions permitting the successful regeneration of damaged PAFs. Possibly due to a relative absence of macrophages, genetic strains of mice in which Wallerian processes are defective show a delayed time-course of hyperalgesia following nerve injury (Myers et al., 1996; Ramer et al., 1997). Indeed, macrophages secrete several pro-inflammatory and pronociceptive cytokines. One important example, TNFa (Section 7.7), is involved in demyelination processes (Redford et al., 1995) and exerts trophic or toxic effects dependent on the cell type targetted and several other factors (Chao et al., 1996; Shen et al., 1997). Levels of TNF α are increased by nerve injury (Wagner and Myers, 1996a,b; Wagner et al., 1998; Wells *et al.*, 1992). Notably, then, $TNF\alpha$ induces ectopic (burst) firing in nocisponsive PAFs, while its local injection into the sciatic nerve elicits mechanical allodynia and associated pathophysiological changes resembling those provoked by PAF injury (Sorkin et al., 1997; Wagner and Myers, 1996a,b). Further, inhibition of the actions of $TNF\alpha$ reduces hyperalgesia due to PAF injury (Redford et al., 1995; Safieh-Garabedian et al., 1995). These data suggest a pronociceptive function of $TNF\alpha$ at damaged PAFs. There are several potential mechanisms. TNF α enhance the activity of COX 2 and increases the density of B_1 receptors on PAFs and sympathetic terminals. Further, it elevates tissue levels of IL-1 β which, in turn, stimulates the synthesis and release of NGF from Schwann cells, and NGF may *intervene* in the induction of B_1 receptors on PAF terminals by TNFa (Hikawa and Takenaka, 1996; Horie et al., 1997b; Lindholm et al., 1987; Petersen et al., 1998b; Ramer et al., 1997; Rueff et al., 1996; Watkins et al., 1995) (Section 7.7). The complex role of NGF in the modulation of nociception at intact and damaged PAFs is described in Sections 7.6.1, 7.7, 8.2.4 and 8.2.6. A final, intriguing aspect is that cytokines, such as $TNF\alpha$ and ILs, may actually be induced in PAF themselves following their injury. A putative functional role of such intrinsic PAF pools of cytokines would be of interest to delineate (Murphy et al., 1995).

8.2.2. Alterations in the Density and Function of Ion Channels in Sensory Neurones

8.2.2.1. Ectopic activity of damaged PAFs

Injured small and large diameter PAFs show marked alterations in their patterns of excitability and conduction properties, both spontaneously and in response to excitation. Such changes reflect alterations in the density and/or operating characteristic of multiple ion channels and result in the generation of abnormal patterns of electrical impulses and afferent input to the DH (Fields et al., 1997; Won et al., 1997; Zhang et al., 1997a). Thus, following injury, nerve endings may seal off to form an end-bulb or attempt, unsuccessfully, to sprout, resulting in a confused growth termed a neuroma. Together with their soma in the DRG, neuromas become the site of 'ectopic' (non-terminal) firing (Amir and Devor, 1997; Devor, 1994; Devor and Dubner, 1988; Fields et al., 1997; Han et al., 1994; Kajander and Bennett, 1992; Kajander et al., 1992; Xie et al., 1995b). Such discharges are seen for C and $A\beta$ fibres in each case and occur both spontaneously and in response to stimulation (Amir and Devor, 1992, 1993; Babbedge et al., 1996; Study and Kral, 1996). Indeed, neuromas develop a marked hypersensitivity to mechanical stimuli (and sympathetic input) (Babbedge et al., 1996; Burchiel, 1984; Devor et al., 1994; Matzner and Devor, 1987). Of particular pertinence to mechanical allodynia, the discharge pattern of low threshold, mechanosponsive PAFs is modified (Na et al., 1993). The significance of multiple classes of ion channels in controlling electrical activity in injured PAFs may be summarized as follows.

8.2.2.2. *Ca*²⁺-*Channels*

Voltage-sensitive, N-type Ca²⁺-channels are the principal class involved in controlling release from the terminals of sensory, central and sympathetic neurones (Diaz and Dickenson, 1997; Diversé-Pierluissi *et al.*, 1997; Wright and Angus, 1996). Their blockade by verapamil and conopeptide ω toxin analogues reduces neuropathic pain (for example, mechanical allodynia) and inhibits spontaneous, electrical activity in ectopically-active DRG cells. These findings are consistent with a role of overactive N-type, Ca²⁺-channels in the induction

of ectopic firing and nociception upon PAF injury (Bowersox et al., 1996; Chaplan et al., 1995; Devor, 1994; Xiao and Bennett, 1995; Wright and Angus, 1996; Zhang et al., 1994b). Nevertheless, in addition to actions at peripheral populations of N-type channels on injured PAFs, blockade of *DH-localized* N-(and P-type) Ca²⁺-channels localized on the central terminals of damaged PAFs-and on intrinsic DH neurones-may also be of significance (Bowersox et al., 1996; Chaplan et al., 1995; Miljanich and Ramachandran, 1995; Nebe et al., 1997; Neugebauer et al., 1996; Sluka, 1997). In contrast to N-type Ca2+ -channels, selective blockade of L-type ⁺-channels may be less effective in moderating nociception provoked by PAF injury. However, the comparative significance of specific classes and discretely-localized populations of various Ca2+-channel types in the mediation of neuropathic-as compared to nociceptive-pain remains under investigation (Bowersox et al., 1996; Chaplan et al., 1995; Diaz and Dickenson, 1997; Malmberg and Yaksh, 1994; Miljanich and Ramachandran, 1995; Nebe et al., 1997; Neugebauer et al., 1996; Sluka, 1997). The ability of the anti-convulsants, carbemazepine and gabapentin, to reduce mechanical allodynia, both in the clinic and in experimental models of neuropathic pain, may involve, amongst other mechanisms, an interaction with Ca²⁺-channels localized on the DRG, neuroma and central terminals of injured PAFs (Abrams, 1996; Field et al., 1997; Fields et al., 1997; Galer, 1995; Gee et al., 1996; Hunter et al., 1997b; Stanfa et al., 1997; Todorovic and Lingle, 1998) (Section 13.4).

In line with a pathological role of Ca^{2+} -channels in controlling the electrical activity of injured PAFs, a facilitation of Ca^{2+} -currents and an elevation of $[Ca^{2+}]_i$ levels contributes to the increase in excitability of damaged PAFs elicited by NGF (Jiang *et al.*, 1997; Sherwood *et al.*, 1997).

8.2.2.3. Na⁺-Channels

The multiplicity of Na⁺-channels present in the DRG of PAFs may, in terms of operational characteristics, be classified into two types.

- 1. TTX-sensitive, low threshold channels which rapidly activate and inactivate; and
- 2. TTX-resistant, high threshold channels with slower kinetics of activation and inactivation.

The distribution of these different classes of Na⁺channel on various populations of PAFs is still under evaluation, but certain interesting differences between small calibre, TRK A-bearing, CGRP-positive PAFs and large, TRK C-bearing PAFs have, for example, been established (Friedel et al., 1997). It is likely that a differential alteration in the density and/or properties of specific Na+-channels upon injury to PAFs is of functional importance in determining changes in their discharge properties and AP profiles (Schild and Kunze, 1997). Overall, as outlined below, a relative increase in the ratio of kinetically-rapid, TTX-sensitive vs kinetically-slow, TTXresistant channels may be elicited by PAF injury (Akopian et al., 1996a,b; Black et al., 1996; Friedel et al., 1997; McGiven et al., 1997; Kobayashi et al., 1996; Oyelese et al., 1997; Schaller et al., 1995; Sangameswaran et al., 1996, 1997).

The accumulation of Na⁺-channels in damaged sensory fibres, in particular in areas of demyelination, leads to hyperexcitability and electrical instability: indeed, irrespective of their class, the total density of Na⁺-channels is proportional to the likelihood of repetitive firing in response to mild stimuli. Further, in line with the presence of a high proportion of TTX-sensitive Na⁺-channels, TTX inhibits ectopic impulses from damaged nerves and their DRG (Devor et al., 1992a Devor et al., 1993; Matzner and Devor, 1994; Omana-Zapata et al., 1997a,b). As concerns individual classes of Na⁺channel, the C fibre-localized, TTX-resistant PN 3 channel (also termed the 'SCN 10A' gene or the ' α -SNS' transcript)—which is involved in inflammatory pain (Section 7.5.2.3)—is down-regulated upon PAF damage (Akopian et al., 1996a, b; Dib-Hajj et al., 1996; Cummins and Waxman, 1997; Friedel et al., 1997; Oaklander and Belzberg, 1997). On the other hand, a novel, TTX-sensitive ('a-III') channel which is not usually encountered in adult neuronal tissue, appears in fine calibre C fibres following their injury (Black et al., 1996, 1997; Cummins and Waxman, 1997; Kobayashi et al., 1996; Oyelese et al., 1997; Waxman et al., 1994). This Na⁺-channel quickly reprimes (recovers from inactivation) and may predispose nerves to rapid and abnormal firing (Cummins and Waxman, 1997). Further, damage to large diameter $A\beta$ fibres provokes a similar increase and decrease in TTX-sensitive and TTX-resistant Na⁺ channels, respectively (Rizzo et al., 1994). A recently-cloned, 'PN 1' TTX-sensitive channel is found both in small and large diameter DRG neurones-as well as sympathetic neurones-suggesting a possible role in mechanical allodynia (Toledo-Aral et al., 1997) (Section 7.6.1). Its relationship to the above-mentioned up-regulation of TTX-sensitive Na⁺-channels would be of interest to establish.

In addition to changes in the expression levels of specific Na⁺-channels, their *activity* (passage of current and inactivation kinetics etc) may be rapidly modified by processes of phosphorylation elicited by inflammatory agents (Sections 7 and 8.2.1), NGF and other modulators of $[Ca^{2+}]_i$ levels and PK activity (Berninger and Poo, 1996; Colbert and Johnston, 1998). Indeed, the alteration of access to NGF and other neurotrophins which results from their injury may play an important role in modifying the expression and operating properties of multiple classes of Na⁺-channel and, thereby, modulating the electrical properties of PAFs (Black *et al.*, 1997; McMahon and Bennett, 1997; Toledo-Aral *et al.*, 1997) (Section 7.6.1).

Systemic administration of Na⁺-channel antagonists/local anaesthetics, such as lidocaine and mexilitine, relieves experimental and clinical neuropathic pain. Their actions may, at least partially, reflect blockade of peripheral (TTX-sensitive) Na⁺-channels in damaged PAFs (Barann *et al.*, 1993; Boas *et al.*, 1982; Calcutt *et al.*, 1996; Chapman *et al.*, 1997; Cummins and Waxman, 1997; Kastrup *et al.*, 1987; Khandwala *et al.*, 1997; Kingery, 1997; McGiven *et al.*, 1997; Nishiyama and Sakuta, 1995; Tanelian and MacIver, 1991). A blockade of voltage-depen-

dent (TTX-resistant and -sensitive) Na+-channels in the neuroma and DRG of injured PAFs may also be involved in the anti-allodynic actions of anti-convulsants, such as lamotrigine and phenytoin (Abrams, 1996; Blackburn-Munro and Fleetwood-Walker, 1997; Galer, 1995; Hunter et al., 1997b; Rush and Elliott, 1997; Trezise et al., 1997; though see Chapman et al., 1997) (Section 13.4). Further, the analgesic actions of local anesthetics are expressed without an apparent interference with axonal conductance, consistent with the argument that they act at DRG-localized, Na⁺-channels mediating processes underlying ectopic activity (Devor et al., 1992a). Nevertheless, in analogy to Ca^{2+} -channels, an additional blockade of Na⁺-channels localized on the central terminals of PAFs-or on intrinsic DH neurones-might also be involved in the reduction of neuropathic pain by such agents (Calcutt et al., 1996; Fraser et al., 1992; Omana-Zapata et al., 1997a,b; Pertovaara et al., 1996).

8.2.2.4. *K*⁺-*Channels*

In addition to changes in the density and properties of cation-permeable channels in damaged PAFs, a reduction in the density, or an alteration in the functional properties, of K+-channels may also increase excitability (Gold et al., 1996b; Rasband et al., 1998). Indeed, it has been shown that the excitation of damaged nerves by NAD involves a decrease in K⁺-conductance (Abdulla and Smith, 1997) (Section 8.2.4). It is not, as yet, clear whether changes in K⁺-channel activity are a primary response to injury, or whether they are contingent upon changes in the properties of Ca²⁺ - and Na⁺ channels (Devor, 1983; Elliott, 1997). In fact, it has been suggested that a Ca^{2+} -dependent *increase* in K⁺-conductance may be triggered by ectopic firing leading to a period over which burst firing is suppressed (Amir and Devor, 1997). In addition to Na⁺-channel blockade (Section 8.2.2.3), a facilitation in K⁺-currents may be involved in the attenuation of neuropathic pain by the anaesthetic agent, mexilitine (Khandwala et al., 1997; Kingery, 1997; Sato et al., 1995). Sensory neurones possess multiple, voltage-gated K⁺-currents, the significance of which would be of interest to examine (Gold et al., 1996b).

8.2.2.5. Spontaneous electrical activity, ion currents, $[Ca^{2+}]_i$ and PAF regeneration

One final and intriguing aspect of the electrical activity of damaged PAFs should be mentioned. Although excessive $[Ca^{2+}]_i$ can kill cells (Choi and Rothman, 1990; Lerea, 1997), a certain level of $[Ca^{2+}]_i$ may be required for DRG survival. Indeed, $[Ca^{2+}]_i$ plays a diverse and key role in controlling the activity and growth of injured PAFs, and their electrical activity may act as a mode of 'trophic' support for DRG cells independently of NGF and other neurotrophins (Benowitz and Routtenberg, 1997; Bito *et al.*, 1997; Ghosh and Greenberg, 1995; Gu and Spitzer, 1995; Hegarty *et al.*, 1997; Sjaastad and Nelson, 1997). Thus, inasmuch as damaged PAFs (transiently) lose access to NGF and other trophic factors, a continued ionic flux-maintained spontaneously and/or in response to stimulation may *itself* be of significance in processes underlying PAF survival and regeneration (Section 11.3).

8.2.3. Abnormal Patterns of Inter-neuronal Communication

The ectopic activity of DRG cells and neuromas may be intensified (and hyperexcitability accentuated) by 'non-synaptic' interactions between neurones. This refers to the phenomenon whereby activity in an individual neurone (or a group of neurones) modifies activity in adjacent neurones. One mode of inter-neuronal communication, termed 'crossed-after-discharge', involves the depolarization of neurones upon the repetitive firing of their neighbours (Amir and Devor, 1996; Devor, 1994, 1995; Fried et al., 1993; Lisney and Devor, 1987). This process, which primarily drives large calibre neurones, is seen at the level of the DRG. Therein, under normal conditions, the consequences of minor depolarizations are limited but, following nerve damage, they may exaggerate ectopic activity, set up reverberating (bursting) circuits and recruit silent neurones (Amir and Devor, 1997). The mechanisms underlying cross-excitation in the DRG are unclear, although there is evidence that diffusible factors, such as ATP or, more likely, an elevation in extracellular K⁺-concentrations, may be involved (Amir and Devor, 1996, 1997; Shinder and Devor, 1994; Utzschneider et al., 1992). Notably, following injury, neurones and neuromas become more accessible to diffusible substances owing to a loss of their normal insulation (Shinder and Devor, 1994).

Electronic coupling has been recorded in peripheral neurones, and a further mode of excitatory cross-talk is termed 'ephaptic' (Lekan *et al.*, 1996). This is seen both at neuromas and at axons in areas of demyelination. It may develop both transiently and rapidly following injury, or several weeks thereafter. The apposition of the membranes of adjacent axons, leading to direct current transfer from one to another, has been implicated (Amir and Devor, 1992; Blumberg and Jänig, 1982; Seltzer and Devor, 1979) (Section 4.7).

These modes of inter-neuronal communication are of particular interest since they might occur between fibres of *different* classes. Should low threshold $A\beta$ fibres activate high threshold C fibres, this mechanism could contribute to the mechanical allodynia whereby a normally innocuous touch stimulus would be perceived as painful (Section 9.1.2). [Interestingly, an alternative theory suggests that, where PAFs are *intact*, $A\beta$ fibres may evoke allodynia by indirectly recruiting C fibres at the level of the DH (Cervero and Laird, 1996a,b; Sluka *et al.*, 1993) (Section 9.2.3).]

8.2.4. Sympathetic Excitation of Damaged Sensory Neurones

8.2.4.1. Increased sympathetic innervation and excitation of the DRG upon PAF injury

The sympathetic nervous system may be involved in sensory changes resulting from damage to peripheral nerves (Coyle, 1996; Drummond, 1996; Fields and Rowbotham, 1994; Kinnman and Levine, 1995; Hu and Zhu, 1989; Koltzenburg, 1995; Raja, 1995; Raja *et al.*, 1996; Treede *et al.*, 1990b). Under normal conditions, sympathetic fibres exert little direct influence upon peripheral, sensory nerve endings (Section 7.4.7). However, following injury, PAF responsiveness to sympathetic stimulation is markedly augmented. Such changes are apparent for both the neuroma and the DRG. In addition, some intact, PAF endings in their vicinity may also manifest a (less pronounced) increase in sympathetic sensitivity. Affected PAFs develop a pronounced response to stimulation of the appropriate sympathetic nerve trunk and to (topical or systemic) administration of α -AR agonists, such as NAD (Davis *et al.*, 1991; Devor *et al.*, 1994; Sato and Perl, 1991).

The sympathetic innervation of the DRG normally terminates within the surrounding vasculature and its role is largely limited to the control of blood supply. However, upon nerve damage, sympathetic afferents-with a time-course corresponding to the appearance of mechanical allodynia-extend into the DRG to form basket-like structures around cell bodies, in particular large soma corresponding to low threshold $A\beta$ (and proprioceptive) fibres. Smaller neurones may also-at least transiently-be affected (Chung et al., 1996; McLachlan et al., 1993; Ramer and Bisby, 1997a,b; Zhou et al., 1996b). Following PAF injury, electron microscope studies have also revealed close contacts between sympathetic endings and sensory neurones in the DRG (Chung et al., 1997). As a consequence, both NAD-and, in theory, other transmitters released by stimulation of sympathetic terminals-can now modulate the activity of DRG cells. While some hitherto-silent soma in the $A\beta$ fibre range are recruited, it is the activity of spontaneously firing cells which is most markedly accelerated.

The above observations suggest that, following injury to PAFs, the development of an abberrant sympathetic input to the DRG amplifies both their spontaneous and evoked activity and, correspondingly, enhances the ectopic barrage to the DH. Correspondingly, chemical or surgical sympathectomy (which is more effective if performed prior to nerve injury) can alleviate mechanical allodynia and other symptoms of neuropathic pain in the majority-though not all-experimental models (Chung et al., 1996; Coyle, 1996; Kim and Chung, 1991; Lee et al., 1997a). Further, guanethidine, which depletes peripheral catecholamines and other sympathetic transmitters, can relieve the pain of sympathetic-dependent painful states in some-but not all-human subjects (Torebjörk et al., 1995). Similar results are, generally, obtained with adrenergic antagonists (Ghostine et al., 1984; Raja, 1995; Raja et al., 1996; Xie et al., 1995a; Wahren et al., 1995). Contrariwise, pain can be rekindled by local administration of NAD into the injured tissue (Coyle, 1996; Torebjörk et al., 1995; Wallin et al., 1976; Xie et al., 1995a).

Irrespective of the mechanisms underlying sympathetic activation of damaged PAFs (Section 8.2.4.2), the ability of chemical sympathetic blockade to *rapidly* alleviate sympathetically-maintained pain (even within minutes) is consistent with the concept that the central sensitization underlying pain is maintained by peripheral input (Section 9.1.3) and, in analogy to other painful states, C *fibres* likely fulfil this role (Cervero and Laird, 1996a,b; Chen *et al.*, 1996; Gracely *et al.*, 1992; Woolf and Wall, 1986). Thus, while an excitation by NAD/ α -ARs—and other sympathetic mechanisms (Section 8.2.4.2)—of large calibre A β fibres may directly provoke mechanical allodynia, the concurrent facilitation of a continuous, low level of C *fibre* input into the DH contributes to the maintenance of *permissive*, central processes of sensitization (Andersen *et al.*, 1995; Choi and Rowbotham, 1997; Koltzenburg *et al.*, 1994) (Section 7.4.7).

An interesting question concerns those factors which permit and encourage the extension of sympathetic fibres into the DRG and neuroma of damaged nerves. In this regard, there is evidence that NGF (a trophic factor for sympathetic nerves) and cytokines may be involved (Cowen and Gavazzi, 1998; Davis et al., 1997a; Kawaja and Crutcher, 1997). Following PAF damage (and Wallerian degeneration of myelinated neurones), proliferating Schwann cells and fibroblasts release NGF, and an increase in the levels of mRNA encoding NGF has been detected in the DRG, in parallel with an increase in levels of p75^{NTR}, the low affinity NGF receptor, in glial cells (Ramer and Bisby, 1997a,b). Thus, NGF may attract and initiate the sprouting of sympathetic fibres (Davis et al., 1994, 1996). This action is presumably exerted in cooperation with other extrinsic, trophic factors, and with growth-associated substances contained in sympathetic terminals (Cowen and Gavazzi, 1998; Heumann et al., 1987a Heumann et al., 1987b; Michael and Priestley, 1995; Ramer et al., 1997; Sebert and Shooter, 1993) (Section 11.3). Interestingly, evidence for a reciprocal interaction between sympathetic neurones and NGF under conditions of neuropathic nerve damage was provided by the observation that the activation of β -ARs enhances the synthesis of NGF in a model of diabetic neuropathy (Devor et al., 1994; Riaz and Tomlinson, 1997; Xie et al., 1995a,b). Sympathetic fibres also bear TRK C receptors and a potential, synergistic role of NT 3 with NGF in encouraging sympathetic sprouting would be of interest to evaluate (Tafreschi et al., 1998). As regards cytokines, the release of ILs, such as IL-6, by immunocompetent cells involved in Wallerian degeneration may also encourage the sprouting of sympathetic terminals into damaged PAFs and their DRG (Bolin et al., 1995; Ludlam et al., 1995). Further, the cytokine, LIF, can also provoke the sprouting of sympathetic fibres into the DRG and it is massively induced in Schwann cells following PAF injury (Thompson and Majithia, 1997) (Sections 8.2.6.2 and 11.3.4). Intriguingly, ILs can also elicit LIF synthesis in sympathetic terminals themselves (Jonakait, 1993), suggesting that they might further 'auto-enhance' their own sprouting.

Notwithstanding the extension of sympathetic terminals into regions where they may directly access injured (and intact) PAFs, there is little or *no* evidence for an increase in local sympathetic outflow into the circulation. On the contrary, under certain conditions, there may even be a *diminution* in sym-

pathetic tone (Blumberg and Jänig, 1994; Devor, 1994, 1995; Drummond *et al.*, 1991; Harden *et al.*, 1994; Wallin *et al.*, 1976) (Section 8.2.4.3).

8.2.4.2. Mechanisms underlying sympathetic excitation of damaged nerves: a role for α -ARs on PAFs or sympathetic terminals

It is currently unclear how damaged PAFs develop an increase in their sensitivity to sympathetic input. One possibility is an upregulation in the density of a-ARs (Bossut et al., 1996; Cho et al., 1997a; Nishiyama et al., 1993; Perl, 1994; Petersen et al., 1996; Sato et al., 1993a, 1994). Such an increase could be triggered by a *reduction* of local, sympathetic outflow which has, in analogy, been suggested to provoke a supersensitivity (up-regulation) of vascular *a*-ARs (Arnold et al., 1993) (Section 8.2.4.3). Further, in line with this possibility, although sympathetic terminals may eventually sprout into the DRG and re-access injured PAFs (Section 8.2.4.1), this may not occur immediately following PAF injury and only a subpopulation of PAFs are affected. In addition to an increase in the *density* of α -ARs, their *responsiveness* to stimulation may be enhanced. Indeed, it was recently shown that PAF injury modifies the activity of a population of α_2 -ARs coupled to N-type Ca²⁺channels and a topographically-linked, Ca2+-dependent K⁺-channel (Abdulla and Smith, 1997). Thus, following PAF injury, the activation by NAD of α_2 -ARs on small DRG neurones ultimately leads to a reduction in K⁺-currents. This results in an increase in PAF excitability and an enhancement in the probability of firing upon stimulation.

The findings of this latter study are consistent with the majority—though not all—reports that α_2 -ARs are involved in the excitation of PAFs following injury (Campbell et al., 1992; Chen et al., 1996; Gold et al., 1997; O'Halloran and Perl, 1997; Hong and Abbott, 1996; Leem et al., 1997; Millan, 1997; Nicholas et al., 1996; Roberts and Elardo, 1985; Sato and Perl, 1991; Xie et al., 1995a; Zhang et al., 1997b). A role of α_2 -ARs in the activation of damaged PAFs would, further, be consistent with the presence of mRNA encoding α_{2A} -, α_{2B} - and predominantly, α_{2C} -AR in DRG (Gold *et al.*, 1997; Millan, 1997; Nicholas et al., 1996). Indeed it is of interest that an increase and decrease in levels of α_{2A} - and α_{2C} -ARs, respectively, was documented in the DRG of rats with peripheral nerve damage (Cho et al., 1997a). However, the respective role(s) of peripheral α_{2A} -, α_{2B} - and α_{2C} -AR subtypes is difficult to experimentally determine in view of the confounding antinociceptive actions of α_2 -AR agonists at central α_{2A} -ARs (Hayashi and Maze, 1993; Hunter *et al.*, 1997a; Millan, 1997; Millan et al., 1994). Anatomical support for the existence of α_1 -ARs in PAFs is weak (Nicholas et al., 1996; Pieribone et al., 1994) and there is only limited clinical and pharmacological evidence for a role of α_1 -ARs on (damaged or intact) sensory neurones in mediating the excitatory actions of NAD (Chen et al., 1996; Choi and Rowbotham, 1997; Davis et al., 1991; Drummond et al., 1991; Hong and Abbott, 1996; Pieribone et al., 1994; Ouseph and Levine, 1995) (Section 7.4.7).

Thus, although the pharmacology of the actions of NAD on the neuroma and DRG of damaged nerves requires further exploration, currently, the data favour a principal role of α_2 - rather than α_1 -ARs (Section 7.4.7).

Apart from actions at α_1 - and α_2 -ARs on PAFs, NAD may also control the activity of injured PAFs *indirectly* via actions at *sympathetic* terminals themselves. Indeed, like PAFs, sympathetic terminals express multiple α_2 -AR subtypes (Campbell *et al.*, 1992; Chen *et al.*, 1996; Gold *et al.*, 1997; Hong and Abbott, 1996; Hunter *et al.*, 1997a; Roberts and Elardo, 1985; Nicholas *et al.*, 1996; Sato and Perl, 1991; Tracey *et al.*, 1995). Such mechanisms may involve:

- 1. modulation of NAD release by α_{2A} -AR (and β -AR) autoreceptors; and
- 2. an increase in the release of PGs elicited by α_{2C} -ARs (Burnstock, 1996; Davisson *et al.*, 1996; Khasar *et al.*, 1995; Levine and Taiwo, 1994; Levine *et al.*, 1986b; Tracey *et al.*, 1995) (Section 7).

However, the generality of this latter mechanism has been challenged by observations that α_2 -AR agonists elicit ectopic discharges in intact and injured A β and C fibres of DRG even in the *absence* of an intact sympathetic innervation (Rubin *et al.*, 1997; Zhang *et al.*, 1997b) (Section 8.2.4.3).

To summarize, it appears that the predominant mechanism for the excitation of damaged PAFs by NAD is a direct action of NAD at upregulated and/ or oversensitive α_2 -ARs on PAFs, through the underlying molecular mechanisms remain unclear.

8.2.4.3. Other sympathetic mechanisms potentially modulating the activity of PAFs

Several other general mechanisms whereby the sympathetic system may modulate the activity of both injured- and intact-PAFs are most appropriately considered here.

Sympathetic fibres may indirectly affect the activity of PAFs via changes in vascular permeability and blood flow—and a role of the sympathetic system in the modulation of plasma extravasation and other inflammatory events has been proposed (Miao *et al.*, 1996; Takahashi *et al.*, 1993). Specifically, the vascular (vasoconstrictor) actions of sympathetically-released NAD may encourage pain by:

- 1. reducing dispersion of inflammatory substances; or
- provoking an oedema which activates pressuresensitive nociceptors (Blumberg and Jänig, 1994; Drummond, 1996).

Such actions may be of importance in diabetic neuropathies and sympathetically-maintained painful states in which an exaggerated vasoconstrictor response to NAD has been observed (Arnold *et al.*, 1993; Blumberg and Jänig, 1994; Drummond, 1996; Paul and Page, 1997; Vo and Tomlinson, 1997). This probably reflects an above-mentioned compensatory overactivity of vascular α -ARs provoked by a decrease in sympathetic tone (Arnold *et al.*, 1993). Nevertheless, vascular mechanisms *cannot* fully account for the hyperalgesic actions of NAD and

the pronociceptive influence of sympathetic transmission since the hyperalgesic actions of NAD (in inflamed tissue) are *greater* than those provoked by ischaemia alone (Drummond, 1996; Paul and Page, 1997).

The trophic factor for sympathetic fibres, NGF, increases the gene expression of NPY in sympathetic neurones (Sabol and Higuchi, 1990; Tracey et al., 1995). Thus, a NGF-induced increase in the release of NPY from sympathetic terminals may be involved in the modulation of nociception, either via actions on small calibre PAFs or via feedback actions to release NAD, PGs and other mediators from sympathetic terminals themselves (Munglani et al., 1996). In addition, the release of other trophic and/or vasoactive factors from sympathetic terminals, such as PACAP, VIP, GAL and SP, may modify PAF activity as well as vascular and immune function, thereby indirectly altering both the electrical activity and the regenerative capacity of injured PAFs (Brandenburg et al., 1997; Munglani et al., 1996). In addition, the release of NO and ATP from sympathetic terminals may directly (and indirectly) modify the activity of damaged nerves in a manner similar to that described for inflamed tissue (Sections 7.4.4 and 7.5.2.5).

The above discussion focuses on events integrated in the peripheral terminals of sympathetic fibres. However, the activity of sympathetic neurones may be modulated *centrally* at the level of preganglionic, sympathetic neurones in the intermediolateral cell column of the spinal cord. Therein, several receptorial mechanisms may potentially modify nociception via an enhancement of sympathetic outflow, including facilitatory mGlu, α_1 -AR and 5-HT_{2A} receptors activated by descending pathways (Hao *et al.*, 1996b; Helke *et al.*, 1997; McCall and Clement, 1994; Millan, 1997; Nicholas *et al.*, 1996; Spanswick *et al.*, 1995; Yaksh *et al.*, 1995).

It should, finally, be recalled that increases in systemic, arterial pressure provoked by i.v. administration of NAD can *decrease* nociception via activation of sinoaortic baro-receptors and vagal afferents, subsequently activating brainstem mechanisms engaging descending inhibitory pathways to the DH (Lovick, 1993; Millan, 1997; Thurston and Helton, 1996).

8.2.4.4. Sympathectomy-Induced pain

The above discussion provides evidence that sympathetic mechanisms contribute to neuropathic pain by facilitating the abnormal pattern of input from damaged PAFs onto sensitized neurones in the DH. It should, nevertheless, be mentioned that neuropathic, painful states are not invariably sympatheticdependent. Indeed, a generalized role of sympathetic mechanisms in their induction has been questioned by some authorities and, at least clinically, 'sympathetic' and 'non-sympathetic' painful states can be differentiated (Blumberg and Jänig, 1994; Campbell et al., 1994; Kingery, 1997; Schott, 1995; Verdugo and Ochoa, 1994; Verdugo et al., 1994). Further, surgical sympathectomy can *itself* trigger a painful syndrome in some patients. Several mechanisms may be involved:

- 1. As described in Section 8.2.4.2, a decrease in sympathetic tone may provoke a compensatory increase in the density and/or responsiveness of α_2 -ARs on PAFs and blood vessels, that is, a perturbation of sensory and vascular function.
- Sympathectomy may interrupt visceral nocisponsive afferents in sympathetic trunks and inadvertently damage other types of adjacent sensory fibre. Conceivably, then, processes underlying nerve damage-related pain may actually be *triggered* by sympathectomy.
- 3. Sympathectomy may lead to abnormal growth patterns and alterations in the phenotype of PAFs, including an induction of SP, CGRP and intrinsic trophic factors: these changes possibly involve the induction of NGF in non-neuronal cells (Aley *et al.*, 1996; Baron and Maier, 1996; Bolden *et al.*, 1997; Cowen and Gavazzi, 1998; Kramis *et al.*, 1996; Schott, 1995) (Sections 8.2.5. and 11.3).
- 4. Damage to sympathetic nerves results in marked alterations in their *own* phenotype, including increases in the expression of neuropeptides such as VIP and PACAP: such changes are mediated by IL-1β, LIF, NGF and several other factors.

Although this response to injury may play a trophic role in assisting sympathetic regeneration, neuropeptides, and other mediators may exert actions at nocisponsive PAFs leading to phenotype changes and functional effects provoking nociception (Cowen and Gavazzi 1998; Jonakait, 1993; Moller *et al.*, 1997a,b; Rao *et al.*, 1993) [Section 7.6]. Thus, paradoxically, surgical sympathectomy may *itself* elicit several pronociceptive mechanisms resembling those triggered by PAF injury. Such effects of sympathectomy require additional clarification.

8.2.5. Altered Phenotype of Damaged PAFs: Functional Consequences

The level of many neuropeptides (and certain neuropeptide receptors) changes in small and large calibre primary afferent neurones following their injury. Species differences, time-dependent patterns of changes and differences between various subpopulations of small and large calibre fibres should be emphasized in this regard (Goff et al., 1998; Hökfelt et al., 1994; Rydh-Rinder et al., 1996; Xu et al., 1997d; Zhang et al., 1993a,b). The importance of an increase in tissue levels of NGF in determining alterations in C fibre phenotype under conditions of inflammation was described in Section 7.6.1. Contrariwise, a decrease in the provision of NGF to the DRG of damaged C fibres, due to an-at least transient-loss of TRK A receptors on their peripheral terminals and a reduction in axonal flow, results in a pattern of phenotype changes broadly opposite to inflammation (Section 8.2.6) (McMahon and Bennett, 1997). Such phenotype changes in injured PAFs may have major, functional consequences at their central (nociception) and peripheral (nociception and trophic effects) terminals. In this regard, certain changes in PAF phenotype occurring upon injury have deleterious consequences, while others may be favourable. Indeed, some authors have interpreted the overall pattern of alterations in the phenotype of axotomized neurones as a globally adaptive response to injury inasmuch as such changes contribute to their recovery by promoting protective and regenerative mechanisms (Fu and Gordon, 1997; Hökfelt *et al.*, 1994; Lieberman, 1971; Tanaka *et al.*, 1997a) (but see Section 2.2). Further, certain—though *not* all—changes in PAF phenotype may palliate the sensory (hyperalgesia and allodynia) consequences of injury as concerns transmission in the DH. That is, certain excitatory (pronociceptive) and inhibitory (antinociceptive) neuropeptides are down- and up-regulated respectively in damaged PAFs (Hökfelt *et al.*, 1994).

An interesting example is provided by GAL, the trophic actions of which may encourage nerve regeneration in the periphery (Kask et al., 1997). Although GAL generally enhances nociceptive reflexes in intact animals, it exerts predominantly inhibitory, antinociceptive actions in the DH following PAF injury and it is up-regulated in both small and (to a lesser extent) large PAFs following their damage (Luo and Wiesenfeld-Hallin, 1995; Ma and Bisby, 1997; Villar et al., 1989; Wiesenfeld-Hallin et al., 1989 and 1992). The release of GAL in superficial DH is, correspondingly, elevated ipsilateral to a peripheral mononeuropathy (Colvin et al., 1997). Interestingly, the levels of both GAL₁ and GAL₂ receptors in damaged neurones are decreased (Xu et al., 1996b, 1997d). This reduction in their density may limit the excitatory, pronociceptive actions of GAL on PAFs (Shi et al., 1997). Thus, the antinociceptive properties of GAL at intrinsic DH neurones may be privileged, actions probably mediated by GAL₁ receptors which are negatively coupled to AC and which enhance and inhibit Ca2+ and K-currents, respectively (Kask et al., 1997; Xu et al., 1997d). Nevertheless, the receptorial mechanisms underlying the complex role of GAL at PAFs and intrinsic DH neurones requires further clarification (Section 3.2.6). The levels of a GAL-related neuropeptide, 'GMAP', which may likewise play a role in the modulation of nociception, are augmented in parallel to those of GAL upon PAF injury (Xu et al., 1995a).

The level of mRNA encoding PACAP is similarly increased in small DRG neurones containing GAL—as well as a subpopulation of large neurones containing NPY (Zhang *et al.*, 1996b) (see below). Like GAL, PACAP may express antinociceptive actions in the DH (Yamamoto and Tatsuno, 1995; Zhang *et al.*, 1993c) and, peripherally, it elicits extravasation, promotes neuronal survival and differentiation and is implicated in repair processes post-injury (Cardell *et al.*, 1997; Deutsch and Sun, 1992; Tanaka *et al.*, 1997a).

Less favourable than the above changes may, however, be the induction of CCK in as many as 30% of DRG neurones, and of CCK_B receptors in both small and large calibre PAFs (Hökfelt *et al.*, 1994; Xu *et al.*, 1993b, 1994, 1997d). CCK interferes with the central antinociceptive actions of opioids, probably via the induction of PLC and an increase in $[Ca^{2+}]_i$ levels. There is evidence from experimental models in rodents that the activation at CCK_B (CCK_A) receptors contributes to the relative inefficacy of opioids in clinically managing neuropathic

68

pain, and blockade of CCKA receptors increases nociception in man (Arnér and Meyerson, 1993; Benedetti, 1997; Nichols et al., 1995; Portenoy et al., 1990; Stanfa et al., 1994; Xu et al., 1993b, 1994; Wiesenfeld-Hallin et al., 1990; Zhang et al., 1993a, 1998b). In line with these observations, CCK_B antagonists enhance opioidergic antinociception in rodent models of neuropathic pain (Idänpään-Heikkilä et al., 1997; Nichols et al., 1995, 1996). Their actions reflect the blockade of upregulated pools of CCK not only in PAFs, but also in intrinsic DH neurones and, possibly, descending fibres. In parallel to the appearance of CCK, levels of somatostatin are decreased in injured, small calibre fibres (Hökfelt et al., 1994). This reduction may also be disadvantageous inasmuch as somatostatin exerts antinociceptive actions in the DH and periphery (Heppelmann and Matthias, 1997; Mollenholt et al., 1990).

Interestingly, despite observations of a *decrease* of SP and CGRP levels in damaged C fibres, nociceptive transmission may be maintained (Hökfelt et al., 1994; Mulder et al., 1997). In addition to peripheral mechanisms discussed in Sections 8.2.1, 8.2.2, 8.2.3 and 8.2.4, this finding may reflect the actions of residual stores of SP and CGRP at up-regulated NK₁ and CGRP receptors on sensitized DH neurones (Abbadie et al., 1996; Carlton and Coggeshall, 1997; Goff et al., 1998) [note, also, that recent studies of the evolution of changes in neuropeptide levels postinjury suggest that SP levels may not be reduced at all time periods post-injury (Goff et al., 1998)]. A further intriguing explanation is that VIP assumes a major, excitatory role in nociceptive transmission at neurones in the DH following its induction by PAF injury (Hökfelt et al., 1994; Shehab and Atkinson, 1986a,b). Indeed, VIP antagonists attenuate the induction of central hyperexcitability from injured (but not intact) PAFs (Wiesenfeld-Hallin et al., 1992). This central, 'deleterious' effect of VIP may be secondary to its beneficial, vasodilatory, trophic and glyconogenic actions in the periphery which encourage nerve survival and regeneration (Cardell et al., 1997; Colbert et al., 1994; Hökfelt et al., 1994; Tanaka et al., 1997a; White and Mansfield, 1996).

A further neuropeptide, NPY, mimics the trophic actions of VIP on blood vessels and nerves and its gene expression can be enhanced by VIP (Kashiba et al., 1997; Shigeri and Fujimoto, 1993; White and Mansfield, 1996; Youn et al., 1997). However, it remains to be elucidated whether VIP plays a role in the modest increase in NPY levels shown by injured C fibres. MRNA encoding NPY shows, in fact, a more marked increase in large calibre fibres following peripheral nerve injury (Noguchi et al., 1993; Wakisaka et al., 1991). Irrespective of its origins, PAF injury markedly enhances the spontaneous release of NPY in the DH (Mark et al., 1998). Further, levels of Y_1 and Y_2 receptors on *large* (but not small) calibre neurones are increased following PAF damage (Mantyh et al., 1994; Zhang et al., 1997e). It has been suggested that NPY exerts antinociceptive actions in the DH, possibly via Y₂ receptors negatively coupled to AC and localized on intrinsic DH neurones and PAF terminals (Munglani et al., 1996). Thus, an inhibitory action

of in increased release of NPY at up-regulated Y2 receptors on central A β fibre terminals might, theoretically, moderate mechanical allodynia (Zhang et al., 1994c, 1997e). In addition, NPY modulates nociception by multiple actions in the periphery, certain of which are expressed in interaction with NAD released (like NPY) from sympathetic neurones. In fact, the prevailing peripheral action of NPY may be an *increase* in nociception via a Y₂-receptor mediated release of PGs from sympathetic terminals (Munglani et al., 1996). Evidently, the relationship between multiple receptors for NPY and its spectrum of peripheral and central actions is still poorly understood. However, this issue would be of interest to explore further within the framework of the changes in NPY levels provoked not only in PAFs, but also in intrinsic DH neurones, by peripheral nerve injury (Ji et al., 1994; Munglani et al., 1996) (Sections 10.4 and 10.6).

Neurotensin generally exerts antinociceptive actions at intrinsic DH neurones and, possibly, PAF terminals. Inhibitory 'neurotensin' receptors (negatively coupled to AC) may be involved in these effects. However, neurotensin receptors in the DH are *down*-regulated by PAF injury revealing additional, *excitatory* actions of neurotensin on C (and $A\beta$) fibres, effects which are presumably mediated via a further, unknown neurotensin receptor type, which is possibly coupled to PLC (Carlton and Coggeshall, 1997; Xu *et al.*, 1997d).

Simultaneous to the loss of SP in fine calibre C fibres, SP actually *appears in* axotomized, *large* calibre A β fibres upon their injury (Miki *et al.*, 1998; Neumann *et al.*, 1996; Noguchi *et al.*, 1994). The consequences of this induction of SP synthesis in A β fibres is compounded by:

- 1. their release of EAAs;
- concurrent, structural changes whereby they now sprout into superficial laminae of the DH to make inappropriate contacts with nocisponsive neurones; and
- 3. an upregulation of NK₁ receptors in the DH (Abbadie *et al.*, 1996; Woolf and Doubell, 1994) (Section 11.2).

Further, $A\beta$ fibres display an increase in levels of VIP and NO following injury (Hökfelt et al., 1994). This induction of VIP and NO may, thus, further contribute to abnormal, pro-nociceptive actions of migrated $A\beta$ fibres at NS neurones in the superficial DH laminae (Section 11.2). Large calibre $A\beta$ fibres projecting to the gracile nucleus of the PSDC pathway (Section 6.1) may play a role in the mediation of dysthesias and mechanical allodynia following PAF injury. Thus, it is of interest that an increase in CGRP levels was detected in large, myelinated PAFs projecting to the gracile nucleus following peripheral axotomy. This increase may, at least partially, reflect its induction in A β fibres (Miki *et al.*, 1998). Further, this study showed that the increase in mRNA encoding CGRP was restricted to the αvs β -isoform suggesting that they can be differentially regulated.

To summarize, damage to PAFs elicits a complex pattern of phenotype changes, of which certain are involved in the induction of neuropathic painful states, while others may contribute to their attenuation.

8.2.6. The Significance of a Reduction in DRG Access to NGF and Other Trophic Factors in Modifying the Phenotype of Damaged PAFs

8.2.6.1. Phenotypic consequences of a loss of DRG access to NGF for C fibres

An-at least temporary-loss of DRG access to NGF likely plays a major role in determining altered patterns of gene expression in injured, fine calibre PAFs (Woolf, 1996). Indeed, in addition to a generalized reduction in axonal transport, both diabetic neuropathies and PAF lesions are characterized by decrease in the DRG density of mRNA encoding TRK A and p75^{NTR} receptors: this further compromises delivery of NGF to the DRGalthough a compensatory increase in mRNA encoding TRK A receptors may eventually occur (Delcroix et al., 1997; Heumann et al., 1987a,b; Maeda et al., 1996; McMahon and Bennett, 1997; Vergé et al., 1992a, 1995a; Zhou et al., 1996b). Certain of the phenotype changes displayed by C fibres can be pre-empted by administration of NGF (Delcroix et al., 1997; Funakoshi et al., 1993; Fernyhough et al., 1994; Gold et al., 1991; McMahon and Bennett, 1997; Ren et al., 1995; Vergé et al., 1992a, 1995a,b) and the accompanying hyperalgesia/allodynia may be reduced in parallel (Dray and Urban, 1996; Herzberg et al., 1997; Thomas et al., 1993). The argument that a reduction in the availability of NGF to the DRG of injured PAFs plays a role in mediating phenotype changes in injured C fibres is well illustrated by GAL, SP and CGRP.

Thus, consistent with observations that NGF exerts a suppressive influence upon the gene expression of GAL, the levels of mRNA encoding GAL increase in C fibres following injury (Section 8.2.5). This increase can be reversed by intrathecal administration of NGF which, via intact TRK A receptors on the *central* terminals of damaged PAFs, regains access to the DRG (Vergé *et al.*, 1995a,b; Xu *et al.*, 1996b; Zhang *et al.*, 1995). High doses of NGF also decrease GAL levels in cultured DRG neurones (Kerekes *et al.*, 1997).

NGF normally enhances the expression of mRNA encoding SP, levels of which generally decrease upon nerve damage (Goff et al., 1998; Hökfelt et al., 1994; Maeda et al., 1996; Munglani et al., 1996; Vergé et al., 1995a,b). This action of NGF may be explained by the presence in the promoter region of the preprotachykinin gene (which encodes SP) of a NGF-responsive element controlling SP transcription (Heumann, 1994; Watson et al., 1995). Thus, whereas an accumulation of NGF contributes to an increase of SP synthesis in C fibres in inflammatory pain (Donnerer et al., 1992; Kuraischi et al., 1989; Woolf et al., 1994), a reduction in the DRG access of damaged nerves to NGF leads to a decrease in SP production in neuropathic pain (Cameron et al., 1997; Hökfelt et al., 1994; Marchand et al., 1994; Nahin et al., 1994; Wong and Oblinger, 1991). Indeed, the application of NGF to DRG and the administration of anti-NGF

antibodies, respectively, up-regulates (cf inflammation) and down-regulates (cf nerve injury) SP synthesis (Donnerer et al., 1992; Lindsay and Harmar, 1989; Woolf et al., 1994). Further, NGF treatment corrects the reduction in DRG levels of mRNA encoding SP and CGRP seen in diabetic neuropathies (Diemel et al., 1994). As regards this influence of NGF upon SP expression, one remarkable indication of PAF plasticity and of the role of NGF is revealed by the surgical re-routing of muscle nerves to skin-the latter tissue contains far higher level of NGF. These reoriented muscle nerves display a marked increase in their content of SP-and develop an intensification of synaptic contacts in the spinal cord. (Lewin and McMahon, 1991; McMahon and Gibson, 1987). Such findings indicate that, if damaged nerves are allowed to regenerate to their normal peripheral targets, certain pathological changes in their phenotype and other changes may, at least partially, be normalized (Lisney, 1983) (Section 11).

Interestingly, despite *opposite* changes in SP synthesis in the DRG, NK_1 receptor levels *increase* in superficial laminae of the DH following both PAF injury and (generally) inflammation. The underlying mechanisms are unclear, but SP may itself elicit this increase. Thus, in the former case, the increase may reflect the initial and transient intense discharge of PAFs which follows their damage (Abbadie *et al.*, 1996; Coggeshall and Carlton, 1997; Goff *et al.*, 1998). In any case, increases in DH levels of NK₁ receptors are temporally correlated with the timecourse of those changes in nociception which accompany both PAF injury and peripheral inflammation (Goff *et al.*, 1998).

The gene encoding CGRP similarly possesses a promoter element for NGF. Thus, it is likely that a reduction in the DRG provision of NGF is involved in the decrease in CGRP levels provoked by injury to C fibres, although changes in CGRP levels following peripheral nerve injury are more variable than those of SP (Cameron *et al.*, 1997; Watson *et al.*, 1995).

In analogy to GAL, administration of NGF can reduce the number of small calibre PAF neurones displaying an upregulation in mRNA encoding CCK and NPY following their injury (Vergé *et al.*, 1995b).

8.2.6.2. A potential role of other trophic factors: focus on LIF

Upon nerve damage, VIP is co-expressed and coinduced with GAL in a subpopulation of C fibres, a change probably reflecting a reduction in access to NGF, the application of which reverses this increase (Kashiba *et al.*, 1992; Vergé *et al.*, 1995b).

However, changes in fine calibre PAF phenotype do *not* all reflect a reduction in access to NGF. Indeed, a further pool of VIP is present in a subpopulation of somatostatin-containing C fibres *lacking* TRK A receptors. The increase and decrease, respectively, in VIP and somatostatin levels in these neurones following injury is, thus, probably NGFindependent (Kashiba *et al.*, 1996; Mulderry and Lindsay, 1990; Noguchi et al., 1993; Vergé et al., 1995b).

One interesting candidate for mediating changes in VIP expression in such somatostatin-containing C fibres lacking TRK A receptors is the cytokine, LIF, which encourages VIP synthesis in sympathetic neurones (Jonakait, 1993; Rao et al., 1993). Thus, the expression of LIF, which is normally very low, is markedly induced in Schwann cells upon peripheral nerve injury via factors derived both from injured PAFs and from macrophages (Banner and Patterson, 1994; Livesey et al., 1997). Further, although LIF does not induce VIP expression in small diameter, TRK A-positive PAFs (Matsuoka et al., 1997) (vide supra), it may contribute-together with the loss of NGF-to the induction of GAL in these neurones (Corness et al., 1996; Curtis et al., 1993; Southall et al., 1996). In line with a modulatory influence of LIF on gene expression in PAFs, it has recently been shown that LIF is retrogradely transported to cell bodies of small calibre, nocisponsive C fibres containing CGRP (Thompson et al., 1997). Furthermore, as described in Section 11.3.4, LIF (derived from Schwann cells or sympathetic terminals) may induce the expression of specific proteins in damaged PAFs, the secretion of which reciprocally encourages further Schwann cell proliferation (Jonakait, 1993; Livesey et al., 1997). Additional support for a modulatory role of LIF is also available. Thus, in analogy to NGF, the peripheral administration of LIF induces mechanical allodynia (Thompson et al., 1996, 1997). Further, the recent discovery that mRNA encoding LIF is present in superficial DH laminae I/II, and that it is upregulated by inflammation and PAF injury, raises the possibility of central actions of LIF in the modulation of nociception (Heppenstall et al., 1997).

These findings underscore the fact that NGF is *not* the only element responsible for changes in the phenotype of damaged C fibres (Aldskogius *et al.*, 1992; Kerekes *et al.*, 1997; Raivich *et al.*, 1991). A reduction of axonal flow upon nerve injury limits access to the DRG of multiple trophic factors and an interruption of axonal flow *per se*, in the *absence* of damage to PAFs, has been shown to markedly modify their phenotype (Kashiba *et al.*, 1992; Knyihar-Csillik *et al.*, 1991) (Section 8.2.2.5). It would be of interest to clarify the role of GDNF in the response of NGF-independent, fine calibre PAFs to injury (Section 3.2).

8.2.6.3. *Gene transcription factors: immediate early genes*

The levels of transcription factors, such as NFkB, and of immediate early genes (IEGs), such as cjun and jun-d, are altered in injured PAFs. A putative role of these factors in the modulation of gene expression by NGF, LIF and other trophic molecules is an interesting issue (Doyle and Hunt, 1997; Herdegen *et al.*, 1991, 1997; Jenkins *et al.*, 1993; O'Neill and Kaltschmidt, 1997; Vergé *et al.*, 1995b). Axonal damage induces the expression of c-jun, enhances its phosphorylation by specific kinases and encourages its functional dimerization with jun-d, c-

fos and other transcription factors. These conformational changes enable c-jun to modify the transcription of several effector genes which act in synergy with other regulatory elements to control the precise pattern of phenotype changes initiated by nerve injury (Fiallos-Estrada et al., 1993; Herdegen et al., 1997; Hughes and Dragunow, 1995). One interesting illustration of the role of ciun is provided by the increase in GAL synthesis detected in damaged C-fibres possessing TRK A receptors (Section 8.2.5). C-jun stimulates the transcription of GAL and a reduction in the suppressive influence of NGF upon c-jun expression in damaged nerves may underlie a secondary increase in GAL gene expression (Gold et al., 1993; Herdegen et al., 1991, 1997; Vergé et al., 1995a,b). Further, inasmuch as NGF activates the transcription factor, NF-kB, in the DRG, a loss of access to NGF may explain the reduction in NF-kB levels observed in damaged PAFs (Dechant and Barde, 1997; Doyle and Hunt, 1997; O'Neill and Kaltschmidt, 1997). The role of other elements, such as BDNF and the cytokines, TNF α and LIF, in modulating the expression of c-jun, NF-kB and other transcription factors in injured PAFs, would be of interest to establish (Cheema et al., 1994; Doyle and Hunt, 1997). However, damage to PAFs proximal to the DRG can also induce c-jun expression, even though DRG cells do not degenerate and trophic factors are still available to their PAF terminals. This observation, and the induction of c-jun in many cell types upon PAF injury, suggest that the induction of cjun and several other regulators of gene expression may be a generalized response to their injury (Kenney and Kocsis, 1997).

8.2.6.4. Factors involved in phenotype changes in large calibre $A\beta$ fibres

As concerns phenotype changes in damaged, *large* $A\beta$ fibres, since these lack TRK A receptors, the above-mentioned increases in levels of SP, NPY and, to a less marked degree, VIP and GAL, are unlikely to directly involve NGF. Rather, they may be due to alterations in the provision of other growth factors for large neurones such as BDNF, NT 3, LIF or fibroblast growth factor (Fu and Gordon, 1997; Lewin and Barde, 1996; Neumann et al., 1996; Ma and Woolf, 1996b; Sterne et al., 1997). Indeed, direct evidence that a loss of access to fibroblast growth factor and BDNF may be involved in the upregulation of GAL in injured, large PAFs has been provided by Kerekes et al. (1997). Further, the availability of LIF from Schwann cells may also contribute to the increase in GAL synthesis therein (Southall et al., 1996). These mechanisms require considerable further clarification.

8.2.6.5. Functional considerations

Notwithstanding the wealth of information discussed above concerning the relationship between NGF and the phenotype of injured C fibres, the precise functional and clinical significance of these observations remain unclear. Some authorities have suggested, in line with the above-described changes in phenotype triggered by a loss of NGF to DRG loss, that the provision of exogenous NGF to injured PAFs might be beneficial in restoring a normal phenotype (Dray and Urban, 1996; McMahon and Priestley, 1995). However, *pronociceptive* effects of NGF at damaged (and intact) PAFs may be expressed independently of changes in phenotype via mechanisms discussed in Section 7. Further, it has been reported that the acute neutralization of NGF (by specific antibodies) can *reduce* neuropathic pain due to PAF injury (Dray and Urban, 1996; Herzberg *et al.*, 1997).

The acute and long-term functional effects of NGF, BDNF, LIF and other trophic factors under conditions of PAF injury require considerable further clarification before the therapeutic exploitation of such strategies can realistically be envisaged.

8.2.7. Phenotype Changes in Damaged PAFs: Neuromodulators other than Neuropeptides

Damage to PAFs provokes changes in the levels of many substances other than neuropeptides. For example, the cytoskeletal protein, tubulin, which plays a role in growth cone support and other regenerative mechanisms of damaged PAF terminals (Fournier and McKerracher, 1997; Wong and Oblinger, 1990, 1991) (Section 11.3).

Concerning nociception, the spinal release of EAAs may be enhanced by nerve injury, a change which enhances nociceptive transmission in the DH (Al-Ghoul *et al.*, 1993; Cui *et al.*, 1997b; Dickenson, 1997—but see Xu *et al.*, 1995b).

An induction of NO synthase has also been seen in the DRG following PAF injury-particularly in small calibre, CGRP-containing neurones-whereas, in a model of diabetic neuropathy, NO levels were not changed (Fiallos-Estrada et al., 1993; Sasaki et al., 1998; Vergé et al., 1992b; Zhang et al., 1993b), NO may play a role in modulating (primarily in facilitating) nociception via both central (DH) and peripheral actions following PAF injury (Sections 3.2.8.3 and 7.7). Interestingly, a loss of access to NGF has been speculated to underlie this increase in NO synthase-in analogy to these changes in neuropeptide levels in C fibres detailed in Sections 8.2.5 and 8.2.6. (Thippeswamy and Morris, 1997). In addition to alterations in nociceptive transmission via actions at PAFs, the peripheral (trophic) actions of NO should also be mentioned: for example, its ability to increase vascular flow (Section 7.9.2) (Hökfelt et al., 1994; Vergé et al., 1992b; Wiesenfeld-Hallin et al., 1993; Zhang et al., 1993b).

As discussed in Section 3.2.8.2, adenosine may play an important role in modulating nociception in the DH. Notably, the engagement of spinal A_1 receptors suppresses both the allodynia and the activation of DH neurones which is associated with PAF injury (Section 3.2.2). Whether the release of adenosine from central or peripheral terminals of damaged PAFs is modified would, thus, be of interest to determine. In line with such a possibility, nerve damage is clinically associated with a *decrease* in CSF levels of adenosine, the administration of which alleviates the accompanying pain. These observations raise the possibility that a deficiency in the central release of adenosine from damaged PAFs may be involved in certain neuropathic, painful states (Cui *et al.*, 1997a; Guieu *et al.*, 1996; Sollevi *et al.*, 1993).

Finally, changes in the density of multiple opioid receptors elicited by PAF injury are of potential significance, in particular a decrease in the levels of μ opioid receptors on PAF terminals and intrinsic neurones in the DH. This reduction in μ -opioid receptor availability may dampen endogenous, antinociceptive mechanisms and contribute to the poor responsiveness of opioids in treating neuropathic pain states (Goff *et al.*, 1998; De Groot *et al.*, 1997; Zhang *et al.*, 1998b).

A further exploration, thus, of alterations in the synthesis and release of substances other than neuropeptides in damaged PAFs would be of interest. In addition, the role(s) of NGF, BDNF, LIF and other trophic factors in mediating such changes would profit from examination.

9. NOCICEPTIVE AND NEUROPATHIC PAIN: CENTRAL MECHANISMS AND ALLODYNIC STATES

9.1. Evidence for Central Mechanisms of Sensitization

9.1.1. The Significance of Central Events

The previous sections illustrate the complexity of peripheral events involved in the generation and maintenance of painful states provoked by tissue inflammation or PAF injury. An enhanced, C fibremediated, afferent barrage from sensitized nociceptors (including the recruitment of silent nociceptors) is involved in the induction-and maintenance-of inflammatory pain states. Moreover, following an initial, intense discharge of injured PAFs, the generation of abnormal (ectopic) impulses is involved in the initiation-and maintenance-of neuropathic, painful states. Correspondingly, interference with peripheral inflammatory events by, for example, inhibition of COX 2 (and 1) synthase, or antagonism of B₂ receptors, may attenuate nociceptive pain (Appleton, 1997; Dray and Urban, 1996). On the other hand, sympathetic blockade, or systemic administration of local anesthetics, may alleviate neuropathic pain (Arnér et al., 1990; Koltzenburg et al., 1994). Indeed, peripheral events certainly contribute to two principal features of tissue damage. That is, spontaneous pain and an exaggerated (hyperalgesic) response to heat and other modes of noxious stimulation (primary hyperalgesia). However, adaptive events in the CNS, which amplify the consequences of an abnormal and/or excessive PAF input, also play an important role in determining changes in pain sensation. Indeed, an additional key element of clinical pain, mechanical allodynia, which is displayed by uninjured tissue surrounding the area of inflammation (secondary hyperalgesia), and which accompanies injury to PAFs, cannot be satisfactorily explained without taking into account central mechanisms (Cervero and Laird, 1996b; Simone et al., 1991; Woolf et al., 1994). Likewise, the phenomenon of spatially-referred ('remote') pain cannot be convincingly ascribed to peripheral processes. The fol-
lowing paragraphs consider, thus, evidence that processes integrated in the CNS contribute to the induction of chronic, painful states.

Implicit to many discussions of central sensitization is the assumption that an increase in the excitability of nociceptive pathways in the periphery, DH and higher structures is deleterious. However, certain events underlying central sensitization are triggered very rapidly (over minutes) and a normalization of neuronal activity may occur once peripheral input is terminated. Such shorter-term, reversible states of sensitization, maintained by peripheral input and correlating temporally with, for example, acute, peripheral noxious stimulation, may not involve permanent phenotypic/morphological changes. They likely play a role in encouraging protective behaviours and avoiding aggravation of tissue injury (Traub, 1997; Wiertelak et al., 1994c) (Section 2, Table 2). Thus, transient processes of sensitization may be related to the adaptive role of subchronic pain. On the other hand, chronic, clinical pain may reflect long-term, possibly irreversible, processes of central sensitization which become at least partially independent of peripheral input, and which ultimately involve morphological/phenotype changes in pathways mediating and modulating nociceptive transmission (Table 2) (Section 10.2.1).

9.1.2. Mechanical Allodynia: Evidence for Mediation by $A\beta$ Fibres

As indicated above, mechanical allodynia is a prominent feature of secondary hyperalgesia and, in particular, of neuropathic, painful states due to PAF or CNS injury. There is compelling evidence that the activation of mechanosensitive, low threshold $A\beta$ fibres, which normally mediate sensations of touch or vibration etc, but *not* pain, underlies the mechanical allodynia associated with, for example, light brushing of the skin. This is the major, clinical form of allodynia (Ma and Woolf, 1996b; Woolf and Doubell, 1994).

- 1. First, as described in Section 8.2.2, PAF injury is associated with abnormal firing patterns in both fine calibre C fibres *and* large diameter, $A\beta$ fibres.
- 2. Following their injury, $A\beta$ fibres become particularly responsive to sympathetic stimulation, and chemical or surgical sympatheteromy often relieves (though not invariably) clinical, mechanical allodynia (Ghostine *et al.*, 1984; Handwerker and Kobal, 1993; Raja *et al.*, 1991; Treede *et al.*, 1992b; Wallin *et al.*, 1976) (Section 8.2.4).
- 3. Clinical and experimental studies of secondary hyperalgesia indicate that, despite the marked increase in pain elicited by mechanical stimulation under conditions of chemical irritation (e.g. application of capsaicin to the skin), the transduction sensitivity of individual, nocisponsive, fine calibre *mechanoceptors* is *not* markedly modified (Gracely *et al.*, 1992; Handwerker and Kobal, 1993; Koltzenburg *et al.*, 1992, 1994; LaMotte *et al.*, 1991; Meyer *et al.*, 1994; Torebjörk *et al.*, 1992; Treede *et al.*, 1992b). High threshold, mechanosensitive, nocisponsive C (or A δ fibres) do *not*, thus, seem to be primarily involved. Further, the unchanged transduction

sensitivity of mechanosponsive PAFs implicates *central* events in the altered perception of $A\beta$ fibre input and the induction of allodynia.

- 4. The delay between cutaneous, mechanical stimulation and the onset of allodynia corresponds to the rapid conduction velocity of myelinated, $A\beta$ fibres (Handwerker and Kobal, 1993; Lindblom, 1985; Treede *et al.*, 1992b).
- 5. In line with this argument, under condition of nerve injury (or capsaicin treatment), selective electrical activation of $A\beta$ fibres elicits pain and dysthesias, while selective blockade of $A\beta$ fibres suppresses brush-evoked allodynia (Koltzenburg, 1995; LaMotte *et al.*, 1991; Simone *et al.*, 1989b; Torebjörk *et al.*, 1992; Wallin *et al.*, 1976).
- 6. A β fibres exert their actions, at least partially, via the activation of DH-localized AMPA receptors, the pharmacological blockade of which reduces tactile allodynia in rats. Although it is difficult to completely exclude a role of AMPA sites coupled to C fibres, AMPA receptors activated by A β fibres are likely involved in this action (Millan, 1995; Sang *et al.*, 1997; Sherman and Loomis, 1996; Xu *et al.*, 1993a; Zhang *et al.*, 1994a) (Section 3.2.7).
- 7. Following peripheral nerve injury, the density and Ca^{2+} -permeability of AMPA receptors is enhanced in the DH with a time-course paralleling mechanical supersensitivity and allodynia (Hargett *et al.*, 1997; Harris *et al.*, 1996) (Section 10.3.2.1).
- 8. Non-noxious touch and electrical stimulation of $A\beta$ fibres modifies gene expression in DH neurones *only* under conditions of inflammation (Molander *et al.*, 1992; Ma and Woolf, 1996a).

The above observations provide convincing evidence for a role for $A\beta$ fibres in mediating (at least some forms of) mechanical allodynia. It is conceivable that damage to PAFs allows for the transactivation of C fibres by $A\beta$ fibres via processes of interneuronal communication in the DRG and neuroma (Section 8.2.3). This could lead to the perception of an innocuous stimulation as noxious. However, this mechanism is of little relevance to the allodynia (secondary hyperalgesia) which manifests itself independently of nerve injury. Further, although it has been proposed that $A\beta$ fibres activate sensitized WDR neurones via the indirect recruitment of C fibre terminals in the DH (Cervero and Laird, 1996b) (Section 9.2.3), this mechanism may be of limited pertinence to the allodynia elicited by PAF injury. In any case, the indirect recruitment of C fibres would *not* question the crucial role of $A\beta$ fibres—and mechanisms of central sensitization—in the mediation of mechanical allodynia.

9.1.3. Secondary Hyperalgesia: Peripheral Maintenance of Central Sensitization

The secondary hyperalgesia encountered in the zone circumferential to the area of injured tissue (primary hyperalgesia) is characterized by an enhanced sensitivity to mechanical rather than heat stimuli (Ali *et al.*, 1996; Treede *et al.*, 1992a).

It might be argued that peripheral events underlie this secondary, mechanical hyperalgesia. Indeed, many years ago, Lewis (1942) hypothesized than a peripheral mechanism was involved whereby APs triggered in PAFs by a noxious stimulus travelled both orthodromically to the DH and also antidromically, via a network of branches or current spread amongst coupled fibres, to invade nerve terminals: these terminals would then release sensitizing substances, such as CGPP and SP. In fact, the interconnected network of fibres imagined by Lewis has not been demonstrated to exist. Further, although interneuronal communication may occur between injured nerves (Section 8.2.3), such processes are uncommon for intact, peripheral fibres (Meyer et al., 1985, 1994) (Section 3.1.1). The antidromic activation of PAFs which occurs within regions of tissue damage/ primary hyperalgesia is involved in processes of NI and the axonal 'flare' response of the skin to injury (Dray and Urban, 1996; Gamse and Saria, 1987; Lynn, 1990; Reeh and Kress, 1995) (Section 7.9). However, it is unclear whether the population of PAFs which mediates the flare response is identical to that which mediates an increase in nociception (Lynn et al., 1996). Further, the boundary of secondary hyperalgesia generally extends beyond the flare zone, and there is no spatial correlation between the flare response and the mechanical hyperalgesia elicited by application of capsaicin to human skin. In addition, antidromic nerve stimulation in the rat does not elicit mechanical hypersensitivity or nociceptor sensitization (Reeh et al., 1986; Treede et al., 1992b). As mentioned above, moreover, the transduction sensitivity of small calibre, mechanosponsive fibres is not changed in secondary hyperagesia. Finally, histamine can produce a flare response with little or no pain and does not, in general, excite mechanosensitive PAFs (Ali et al., 1996; LaMotte et al., 1991). In conclusion, there is little evidence that antidromic activation of PAFs contributes to the secondary, mechanical allodynia of uninjured tissue.

The observations in Section 9.1.2 provide compelling evidence that stimulation of $A\beta$ fibres mediates the mechanical allodynia of secondary hyperalgesia. They also suggest that central mechanisms are required for its expression: such mechanisms are discussed in detail in Sections 10 and 12. Assuming that central processes underlie secondary hyperalgesia and altered responsiveness to $A\beta$ fibre stimulation, it is reasonable to ask whether they are phasically triggered by tissue injury and persist in the absence of additional peripheral input, or whether they require a continuous, peripheral drive for their maintenance.

Following early work by Hardy *et al.* (1950), LaMotte *et al.* (1991) showed that the hyperalgesia and allodynia to mechanical stimuli provoked in human subjects by application of capsaicin to the skin extends outside the region of injection. The hyperalgesia could be prevented by anaesthetic blockade of the peripheral nerve innervating the injected site *before* capsaicin administration whereas, *following* the induction of pain, cooling of the capsaicin-treated region was only partially effective in its relief (Dahl *et al.*, 1993; LaMotte *et al.*, 1991; Treede *et al.*, 1992b). Similar data have been found employing electrical stimulation (Torebjörk *et al.*,

1992). Thus, under some conditions, central changes and a corresponding induction of secondary hyperalgesia, may become at least partially independent of subsequent C (and A β) fibre input (Baron and Saguer, 1993). Nevertheless, under other conditions, a sustained, low level of PAF drive appears to maintain those persistent states of central sensitization which contribute to the cutaneous and visceral hyperalgesia/allodynia provoked by tissue damage (Andersen et al., 1995; Cervero, 1995a; Cervero and Laird, 1996b; Cervero et al., 1994; Dickenson, 1997; Gracely et al., 1992; Koltzenburg et al., 1994; LaMotte et al., 1991). For example, as mentioned in Sections 8.2.2 and 8.2.4, local anaesthesia (or sympathectomy) can rapidly alleviate some forms of secondary hyperalgesia and neuropathic pain due to PAF injury. In addition, these interventions can even prevent the re-appearance of pain upon intense A β fibre stimulation (Gracely *et al.*, 1992; LaMotte et al., 1991; Treede et al., 1992b).

In analogy to clinical investigations, experimental studies have shown that capsaicin enhances the responsiveness of WDR neurones to low intensity, mechanical stimuli and $A\beta$ fibre stimulation *outside* the region of tissue injury in rats. Further, their responsiveness to $A\delta$ fibre stimulation may also be enhanced (Simone et al., 1989a). Both in rats and in primates, heat injury of the skin leads to a facilitation in the response of DH (including STT) neurones to adjacent, non-noxious, tactile stimuli (Dougherty et al., 1992b; Kenshalo et al., 1982; Lin et al., 1997; McMahon and Wall, 1984). In parallel, a reorganization or expansion in the RFs of individual DH neurones may be seen-including those in the cuneate nucleus of the PSDC pathway (Laird and Cervero, 1989; Pettit and Schwark, 1996). Such changes, which are considered to reflect central sensitization, are most pronounced for WDR PNs, in particular, those in deep laminae. However, it is possible that WDR and NS classes of PNs and EXINs in superficial and deeper laminae become sensitized and responsive to low-threshold, $A\beta$ fibre input. Indeed, studies of intracellular markers of neuronal activity bear out the possibility that neurones throughout the DH can be sensitized by nociceptive input (Cervero, 1995b; Chapman and Besson, 1997; Simone et al., 1991; Treede and Magerl, 1995) (Section 10.4.2.5). In analogy to several, above-discussed clinical studies, there is evidence that central, sensitized states in the rat may be further facilitated by a persistent, peripheral input from injured tissue (Dickenson, 1997; Taylor et al., 1995).

To summarize, then, one component of central sensitization may be expressed independently of further PAF input, whereas a further, superimposed component (mediated by contrasting mechanisms) may be maintained by a persistent peripheral drive. Nevertheless, the quantitative relationship between the duration, intensity, temporal patterns and other characteristics of peripheral input to the DH and the maintenance of central, sensitized states requires further elucidation.

9.1.4. Referred Pain

9.1.4.1. Characteristics of referred pain

Injury to superficial and (more markedly) deep nerves, as well as noxious stimulation of cutaneous C fibres and (more effectively) nocisponsive muscle afferents, increases both ipsilateral and contralateral flexor reflexes. This increased excitability of contralateral reflexes is maintained even after interruption of input from the ipsilateral (affected) paw (Wall and Woolf, 1984; Woolf and McMahon, 1985; Woolf and Wall, 1986) and is independent of changes in the excitability of motoneurones and PAFs innervating the region of referred pain (Cook et al., 1986). These observations point to a role of central mechanisms in the mediation of potentiated reflexes from 'remote' tissue regions. Tissue damage may also elicit a hypersensitivity to innocuous, mechanical stimuli in remote regions (cf allodynia) and increase the RF size of the corresponding population of DH neurones (Cook et al., 1987; McMahon and Wall, 1984; Ness and Gebhart, 1990; Neugebauer and Schaible, 1990). In analogy, localized noxious stimulation of a single paw in rats can elicit a hyperalgesia of the contralateral paw which is prevented, but not reversed, by anaesthesia of the stimulated paw (Coderre and Melzack, 1991; Coderre et al., 1993: Wiertelak et al., 1994a). These data suggest that noxious stimulation has triggered persistent changes in the central processing of nociceptive input from other regions.

Many other examples of referred pain involving tenderness, allodynia and hyperalgesia of intact tissues are seen clinically. In general, referred pain develops slowly, like secondary hyperalgesia, and, clinically, it is triggered by deep somatic and visceral rather than superficial pain (Graven-Nielsen et al., 1997; Mohammadian et al., 1997; Procacci and Zoppi, 1983). Examples include the referred pain which is felt segmentally upon stimulation of the viscera, and the referral of muscle pain to other muscle groups, tendons or joints. The most familiar illustration is, however, the referred pain from the heart which is experienced in the inner aspect of the leftarm and shoulder, and which may signal cardiac infarction (Graven-Nielsen et al., 1997). This example conforms to a 'dermatomal' rule whereby pain is often referred to regions derived from the same embryonic segment, and which possess nerves entering the spinal cord at the same level-an observation consistent with a role of convergent DH neurones in the mediation of remote pain (Section 9.1.4.2). A classic illustration is provided by pain in the diaphragm which is often referred to the shoulder. PAFs supplying these tissues share a common embryonic origin and, in each case, enter the DH via the third and fourth cervical DRG to target a common population of neurones. Nevertheless, referred pain can exceed segmental boundaries and its referral may be idiosyncratic. For example, some subjects display referred heart pain exclusively in the abdominal region. The non-stereotyped nature of referred pain is also illustrated by the frequent referral of pain to sites of previous painful trauma: for example, a surgical scar in abdominal disease, or previously-operated teeth during dental surgery.

This 'learnt' and variable component of pain is consistent with a role of central (cerebral) processes in its mediation.

Although the issue is still somewhat controversial, referred pain in the absence of long-term changes in tissue pathology (Section 9.1.4.2), is generally prevented or interrupted by anaesthesia of the region exposed to noxious stimulation, rather than by anaesthesia of the region where pain is experienced (Vecchiet *et al.*, 1993). Further, intact tissue regions usually do *not* display alterations in the transduction sensitivity of their nociceptors. These observations, together with the above-mentioned findings, point to a role of *central* mechanisms in the mediation of referred pain. There are several, possible mechanistic explanations which should be regarded as complementary rather than mutually exclusive.

9.1.4.2. Potential mechanisms underlying referred pain

- 1. Referred pain has been attributed to the peripheral branching of PAFs. This hypothesis assumes that individual sensory nerves send collaterals both to deep tissues and to the skin. However, little anatomical support for this possibility is available inasmuch as dichotomizing fibres are rare and there is considerable inter-individual variability in their occurrence (Bahr *et al.*, 1981; Dawson *et al.*, 1992; Häbler *et al.*, 1988; Mense *et al.*, 1981; Takahashi *et al.*, 1993). Further, the peripheral branching of sensory afferents cannot satisfactorily account for either the temporal *delay* in the onset of referred as compared to local pain nor for its *differing* characteristics.
- 2. A minor population of PAFs projecting to the *contralateral* DH (Culberson *et al.*, 1979; Light and Perl, 1979; Sugimoto *et al.*, 1987; Sugiura *et al.*, 1986) might be involved in instances of segmentally-referred contralateral pain, but this mechanism is not relevant to referred pain from other regions.
- 3. In many cases of segmentally-referred pain, the underlying mechanisms may involve convergent (e.g. visceral/cutaneous) input onto a common population of sensitized DH neurones (Al-Chaer et al., 1996a,b; Berkley et al., 1993; Cervero, 1995a; Cook et al., 1987; McMahon, 1994; Mense, 1993; Ness and Gebhart, 1990; Schaible et al., 1987). Indeed, all DH neurones receiving input from the viscera and deep somatic regions are also targetted by PAFs innervating the skin. However, a role of convergent neurones cannot provide a complete explanation inasmuch as sensation may be *differentially* modified in the region of referred as compared to local pain (Graven-Nielsen et al., 1997). Further, referred pain may exceed segmental boundaries and it also occurs between muscle and deep tissues for which convergence of sensory input at a common population of DH neurones is rare (Hoheisel and Mense, 1990; Mense, 1993). Thus, at least under certain conditions, additional mechanisms may also intervene.
- 4. The engagement of (ipsilateral or contralateral) DH neurones corresponding to remote tissue

regions may occur via a supraspinal loop. That is, a strengthening and weakening of descending facilitation and inhibition, respectively (Section 10.8). The putative implication of supraspinal structures in mediating referred pain is reinforced by observations that the noxious stimulation of PAFs is associated with distinctive patterns of adaptive changes in neuronal properties in the thalamus and cortex: notably, the appearance of neurones with bilateral RFs (Apkarian, 1995b; Apkarian et al., 1995; Guilbaud and Benoist, 1995) (Section 12). Further, in models of unilateral, inflammatory and neuropathic pain, a hyperalgesia may develop in the opposite, untreated limb in parallel with the appearance of a corresponding subpopulation of sensitized neurones in the thalamus and cortex (Guilbaud and Benoist, 1995). One anatomical substrate permitting such changes would be the convergence of both ipsi- and contralateral cutaneous, muscle and visceral nociceptive input from multiple ascending pathways in specific, cerebral processing centres: for example, the VPL and other thalamic nuclei, the nucleus gracilis of the PSDC or the cortex itself (Al-Chaer et al., 1997a; Apkarian, 1995b; Apkarian et al., 1995; Brüggemann et al., 1997) (Section 6). Indeed, thalamic neurones have been identified which display increased responsivity to mechanical stimuli in RF regions remote from the locus of injury (Apkarian et al., 1995; Guilbaud et al., 1986). Further, interhemispheric communication of sensory information-including sensitized statescan occur at the level of the cortex and other supraspinal structures, possibly via the engagement of intracortical and reciprocal corticothalamocortical circuits (Sections 6.3.2 and 12). As concerns the role of individual, ascending pathways, Apkarian et al. (1995) have, intriguingly, suggested that 'true' and 'referred' visceral pain are predominantly mediated by the PSDC and STT, respectively. Within the context of referred pain, the present discussion focuses on remote increases in nociception. However, C fibre input from damaged tissue can heterotropically activate descending inhibitory mechanisms. This may account for certain instances of modality-specific decreases in nociception (Bouhassira et al., 1995; Graven-Nielsen et al., 1997; Le Bars, 1988; Romita and Henry, 1996). Although mechanisms of heterotropic inhibition do not necessarily require a supraspinal loop (McGaraughty and Henry, 1997; Romita and Henry, 1996), these observations are consistent with the notion that cerebral mechanisms may be involved in remote alterations in nociception.

5. Persistent, intense nociceptive input—or PAF injury—may lead to generalized and widespread increases in DH excitability via mechanisms involving sensitization, disinhibition and possibly, the emergence of 'generalized, reverberating' circuits (Coghill *et al.*, 1991) (Section 10.6). Such excited neuronal states, and corresponding alterations in RF properties, could be communicated intrasegmentally, intersegmentally and contralaterally by ININs and propriospinal neurones (Willis and Coggeshall, 1991). In addition, a supraspinal loop (vide supra) and the minor population of PAFs with collaterals running to the contralateral DH may also be involved. Following disinhibition, normally-subthreshold stimuli may elicit frank APs in DH neurones innervated by remote tissues (Cook et al., 1987; McMahon and Wall, 1984; Pettit and Schwark, 1996). In line with this concept of spreading sensitization, SP released from PAF terminals persists in the DH wherein it may be widely dispersed and act via volume transmission (Section 4.7). Moreover, a localized, noxious stimulus has been observed to elicit a widespread, 'non-topographic' upregulation of NK1 receptors in the DH (Abbadie et al., 1996; Goff et al., 1998). In addition, unilateral PAF stimulation or damage can elicit a variety of bilateral changes including alterations in:

- 1. nociceptive thresholds and the influence of drugs upon nociception;
- 2. sensitivity and RF characteristics of DH neurones; and
- levels of various transmitters in the DH (Ji et al., 1994; Leah et al., 1992; Leslie et al., 1995; Löfgren et al., 1997; Mao et al., 1992b, 1993a; Millan, 1993; Mohammadian et al., 1997; Presley et al., 1990; Sugimoto et al., 1990; Takaishi et al., 1996) (Section 10.6).

In this light, it is of note that central, DH-integrated mechanisms can modify the activity of sensory fibres via an influence upon their central terminals. This leads, in turn, to the antidromic invasion of their peripheral branches. Indeed, as described in Section 7.9, the efferent activation of polymodal C fibres provokes extravasation and other processes of NI. For example, there is evidence that peripheral inflammation and the accompanying nociception can be modified via DHlocalized, GABAA- and non-NMDA receptormediated mechanisms eliciting 'dorsal root potentials' (DRP) (or 'dorsal root reflexes') in fine calibre PAFs (Bagust et al., 1997; Cervero and Laird, 1996b; Sluka et al., 1993, 1994a,b, 1995; Thompson and Wall, 1996; Wall and Lidierth, 1997). The engagement of such mechanisms might underlie the contralateral vasodilation seen in response to unilateral sciatic ligation in rats, a procedure which is accompanied by bilateral changes in neuronal activity the DH (Kurvers et al., 1996; Mao et al., 1992c). Further, unilateral, arthritic joint inflammation may be accompanied by bilateral DRPs and hyperalgesia (Rees et al., 1996), while the eventual transfer of unilateral, arthritic joint inflammation of the paw to the contralateral side may well involve such DHintegrated events (Millan and Colpaert, 1991). In addition to the antidromic activation of peripheral PAF terminals via actions mediated in the DH, sympathetic mechanisms may also modulate the activity of PAF terminals, providing a sixth potential mechanism underlying referred changes in nociception. Changes in sympathetic outflow to regions of referred pain might be mediated either via:

Table 7. Summary of mechanisms potentially involved in the induction of referred (remote) pain (see Section 9.1.4)

- A. PAF organization
- 1. Peripheral branching of PAFs: collaterals to both deep (e.g. visceral) and cutaneous tissue
- 2. PAFs projecting to the contralateral as well as ipsilateral DH

- 1. Convergent PAF input from different tissues (e.g. muscle and skin) onto a common population of (sensitized) DH neurones
- 2. Development of heterosegmental, excited circuits in the DH leading to a generalized pattern of disinhibition and sensitization over widespread regions of the DH
- 3. Convergent input of nociceptive information from different tissues (e.g., muscle and skin), and transmitted by multiple, ascending pathways, onto a common populations of neurones in the thalamus or other supraspinal structures involved in second order, nociceptive processing
- Exchange of nociceptive information amongst specific, supraspinal pain-processing regions via corticothalamic loops, interhemispheric links etc: activation of cortical neurones with large and/or bilateral receptive fields
- 5. Triggering by ascending nociceptive information of a supraspinal loop leading to a heterosegmental modulation of descending inhibition (reduction) or descending facilitation (reinforcement). Regions of the DH affected include both those receiving PAF input from damaged tissue as well as *other* areas
- 1. spinal reflexes directly connecting the DH to preganglionic sympathetic neurones in the intermediolateral cell column; or
- engagement of a supraspinal loop leading to activation of descending pathways innervating preganglionic neurones (Jänig, 1993; Miao *et al.*, 1996; Millan, 1997; Takahashi *et al.*, 1993).

Within this context, it is also of note that unilateral injury to PAFs elicits *bilateral* alterations in their phenotype, possible by DH-integrated mechanisms comparable to those evoked above (Oaklander and Belzberg, 1997; Petersen *et al.*, 1998a).

Alterations in the sensitivity and activity of PAFs innervating regions of referred pain (whether antidromically- or sympathetically-mediated) are of particular interest. They may, thus, underlie the abovementioned, rare cases where anaesthesia of the intact region, rather than the site of injury, reduces referred pain. Further, changes in the activity of PAFs from areas of remote pain may underlie not only changes in nociception but also slowly-developing trophic phenomena, such as a perturbation of blood flow, oedema and alterations in skin texture (Jänig, 1993).

To summarize (Table 7), there is considerable evidence that *central* events integrated in the DH and/ or higher structures contribute to alterations in nociception from remote, intact tissue regions. These involve: sensitization of convergent neurones in the DH (and higher structures), widespread, non-topographical increases in DH excitability, modulation of the activity of descending pathways and, in certain cases, secondary, spinally- or sympatheticallymediated changes in the activity of PAFs innervating the region of referred pain.

9.2. C Fibres and Allodynic States

9.2.1. Involvement of C Fibres in Allodynic States

The above-discussed observations suggest that mechanical allodynia is predominantly mediated by large calibre, $A\beta$ fibres. However, they do *not* necessarily exclude a role of C fibres in the induction or maintenance of central sensitized states underlying mechanical allodynia, or in the development of

clinical, neuropathic painful states. For example, a recent study revealed a deep, C fibre-mediated component of clinical, neuropathic pain which was *not* affected by manipulations of $A\beta$ fibre/sympathetic fibre activity, despite the disappearance of tactile allodynia (Mailis *et al.*, 1997). The following points underline the potential importance of C fibres in mediating and/or maintaining painful states.

- 1. As discussed in Section 9.1.3, an afferent barrage from C fibres triggers the sensitization of WDR neurones in the DH and a persistent, low level of C fibre-mediated afferent drive may be necessary for maintaining central mechanisms underlying certain hyperalgesic and allodynic states (Andersen et al., 1995; Cervero et al., 1993; Cervero, 1995b; Dickenson, 1997; Gracely et al., 1992; Koltzenburg et al., 1994; LaMotte et al., 1991; Qian et al., 1996; Xu et al., 1995b). Mechanisms involved in maintaining a background state of central neuronal excitation may not be identical to those responsible for the effects of phasic superimposition of further, intense stimulation (Hoheisel et al., 1995). Thus, C fibre input to the DH may indirectly facilitate A β fibre-mediated allodynia by inducing and accentuating central processes of sensitization. Further, at C fibre-excited DH neurones, $A\beta$ fibre stimulation may be able to elicit processes of further sensitization (Section 10.5). In addition, excitatory C and A β fibre input may even synergize at the level of WDR DH neurones (Andersen et al., 1995; Choi and Rowbotham, 1997; Hoheisel et al., 1995, 1997).
- 2. Several, different forms of mechanical allodynia exist and it has been proposed that a non-A β class of PAF is involved in mediating a specific form of 'punctate' mechanical allodynia (Ali *et al.*, 1996; LaMotte *et al.*, 1991) (Section 9.2.2).
- 3. A recent theory has suggested that $A\beta$ fibres elicit mechanical allodynia (secondary hyperalgesia) by the (indirect) recruitment of C fibre terminals in the DH (Cervero and Laird, 1996b) (Section 9.2.3).
- 4. There is evidence that a disinhibition of small calibre, 'cooling-sensitive', C fibres mediates *cold* allodynia, a process involving adaptive changes

B. Central mechanisms

integrated at the level of the thalamus (Craig, 1995; Craig et al., 1996) (Sections 9.2.4 and 13.4).

5. It is of note that CNS damage or PAF injury may be accompanied by a C fibre-mediated hyperalgesia to heat which occurs simultaneously with a A β fibre-mediated, mechanical allodynia (Dickenson, 1997; Hao *et al.*, 1996a,b; Hunter *et al.*, 1997b; Kim *et al.*, 1997b; Qian *et al.*, 1996a; Shir and Seltzer, 1990; Wegert *et al.*, 1997).

Irrespective of the relative roles of C and $A\beta$ fibres, it should be emphasized that central, sensitized states are a precondition for *each* of the possible mechanisms underlying allodynia summarized in the following paragraphs.

9.2.2. 'Punctate' Hyperalgesia/allodynia

The discussion in Section 9.1.2 focused on the role of $A\beta$ fibres in mediating the mechanical allodynia which accompanies nerve injury, and which can be elicited from the area surrounding a region of peripheral tissue damage. This form of allodynia is evoked by normally innocuous, mechanical stimulation, such as light touch or brushing of the skin, and has been termed 'dynamic hyperalgesia'. It may be distinguished from a further form of mechanical allodynia, termed 'punctate', which is elicited by non-noxious, localized, mechanical stimuli: for example, Von Frey hairs (Koltzenburg et al., 1992; LaMotte et al., 1991; Warncke et al., 1997a). Like its dynamic counterpart, punctate allodynia requires the induction of central sensitization for its manifestation. However, punctuate allodynia develops more rapidly, is more persistent and occupies a larger tissue region.

It has been proposed that punctate allodynia is mediated by small calibre fibres (Cervero et al., 1993, 1994; Treede and Cole, 1993; LaMotte et al., 1991). However, if polymodal C receptors were involved, a thermal allodynia should also be apparent, yet this is not so. Further, the distribution of punctate allodynia might be anticipated to correspond to the zone of axonal flare, which is also not the case (Ali et al., 1996)-and the population of PAFs which mediates axonal flare may not be the same as that which mediates nociception (Lynn et al., 1996). One possible explanation would be the involvement in punctate allodynia of a 'mechanoselective channel': that is, of mechanospecific, small calibre (possibly type I A δ) fibres projecting onto sensitized WDR neurones-which may, in turn, selectively respond to mechanical input (Ali et al., 1996; Fitzgerald and Lynn, 1977; Treede and Magerl, 1995; Willis and Coggeshall, 1991) (Section 3.1.1). A further proposition by Ali et al. (1996) is that mechanical stimuli activate not only the same population of polymodal C fibres as thermal stimuli but also large diameter, mechanosensitive afferents which normally activate ININs in the DH, thereby counterbalancing the activation of WDR neurones. Under conditions of PAF injury, the activity of these ININs is reduced (by mechanisms discussed in Section 10.6) and mechanical nociception is selectively enhanced (Fig. 2). However, as discussed in Section 10.6, inflammatory tissue injury is not accompanied by a decrease in ININ tone in the DH,

questioning the pertinence of this hypothesis to secondary hyperalgesia. A related proposition of Treede and Magerl (1995), which also provides a more general explanation for selective, secondary mechanical vs thermal hyperalgesia, proposes that a generalized sensitization of *all* PNs may *not* occur in the DH. Rather, mechanosensitive, but not thermosensitive, afferents may impinge upon a subset of sensitized EXINs which themselves project onto PNs.

Thus, although the precise mechanisms involved in punctate allodynia require further elucidation, central, adaptive events emerge as of key importance in its expression. Further, consistent with a role of C fibres in its mediation, the NMDA receptor channel blocker, ketamine, was recently shown to suppress punctate, secondary hyperalgesia in man (Bouhassira *et al.*, 1997; Warncke *et al.*, 1997b).

9.2.3. An Aβ-C Fibre Link in the DH Underlying Allodynia?

As mentioned above, the absence of thermal allodynia presents a challenge for all theories of the mechanical allodynia (whether dynamic or punctate) which accompanies secondary hyperalgesia and peripheral nerve damage, since it is difficult to understand why WDR neurones in the DH should be selectively sensitized to specific, sensory modalities. Further, notwithstanding a key role of mechanoselective A β fibres, *the rapidity* with which mechanical allodynia can be induced by, for example, capsaicin treatment of human skin, suggests that this phenomenon occurs within a time-scale too rapid to allow for the establishment of novel patterns of $A\beta$ connectivity in the DH (Section 11.2). Based on a series of studies in hyperalgesic human skin, Cervero and Laird (1996b) have recently proposed a model of mechanical allodynia which helps account for these observations.

The hypothesis suggests, in principle, that mechanical allodynia depends upon activity in both $A\beta$ and C fibres, and that the former indirectly enhance input of the latter onto sensitized WDR (or NS) neurones. This is achieved by a mechanism involving A β fibre-mediated activation of a population of ININs in the DH which interact with the central terminals of small calibre, C fibres. This theory is, thus, based upon older studies showing that DRPs (that is, antidromic activity) can be evoked in the central terminals of C fibres by stimulation of INs interacting with their terminals (Section 9.1.4.2). The mild, depolarizing action of INs at C fibre terminals, termed 'primary afferent depolarization' (PAD), results in presynaptic inhibition of neurotransmitter release and, under normal conditions, antinociception (Melzack and Wall, 1965; Schmidt, 1971; Wall, 1995). This action is mediated via the activation of both GABAA receptors (coupled to chloride channels) and of non-NMDA (kainate or AMPA) ionotropic receptors (Cervero and Laird, 1996a; Rees et al., 1995a; Rudomin, 1990; Sluka et al., 1993, 1994a,b, 1995). The model of Cervero and Laird (1996a) assumes that these INs also possess a C fibre input. Thus, following tissue injury and inflammation, DH-localized INs are excited by C

fibres originating in the region of primary hyperalgesia. Activation of these INs triggers DRPs in the terminals of C fibres emanating from the neighbouring, uninjured zone of *secondary* hyperalgesia (Rees *et al.*, 1995a). As indicated above, this low level of C fibre terminal depolarization normally discourages propagation of APs and decreases transmitter release. However, the superimposition of an *additional* volley of excitatory input onto pre-excited INs by activation of A β fibres from the region of secondary hyperalgesia results in spike activity in C fibre terminals. This, in turn, results in:

- 1. a localized C fibre-mediated flare response peripherally (antidromic transmission); and
- 2. activation of DH neurones and the misinterpretation of $A\beta$ fibre-triggered, yet C fibre-mediated, excitation of WDR and NS as noxious.

This mechanism provides a more rapid mechanism for A β fibre-mediated activation of sensitized DH neurones than that of the reorganization of $A\beta$ input in the DH following PAF injury (Section 11.2). Further, an (indirect) activation of the central terminals of C fibres by $A\beta$ fibres would be consistent with the sensitization of DH neurones provoked by $A\beta$ fibre stimulation in inflamed states (Section 10.5), and with the induction of DH gene expression by A β fibre stimulation *only* following PAF nerve injury (Ma and Woolf, 1996a; Molander et al., 1992) (Section 11.2). The hypothesis is also consistent with the spontaneous activity, and increased sympathetic responsiveness, of large calibre $A\beta$ fibres following PAF injury (Section 8.2.4). Moreover, under conditions of inflammation, there is evidence for an increase in the activity of GABAergic ININS which mediate PAD via GABAA receptors on PAF terminals (Cervero and Laird, 1996a; Coggeshall and Carlton, 1997) (Sections 9.1.4.2 and 10.6). Nevertheless, several points require clarification.

- 1. Following PAF *injury*, notwithstanding the occurrence of allodynia, there is an—at least transient—decrease in the functional activity of GABAergic ININs in the DH, and DRPs may also be reduced by nerve injury (Laird and Bennett, 1992; Wall and Devor, 1981) (Sections 9.1.4.2 and 10.6).
- 2. A pronounced depolarization of C fibre terminals could result in a *resistance* to invasion by APs: that is, a shunt resulting in the *termination*—rather than potentiation of—forward transmission onto DH neurones (Cattaert *et al.*, 1994; Wall, 1995).
- 3. As mentioned in Section 9.1.3, there is evidence from other studies that *different* areas of hyperalgesic skin present flare responses *or* allodynia and that the underlying neuronal mechanisms may not be identical (Ali *et al.*, 1996; Lynn *et al.*, 1996).
- 4. It might be expected that, even in the absence of additional $A\beta$ fibre stimulation, the induction of PAD by INs should alter the effects of C fibre stimulation in the region of secondary hyperalgesia.

5. This model of allodynia, while providing an attractive, potential explanation of *rapidly-evoked* secondary hyperalgesia seems less pertinent to the allodynia associated with long-term PAF injury in which there is a reorganization of large calibre $A\beta$ input in the DH. This offers an alternative mechanism whereby normally innocuous, mechanical information can access and excite sensitized nocisponsive WDR (and NS) neurones and evoke the sensation of pain (Section 11.2).

Thus, despite the insights afforded by this novel theory, further work is required to improve our understanding of processes underlying mechanical allodynia. Irrespective of the precise underlying mechanisms, there is clearly a consensus that at least one important, clinical form of mechanical allodynia involves both the activation of A β fibres and central mechanisms of sensitization triggered—and possibly maintained by—C fibre-input to the DH.

9.2.4. Cold Allodynia

Cold allodynia refers to the induction of pain by normally non-noxious, cold stimuli. This abnormal responsiveness to cold is seen both experimentally and clinically from regions of secondary hyperalgesia and, in particular, from areas innervated by damaged nerves. It may be distinguished from the reduction in pain obtained upon the cooling of areas of primary hyperalgesia. Unmyelinated axons are the predominant route for the signalling of noxious, cold information in rats and, like thermal (heat) hyperalgesia, cold allodynia appears to be mediated by capsaicin-sensitive C fibres interacting with sensitized WDR neurones in the DH (Cervero et al., 1994; Craig et al., 1996; Hao et al., 1996a; Kim et al., 1995; Meller et al., 1992b,c; Shir and Seltzer, 1990; Wegert et al., 1997). It cannot be discounted that the population of C fibres which mediates cold allodynia is mechanosensitive and polymodal in nature, although this is unlikely.

In any case, cold-sensitive C fibres mediating allodynia appear to be independent of the population of cold-sensitive, $A\delta$ fibres which intensity-code temperature information (Sections 3.1.1 and 13.4) (Craig, 1995; Craig *et al.*, 1994; LaMotte and Thalhammer, 1982). Interestingly, cold-temperature/ $A\delta$ and cold-pain/C channels normally interact at the thalamic level and this mechanism may be compromised upon damage to the thalamus. That is, a lesion-induced disinhibition of a C fibre-activated medial thalamus–cingulate cortex circuit may underlie the cold allodynia provoked by stroke damage to the thalamus (Boivie *et al.*, 1989; Craig *et al.*, 1994 and 1996; Vanhoesen, 1995) (Section 13.4).

10. NOCICEPTIVE AND NEUROPATHIC PAIN: PROCESSES INTEGRATED IN THE DH

10.1. Multiple Mechanisms Integrated in the DH

From the above-discussed observations (Section 9), the most satisfactory explanation of the phenomena of mechanical allodynia, secondary hyperalgesia and referred pain is that they involve *central*, adaptive changes triggered by peripheral events and involving both C and $A\beta$ fibre input.

Several processes integrated in the DH have been implicated in the induction of painful states, and each of these evolves with a distinctive time-course (Table 5):

- 1. Neuronal sensitization; that is, an increased excitability of WDR neurones to both C and A β fibre input. This can occur rapidly (within minutes) and is triggered, and usually enhanced by C fibre input. NMDA receptors play a key role in both the induction and maintenance of such sensitized states.
- 2. As concerns pain due to PAF injury, a reduction in the activity of ININs targeting PNs, reflecting either transient functional changes or their physical degeneration.
- 3. Likewise for PAF damage, a reduction in excitatory $A\beta$ input onto ININs in the DH.
- 4. A modulation in the activity of descending pathways.

These processes, which are likely interactive and synergistic, are described below. In addition, as outlined in Section 11.2, nerve injury may trigger a structural reorganization of the DH involving the inappropriate sprouting of damaged $A\beta$ afferents from deeper into superficial laminae, a mechanism lying at the interface of central and peripheral events contributing to neuropathic pain.

10.2. Sensitization and Increased Excitability of DH Neurones

10.2.1. LTP and Neuronal Sensitization

Studies of the hippocampus and other cerebral regions have established that persistent alterations in synaptic efficacy may be produced postsynaptically by alterations in patterns of presynaptic stimulation. Correspondingly, LTP and long-term depression (LTD) refer to an increase and a decrease in synaptic strength, respectively, with changes in postsynaptic, cytosolic levels of $[Ca^{2+}]_i$ and PKs playing a key role in their generation (Barria et al., 1997; Bito et al., 1997; Bliss and Collingridge, 1993; Deisseroth et al., 1998; Linden and Connor, 1996; Lisman, 1994, 1997; Malenka and Nicoll, 1997; McEachern and Shaw, 1996; Wang et al., 1997g; Xu et al., 1997a). Recent studies have suggested that changes resembling LTD can be induced in the substantia gelatinosa by low frequency stimulation of afferent A δ fibres. On the other hand, high frequency, tetanic stimulation of C (and, probably, $A\delta$) fibres may elicit processes similar to LTP in DH neurones (Bouhassira et al., 1997; Liu and Sandkühler, 1997; Randic et al., 1993; Sandkühler et al., 1997; Svendsen et al., 1997; Willis, 1997). Furthermore, there is considerable evidence that intense, recurrent and/or sustained electrical or noxious stimulation of C fibres leads to an increase in synaptic efficacy and WDR neurone excitability in the DH. Such processes of 'central sensitization' also appear related to processes underlying LTP and similarly involve the engagement of NMDA receptors and alterations in levels of $[Ca^{2+}]_i$ and PKs (Baranauskas and Nistri, 1998; Bouhassira *et al.*, 1997; Dickenson, 1997; Liu and Sandkühler, 1997; McEachern and Shaw, 1996; Lisman *et al.*, 1997; Svendsen *et al.*, 1994, 1997; Willis, 1997) (Sections 10.3 and 10.4).

Supraspinal processes of LTP and memory formation can persist for days or longer even in the absence of additional stimulation, and they may be associated with alterations in gene expression and, ultimately, morphological changes. Further, the induction of LTP, alterations in synaptic strength and memory storage may involve a multiplicity of preand postsynaptic mechanisms operating in parallel and in synergy (Bliss and Collingridge, 1993; Shaw, 1996; McEachern and Schulz and Fitzgibbons, 1997; Turrigiano et al., 1998). On the other hand, the sensitization of DH neurones can be expressed very rapidly and reversibly in the absence of morphological changes, and may require peripheral input for its maintenance. It might, thus, be questioned whether rapid DH sensitization can genuinely be considered as equivalent to processes of LTP. However, although DH mechanisms of neuronal sensitization are not invariably comparable to processes of LTP, they possess several common underlying mechanisms (vide supra) and recent studies have re-asserted the possibility that LTP-like phenomena can be triggered in DH neurones by nociceptive input to the DH (Baranauskas and Nistri, 1998; Dickenson, 1997; Liu and Sandkühler, 1997; Svendsen et al., 1997; Wilcox, 1993b; Willis, 1997). On teleological grounds, it might be argued that LTP-like events are *inconsistent* with a physiological, reversible and short-term protective role of DH sensitization (Section 9.1.1). Indeed, the induction of prolonged-and perhaps irreversible-LTP-like states of enhanced excitability, involving structural modifications of synaptic architecture, may underlie chronic, pathological, painful states due to, for example, PAF injury (Section 10.2.1).

In analogy to processes of peripheral nociceptor sensitization and the recruitment of 'silent nociceptors' (Sections 3.1.3 and 7.5), those central processes of response amplification and sensitization which underlie prolonged, painful states may reflect both increases in synaptic efficacy as well as the transformation of synapses from a 'silent' to an 'active' mode (Isaac *et al.*, 1997; Malenka and Nicoll, 1997).

10.2.2. Induction of Sensitization in DH Neurones: Characteristics of Sensitized Neurones

Damage to the skin and other organs, cutaneous application of capsaicin, inflammation, intrathecal administration of GLU or brief electrical activation of C (and possibly $A\delta$, but not $A\beta$) fibres elicits a state of enhanced excitability in the DH, in particular of WDR neurones in deep laminae (Clément *et al.*, 1996; Coutinho *et al.*, 1996; Dickenson, 1997; Dougherty and Willis, 1992; Ide *et al.*, 1997; King and Thompson, 1995; Kolhekar and Gebhart, 1996; Mense, 1993; Schaible and Grubb, 1993). Stimulation at 1 Hz for 20 sec, for example, elicits an increase in excitability which persists for about 1 hr (Baranauskas and Nistri, 1998; King and Thompson, 1995; Owens *et al.*, 1992; Wall and Woolf, 1984; Woolf and Wall, 1986; Woolf, 1983; Woolf and Wiesenfeld-Hallin, 1986). Increased neuronal excitability manifests several general characteristics.

- 1. A transient stimulus now provokes a response of greater duration and intensity involving a greater number of APs: that is, the magnitude of the response is enhanced (cf hyperalgesia).
- 2. RFs are magnified such that responses can now be evoked from a larger area. Normally, only stimulation in a spatially-limited, discrete, 'trigger' region evokes APs. In contrast, stimulation of the surrounding, 'subliminal' zone, which provides a convergent but less strong input to DH neurones, is ineffective—at most eliciting EPSPs but not provoking firing. This fringe zone now *also* becomes the origin of frank discharges (cf secondary hyperalgesia).
- 3. The threshold for triggering firing is reduced such that neurones can be activated by lower, normally sub-noxious, stimulus intensities. There is also an increase in, and an appearance of novel, responses to $A\beta$ fibres (cf allodynia).

The relationship of these changes to experimental and clinical observations of primary and *secondary* hyperalgesia, and of $A\beta$ fibre-mediated mechanical allodynia is, thus, evident.

The increased excitability of deep DH laminae provoked by repetitive C fibre stimulation may be explained by the temporal and spatial summation of rapid and, more importantly, slow (many seconds) EPSPs. This process reflects the sequential yet intercalated activation of several receptor types. Thus, upon activation of C fibres, an initial and rapid EPSP lasting a few milliseconds is provoked by the engagement of AMPA receptors. This triggers the opening of VDCCs and further depolarization. Further, it is succeeded by the activation of $NK_{1/2}$, group I MTB and CGRP receptors mediating sustained EPSPs over tens of seconds. The differential mechanisms underlying these various components of excitation, and their cooperative nature, is explained further in Sections 10.3 and 10.4. Evidently, where noxious stimulation is maintained and firing is rapid, EPSPs will summate, leading to an incremental depolarization, induction of APs and an increase in excitability. In this regard, two related events should be recognized.

- 1. C fibre-mediated 'wind-up' refers to the gradual neuronal depolarization and increased number of APs evoked by C fibre stimulation (King and Thompson, 1995).
- 2. 'Heterosynaptic facilitation' refers to the process whereby a progressive increase in neuronal excitability leads to an increased amplitude in the response to *other* inputs, specifically $A\beta$ fibres (Thompson *et al.*, 1993b; Woolf and Doubell, 1994).

These processes might, therefore, be considered as substrates of hyperalgesia and allodynia, respectively, and would both contribute to the expansion of RFs.

To summarize, sensitization of WDRs is associated with an increase in their spontaneous neuronal activity, an expansion of their RFs (too extensive to be accounted for by changes in the RFs of their corresponding PAFs), an increased responsiveness to noxious, C-fibre-mediated input and the induction of responses by low threshold (normally innocuous) A β fibre input (Baranauskas and Nistri, 1998; Dougherty et al., 1992c; Dubner, 1992; Hylden et al., 1989; McMahon and Wall, 1984; Woolf and King, 1990). In correspondence with this schema, diverse noxious stimuli (such as capsaicin) applied to the skin within the RFs of STT and other deep WDR neurones in rodents and primates have been shown to decrease their thresholds for responding to mechanical stimuli, expand their RFs and, in particular, enhance their responsiveness to innocuous A β fibre-mediated mechanical stimulation from the area of secondary hyperalgesia (Cook et al., 1987; Dougherty et al., 1992c; Laird and Cervero, 1989; Lin et al., 1997; Simone et al., 1991; Thompson et al., 1993b).

The above description of sensitization is largely based on observations obtained from WDR neurones in deep DH laminae of rats and primates. In principle, assuming that the necessary intracellular machinery is present, sensitization might be initiated by C fibre/NMDA stimulation of other neuronal classes in the DH. Thus, although monosynaptic sensitization of PNs definitely does occur, they may also be secondarily (polysynaptically) excited and sensitized by (presensitized) EXINs. Further, ININs might themselves be sensitized by recurrent and intense C fibre input, a possibility supported by neurochemical and functional observations of increases in the activity of opioidergic (DYN-containing) and GABAergic ININs in the DH under conditions of inflammation (Chapman and Besson, 1997; Menétrey et al., 1989; Millan, 1990, 1993; Todd et al., 1994a) (Sections 10.4.2.5 and 10.6). In addition to WDR units, superficial or deep laminae NS neurons may also be sensitized although, to date, direct evidence in support of this possibility is limited (Chapman and Besson, 1997; Dougherty et al., 1992b,c; Cervero, 1995b; Simone et al., 1991; Tavares et al., 1993). Thus, the following description of molecular mechanisms underlying DH sensitization concerns events documented primarily in deeper laminae WDR neurones, such as PNs of the STT. Although the following paragraphs focus on EAAs, tachykinins and CGRP, a potential role of other excitatory neuropeptides, PGs, ATP, neurotrophins, cytokines and other substances released from PAF terminals in mediating (or modulating) the sensitization of DH neurones should not be neglected (Sections 3.2.6, 4.8 and 14). In any case, all such potential mediators likely converge on common, intracellular transduction mechanisms involved in the sensitization of SH neurones and illustrated in Fig. 5 (Section 10.4).

10.3. The Role of EAAs, Tachykinins and CGRP in the Sensitization of DH Neurones

10.3.1. Molecular Mechanisms Underlying Neuronal Sensitization: The Key Role of NMDA Receptors

Recent studies have begun to unravel the mutually-interactive pattern of cellular events which underlies the sensitization of DH neurones by C fibre input from the skin and other organs (Baranauskas and Nistri, 1998; Coderre et al., 1993; Coutinho et al., 1996; Dickenson, 1997; Ide et al., 1997; Kolhekar and Gebhart, 1996; King and Thompson, 1995; Mense, 1993; Schaible and Grubb, 1993; Urban et al., 1994a,b; Wilcox, 1991; Woolf, 1994). As mentioned above, the following account concentrates upon the actions of EAAs, tachykinins and SP, but the possible role of other neurotransmitters, in particular excitatory neuropeptides, in the sensitization of DH neurones, should not be forgotten (Baranauskas and Nistri, 1998; Coggeshall and Carlton, 1998). Indeed, the summation of multiple, slow and sustained EPSPs and depolarizations plays a key role in the sensitization of DH neurones. The underlying processes are of considerable complexity and the following key points should be emphasized.

- 1. The release of EAAs, tachykinins and CGRP from a subset of nocisponsive C fibres innervating the skin and other organs fulfils a key and cooperative role in eliciting DH sensitization.
- 2. The influence of PAF-localized transmitters upon the activity of DH neurones is mediated via:
 - 1. a direct alteration in ion flux at cation-permeable channels;
 - 2. an interaction with intracellular transduction mechanisms leading to the secondary modification of ionic currents (for example, via receptor/ion channel phosphorylation) and
 - 3. longer-term effects involving processes of receptor recycling and alterations in the gene transcription of receptors, transmitters and other molecules (Mantyh *et al.*, 1995; Sluka *et al.*, 1997a,b; Lin *et al.*, 1997).
- 3. Sensitizing mechanisms may all ultimately converge to reinforce transmission at NMDA receptors, thereby resulting in a pronounced and sustained elevation in Ca^{2+} -influx and $[Ca^{2+}]_i$ levels.
- 4. NMDA receptors display several, distinctive attributes underlying their key importance in both the induction and maintenance of sensitized states (Chapman *et al.*, 1996; Chen and Huang, 1992; Coderre and Melzack, 1991, 1992a,b; Coderre *et al.*, 1993; Kolhekar and Gebhart, 1996; King and Thompson, 1995; Mayer and Miller, 1990; Seguin *et al.*, 1995; Woolf and Thompson, 1991; Yoshimura and Jessel, 1990). These properties may be summarized as follows:
 - 1. under resting conditions, NMDA receptors are quiescent owing to an intra-channel Mg^{2+} -block, the relief of which by depolarization is required to permit the sustained passage of current (Chizh *et al.*, 1997; Sharma and Stevens, 1996);

- the NMDA receptor-coupled ion channel is permeable to Ca²⁺-ions. Thus, NMDA receptor activation not only depolarizes cells, but also increases levels of [Ca²⁺]_i (Mayer and Miller, 1990; Reichling and MacDermott, 1996);
- 3. the phosphorylation of NMDA receptors by PKC counteracts the Mg²⁺-block and allows NMDA receptors to operate at more negative (hyperpolarized) potentials (Ben-Ari *et al.*, 1992; Lerea, 1997);
- the activity of PKC is synergistically facilitated by increases in [Ca²⁺]_i and the generation of DAG following activation of PLC by NK_{1/2} and group I mGlu receptors (Ramakers *et al.*, 1997; Zheng *et al.*, 1997);
- activation of NMDA receptors can elicit a 'burst'-like pattern of firing in DH neurones, an action which is enhanced by the interruption of glycinergic/GABAergic ININ transmission (Grubb *et al.*, 1996) (Section 10.6). Based on studies elsewhere in the CNS (Johnson *et al.*, 1992; Lejeune *et al.*, 1997), 'burst' firing markedly enhances neurotransmitter release. Further, it may set-up reverberatory, hyperexcitable circuits and reinforce the rostral transmission of nociceptive information by PNs;
- 6. presynaptic NMDA receptors on central PAF terminals may enhance the release of SP and EAA, thereby further accentuating sensitizing events at PNs in the DH (Coggeshall and Carlton, 1997; Liu et al., 1997b) (Section 3.2.4). This action is complemented by a further component of NMDA receptormediated PAF release indirectly mediated via NO (Sorkin, 1993). There is also evidence for an increase in PAF levels of GLU under conditions of peripheral inflammation (Sluka and Westlund, 1993; Westlund et al., 1992). However, there are contradictory data concerning changes in the density of NMDA receptors in the DH in painful states. Minor reductions in NMDA receptor density might be attributable to their down-regulation by an enhanced PAF input, or to their loss upon the transynaptic degeneration of ININs triggered by PAF injury (Coggeshall and Carlton, 1997; Harris et al., 1996; Dickenson, 1997; Hama et al., 1995; Pellegrini-Giampietro et al., 1994; Sloan et al., 1991) (Section 10.6).

Overall, then, upon repetitive C fibre stimulation, activity at NMDA receptors is augmented by several mechanisms of response *amplification*. Indeed, for the above reasons, NMDA receptors and increases in $[Ca^{2+}]_i$ levels fulfil a particularly important role in triggering and maintaining neuronal sensitization in the DH, a process underlying the development of hyperalgesic and allodynic states. In addition, under conditions of PAF injury, the increase in DH excitability mediated via NMDA receptors may contribute to the diminished antinociceptive efficacy of μ -opioid agonists. Correspondingly, the administration of NMDA receptor antagonists potentiates

the analgesic properties of opioidergic agents (Dickenson, 1997; Felsby *et al.*, 1995; Hama *et al.*, 1995; Hoheisel *et al.*, 1997; Lewin *et al.*, 1994a; Mao *et al.*, 1993b; Smith *et al.*, 1994b; Ossipov *et al.*, 1995, 1997; Thompson *et al.*, 1995; Wiesenfeld-Hallin, 1998).

A recent study has elegantly exemplified the functional interrelationships amongst NMDA receptor activation, increases in $[Ca^{2+}]_i$ levels and the acute modulation of neuropathic pain. Thus, Kawamata and Omote (1996) showed that PAF damage elicits an increase in the GLU content of superficial and deep laminae of the ipsilateral DH together with a parallel elevation in levels of $[Ca^{2+}]_i$: the NMDA channel blocker, dizolcipine, blocked both the rise in $[Ca^{2+}]_i$ levels and the accompanying hyperalgesia.

The contribution of specific receptor types to intracellular mechanisms underlying C fibre-mediated sensitization of WDR neurones in the DH is outlined in the following paragraphs. A graphic summary of events is presented in Fig. 5. The present description is based primarily on observations acquired in DH neurones themselves. However, it also incorporates a few, speculative elements (see legend to Fig. 5) derived from LTP and other processes of synaptic plasticity and response amplification contributing to memory formation in the hippocampus and cortex (Section 10.2.1).

10.3.2. The Roles of AMPA, $NK_{1/2}$, MGlu and CGRP Receptors in Neuronal Sensitization

10.3.2.1. AMPA receptors

Upon C fibre stimulation, the release of GLU, aspartate or other EAAs evokes a fast (milliseconds) EPSP at ionotropic AMPA sites mediating cationic currents, predominantly to Na⁺ and only to a lesser degree to Ca²⁺ (Dickenson, 1997; Lerea, 1997; Wood and Docherty, 1997) (Section 3.2.3). The instantaneous engagement of AMPA receptors likely fulfils an important role in the phasic transmission of nociceptive information under 'baseline' conditions and at low frequency rates (Dickenson, 1997; Hunter and Singh, 1994). The activation of AMPA receptors leads to the secondary activation of VDCCs which further amplify depolarization and trigger an increase in $[Ca^{2+}]_i$ (Wood and Docherty, 1997). Thus, processes which eventually lead to a relief of the voltage-dependent Mg^{2+} block at NMDA receptors are initiated via AMPA receptor stimulation. Nevertheless, the ephemeral (msec) nature of AMPA receptor-mediated currents (which may rapidly desensitize), and the topographicallyrestricted extent of alterations in intracellular levels of $[Ca^{2+}]_i$, is incompatible with a major role in the activation of NMDA receptors and the induction of sensitization (Sections 3.2.3 and 10.5). Furthermore, the gene expression of AMPA receptors (gluRI subunit) is diminished under conditions of peripheral inflammation-perhaps suggesting a decreased Ca^2 ⁺-permeability (Pellegrini-Giampietro *et al.*, 1994). Indeed, over a longer time-scale, sustained NDMA receptor activation may, as described below, be more convincingly be ascribed to the subsequent engagement of NK1 and other receptor

types eliciting prolonged alterations in intracellular signals in the vicinity of NMDA receptors. In line with these considerations, the balance of functional evidence suggests that AMPA receptors play a comparatively minor role in the mediation of persistent painful states (Coderre and Melzack, 1991, 1992a,b; Dickenson, 1997; Hunter and Singh, 1994; King and Thompson, 1995; Nasström *et al.*, 1992; Székely *et al.*, 1997). Further, although AMPA receptors are implicated in mediating mechanical allodynia, the population involved is likely activated by EAAs released from $A\beta$ —rather than C-fibres onto *presensitized* WDR (or NS) neurones (Cumberbatch *et al.*, 1994; Millan, 1995; Sherman and Loomis, 1996; Xu *et al.*, 1993a; Zhang *et al.*, 1994a) (Section 9.1.2).

Notwithstanding the above observations, recent studies of the interrelationship between NMDA and AMPA receptors in the induction of hippocampal LTP suggest that the activity of AMPA receptors may be reciprocally enhanced via the engagement of NMDA receptors. Thus, a NMDA receptor-mediated increase in $[Ca^{2+}]_i$ levels results in the autophosphorylation of calmodulin kinase II which subsequently phosphorylates and increases the responsiveness of AMPA receptors. Such processes, which occur over a time-scale of 10–15 min, suggest that AMPA receptors play a significant role in sustaining LTP and mediating synaptic plasticity (Barria et al., 1997; Lisman, 1994; Lisman et al., 1997). An exploration of a possible role of such mechanisms in the sensitization of DH neurones would be of interest. Further, the influence of PAF injury upon AMPA receptors in the DH differs to that of inflammatory, painful states. Thus, there may be an increase in the density of AMPA sites on intrinsic neurones in the DH while the ratio of the gluR1 to the gluR2 subunit is increased, suggestive that their permeability to Ca^{2+} is enhanced (Carlton *et al.*, 1998b; Coggeshall and Carlton, 1998; Harris et al., 1996). These observations suggest that, under conditions of neuropathic pain, AMPA receptors-activated by small or large calibre fibre input-may make a more important contribution to processes of sensitization in the DH.

10.3.2.2. $NK_{1/2}$ receptors

Upon activation of C fibres, SP and NKA are released in the DH. Via actions at NK₁ and NK₂ receptors, respectively, they together make an important contribution to the induction of DH sensitization (Abbadie et al., 1997; Aicher et al., 1997; Donnerer et al., 1993; McCarson and Krause, 1994, 1996; Krause et al., 1995; Liu and Sandkühler, 1997; Maggi and Schwartz, 1997; Noguchi and Ruda, 1992; Noguchi et al., 1993, 1994; Traub, 1996). As compared to NMDA receptors, however, the selective engagement of NK1 and NK2 receptors likely plays a less marked role in the maintenance of sensitized (inflammatory) states once established (De Koninck and Henry, 1991; Liu and Sandkühler, 1997; Mantyh et al., 1997; Rusin et al., 1992; Thompson et al., 1993a, 1994; Traub, 1996). It is possible that this lesser importance of NK₁ receptors may, at least partially, reflect their rapid, PKCmediated phosphorylation-although the significance of this mechanism of desensitization in the DH remains unclear (Quartara and Maggi, 1997). In any case, over a time-scale of hours, the endosomic internalization of NK1 receptors on neurones in superficial laminae (I/II₀) of the DH and on the dendrites of neurones in deeper laminae results in their disappearance from the cell membrane. However, this process is only transient and, following degradation of SP, functionally-active NK1 receptors are rapidly recycled to the cell surface. Indeed, PAF damage and-less consistently-peripheral inflammation may be accompanied by an increase in the density of NK1 receptors in the DH, while the release of SP from PAFs is also enhanced by peripheral inflammation (Abbadie et al., 1996; Allen et al., 1997a; Coggeshall and Carlton, 1997; Garland et al., 1996; Goff et al., 1998; Liu et al., 1997b; Mantyh et al., 1995; Tao et al., 1997) (Section 7.6).

The principal NK_1/NK_2 receptor-coupled, cellular mechanism underlying the induction of slow depolarizations and sustained EPSPs by SP and NKA is the activation of PLC leading to the formation of:

- 1. IP₃, which releases Ca²⁺ from intracellular sequestration in the endoplasmic reticulum; and
- DAG, which, synergistically with [Ca²⁺]_i, activates PKC leading to the phosphorylation of NMDA receptors and a further relief of their Mg²⁺-block (Ben-Ari *et al.*, 1992; Chen and Huang, 1992; Mantyh *et al.*, 1989; Otsuka and Yoshioka, 1993; Quartara and Maggi, 1997; Rusin *et al.*, 1993; Urban *et al.*, 1994a,b).

As described in Section 10.2, an increase in $[Ca^{2+}]_i$ levels may exert several other actions contributing to an enhancement of neuronal excitability, including the generation of NO, which has been implicated in the pronociceptive actions of SP on DH neurones (Radhakrishnan et al., 1995; Tao et al., 1997) (Section 10.4.2.2). In addition to the induction of PKC, several other intracellular transduction mechanisms may potentially intervene in the excitatory actions of SP at NK1 receptors (Quartara and Maggi, 1997). For example, a component of the slow membrane depolarization which is triggered by SP in certain neuronal types is effected-possibly independently of PLC—via a decrease in an inwardly-rectifying, Ca^{2+} -dependent K⁺-current (Baranauskas and Nistri, 1998; Bentley and Gent, 1995; Murase and Randic, 1984; Nowak and MacDonald, 1982; Otsuka and Yoshioka, 1993; Quartara and Maggi, 1997; Shen and North, 1992). However, the potential role of intracellular transduction mechanisms other than PLC in the actions of tachykinins at DH neurones requires clarification. In addition to effects on WDR themselves, SP exerts a (partially indirect) positive feedback action at PAF terminals to further facilitate the release of GLU and SP (Coggeshall and Carlton, 1997; Hu et al., 1997; Kangrga and Randic, 1990; Skilling et al., 1990; Smullin et al., 1990; Sorkin, 1993).

The relative importance of NK_1 and NK_2 receptors in mediating the facilitory, sensitizing actions of tachykinins on DH neurones is still under discussion. In fact, it has proven difficult to visualize NK_2 receptors in the DH. Further, although there is evidence suggesting that DH-localized NK_2 receptors

are involved in spinal nociceptive transmission, this role appears to be exerted phasically. Indeed, under more prolonged conditions, behavioural and electrophysiological studies from a broad variety of tissues suggest that NK₁ receptors assume the major role (Coggeshall and Carlton, 1997; De Koninck and Henry, 1991; Fleetwood-Walker et al., 1990; Jia and Seybold, 1997; Liu and Sandkühler, 1997; Nagy et al., 1994; Neugebauer et al., 1995; Parsons et al., 1996; Radhakrishnan and Henry, 1997; Seguin et al., 1995; Sluka et al., 1997a; King and Thompson, 1995; Urban et al., 1994a,b). There is substantial evidence that activation of NK1 receptors reinforces excitatory actions mediated by NMDA receptors in the DH. Indeed, the *cojoint* activation of NK_1 and NMDA receptors may be essential for a maximal degree of neuronal excitation. Correspondingly, persistent painful states reflecting sensitization of DH neurones can be synergistically (or additively) blocked by co-administration of antagonists at NMDA and NK₁ receptors (Chapman et al., 1996; Chizh et al., 1995; Clayton et al., 1997; Dougherty et al., 1995; Heppenstall and Fleetwood-Walker, 1997a,b; Mjellem-Joly et al., 1991; Nagy et al., 1993; Ren et al., 1996; Rusin et al., 1992; Seguin and Millan, 1994; Xu et al., 1992).

Interestingly, under conditions of PAF injury, the role of SP in nociceptive transmission and DH sensitization may, at least partially, be assumed by VIP, the levels of which are up-regulated in parallel with a decrease in those of SP. However, unlike NK_{1/2} receptors, VIP receptors are positively coupled to AC (Hökfelt *et al.*, 1994; Vertongen *et al.*, 1997) (Section 3.2.6). The relationship between actions mediated by VIP and NMDA receptors under conditions of PAF damage would be of interest to examine.

10.3.2.3. mGlu receptors

In analogy to NK_{1/2} receptors, group I mGlu₁ and mGlu₅ receptors are positively coupled to PLC. Thus, their activation ultimately results in an increase in $[Ca^{2+}]_i$ and a stimulation of PKC—a feedback action of which may, however, result in the rapid desensitization of mGlu receptors by their phosphorylation (Conn and Pin, 1997; Gereau and Heinemann, 1998; Stefani et al., 1996; Toms et al., 1996; Valerio et al., 1997; Young et al., 1995). Correspondingly, the activation of group I mGlu₁ (and mGlu₅) receptors likely initiates events in DH neurones similar to those outlined above for SP (Section 10.3.2.2). Further, evidence from cerebral structures that group I mGlu receptors are involved in processes eliciting LTP under certain-though not all-conditions, supports the concept that they potentiate NMDA receptor-mediated processes of sensitization in the DH (Bortolotto et al., 1994, 1995; Huber et al., 1998; Martin and Morris, 1997; Vickery et al., 1997). Studies in other tissues have shown that PKC inhibitors interfere with the ability of mGlu receptors to potentiate activity at NMDA receptors (Ben-Ari et al., 1992; Kelso et al., 1992).

In correspondence with these observations, there is electrophysiological and behavioural evidence that the activation of DH-localized mGlu receptors by noxious, inflammatory stimuli elicits a slow excitation of spinal neurones (via an increase and decrease in cation- and K⁺-currents, respectively) and cooperatively facilitates the excitatory actions of NMDA receptors. Interestingly, further, mGlu receptors also potentiate the excitatory actions of AMPA receptors, with which they act collaboratively in recruiting NMDA receptors (Budai and Larson, 1998; Coderre, 1993; Dickenson, 1997; Fisher and Coderre, 1996; Hunter and Singh, 1994; Meller *et al.*, 1993; Neugebauer *et al.*, 1994; Young *et al.*, 1994, 1995, 1997).

MGlu₁ and mGlu₅ receptors may, then, by virtue of their positive coupling to PLC and, possibly, NO synthase, activate DH neurones. In line with their high density in the DH-especially lamina II-pharmacological evidence for a key role of mGlu₁ receptors in nociceptive transmission was recently presented (Coggeshall and Carlton, 1997; Conn and Pin, 1997; Young et al., 1997). The potential significance of other mGlu subtypes in modulating nociception requires further elucidation (Budai and Larson, 1998; Fisher and Coderre, 1996; Stefani et al., 1996; Toms et al., 1996; Young et al., 1997). Indeed, group II and III mGlu receptor subtypes are all negatively coupled to AC. This suggests that, should they be present on WDR neurones, their activation may counter increases in WDR excitability provoked by other mechanisms. Nevertheless, it remains possible that certain group II mGlu receptor subtypes increase cAMP levels and activate PKA independently of AC (Winder and Conn, 1995). Further, an influence of certain group II and/ or group III mGlu receptors upon Ca^{2+} and K⁺currents may increase neuronal excitability (Conn and Pin, 1997). In addition, there is evidence from other tissues that group II receptors can facilitate the induction of PLC by group I receptors, suggesting a putative synergistic role of group I and II mGlu receptors in neuronal sensitization (Mistry et al., 1998). Such potential mechanisms require direct evaluation in the DH. In a recent study of the influence of peripheral inflammation upon mRNA levels encoding various mGlu receptor types in the DH, only the density of mGlu₃ receptors was affected (Boxal et al., 1998).

An additional level of complication is provided by observations that mGlu receptors modulate EAA release from fine calibre PAFs in the DH. For example, there is evidence for an involvement of group II mGlu₇ (or L-AP₄-sensitive receptors). Indeed, group II (and mGlu III) receptors, which are negatively coupled to AC, exert a predominantly inhibitory influence upon presynaptic release, via actions involving a reduction in Ca²⁺-currents and, possibly, a potentiation of K⁺-currents (Conn and Pin, 1997; Crawford *et al.*, 1997; Fisher and Coderre, 1996; Pin and Duvoisin, 1995; Li *et al.*, 1997a). The identity of excitatory mGlu receptors on PAF terminals remains unclear (Coggeshall and Carlton, 1997; Sánchez-Prieto *et al.*, 1996).

10.3.2.4. CGRP receptors

The role of CGRP in the DH is less well-defined than those of EAAs and SP, but it probably facili-

tates their actions. Thus, CGRP elicits slow membrane depolarizations in parallel with an increase in $[Ca^{2}]$ ⁺]_i and an enhancement of Ca²⁺-influx through (possibly L-type) VDCCs. These actions potentiate the effects of NK1 and NMDA receptor activation and further strengthen processes of sensitization (Miletic and Tan, 1988; Oku et al., 1988; Woolf and Wiesenfeld-Hallin, 1986). The mechanism(s) underlying these actions of CGRP of DH neurones are unclear, but presumably involve the positive coupling of CGRP1 and CGRP2 receptors to AC leading, via cAMP, to the activation of PKA. In addition, an increase in NO synthesis, leading to the formation of PKG, may be involved (Sluka and Willis, 1997; Wimalawansa, 1996).

In addition to these poorly-characterized postsynaptic actions at CGRP at DH neurones, CGRP facilitates nociception by directly retarding the metabolism of SP, thereby encouraging 'volume transmission' (Section 4.7). Further, CGRP may enhance the release of GLU and SP from PAF terminals (Kangrga and Randic, 1990; Schaible et al., 1992; Wimalawansa, 1996). An involvement of CGRP in inflammatory nociception is supported by the increase in DRG and DH levels of CGRP which is elicited by chronic inflammation and by the reduction in the accompanying nociception effected both by the antagonist, GGRP₈₋₃₇, and by sequestration of spinal pools of CGRP with specific antibodies (Hökfelt et al., 1994; Kuraischi et al., 1989; Löfgren et al., 1997; Malcangio and Bowery, 1996a; Satoh et al., 1992; Wimalawansa, 1996: though see Yu et al., 1996) (Section 7.6).

10.4. Cellular Processes Underlying Sensitization of DH Neurones: Focus on Ca²⁺ and PKs

As described in this section, the rapid and longterm sensitization of DH neurones by EAAs, tachykinins and other excitatory mediators reflects their influence upon multiple intracellular transduction mechanisms. In this regard, alterations in the levels of $[Ca^{2+}]_i$ and the activity of various PKs play particularly crucial roles.

10.4.1. Increases in Ca^{2+} -Influx and in $[Ca^{2+}]_i$ Levels in DH Neurones

The above observations point to a critical role of Ca^{2+} -influx and increases in $[Ca^{2+}]_i$ levels in the sensitization of DH neurones. Consistent with this contention, Ca²⁺-channel antagonists and Ca² ionophores, respectively, inhibit and potentiate inflammatory nociception. In distinction they do not modify the effects of transient, intense, noxious stimuli (Coderre and Melzack, 1992a,b; Coderre et al., 1993; Diaz and Dickenson, 1997; Malmberg and Yaksh, 1994, 1995; Miljanich and Ramachandran, 1995; Miranda et al., 1992; Neugebauer et al., 1996). The activation of L-type Ca²⁺-channels on intrinsic DH neurons may be of particular significance in the mediation of slow, summating Ca²⁺-currents contributing to gradual and sustained increases in excitability (Baranauskas and Nistri, 1998; Morisset and Nagy, 1996). Further, L-type Ca²⁺-channels are an important source of those intracellular pools of $[Ca^{2+}]_i$ which modulate gene transcription via various nuclear regulatory elements (Section 10.4.2.5). On the other hand, the effects of antagonists at N-(and P-) type Ca^{2+} -channels—which are concentrated in superficial lamina I/II—also reflect actions at the terminals of PAFs (Diaz and Dickenson, 1997; Malmberg and Yaksh, 1994; Miljanich and Ramachandran, 1995; Nebe *et al.*, 1997) (Section 8.2.2). The respective roles of individual Ca^{2+} -channel subtypes requires, thus, further characterization.

To summarize the findings discussed above, increases in $[Ca^{2+}]_i$ levels may be derived from several sources (Berridge, 1993; Bito *et al.*, 1997; Carafoli *et al.*, 1997; Ogden, 1996; Pellegrini-Giampietro *et al.*, 1997; Rodriguez-Alvarez *et al.*, 1997; Simpson *et al.*, 1995; Tsumoto and Yasuda, 1996):

- 1. ligand-gated ion channels, in particular, the NMDA-receptor-gated ion channel (Kawamata and Omote, 1996; Ozawa, 1996);
- VDCCs, the operation of which is enhanced by the engagement of depolarizing AMPA and NMDA receptors, by increases in [Ca²⁺]_i, levels and by the actions of specific phosphorlyating PKs (Ben-Ari *et al.*, 1992; Olivera *et al.*, 1994);
 mobilization of Ca²⁺ from intracellular stores
- mobilization of Ca²⁺ from intracellular stores following activation of PLC by NK₁, MGlu₁ and MGlu₅ receptors; and
- 4. in a 'feed-forward' fashion, an influx of Ca^{2+} may further enhance $[Ca^{2+}]_i$ levels via the activation of Ca^{2+} -mobilizing ryanodine receptors on the endoplasmic reticulum (Dettbarn and Palade, 1997; Simpson *et al.*, 1995; Usachev and Thayer, 1997; Wang *et al.*, 1996b).

Recent studies have shown that specific pools of intracellular $[Ca^{2+}]_i$, for example, those which enter cells via VDCCs vs NMDA receptors, exert a contrasting influence upon modulation of neuronal activity. For example, in the generation of the cellular mediators, NO and PG (Sections 10.4.3.2 and 10.4.3.3), and in the modulation of gene expression (Carafoli *et al.*, 1997; Hughes and Dragunow, 1995; Lerea, 1997) (Section 10.4.2.5).

Unrestrained increases in $[Ca^{2+}]_i$ are potentially dangerous and may lead to excitotoxic degeneration (Choi and Rowbotham, 1997; Lerea, 1997) (Section 10.6.2). Studies in a variety of tissues suggest that homeostatic mechanisms may be concurrently engaged by increases in Ca^{2+} -influx and $[Ca^{2+}]_i$ levels which *counterregulate* the above processes. These include the activation of an APTase-dependent pump for the extrusion of $[Ca^{2+}]_i$, and the stimulation of several phosphatases leading to a reduction in neuronal excitability and Ca^{2+} -influx as outlined in Section 10.4.2.4 (Kyrozis *et al.*, 1996; Montieh and Roufogalis, 1995; Robello *et al.*, 1997; Young *et al.*, 1998).

In general, increases in $[Ca^{2+}]_i$ levels result in an enhancement of transmitter releases (Stanley, 1997). However, an increase in $[Ca^{2+}]_i$ levels in DH neurones may play a more specific role in their sensitization. Several of the mechanisms summarized below result in the induction of specific PKs, the significance of which for neuronal sensitization is discussed in Section 10.4.3.

10.4.2. Functional Effects of Increases in $[Ca^{2+}]_i$: Focus on Phenotype Changes in DH Neurones

10.4.2.1. Enhancement of the activity of PKA, PKC and other PKs

PLC-mediated increases in levels of DAG and [Ca²⁺]_i synergistically enhance the activity of PKC and increase its translocation to the cell membrane. Moreover, an increase in [Ca²⁺]_i levels may directly enhance the ability of PKC to phosphorylate NMDA receptors (Chen and Huang, 1992; Coderre et al., 1993; Rostas et al., 1996; Zheng et al., 1997). Elevations in $[Ca^{2+}]_i$ levels generally also increase the activity of NO synthase leading, via the production of NO, to the activation of PKG (Clementi and Meldolesi, 1997). Further, they favour the generation of cAMP by AC and the subsequent activation of PKA which, together with a calmodulindependent PK II, has been shown to phosphorylate AMPA receptors in certain populations of neurones (Barria et al., 1997; Ben-Ari et al., 1992; Lisman, 1994, 1997; Section 10.3.7). As concerns longer-term events, these and other Ca^{2+} -dependent PKs play crucial roles in modulating the activity of nuclear transcription factors, IEG expression and gene transcription (Bito et al., 1997; Carafoli et al., 1997; Deisseroth et al., 1998; Hughes and Dragunow, 1995; Lerea, 1997; Lisman et al., 1997; Stevens et al., 1994).

10.4.2.2. Induction of NO synthesis

There is a complex pattern of interplay between intracellular pools of [Ca²⁺]_i and NO (Clementi and Meldolesi, 1997; Horie et al., 1997a; Garthwaite and Boulton, 1995) and, via an action involving calmodulin, [Ca²⁺]_i increases the activity of neuronal NO synthase. Indeed, both peripheral tissue damage and PAF injury are associated with increases in DH levels of NO (Dickenson, 1997; Herdegen et al., 1993; Lin et al., 1997). The elevation in NO synthesis which follows NMDA receptor activation and an increase in $[Ca^{2+}]_i$ levels is of particular significance inasmuch as NO plays a broad role in NMDA receptor-associated processes of synaptic plasticity in other tissues (Zhang and Snyder, 1995). NO participates in the modulation of nociceptive transmission via several mechanisms outlined in Section 3.2.8.1 and, of special note, it increases intracellular levels of cGMP which, via the activation of PKG, plays an important role in neuronal sensitization (Section 10.4.3.3). Alterations in intracellular $[Ca^{2+}]_i$ levels may be spatially restricted. This compartmentalization reflects the differential origins of $[Ca^{2+}]_{i}$, for example via VDCCs or the activation of NMDA receptors (Allbritton et al., 1992; Carafoli et al., 1997; Deisseroth et al., 1998; Lerea, 1997). A recent study in striatal neurones indicated that the pool of $[Ca^{2+}]_i$ which activates NO synthase may be primarily derived from VDCCs (Rodriguez-Alvarez et al., 1997). This finding was unanticipated inasmuch as the activation of NO synthase triggered by activation NMDA receptors might reasonably be assumed to reflect an accumulation of $[Ca^{2+}]_i$ entering neurones via NMDA receptors themselves. It would be of interest to perform such studies in DH neurones. Apart from the modulation of nociception following injury to PAFs or to the spinal cord itself, NO may, possibly independently of cGMP, also modulate processes of cell survival and degeneration (Minghetti and Levi, 1998; Nishio and Watanabe, 1998) (Section 10.6.2).

10.4.2.3. Generation of PGs

 $[{\mbox{Ca}}^{2\,+}]_i$ increases arachidonic acid production by facilitating the activity of PLA2. There is evidence from the above-mentioned studies in striatal neurones that the major pool of $[Ca^{2+}]_i$ which activates PLA₂, in distinction to that inducing NO synthase, enters neurones via NMDA receptors themselves (Dumuis et al., 1988; Lerea et al., 1997; Rodriguez-Alvarez et al., 1997). The generation of arachidonic acid via PLA₂ and, subsequently, of PGs via COX, provides several potential mechanisms for an increase in cellular excitability. Notably, a facilitation in the activity of PKC (Section 10.4.3.1), an action expressed both in cells of origin and, following extracellular release and diffusion, in PAF terminals from which GLU and SP release may be enhanced (Section 3.2.8.4). In intrinsic DH neurons, gene expression may also be modified (Casabona, 1997; Collins and Davies, 1998; Lerea, 1997; Lerea et al., 1997; Nicholls and Attwell, 1990; Sánchez-Prieto et al., 1996; Yaksh and Malmberg, 1994).

10.4.2.4. Counter-regulatory processes: induction of phosphatases

In an opposite manner, however, $[Ca^{2+}]_i$ may activate several Ca²⁺- (and calmodulin)-dependent protein phosphatases, such as calcineurin, which has been detected in intrinsic neurones in superficial DH laminae (Strack et al., 1996; Wang and Kelly, 1997). The dephosporylation of DH-localized NMDA receptors reduces their functional activity (Kyrozis et al., 1996; Lieberman and Mody, 1994; Wang and Salter, 1994). In addition, the dephosphorylation of GABA_A receptors results in an *increase* in their activity (Robello et al., 1997). As mentioned above, the relationship between NO synthesis and [Ca² [⊤]]; is complex and, under certain conditions, an increase in [Ca²⁺] may, likewise via a mechanism involving activation of phosphatases, suppress NO synthesis, (Horie et al., 1997a). Thus, in addition to the possible activation of a Ca²⁺-extruding-pump (Montieh and Roufogalis, 1995; Young et al., 1998), increases in [Ca²⁺]_i may trigger several, phosphatase-mediated, mechanisms resulting in an inhibition of neuronal excitability and sensitization (Winder et al., 1998). These actions clearly counterbalance the phosphorylating, PK-mediated excitatory effects of increases in $[Ca^{2+}]_i$ levels (Section 10.4.2). The precise balance between such processes of phosphorylation/excitation and dephosphorylation/inhibition as a function of the levels of $[Ca^{2+}]_i$ and other factors remains to be established. A perturbation of this homeostatic regulation of $[Ca^{2+}]_i$ levels may well have grave consequences. Thus, an uninhibited increase in $[Ca^{2+}]_i$ levels in PNs or EXINs may result in an excessive output of nociceptive information to higher centres. Further, the unrestrained accumulation of [Ca2+]i in ININs, which are particularly vulnerable to excitotoxic damage, may lead

to their degeneration following PAF or spinal cord injury (Section 10.6). Ca^{2+} -dependent phosphatases likely play a more generalized and complex role in the modulation of neuronal excitability, receptor sensitivity, gene expression (in a fashion opposite to Ca^{2+} -dependent PKs) and other events underlying sensitization and control of neuronal function: such actions remains to be further explored in the DH (Bito *et al.*, 1997; Guerini, 1997; Sandkühler *et al.*, 1997; Winder *et al.*, 1998).

10.4.2.5. Modulation of IEG expression and gene transcription in DH neurones

Via multifarious mechanisms, including interactions with NO, PGs, cAMP and several PKs, ⁺]_i plays a key role in modifying the expression [Ca² of nuclear transcription factors controlling IEG and gene expression (Bito et al., 1997; Carafoli et al., 1997; Chapman and Besson, 1997; Chapman et al., 1995, 1996; Deisseroth et al., 1998; Eder, 1997; Hardingham et al., 1997; Hughes and Dragunow, 1995; Lerea, 1997; Lerea et al., 1997; Lin et al., 1994c; O'Neill and Kaltschmidt, 1997; Roche et al., 1996). The precise origin of various pools of ⁺]_i, whether derived from VDCCs or NMDA $[Ca^{2}]$ receptors, may be important in determining their influence upon gene expression. This distinction reflects not only their differential modulation of intracellular levels of NO and PG (Sections 10.4.2.2 and 10.4.2.3) but also their differential interaction with PKs such as calmodulin-dependent PK II. This results in a contrasting influence upon the activity of the nuclear transcription factor Ca²⁺/cAMP-response element binding protein (CREB) and, thereby, the expression of IEGs such as c-fos and, ultimately, synaptic plasticity. Intriguingly, it appears that $[Ca^{2+}]_i$ which enters neurones via Ltype Ca²⁺ channels plays a particularly significant and sustained role in triggering the phosphorylation of CREB and, thereby, modulating gene expression (Bito et al., 1997; Carafoli et al., 1997; Hughes and Dragunow, 1995; Lerea, 1997) A localized increase in $[Ca^{2+}]_i$ at the level of the cell membrane triggers the translocation of calmodulin into the nucleus: calmodulin then phosphorylates a specific PK IV which subsequently phosphorylates CREB (Deisseroth et al., 1998). In addition, PKA and several other PKs converge onto CREB to modify IEG expression and gene transcription (Bito et al., 1997).

The CREB-activated IEG, c-fos, has been extensively exploited as a marker of neuronal activation in the DH-although DH expression of c-fos does not invariably correlate with levels of nociception (Chapman and Besson, 1997; Millan, 1993; Ossipov et al., 1997; Presley et al., 1990; Todd et al., 1994a). The widespread appearance of c-fos in the DH in response to peripheral nociceptive input is consistent with the notion that cellular mechanisms underlying sensitization are not confined to WDR neurons in deeper laminae. In this respect, much attention has been directed toward the apparent induction by cfos of gene expression for the endogenous opioids, enkephalin and DYN, under conditions of acute and long-term noxious stimulation and PAF damage (Chapman and Besson, 1997; Clayton et al.,

1997; Clément et al., 1996; Coderre, 1992; Dubner and Ruda, 1992; Hunt et al., 1987; Hunter et al., 1995; Kawakami et al., 1994; Millan, 1990, 1993; Naranjo et al., 1991; Ossipov et al., 1997). A c-fos complex and a further IEG upregulated by noxious stimulation, c-jun, may be involved in the alteration of the gene transcription of opioids peptides and other proteins (Wisden et al., 1990). Although a subpopulation of PNs in superficial DH laminae contains DYN (Section 6.1.1), both DYN and other opioid peptides inhibitory to nociceptive transmission in the DH are preferentially localized (with c-fos) in ININs rather than WDR PN in deep laminae (Hunter et al., 1995; Millan, 1993; Todd et al., 1994a) (Sections 4.6 and 10.6). Thus, the question remains as to which putative transmitters are upregulated in parallel with, and possibly by, c-fos in sensitized WDR PNs. Presumably, they play a role in the ascending transmission of nociceptive information. Their characterization would be of particular interest inasmuch as a recent study reported that neutralization of c-fos expression by spinal antisense treatment decreases inflammatory nociception (Hou et al., 1997). However, under other conditions, a similar approach has been shown, to enhance nociception, possibly via an interference with the induction of DYN (Ossipov et al., 1997-though see Hunter et al., 1995). The reasons underlying these contrasting results are unclear.

As concerns changes in the phenotype of DHlocalized neurones, one could imagine that an increased expression of pronociceptive transmitters in EXINs might aggravate painful states and several lines of evidence might be advanced to support this possibility:

- 1. An increase in the synthesis and release of CCK from EXINs may be involved in the reduced antinociceptive efficacy of opioids following injury to the spinal cord or damage to PAFs (Ossipov *et al.*, 1997; Xu and Wiesenfeld-Hallin, 1997).
- 2. An increase in DH levels of the 'anti-opioid' neuropeptide FF under conditions of peripheral inflammation was reported by Kontinen *et al.* (1997). However, under these conditions, Pertovaara *et al.* (1998) have suggested that neuropeptide, this may actually fulfil an *antinociceptive* role.
- 3. Inflammatory tissue damage and PAF injury is, as mentioned above, accompanied by an increase in DH levels of DYN.

Certain authors have suggested that high concentrations of DYN *enhance* nociception by antagonist actions at NMDA receptors. However, the principal, physiological role of DYN is the inhibition of nociceptive transmission via actions at κ -opioid receptors in the DH (Millan, 1990, 1993; Ossipov *et al.*, 1997) (Section 10.6). Further, DH levels of ININ-localized enkephalins are also increased under conditions of PAF inflammation (Dubner and Ruda, 1992; Millan, 1993; Ossipov *et al.*, 1997) and there is additional evidence that phenotype changes in DH-localized ININs may counter nociception. Thus, inflammation also elicits an increase in the synthesis of GABA in the DH, while DH levels of GAL and NPY, which are colocalized with GABA and which may also inhibit nociception, are likewise elevated (Mantyh *et al.*, 1994; Munglani *et al.*, 1996; Xu and Wiesenfeld-Hallin, 1997) (Section 10.6). Further, Stanfa *et al.* (1994) suggested that a *reduction* in CCK generation by EXINs may account for the potentiation of opioidergic antinociception in inflammation. The relationship of such changes to alterations in c-fos, other regulators of gene expression and $[Ca^{2+}]_i$ levels remains to be clarified.

To summarize, the induction of c-fos in PNs and EXINs may, in theory, trigger events leading to the upregulation of pronociceptive transmitters in the DH. However, there is, to date, *little* evidence that neuronal phenotype changes in the DH contribute to the induction of inflammatory, painful states, in which GABAergic ININ activity is enhanced (Section 10.6.2.2) and the activity of CCK (probably) decreased. On the other hand, an increase and decrease in DH levels of CCK and GABA, respectively, may be involved in the induction of neuropathic, painful states due to PAF or spinal injury. Further, the increase in the DH density and Ca²⁺ permeability of AMPA receptors should also be mentioned in this regard (Section 10.3.2.1), as well as the elevation in the DH density of NK1 and CGRP receptors (Abbadie et al., 1996; Coggeshall and Carlton, 1997).

A further exploration of changes in the phenotype of various classes of DH neurones in relationship to alterations in levels of $[Ca^{2+}]_i$, IEGs and other intracellular signals controlling gene expression would be of considerable interest. Along these lines, it has recently been speculated that a reduction in activity at tonically-active, antinociceptive cannaboid receptors in the DH may be involved in the induction of certain painful states (Richardson *et al.*, 1998).

10.4.3. The Roles of Specific PKs

10.4.3.1. Actions of PKC

As indicated above, one particularly important action of $[Ca^{2+}]_i$ in nocisponsive DH neurones is (together with DAG) the induction of PKC (Newton, 1997; Ramakers *et al.*, 1997). The following observations support the concept that PKC plays a pivotal role in processes underlying neuronal sensitization and an enhancement of nociception.

- 1. Stimulation and inhibition of spinal pools of PKC respectively enhances and reduces the inflammatory hyperalgesia provoked by intraplantar injection of formalin (Coderre, 1992; Coderre and Yashpal, 1994; Munro *et al.*, 1994).
- PKC inhibitors attenuate the capsaicin-induced sensitization of WDR STT cells to mechanical stimuli (Lin *et al.*, 1996a,b; Sluka *et al.*, 1997a,b). In addition, the PKC inhibitor, staurosporin, blocks the facilitatory influence of NK₁ receptors upon NMDA receptor-mediated currents in DH neurones whereas phorbol esters, which enhance PKC activity, mimic the facilitatory influence of PKC on NMDA receptor-mediated ion currents (Gerber *et al.*, 1989; Rusin *et al.*, 1992; Urban *et al.*, 1994a,b).
- Activation of PKC excites and sensitizes WDR PNs of the STT (Lin *et al.*, 1996a; Palecek *et al.*, 1994; Sluka and Willis, 1997; Sluka *et al.*, 1997b)

and inhibits opioidergic mechanisms of antinociception (Zhang et al., 1990).

- Levels of membrane-bound (translocated) PKC are increased in superficial and deep DH laminae during inflammatory hyperalgesia (Mao *et al.*, 1992d; Yashpal *et al.*, 1995).
- 5. Gangliosides may reduce pain due to PAF injury by interfering with the translocation of PKC (Hayes *et al.*, 1992; Mao *et al.*, 1992a; Vaccarino *et al.*, 1987).

Together with Ca²⁺, a key role has been established for PKC in modulating the activity of c-fos and other mechanisms controlling gene expression: such actions are of importance to long-term increases in DH neuronal excitability (Hughes and Dragunow, 1995). Further, over a more rapid timescale, PKC is strongly implicated in NMDA-receptor dependent mechanisms of synaptic plasticity underlying memory formation (LTP). Indeed, the most important sensitizing action of PKC in WDR neurones is undoubtably the phosphorylation of NMDA receptors (Rostas et al., 1996; Lisman, 1994; Zheng et al., 1997). This leads to a relief of Mg^{2+} -block, thereby permitting their operation at lower potentials and enhancing passage of current. Although the underlying molecular mechanisms remain unclear, PKC likely fulfils a more general contribution to the control of ionic currents and intracellular Ca2+-homeostasis. For example, PKC can also increase cellular excitability via a prolongation of the Ca²⁺-dependent inactivation of K⁺-currents or via a reduction in the inactivation of Na⁺-currents, events which may occur in DH neurones (Alkon et al., 1986; Colbert and Johnston, 1998; De Riemer et al., 1985; Madison et al., 1986; Pedrosa-Ribeiro and Putney, 1996). A further mechanism underlying the sensitization of WDR PNs by PKC may-in analogy to PKG and PKA (Section 10.4.3)-be a decrease in the efficacy of ININ transmission effected via the phosphorylation of inhibitory GABAergic receptors (Leidenheimer et al., 1992; Lin et al., 1996a,b,c).

On the other hand, certain actions of PKC may decrease neuronal activity. Thus, a PKC-mediated phosphorylation of group I mGlu and NK₁ receptors, thereby enhancing their desensitization, might counter excitatory actions at DH neurons (Gereau and Heinemann, 1998; Quartara and Maggi, 1997). Further, the induction of PKC by 5-HT₂ agonists leads to an *increase* in currents at *inhibitory* glycinergic receptors targeting WDR PNs. Via this mechanism, PKC may also *reduce* nociception (Xu *et al.*, 1996a). Finally, as mentioned above, increases in $[Ca^{2+}]_i$ levels may enhance the activity of a compensatory ATPase-coupled pump for Ca^{2+} -extrusion and PKC appears to be involved in this effect (Montieh and Roufogalis, 1995; Young *et al.*, 1998).

Many isoforms and cellular pools of PKC have been identified (Casabona, 1997; Newton, 1997; Ramakers *et al.*, 1997) and, as implied by the preceding comments, they may play differential roles in the modulation of nociceptive transmission. Interestingly, a Ca²⁺-*independent* isoform of PKC appears to be involved in the above-mentioned facilitation of glycine currents (Xu *et al.*, 1996a). This is in contrast to the Ca²⁺-dependent isoform of PKC which phosphorylates NMDA receptors although its precise identity remains to be defined (Nagashima *et al.*, 1991). Recently, a PKC γ isoform was detected in lamina II-localized ININs receiving input from C fibres expressing P_{2X3} receptors (Section 7.4.4). Transgenic mice lacking this PKC γ isoform failed to develop alterations in nociception following PAF injury suggesting that it plays an important role in neuropathic pain (Malmberg *et al.*, 1997b). A characterization of the roles of various DH-localized pools and isoforms of PKC remains a challenging task for the future.

10.4.3.2. Actions of cAMP and PKA

There is evidence that PKA plays a role in the sensitization of DH neurones. Increases in $[Ca^{2+}]_i$ favour the activity of AC, and the positive coupling of CGRP, other neuropeptides and A_{2A} receptors to AC suggests that, via the induction of cAMP, they can induce PKA (Section 3.2). In addition, descending pathways may be involved in the activation of PKA: for example, dopamine D₁ receptors and several subtypes of serotoninergic receptor (5-HT₄, 5-HT₆ and 5-HT₇) all generate cAMP via the stimulation of AC (Boess and Martin, 1994; Millan, 1997).

Consistent with a role of PKA in enhancing nociception and sensitization, selective inhibition of PKA attenuates the induction of mechanical allodynia by cutaneous capsaicin (Sluka and Willis, 1997). Further, the spinal administration of cAMP, stabilized cAMP analogues or the catalytic subunit of PKA potentiates the excitatory influence of EAAs upon DH neurones, and diminishes opioidergic mechanisms of antinociception (Cerne *et al.*, 1992; Malmberg *et al.*, 1997a).

Although the above observations would be consistent with an enhancement of NMDA receptormediated currents, it is unclear how this could be caused by cAMP/PKA. Indeed, the mechanistic bases of their facilitatory influence upon DH neurones is currently under exploration. However, the phosphorylation by PKA (in analogy to PKC) of inhibitory GABAergic receptors would lead to an increase in neuronal sensitivity (Robello et al., 1993, 1997; Sluka and Willis, 1997; Zarri et al., 1994). There is also evidence from studies in other tissues that PKA (together with a calmodulin-dependent K II) may enhance excitability by the phosphorylation of AMPA receptors, and the possible phosphorylation of certain VDCCs should also be considered (Ben-Ari et al., 1992; Blackstone et al., 1994; Barria et al., 1997). PKA may also enhance the activity of COX 2 in certain cell types (Minghetti and Levi, 1998). Finally, via phosphorylation of the nuclear regulatory element, 'CREB' (Section 10.4.2.5), PKA, together with calmodulin, plays an important role in LTP and other longer-term plastic events potentiating neuronal function in cerebral structures and, likely, the DH (Ardenghi et al., 1997; Carafoli et al., 1997; Deisseroth et al., 1998; Hughes and Dragunow, 1995: Yin and Tully, 1996).

10.4.3.3. Actions of cGMP and PKG

A NO-triggered increase in cGMP levels in WDR neurones contributes to mechanisms underlying sensitization by the induction of PKG, which subsequently phosphorylates inhibitory GABAergic and glycinergic receptors (Leidenheimer, 1996; Lin et al., 1996a,b,c, 1997; Meller et al., 1992a,b,c; Robello et al., 1996, 1997; Sluka and Willis, 1997). As indicated in Sections 10.4.3.1 and 10.4.3.2, the deactivation of these inhibitory receptors by their phosphorylation is a property shared with PKC and PKA. Indeed, it has been shown that PKG phosphorylates GABAA receptors in other tissues (cortex and cerebellum) at a consensus site shared with PKA (Leidenheimer, 1996; Robello et al., 1996, 1997; Sluka and Willis, 1997; Zarri et al., 1994). The inhibition of GABAergic transmission by PKG and other PKs exacerbates processes of sensitization triggered by nociceptive PAF input to WDR DH neurones. Further, a component of descending inhibition exerted via the recruitment of GABAergic ININs in the DH is compromised by PKG-mediated phosphorylation of GABAergic receptors (Lin et al., 1997: Peng et al., 1996). This leads to a shift in the overall pattern of descending modulation towards facilitation (Section 10.8). Two additional actions of PKG and cGMP which result in elevations of $[Ca^{2+}]_i$ may also contribute to the excitation of DH neurones. Thus, PKG can modify the operation of VDCCs (Van Coppenolle et al., 1997), while observations in DRG neurones suggest that cGMP can mobilize [Ca²⁺]_i from ryanodine-sensitive intracellular stores, thereby further accentuating processes underlying sensitization (Crawford et al., 1997).

In line with a putative role in enhancing nociception, levels of cGMP are increased in the DH under conditions of peripheral inflammation (Garry *et al.*, 1994b; Igwe and Ning, 1994). Further, the spinal administration of cGMP analogues elicits hyperalgesia (Garry *et al.*, 1994a) and potentiates the responsiveness of deep laminae WDR neurones to mechanical stimulation (Lin *et al.*, 1997). Contrariwise, blockade of the activity of cGMP prevents the sensitization of DH neurones by capsaicin (Lin *et al.*, 1997).

10.4.3.4. Key roles of $[Ca^{2+}]_i$ and PKs in the sensitization of DH neurones: a summary

To summarize (see Fig. 5), increases in the levels of $[Ca^{2+}]_{i}$, and in the activity of PKC PKA, PKG and, probably, calmodulin-dependent PKs—likely represent key, intracellular substrates underlying the rapid and sustained sensitization of WDR neurones. Nevertheless, their roles, together with those of other intracellular signalling systems, in the control of neuronal excitability in the DH and the induction of painful states require considerable further clarification. In this respect, it is unlikely that mechanisms of sensitization are restricted to WDR PNs in deeper laminae. The significance of alterations in the levels of $[Ca^{2+}]_i$ and the induction of various PKs in other classes of neurones, in particular ININs and superficial NS neurones, would be of importance to establish.

10.5. A β Fibre-Mediated Sensitization of DH Neurones in Inflammatory States

Inasmuch as $A\beta$ and C fibre input excites a common population of WDR neurones in the deep DH, the possibility that activation of $A\beta$ receptors can likewise elicit processes of sensitization must be considered. However, $A\beta$ fibre stimulation is *not* normally perceived as noxious (Sections 3.1.1 and 4.6.2) (Fig. 2) and, under *normal* circumstances, the following arguments indicate that activation of $A\beta$ fibres is unlikely to sensitize DH neurones.

- 1. The stimulation of $A\beta$ fibres activates AMPA channels of comparatively low Ca2+-yet high Na+-, permeability and possessing very rapid, transient kinetics (Dickenson, 1997; Fletcher and Lodge, 1996; Furuyama et al., 1993: Pellegrini-Giampietro et al., 1997; Tomiyama et al., 1996) (Section 3.2.3). As such, even though the depolarization elicited by engagement of AMPA receptors activates VDCCs, there is no sustained or marked increase in $[Ca^{2+}]_i$ levels in DH neurones and any minor increases in $[Ca^{2+}]_i$ may be confined to a discrete region of the cytosol (Berninger and Poo, 1996; Kharazia and Weinberg, 1997; Lerea et al., 1997; Morris, 1997; Ottersen and Landsend, 1997; Sakai et al., 1997; Simpson et al., 1995). Further, in the absence of a pronounced increase in $[Ca^{2+}]_i$, levels the activation (phosphorylation) of AMPA receptors by calmodulin-dependent PK II and PKA does not occur (Sections 3.2.3 and 10.3.2.1).
- 2. Upon the *selective* activation of $A\beta$ fibres, C fibre-coupled NMDA and NK_{1/2} receptors are quiescent. Thus, there is no subsequent increase in Ca²⁺-influx, [Ca²⁺]_i levels or the activity of PKs. [In this respect, the differential localization of AMPA and NMDA receptors in specific populations of ININ in the superficial DH should also be pointed out (Coggeshall and Carlton, 1997).]
- 3. Since $A\beta$ fibres excite ININs, their cumulative mono- and *poly*-synaptic impact upon WDR neurones is less pronounced than that of C fibres which excite WDR neurons both directly and indirectly via EXINs (Fig. 2). Thus, $A\beta$ fibre stimulation generally elicits only rapid EPSPs and localized, minor increases in neuronal [Ca²⁺]_i, which do not, under normal conditions, lead to neuronal sensitization.

Nevertheless, the Ca²⁺-permeability and density of DH-localized AMPA receptors is enhanced by PAF injury, and NMDA receptor activation may, at least in certain neurones, activate AMPA receptors via a calmodulin-dependent PK II (Section 10.3.2.1). Indeed, Herrero *et al.* (1995) recently suggested that stimulation of $A\beta$ fibres *can* elicit wind-up under conditions of inflammation. In a further study, Ma and Woolf (1996b) also raised the possibility that $A\beta$ fibre stimulation may, likewise under conditions of inflammation, acquire the capacity to provoke central sensitization. This phenomenon, termed 'progressive tactile hypersensitivity' refers to a *slowly*-developing (hours), $A\beta$ fibre-mediated allodynia to non-noxious mechanical stimulation. Notably, as compared to *rapid* (minutes) C fibre-mediated mechanisms of wind-up, this time-scale is similar to that seen for the appearance of inflammatory allodynia in man following tissue injury. Indeed, these findings raise the possibility of temporally-different patterns of hypersensitivity involving both C *and* A β PAFs under conditions of inflammation.

The roles of ionic and soluble transduction mechanisms in $A\beta$ fibre-mediated sensitization of DH neurones remain to be further explored (Ma and Woolf, 1997). However, apart from AMPA receptors, a possible implication of SP and NK₁ receptors would be of particular interest in the light of

- 1. the recently-described phenotypic switch whereby inflammation leads to the synthesis of SP in $A\beta$ fibres (Neumann *et al.*, 1996) (Section 8.2.5);
- 2. the upregulation of NK_1 receptors in the DH seen following peripheral inflammation and PAF injury (Abbadie *et al.*, 1996; Goff *et al.*, 1998); and
- 3. the migration, following PAF injury, of A β fibres from deeper laminae receiving principally innocuous, sensory input into more superficial lamine containing NS and WDR neurones and receiving nociceptive information from C fibres (Woolf *et al.*, 1992) (Section 11.3.2) (Fig. 2).

Indeed, NK₁ receptor antagonists can prevent the emergence of progressive, mechanical hypersensitivity under conditions of inflammation (Ma and Woolf, 1997). This observation is consistent with a role of SP released from $A\beta$ fibres in its mediation. However, a C fibre-mediated induction of sensitization in DH neurones may be a pre-condition for A β fibre-mediated processes of sensitization to occur. Thus, it cannot be excluded that the reduction by NK1 antagonists of tactile hypersensitivity reflects the blockade of C fibre-coupled NK1 receptors involved in the induction and maintenance of neuronal sensitization in the DH (Section 9.1.3). Moreover, as discussed in Section 9.2.3. Cervero and Laird (1996b) suggested that $A\beta$ fibre-mediated allodynia may reflect an indirect ININ-mediated engagement of C fibre terminals in the DH. This mechanism would also be consistent with a blockade of tactile hypersensitivity by NK1 receptor antagonists.

10.6. Reduction in the Functional Activity of ININs

10.6.1. Inhibition of ININ Activity Mimics and Enhances Sensitization in the DH

In Section 10.7, evidence was presented that an induction of PKA, PKC and/or PKG in deep laminae WDR PNs contributes to their sensitization by phosphorylating inhibitory GABAergic and/or glycinergic receptors. Such mechanisms are likely triggered both by the prolonged stimulation of nociceptors upon inflammatory tissue damage and, although direct evidence is lacking, by PAF injury. The attenuation of ININ tone at WDR neurones via such *intrinsic* intracellular mechanisms enhances their excitability, and contributes to processes underlying sensitization, hyperalgesia and allodynia (Lin et al., 1996a,c, 1997; Sluka and Willis, 1997).

The possibility that other mechanisms may more directly lead to a reduction in the activity of GABAergic and glycinergic ININs in the DH has attracted much attention, in particular as concerns the effects of PAF injury. This interest is underpinned by observations that pharmacological interruption of glycinergic and GABAergic transmission in the DH produces a long-lasting increase in the responsiveness of DH neurones to low intensity, $A\beta$ fibre stimulation which resemble hyperexcitabilitybased, allodynic states (Sivilotti and Woolf, 1994). A loss of GABAergic tone in the DH following PAF injury may also facilitate the ability of $A\beta$ fibre input to sensitize DH neurones (Section 10.5). Further, behavioural evidence that a loss of functional ININ activity alters nociceptive processing in the DH is provided by studies of the spinal administration of strychnine and bicuculline, which block glycinergic and GABA_B receptors, respectively. The resultant touch-evoked allodynia and increased flexor withdrawal is comparable to that seen upon induction of central sensitization by nerve damage: further, this allodynia-like response to mechanical $A\beta$ fibre input is similarly prevented by NMDA and AMPA antagonists (Hao et al., 1992; Sorkin and Puig, 1996; Wall and Bennett, 1994; Woolf and Doubell, 1994; Xu et al., 1993a; Yaksh, 1989). Strychnine and bicuculline also potentiate the hyperalgesia/allodynia associated with nerve injury, actions likewise prevented by NMDA receptor antagonists (Milne et al., 1996; Satoh and Omote, 1996; Seltzer et al., 1991; Yamamoto and Yaksh, 1993). Conversely, the GABA_A and GABA_B agonists, baclofen and muscimol, respectively, antagonize the allodynia provoked by PAF injury (Hwang and Yaksh, 1997).

As described in Section 13.2, similar changes in nociception due to a loss of GABAergic ININ activity have been reported following focal, spinal cord ischaemia (Hao et al., 1992). Further, the spinal administration of 5-HT_{1A} receptor agonists elicits a behavioural state mimicking mechanical allodynia, apparently via (the hyperpolarization of) GABAergic (or other types) of ININ in the DH. The induction of such an allodynia-like state by 5-HT_{1A} agonists is prevented both by GABAergic agonists, which restore ININ activity, as well as by antagonists at NMDA receptors (Millan et al., 1996; Yang et al., 1998). AMPA receptor antagonists are also effective, presumably via blockade of actions mediated by $A\beta$ fibres transmitting normally nonnoxious, mechanical input (Millan et al., 1996). Interestingly, further, spinal administration of nociceptin, the heptadecapeptide ligand of a recentlycloned 'opioid receptor homologue', elicits allodynia via inhibition of glycinergic ININs (Hara et al., 1997; Meunier, 1997).

Pharmacological blockade of activity at inhibitory GABAergic and (glycinergic) receptors elicits rhythmic, burst-like firing in neurones in the VH and other tissues (Forti *et al.*, 1997). Burst activity increases the efficacy of synaptic transmission and enhances transmitter release, thereby improving synaptic signalling and information transfer (Lejeune *et al.*, 1997; Lisman, 1997; Woolf and King, 1989). Further, burst-like activity mediated by NMDA receptors has been specifically implicated in LTP, a process of synaptic strengthening underlying hippocampal memory consolidation, and which is analogous to mechanisms involved in the sensitization of DH neurones. Notably, both activation of NMDA receptors *and* inhibition of GABAergic receptors appears necessary for optimal generation of LTP (Huerta and Lisman, 1995) (Section 10.2.1).

A reduction in ININ tone in the DH may, then, together with an increase in C fibre input, synergistically:

- 1. increase the spontaneous activity and excitability of WDR PNs;
- 2. perturb and extend their RFs; and
- trigger synchronized and oscillating circuits reflecting a generalized increase in neuronal excitability, and resulting in a disruption of nociceptive processing in extensive regions of the DH (Biella *et al.*, 1997; Bracci *et al.*, 1996a,b; Brewer and Ray, 1995; Castro-Alamancos and Connors, 1997; Dubner, 1992; Eblen-Zajjur and Sandkühler, 1997; Forti *et al.*, 1997; Kaneko and Hammond, 1997; Khandwala *et al.*, 1997; Lin *et al.*, 1996a,b,c) (Section 9.1.4.2).

It is likely that similar phenonema contribute to the sensitization of nocisponsive neurones at the thalamic and cortical level (Sections 12.3.2 and 13.3).

To summarize, an increase in the excitability of WDR PN neurones in the DH can be directly initiated or aggravated by a suppression in the activity of GABAergic and/or glycinergic ININs. In addition, ININs reduce release from the central terminals of nocisponsive C fibres and mechanosensitive $A\beta$ fibres involved in mediating mechanical allodynia. The loss of this inhibitory influence upon PAF input into the DH further amplifies processes eliciting hyperexcitability in WDR PNs (Malcangio and Bowery, 1996b; Sorkin and Puig, 1996; Todd and Spike, 1993; Todd *et al.*, 1996).

10.6.2. Mechanisms Underlying a Reduction of ININ Activity

10.6.2.1. Evidence for a reduction in GABAergic ININ activity upon PAF injury

The above observations suggest that a (pharmacological) reduction in the activity of GABAergic or glycinergic ININs mimics and/or exacerbates central sensitization and the accompanying hyperalgesia/ allodynia. The question arises, then, as to whether a *pathophysiological* loss of ININ tone in the DH may contribute to the *induction* of painful states.

As concerns PAF injury, there is evidence for a functional reduction in the GABAergic inhibition of DH-localized C fibre terminals (Bhisitkul *et al.*, 1990). Indeed, the slow depolarizing waves (DRPs) elicited in PAFs by ININs seem to be reduced in the first few weeks following nerve injury, suggestive that the functional influence of ININs upon central PAF terminals is reduced (Laird and Bennett, 1992; Wall and Devor, 1981)

(Section 9.1.4.2). These findings may be related to a decrease in the density of $GABA_B$ receptors documented in the DRG ipsalateral to PAF injury (Castro-Lopes *et al.*, 1995; Oyelese *et al.*, 1997; Towers *et al.*, 1997).

Much interest has been focused on the question of whether PAF injury precipitates a reduction in the number of GABAergic or other types of ININ in the DH (Castro-Lopes et al., 1993; Dumoulin et al., 1996; Gu et al., 1997; Hammond, 1997; Kelly et al., 1997; Oliveira et al., 1997). Indeed, based upon the appearance of 'dark' neurones in the DH, it has been concluded that neurectomy can provoke the transynaptic degeneration of intrinsic ININs, primarily in laminae I and II (Ibuki et al., 1997; Mao et al., 1997; Nachemson and Bennett, 1993; Sugimoto et al., 1989) (Fig. 2). This apparent loss is, further, potentiated by administration of strychnine or bicuculline, possibly due to disinhibition of reverberating, excitatory circuits (Bracci et al., 1996a and b; Eblen-Zajjur and Sandkühler, 1997) (Sugimoto et al., 1989) (Sections 10.6.1, 12.3.2 and 13.3). The supersensitivity displayed by mononeuropathic rats to the analgesic actions of baclofen (Abrams, 1996; Anghinah et al., 1994; Smith et al., 1994a) may also be taken as evidence for a degeneration-or reduced activity-of GABAergic neurones, since this would lead to supersensitive, postsynaptic GABAergic receptors in the DH. Indeed, PAF axotomy increases the density of GABAA receptors in the superficial DH (Castro-Lopes et al., 1995). Further, in analogy to the burst firing which is induced in the DH by a suppression of ININ activity (Section 10.6.1), rhythmic discharges of DH neurones have been reported in the DH upon peripheral nerve damage, consistent with a loss of ININ control (Lombard and Larabi, 1983).

Pretreatment with NMDA receptor antagonists can prevent certain of the neurochemical and behavioural changes, such as DH sensitization and allodynia, which accompany PAF injury (Carlton et al., 1997; Kim et al., 1997b; Satoh and Omote, 1996; Smith et al., 1994a; Woolf and Chong, 1993; Yoshimura and Nishi, 1995). Further, upon PAF injury, the activation of NMDA receptors by C fibres triggers the sensitization of WDR neurones in the DH (Section 10.3). On this basis, it has been contended that PAF injury elicits a transient but massive release of EAAs at NMDA receptors onto small, vulnerable *ININs*. This may lead to an excessive accumulation of $[Ca^{2+}]_i$, and the induction of NO and other mechanisms provoking their excitotoxic degeneration (Choi and Rothman, 1990; Gu et al., 1997; Lerea, 1997; Mao et al., 1997; Melino and Bernassola, 1997; Minghetti and Levi, 1998; Nishio and Watanabe, 1998). In certain studies, a modest decrease in gene expression for NMDA receptors has been detected in the DH following PAF injury (Coggeshall and Carlton, 1997; Hama et al., 1995). This may reflect a loss of NMDA sites on degenerated ININs, although it cannot be discounted that a functional down-regulation of NMDA receptors by a transient enhancement in input from damaged PAFs is involved.

AMPA receptors have also been implicated in degenerative processes in CNS neurones (Pellegrini-Giampietro et al., 1997) and other processes may be involved in the loss-or rescue-of DH-localized ININs following PAF injury. For example, it is possible that damage to PAFs leads to an atrophy of ININs by deprivation of neurotrophic factors normally delivered from their central terminals (Oliveira et al., 1997) (Section 14.5.3). Interestingly, it was recently suggested that PGs (PGE₂), despite their pronociceptive roles in the DH (Section 3.2.8.2), may play a role in *diminishing* (apoptotic) cell loss in the superficial DH following peripheral nerve injury, possibly via the activation of a mitogen-activated PK (Kawamura et al., 1997a,b). Further, it has been shown that PGE₂ can protect neurones against NMDA receptor (and NO)mediated toxicity (Akaike et al., 1994). Evidently, then, the mechanisms involved in the putative lossand protection-of DH-localized ININs following PAF injury require further elucidation.

10.6.2.2. Temporal evolution of changes in ININ activity: lack of a decrease upon peripheral inflammation

Based on the above findings concerning PAF damage, Dubner and Ruda (1992) speculated that, under conditions of prolonged, inflammatory pain, small GABAergic ININs may degenerate due to an excessive, excitatory, NMDA receptor-mediated C fibre input (Mattson and Mark, 1996; Solum *et al.*, 1997; Zhang *et al.*, 1998a). Further, they proposed that GABAergic ININs inhibit the activity of dynorphinergic '*EXINs*'. A disinhibition of the latter would compound sensitization by:

- 1. further exciting sensitized WDR PNs; and
- 2. enhancing release from terminals of C fibres targeting WDR neurones.

In line with this hypothesis, DYN gene expression is enhanced in the DH under conditions of PAF damage and peripheral inflammation (Draisci et al., 1991; Hunter et al., 1995; Kajander et al., 1990; Millan, 1993; Ossipov et al., 1997). Further, DYN has been shown to enhance C fibre reflexes and it may also enhance spinal cord levels of SP and EAAs, presumably derived from PAF term-inals (Ossipov et al., 1997; Suarez-Roca and Maixner, 1993). The intrathecal injection of DYN was, in addition, reported to elicit allodynia, while spinal administration of DYN and antibodies to DYN respectively inhibited and potentiated opioidergic analgesia following PAF injury. These actions of DYN are likely exercised via a facilitatory interaction at the modulatory glycine site of NMDA receptors (Brauneis et al., 1996; Ho et al., 1997; Jarvis et al., 1997; Laughlin et al., 1997; Ossipov et al., 1997; Vanderah et al., 1996b; Zhang et al., 1997c). Despite the potential interest of this hypothesis, however, there remain several questions:

1. While exogenous administration of *high* concentrations of DYN may, indeed, exert excitatory and pronoceptive actions via an interaction with NMDA receptors, the principal, physiological κ -opioid receptor-mediated role of endogenous pools of DYN in the DH is *antinociceptive*. Further, an action of DYN at DH-localized κ -receptors *counters* rather than exacerbates inflammatory pain, while administration of DYN-specific antisera to neutralize the actions of DYN exerts little or *no* anti-allodynic actions against neuropathic pain (Hunter *et al.*, 1995; Mackawa *et al.*, 1995; Millan, 1990, 1993; Nichols *et al.*, 1997; Ossipov *et al.*, 1996, 1997; Schaible *et al.*, 1992; Riley *et al.*, 1996; Stiller *et al.*, 1993).

- 2. Acute and chronic noxious stimulation, including inflammation, actually *increases* the number of GABAergic ININs, and levels of GABA, in the DH at times when sensitization and hyperalgesia are *already* apparent: synthesis of the co-localized, antinociceptive neuropeptides, GAL and NPY, is increased in parallel (Castro-Lopes *et al.*, 1992, 1994; Hylden *et al.*, 1991; Ji *et al.*, 1994, 1995; Malcangio and Bowery, 1996b; Mantyh *et al.*, 1994; Nahin and Hylden, 1991).
- 3. Pharmacological blockade of activity at $GABA_B$ receptors *potentiates* inflammatory hyperalgesia (Malcangio and Bowery, 1994, 1996b).
- 4. The density of GABA_B receptors in the DH is slightly reduced by inflammation and the actions of GABA_B agonists are attenuated (Castro-Lopes *et al.*, 1995; Coggeshall and Carlton, 1998; Malcangio and Bowery, 1994, 1996b). This may reflect a down-regulation of GABA_B receptors by an *increase* in the release of GABA.

The above observations suggest that an *enhancement* in GABAergic tone under inflammatory conditions may *counter* increases in nociception. Evidence that GABAergic ININ activity can be decreased by inflammatory pain is, thus, *lacking*.

Moreover, a recent study of PAF injury similarly observed an *increase* in DH levels of both glycine and GABA, and blockade of transmission at these sites potentiated thermal hyperalgesia (Satoh and Omote, 1996). These changes were preventable by NMDA antagonists in each case. Consequently, several further questions must also be raised which likewise concern a putative role of ININ loss following PAF damage:

- 1. In line with the above observations, bicuculline enhances the thermal hyperalgesia elicited by PAF injury, suggesting that GABAergic ININs *are* functionally active and that—like peripheral inflammation—they act to moderate nociception (Yamamoto and Yaksh, 1993).
- Transynaptic 'degeneration' and other neurochemical changes seen in the DH, although predominantly ipsilateral, are also seen *bilaterally* even where changes in nociception are unilateral (Ibuki *et al.*, 1997; Mao *et al.*, 1997; Satoh and Omote, 1996; Sugimoto *et al.*, 1990) (Section 9.1.4.2).
- 3. The long-term *recovery* of GABAergic profiles in the DH following their reduction by PAF injury—or damage to the DH itself—suggests that they do *not* actually degenerate, but rather

decrease their activity (Hao et al., 1992; Ibuki et al., 1997).

It cannot be entirely excluded that a compensatory increase in GABA synthesis in residual, surviving GABAergic ININs underlies the recovery of GABA levels. However, this seem unlikely and evidence that (intact) GABAergic neurones in superficial lamina *transiently* display a *reduction* in GABA synthesis following PAF damage was recently presented (Dumoulin *et al.*, 1996).

Thus, although a proportion of ININs may be eliminated following PAF injury by an NMDA receptor-mediated excitotoxic C fibre input, it is unlikely that there is a generalized degeneration of ININs in the DH (Mattson and Mark, 1996). In addition, there is no evidence for a reduction in the activity of GABAergic ININs in the DH under conditions of peripheral inflammation. An alternative interpretation of these findings, which does not evoke NMDA receptor-mediated excitotoxic damage, is that the activity of ININs is tonically enhanced by PAF input. Thus, a sustained increase in PAF input under conditions of peripheral inflammation enhances GABAergic activity in the DH. Further, as concerns PAF injury, following an initial, intense discharge, and prior to the development of ectopic activity (Section 8.2.2), damaged PAFs may traverse a period of quiescence whereby their stimulatory input onto ININs is reduced. This would lead to a transient and reversible reduction in the activity GABAergic ININs. The reduction in of GABAergic activity subsequently recovers by compensatory mechanisms and in parallel with the restoration of input from damaged PAFs due to their ectopic firing and other mechanisms discussed in Section 8.2. In some cases, a lack of recovery of PAF input may be associated with a more prolonged reduction of ININ activity (Castro-Lopes et al., 1993; Kelly et al., 1973).

10.6.3. Alterations in ININ Activity upon PAF Injury: A Summary

To summarize, an (at least) transient, functional reduction of tonic GABAergic and glycinergic ININ activity can mimic and accentuate processes of DH sensitization, thereby contributing to the hyperalgesia and allodynia elicited by PAF injury. It is unclear which mechanisms might provoke such a change in ININ activity. The destruction of a subpopulation of ININs in the DH may be attributable to NMDA receptor-mediated, excitotoxic mechanisms. In this case, a compensatory acceleration of synthesis in surviving populations of ININs would account for the recovery of normal levels of GABA. On the other hand, the loss of a facilitatory PAF input onto ININs provides an alternative explanation for a temporary reduction in ININ tone in the DH upon following PAF injury. Further, this possibility is consistent with the induction of GABAergic activity in the DH induced by an increase in PAF input under conditions of peripheral inflammation. An assessment of the relative merits of these theories for the loss of ININ tone in the DH upon PAF damage will require further experimental analysis. Irrespective of the underlying mechanisms, there is a consensus that a reduction in ININ tone in the DH has highly undesirable consequences for nociceptive processing and pain. Moreover, as discussed in Sections 12.3.2 and 13.3, there is evidence that a suppression of ININ activity in the thalamus and other cerebral structures involved in the processing of nociceptive information, may also play an important role in the induction of painful states.

As a final comment, GABA plays a neurotrophic role in the developing CNS. In this light, it might be speculated that the consequences of an alteration in GABAergic activity in the DH following PAF injury may not be limited to the modulation of nociception. Rather, GABA may be involved in the modulation of trophic processes including, perhaps, the reorganization of input from injured PAFs (Dumoulin *et al.*, 1996) (Section 11.2).

10.7. Reduction of Stimulatory A β Fibre Input to ININs upon PAF Injury

In accordance with traditional formulations of the 'gate' theory of nociceptive processing, it has been suggested that stimulation of $A\beta$ fibres provides a substrate for the analgesic effects of non-noxious manipulations, such as rubbing the site of cutaneous tissue injury. Since $A\beta$ fibres do not directly inhibit PNs, their actions could be mediated via the activation of ININs, which subsequently interact with C fibre terminals (Wall and Devor, 1981) or PNs themselves (Sections 4.6 and 9.1.4) (Fig. 2). Indeed, there is evidence for a reduction of ININ-mediated actions at fine calibre PAF terminals following PAF injury (Sections 9.1.4 and 10.6.2.1). Thus, the reduction of a $A\beta$ fibre-mediated stimulation of ININs upon PAF injury might disinhibit and further sensitize nocisponsive neurones in the DH. There are several possible mechanisms via which an attenuation of $A\beta$ input to ININs might take place.

- 1. There is some evidence that low threshold mechanoceptors show an altered (decrease) in their responsiveness following tissue injury (Na *et al.*, 1993; Treede *et al.*, 1992b; Wahren *et al.*, 1989).
- 2. A component of $A\beta$ fibre-mediated input to the DH may, at least temporarily, be directly incapacitated by lesions to PAFs (Laird and Bennett, 1992; Wall and Devor, 1981).
- 3. A reduction in the facilitatory $A\beta$ input at ININs in deeper laminae would occur upon the re-orientation of $A\beta$ fibres into superficial DH laminae (Section 11.2).

A loss of excitatory $A\beta$ input would provide a putative mechanism for a—possibly *rapid*—loss of ININ tone in the DH following PAF injury (Section 10.6). Nevertheless, under normal conditions, evidence that $A\beta$ fibres exert a major, inhibitory role upon PNs via stimulation of ININs is equivocal. Further, under conditions of secondary hyperalgesia and neuropathic pain (Sections 9.1.3 and 11.2), *direct, excitatory* actions of $A\beta$ fibres at WDR neurones in deep DH laminae—and, following rewiring, at NS neurones in superficial laminae—are of key importance in mediating mechanical allodynia. These actions of A β fibres *facilitate* rather than inhibit nociception and, overall, likely outweigh the functional effects of any reduction in excitatory input onto ININs (Sections 9.1.2 and 10.5).

10.8. Modulatory Role of Descending Pathways

10.8.1. A Decrease in Descending Inhibition and an Increase in Descending Facilitation

As indicated in Sections 6.4 and 9.1.4.2, exposure to noxious stimuli may, via a supraspinal loop, modify the activity of pathways descending from cerebral centres to the DH. Indeed, under conditions of inflammation, a transient or sustained role of descending, inhibitory pathways in dampening the activity of sensitized PNs in the DH has been proposed (Basbaum and Fields, 1984; Cervero and Laird, 1996a,b; Le Bars, 1988; Lin *et al.*, 1997; Millan, 1986, 1990, 1993, 1995, 1997; Owens *et al.*, 1992; Ren and Dubner, 1996; Sandkühler, 1996; Schaible and Grubb, 1993; Stanfa *et al.*, 1994; Traub, 1997; Zhang *et al.*, 1997f). These descending mechanisms operate via direct inhibition of PNs, suppression of release from PAF terminals and the engagement of ININs (Section 5.1).

Contrariwise, under certain conditions, a decrease in descending inhibition might, either directly or indirectly, contribute to an increase in the excitability of DH-localized PNs. Although there is little evidence for such a reduction in the activity of descending pathways upon exposure to noxious stimulation, intracellular events triggered in WDR neurones by noxious stimulation can lead, via the activation of PKs, to a weakening of GABAergic and/or glycinergic tone (Section 10.4.2). Consequently, the component of descending inhibition which is mediated via activation of DH-localized ININs is compromised, and processes of sensitization amplified (Lin et al., 1997). Further, spinal and supraspinal actions of morphine may act synergistically in exerting antinociceptive actions. It has been suggested that this interaction, which is expressed in the rostral ventromedial medulla, is perturbed under conditions of neuropathic pain, resulting in a reduced efficacy of morphine (Heinricher et al., 1994; Ossipov et al., 1997). Whether this loss of synergy reflects a reduction in descending inhibition, or the parallel activation of descending facilitation, is unclear. In any case, as described in Section 10.8.5., the rostral ventromedial medulla is a key site for the initiation of facilitatory networks running to the DH.

As discussed in the following paragraphs, an *enhancement* in the activity of descending, *facilita-tory* mechanisms may contribute to an increase in nociception and the induction of sensitization.

10.8.2. Adrenergic Mechanisms

There is extensive evidence for a facilitatory influence of both acute and prolonged, noxious stimulation upon the activity of centrifugal, adrenergic pathways. Further, a tonic, antinociceptive role of NAD, exerted via inhibitory DH-localized α_{2A} -ARs negatively coupled to AC, inhibits inflammatory nociception (Schaible *et al.*, 1991; Stanfa *et al.*, 1994; Millan, 1997). However, there is some evidence that NAD mediates *pro*-nociceptive actions in the DH via activation of excitatory α_1 -ARs positively coupled to PLC and increases in $[Ca^{2+}]_i$ (Jordan *et al.*, 1979; Jones, 1992; Millan, 1997). Thus, an increase in activity at PN-localized α_1 -ARs might be involved in mechanisms underlying sensitization.

10.8.3. Serotonergic Mechanisms

The activity of descending, serotoninergic pathways in the DH is likewise modified under conditions of acute and prolonged nociceptive and neuropathic pain, and spinal 5-HT turnover is modified in chronic pain patients (Von Knorring, 1989; Weil-Fugazza, 1989; Wolfe et al., 1997; Millan, 1995). Inasmuch as serotoninergic mechanisms in the DH exert both antinociceptive and pronociceptive actions via different 5-HT receptor subtypes, it is conceivable that an increase in activity at the latter contributes to DH sensitization (Millan, 1995, 1997). Further, stimulation of serotoninergic neurones projecting to the DH from brainstem raphe nuclei results in a biphasic excitatory and inhibitory influence upon DH neurones (Zhuo and Gebhart, 1997). In this regard, 5-HT_{2C} and 5-HT_{2A} receptors are positively coupled to PLC and their activation suppresses K⁺-currents (Boess and Martin, 1994; Millan, 1997). Further, 5-HT₃ receptors directly activate a cation-permeable ion channel which triggers the opening of VDCCs, and their engagement can also induce PLC and further increases in $[Ca^{2+}]_i$ (Boess and Martin, 1994; Millan, 1995; Rondé and Nichols, 1997). Thus, 5-HT $_{\rm 2A},$ 5-HT $_{\rm 2C}$ or 5-HT $_{\rm 3}$ receptors localized on PNs, EXINs or the terminals of PAF fibres may play a role in mediating pronociceptive actions of 5-HT and mechanisms of descending facilitation (Inoue et al., 1997; Kjørsvik et al., 1997; Millan, 1995, 1997; Saria et al., 1990). On the other hand, $5-HT_{1A}$ receptors which, like 5-HT₃ receptors, are enriched in the superficial DH, are negatively coupled to AC and their stimulation enhances and suppresses K⁺-Ca²⁺-currents, respectively (Boess and and Martin, 1994; Millan, 1995; Yang et al., 1998). Thus, it has been proposed that inhibition of 5-HT1A receptors localized on ININs play a role in processes underlying DH sensitization and allodynia by the disinhibition of WDR PNs (Millan et *al.*, 1994, 1996; Millan, 1995, 1997) (Section 10.6.2). In this light, thus, the putative actions of 5-HT_{1A}, 5-HT_{2A}, 5-HT_{2C} and 5-HT₃ antagonists in models of neuropathic pain would be of interest to evaluate.

Interestingly, there is evidence for plasticity in descending serotoninergic pathways under conditions of noxious stimulation or nerve damage. Thus, in addition to changes in spinal levels of 5-HT, the sprouting of serotoninergic neurones has been seen in the DH following peripheral capsaicin treatment or PAF injury. Notably, serotonergic terminals in deeper laminae extend into lamina II (Godefroy *et al.*, 1987; Marlier *et al.*, 1992; Polistina *et al.*, 1990; Wang *et al.*, 1991). These changes may involve specific, intrinsic growth-promoting proteins, the significance of which in encouraging the regeneration of (injured) nerves is considered in Section 11.3 (Ching *et al.*, 1994). In addition, serotoninergic neurones projecting to the DH bear TRK C receptors for NT 3, the potential role of which is considered in Section 14.5.3 (Arvidsson *et al.*, 1994).

The above observations suggest that a perturbation of serotoninergic transmission in the DH may be involved in modulating (enhancing) nociception under conditions of nociceptive and neuropathic pain. In this context, it is of note that descending serotoninergic pathways modulate the rostral transmission of normally-innocuous, tactile information via an influence upon PSDC neurones (Wu and Wessendorf, 1992) (Section 6). Whether this role can be related to the induction of mechanical allodynia, and whether 5-HT modulates transmission of nociceptive (visceral) information by PSDC pathways, would be of interest to evaluate.

10.8.4. Dopaminergic Mechanisms

Dopaminergic networks in the DH are also involved in the modulation of nociception, and multiple dopaminergic receptors exert a contrasting influence upon nociception. Thus, D_2 and D_1 receptors, which are positively and negatively coupled to AC, mediate antinociceptive and pro-nociceptive actions respectively, presumably via direct actions at WDR neurones in the DH (Millan, 1997; Verma and Kulkarni, 1993). Correspondingly, an increase in activity at dopamine D_1 receptors, or a reduction in activity at dopaminergic D_2 receptors, might contribute to the sensitization of WDR neurones in the DH.

10.8.5. Role of Cerebral Mechanisms in Triggering Descending Facilitation

A loss of cerebral GABAergic inhibitory tone may result in the disinhibition of descending, (Hammond, 1997: facilitatory pathways Monhemius et al., 1997) and several-non-monoaminergic-mechanisms may play a role in engaging descending facilitatory pathways. Notably, neuropeptide FF and CCK, each of which counters antinociceptive mechanisms in the DH (Sections 3.2.6 and 4.5). Several studies have suggested that the rostro-medial region of the ventral medulla is a major centre for the initiation of mechanisms of descending facilitation and hyperalgesia. Indeed, the temporary interruption of neuronal transmission in this structure by local application of lidocaine attenuates the allodynia associated with nerve damage or inflammation (Almeida et al., 1996; Morgan et al., 1994; Wiertelak et al., 1994a,b,c,d; Watkins et al., 1994). Recent evidence suggests that the novel opioid peptide, nociceptin, may play a role in hyperalgesic mechanisms in this and other structures, such as the PAG (Heinricher et al., 1997; Morgan et al., 1997). Further, neurotensin is implicated in pronociceptive mechanisms in the rostro-medial ventral medulla, and a circuit invol-

ving medullary pools of neurotensin, and CCK in the DH, was suggested to mediate the hyperalgesia of the tail elicited by noxious, chemical stimulation of the paw (Herrero and Cervero, 1996; Skinner et al., 1997; Smith et al., 1997a; Urban et al., 1996a,b; Urban and Gebhart, 1997; Zhuo and Gebhart, 1997). Whether the DH pool of CCK involved is derived from descending terminals, EXINS or PAFs is unclear. In any case, this observation supports the hypothesis that supraspinal mechanisms and descending facilitation are involved in the induction of referred pain (Section 9.1.4.2). A further supraspinal structure excited by ascending nociceptive input, the anterior pretectal area of the brainstem, may also, via descending facilitatory pathways, exert an ex-citatory influence upon NS neurones in the superficial DH. This region has likewise been implicated in the mediation of painful states due to PAF injury, as well as in the induction of referred pain (Rees and Roberts, 1993; Rees et al., 1995b).

10.8.6. Physiological Activation of Mechanisms of Descending Facilitation

The above findings suggest that a supraspinal loop engaging descending, facilitatory mechanisms may contribute to the induction of nociception and DH sensitization (Mansikka and Pertovaara, 1997a; Pertovaara et al., 1996). Notably, such descending pronociceptive mechanisms may be elicited not only by exposure to noxious stimuli, but also by other stimuli such as vagal stimulation and/or 'illness' (Lovick, 1993; Urban *et al.*, 1996a,b; Watkins *et al.*, 1994; Wiertelak *et al.*, 1994a,b,c,d; Zhuo and Gebhart, 1992, 1997). Indeed, as concerns the response to 'illness', the release of IL-1 β by macrophages has been suggested to activate vagal, sensory afferents running to the nucleus tractus solitarius. Therefrom, a descending facilitatory pathway is activated which elicits nociception by the recruitment of NO and NMDA receptors in the DH (Watkins et al., 1994, 1995; Wiertelak et al., 1994a,b,c,d). It was pointed out in Sections 10.4.2 and 10.8.1 that mechanisms of descending inhibition at WDR PNs may be defective in prolonged, painful states due to PAF- or CNS-injury. A parallel enhancement in mechanisms of descending facilitation implies a displacement in the overall influence of modulatory, descending pathways towards an enhancement of nociceptive transmission. This disequilibrium will amplify the excitation and sensitization of DH neurones transmitting nociceptive information to cerebral structures (Traub, 1997).

11. NEUROPATHIC PAIN: REORGANIZATION OF CENTRAL AND PERIPHERAL PROJECTIONS OF PAFS FOLLOWING INJURY

11.1. Peripheral-Central Interface in the DH

A distinctive mechanism involved in the induction of prolonged, neuropathic pain consists ofprobably permanent-changes in the DH projection targets of PAFs following peripheral nerve injury. This reorganization of PAF input to the DH is a process which lies at the interface of 'peripheral' and 'central' mechanisms underlying chronic pain. Processes of atrophy and regeneration following destruction of PAFs may be attributed to a complex and interactive mesh of factors both intrinsic and extrinsic to damaged PAFs, full details of which are beyond the scope of this review (Deller and Frotscher, 1997; Fu and Gordon, 1997). Attention here is focused on the relationship of such events to alterations in nociception. In this light, the implication of two, specific molecules is of particular pertinence.

- 1. neurotrophins, in view of their crucial role as trophic factors for sensory nerves, and of the modulatory influence of NGF upon nociception via actions at the peripheral terminals of small calibre C fibres (Sections 7.6 and 7.7);
- 2. the growth associated phosphoprotein (GAP) 43, which plays a key role in the growth and orientation of injured—and, probably, intact—small calibre C fibres (Section 11.3).

11.2. Central Degeneration and Regrowth of Injured PAFs: Abnormal Patterns of $A\beta$ Fibre Connectivity

The transection of peripheral nerves leads to marked degenerative/atrophic and regenerative changes in their central terminals. Following destruction of the peripheral branches of PAFs, a subpopulation of terminals in the DH may persist, while others die by processes resembling apoptosis. Still others may undergo degenerative ('chromolytic') changes followed by a switch to a regenerative, 'growth mode'-although the latter processes does not necessarily depend on the former (Cameron et al., 1997; Fu and Gordon, 1997; Lieberman, 1971; McMahon and Priestley, 1995). Most notably, as concerns nociceptive processing, A β fibres sprout within a dorsoventral plane-to inappropriate targets within the superficial DH (Ambron and Walters, 1996; Cameron et al., 1997). That is, following axotomy of $A\beta$ fibres distal to the DRG, their central projections in laminae III and IV begin to penetrate superficial laminae (in particular lamina, II₀). Therein, they may potentially interact, either synaptically or via volume transmission, with NS (and WDR) neurones which they would not otherwise access (Woolf et al., 1992) (Fig. 2). This reorientation of $A\beta$ fibres into superficial laminae is compounded by a phenotypic switch whereby they begin to synthesize the pronociceptive and excitatory transmitters, SP and VIP—in addition to EAAs (Section 8.2.5). Further, there is an upregulation of NK₁ receptors and an increased Ca²⁺-permeability of AMPA receptors in superficial DH (Abbadie *et al.*, 1996; Coggeshall and Carlton, 1998; Goff *et al.*, 1998). Thus, an aberrant pattern of A β input to nocisponsive neurones in the superficial DH provides a substrate for the mediation of mechanical allodynia and hyperalgesia (Doubell *et al.*, 1997; Lekan *et al.*, 1996, 1997; Woolf and Doubell, 1994; Woolf *et al.*, 1992) (Section 9.1.2).

In addition to the extra-territorial sprouting of damaged $A\beta$ fibres, there is some evidence that regenerating, damaged C fibres may expand into deeper DH laminae (Cameron *et al.*, 1992; Florence *et al.*, 1993; Lekan *et al.*, 1997; LaMotte *et al.*, 1989; McMahon and Kett-White, 1991). Should C fibres infiltrating deeper laminae contact (or volume activate) WDR and NS neurones, they would also contribute to patterns of abnormal signalling and hyperalgesic states.

PAF sprouting following injury is, in general, far more marked in immature than in adult animals (Fitzgerald *et al.*, 1994; McMahon and Priestley, 1995; Shortland and Fitzgerald, 1994; Wilson and Kitchener, 1996). This difference provides several, important insights into processes potentially promoting the regeneration of damaged nerves (Fawcett, 1992; Schwab, 1990).

- 1. Immature nerves developmentally express high levels of several 'growth-associated-factors' which are less prominent in uninjured, adult, peripheral neurones. These include GAP 43, which, as described in Section 11.3, plays an important role in the regeneration of damaged (and intact) C fibres, and cytoskeletal proteins such as tubulin, which support GAP 43-coordinated growth cones (Benowitz and Routtenberg, 1997; Coggins and Zwiers, 1991; Fournier and McKerracher, 1997; Meiri and Burdick, 1991; Strittmatter *et al.*, 1995).
- There is a lesser degree of myelination and a generally more permissive environment for nerve growth in the immature DH (Kolodkin, 1996; Nieto, 1996).
- 3. Immature neurones are more dependent upon peripheral stores of neurotropins: the loss of this peripheral source of neurotrophins upon nerve section leads, thus, to a greater degree of central degeneration in the immature DH and to the liberation of a greater number of potential synaptic sites for regenerating axons (Wilson and Kitchener, 1996).

Despite these differences, as indicated above, the reorientation of PAFs to inappropriate targets in the DH *also* occurs following peripheral nerve lesions in mature subjects.

To summarize, following transection of PAFs, axonal transport is reduced, the DRG loses access to neurotropins and other essential, trophic factors (Gold *et al.*, 1991; Rich *et al.*, 1987; Wong and Oblinger, 1991), PAF phenotype is modified (Section 8.2.5), intraneuronal structural changes occur due to an altered expression of cytoskeletal proteins, and, ultimately, DH terminals of PAFs

display morphological alterations in their structure and 'transganglionic' atrophy. An important consequence of these events is a decrease in the density of functional synaptic contacts and vesicles in the DH, resulting in the appearance of so-called 'vacant' synapses potentially accessible to other neurones, such as migrating $A\beta$ fibres. (Mechanisms triggering the migration of $A\beta$ fibres into superficial laminae are discussed in detail in Section 11.3.3).

Although, *sub-total*, peripheral nerve damage (as compared to complete transection) does *not*, at least, initially, incur the degeneration of central DH terminals, this can occur in the absence of successful *peripheral* regeneration (Aguayo, 1985; Castro-Lopes *et al.*, 1990; Hachisuka *et al.*, 1989; Kitchener *et al.*, 1994; Knyihar-Csillik and Török, 1989; Knyihar-Csillik *et al.*, 1987, 1989, 1990, 1992; Snow and Wilson, 1991). Indeed, the regeneration and re-establishment of the *peripheral* targets of injured PAFs encourages, and may be a pre-condition for, restoration of their central regrowth (Ambron and Walters, 1996; Chong *et al.*, 1996; McMahon and Kett-White, 1991; Winter *et al.*, 1992).

11.3. The Role of GAP 43 and other Factors in the Central and Peripheral Regeneration of Injured PAFs

11.3.1. The Induction of GAP 43 in Injured C Fibres: Short-Term and Long-Term Role in Neuronal Sensitization

GAP 43 is a calmodulin-binding phosphoprotein activated (phosphorylated) by PKC and deactivated (dephosphorylated) by calcineurin, which fulfils a crucial role in controlling the growth and connectivity of axons (Aigner and Caroni, 1995; Benowitz and Routtenberg, 1997; Li et al., 1995; Ramakers et al., 1997). It is found almost exclusively in their presynaptic terminals, wherein it is closely-attached to the terminal membrane. GAP 43 is virtually undetectable in large diameter mature PAFs, and modestly-expressed in their mature, small calibre counterparts (Benowitz and Routtenberg, 1997; Leslie et al., 1995). It is a principal component of the growth cones of developing neurones, sprouting, intact neurones and injured, regenerating C fibres (Benowitz and Routtenberg, 1997; Buffo et al., 1997). The precise signals triggering GAP 43 activation upon nerve injury and other conditions remain to be defined. However, a PKC-mediatedand calmodulin-repressible-phosphorylation of GAP 43 at residue Ser⁴¹ transforms cells into a growth mode (Benowitz and Routtenberg, 1997; Ghosh and Greenberg, 1995; Gu and Spitzer, 1995; Siaastad and Nelson, 1997). The actions of GAP 43 may involve a modification of its interaction with the cytoskeleton, the activation of specific G proteins and/or the induction of various, nuclear transcription factors (Hens et al., 1993; Namgung et al., 1997; Ramakers et al., 1997; Verkade et al., 1997). An activation of various Ca²⁺-dependent and -independent isoforms of PKC has been observed in injured nerves. Of these, a PKC β isoform is likely implicated in the activation of GAP 43, although additional isoforms, acting via GAP 43 and/or other mechanisms, may also be involved in neuronal regeneration (Benowitz and Routtenberg, 1997; Kawano *et al.*, 1997; Okajima *et al.*, 1995).

In the light of this role of PKC in the activation of GAP 43, it is of interest to recall the positive coupling to PKC of several, pronociceptive, inflammatory mediators at C fibre terminals, including SP, BK, 5-HT and PGE₂. Further, GAP 43 is up-regulated in C fibre terminals upon prolonged peripheral inflammation (Leslie et al., 1995). Nevertheless, in addition to PKC, other permissive and/or obligatory factors are likely involved in the regulation of GAP 43 activity in PAFs (and other neurones) inasmuch as their transient excitation is not sufficient for sustained GAP 43 activation and the induction of neuronal growth (Dekker et al., 1989; Namgung et al., 1997). In this regard, one potentially important, intracellular regulator element should be mentioned: the IEG, c-jun. As discussed in Section 8.2.6.3, c-jun plays a role in modulating patterns of gene transcription following its induction in damaged PAFs (Herdegen et al., 1997). The GAP-43 gene possesses a highly-conserved, c-jun-sensitive, regulatory element which modulates its transcription. Further, levels of c-jun are increased and co-expressed with GAP 43 in a subset of neurones following their injury (Schaden et al., 1994). Thus, c-jun may be involved in the induction of GAP 43 expression following PAF damage. In this regard, moreover, c-jun (and PKC) may intervene in the induction of GAP 43 by NGF (Kaplan and Stephens, 1994; Leslie et al., 1995) (Section 11.3.4.2).

There is evidence that GAP 43, once phosphorylated by PKC, may also modulate transmitter release. This suggests that, at central and/or peripheral terminals of regenerating neurones, in addition to its long-term role in the modification and reestablishment of synaptic connectivity, GAP 43 might rapidly modulate nociception via alterations in the release of SP or other pronociceptive sub-stances (Dekker et al., 1989; Hens et al., 1993; Ramakers et al., 1997). In the DH, this might occur upon NMDA receptor-mediated activation of intrinsic neurones to release a retrograde factor which then enhances PKC activity in PAF terminals: for example, arachidonic acid metabolites, PGs or EAAs acting at PLC-coupled, group I mGlu receptors (Sánchez-Prieto et al., 1996). Further, in contrast to a lack of effect of transient depolarization, presynaptic pools of GAP 43 are induced/phosphorylated in parallel with the expression of LTP in the hippocampus (Benowitz and Routtenberg, 1997; Namgung et al., 1997) (Section 10.2.1). This observation is consistent with a broader role of GAP 43 in adaptive processes underlying prolonged and rapid alterations in synaptic transmission. Further, it provides an interesting parallel to the rapid and long-term actions of neurotrophins in modulating synaptic transmission (Section 14.4).

In conclusion, an exploration of the role of GAP 43 in DH-localized PAF terminals in the sustained *and* rapid enhancement of neuronal sensitization and nociception upon prolonged tissue inflammation, or PAF injury, would be of interest. Such a potential role of GAP 43 in supraspinal structures might also be envisaged.

11.3.2. The Role of GAP 43 and Other Intrinsic Factors in the Central Regeneration of Damaged PAFs

GAP 43 is present a minority of PAFs under normal conditions but it reappears in virtually all damaged, small calibre C fibres following their injury (Chong et al., 1992; Leslie et al., 1995; Namgung et al., 1997; Schreyer and Skene, 1991; Sommervaille et al., 1991; Tetzlaff et al., 1991; Van der Zee et al., 1989) and plays a role in the regeneration of their peripheral and central terminals, to both of which it is transported. Anatomical and ultrastructural studies have, indeed, shown that GAP 43 is concentrated in growth cone-like structures of C fibre terminals in the DH following their peripheral injury: further, its appearance coincides with the onset of their regenerative, central regrowth (Cameron et al., 1997; Chong et al., 1994; Coggeshall et al., 1991; Doubell and Woolf, 1997; Knyihar-Csillik et al., 1992; Schreyer and Skene, 1991; Woolf et al., 1990).

It might reasonably be assumed that GAP 43 is also present in the terminals of $A\beta$ fibres which infiltrate Lamina II following their migration from deeper laminae (Koerber, 1996; McMahon and Kett-White, 1991; Shortland and Woolf, 1993; Wilson and Kitchener, 1996; Woolf et al., 1992). However, a recent study has, surprisingly, indicated the contrary (Doubell and Woolf, 1997). This observation suggests that GAP 43 may not be critical for the sprouting of the central terminals of $A\beta$ fibres. Other classes of growth-associated protein may, thus, be involved (Doubell and Woolf, 1997). In this context, it is of note that several other substances contained in PAF terminals and potentially involved in axonal orientation and growth appear in the DH territories of injured nerves following their lesioning. These include soybean agglutinin, the presence of which is associated with sensory neurone growth; the carbohydrate-binding lectin, R2-14.5, which promotes neuronal regrowth and the neuronal adhesion molecule RL-29, which regulates mRNA expression in PAFs (Barondes et al., 1994; Cameron et al., 1997; Fields and Itoh-Kouichi, 1996; Fu and Gordon, 1997) (Section 11.3.4.4). In addition, the induction of mRNA encoding the cytoskeletal protein, tubulin, suggests that this offers physical support for the regeneration and orientation of the growth cones of injured neurones (Challacombe et al., 1996; Fournier and McKerracher, 1997). The roles of GAP 43, cytoskeletal proteins and other factors in synaptic remodelling and axonal regeneration following damage to PAFs is the subject of intensive investigation (Fu and Gordon, 1997).

11.3.3. Mechanisms Underlying Aβ Fibre Migration into Foreign DH Laminae

The mechanisms underlying the reorientation of $A\beta$ fibres into alien territories remain unclear. However, it is likely that several, interrelated and interactive processes are involved.

- 1. Synaptic sites in superficial laminae (II₀) are vacated by the degeneration and/or withdrawal of damaged C fibres (Castro-Lopes *et al.*, 1990; Deller and Frotscher, 1997; Knyihar-Csillik *et al.*, 1987) (Section 11.2). The argument that a loss of 'competing' C fibres in lamina II permits expansion of $A\beta$ fibres away from their normal sites of termination is supported by developmental studies showing that:
 - 1. A β fibres *do* initially run to superficial lamina but they are eventually displaced by incoming C fibres; and
 - 2. prevention of C fibre innervation by capsaicin treatment allows $A\beta$ fibres to maintain their presence in superficial laminae (Cameron *et al.*, 1997; Fitzgerald *et al.*, 1994; Shortland *et al.*, 1990). Further, NGF can *prevent* the sprouting of $A\beta$ fibres into lamina II following PAF lesions, possibly since it acts as a trophic factor for C fibres, thereby preventing their withdrawal from superficial lamina (Bennett *et al.*, 1996a). However, as discussed in Section 11.3.4.2, the role of NGF as a trophic factor for the central regrowth of injured C fibres is unclear (Zhou *et al.*, 1996b; Gallo *et al.*, 1997; Song *et al.*, 1997).
- 2. Lamina II-localized C fibre terminals, or nonneuronal glial and microglial cells in their vicinity, may release substances attracting $A\beta$ fibres and encouraging their growth (Doubell *et al.*, 1997; Goodman *et al.*, 1996; Song *et al.*, 1997). For example, an increase in the release of C fibre-localized neuropeptides possessing trophic properties, such as NPY or GAL, could be involved (Hökfelt *et al.*, 1994; Xu and Wiesenfeld-Hallin, 1997). However, for several reasons, a more specific and attractive candidate would be BDNF.
 - 1. Unlike NGF, BDNF is a trophic factor for $A\beta$ fibres (Lexin and Barde, 1996; Mannion *et al.*, 1996).
 - BDNF can induce the expression of molecules encouraging axonal regrowth, such as tubulin and GAP 43, and it encourages branching in some peripheral neurones (Fournier *et al.*, 1997; Kobayashi *et al.*, 1997a);
 - following damage to PAFs, BDNF is induced in (TRK A-bearing) injured C fibres, wherein it is transported to their spinal terminals (Barakat-Walter, 1996; Cho *et al.*, 1997b,c; Funakoshi *et al.*, 1993; Michael *et al.*, 1997a; Vergé *et al.*, 1995b; Zhou and Rush, 1996);
 - BDNF levels are concentrated in lamina II of the DH (Cho *et al.*, 1997b,c; Conner *et al.*, 1997; Ernfors *et al.*, 1990; Furukawa *et al.*, 1997); and
 - 5. in a cAMP-dependent manner, BDNF modifies the turning behaviour of the growth cones of spinal neurones (Song *et al.*, 1997).

Although the density of TRK B receptors is decreased in the *peripheral* branches of damaged PAFs, this is probably not the case for their central terminals (Fu and Gordon, 1997). Thus, the release of BDNF by the central terminals of damaged C

fibres in laminae II may—in analogy to the orienting response of TRK A receptor-bearing C fibres to NGF (Gallo *et al.*, 1997; Song *et al.*, 1997)—attract TRK B receptor-bearing $A\beta$ fibres into lamina II₀. However, additional factors may be involved inasmuch as it is difficult to explain the *dorsoventral* orientation of $A\beta$ infiltration of lamina II by a mechanism *exclusively* implicating BDNF (Doubell and Woolf, 1997).

- In superficial laminae vacated by degenerated C fibre terminals, there may be a relative absence of molecules inhibitory to growth cones, such as myelin, sulphated proteoglyans and other proteins synthesized by astrocytes and oligodendrocytes: thus, the extracellular matrix is relatively favourable to neuronal growth (Fazeli and Walsh, 1996; Fawcett, 1992; Fu and Gordon, 1997; Golding *et al.*, 1997; Kolodkin, 1996; Nieto, 1996; Schwab, 1990, 1996).
- 4. A modulation in the expression, or Ca^{2+} CAMP-dependent actions, of 'attractant and repellant' molecules involved in central axonal guidance and target recognition, such as netrins and semaphorins, may be involved (Kobayashi et al., 1997a,b; Kolodkin, 1996; Ming et al., 1997; Olson, 1997; Tanelian et al., 1997; Tessier-Lavigne and Goodman, 1996). For example, during development, semophorin III, which acts via a recently-cloned neuropilin receptor (Kitzukawa et al., 1997; Kolodkin et al., 1997), induces growth cone collapse in sensory, small calibre C and A δ PAFs in the VH (Tanelian et al., 1997). If PAF damage results in a reduced expression of repellant factors for $A\beta$ fibres in the superficial VH, their infiltration of lamina II would no longer be discouraged (Messersmith et *al.*, 1995; Püschel, 1996; Püschel *et al.*, 1996; Wright *et al.*, 1995; Tanelian *et al.*, 1997). On the other hand, a repellant factor expressed upon PAF injury in deeper laminae may drive $A\beta$ fibres out of this region. Contrariwise, an increase and decrease in the expression of a chemoattractant for $A\beta$ fibres in superficial and deep lamina, respectively, would also facilitate migration of $A\beta$ fibres in lamine II.

Mechanisms analogous to those outlined above are probably involved in the reorientation of C fibres into deeper laminae following their peripheral injury (Section 11.2). The nature of multiple mechanisms underlying patterns of regeneration and reorganization of PAF terminals in the DH following destruction of their peripheral branches will provide fertile territory for further investigation (Fazeli and Walsh, 1996; Fields and Itoh-Kouichi, 1996; Kikuchi *et al.*, 1997; Kolodkin, 1996; Kobayashi *et al.*, 1997a; Naciff *et al.*, 1996; Nieto, 1996; Taniguchi *et al.*, 1997; Tessier-Lavigne and Goodman, 1996).

11.3.4. Peripheral Regeneration of Damaged PAFs

11.3.4.1. The importance of Schwann cells

Regeneration of the peripheral branches of damaged PAFs is probably imperative for the regrowth of their *central* branches and the restitution of a normal phenotype. Peripheral regeneration involves the re-access of damaged PAFs to neurotrophins and many other nutrient factors (Aigner and Caroni, 1995; Berninger and Poo, 1996; Benowitz and Routtenberg, 1987; Deller and Frotscher, 1997; Fu and Gordon, 1997; Terenghi *et al.*, 1997; Vergé *et al.*, 1995a; Wells *et al.*, 1994).

For the successful peripheral regeneration of damaged (myelinated) neurones, Wallerian degeneration is probably a prerequisite. That is, the removal of myelin and other factors inhibiting nerve regeneration. This process is effected primarily by macrophages which, together with damaged PAFs themselves, release mitogens triggering the proliferation of Schwann cells (Brown et al., 1997; Fields and Itoh-Kouichi, 1996; Fu and Gordon, 1997; Matsuoka et al., 1997). In parallel to the switch of injured neurones from a 'transmission' to a 'growth' mode, Schwann cells switch from a 'sheath' to a 'support' mode for PAFs. Thus, in addition to the elaboration of myelin for larger calibre neurones, they play a more general role in supporting the growth of injured PAF.

- 1. Schwann cells become a rich source of the neurotrophins, NGF, BDNF and GDNF as well as the cytokine, LIF, and other molecules permitting the regeneration of damaged PAFs (Bunge, 1993).
- 2. They secrete cellular adhesion molecules (CAMs), such as *N*-cadherin and ninjurin, extracellular matrix molecules, such as laminin and fibronectin, and other elements providing an environment conducive to the growth of damaged axons (Section 11.3.4.4).
- 3. Peripheral nerves likely regenerate through columns of basal laminae (endoneurial tubes or 'Bands of Bungner') filled with Schwann cells which have proliferated following axonal loss. Schwann cells also provide, thus, the physical support necessary for the regrowth of injured neurones (Fu and Gordon, 1997; Martini, 1994; Venstrom and Reichardt, 1993; Xu et al., 1995c).
- Schwann cells may contribute to the distribution and clustering of axonal Na⁺- and K⁺-channels in myelinated axons, thereby allowing for optimal neuronal conduction properties (Kaplan *et al.*, 1997; Joe and Angelides, 1992; Rasband *et al.*, 1998).

One particularly interesting mode of reciprocal interaction between regenerating PAFs and surrounding Schwann cells has recently been brought to light by Livesey et al. (1997). Damage to PAFs results in a massive induction of the cytokine, LIF, in Schwann cells (Jonakait, 1993). As outlined in Section 8.2.6.2, LIF can modify the phenotype of certain, injured PAFs. In addition, it prompts the expression and secretion of the protein Reg-2 in injured PAFs which, together with other substances, acts as a potent mitogen for Schwann cells (Livesey et al., 1997; Matsuoka et al., 1997). Thus, via injured PAFs, Schwann cells actually promote their own proliferation thereby creating a physical and chemical environment favourable to PAF regeneration (vide supra). In line with these findings, LIF has been shown to rescue damaged PAFs from degeneration (Cheema *et al.*, 1994). Interestingly, LIF may also play a role, together with NGF, in promoting the sprouting of sympathetic neurones into the PAF following PAF injury (Thompson and Majithia, 1997) (Section 8.2.4.2).

11.3.4.2. NGF and processes of regeneration in GAP 43-Positive C fibres

Following lesions of C fibres, a reduced provision of NGF to their DRG results in an alteration of their phenotype (Section 8.2.6). Correspondingly an eventual reaccess to NGF may play a role both in the restitution of a normal phenotype and in mechanisms underlying their regeneration. As discussed above (Section 11.3.4.1), Schwann cells constitute a rich source of NGF, and exogenous NGF administration was reported to reduce the transgangionic degeneration of C fibre terminals in the DH provoked by axotomy (Fitzgerald et al., 1985; McMahon and Bennett, 1997). Further, studies performed both in vivo and on cultured sensory neurones suggest that, via an action at TRK A receptors, NGF can increase the stability of GAP 43 and enhance growth in intact C fibres (Benowitz and Routtenberg, 1997; Leslie et al., 1995; Mohiuddin et al., 1995) (Section 11.3.1). The 'spreading' response elicited in developing, sensory neurones by NGF is also a GAP 43-dependent process (Aigner and Caroni, 1995). A more general role of NGF in encouraging PAF regrowth may also be derived from its promotion of the proliferation and migration of non-neuronal cells, thereby providing an environment conducive to regrowth and improving the irrigation of damaged PAFs. Further, there has been recent interest in the possibility that low affinity p^{75NTR} receptors for NGF on Schwann cells are involved in 'constitutive' apoptosis, although it is still controversial as to whether their occupation by NGF suppresses or activates such processes (Bredesen and Rabizadeh, 1997; Dechant and Barde, 1997; Fu and Gordon, 1997).

Notwithstanding these observations, several phenotype changes in damaged PAFs are mediated independently of NGF (Section 8.2.6) and NGF is not likely to be an absolute (or sufficient) requirement for *initiating* the regeneration of damaged PAFs (Diamond et al., 1987, 1992a). Notably, NGF was found not to modify GAP 43 expression following nerve transection (Vergé et al., 1990). Further, in view of the initial loss of TRK A receptors and the reduction in access to NGF upon transection of PAFs, NGF is unlikely to be essential for triggering either GAP 43 expression or nerve regeneration per se. Indeed, there is even evidence to suggest that, under certain conditions, NGF may retard the regrowth of some damaged neurones, perhaps by mechanisms involving the induction of NO (Diamond et al., 1987, 1992a,b; Gold and Spencer, 1993; Melino and Bernassola, 1997; Gold et al., 1991, 1993). Thus, the significance of NGF for regenerative processes in damaged, adult C fibres remains unclear.

11.3.4.3. Roles of GDNF and other trophic factors

As mentioned in Section 11.3.2, a substantial majority of small calibre PAFs display GAP 43 expression. Evidently, they cannot all be TRK Apositive, NGF-dependent cells. Indeed, a further population of fine calibre, nociceptive PAFs, characterized by the expression of P_{2X3} receptor on their peripheral terminals, does not possess TRK A receptors (Sections 3.2.1. and 8.2.5). Rather, following an early developmental switch, these neurones become dependent on, and regulated by, GDNF. GDNF acts via a receptorial complex consisting of a GFR α -1 (GDNFR α) binding site (of which multiple forms may exist), together with a further 'Ret' TRK protein product (Jing et al., 1997; Kotzbauer et al., 1996; Leclere et al., 1997; Liebl et al., 1997; Molliver et al., 1997). GDNF can prevent the degeneration of this population of PAFs following their injury. Thus, GDNF likely fulfils an important role in the protection, rescue and subsequent regeneration of certain classes of C fibre following their peripheral damage (Leclere et al., 1997; Matheson et al., 1997; Naveilhan et al., 1997). Studies of motoneurones suggest that mRNA encoding Ret and GFRa-1 is induced by nerve lesions, and GDNF may even be synthesized by injured motoneurones themselves, although such finding remain to be documented for sensory nerves (Naveilhan et al., 1997).

As concerns those neurotrophins which act predominantly on large calibre PAFs (BDNF, NT 4/5 and NT 3), their roles remain comparatively obscure. Nevertheless, studies of central axons and motoneurones indicate that BDNF can improve their regenerative capacity and enhance the expression of growth-supporting molecules such as GAP 43 and tubulin (Fournier et al., 1997; Kobayashi et al., 1997b), raising the possibility that such actions may be exerted at injured, large calibre PAFs. Further, BDNF may fulfil a protective (autocrine) role in protecting sensory neurones in the DRG from degeneration (Acheson et al., 1995). NT 3 (together with NGF) induces GAP 43 synthesis and neurite extension in cultured, sensory neurones and, in vivo, it encourages the regeneration of injured (NPY-containing), large calibre PAFs-as well as sympathetic neurones (Mohiuddin et al., 1995; Molliver et al., 1997; Sawai et al., 1996; Sterne et al., 1997; Tafreschi et al., 1998) (Section 8.2.4.2). Further, TRK C receptors for NT 3 are up-regulated following damage to PAFs (Funakoshi et al., 1993).

11.3.4.4. Cellular adhesion and extracellular matrix molecules

Apart from growth factors, many other classes of molecule are involved in triggering and coordinating the regrowth of damaged (and intact) PAFs (Deller and Frotscher, 1997; Fu and Gordon, 1997; Smith *et al.*, 1993). In this regard, CAMs have attracted particular attention (Fields and Itoh-Kouichi, 1996; Fu and Gordon, 1997). As mentioned above, CAMs are released by Schwann cells and certain other cell types, and they play a generalized role in axonal regrowth and navigation by permitting the adhesion of axons to other axons, to Schwann cells and to

basement membranes, etc. Further, several CAMs are transiently induced in Schwann cells following PAF injury and, once PAF regrowth is achieved and axonal contact re-established, Schwann cell phenotype reverts to normal and their induction is terminated (Bunge, 1993; Fazeli and Walsh, 1996; Kolodkin, 1996). That is, the expression of CAMs may be 'axon damage-dependent'. One example of such a potentially-important CAM is provided by the recently-identified 'ninjurin', which is induced in Schwann cells (and DRG neurones) upon nerve injury and which promotes axonal growth (Araki and Milbrandt, 1996). Interestingly, CAMs are also transiently induced in the DH upon PAF transection (Bonfanti et al., 1996) suggesting that they may also play a role in central processes of regrowth. CAMs also play an important role in facilitating the infiltration of injured nerves-and the adhesion to damaged PAFs-of monocyte-derived macrophages. These cells are key participants in the phagocytosis of the myelin of injured PAFs (Wallerian degeneration), a process which anticipates regenerative processes of injured PAFs (Brown et al., 1997) (Section 11.3.4.1).

In addition to their structural roles, extracellular matrix molecules such as fibronectin and, in particular, lamanin, may fulfil a more direct role in permitting the regeneration and navigation of injured PAFs following Wallerian degeneration (Cowen and Gavazzi, 1998; Diamond and Springer, 1994; Luckenbill-Edds, 1997). Certain actions of lamanin appear, interestingly, to reflect a modulation of Ca^{2+} influx and $[Ca^{2+}]_i$ levels in growth cones (Agius and Cochard, 1998; Kuhn et al., 1998). Indeed, via interactions with GAP 43, calmodulin and various kinases, and in response to laminin and multiple types of stimuli, alterations in the levels of $[Ca^{2+}]_i$ play an important and diverse role in controlling the morphology and behaviour of the growth cones of regenerating injured PAFs (Ghosh and Greenberg, 1995; Gu and Spitzer, 1995; Komuro and Rakic, 1996; Sjaastad and Nelson, 1997) (Sections 8.2.2.5 and 11.3.1).

11.3.4.5. NO and neuropeptides

NO justifies brief mention inasmuch as its synthesis in PAFs is enhanced by their injury (Vergé et al., 1992b; Xu and Wiesenfeld-Hallin, 1997), and since it plays an important role both centrally and peripherally in the modulation of nociception (Sections 3.2.8.3 and 7.5.2.5). Schwann cells provide an important source of NO which, by improving local blood flow and other actions, may indirectly improve the immediate environment for PAF regeneration. However, NO participates in processes underlying (apoptotic) cell death and, by virtue of its complex interactions with PAFs, Schwann cells and other tissues types, as well as with molecules such as NGF, NO likely exerts both beneficial and deleterious effects as concerns PAF recovery (Melino and Bernassola, 1997; Minghetti and Levi, 1998; Nishio and Watanabe, 1998; Zochodne et al., 1997; Yonehara et al., 1997).

As discussed in Section 8.2.5, the gene expression of many neuropeptides, such as VIP, PACAP, GAL and NPY, is modified in damaged PAFs—as well as in sympathetic fibres (Jonakait, 1993). These neuropeptides exert a diversity of peripheral trophic actions which likely play an important role in creating an environment favourable for PAF regeneration (Hökfelt *et al.*, 1994) (Section 3.2.6).

11.3.4.6. The interactive nature of regenerative mechanisms

To summarize, following transection of PAFs, peripheral nerve regeneration is probably indispensable for the subsequent regrowth and connectivity of their atrophied, central branches. Ultimately, both peripheral and central processes of regeneration play a key role in determining those changes in nociception which ensue upon PAF damage. Their further characterization should, thus, provide important insights into the mechanisms underlying neuropathic pain and its potential management. In this regard, many novel molecules, both intrinsic and extrinsic to injured PAFs, likely await discovery (Ambron and Walters, 1996; Camborieux et al., 1998; Fu and Gordon, 1997; Kotzbauer et al., 1996; Kolodkin, 1996; Nieto, 1996; Reinhardt et al., 1994). Notably, recent evidence suggests that cooperative and convergent, mechanisms triggered by a multiplicity of factors, including neurotrophins, GAP 43, cytokines, CAMs, extracellular matrix molecules and neuropeptides may be of importance in determining the regeneration and functional activity of damaged PAFs (Diversé-Pierluissi et al., 1997; Rozenburt, 1985; Yao et al., 1997).

Notwithstanding the diversity of above-discussed mechanisms encouraging the peripheral regeneration of damaged PAFs, they do not generally regain their original configuration in (cutaneous) tissue. Rather, an alteration of their topography is associated with a reorganization of the RFs of DH neurones which become large and diffuse. These expanded RFs eventually coalesce into smaller, contiguous units and this process of central compensation may reflect a role of spinal NMDA receptors in strengthening synaptic input from specific areas of reinnervated tissue (Lewin et al., 1994a). Further, such a retuning of DH synapses has been implicated in the reconstruction of cortical, somatosensory maps following peripheral nerve damage (Kaas, 1991). These findings further underpin the key role of NMDA receptors in the plastic response of the DH to PAF injury (Section 10.3).

11.4. Reorganization of PAF Input to the DH Following Injury: Overview of Functional Consequences for Nociceptive Transmission

The arrival of foreign, large calibre $A\beta$ fibres in laminae II following PAF injury (Section 11.2) may have profound functional consequences inasmuch as this organizational change is compounded by the transformation of their phenotype to a SP-synthesizing mode (Marchand *et al.*, 1994; Neumann *et al.*, 1996) (Section 8.2.5). Since damaged $A\beta$ fibres continue to produce EAAs, they now possess two key elements underlying the induction of sensitization in DH neurones (Section 10.3.2.1). Stimulation of $A\beta$ fibres will therefore, strengthen synaptic signalling and encourage response amplification at sensitized NS (and WDR) neurones bearing NMDA, AMPA, $mGlu_1$ and (up-regulated) NK_1 receptors in superficial DH laminae (Section 11.2). Further, the stimulation (or spontaneous ectopic firing) of A β fibres will be misinterpreted as noxious. This abnormal pattern of signalling by $A\beta$ fibres likely contributes to both hyperalgesia and, most importantly, mechanical allodynia following PAF damage (Woolf and Doubell, 1994; Ma and Woolf, 1996b). Indeed, Molander et al. (1992) have shown that stimulation of A β fibres can elicit c-fos expression in the DH following sciatic nerve damage, an effect normally seen only for C fibres. Other factors may also intervene in the enhancement of nociceptive transmission by errant A β fibres. Thus, their re-routing to superficial laminae may allow them to 'escape' presynaptic inhibitory controls from ININs and descending pathways (Laird and Bennett, 1992; Wall, 1995). Further, following their re-orientation from deeper to superficial laminae, $A\beta$ fibres themselves will no longer be able to stimulate ININs, thereby contributing to a loss of inhibitory ININ tone and aggravating WDR neuronal sensitization in deep DH laminae (Section 10.7) (Fig. 2). In addition, there is limited evidence that regenerating, damaged C fibres may expand into deeper DH laminae in which case, they may also contribute to patterns of abnormal signalling and hyperalgesic states (Section 11.2).

11.5. Peripheral and Central Sprouting of Intact PAFs

11.5.1. Sprouting of Intact Collaterals into Denervated, Peripheral Tissue: A Role for NGF

As mentioned in Section 11.3.4.2, the regeneration of damaged, sensory C fibres may not imperatively depend upon the availability of NGF (Diamond et al., 1987, 1992b). However, developmentally, NGF augments the density of peripheral, thermosensitive nociceptors (Lewin and Mendell, 1994) and, under certain conditions, NGF plays a role in the reinnervation of denervated skin by adjacent, intact, cutaneous C fibres (Goodness et al., 1997). This process can be reversibly interrupted by administration of antibodies against NGF (Mearow and Kril, 1995; Ro et al., 1996). Indeed, there is an increase in levels of mRNA encoding NGF in denervated skin together with an induction of mRNA encoding TRK A and p75^{NTR} receptors—as well as GAP 43-in the peripheral branches of sprouting neurones (Leslie et al., 1995; Mearow et al., 1994; Mearow and Kril, 1995). An elevation in the density of TRK A receptors on non-neuronal cells in the skin has also been documented (Terenghi et al., 1997). These observations collectively suggest that the collateral, *peripheral* sprouting of *intact* C fibres, in contrast to the regeneration of their damaged neighbours-and in analogy to sympathetic neurone sprouting-is markedly NGF-dependent (Davis et al., 1997a; Diamond et al., 1987, 1992a,b; Kawaja et al., 1997; Miller et al., 1994).

An important question arises at to whether the sprouting of intact C fibres may be involved in changes in nociception. As concerns areas of denervated skin, there are indications from studies of sciatic ligation and the subsequent sprouting of adjacent, saphenous nerves that collaterals from the latter play a role in the accompanying hyperalgesia and allodynia. However, their contribution appears to be time- and response-dependent and may depend upon a continuing, sensitizing input to the DH from the damaged sciatic nerve itself (Ro and Jacobs, 1993; Sotgiu et al., 1996; Tal and Bennett, 1994). A related question concerns the 'extra-territorial' hyperalgesia and allodynia evoked from the adjacent area of skin innervated by the intact saphenous nerve itself (Tal and Bennett, 1994). Since saphenous and sciatic nerves are subserved by largely separate spinal ganglia, mechanisms of inter-nerve communication cannot be involved (Section 8.2.3) and it appears that capsaicin-sensitive fibres of the saphenous nerve mediate this increased nociception from their own territory (Mansikka and Pertovaara, 1997b; Sotgiu and Biella, 1997; Tal and Bennett, 1994). While the underlying mechanisms remain unclear, there is no evidence for supersensitivity of saphenous PAFs per se. It is, thus, possible that widespread, intersegmental mechanisms of disinhibition and sensitization in the DH (Sections 9.1.4.2, 10.3 and 10.6) are triggered via damage to the sciatic nerve. This may lead to either:

- 1. secondary alterations in the activity of saphenous nerve via antidromic mechanisms described in Section 9.1.4.2;
- an increase in the responsiveness of WDR neurones in the DH receiving nociceptive input from the saphenous nerve (Mansikka and Pertovaara, 1997b; Sang *et al.*, 1996; Sotgiu and Biella, 1997).

These observations may, thus, provide a special case of the phenomenon of 'referred' pain discussed in Section 9.1.4, and they likewise implicate central processes of sensitization in the DH.

As indicated in Section 11.5.2, it is possible that changes in nociception (allodynia) from the region bordering the sciatic nerve-denervated area involve the collateral, *central* sprouting of saphenous $A\beta$ afferents in the L4 segment of the DH (Doubell *et al.*, 1997).

Interestingly, Leslie *et al.* (1995) reported that the increase in NGF levels evoked by inflammation upregulates GAP 43 in the peripheral terminals of C fibres. This observation correlates well with a role of NGF/GAP 43 in the collateral sprouting of *intact* nerves into denervated skin (Section 11.5) and raises the possibility of a reorganization of the C fibre innervation of inflamed tissue.

11.5.2. Collateral Sprouting of Intact PAFs to the DH

The reorganization of synaptic architecture in the DH following peripheral axotomy appears *primarily* to reflect central re-innervation by *damaged* axons (those with regenerating peripheral branches). The collateral sprouting ('reactive synaptogenesis') of intact PAFs has long been considered rare (Chong *et al.*, 1994; Coggeshall, 1994; Doubell *et al.*, 1997;

Knyihar-Csillik and Török, 1989; Knyihar-Csillik et al., 1990; McMahon and Kett-White, 1991; Pubols and Bowen, 1988). This contention is largely based on early studies in which the peripheral regeneration of surgically-destroyed nerves was artificially prevented. Under these conditions, the virtual disappearance of PAF markers, such as fluoride-resistant acid phosphatase, was indefinitely maintained (Devor and Claman, 1980). In line with these findings, other studies indicated that the loss of DH pools of SP following peripheral nerve injury was permanent, likewise indicating a lack of collateral sprouting, at least of C fibres (Shehab and Atkinson, 1986a,b; Barbut et al., 1981). Further, the 'priming' or 'conditioning' of PAFs by injury to their peripheral branches is, generally, necessary for the triggering of their sprouting into the DH territory of an adjacent nerve (McMahon and Kett-White, 1991: Pubols and Bowen, 1988).

Other studies have, nevertheless, raised the possibility that a subpopulation of *non*-injured nerves may sprout into DH regions vacated following injury to an *neighbouring* nerve (Coggeshall, 1994; Doubell *et al.*, 1997).

- 1. Neurotoxic (intraneural pronase) lesions of PAFs initiated the collateral sprouting of *undamaged*, adjacent nerves to regions outside their normal, somatotropic DH territories in certain—though not all—studies (LaMotte *et al.*, 1989; LaMotte and Kapadia, 1993).
- 2. Capsaicin-induced destruction of C fibres in the sciatic nerve elicits the (non-collateral) sprouting of A β fibres into lamina II regions vacated by degenerating C fibres (Mannion *et al.*, 1996).

This observation suggests that 'priming' of $A\beta$ fibres may *not* be imperative for their central sprouting. Indeed, Doubell *et al.* (1997) recently showed that, following injury to the sciatic nerve, a subpopulation of saphenous, myelinated $A\beta$ fibres sprouts, in a dorsoventral plane, into the lateral part of lamina II which is normally occupied by sciatic terminals. Assuming that these $A\beta$ fibres contact those NS cells in lamina II which are normally targetted by C fibres input, this may contribute to the hypersensitivity/allodynia seen at the border of denervated skin (Section 11.5.1).

Genuine collateral sprouting of intact PAFs into DH territories of adjacent nerves appears, therefore, to be limited and uncommon. Nevertheless, these observations indicate that intact, myelinated PAFs do have the capacity to sprout following damage to C fibres or adjacent nerves. The induction of intracellular factors required for PAF sprouting may not, thus, be critically dependent upon their damage. Indeed, the nature of those intrinsic and extrinsic mechanisms which permit the central sprouting of intact as compared to injured PAFs would be of interest to delineate. The sprouting of descending serotoninergic neurones following PAF damage (Marlier et al., 1992) (Section 10.8.3) further indicates that injury to neurones themselves may not be critical for the initiation of processes underlying their regrowth. Whether intact C fibres sprout collaterally in the DH remains to be determined.

11.6. Surgical and Chemical Destruction of PAF Input into the DH

11.6.1. Surgical Interruption of PAF Input to the DH via Lesions Proximal to the DRG

There are several important differences between the effects of distal, peripheral nerve axotomy and those of PAF transection proximal to the DRG (that is, on the DH side). Thus, transection of PAFs proximal to the DRG leads neither to degeneration of cells bodies in the DRG, nor to growth of their peripheral branches. Moreover, sectioned, central terminals immediately disappear and the regenerating central processes of their DRG cells generally do not re-gain access to the DH. This likely reflects a variety of external and internal factors including a relative lack of GAP 43, which is necessary for sustained, long axon growth (Chong et al., 1994; Ramakers et al., 1997). Further, the barrier of glial cells in the dorsal re-entry zone of the DH physically prevents access of PAFs to the DH. Indeed, in contrast to neurotrophin- and CAM-rich Schwann cells (the peripheral environment of regenerating PAFs) (Section 11.3.4.1), DH-localized oligodendrocytes and astrocytes are hostile to axonal re-entry and regrowth due to their expression and display of molecules inhibitory to growth cones (Golding et al., 1997; Kolodkin, 1996; Kozlova et al., 1995; Lekan et al., 1997; Olson, 1997; Schwab, 1990, 1996; Stensaas et al., 1987; Tessler et al., 1988; Weiss et al., 1996). Under these conditions of complete and rapid loss of PAF input to the DH, then, the contribution of structural changes to pain differ from those provoked by *peripheral* nerve lesions.

Observations of PAF markers, such as CGRP and fluoride-resistant acid phosphatase, indicate that, under these conditions, the collateral sprouting of *intact*, unmyelinated PAFs and myelinated $A\beta$ fibres into deafferented, superficial DH regions does occur (Lekan et al., 1997; LaMotte et al., 1989; McMahon and Kett-White, 1991; McNeill et al., 1990: McNeill and Hulsebosch, 1987). Correspondingly, a reappearance of lost synapses has been documented (Chung et al., 1989). Thus, the occurrence of spontaneous pain and the continued (or even increased) responsiveness of DH neurones to high intensity, sensory input despite the loss of original PAF input may be explained by several mechanisms:

- 1. sensitization of DH neurones triggered by the transient and intense activation of PAFs upon their injury;
- a reduction in ININ tone reflecting the loss of facilitatory PAF input and their partial excitotoxic degeneration: this leads to chronically-disinhibited, spontaneously-active and hyperexcited neuronal circuits in the DH;
- 3. supraspinal mechanisms of sensitization and an enhancement of descending facilitation; and
- 4. a novel source of sensory PAF input to denervated DH neurones from collaterally sprouted intact nerves.

The mechanisms underlying the collateral sprouting of intact neurones under these conditions remain to be examined (Fu and Gordon, 1997; Janscó and Lawson, 1990; Kapadia and LaMotte, 1987; LaMotte *et al.*, 1982). As concerns the increased sensitivity of DH neurones, it is possible that an elevation in the DH density of NK₁ and CGRP receptors is involved (Coggeshall and Carlton, 1997). Further, an increase in the density and Ca²⁺-permeability of AMPA receptors in the DH may also be implicated in the enhanced sensitivity of DH neurones (Section 10.3.2.1).

11.6.2. Chemical Destruction of Small Calibre C Fibres by Capsaicin

As discussed in Section 9.1, the acute, cutaneous application of capsaicin sensitizes nocisponsive neurones in the DH and provokes a $A\beta$ fibre-mediated mechanical allodynia in a manner similar to that seen for painful states due to inflammation of PAF injury.

The permanent elimination of small calibre C fibres by developmental treatment with capsaicin leads to a variety of neuroplastic changes, including an increase in the spontaneous activity and RF size of DH neurones. These effects can be inhibited by antagonists at NMDA receptors indicating their involvement in mediating DH sensitization under these conditions (Chiang et al., 1997). The destruction of C fibres by capsaicin results in a pronounced decrease in thermal sensibility and a loss of peripheral NI. Further, the administration of capsaicin markedly depletes TRK A and vanilloid receptors and reduces SP and CGRP levels in nocisponsive, small calibre PAFs in parallel with an induction of GAL, VIP and NO synthase (Farkas-Szallasi et al., 1995; Kashiba et al., 1997; Xu et al., 1997c; Vizzard et al., 1995). These changes resemble those seen upon peripheral nerve damage (Hökfelt et al., 1994) (Section 8.2.5). An induction of GAP 43 synthesis has also been observed, as well as a (NGF-reversible) reduction in axonal flow, effects likewise common to peripheral nerve damage (Kashiba et al., 1997; Taylor et al., 1985) (Sections 8.2.5 and 11.3.2). Further, NGF can protect against and restore the changes in neuropeptide content and function provoked by capsaicin treatment (Donnerer et al., 1996). The marked and sustained disappearance of C fibre terminals in the DH following capsaicin treatment suggests that they do not recover. However, in analogy to the surgical destruction of PAFs, the capsaicin-induced lesioning of C fibres may be a sufficient stimulus to trigger $A\beta$ fibre sprouting into lamina II (Mannion et al., 1996) (Section 11.2).

Notwithstanding these similarities to peripheral nerve injury, there are several distinctive differences between capsaicin treatment and surgical damage to PAFs. For example, capsaicin administration does not modify the levels of somatostatin, which is contained in a population of small calibre PAFs differing to those expressing VIP, GAL or CGRP (Farkas-Szallasi *et al.*, 1995, 1996; Hökfelt *et al.*, 1994; Kashiba *et al.*, 1996, 1997). Additional comparisons of the functional and neurochemical consequences of selective, neurotoxic destruction of C fibres, as compared to surgical axotomy of peripheral nerves (containing all classes of PAF) should yield novel insights into mechanisms controlling PAF phenotype, regrowth and the modulation of nociception.

12. NOCICEPTIVE AND NEUROPATHIC PAIN: PROCESSES OCCURRING IN SUPRASPINAL CENTRES

12.1. Sensitization of Neurones in the S I and Thalamus

Central adaptive responses to nociceptive and neuropathic pain, involving alterations in neuronal responsiveness together with a reorganization of patterns of synaptic connectivity, are not restricted to the DH. They may likewise occur in supraspinal centres (Besson et al., 1995; Castro-Alamancos, 1997: Darian-Smith and Gilbert, 1995; Faggin et al., 1997; Guilbaud and Benoist, 1995; Guilbaud et al., 1994; Pettit and Schwark, 1996; Sherman and Guillery, 1996; Zhang and Rowe, 1997). As yet, data are limited but there is evidence that such changes occur both at the cortical level as well as in several thalamic nuclei, including the 'ventrobasal complex' (VPL/VPM), the nucleus centrolateralis of the intralaminar thalamus and several components of the medial thalamus-although no data appear, as yet, to be available for posterior thalamic nuclei (Guilbaud et al., 1994; Ralston et al., 1995). For example, the induction of acute hind-limb inflammation by carrageenin increases the responsiveness of S I and thalamic neurones to stimulation of the inflamed joint in parallel with an increased responsiveness of DH neurones and of joint nociceptors. These changes evolve in synchrony with the development of hyperalgesia and can be reversed by administration of anti-inflammatory agents into the treated joint (Guilbaud et al., 1986, 1994; Hoheisel and Mense, 1989; Mense, 1993). Alterations in peripheral sensitivity are, thus, reproduced in supraspinal centres responsible for the sensation of pain and, in line with such an onward transmission of changes in sensitivity, it has been shown that an increased excitability of thalamic neurones is accompanied by a potentiation in responses at the cortical level (Lee and Ebner, 1992). Further, the existence of bi-directional circuits interconnecting the thalamus and cortex allows for the reciprocal modulation of their activity (Eaton and Salt, 1995; Gil and Amitai, 1996a,b; Nothias et al., 1988).

12.2. Adaptive Changes in Higher Centres

12.2.1. Distinctive Changes in Supraspinal Structures

The above-mentioned findings might be considered to imply a 'domino-like' phenomenon whereby peripheral increases in sensitivity are progressively transferred to higher levels of processing in a comparatively unmodulated fashion. However, the sequential, supraspinal transfer of nociception should *not* be regarded as a purely passive process inasmuch as several studies have provided evidence that supraspinal changes in nociceptive processing, in particular in the S I, may occur *differentially* to those in downstream structures (Dong and Chudler, 1995; Guilbaud et al., 1994; Guilbaud and Benoist, 1995; Sherman et al., 1997a).

12.2.2. Adaptive Changes at the Cortical Level (S I)

Further evidence for changes in higher centres has been obtained in studies of the long-term effects of arthritic inflammation of the joint, and of peripheral nerve damage (Guilbaud et al., 1994; Schaible and Grubb, 1993; Schaible and Schmidt, 1988). Thus, in a neuropathic model of sciatic ligation (2 weeks following injury), the responsiveness of thalamic and S I neurones to mechanical and cold thermal stimuli of the paw was amplified (cf allodynia) and the S I displayed a reorganization of somatic input. Moreover, there was an increased participation from the saphenous nerve-though this change may have been secondary to an abnormal position or gait of the paw rather than reflecting the peripheral or central sprouting of saphenous nerves (Guilbaud et al., 1990, 1992; Guilbaud and Benoist, 1995) (Section 11.5.1).

There is evidence that the superimposition of distinctive, adaptive changes at higher levels can either accentuate, or attenuate, changes in neuronal responsivity. Notably, one class of S I neurone has been described in the primate for which the adaptive (sensitized) response to sustained noxious thermal stimulation is less marked than in either the thalamus or DH, indicating the involvement of counteradaptive processes (Kenshalo et al., 1979, 1980; Kenshalo and Isensee, 1983; Dong and Chudler, 1995). Instructive results were also acquired in a model of chronic (3 weeks) arthritic inflammation of a single hind-paw. In contrast to normal rats, mild mechanical stimulation of the joints and contiguous skin activated neurones in both thalamic and S I neurones (Dostrovsky and Guilbaud, 1990; Lamour et al., 1983; Guilbaud et al., 1994; Schaible and Grubb, 1993). Further, in the studies of Lamour et al. (1983) several excitatory changes detected in the thalamus, including paroxysmal discharges, responsiveness to brushing of the skin, and after-discharges exceeding the duration of stimulation were not detected in S I of the cortex. These observations also suggest the overlay of additional processes at the S I cortical level moderating the intensity and extent of sensitization. In support of the contention that cortical adaptivity does not merely mimic antecedent changes in the thalamus, there was a striking displacement of the laminar, topographic organization of the responsiveness of NS and WDR cortical neurones. Whereas these were normally encountered in the infragranular layer, they became apparent in lamina V which otherwise contains NON-N neurones (Lamour et al., 1983). Further, many S I neurones developed bilateral RFs as compared to the contralateral RFs seen in normal animals. Corticospinal neurones have also been shown to display functional and structural changes upon PAF damage (Tseng, 1996) and a functional reorganisation of the cortex has been documented following injury to PAFs mediating other modes of sensory input (Darian-Smith and Gilbert, 1995; Pons et al., 1991; Zhang and Rowe, 1997).

The above findings strongly suggest that nociceptive input may initiate distinctive patterns of cortical adaptation, possibly involving novel patterns of synaptic strength or connectivity, in analogy to the marked influence of other modes of sensory input upon cortical function (Darian-Smith and Gilbert, 1995; Pons et al., 1991; see Zhang and Rowe, 1997, Section 14). Further, they suggest that the further elaboration of adaptive changes at the cortical level can lead to a *reduction* in alterations in sensitivity as compared to the DH and thalamus (Hsu and Shyu, 1997). Nevertheless, a recent discussion of comparative changes in the thalamus and S I in response to noxious stimulation (Guilbaud and Benoist, 1995) has pointed out that some changes may be more pronounced in the S I than in the thalamus, suggesting a further amplification of increases in sensitivity. Indeed, the involvement of EAAs and NMDA receptors in the transfer of nociceptive information via thalamocortical and intracortical networks provides a mechanistic basis for processes of rapid sensitization and more prolonged forms of synaptic plasticity at the cortical level (Section 12.3.3).

To summarize, the question of comparative changes in neuronal responsiveness at the cortical and thalamic as compared to spinal level requires further examination. Nevertheless, it is evident that the S I, thalamus and other higher centres do *not* merely behave as passive recipients and relayers of information from the DH. Rather, they are themselves involved in the further integration of adaptive, neuronal changes in acute and chronic painful states due to either inflammation or peripheral nerve damage (Sherman and Guillery, 1996).

12.2.3. Adaptive Changes at the Thalamic Level

In this light, there is also evidence for distinctive, adaptive changes at the level of the thalamus in response to alterations of sensory input (Faggin et al., 1997; Guilbaud et al., 1989b; Nicolelis et al., 1993; Sherman et al., 1997a,b). For example, as discussed in Sections 9.2.1 and 13.4, a disruption of the interaction between thermosensory and cold channels in thalamic posterior nuclei may contribute to cold allodynia (Craig et al., 1996). Indeed, such forms of 'thalamic' pain may reflect organizational (possibly structural) and adaptive changes triggered by damage to the thalamus itself, to the STT, to other ascending pathways or to PAFs (Ralston et al., 1995) (Section 13). In support of the argument that changes in thalamic function triggered at the level of PAFs are expressed in a more complex fashion than can be accounted for by simple extrapolation of alterations in the activity of DH neurones, Nicolelis et al. (1993) showed that the anaesthetic block of tactile, mechanical information can immediately elicit distinctive, spatiotemporal alterations in the RFs and responsiveness of thalamic neurones. Further, Sherman et al. (1997a,a) have reported alterations in the response properties and RFs of thalamic neurones upon blockade of glycinergic ININ activity in the DH, supporting a role of thalamic (and cortical) changes in the mediation of allodynia. More persistent, and perhaps irreversible, alterations in thalamic

106

function may ensue upon prolonged alterations in peripheral nociceptive input. In view of the existence of somatovisceral convergence at the thalamic level, such changes may be pertinent not only to pain resulting from damage to the thalamus itself and from PAF injury, but also to the phenomenon of referred pain (Apkarian *et al.*, 1995) (Section 9.1.4.2).

12.2.4. Other Structures

To summarize, distinctive patterns of adaptive changes in the response to acute and prolonged nociceptive and neuropathic pain occur in both the thalamic and S I. Further, adaptive processes in the cortex do not simply reproduce adaptive changes in subcortical structures (Faggin et al., 1997). The occurrence of such distinctive, supraspinal changes is consistent with a role of EAAs/NMDA receptors as transmitters/receptors in ascending projections to the thalamus, in thalamocortical projections and in intracortical pathways, as well as with a modulatory influence of GABAergic mechanisms upon thalamic and cortical sensory processing (Gil and Amitai, 1996b) (Sections 6.1.1, 12.3 and 14). Nevertheless, the underlying mechanisms require further elucidation. Further, an extension of such observations to other tissues would be of interest, in particular to those corresponding to the affective axis of nociceptive processing and sensation-the VPI/VMpo of the thalamus and the S II, anterior cingulate and insular cortices. Along these lines, Matsumoto et al. (1996) recently reported increases in spontaneous activity and responsiveness to mechanical stimulation in PBN neurones following chronic inflammation. This suggests that adaptive changes do, indeed, occur in supraspinal structures other than the thalamus.

12.3. Roles of Neurotransmitters in Adaptive Changes in Higher Centres

12.3.1. GABA and EAAs

Many neurotransmitters involved in the cerebral processing of nociceptive information, and in the above-described patterns of adaptive changes in supraspinal centres, likely remain to be identified. However, based on observations in the DH (Sections 10.3 and 10.6), it is not unreasonable to suspect an involvement of GABA and NMDA receptors.

12.3.2. GABAergic Mechanisms

The importance of GABAergic ININs in suppressing neuronal excitability (Zhang *et al.*, 1997d) and in negatively controlling nociceptive output from both thalamic relay and cortical neurones should be underlined (Kao and Coulter, 1997; Kapur *et al.*, 1997; Kim *et al.*, 1997a; Paré *et al.*, 1991; Roberts *et al.*, 1992; Oliveras and Montagne-Clavel, 1994; Alloway *et al.*, 1989) (Section 6.2.3.). Indeed, as described in Section 13.3., the interruption of GABAergic transmission in the cortex or thalamus can provoke pain and expand RFs (Dykes *et al.*, 1984; Hicks *et al.*, 1986), while antagonism of local GABAergic receptors increases the RF size of neurones in the gracile nucleus (Berkley, 1997). Further,

a reduction in the inhibitory control of thalamic input to the cortex by cortical ININs may trigger synchronized, widely-dispersed, excited, seizure-like circuits (Contreras *et al.*, 1997).

In analogy to processes occurring in the DH (Section 10.6), thus, a disruption of GABAergic transmission in the thalamus or other cerebral structures may disinhibit the transfer and modulation of nociceptive information and contribute to painful states. A degeneration of GABAergic ININs might, in theory, be provoked by an excessive, excitoxic release of EAAs from an overactive STT, other sources of nociceptive input or corticothalamic pathways (Ralston et al., 1995). Alternatively, one might argue that, in an opposite fashion, a loss of an excitatory input from the STT onto thalamic GABAergic ININs following peripheral nerve damage may result in a temporary and reversible diminution in their activity (Section 10.6.2). As yet, no concrete data are available in this regard (Ralston et al., 1995).

Indeed, it remains to be determined whether events in peripheral nerves can transiently or permanently disable the activity of GABAergic-or other classes of-ININ in the thalamus and/or cortex. Nevertheless, thalamocortical pathways involved in the synchronization of neocortical activity, as well as intrinsic cortical and thalamic networks themselves, may be involved in the generation of burstlike activity following breakdown of GABAergic inhibition. It might be speculated, then, that, in analogy to the DH (Sections 9.1.4.2 and 10.6), neuronal disinhibition provoked by the discrete or generalized loss of supraspinal GABAergic tone may set off reverbatory firing patterns in cerebral nociceptive circuits (Ben-Ari et al., 1997; Bramham et al., 1996; Berzaghi et al., 1995; Castro-Alamancos and Connors, 1996; Crick and Koch, 1998; Forti et al., 1997; Witte et al., 1997; Zhuo et al., 1997b-though see Liu et al., 1991) (Section 13.3).

12.3.3. EAAs, NMDA and MGlu Receptors

Both NMDA and group I MGlu receptors are involved in the induction and maintenance of processes of sensitization in the DH (Section 10.3). Further, via the induction of LTP and other processes, NMDA-and, under certain conditions, group I mGlu-receptors play a key role in modulating synaptic plasticity in the hippocampus and other cerebral structures (Bliss and Collingridge, 1993; Bortolotto et al., 1994; Huber et al., 1998; Martin and Morris, 1997; McEachern and Shaw, 1996: Vickery et al., 1997) (Section 10.2.1). These observations suggest that NMDA and, possibly, group I mGlu receptors may mediate adaptive events in the thalamus and other higher centres underlying prolonged, painful states. In line with this contention, NMDA (and group I mGlu) receptors fulfil a crucial role in the excitation of thalamic neurones: by

- 1. direct nociceptive input from the STT and other ascending pathways;
- 2. corticothalamic circuits; and
- nociceptive input received via other regions involved in the integration of nociceptive information, such as the PAG (Dougherty et al., 1996;

Eaton and Salt, 1995, 1996; Kao and Coulter, 1997; Salt and Eaton, 1995; Vaccarino *et al.*, 1997).

In addition, NMDA receptors play a major role in thalamocortical transmission of sensory information (Isaac *et al.*, 1997), and the stimulation of thalamocortical afferents can evoke a NMDA-receptor dependent form of LTP in the cortex (Castro-Alamancos, 1997; Castro-Alamancos and Connors, 1996) (Section 6.2.3).

The above observations are consistent with the notion that NMDA receptors play a key role in sensitizing neurones in the thalamus, cortex and other supraspinal structures under conditions of excessive or prolonged nociceptive stimulation (Conti and Hicks, 1996; Salt *et al.*, 1995; Ralston *et al.*, 1995). Indeed, there is direct evidence that NMDA receptors in the thalamus are involved in the development and maintenance of inflammatory hyperalgesia, possibly via mechanisms involving the induction of NO, c-fos or nuclear transcription factors (Bester *et al.*, 1997; Do *et al.*, 1994; Kolhekar *et al.*, 1997; Lin *et al.*, 1997c; Redburn and Leah, 1997). Indeed, NO may itself modulate responses to sensory input and EAAs in the thalamus (Shaw and Salt, 1997).

Administration of COX inhibitors to the PAG elicits antinociception, an action reflecting the activation of descending pathways to the DH (Vanegas *et al.*, 1997) (Section 5). Further, activation of NMDA receptors can increase the cerebral production of PGs via the activation of COX 2 (Miettinen *et al.*, 1997) (Section 3.2.8.4). This mechanism is triggered in cortical neurones by 'spreading depression', a process implicated in the induction of migraine blockade (Sections 4.7 and 7.9). These findings suggest a possible interplay between cerebral populations of NMDA receptor and PGs in processes underlying painful states, an issue requiring further evaluation.

12.3.4. Other Potential Mediators

As proposed in Section 14, the neurotrophin, BDNF, which displays reciprocal interactions with NMDA and GABAergic receptors, is an attractive candidate for a modulatory role in nociceptive transmission in the cortex and/other supraspinal regions (Berninger and Poo, 1996; Gwag *et al.*, 1993; Li *et al.*, 1994; Marty *et al.*, 1997). Interestingly, there is recent evidence that GAP 43, an intrinsic component of neuronal terminals involved in axonal growth and regeneration (Section 11.3.2), may be able to induce LTP (Namgung *et al.*, 1997). This suggests that it may play a more rapid and multifarious role in remodelling central synaptic transmission than has hitherto been assumed.

As discussed in Section 4.8, a possible role of cytokines, such as IL-1 β and TNF α , in the modulation of sensory processing in the cortex and other higher tissues must be evoked. For example, TNF α can enhance EAA-mediated ion currents and IL-6 modulates the response to sensory input of neurones in the S I region of the cortex (Grassi *et al.*, 1994; Oka *et al.*, 1995; Shin *et al.*, 1997). It is also likely

that neuropeptides are involved in adaptive processes in higher centres.

Data concerning plastic changes at higher levels, and the neurotransmitters and other factors involved in their mediation, clearly remain fragmentary and require considerable clarification.

13. PAIN DUE TO CNS DAMAGE: CENTRAL PAIN

13.1. The Pathological Origins of Central Pain

Central pain-the pain provoked by damage to (or a dysfunction of) the CNS itself-displays a diversity of pathologies. For example, strokeinduced ischaemic damage, multiple sclerosis, malignant disease and Parkinson's disease (Boivie, 1994: Boivie and Östeberg, 1995; Bowsher, 1996; Chulder and Dong, 1995). Onset of pain following injury may be immediate or delayed (months). Although there may be a 'hypoalgesia' to some stimulus modalities, spontaneous pain and abnormal sensations of temperature are characteristic, in particular cold allodynia. A dysfunction of the posterior region of the thalamus (VMpo nuclei) may play a crucial role in this regard (Sections 9.2.4 and 13.4). 'Thalamic' pain is not, nevertheless, synonymous with 'central' pain which, as implied by its multiple origins, does not invariably involve damage to the thalamus itself. For example, damage to the STT can result in deficits in nociceptive and thermosensory sensation, as well as pain (Casey et al., 1996a). Pain resulting from damage to the STT (which provides nociceptive input to the thalamus) might be considered as analogous to the pain caused by damage to PAFs (which transmit nociceptive information to the DH). Indeed, several potential mechanisms underlying central pain resemble those spinal events involved in the induction of pain due to peripheral nerve injury. Notably, in analogy to the reduction in GABAergic tone in the DH upon PAF injury (Section 10.6), a dysfunction of thalamic or cortical GABAergic inhibitory mechanisms, thereby disinhibiting local and remote nociceptive circuits, may be involved in the induction of central pain.

13.2. A Model of Pain Due to Spinal Damage: Loss of GABAergic Inhibition in the DH

A photochemically-induced, ischaemic lesion of the spinal cord elicits a mechanical allodynia resembling that associated with peripheral nerve injury (Hao et al., 1992, 1996a,b; Zhang et al., 1994a). Following this manipulation, WDR neurones in the DH display a prolonged and enhanced response to electrical or innocuous, mechanical stimulation, whereas their response to thermal stimuli is normal. This suggests that a perturbation of $A\beta$ fibre input and/or processing underlies the mechanical allodynia which can, further, be attenuated by AMPA antagonists. The GABA_B agonist, baclofen, also reverses both allodynia and the abnormal responsiveness of WDR neurones, suggesting that their enhanced excitability and the accompanying allodynia reflect a loss of GABAergic inhibition in the DH. In line with
this hypothesis, ischaemia has been shown to decrease both GABA levels in the spinal cord, and the number of GABAergic cells in laminae I and II (Zhang *et al.*, 1994a; Martiniak *et al.*, 1991). Interestingly, NMDA antagonists, such as dizolcipine, prevent these behavioural changes. Although this finding might be taken as evidence that an excitotoxic action of EAAs is involved in the damage of GABAergic ININs, the underlying mechanisms remain unclear. Further, the loss of GABAergic neurones is *transient* and they recover in parallel with the disappearance of allodynia, suggesting that they do *not* irreversibly degenerate (Hao *et al.*, 1996a; Zhang *et al.*, 1994a) (Section 10.6).

In a small population of lesioned animals, a more chronic mechanical allodynia develops. This is irresponsive to baclofen but naloxone-reversibly inhibited by the CCK_B antagonist, CI 988, which also enhances morphine antinociception. This provides a further parallel to peripheral nerve damage in suggesting that an upregulation of the activity of CCK-containing neurones in the DH—or PAF terminals—interferes with an endogenous, opioidergic tone inhibiting pain and allodynia (Xu *et al.*, 1994; Section 3.2.6). The inhibition of this mechanical allodynia by A_1 agonists provides a further parallel to models of PAF injury (Section 3.2.8.2) (Sjölund *et al.*, 1998).

13.3. Reduced GABAergic Transmission in the Thalamus and Cortex: Burst Firing and Reverbatory Circuits

Disinhibitory mechanisms involving a loss of GABAergic tone may play a more general role in the induction of central, painful states. Indeed, the GABA_A antagonist, picrotoxin, elicits pain-like behaviour upon local introduction both into the trigeminal nucleus (the cerebral homologue of the DH) and the 'ventrobasal complex' of the thalamus, wherein GABAergic mechanisms modulate the output of thalamocortical pathways running to the S I (McCormick, 1992; Roberts et al., 1992; Sakai et al., 1979; Oliveras and Montagne-Clavel, 1994; Tasker, 1990) (Section 12.3.2). The GABA_B antagonist, bicuculline, also alters RFs and neuronal responsiveness upon application into the thalamus (Hicks et al., 1986). Further, a lesion-induced, neuronal loss (presumably of ININs) in the thalamus has been shown to modify sensory responsiveness therein (Kayser et al., 1985).

These findings suggest that a dysregulation of GABAergic mechanisms in the thalamus may contribute to chronic pain states. Indeed, vascular lesions of the thalamus might result in a preferential loss of GABAergic ININs inasmuch as small ININs are particularly vulnerable to ischaemic damage triggered by activation of NMDA receptors coupled to increases in $[Ca^{2+}]_i$, NO and other mechanisms (Choi and Rothman, 1990; Lin *et al.*, 1997; Marcoux, 1994; Mattson and Mark, 1996; Solum *et al.*, 1997). In addition to damage to the thalamus itself, disruption of afferent input to the thalamus from the STT can elicit 'thalamic' pain (Boivie *et al.*, 1989; Bowsher, 1996). Thus, extrapolating from the hypothesized degeneration of GABAergic ININs

in the DH by the excitotoxic actions of EAAs released from injured PAFs (Section 10.6), one might speculate that a liberation of EAAs from intensely-activated, damaged STT fibres might similarly lead to a loss of GABAergic ININs in the thalamus. This possibility remains to be directly examined. An alternative hypothesis is that the STT physiologically stimulates GABAergic ININs. A loss of STT input following its damage might thus, result in a diminished GABAergic tone in the thalamus permitting the escape of nociceptive information transmitted by other pathways under GABAergic control (Section 12.3). In line with this possibility, the density of GABAergic dendrites is reduced in the thalamus following damage to the ML pathway (Ralston et al., 1995).

Ralston et al. (1995) have generated a specific hypothesis concerning GABAergic ININs which may account for the pain arising from vascular damage to the thalamus or from lesions of the STT. They suggest that GABAergic modulation of sensory input in the thalamus is more pronounced for innocuous, mechanical input originating in the PSDC and accessing the thalamus via the ML than for nociceptive input derived from the STT. Lesions of the STT, by liberating synaptic targets, encourage sprouting of ML neurones into vacant sites. As a consequence, they escape GABAergic control and attain nocisponsive neurones which they would not normally target. This hypothesis clearly resembles the sprouting of $A\beta$ fibres into lamina II of the DH following PAF injury (Section 11.2). In addition, it is conceivable that ML fibres mediating nociceptive information from the viscera may be involved in such organizational changes and contribute to spontaneous pain and hyperalgesia (Section 6.1).

Neurones in S I and S II are also under GABAergic control (Alloway et al., 1989) and application of picrotoxin to these regions provokes a pain-like behaviour which is accompanied by epileptic, electroencephalographic activity (Forti et al., 1997; Oliveras and Montagne-Clavel, 1996). In analogy, iontophoretic application of GABAergic antagonists to nocisponsive, cortical neurones increases their RF size and provokes burst firingthis pattern of discharge being critical for persistent, epileptic-like, excitability changes (Alloway et al., 1989; Forti et al., 1997; Dykes et al., 1984). The cortex also provides a reciprocal pathway to the thalamus, via the VMpo, which is likely subject to GABAergic inhibition. Thus, in analogy to the DH, a loss of GABAergic tone might set up a network of self-sustaining, excitatory and synchronized bursting circuits both intrinsic to and inter-connecting the thalamus and cortex and dispersing both ipsi- and contralaterally (Benardo, 1997; Ben-Ari et al., 1997; Bramham et al., 1996; Crick and Koch, 1998; Lytton et al., 1997; Parker et al., 1998; Plenz and Kitai, 1996; Steriade, 1998; Zhou et al., 1997b; Zhang et al., 1997d). Such changes would be analogous to epileptic seizures caused by a loss of ININ tone, and rhythymic, neuronal circuits triggered by ischaemic damage to the CNS (Berzaghi et al., 1995; Castro-Alamancos and Connors, 1997; Forti et al., 1997; Witte et al., 1997-though see Liu et al., 1991). These findings are of pertinence to clinical

observations of bursting-like patterns in the thalamus of patients with central pain (Lenz et al., 1989). In this light, further, the utility of anti-epileptic and anti-convulsant agents, such as carbamazepine, felbamate, gapabentin and valproate, in the treatment of central and other neuropathic pain syndromes is of relevance. Although their mechanisms of action are unclear, they may suppress excitation in bursting populations of intrinsic neurones and thalamocortical/corticothalamic networks via their Ca²⁺-/Na⁺channel blocking properties, strengthening of GABAergic tone and/or a reduction in GLU release (Fields et al., 1997; Gee et al., 1996; Hunter et al., 1997b; Shimoyama et al., 1997; Stanfa et al., 1997; Tasker, 1990; Boivie et al., 1989; Todorovic and Lingle, 1998).

Burst firing and the synchronous and cooperative activity of neural circuits may fulfil a *physiological* role in facilitating information transfer and response patterns in the cortex and other structures (Chergui *et al.*, 1997; Lisman, 1997; Lumer *et al.*, 1997). However, under *pathological* conditions, due to a loss of ININs, this may lead to excessively-active, sensitized circuits contributing to chronic pain (Chergui *et al.*, 1997; Crick and Koch, 1998; Lisman, 1997; Lumer *et al.*, 1997) (Sections 9.1.4.2 and 12.3.2).

13.4. Disinhibition of Thalamocortical Pathways by Thalamic Damage: Cold Allodynia

One frequent complication of stroke-induced thalamic damage is cold allodynia. There is evidence that the cold allodynia associated with secondary hyperalgesia and PAF damage reflects activation of cold-sensitive C rather than $A\beta$ fibres (Sections 3.1.1 and 9.2.4). Such C fibres are also involved in mediating the cold allodynia due to vascular lesions of the thalamus. Information concerning cutaneous, cold stimuli originating in lamina I NS neurones is transmitted to the thalamus via the dorsal STT which targets the MDvc of the medial thalamus, which itself projects to the anterior cingulate cortex (Sections 6.2.2 and 6.3) (Dostrovsky and Craig, 1996; Craig et al., 1996; Castro-Alamancos, 1997; Hsu and Shyu, 1997). On the other hand, thermosensory information pertaining to cold is mediated by $A\delta$ fibres running via the ventral STT to the VMpo of the posteriolateral thalamus, and therefrom to the insular cortex. Craig et al. (1996) have shown that a conduction block of $A\delta$ fibres transmitting cold stimuli can elicit a cold allodynia mediated via cold-sensitive C fibres. Further, the syndrome of cold allodynia due to thalamic damage is invariably associated with vascular lesions of the thalamus incorporating the VMpo but not the medial thalamus. Craig et al. (1996) have postulated that a disinhibition of this latter channel is involved. Thus, under normal circumstances, $A\delta$ fibremediated thermosensory input to the VMpo inhibits activation of the medial thalamus by C fibremediated cold input. However, this modulatory influence is eliminated by lesion-induced damage to the VMpo. This possibility is supported by an acute, clinical model of cold allodynia in which a specific activation of the anterior cingulate cortex is seen

(Craig *et al.*, 1996). Interestingly, as pointed out in Section 6.2.2, the VMpo forms part of an inhibitory corticothalamic feedback loop, the dysregulation of which by damage to the VMpo, might also contribute to persistent cold allodynia (Nothias *et al.*, 1988).

Whether a comparable disruption of thalamic integration between thermosensory and pain signalling underlies cold allodvnia due to events at the level of peripheral nerves would be of interest to determine. Although no pertinent data are, as yet, available, it is conceivable that the overactivation of the thalamo-cingulate pathway in cold allodynia involves the disabling of GABAergic or other types of ININ interconnecting the VMpo and the medial thalamus. In this light, it is of note that the novel anticonvulsants, lamotrigine, felbamate and gabapentin, inhibit cold allodvnia in rodents. Their effects likely involve an (indirect) potentiation of GABAergic transmission, although other actions, such as an inhibition of GLU release and blockade of Ca² channels may also be involved (Section 8.2.2).

13.5. Mechanisms Involved in the Response of the Thalamus and other CNS Tissues to Injury

As for peripheral nerve injury (Section 11), important insights into processes underlying central pain will likely be gained by an improved understanding of mechanisms underlying both neuronal damage and neuronal recovery. A brief consideration of certain of the factors involved in the response to thalamic injury is, thus, justified (Aguayo, 1985; Ambron and Walters, 1996; Marty *et al.*, 1994; Minghetti and Levi, 1998; Olson, 1997; Schwab, 1996). Indeed, there is preliminary evidence that GAP 43, neurotrophins, other growth factors, cytokines, NO and PGs all play a significant role in the response of cerebral tissue to injury, in analogy to their important functions in the periphery upon PAF damage (Deller and Frotscher, 1997; Ebadi et al., 1997; Fu and Gordon, 1997; Lin et al., 1997; Minghetti and Levi, 1998; Olson, 1997) (Sections 8 and 11).

Lesions of the thalamus induce the expression of GAP 43, the activation of which encourages neuronal growth, coordinates the reorganization of synaptic connectivity and enhances the capacity of neurones to respond to trophic factors such as NGF (Benowitz and Routtenberg, 1997). Indeed, as concerns regenerative changes following thalamic injury, the induction of GAP 43 by individual neurones is proportional to their ability to regenerate their axons. Further, the targetted over-expression of GAP 43 can elicit sprouting following lesions to the CNS, and GAP 43 can also modify the efficacy of transmission at synapses displaying LTP and other plastic changes (Buffo *et al.*, 1997; Fawcett, 1992; Namgung *et al.*, 1997; Vaudano *et al.*, 1995).

Following thalamic damage, NGF is produced both in intrinsic glial and microglial cells, as well as in mast cells and macrophages which migrate into injured tissue from leaky vessels (Deller and Frotscher, 1997; Ebadi *et al.*, 1997; Fu and Gordon, 1997). One interesting, potential mechanism for the induction of NGF in microglial cells involves a role of A2A receptors activated by adenosine released (as ATP) from degenerating neurones (Heese et al., 1997). A putative role of thalamic, neuronal pools of BDNF should also be borne in mind as concerns both structural changes and the modulation of synaptic transmission (Section 14). Indeed, BDNF expression is enhanced by neuronal damage, while knock-out mice lacking TRK B receptors show both an increased vulnerability to neuronal CNS lesions and a lower regenerative capacity of neuronal tissue (Alcántara et al., 1997). As a further example, GDNF can attenuate CNS damage due to ischaemia by a reduction in the synthesis of NO (Wang et al., 1997f). The role of these and other neurotrophins in modulating neuronal phenotype/regeneration and nociception following CNS injury remains to be explored in detail.

Inflammatory responses in the CNS, involving gliosis and the above-mentioned infiltration of damaged tissue by immune-competent cells, occur following injury (Köller et al., 1997; Ridet et al., 1997; Rothwell, 1997). Correspondingly, cytokines such as TNF_{α} and $IL-1\beta$ are generated both by invading macrophages as well as by resident glial and microglial cells, and may reciprocally modify their activity (Arvin et al., 1996; Ebadi et al., 1997; Shen et al., 1997; von Zahn et al., 1997) (Section 4.8). These cytokines likely fulfil a diversity of interactive roles associated with tissue repair and damage (Merrill and Benveniste, 1996). Further, as discussed in Section 4.8, ILs, TNF α and other cytokines may exert a complex modulation of nociception via an influence upon synaptic transmission, including NMDA receptor-mediated processes of sensitization (Grassi et al., 1994; Heyser et al., 1997).

Cytokines can induce glial and microglial expression of NO synthase and COX 2 synthase to yield NO and PGs, respectively (Section 4.8). This of significance since the generation of (neuronal and non-neuronal pools of) NO and PGs in the DH by C fibre input enhances nociception by mechanisms described in Section 3.2.8, while a role for NO in the sensitization of nocisponsive neurones in the thalamus has also been proposed (Do et al., 1994; Kolhekar et al., 1997). Further, although NO may exert certain neuroprotective actions in the CNS, excessive levels of NO can provoke and aggravate tissue damage, and NO may also indirectly modify the response to tissue damage via its influence upon the cerebral microvasculature (Iadecola et al., 1996; Ladiwala et al., 1998; Merrill and Benveniste, 1996; Minghetti and Levi, 1998; Nogawa et al., 1997; Salvemini et al., 1995; Thippeswamy and Morris, 1997). There is some evidence that the induction of COX 2 expression is associated with degenerative processes. However, it has been suggested that PGs generally play an opposite role to NO in exerting primarily neuroprotective actions, for example in countering the excitotoxic effects of overstimulation of NMDA receptors (Kawamura et al., 1997a,b; Minghetti and Levi, 1998).

Indeed, via the production of NO and an excessive accumulation of $[Ca^{2+}]_i$, NMDA receptors are strongly implicated in processes of cell degeneration provoked by ischaemic tissue damage (Choi and Rothman, 1990; Lerea, 1997; Liu *et al.*, 1994c; Lin

et al., 1997). On the other hand, there is evidence that they may play a more constructive role in the control of neuronal growth and migration via actions controlling $[Ca^{2+}]_i$ levels in the growth cones of sprouting neurones (Sjaastad and Nelson, 1997; Ghosh and Greenberg, 1995; Komuro and Rakic, 1993). Further, the proliferation, differentiation and synthetic activity of glial cells may be modified by glutamatergic mechanisms (Steinhäuser and Gallo, 1996).

Finally, tissue acidosis provoked by ischaemia may also modify the activity of many receptor types, both facilitatory and inhibitory to nociceptive transmission (King *et al.*, 1997a) (Section 7.4.5.4). Of particular note are GABA_A receptors, the activity of which is either potentiated or inhibited dependent upon their subunit composition (Krishek *et al.*, 1996).

Thus, following damage to the thalamus and other CNS regions, degenerative and regenerative events—and the resultant changes in nociceptionlikely involve a multitude of factors, many of which resemble those encountered following PAF injury, and many of which may directly modulate nociceptive processing (Section 11). However, these processes require considerable further characterization.

14. CENTRAL ACTIONS OF NEUROTROPHINS AND THEIR RELEVANCE TO NOCICEPTIVE PROCESSING

14.1. A Role for Neurotrophins in the Modulation of Nociception in the CNS?

In Sections 10 and 13, several mechanisms emerged as of particular importance in determining those alterations in nociceptive processing and transmission which underlie the induction of painful states. These include:

- 1. amplification of responsiveness at sensitized neurones in the DH, thalamus and cortex;
- a key role of NMDA receptors both in sensitization processes and in the transfer of nociceptive information from PAFs to the DH, and therefrom to higher centres;
- 3. a widespread inhibitory influence of GABAergic ININs upon nociceptive transmission in the DH and supraspinal structures; and
- 4. a pronounced influence of neurotrophins, such as NGF, upon the activity, phenotype and growth status of PAFs.

In this section, these themes are drawn together within a hypothetical framework which proposes that *central* actions of neurotrophins, in particular BDNF, may play an important role, in interaction with glutaminergic (NMDA) and GABAergic mechanisms, in strengthening or otherwise modulating transmission at synapses involved in the transfer and integration of nociceptive information at supraspinal and spinal loci.

14.2. Neuronal Sources of Neurotrophins: Physiological Modulation

Neurotrophins are generated in discrete populations of neurones localized in several structures involved in nociceptive processing, and including the S I cortex, the thalamus, the striatum and the DH. Further, neuronal pools of neurotrophins are generated both constitutively as a function of 'basal', neuronal activity as well as in response to appropriate-excitatory-stimuli (Conner et al., 1997; Lo, 1995; Lu et al., 1991; Marty et al., 1997; Thoenen, 1995; Zhou and Rush, 1996). A detailed discussion of the diverse roles of neurotrophins, in particular the widely-distributed BDNF, in the transient and longterm sculpting of synaptic architecture and modulation of synaptic transmission is beyond the scope of the present review (Thoenen, 1995; Marty et al., 1997; Lewin and Barde, 1996; Berninger and Poo, 1996; Lo, 1995; Patterson and Nawa, 1993; Schmidt-Kastner et al., 1996). Nevertheless, the following points support a role of neuronal pools of neurotrophins in the modulation of nociception in the CNS.

14.3. Induction and Release of Neuronal Pools of Neurotrophins: NMDA Receptor-Mediated Stimulation

Neurotrophins are released from neurones by an unconventional mechanism which requires an increase in $[Ca^{2+}]_i$ levels, yet which is dependent on extracellular Na^+ rather than Ca^{2+} (Blöchl and Thoenen, 1995). A rise in cAMP levels may also trigger the release of NGF (Thoenen, 1995). Increases in neuronal excitability are associated with an enhanced secretion of neurotrophins. In this regard, it is of particular significance that glutaminergic mechanisms, involving the activation of NMDA receptors and in increase in $[Ca^{2+}]_i$, induce the release and synthesis of neurotrophins (Castrén et al., 1993a; Goodman et al., 1996; Gwag et al., 1993; Zafra et al., 1991; Thoenen, 1995; Lu et al., 1991). Although the decrease in cAMP levels provoked by group II and III mGlu receptors suggests that they may suppress neurotrophin production, the activation of group I mGlu receptors, which are positively coupled to PLC and increases in [Ca²⁻ may accelerate neurotrophin synthesis and release.

To date, a systematic evaluation of the potential influence of nociceptive stimulation upon the activity of CNS pools of neurotropins has not been undertaken. Nevertheless, the generation and secretion of neuronal pools of BDNF is facilitated under the following conditions, each of which is of relevance to the induction of painful states:

- ischaemia, provoked by stroke-induced vascular damage to the thalamus, induces BDNF gene expression, and NMDA antagonists can prevent this effect (Hughes *et al.*, 1993);
- 2. cortical spreading depression, a process implicated in the triggering of migraine attacks (Sections 4.7 and 7.9), also enhances BDNF synthesis (Kokaia *et al.*, 1993; Lindvall *et al.*, 1992; Moskowitz, 1992); and

3. epileptiform activity in the cortex and hippocampus, which mimics the excited, oscillating neuronal circuits and burst firing triggered by a loss of ININ tone, also elicits BDNF synthesis (Elmér *et al.*, 1998) (Sections 10.6.2 and 13.3).

Interestingly, in contrast to NMDA receptors, and consistent with their *inhibitory* influence on neuronal activity, activation of GABAergic receptors *downregulates* neurotrophin production (Zafra *et al.*, 1991; Thoenen, 1995).

In correspondence with the above findings, LTP in the hippocampus and memory formation, which is sustained and suppressed by NMDA- and GABA-dependent processes, respectively, is associated with an induction of neurotrophins, notably BDNF (Bliss and Collingridge, 1993; Castrén et al., 1993b; Collingridge and Singer, 1990; Gordon et al., 1997a; Ma et al., 1998; Patterson et al., 1992; Wilson and Tonegawa, 1997). These observations concerning LTP are of particular importance since hippocampal LTP is analogous to NMDA receptormediated processes of wind-up implicated in processes of neuronal sensitization in the DH, thalamus and S I (Section 10.2.1). Further, with respect to the role of supraspinal mechanisms in mediating referred pain (Section 9.1.4), unilateral induction of LTP results, via intrahemispheric connections, in a bilateral induction of mRNA encoding neurotrophins and TRK A receptors (Bramham et al., 1996).

The above findings indicate, thus, that neurotrophin levels are elevated by events (including activation of NMDA receptors) which enhance synaptic transmission and which trigger processes underlying synaptic plasticity. In addition, as outlined below, by a positive feedback (retrograde) mechanism, the release of neurotropins initiates processes further reinforcing synaptic signalling.

14.4. Actions of Neurotrophins: Modulation of Synaptic Transmission

14.4.1. Anterograde Transport of BDNF

Recent studies (Altar et al., 1997; Conner et al., 1997; Smith et al., 1997c; Von Bartheld et al., 1996), have shown that BDNF is anterogradely transported to nerve terminals. This raises the possibility that anterograde release of BDNF may mediate rapid, postsynaptic actions involved in the modulation of LTP and memory formation (Section 10.2.1). Indeed, many hippocampal neurones express both BDNF and TRK B receptors raising the possibility of retrograde and anterograde mechanisms of actions (Smith et al., 1997c). Similar findings also suggest an anterograde release of BDNF from PAF terminals in the DH (Canossa et al., 1997; Cho et al., 1997a,b; Michael et al., 1997a; Zhou and Rush, 1996) (Section 14.5.3). Thus, at specific (glutaminergic) synapses, it is conceivable that BDNF is released both pre and postsynaptically, and that it exerts actions at both pre and postsynatic sites.

14.4.2. Modulation of Glutaminergic Transmission: Presynaptic and Postsynaptic Actions

Neurotrophins exert a marked influence upon gene expression and synaptic transmission. Such actions are *not* restricted to delayed alterations in synaptic morphology and connectivity. Rather, neurotrophins also elicit rapid, TRK-mediated effects via multiple intracellular transduction mechanisms (Berninger and Poo, 1996). These actions of neurotrophins are expressed both *presynaptically* in a retrograde fashion at those synapses which provoke their release, as well as *postsynaptically*.

As concerns *postsynaptic* actions, a phosphorylation of ion channels and transmitter receptors is likely involved in the potentiation of postsynaptic ion currents (Holm et al., 1997; Kaplan and Stephens, 1994; Levine et al., 1995; Berninger and Poo, 1996; Swope et al., 1992)-although other types of alteration in neuronal activity may also occur. Most pertinently, NMDA receptor-mediated currents are potentiated by their neurotrophinmediated (phosphorylation). For example, BDNF rapidly phosphorylates postsynaptic NMDA receptors in the hippocampus (which are involved in LTP) by an intracellular mechanism involving increases in $[Ca^{2+}]_i$, and, probably, the activation of PLC or other PKs (Blanquet and Lamour, 1997; Sakai et al., 1997; Suen et al., 1997; Wang and Salter, 1994). In a recent study of hippocampal neurones, a further postsynaptic mechanism for the potentiation of NMDA receptor-mediated activity by BDNF was proposed. Jarvis et al. (1997) suggested that BDNF enhances NMDA receptormediated currents (and reduces their desensitization) by acting at the allosteric, facilitatory glycine_B site, stimulation of which is obligatory for full activation of the ion channel coupled to NMDA receptors. The source of BDNF mediating this action is likely presynaptic terminals, from which BDNF can be anterogradely released (Section 14.4.1). However, a feedback action of BDNF following its release from postsynaptic neurones themselves is also a possibility.

More extensive data concerning presynaptic actions of neurotrophins have shown that they increase synaptic currents via an increase in the release of transmitters, including GLU (Berninger and Poo, 1996; Knipper et al., 1994). These presynaptic actions reflect a rapid increase in $[Ca^{2+}]_i$ levels. This rise is dependent upon extracellular Ca²⁺ and possibly involves L-type VDCCs which are susceptible to phosphorylation following induction of PKs via TRK receptors (Berninger et al., 1993; Kaplan and Stephens, 1994; Sakai et al., 1997; Sherwood et al., 1997; Wildering et al., 1995). These presynaptic actions imply, then, a retrograde role of neurotrophins to increase activity at those excitatory synapses which provoke their own release: that is, a positive feedback mechanism. This is reminiscent of the positive feedback action of GLU at presynaptic NMDA receptors on the central terminals of PAFs in the DH (Section 3.2.4). On the other hand, BDNF released in an anterograde fashion may also mediate actions at presynaptic terminals.

BDNF, NT 3 and NT-4/5 all potentiate transmission at hippocampal CA 1 synapses in a manner resembling LTP and involving the recruitment of glutamatergic mechanisms (Kang and Schuman, 1995; Messaouidi, 1998; Thoenen, 1995). In line with a physiological role of BDNF in this regard, LTP and memory formation is accompanied by an increase in BDNF levels in hippocampus (Section 14.3). Further, LTP (and memory retention) is impaired upon antagonism of TRK B receptors by fusion products and by administration of antisense probes neutralizing BDNF. Similar findings have been obtained in BDNF knock-out mice in which adenovirus-mediated substitution of the BDNF gene results in a restoration of LTP (Figurov *et al.*, 1996; Korte *et al.*, 1995; Lu and Figurov, 1997; Ma *et al.*, 1998; Thoenen, 1995; Patterson *et al.*, 1996). Interestingly, BDNF has also been shown to block *LTD* in the cortex (Akaneya *et al.*, 1996).

Very recently, a novel aspect of the positive feedback actions of neurotrophins was revealed with the demonstration that BDNF can induce its own release from hippocampal neurones (Canossa *et al.*, 1997; Furokawa *et al.*, 1997).

14.4.3. Modulation of GABAergic Transmission

In addition to the above-described, facilitatory influence of BDNF upon NMDA-mediated transmission, neurotrophins may also interact with GABAergic mechanisms. GABAergic ININs target hippocampal pyramidal neurones containing BDNF and displaying LTP. BDNF postsynaptically decreases GABAergic currents in these neurones by an IP₃-mediated mobilization of [Ca²⁺]_i which presumably leads to the phosphorylation of GABAA receptors (Section 10.4.3). Thereby, the induction of hippocampal LTP is facilitated (Jiang et al., 1997; Tanaka et al., 1997b). Further, NT 3 decreases GABAergic transmission in the hippocampus and potentiates both synaptic transmission and processes of LTP (Kim et al., 1994), although it is not known whether this action is expressed pre- or post-synaptically. In fact, information concerning the modulation of GABAergic transmission by neurotrophins is limited, and they do not decrease GABAergic transmission in all structures examined to date (Section 14.5.1). Thus, a putative modulation by neurotropins of synaptic events underlying nociception should not necessarily be assumed to reflect inhibition of GABAergic transmission.

14.4.4. Overview of Short- and Long-Term Actions of BDNF

It may be concluded from the above paragraphs that BDNF and/or other neurotrophins rapidly modify neuronal excitability and processes of activity-dependent plasticity via mechanisms involving the potentiation of actions mediated by NMDA receptors and, possibly, a modulation of GABAergic transmission (Marty et al., 1997; Rutherford et al., 1997). Most significantly, via both presynaptic (retrograde) and postsynaptic actions, neurotrophins increase transmission at glutamatergic (NMDA receptor-coupled) synapses involved in plastic, long-term changes increasing neuronal sensitivity. Such rapid actions may be succeeded by persistent changes in synaptic efficacy due to phenotypic alterations in synaptic architecture and in patterns of synaptic connectivity, in particular under conditions of damage to nervous tissue and/ or prolonged excitatory stimulation (Thoenen, 1995). As concerns longer-term changes in neuronal morphology and phenotype, via the modulation of $[Ca^{2+}]_i$, PKs and various nuclear regulatory elements involved in plastic synaptic events, neurotrophins may modify the CNS gene expression of many substances involved in the modulation of nociception, including NO and various neuropeptides (Berninger and Poo, 1996; Carmignoto *et al.*, 1997; Finkbelner *et al.*, 1997; Holm *et al.*, 1997; Jones *et al.*, 1994; Liu *et al.*, 1997a; Planas *et al.*, 1997).

14.4.5. Summary

To summarize, BDNF and other neurotrophins are released from neurones by excitatory stimuli, including the activation of NMDA receptors. BDNF plays an important role, by both pre- and postsynaptic mechanisms, in the strengthening of synaptic transmission via a reinforcement in NMDA receptor-mediated activity. Further, at certain, through not all, neurones, an inhibition of GABAergic activity is apparent. These actions are associated with a facilitation of NMDA-dependent and GABA-suppressible mechanisms of LTP in the hippocampus, a NMDA receptor-mediated and GABA receptor-inhibited process of synaptic amplification analogous to mechanisms of neuronal sensitization underlying painful states (Section 10.2.1). In addition, the recent demonstration of anterograde BDNF transport in neurones suggests that it may also mediate actions via 'conventional' synaptic release onto postsynaptic targets. It is not unreasonable, then, to hypothesize that BDNF and other neurotrophins may be involved in modulating synaptic events underlying the sensitization of spinal and supraspinal neurones involved in the transmission of nociceptive information.

14.5. Sites at Which Neurotrophins May Modulate Nociceptive Processing

14.5.1. The S I of the Cortex

One locus where neurotrophins may modulate synaptic processes involved in nociceptive processing is the S I cortex, which possesses both a rich population of TRK B receptors and a high concentration of BDNF (Yan et al., 1997; Zhou et al., 1993). Indeed, thalamocortical afferents transmitting sensory information to the cortex exert their actions via EAAs/NMDA receptors (Section 6.3.1). Further, NMDA receptors play a major role in neuronal plasticity in the cortex and their activation modifies the expression of BDNF therein (Section 14.3). Thus, BDNF may, via pre and postsynaptic actions described in Section 14.4, reinforce NMDA receptor-mediated transmission of nociceptive information in the cortex (Armstrong-James et al., 1993; Castrén et al., 1992, 1993a; Conti and Hicks, 1996; Gordon et al., 1997a; Li et al., 1994; Marty et al., 1997; Thoenen, 1995; Zafra et al., 1991). On the other hand, BDNF modifies synaptic plasticity and connectivity in the S I cortex via a facilitatory (retrograde?) influence upon the activity of TRK Bbearing GABAergic ININs which, like thalamocortical afferents, target pyramidal neurones secreting BDNF (Jones et al., 1994; Marty et al., 1997;

Rocamora *et al.*, 1996; Rutherford *et al.*, 1997; Thoenen, 1995). Such actions upon GABAergic ININs may include alterations in their patterns of connectivity as well as changes in the levels of GABA and of other co-expressed transmitters, such as NPY (Marty *et al.*, 1997). Two modes of sensory input have been examined in detail as regards the influence of BDNF upon plastic events in the cortex.

- 1. Mechanical stimulation (deflection) of whiskers in mice leads to a specific up-regulation of BDNF in topographically-organized structures of the S I termed 'barrel columns'. Particularly pronounced effects are seen in layer IV which receives the most intense input from thalamocortical afferents originating in the ventrobasal complex (Chmielowska et al., 1989; Rocamora et al., 1996). In contrast, the opposite manipulation of sensory deprivation causes a reduction in cortical BDNF expression. This may lead to a reduction in GABAergic activity resulting in enlarged RFs and increases in spontaneous neuronal activity (Marty et al., 1997). Such effects are reminiscent of changes triggered in the DH upon PAF damage due to a loss of GABAergic inhibitory transmission (Section 10.6).
- The thalamus (lateral geniculate nucleus) is the 2. major source of sensory input to the visual cortex. Like nociceptive input to the cortex, thalamocortical afferents transmitting visual information employ EAAs. Temporary, developmental deprivation of visual input leads to marked alterations in the RFs of neurones in the visual cortex and a to (reversible) reduction in the levels of BDNF and GABAergic transmission (Castrén et al., 1992). It has been suggested that the down-regulation of BDNF may intervene in this diminution of GABAergic ININ activity (Marty et al., 1997; Rocamora et al., 1996).

To summarize, nociceptive (and other forms of sensory) input to the cortex mediated via thalamocortical pathways utilizing NMDA receptors may trigger the induction of BDNF which further reinforces excitation by both postsynaptic and (feedback, retrograde) presynaptic actions. Depending upon stimulus duration and other factors, this may lead to both rapid and persistent functional and phenotypic/morphological mechanisms underlying sensitization and increases in synaptic efficacy. However, in analogy to the parallel activation of PNs and ININs in the DH by C fibre input, BDNF may temper these actions via an enhancement of GABAergic tone. As described in Section 12.2, under diverse conditions, the increase in neuronal excitability provoked by nociceptive input may be either less or more marked in cortical than in thalamic regions. This raises the possibility that the precise balance between excitatory and inhibitory mechanisms involving NMDA receptors, neurotrophins and GABAergic ININS may be involved in such differences. On the other hand, a loss of excitatory, sensory (nociceptive) input to the cortex following damage to PAFs, the STT or thalamus may lead to a down-regulation in cortical activity of BDNF, a reduction in GABAergic tone and neuronal disinhibition (Cabelli et al., 1995; Gordon et al., 1997a; Marty et al., 1997; Rocamora et al., 1996; Thoenen, 1995).

Similar events involving reciprocal interactions of BDNF and glutamatergic/NMDA and GABAergic transmission likely occur in other cortical regions receiving *nociceptive* input from the thalamus and other structures. The precise nature and time-course of such interactions, and their global influence upon nociceptive transmission and processing, remains to be established.

14.5.2. The Thalamus and Other Cerebral Structures

The thalamus contains a high density of TRK B receptors and a substantial population of BDNFpositive neurones (Furukawa et al., 1997; Thoenen, 1995; Yan et al., 1997; Zhou et al., 1993). Thalamic levels of BDNF are elevated both by selective activation of NMDA receptors and by generalized increases in the electrical activity of the thalamus (Castrén et al., 1992; Schmidt-Kastner et al., 1996). Further, ischaemic CNS damage can induce BDNF gene expression and, as discussed in Section 13.5, neurotrophins play a role in neuronal regeneration of the thalamus following its injury. Thus, vascular accidents underlying stroke-induced thalamic pain likely also induce an expression of neurotrophins in the thalamus. Both NMDA receptors and GABAergic mechanisms are involved in the induction of pain due to thalamic damage (Section 13). An interaction of BDNF with these mechanisms may, therefore, play a role in the modulation of painful states due either to damage to the thalamus itself, or to input from ascending nociceptive channels.

The striatum, amygdala and lateral hypothalamus, additional sites of termination of ascending nociceptive input, also possess TRK B receptors and neurones synthesizing BDNF. Thus, they also represent further potential sites at which neurotrophins may modify synaptic events involved in the processing of nociceptive information (Castrén *et al.*, 1993b; Furukawa *et al.*, 1997; Schmidt-Kastner *et al.*, 1996; Yan *et al.*, 1997). The midbrain is also rich in BDNF-immunoreactivity (Furukawa *et al.*, 1997). Although the underlying mechanisms are obscure, administration of BDNF or NT 3 into the midbrain elicits antinociception in parallel with an alteration in cerebral and spinal levels of 5-HT, β -endorphin, SP and NPY (Siuciak *et al.*, 1994, 1995).

14.5.3. The DH of the Spinal Cord

The most intriguing, potential site of action of BDNF and other neurotrophins in the modulation of nociception is the DH. Although data are limited, recent studies support this possibility.

Both the DH and the (homologous) trigeminal nucleus possess a substantial population of TRK B receptors (Michael *et al.*, 1997b; Zhou *et al.*, 1993). As concerns potential sources of BDNF acting at these sites, there is little evidence for BDNF in intrinsic neurones in the DH or in the terminals of descending pathways. Thus, the most likely source of BDNF for the DH is PAFs (Acheson *et al.*, 1995; Zhou and Rush, 1996). Indeed, there is compelling evidence for anterograde transport of BDNF to

nerve terminals in the CNS and this process has been demonstrated to occur in PAFs, probably small calibre C fibres, projecting to the DH (Conner et al., 1997; Zhou and Rush, 1996; Cho et al., 1997a,b). Further, it has recently been demonstrated that axon terminals intensely labelling for BDNF are present in superficial DH laminae I and II-and to a lesser extent-in deeper laminae IV/V and lamina X (Conner et al., 1997; Cho et al., 1997a,b). That is, the principal laminae targetted by nocisponsive C fibres and transmitting nociceptive information to higher centres (Section 4.1). PAF damage has been shown to enhance BDNF levels in the DRG (Ernfors et al., 1990) and peripheral tissue inflammation elicits a parallel increase in DRG and DH staining for BDNF (Cho et al., 1997a,b). This induction of BDNF expression by noxious stimulation or PAF injury is consistent with its occurrence in nocisponsive, small calibre neurones, and can be prevented by antibodies to NGF (Cho et al., 1997b,c; Michael et al., 1997a). Those observations reinforce the argument that an anterograde transport of BDNF in C fibres is the primary source of BDNF in the DH and indicate that this process is accelerated by noxious, peripheral manipulation.

There is evidence from other tissues for the neuronal storage and secretion of BDNF (Fawcett, 1992; Fawcett et al., 1997) (Section 14.2). Thus, it is likely that BDNF is released from C fibre terminals in the DH to exert acute, functional actions at intrinsic DH neurones. In this respect, one potential target of BDNF is the terminals of $A\beta$ fibres which bear TRK B receptors. This would potentially implicate BDNF in the modulation of mechanical allodynia. Further, as discussed in Section 11.3.3, following PAF injury, BDNF release from C fibre terminals may attract $A\beta$ fibres into Laminae II. It is also possible that BDNF acts directly on intrinsic DH neurones bearing TRK B receptors. In this respect, a potential interaction of BDNF with NMDA receptor-bearing PNs, EXINs or GABAergic ININs targetted by C fibres would be of special interest. Consistent with a putative role of BDNF in the enhancement of nociception, it was recently shown to increase the expression of both c-fos and NO synthase in DH neurones (Bennett et al., 1996b). Further, BDNF reduces GABAA receptor-mediated conductances in injured PAFs, suggesting that it may facilitate nociceptive input from their terminals into the DH (Oyelese et al., 1997). Thus, a potential, pronociceptive role of C fibre-derived pools of BDNF in the DH should provide fertile territory for future exploration. Further, an interesting possibility is that the central release of C fibre-localized pools of BDNF in the DH may be modulated by NMDA receptors on their terminals (Liu et al., 1997b) (Section 3.2.4).

Comparatively little is known concerning potential actions of NGF at TRK A receptors in the spinal cord. Potential sources of NGF comprise both glial cells and intrinsic neurones. Anterograde delivery of NGF from PAFs to the DH has not, to date, been demonstrated (Anand *et al.*, 1995). In fact, evidence for TRK A receptors on intrinsic DHlocalized neurones is limited, although they may be present both in superficial laminae (a principal target of nociceptive PAF input) as well as in lamina X (indicating a potential role in the integration of visceral sensory input and autonomic outflow) (Michael *et al.*, 1997b; Molliver *et al.*, 1995). On the other hand, fibres showing dual labelling for TRK A receptors and CGRP have been visualized in superficial DH laminae and the dorsal columns. This suggests a possible modulation by DH pools of NGF of the activity of the central terminals of nocisponsive C fibres. However, superfusion of the spinal cord with NGF has been reported *not* to modify the release of SP therein (Averill *et al.*, 1995; Malcangio *et al.*, 1997a,b; Michael *et al.*, 1997a; Molliver *et al.*, 1995).

Administration of NT 3 into the midbrain elicits antinociception via a mechanism involving serotoninergic transmission (Siuciak et al., 1995). In this light, it is of interest that TRK 3 receptors have been detected in descending serotoninergic fibres in the spinal cord (Arvdisson et al., 1994). Further, binding sites for NT 3 have been visualized throughout the spinal cord, including superficial laminae (Zhou and Rush, 1994). The origins of putative DH pools of NT 3 remain, however, unclear. The demonstration that large, myelinated, mechanosensitive PAFs retrogradely transport NT 3 from the peripheral terminals to their DRG raises the possibility of its subsequent anterograde delivery to the DH, though this remains to be demonstrated (Di Stefano et al., 1992). Malcangio et al. (1997b) showed that both the systemic and spinal administration of NT 3 reduced the enhancement of DH SP release elicited by electrical stimulation of the DRG, an action apparently mediated by activation of opioiderigic ININs. The authors tentatively related this observation to the ability of peripherally-administered NT 3 to attenuate sensory deficits associated with a chemically-induced large fibre neuropathy (Helgren et al., 1997). However, the systemic administration of NT 3 has been documented to elicit a transient hyperalgesia (Malcangio et al., 1997b). Thus, the potential role of NT 3 in the DH and elsewhere in the modulation of nociception remains to be clarified.

14.6. Summary

An examination of the potential role of BDNF and other neurotrophins in modulating rapid and prolonged synaptic events underlying neuronal sensitization and nociception would appear to be of considerable interest, in particular with regard to the DH and cortex. Although the above discussion focused on neuronal pools of neurotrophins, the production of neurotrophins by glial cells and immunecomponent cells provides a further potential source via which synaptic events might be modulated, for example following CNS injury. In this light, one final aspect should be noted. BDNF and other neurotropins exert a marked modulatory influence upon processes of neuronal death elicited by ischaemic damage and the activation of excitotoxic NMDA receptors, effects in which NO may be involved (Samdani et al., 1997) (Section 13.5). It was pointed out in Section 13.3 that a loss of vulnerable GABAergic ININS upon ischaemic damage to the CNS may be involved in the induction of central

pain. Thus, the possibility that neurotrophins may indirectly modify nociception via long-term, trophic actions at GABAergic and other cell types following tissue injury, should also be considered.

In conclusion, it is reasonable to hypothesize a role of BDNF-and, perhaps, other neurotrophins-in the modulation of plastic, synaptic events involved in the transfer and modulation of nociceptive information in the CNS. Such actions may be mediated both rapidly, via modification of transmitter release and receptor function, and more slowly and persistent, via a reshaping of synaptic morphology and patterns of neuronal connectivity. The actions of BDNF and other neurotrophins are likely expressed in interaction with glutamatergic neurones/NMDA receptors and GABAergic transmission. That is, excitatory and inhibitory systems playing crucial roles in the transmission and processing of nociception in DH the thalamus and the cortex. Although the DH and cortex are the most promising targets for an exploration of such potential mechanisms, the thalamus and other supraspinal regions also justify investigation. However, in particular as concerns interactions with GABAergic mechanisms, only direct study will allow for a characterization of the precise influence of neurotrophins upon nociceptive processing at various levels of the neuroaxis.

15. GENERAL DISCUSSION AND CONCLUSIONS: A SUMMARY OF KEY ISSUES

15.1. The Complexity of Mechanisms Involved in the Induction of Pain

The foregoing discussion exemplifies the extraordinary complexity and diversity of mechanisms involved in the induction of pain. Over the last decade, enormous advances have been made in our understanding of these processes and several, global aspects deserve brief emphasis.

- 1. A knowledge of:
 - the intracellular transduction mechanisms activated by specific receptor types; and
 - 2. the localization of these receptors at the cellular level,

provides a powerful basis for a rational prediction of their role in the modulation of nociceptive transmission. For example, the detection of excitatory P_{2x3} receptors coupled to cationpermeable channels on peripheral, nocisponsive C fibres terminals leads inevitably to the prediction that they fulfil a prononceptive role at this locus.

2. Over recent years, the roles of many mediators in the response to, and the induction of, pain have emerged to be more diverse than previously realized. This increasing complexity reflects several factors: in particular the proliferation of receptorial and intracellular transduction mechanisms mediating the actions of specific transmitters, such as 5-HT, NPY and ATP. Further, the identification of previously-unsuspected loci of action is also of significance: for example, actions of PGs in the DH, of EAAs (and opioids) in the periphery and of SP in the limbic system. Temporal aspects are also of importance as concerns adaptive events, in particular neuronal sensitization, underlying prolonged pain. Mechanisms which instantaneously modify neuronal activity via alterations in ionic flux, such as the activation of NMDA receptors or L-type VDCCs, have been shown to express long-term actions: thus, they sequentially induce increases in $[Ca^2]$ ⁺]_i levels and activate specific PKs, nuclear transcription factors, IEGs and gene expression. Such intracellular cascades lead to sustained changes in neuronal phenotype and synaptic plasticity. Contrariwise, neurotrophins, such as BDNF and NGF, and intracellular growth-associated factors, such as GAP 43, may, in addition to their long-term modification of synaptic morphology, exert a rapid influence upon synaptic transmission via interactions with [Ca² various PKs and other intracellular and extracellular signals.

- 3. Considerably more work is needed at the *cerebral* level for an improved understanding of supraspinal mechanisms involved in the induction of pain, in particular its *emotional-cognitive* aspects. Initial analyses at the cellular level of supraspinal mechanisms triggered by peripheral inflammation or PAF injury have suggested several striking similarities to events occurring in the DH. Notably, as concerns the involvement of NMDA receptor-mediated processes of synaptic plasticity and the importance of inhibitory GABAergic transmission. Nevertheless, adaptive events occurring in suprapinal centres are distinctive and do *not* simply duplicate changes seen at the spinal level.
- 4. Experimental models of PAF injury have proven of decisive importance in furthering our understanding of processes underlying neuropathic pain and the regenerative capacity of damaged neurones (Bennett and Xie, 1988; Coyle, 1996; Kim and Chung, 1992). Such models have revealed certain important differences between nociceptive pain (due to nociceptor stimulation) and neuropathic pain (due to PAF damage) as concerns their induction and maintenance. For example, in terms of the phenotype changes provoked in small calibre C fibres, the (transient) reduction in the activity of DH-localized ININs elicited by PAF injury, abnormal patterns of ectopic firing in damaged PAFs and the reorientation of injured A β fibres from deep into superficial laminae of the DH. Correspondingly, certain strategies for the relief of inflammatory as compared to neuropathic pain may well differ.
- 5. Perhaps more surprisingly, many *similarities* exist between painful states elicited by the stimulation as compared to the injury of PAFs. For example, the key role of large calibre, $A\beta$ fibres in mediating mechanical allodynia. Further, the emergence of nociceptive and neuropathic pain, in each case, critically depends upon the development of *central* processes of *sensitisation* in the DH and supraspinal structures. Indeed, in some respects,

the similarities between the pain provoked by PAF injury as compared to tissue inflammation are more striking that the differences. Consequently, it may be possible to elaborate strategies for the relief of both nociceptive and neuropathic pain based on countering such common processes as central sensitization, peripheral C fibre drive maintaining certain sensitized states and $A\beta$ fibre mediated mechanical allodynia. In this regard, it may be desirable to reduce *both* the initial PAF volley which triggers central sensitization and the subsequent C fibre, barrage drive underlying its persistence. This issue is currently undergoing clinical evaluation employing protocols of pre-emptive as compared to post-hoc administration of opioids, NMDA antagonists, Na⁺-channel blockers and other classes of analgesic agent (Andersen et al., 1996; Dahl et al., 1993; Gordon et al., 1997b; Katz et al., 1996; Smith et al., 1994b; Rice, 1995a; Woolf and Chong, 1993).

- 6. A fundamental process underlying the induction of prolonged, painful states is sensitization peripheral nocisponsive terminals, and localized CNS neurones involved in the transmission of nociceptive information to successively higher centres. Sensitization is elicited by sustained and repetitive rather than phasic stimulation and, within a temporal framework, one might conceive of essentially three phases of neuronal activation (Tables 2, 5 and 6).
 - 1. The phasic engagement of a discrete population of DH neurones, mediated by AMPA receptors, and involving a spatially restricted influence upon $[Ca^{2+}]_i$ concentrations and other intracellular signals, leading to a brief discharge of APs and the activation of 'discriminative-sensory' channels. This process corresponds to the adaptive, warning role of acute, localized and transient exposure to (cutaneous) noxious stimulation.
 - 2. More prolonged activation of a more extensive population of DH neurones by processes of temporal and spatial summation, with a greater recruitment of NMDA/mGlu/NK1 receptors and VDCCs etc, leading to more pronounced and widely-distributed increases in $[Ca^{2+}]_i$ levels and the activation of PKs, triggering alterations in ion channel/receptor function by phosphorylation and the activation of nuclear transcription factors, IEG and gene expression. However, these processes remain largely reversible (maintained by peripheral input) and morphological alterations in synaptic architecture and neuronal connectivity do not occur. This corresponds to subchronic, painful states which decline in parallel with the resolution of the inflammation/tissue injury by which they are provoked.
 - 3. Particularly in the case of injury to PAFs (or the CNS), long-term and abnormal patterns (spatially and temporally) or peripheral input, with a loss of ININ control leading to the aggravation and propagation of self-perpetuating sensitized states, accompanied by perma-

nent alterations in neuronal organization. This corresponds to chronic, maladaptive (generally neuropathic) painful states.

- 7. One major consequence of the neuronal excitation provoked by a loss of inhibitory transmission due to PAF or CNS injury is, then, the induction of widespread, generalized and reverberating, overexcited circuits. These may incorporate extensive regions of the DH and supraspinal structures, including oscillating ipsilateral and bilateral, intrathalamic, intracortical and thalamocorticothalamic networks. The induction of such states implies that the notion of anatomically-defined, precisely-localized 'pain centres' may be illusory. Imaging studies of the activation of cerebral regions in man have also indicated that a widespread 'neuromatrix' rather than individual loci are involved in the sensation of pain (Section 6). Thus, the discrete interference with activity in specific cerebral (or DH) regions may not necessarily be effective in interrupting hyperexcitability and affording pain relief.
- 8. This lack of discrete, circumscribed CNS regions underlying pain transmission and sensation is related to, and compounded, by the extensive redundancy of mechanisms transmitting nociceptive information, and the extensive pattern of reciprocal interactions amongst them. Thus, there exists a multiplicity (moving down from the cortex) of: cortical and subcortical structures integrating nociceptive information; ascending pathways for its transmission from the DH to supraspinal centres; pronociceptive neurotransmitters released from multiple classes of nocisponsive onto DH neurones; inflammatatory mediators acting at their peripheral terminals; mechanisms triggering abnormal patterns of firing in injured PAFs and cascades of intracellular signals initiated by peripheral and central mediators of pronociceptive actions (Table 4, Figs 3-5). Thus, selective inactivation of delimited regions of the CNS, or blockade of the actions of specific pronociceptive mechanisms, is unlikely to offer a generalized solution for the countering of processes underlying pain due to contrasting pathologies. Examples of manipulations manifesting limited efficacy are provided by lesions of specific cerebral nuclei, discrete elimination of the STT at the spinal level, sympathectomy, selective blockade of spinal NK1 receptors or specific antagonism of peripheral B₂ receptors.
- 9. In the above light, it is also of significance that noxious stimulation both activates pronociceptive mechanisms and coactivates antinociceptive (or 'anti-pronociceptive') mechanisms counterbalancing their actions. For example: descending inhibition and facilitation; C fibre activation of WDR/NS PNs and GABAergic ININs in the DH; pronociceptive and antinociceptive transmitters in PAFs; and, at the molecular level, Ca²⁺-and PK-mediated processes of sensitization/desensitization and phosphorylation/dephosphorylation. Procedures for the supposedly discrete elimination of pronociceptive mechanisms must

take into account a potential long-term perturbation of intrinsic, *antinociceptive* mechanisms.

As summarized in the three preceding points, there is evidence for

- 1. diffuse cerebral 'neuromatrices' underlying pain sensation;
- 2. redundancy in channels for pain transmission channels: and
- 3. compensatory antinociceptive mechanisms activated by noxious stimulation.

These observations question whether discrete manipulations of specific components of pronociceptive mechanisms can offer a generalized strategy for pain relief. Nevertheless, analgesic strategies appropriate to the effective control of specific painful conditions remain a realistic proposition. For example, the vascular and/or neurogenic actions of 5-HT_{1B} agonists on CBVs in the treatment of migraine or the use of (TTX-sensitive) Na⁺-channel antagonists for controlling ectopic activity of injured PAFs. Indeed, the recent discovery of many hitherto-unsuspected 'unconventional' mechanism underlying the induction of painful states should be emphasized. Notably: sympathetic mechanisms; immunecompetent and glial cells in the periphery and CNS; neurotrophins; cytokines; protons; Na⁺-channels; ATP; NO; vanilloids; and a multiplicity of neuropeptides. Correspondingly, the palette of potential targets and strategies for the development of novel analgesics is far richer than previously imagined. Thus, it is appropriate to conclude with a brief consideration of the implications of an improved understanding of mechanisms underlying pain for the discovery of novel, analgesic agents.

15.2. Mechanisms of Antinociception, Nociception and the Development of Novel Analgesics

In the seventies and eighties, considerable excitement was elicited by the discovery and cloning of endogenous opioid peptides and the characterization of descending, inhibitory monoaminergic pathways to the DH. Correspondingly, efforts were devoted to developing drugs which mimick or reinforce physiologically-engaged, 'endogenous' opioidergic monoaminergic or other mechanisms of antinociception (Basbaum and Fields, 1984; Fields and Basbaum, 1994; Millan, 1986, 1990, 1997). Unfortunately, for a variety of reasons, comparatively little progress in the development of improved analgesic agents has been achieved via this approach. For example, difficulties in dissociating the sedative/hypotensive actions of α_2 -AR agonists from their analgesic properties (Millan, 1998) and the-to date-failure to identify a key, DH-localized 5-HT receptor type mediating antinociception (Millan, 1995). Other factors, such as problems encountered in the cloning of opioid receptors and in the identification of endogenous ligands for μ -opioid receptors—now resolved—may also be relevant (Mansour *et al.*, 1995; Zadina et al., 1997).

The exponential growth of interest in mechanisms underlying the *induction* rather than the inhibition of pain may reflect, thus, a certain frustration with strategies for the development of novel analgesics based upon the principle of reproducing endogenous mechanism of antinociception. Moreover, as indicated in Section 15.1, the availability of experimental models of PAF damage has provided a key impetus for studies of mechanisms underlying neuropathic pain, a condition which responds poorly to traditional analgesics such as opioids and anti-inflammatory agents. Also of key importance is, as mentioned above, an appreciation of the roles of many, novel mechanisms in the induction of nociception, including neurotrophins, cytokines, Na⁺channels, protons, DH mechanisms of sensitization, $A\beta$ fibre-mediated mechanical allodynia, PAF phenotype changes in inflammatory and neuropathic pain, regenerative changes in damaged PAFs and sympathetic transmission.

An improved understanding of such mechanisms has contributed to a resurgence of optimism that it may be possible to prevent or interrupt mechanisms involved in the induction of pain. However, precisely which of this diversity of 'pronociceptive' mechanisms should most appropriately be manipulated remains to be elucidated. As suggested above (Section 15.1), in view of redundancy in nociceptive transmission and the existence of diffuse matrixes underlying pain sensation, any single mechanism may be insufficient. Indeed, it is worth reiterating that the distinctive quality of nociceptive pain relief afforded by μ -opioids may reflect multiple actions at all many hierarchical levels: PAF terminals, the DH; the thalamus and other supraspinal relay centres; the cortex and limbic system and brainstem centres modulating descending inhibition. Via such complementary and diverse actions, morphine effectively modifies both the sensory and emotional-cognitive dimensions of pain.

15.3. Concluding Comments

In conclusion, there is a growing realisation that a characterization of the molecular bases of endogenous mechanisms generating rather than inhibiting nociception may provide more fertile territory for the discovery of novel strategies to alleviate pain. Instead of activating antinociceptive mechanisms or interfering with pronociceptive processes, it would obviously be desirable to terminate the pathological events, such as metastatic disease or arthritis, underlying chronic and clinical pain, thereby rendering analgesics superfluous. Unfortunately, this is not currently foreseeable and remains a task for the next century. Today, the development of novel, improved analgesics remains a major challenge to which an improved understanding of mechanisms underlying the induction of pain will hopefully make a major contribution.

Acknowledgements—The author would like to thank C. Langaney, M. Soubeyran, L. Verrièle, A. Gobert, M. Brocco, E. Gayet and S. Dapremont for invaluable assistance in the preparation of this manuscript. J.-M. Besson, F. Cervero, A. Dickenson, A. Dray, M. Headley and B. Morris are also thanked for helpful comments.

REFERENCES

- Abbadie, C., Brown, J.-L., Mantyh, P. W. and Basbaum, A. I. (1996) Spinal cord substance P receptor immunoreactivity increases in both inflammatory and nerve injury models of persistent pain. *Neuroscience* **70**, 201–209.
- Abbadie, C., Trafton, J., Liu, H., Mantyh, P. W. and Basbaum, A. I. (1997) Inflammation increases the distribution of dorsal horn neurons that internalize the neurokinin-1 receptor in response to noxious and non-noxious stimulation. J. Neurosci. 17, 8049–8060.
- Abbott, F. V., Hong, Y. and Blier, P. (1996) Activation of 5-HT_{2A} receptors potentiates pain produced by inflammatory mediators. *Neuropharmacology* 35, 99–110.
- Abbott, F. V., Hong, Y. and Blier, P. (1997) Persisting sensitization of the behavioural response to formalin-induced injury in the rat through activation of serotonin_{2A} receptors. *Neuroscience* **77**, 575–584.
- Abdulla, F. A. and Smith, P. A. (1997) Ectopic α_2 -adrenoceptors couple to N-type Ca²⁺ channels in axotomized rat sensory neurons. J. Neurosci. **17**, 1633–1641.
- Abrahams, V. C., Lynn, B. and Richmond, F. J. R. (1984) Organization and sensory properties of small myelinated fibres in the dorsal cervical rami of the cat. J. Physiol. 347, 177–187.
- Abrams, B. M. (1996) Tutorial 27: the antidepressants, anticonvulsants and less frequently used primary and adjuvant analgesics. *Pain Digest.* 6, 374–381.
- Acheson, A., Conover, J. C., Fandl, J. P., Di Chiara, T. M., Russel, M., Thadani, A., Squinto, S. P., Yancopoulos, G. D. and Lindsay, R. M. (1995) A BDNF autocrine loop in adult sensory neurons prevents cell death. *Nature* 374, 450–453.
- Acs, G., Palkovits, M. and Blumberg, P. M. (1996) Specific binding of [³H]resiniferatoxin by human and rat preoptic area, locus coeruleus, medial hypothalamus, reticular formation and ventral thalamus membrane preparations. *Life Sci.* 59, 1899–1908.
- Acs, G., Biro, T., Acs, P., Modarres, S. and Blumberg, P. M. (1997) Differential activation and desensitization of sensory neurons by resiniferatoxin. J. Neurosci. 17, 5622–5628.
- Agius, E. and Cochard, P. (1998) Comparison of neurite outgrowth induced by intact and injured sciatic nerves: a confocal and functional analysis. J. Neurosci. 18, 328–338.
- Aguayo, A. J. (1985) Axonal regeneration from injured neurons in the adult mammalian central nervous system. In: *Synaptic Plastiticity*, pp. 457–484. Ed. C. W. Cotman. Raven Press, New York.
- Aguayo, L. G. and White, G. (1992) Effects of nerve growth factor on TTX- and capsaicin-sensitivity in adult rat sensory neurones. *Brain Res.* 570, 61–67.
- Aicher, S. A., Sharma, S., Cheng, P. Y. and Pickel, V. M. (1997) The *N*-methyl-D-aspartate (NMDA) receptor is postsynaptic to substance P-containing axon terminals in the rat superficial dorsal horn. *Brain Res.* **772**, 71–81.
- Aigner, L. and Caroni, P. (1995) Absence of persistent spreading, branching and adhesion in GAP-43 depleted growth cones. J. Cell. Biol. 128, 647–660.
- Airaksinen, M. S., Koltzenburg, M., Lewin, G. R., Masu, Y., Helbig, C., Wolf, E., Brem, G., Toyka, K. V., Thoenen, H. and Meyer, M. (1996) Specific sub-types of cutaneous mechanoreceptor require neurotrophin-3 following peripheral target innervation. *Neuron* 16, 287–295.
- Akaike, A., Kanedo, S., Tamura, Y., Nakata, N., Shiomi, H., Ushikubi, F. and Narumiya, S. (1994) Prostaglandin E₂ protects cultured cortical neurons against *N*-methyl-D-aspartate receptormediated glutamate cytotoxicity. *Brain Res.* 663, 237–243.
- Akaneya, Y., Tsumoto, T. and Hatanaka, H. (1996) Brain-derived neurotrophic factor blocks long-term depression in rat visual cortex. J. Neurophysiol. 76, 4198–4201.
- Akopian, A. N., Abson, N. C. and Wood, J. N. (1996a) Molecular genetic approaches to nociceptor development and function. *Trends Neurosci.* 19, 240–246.
- Akopian, A. N., Sivilotti, L. and Wood, J. N. (1996b) A tetrodotoxin-resistant voltage-gated sodium channel expressed by sensory neurons. *Nature* 379, 257–262.
- Al-Chaer, E. D., Lawand, N. B., Westlund, K. N. and Willis, W. D. (1996a) Visceral nociceptive input into the ventral posterolateral nucleus of the thalamus: a new function for the dorsal column pathway. J. Neurophysiol. 76, 2661–2674.
- Al-Chaer, E. D., Westlund, K. N. and Willis, W. D. (1996b) Potentiation of thalamic responses to colorectal distension by visceral inflammation. *NeuroReport* 7, 1635–1639.
- Al-Chaer, E. D., Westlund, K. N. and Willis, W. D. (1997a) Nucleus gracilis: an integrator for visceral and somatic information. J. Neurophysiol. 78, 521–527.

- Al-Chaer, E. D., Westlund, K. N. and Willis, W. D. (1997b) Sensitization of postsynaptic dorsal column neuronal responses by colon inflammation. *NeuroReport* 8, 3267–3273.
- Al-Ghoul, W. M., Volsi, G. L., Weinberg, R. J. and Rustioni, A. (1993) Glutamate immunocytochemistry in the dorsal horn after injury or stimulation of the sciatic nerve of rats. *Brain Res. Bull* **30**, 453–459.
- Allbritton, N. L., Meyer, T. and Stryer, L. (1992) Range of messenger action of calcium ion and ionositol 1,4,5-trisphosphate. *Science* 258, 1812–1815.
- Alcántara, S., Frisén, J., del Rio, J. A., Soriano, E., Barbacid, M. and Silos-Santiago, I. (1997) TrkB signalling is required for postnatal survival of CNS neurons, protects hippocampal and motor neurons from axotomy-induced cell death. J. Neurosci. 17, 3623-3633.
- Aldskogius, H., Arvidsson, J. and Grant, G. (1992) Axotomyinduced changes in primary sensory neurons. In: Sensory Neurons: Diversity, Development and Plasticity, pp. 363–383. Ed. S. A. Scott. Oxford University Press, Oxford.
- Aley, K. O. and Levine, J. D. (1997) Multiple receptors involved in peripheral α₂, μ, and A₁ antinociception, tolerance and withdrawal. J. Neurosci. 17, 735–744.
- Aley, K. O., Kinnman, E. and Levine, J. D. (1996) Delayed sympathectomy after a prolonged hyperalgesia results in a subsequent enhanced acute hyperalgesic response. *Neuroscience* 71, 1083–1090.
- Ali, Z., Meyer, R. A. and Campbell, J. N. (1996) Secondary hyperalgesia to mechanical but not heat stimuli following a capsaicin injection in hairy skin. *Pain* 68, 401–411.
- Alkon, D. L., Kubota, M., Neary, J. T., Naito, S., Coulter, D. and Rasmussen, H. (1986) C-Kinase activation prolongs Ca²⁺ dependant inactivation of K⁺currents. *Biochem. Biophys. Res. Commun.* 134, 1245–1253.
- Allen, B. J., Rogers, S. D., Ghilardi, J. R., Menning, P. M., Kuskowski, M. A., Basbaum, A. I., Simone, D. A. and Mantyh, P. W. (1997a) Noxious cutaneous thermal stimuli induce a graded release of endogenous substance P in the spinal cord: imaging peptide action in vivo. J. Neurosci. 17, 5921–5927.
- Allen, G., Buxton, R. B., Wong, E. C. and Courchesne, E. (1997b) Attentional activation of the cerebellum independent of motor involvement. *Science* 275, 1940–1943.
- Allen, G. V. and Pronych, S. P. (1997) Trigeminal autonomic pathways involved in nociception-induced reflex cardiovascular responses. *Brain Res.* **754**, 269–278.
- Alloway, K. D., Rosenthal, P. and Burton, H. (1989) Quantitative measurements of receptive field changes during antagonism of GABAergic transmission in primary somatosensory cortex of cat. *exp. Brain Res.* 78, 514–532.
- Almeida, A. and Lima, D. (1997) Activation by cutaneous or visceral noxious stimulation of spinal neurons projecting to the medullary dorsal reticular nucleus in the rat: a c-fos study. *Eur.* J. Neurosci. 9, 686–695.
- Almeida, A., Tjølsen, A., Lima, D., Coimbra, A. and Hole, K. (1996) The medullary dorsal reticular nucleus facilitates acute nociception in the rat. *Brain Res. Bull.* **39**, 7–15.
- Aloe, L., Turveri, M. A., Carcassi, U. and Levi-Montalcini, R. (1992) Nerve growth factor in synovial fluid of patients with chronic arthritis. *Arthritis Rheumat*, 35, 351–355.
- Aloe, L., Moroni, R., Angelucci, F. and Fiore, M. (1997) Role of TNF-α but not NGF in murine hyperalgesia induced by parasitic infection. *Psychopharmacology* 134, 287–292.
- Altar, C. A., Cai, N., Bliven, T., Juhasz, M., Conner, J. M., Acheson, A. L., Lindsay, R. M. and Wiegand, S. J. (1997) Anterograde transport of brain-derived neurotrophic factor and its role in the brain. *Nature* 389, 856–860.
- Altier, N. and Stewart, J. (1996) Opioid receptors in the ventral tegmental area contribute to stress-induced analgesia in the formalin test for tonic pain. *Brain Res.* 718, 203–206.
- Altier, N. and Stewart, J. (1997a) Neuropeptide FF in the VTA blocks the analgesic effects of both intra-VTA morphine and exposure to stress. *Brain Res.* 758, 250–254.
- Altier, N. and Stewart, J. (1997b) Tachykinin NK-1 and NK-3 selective agonists induce analgesia in the formalin test for tonic pain following intra-VTA or intra-accumbens microinfusions. *Behav. Brain Res.* 89, 151–165.
- Amann, R., Schuligoi, R., Lanz, I. and Peskar, B. A. (1996) Effect of a 5-lipoxygenase inhibitor on nerve growth factor-induced thermal hyperalgesia in the rat. *Eur. J. Pharmac.* 306, 89–91.
- Ambron, R. T. and Walters, E. T. (1996) Priming events and retrograde injury signals: a new perspective on the cellular and molecular biology of nerve regeneration. *Mol. Neurobiol.* 13, 61–79.

- Amir, R. and Devor, M. (1992) Axonal cross-excitation in nerveend neuromas: comparison of A- and C-fibers. J. Neurophysiol. 68, 1160–1166.
- Amir, R. and Devor, M. (1993) Ongoing activity in neuroma afferents bearing retrograde sprouts. *Brain Res.* 630, 283–288.
- Amir, R. and Devor, M. (1996) Chemically mediated cross-excitation in rat dorsal root ganglia. J. Neurosci. 16, 4733–4741.
- Amir, R. and Devor, M. (1997) Spike-evoked suppression and burst patterning in dorsal root ganglion neurons of the rat. J. Physiol. 501, 183–196.
- Anand, P., Parrett, A., Martin, J., Zeman, S., Foley, P., Swash, M., Leigh, P. N., Cedarbaum, J. M., Lindsay, R. M., Williams-Chestnut, R. E. and Smicropi, D. V. (1995) Regional changes of ciliary neurotrophic factor and nerve growth-factor levels in post-mortem spinal cord and cerebral cortex from patients with motor disease. *Nature Med.* 7, 168–172.
- Andersen, O. K., Gracely, R. H. and Arendt-Nielsen, L. (1995) Facilitation of the human nociceptive reflex by stimulation of $A\beta$ -fibres in a secondary hyperalgesic area sustained by nociceptive input from the primary hyperalgesic area. *Acta Physiol. Scand.* **155**, 87–97.
- Andersen, O. K., Felsby, S., Nicolaisen, L., Bjerring, P., Jensen, T. S. and Arendt-Nielsen, L. (1996) The effect of ketamine on stimulation of primary and secondary hyperalgesic areas induced by capsaicin—a double-blind, placebo-controlled, human experimental study. *Pain* 66, 51–62.
- Andersen, S. L. V. and Clausen, T. (1993) Calcitonin gene-related peptide stimulates active Na⁺/K⁺ transport in rat soleus muscle. Am. J. Physiol. 264, C419–C429.
- Andoh, T., Nagasawa, T. and Kuraishi, Y. (1996) Expression of tachykinin NK₁ receptor mRNA in dorsal root ganglia of the mouse. *Mol. Brain Res.* 35, 329–332.
- Andoh, T., Itoh, M. and Kuraishi, Y. (1997) Nociceptin gene expression in rat dorsal root ganglia induced by peripheral inflammation. *NeuroReport* 8, 2793–2796.
- Andreev, N. Y., Dimitrieva, N., Koltzenburg, M. and McMahon, S. B. (1995) Peripheral administration of nerve growth factor in the adult rat produces a thermal hyperalgesia that requires the presence of sympathetic post-ganglionic neurones. *Pain* 63, 109– 115.
- Anghinah, R., Oliveira, A. S. B. and Gabbai, A. A. (1994) Effect of baclofen on pain in diabetic neuropathy. *Muscle and Nerve*. 17, 958–959.
- Antal, M., Petkó, M., Polgár, E., Heizmann, C. W. and Storm-Mathisen, J. (1996) Direct evidence of an extensive gabaergic innervation of the spinal dorsal horn by fibres descending from the rostral ventromedial medulla. *Neuroscience* **73**, 509–518.
- Antonijevic, I., Mousa, S. A., Schäfer, M. and Stein, C. (1995) Perineurial defect and peripheral opioid analgesia in inflammation. J. Neurosci. 15, 165.
- Apfel, S. C., Wright, D. E., Wildeman, A. M., Dormia, C., Snider, W. D. and Kessler, J. A. (1996) Nerve growth factor regulates the expression of brain derived neurotrophic factor messenger RNA in the peripheral nervous system. *Mol. Cell Neurosci.* 7, 134–142.
- Apkarian, A. V. (1995a) Functional imaging of pain: new insights regarding the role of the cerebral cortex in human pain perception. *Neuroscience* 7, 279–293.
- Apkarian, A. V. (1995b) Thalamic anatomy and physiology of pain perception: connectivity, somato-visceral convergence and spatio-temporal dynamics of nociceptive information coding. In: *Forebrain Areas Involved in Pain Processing*, pp. 93–118. Eds. J.-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Apkarian, A. V., Stea, R. A., Manglos, S. H., Szeverenyi, N. M., King, R. B. and Thomas, F. D. (1992) Persistent pain inhibits contralateral somatosensory cortical activity in humans. *Neurosci. Lett.* **140**, 141–147.
- Apkarian, A. V., Brüggemann, J., Shi, T. and Airapetian, L. R. (1995) A thalamic model for true and referred visceral pain. In: *Visceral Pain, Progress in Pain Research and Management*, Vol. 5, pp. 217–259. Ed. G. F. Gebhart. IASP Press, Seattle.
- Appleton, I. (1997) Non-steroidal anti-inflamatory drugs and pain. In *The Pharmacology of Pain. Handbook of Experimental Pharmacology*, Vol. 130, pp. 43–60. Eds. A. Dickenson and J-M. Besson. Springer-Verlag, Berlin.
- Araki, T. and Milbrandt, J. (1996) Ninjurin, a novel adhesion molecule, is induced by nerve injury and promotes axonal growth. *Neuron* 17, 353–361.
- Ardenghi, P., Barros, D., Izquierdo, L. A., Bevilaqua, L., Schröder, N., Quevedo, J., Rodrigues, C., Madruga, M., Medina, J. H. and Izquierdo, I. (1997) Late and prolonged post-training memory modulation in entorhinal and parietal

cortex by drugs acting on the cAMP/protein kinase A signalling pathway. *Behav. Pharmacol.* **8**, 745–751. Arias-Negrette, S., Keller, K. and Chadee, K. (1995)

- Arias-Negrette, S., Keller, K. and Chadee, K. (1995) Proinflammatory cytokines regulate cyclooxygenase-2 mRNA expression in human macrophages. *Biochem. Biophys. Res. Commun.* 208, 582–589.
- Armstrong-James, M., Welker, E. and Callahan, C. A. (1993) The contribution of NMDA and non NMDA receptors to fast and slow transmission of sensory information in rat S1 barrel cortex. *J. Neurosci.* 13, 2149–2160.
- Arnér, S. and Meyerson, B. (1993) Opioids in neuropathic pain. Pain Digest 3, 15–22.
- Arnér, S., Lindblom, U., Meyerson, B. A. and Molander, C. (1990) Prolonged relief of neuralgia after regional anesthetic blocks: a call for further experimental and systematic clinical studies. *Pain* 43, 287–297.
- Arnim, T. V. and Maseri, A. (1987) Silent Ischemia: Current Concepts and Management. Eds. T. V. Arnim and A. Maseri, p. 211. Springer Verlag, New York.
- Arnold, J. M. C., Teasell, R. W., MacLeod, A. P., Brown, J. E. and Carruthers, S. G. (1993) Increased venous, alpha-adrenoceptor responsiveness in patients with reflex sympathetic dystrophy. Ann. Intern. Med. 118, 619–621.
- Arvidsson, U., Risling, M., Frisen, J., Piehl, F., Fried, K., Hokfelt, T. and Cullheim, S. (1994) TrkC-like immunoreactivity in the primate descending serotoninergic system. *Eur. J. Neurosci.* 6, 230–236.
- Arvidsson, U., Dado, R. J., Riedl, M., Lee, J. H., Law, P. Y., Loh, H. H., Elde, R. and Wessendorf, M. W. (1995) Deltaopioid receptor immunoreactivity. Distribution in brainstem and spinal cord, and relationship to biogenic amines and enkephalin. J. Neurosci. 15, 1215–1235.
- Arvin, B., Neville, L. F., Barone, F. C. and Feuerstein, G. Z. (1996) The role of inflammation and cytokines in brain injury. *Neurosci. Biobehav. Rev.* 20, 445–452.
- Ault, B. and Hildebrand, L. M. (1993) L-glutamate activates peripheral nociceptors. Agents Actions 39, 142.
- Averill, S., McMahon, S. B., Clary, D. O., Reichardt, L. F. and Priestley, J. V. (1995) Immmunocytochemical localization of TRKA receptors in chemically identified subgroups of adult rat sensory neurons. *Eur. J. Neurosci.* 7, 1484–1494.
- Aylwin, M. L., Horowitz, J. M. and Bonham, A. C. (1997) NMDA receptors contribute to primary visceral afferent transmission in the nucleus of the solitary tract. J. Neurophysiol. 77, 2539–2548.
- Azkue, J. J., Knöpfel, T., Kunh, R., Mateos, J. M. and Grandes, P. (1997) Distribution of the metabotropic glutamate receptor subtype mGluR5 in rat midbrain periaqueductal grey and relationship with ascending spinofugal efferents. *Neurosci. Lett.* 228, 1–4.
- Babbedge, R. C., Sper, A. J., Gentry, C. T., Hood, V. C., Campbell, E. A. and Urban, L. (1996) In vitro characterization of a peripheral afferent pathway of the rat after chronic sciatic nerve section. J. Neurophysiol. 76, 3169–3177.
- Bagust, J., Zhang, L. and Owen, D. (1997) Toxin I, but not 4aminopyridine, blocks the late inhibitory component of the dorsal root reflex in an isolated preparation of rat spinal cord. *Brain Res.* 773, 181–189.
- Bahr, R., Blumberg, H. and Janig, W. (1981) Do dichotomizing afferent fibers exist which supply visceral organs as well as somatic structures? A contribution to the problem of referred pain. *Neurosci. Lett.* **24**, 25–28.
- Bakhle, Y. S. and Botting, R. M. (1996) Cyclooxygenase-2 and its regulation in inflammation. *Med. Inflammation* 5, 305–323.
- Banner, L. R. and Patterson, P. H. (1994) Major changes in the expression of the mRNAs for cholinergic differentiation factor/ leukemia inhibitor factor and its receptor after injury to adult peripheral nerves and ganglia. *Proc. natl Acad. Sci. U.S.A.* 91, 7109–7113.
- Barakat-Walter, I. (1996) Brain-derived neurotrophic factor-like immunoreactivity is localized mainly in small sensory neurons of rat dorsal root ganglia. J. Neurosci. Meth. 68, 281–288.
- Baranauskas, G. and Nistri, A. (1998) Sensitization of pain pathways in the spinal cord: cellular mechanisms. *Prog. Neurobiol.* 54, 349–365.
- Barann, M., Göthert, M., Fink, K. and Bonisch, H. (1993) Inhibition by anesthetics of ¹⁴C-guanidinium flux through the voltage-gated sodium channel and the cation channel of the 5-HT₃ receptor of NIE-115 neuroblastoma cells. *Naunyn-Schmiedeberg's Arch. Pharmac.* 347, 125.
- Barber, L. A. and Vasko, M. R. (1996) Activation of protein kinase C augments peptide release from rat sensory neurons. J. Neurochem. 67, 72–80.

- Barbour, B. and Häusser, M. (1997) Intersynaptic fusion of neurotransmitters. *Trends Neurosci.* 20, 377–384.
- Barbut, D., Polak, J. M. and Wall, P. D. (1981) Substance P in spinal cord dorsal horn decreases following peripheral nerve injury. *Brain Res.* 205, 289–298.
- Bardin, L., Bardin, M., Lavarenne, J. and Eschalier, A. (1997) Effect of intrathecal serotonin on nociception in rats: influence of the pain test used. *Exp. Brain Res.* 113, 81–87.
- Baron, R. and Maier, C. (1996) Reflex sympathetic dystrophy: skin blood flow, sympathetic vasoconstrictor reflexes and pain before and after surgical sympathetcomy. *Pain* 67, 317–326.
- Baron, R. and Saguer, M. (1993) Postherptic neuralgia: are C-nociceptors involved in signalling and maintenance of tactile allodynia? *Brain* 116, 1477–1496.
- Barondes, S. H., Cooper, D. N. W., Gitt, M. A. and Leffler, H. (1994) Galectins. Structure and function of a large family of animal lectins. J. Biol. Chem. 269, 20807–20810.
- Barria, A., Muller, D., Derkach, V., Griffith, L. C. and Soderling, T. R. (1997) Regulatory phosphorylation of AMPA-type glutamate receptors by CaM-KII during long-term potentiation. *Science* 276, 2042–2045.
- Basbaum, A. I. and Fields, H. L. (1984) Endogenous pain control systems: brainstem spinal pathways and endorphin circuitry. A. *Rev. Neurosci.* 7, 309–338.
- Bassilanat, F., Champigny, G., Waldmann, R., de Weille, J. R., Heurteaux, C. and Lazdunski, M. (1997) The acid-sensitive ionic channel subunit ASIC and the mammalian degenerin MDEG form a heteromultimeric H⁺-gated Na⁺ channel with novel properties. J. Biol. Chem. 272, 28819–28822.
- Battaglia, G. and Rustioni, A. (1988) Coexistence of glutamate and substance P in dorsal root ganglion cells of the rat and monkey. J. Comp. Neurol. 277, 302–312.
- Beiche, F., Scheuerer, S., Brune, K., Geisslinger, G. and Goppelt-Struebe, M. (1996) Up-regulation of cyclooxygenase-2 mRNA in the rat spinal cord following peripheral inflammation. *FEBS Lett.* **390**, 165–169.
- Belemonte, C. and Cervero, F. (1996) Neurobiology of Nociceptors, p. 531. Oxford University Press, Oxford.
- Ben-Ari, Y., Anikszteijn, L. and Bregestovski, P. (1992) Protein kinase C modulation of NMDA currents: an important link for LTP induction. *Trends Neurosci.* 15, 333–339.
- Ben-Ari, Y., Khazipov, R., Leinekugel, X., Caillard, O. and Gaiarsa, J. L. (1997) GABA_A, NMDA and AMPA receptors: a developmentally regulated 'ménage à trois'. *Trends Neurosci.* 20, 523–529.
- Benardo, L. S. (1997) Recruitment of GABAergic inhibition and synchronization of inhibitory interneurons in rat neocortex. J. Neurophysiol. 77, 3134–3144.
- Benedetti, F. (1997) Cholecystokinin type A and type B receptors and their modulation of opioid analgesia. *News Physiol. Sci.* 12, 263–268.
- Bennett, D. L., French, J. S., Priestley, J. V. and McMahon, S. B. (1996a) NGF but not NT-3 or BDNF prevents the A fiber sprouting into lamina II of the spinal cord that occurs following axotomy. *Mol. Cell. Neurosci.* 8, 211–220.
- Bennett, D. L., French, J. S., Priestley, J. V. and McMahon, S. B. (1996b) The effect of BDNF on *c-fos* and NOS expression in dorsal horn neurons of the adult rat spinal cord. *Am. Soc. Neurosci.* **22**, 993.
- Bennett, G. J. and Xie, Y. K. (1988) A peripheral mononeuropathy in rat that produces disorders of pain sensation like those seen in man. *Pain* 33, 87–107.
- Benowitz, L. I. and Routtenberg, A. (1987) A membrane phosphoprotein associated with neural development, axonal regeneration, phospholipid metabolism and synaptic plasticity. *Trends Neurosci.* 10, 527–532.
- Benowitz, L. I. and Routtenberg, A. (1997) GAP-43: an intrinsic determinant of neuronal development and plasticity. *Trends Neurosci.* 20, 84–91.
- Bentley, G. N. and Gent, J. P. (1995) Neurokinin actions on substantia gelatinosa neurones in adult longitudinal spinal cord preparation. *Brain Res.* 673, 101–111.
- Berg, T. and Koteng, Ø. (1997) Signalling pathways in bradykininand nitric oxide-induced hypotension in the normotensive rat: role of K⁺-channels. *Br. J. Pharmac.* **121**, 1113–1120.
- Bergmann, I., Priestly, J. V., McMahon, S. B., Brocker, E. B., Toyka, K. V. and Koltzenburg, M. (1997) Analysis of cutaneous sensory neurons in transgenic mice lacking the low affinity neurotrophin receptor. *Eur. J. Neurosci.* 9, 18–28.
- Berkley, K. J. (1997) On the dorsal columns: translating basic research hypotheses to the clinic. *Pain* 70, 103–107.

- Berkley, K. J. and Hubscher, C. H. (1995) Are there separate central nervous system pathways for touch and pain? *Nature Med.* **1**, 766–773.
- Berkley, K. J., Hubscher, C. H. and Wall, P. D. (1993) Neuronal responses to stimulation of the cervix, uterus, colon and skin in the rat spinal cord. J. Neurophysiol. 69, 545–556.
- Bernard, J.-F. and Besson, J.-M. (1990) The spino (trigemino) pontoamygdaloid pathway: electrophysiological evidence for an involvement in pain processes. J. Neurophysiol. 63, 473–490.
- Bernard, J.-F., Huang, G. F. and Besson, J.-M. (1990) Effect of noxious somesthetic stimulation on the activity of neurons of the nucleus centralis of the amygdalia. *Brain Res.* 523, 347–350.
- Bernard, J.-F., Aldén, M. and Besson, J.-M. (1993) The organization of the efferent projections from the pontine parabrachial area to the amygdaloid complex: a Phaseolus vulgaris leucoagglutini (PHA-L) study in the rat. J. Comp. Neurol. 329, 201– 229.
- Bernard, J.-F., Huang, G. F. and Besson, J.-M. (1994) The parabrachial area: electrophysiological evidence for an involvement in visceral nociceptive processes. J. Neurophysiol. 71, 1646–1660.
- Bernard, J.-F., Bester, H. and Besson, J.-M. (1995) The spinoparabrachio-amygdaloid and -hypothalamic nociceptive pathways. In: *Forebrain Areas Involved in Pain Processing*, pp. 27– 48. Eds. J.-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Bernardi, P. S., Valtschanoff, J. G., Weinberg, R. J., Schmidt, H. H. H. W. and Rustioni, A. (1995) Synaptic interactions between primary afferent terminals and GABA and nitric oxide-synthesizing neurons in the superficial laminae of the rat spinal cord. *J. Neurosci.* 15, 1363–1371.
- Berninger, B. and Poo, M. M. (1996) Fast actions of neurotrophic factors. Curr. Opin. Neurobiol. 6, 324–330.
- Berninger, B., Garcia, D. E., Inagaki, N., Hahnel, C. and Lindholm, D. (1993) BDNF and NT-3 induce intracellular Ca²⁺ elevation in hippocampal neurones. *NeuroReport* 4, 1303– 1306.
- Berridge, M. J. (1993) Inositol trisphosphate and calcium signalling. Nature 361, 315–325.
- Berzaghi, M. P., Gutiérrez, R., Heinemann, U., Lindholm, D. and Thoenen, H. (1995) Neurotrophins induce acute transmittermediated changes in brain electrical activity. *Am. Soc. Neurosci.* 21, 545.
- Besson, J.-M. and Chaouch, A. (1987) Peripheral and spinal mechanisms of nociception. *Physiol. Rev.* 67, 67–186.
- Besson, J.-M., Guilbaud, G. and Ollat, H. (1994) Peripheral Neurons in Nociception: Physico-pharmacological Aspects, p. 271. John Libbey Eurotext, Paris.
- Besson, J.-M., Guilbaud, G. and Ollat, H. (1995) Forebrain Areas Involved in Pain Processing, p. 283. John Libbey Eurotext, Paris.
- Bessou, P. and Perl, E. R. (1969) Responses of cutaneous sensory units with unmyelinated fibres to noxious stimuli. J. Neurophysiol. 32, 1025–1043.
- Bester, H., Matsumoto, N., Besson, J-M. and Bernard, J-F. (1997) Further evidence for the involvement of the spinoparabrachial pathway in nociceptive processes: a c-fos study in the rat. J. Comp. Neurol. 383, 439–458.
- Bettler, B. and Mulle, C. (1995) Review: neurotransmitter receptors II. AMPA and kainate receptors. *Neuropharmacology* 34, 123–139.
- Bevan, S. J. and Yeats, J. (1991) Protons activate a cation conductance in a sub-population of rat dorsal root ganglion neurones. *J. Physiol.* **433**, 145–161.
- Bevan, S. J. and Geppetti, P. (1994) Protons: small stimulants of capsaicin-sensitive sensory nerves. *Trends Neurosci.* 17, 509–512.
- Bevan, S. J. and Winter, J. (1995) Nerve growth factor (NGF) differentially regulates the chemosensitivity of adult rat cultured sensory neurons. J. Neurosci. 15, 4918–4926.
- Bevan, S. J., Forbes, C. A. and Winter, J. (1993) Protons and capsaicin activate the same ion channels in rat isolated dorsal root ganglion neurones. J. Physiol. 459, 401.
- Beydoun, A., Dyke, D. B. S., Morrow, T. J. and Casey, K. L. (1996) Topical capsaicin selectively attenuates heat pain and $A\delta$ fiber-mediated laser-evoked potentials. *Pain* **65**, 189–196.
- Bezzi, P., Carmignoto, G., Pasti, L., Vesce, S., Rossi, D., Rizzini, B. L., Pozzan, T. and Volterra, A. (1998) Prostaglandins stimulate calcium-dependent glutamate release in astrocytes. *Nature* **391**, 281–285.
- Bhisitkul, R. B., Kocsis, J. D., Gordon, T. R. and Waxman, S. G. (1990) Trophic influence of the distal nerve segment on GABA_A receptor expression in axotomized adult sensory neurons. *Exp. Neurol.* 109, 273–278.

- Bianchi, M. and Panerai, A. E. (1996) The dose-related effects of paracetamol on hyperalgesia and nociception in the rat. Br. J. Pharmac. 117, 130–132.
- Biella, G., Riva, L. and Sotgiu, M. L. (1997) Interaction between neurons in different laminae of the dorsal horn of the spinal cord: a correlation study in normal and neuropathic pain. *Eur.* J. Neurosci. 9, 1017–1025.
- Bileviciute, I., Ahmed, M., Bergström, J., Ericsson-Dahlstrand, A., Kreicbergs, A. and Lundeberg, T. (1997a) Expression of corticotropin-releasing factor in the peripheral nervous system of the rat. *NeuroReport* 8, 3127–3130.
- Bileviciute, I., Theodorsson, E. and Lundeberg, T. (1997b) Effects of histamine on neuropeptide release into the knee joint perfusate and cerebrospinal fluid in rats. *Neurosci. Lett.* 226, 9–12.
- Birrell, G. J., McQueen, D. S., Iggo, A., Coleman, R. A. and Grubb, B. D. (1991) PGI₂-induced activation and sensitization of articular mechanonociceptors. *Neurosci. Lett.* **124**, 5–8.
- Bisgaard, H. and Kristensen, J. K. (1985) Leukotriene B4 produces hyperalgesia in humans. *Prostaglandins* 30, 791–797.
- Bito, H., Deisseroth, K. and Tsien, R. W. (1997) Ca²⁺-dependent regulation in neuronal gene expression. *Curr. Opin. Neurobiol.* 7, 419–429.
- Black, J. A., Dib-Hajj, S., McNabola, K., Jeste, S., Rizzo, M. A., Kocsis, J. D. and Waxman, S. G. (1996) Spinal sensory neurons express multiple sodium channel α-submit mRNAs. *Mol. Brain Res.* 43, 117–131.
- Black, J. A., Langworthy, K., Hinson, A. W., Dib-Hajj, S. D. and Waxman, S. G. (1997) NGF has opposing effects on Na⁺ channel III and SNS gene expression in spinal sensory neurons. *NeuroReport* 8, 2331–2335.
- Blackburn-Munro, G. and Fleetwood-Walker, S. M. (1997) The effects of Na⁺ channel blockers on somatosensory processing by rat dorsal horn neurones. *NeuroReport* 8, 1549–1554.
- by rat dorsal horn neurones. NeuroReport 8, 1549–1554.
 Blackstone, C., Murphy, T. H., Moss, S. J., Baraban, J. M. and Huganir, R. L. (1994) Cyclic AMP and synaptic activity-dependent phosphorylation of AMPA-preferring glutamate receptors. J. Neurosci. 14, 7585–7593.
- Bland-Ward, P. A. and Humphrey, P. P. A. (1997) Acute nociception mediated by hindpaw P_{2X} receptor activation in the rat. *Br. J. Pharmac.* 122, 365–371.
- Blanquet, P. R. and Lamour, Y. (1997) Brain-derived neurotrophic factor increases Ca²⁺/calmodulin-dependent protein kinase 2 activity in hippocampus. J. Biol. Chem. 272, 24133–24136.
- Bleazard, L., Hill, R. G. and Morris, R. (1994) The correlation between the distribution of the NK₁ receptor and the actions of tachykinin agonists in the dorsal horn of the rat indicates that substance P does not have a functional role on substantia gelatinosa (LAMINA II) neurons. J. Neurosci. 14, 7655–7664.
- Bliss, T. V. and Collingridge, G. L. (1993) A synaptic model of memory: long-term potentiation in the hippocampus. *Nature* 361, 31–39.
- Blöchl, A. and Thoenen, H. (1995) Characterization of nerve growth factor (NGF) release from hippocampal neurons: evidence for a constitutive and an unconventional sodium-dependant regulated pathway. *Eur. J. Neurosci.* 7, 12208–12228.
- Blomqvist, A., Hermanson, O., Ericson, H. and Larhammar, D. (1994) Activation of a bulbospinal opioidergic projection by pain stimuli in the awake rat. *NeuroReport* 5, 461–464.
- Bloom, F. E. and Kupfer, D. J. (1995) *Psychopharmacology: The Fourth Generation of Progress*, p. 2002. Raven Press, New York.
- Blumberg, H. and Jänig, W. (1982) Changes in unmyelinated fibers including sympathetic postganglionic fibers of a skin nerve after peripheral neuroma formation. *Pain* 6, 173–183.
- Blumberg, H. and Jänig, W. (1994) Clinical manifestations of reflex sympathetic dystrophy and sympathetically maintained pain. In: *Textbook of Pain*, 3rd edn, 1524, pp. 685–699. Eds. P. D. Wall and R. Melzack. Churchill- Livingstone, Edinburgh.
- Boas, R. A., Covino, B. G. and Shahnarian, A. (1982) Analgesic responses to i.v. lignocaine. Br. J. Anaesthiol. 54, 501–505.
- Bockman, C. S., Gonzalez-Cabrera, I. and Abel, P. W. (1996) Alpha-2 adrenoceptor subtype causing nitric oxide-mediated vascular relaxation in rats. J. Pharmac. Exp. Ther. 278, 1235– 1243.
- Boess, F. G. and Martin, I. L. (1994) Molecular biology of 5-HT receptors. *Neuropharmacology* 33, 275–317.
- Boie, Y., Stocco, R., Sawyer, N., Slipetz, D. M., Ungrin, M. D., Neuschäfer-Rube, F., Püschel, G. P., Metters, K. M. and Abramovitz, M. (1997) Molecular cloning and characterization of the four rat prostaglandin E₂ prostanoid receptor subtypes. *Eur. J. Pharmac.* 340, 227–241.
- Boivie, J. (1994) Central pain. In *Textbook of Pain*, 3rd edn, pp. 871–902. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.

- Boivie, J. and Östeberg, A. (1995) Central pain syndrome. In: Forebrain Areas Involved in Pain Processing, pp. 239–252, Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Boivie, J., Leijon, G. and Johansson, I. (1989) Central post-stroke pain, a study of the mechanisms through analyses of the sensory abnormalities. *Pain* 37, 173–185.
- Bolden, D. A., Sternini, C. and Kruger, L. (1997) Gap-43 mRNA and calcitonin gene-related peptide mRNA expression in sensory neurons are increased following sympathectomy. *Brain Res. Bull.* 42, 39–50.
- Bolin, L. M., Verity, A. N., Silver, J. E., Shooter, E. M. and Abrams, J. S. (1995) Interleukin-6 production by Schwann cells and induction in sciatic nerve injury. J. Neurochem. 64, 850– 858.
- Bonfanti, L., Merighi, A. and Theodosis, D. T. (1996) Dorsal rhizotomy induces transient expression of the highly siaylated isoform of the neural cell adhesion molecule in neurons and astrocytes of the adult rat spinal cord. *Neuroscience* 74, 619– 623.
- Boolell, M. and Tooke, J. E. (1990) The skin hyperaemic response to local injection of substance P and capsaicin in diabetes mellitus. *Diabetic Med.* 7, 898–901.
- Borszcz, G. S., Johnson, C. P. and Williams, D. H. (1996) Increases in vocalization and motor reflex thresholds generated by the intrathecal administration of serotonin or norepinephrine. *Behav. Neurosci.* 110, 809–822.
- Bortolotto, Z., Bashir, Z. I., Davies, C. H. and Collingridge, G. L. (1994) A molecular switch activated by metabotropic glutamate receptors regulates induction of long-term potentiation. *Nature* 368, 740–743.
- Bortolotto, Z., Bashir, Z. I., Davies, C. H., Taira, T., Kaila, K. and Collingridge, G. L. (1995) Studies on the role of metabotropic glutamate receptors in long-term potentiation: some methodological considerations. *J. Neurosci. Meth.* 59, 19–24.Bossut, D. F., Shea, V. K. and Perl, E. R. (1996) Sympathectomy
- Bossut, D. F., Shea, V. K. and Perl, E. R. (1996) Sympathectomy induces adrenergic excitability of cutaneous C-fiber nociceptors. *J. Neurophysiol.* 75, 514–517.
- Bothwell, M. (1995) Functional interactions of neurotrophins and neurotrophin receptors. A. Rev. Neurosci. 18, 223–253.
 Bouaziz, H., Tong, C., Yoon, Y., Hood, D. D. and Eisenach, J.
- Bouaziz, H., Tong, C., Yoon, Y., Hood, D. D. and Eisenach, J. C. (1996) Intravenous opioids stimulate norepinephrine and acetylcholine release in spinal cord dorsal horn: Systematic studies in sheep and an observation in a human. *Anesthesiology* 84, 143–154.
- Bouchelet, I., Cohen, Z. V. I., Case, B., Séguéla, P. and Hamel, E. (1996) Differential expression of sumatriptan-sensitive 5-hydroxytryptamine receptors in human trigeminal ganglia and cerebral blood vessels. *Mol. Pharmac.* 50, 219–223.
- Bouhassira, D., Gall, O., Chitour, D. and Le Bars, D. (1995) Dorsal horn convergent neurons: negative feedback triggered by spatial summation of nociceptive afferents. *Pain* 62, 195–200. Bouhassira, D., Guirimand, F., Dupont, X. and Chauvin, M.
- Bouhassira, D., Guirimand, F., Dupont, X. and Chauvin, M. (1997) Reduction of temporal summation of nociceptive flexion (RIII) reflex and pain after low systemic doses of ketamine. *Am. Soc. Neurosci.* 23, 1015.
- Bouvier, M. M., Evans, M. L. and Benham, C. D. (1991) Calcium influx induced by stimulation of ATP receptors on neurones cultured from dorsal root ganglia. *Eur. J. Neurosci.* 3, 285–291.
- Bove, G. M. and Moskowitz, M. A. (1997) Primary afferent neurons innervating guinea pig dura. J. Neurophysiol. 77, 299–308.
- Bowersox, S. S., Gadbois, T., Singh, T., Pettus, M., Wang, Y. X. and Luther, R. R. (1996) Selective N-type neuronal voltage-sensitive calcium channel blocker, SNX-111, produces spinal antinociception in rat models of acute, persistent and neuropathic pain. J. Pharmac. Exp. Ther. 279, 1243–1249.
- Bowles, W. R., Jackson, D. L. and Hargreaves, K. M. (1994) Nerve growth factor evokes iGGRP release by a phosphorylation-dependent mechanism. *Am. Soc. Neurosci.* 20, 18.
- Bowsher, D. (1996) Central pain: clinical and physiological characteristics. J. Neurol. Neurosurg. Psychiatry 61, 62–69.
- Boxal, S. J., Berthele, A., Laurie, D. J., Sommer, B., Zieglgänsberger, W., Urban, L. and Tölle, T. R. (1998) Enhanced expression of metabotropic glutamate receptor 3 messenger RNA in the rat spinal cord during ultraviolet irradiation induced peripheral inflammation. *Neuroscience* 82, 591-602.
- Bracci, E., Ballerini, L. and Nistri, A. (1996a) Localization of rhythmogenic networks responsible for spontaneous bursts induced by strychnine and bicuculline in the rat isolated spinal cord. J. Neurosci. 16, 7063–7076.
- Bracci, E., Beato, M. and Nistri, A. (1996b) Afferent inputs modulate the activity of a rhythmic burst generator in the rat disinhibited spinal cord *in vitro*. J. Neurophysiol. 77, 3157–3167.

- Brain, S. D. and Cambridge, H. (1996) Calcitonin gene-related peptide: vasoactive effects and potential therapeutic role. *Gen. Pharmacol.* 27, 607–611.
- Bramham, C. R., Southared, T., Sarvey, J. M., Herkenhan, M. and Brady, L. S. (1996) Unilateral LTP triggers bilateral increases in hippocampal neurotrophin and *trk* receptor mRNA expression in behaving rats: evidence for interhemispheric communication. J. Comp. Neurol. 368, 371–382.
- Brandenburg, C. A., May, V. and Braas, K. M. (1997) Identification of endogenous sympathetic neuron pituitary adenylate cyclase-activating polypeptide (PACAP): depolarization regulates production and secretion through induction of multiple propeptide transcripts. J. Neurosci. 11, 4045–4055.
- Brauneis, U., Oz, M., Peoples, R. W., Weight, F. F. and Zhang, L. (1996) Differential sensitivity of recombinant *N*-methyl-paspartate receptor subunits to inhibition by dynorphin. *J. Pharmac. Exp. Ther.* **279**, 1063–1068.
- Breder, C. D., Dewitt, D. and Kraig, R. P. (1995) Characterisation of inducible cyclooxygenase in rat brain. J. Comp. Neurol. 355, 296–315.
- Bredesen, D. A. and Rabizadeh, S. (1997) p75^{NTR} and apoptosis: Trk-dependent and Trk-independent effects. *Trends Neurosci.* 20, 287–290.
- Brenman, J. E. and Bredt, D. S. (1997) Synaptic signaling by nitric oxide. *Curr. Opin. Neurobiol.* 7, 374–378.
- Brewer, K. L. and Ray, R. H. (1995) Acute reorganization of rat nucleus gracilis following peripheral denervation: role of GABA in functional plasticity. *Am. Soc. Neurosci.* 21, 110.
 Bromm, B., Jahnke, M. T. and Treede, R. D. (1984) Responses of
- Bromm, B., Jahnke, M. T. and Treede, R. D. (1984) Responses of human cutaneous afferents to CO₂ laser stimuli causing pain. *Exp. Brain Res.* 55, 15–166.
- Bromqvist, A. G. and Herzog, H. (1997) Y-receptor subtypes how many more? Trends Neurosci. 20, 294–298.
- Brown, H. C., Castano, A., Fearn, S., Townsend, M., Edwards, G., Streuli, C. and Perry, V. H. (1997) Adhesion molecules involved in macrophage responses to wallerian degeneration in the murine peripheral nervous system. *Eur. J. Neurosci.* 9, 2057–2063.
- Brown, P. B. and Culberson, J. L. (1981) Somatotopic organization of hindlimb cutaneous dorsal root projections to cat dorsal horn. J. Neurophysiol. 45, 137–143.
- Brüggemann, J., Shi, T. and Apkarian, A. V. (1994) Squirrel monkey lateral thalamus. II. Viscero-somatic convergent representation of urinary bladder, colon and esophagus. J. Neurosci. 14, 6796–6814.
- Brüggemann, J., Shi, T. and Apkarian, A. V. (1997) Viscerosomatic neurons in the primary somatosensory cortex (SI) of the squirrel monkey. *Brain Res.* **756**, 297–300.
- Bruinvels, A. T., Landwehrmeyer, B., Gustafson, E. L., Durkin, M. M., Mengod, G., Branchek, T. A., Hoyer, D. and Palacios, J. M. (1994) Localization of 5-HT_{1B}, 5-HT_{1Dz}, 5-HT_{1E} and 5-HT_{1F} receptor messenger RNA in rodent and primate brain. *Neuropharmacology* **33**, 367–389.
- Budai, D. and Larson, A. A. (1998) The involvement of metabotropic glutamate receptors in sensory transmission in dorsal horn of the rat spinal cord. *Neuroscience* 83, 571–580.
- Buffo, A., Holtmatt, A. J. D. G., Savio, T., Verbeek, J. S., Oberdick, J., Oestreicher, A. B., Gispen, W. H., Verhaagen, J., Rossi, F. and Strata, P. (1997) Targeted overexpression of the neurite growth-associated protein B-50/GAP-43 in cerebellar purkinje cells induces sprouting after axotomy but not axon regeneration into growth-permissive transplants. J. Neurophysiol. 17, 8778–8791.
- Bullitt, E. (1989) Induction of c-fos like protein within the lumbar spinal cord and thalamus of the rat following peripheral stimulation. *Brain Res.* 493, 391–397.
- Bullitt, E. (1990) Expression of c-fos-like protein as a marker for neuronal activity following noxious stimulation in the rat. J. Comp. Neurol. 296, 517–530.
- Bunge, R. P. (1993) Expanding roles for the Schwann cell: ensheathment, myelination, trophism and regeneration. *Curr. Opin. Neurobiol.* 3, 805–809.
- Burchiel, K. J. (1984) Effects of electrical and mechanical stimulation on two foci of spontaneous activity which develop in primary afferent neurons after peripheral axotomy. *Pain* 18, 249– 265.
- Burgess, G. M., Mullaney, I., McNeill, M., Dunn, P. and Rang, H. P. (1989) Second messengers involved in the action of bradykinin on cultured sensory neurons. *J. Neurosci.* 9 3314, 3325
- kinin on cultured sensory neurons. J. Neurosci. 9, 3314–3325.
 Buritova, J., Chapman, V., Honoré, P. and Besson, J. M. (1996)
 Selective cyclooxygenase-2 inhibition reduces carrageenan
 oedema and associated spinal c-Fos expression in the rat. Brain
 Res. 715, 217–220.

- Buritova, J., Chapman, V., Honoré, P. and Besson, J. M. (1997) The contribution of peripheral bradykinin B₂ receptors to carrageenan-evoked oedema and spinal c-fos expression in rats. *Eur. J. Pharmac.* 320, 73–80.
- Burkey, A. R., Carstens, E., Wenniger, J. J., Tang, J. and Jasmin, L. (1996) An opioidergic cortical antinociception triggering site in the agranular insular cortex of the rat that contributes to morphine antinociception. J. Neurosci. 16, 6612–6623.
- Burnstock, G. (1996) A unifying purinergic hypothesis for the initiation of pain. Lancet 347, 1604–1605.
- Burnstock, G. and Wood, J. N. (1996) Purinergic receptors: their role in nociception and primary afferent neurotransmission. *Curr. Opin. Neurobiol.* 6, 526–532.
- Burstein, R. and Potrebic, S. (1993) Retrograde labeling of neurons in the spinal cord that project directly to the amygdala or the orbital cortex in the rat. J. Comp. Neurol. 335, 469–485.
- Burstein, R., Dado, R. J. and Giesler, G. J. (1990) The cells of origin of the spinothalamic tract of the rat: a quantitative reexamination. *Brain Res.* 511, 329–337.
- Burstein, R., Dado, R. J., Cliffer, K. D. and Giesler, G. J. (1991) Physiological characterization of spinohypothalamic tract neurons in the lumbar enlargement of rats. J. Neurophysiol. 66, 261–284.
- Burstein, R., Falkowsky, O., Borsook, D. and Strassman, A. (1996) Distinct lateral and medial projections of the spinohypothalamic tract of the rat. J. Comp. Neurol. 373, 549–574.
- Bushnell, M. C. (1995) Thalamic processing of sensory-discriminative and affective-motivational dimensions of pain. In: *Forebrain* Areas Involved in Pain Processing, pp. 63–78. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Busija, D. W. and Thore, C. (1997) Modulation of prostaglandin production by nitric oxide in astroglia. *Prostaglandins*, *Leukotrienes and Essential Fatty Acids* 56, 355–359.
- Bustamante, D., Paeile, C., Willer, J. C. and Le Bars, D. (1997) Effects of intrathecal or intracerebroventricular administration of nonsteroidal anti-inflammatory drugs on a C-fiber reflex in rats. J. Pharmac. Exp. Ther. 281, 1381–1391.
 Buzzi, M. G., Carti, W. B., Shimizu, T., Health, H. and
- Buzzi, M. G., Carti, W. B., Shimizu, T., Health, H. and Moskowitz, M. A. (1991) Dihydroerotamine and sumatriptan attenuate levels of CGRP in plasma in rat superior sagittal sinus during electrical stimulation of the trigeminal ganglia. *Neuropharmacology* 30, 1193–1200.
- Cabelli, R. I., Hohn, A. and Shatz, C. (1995) Inhibition of ocular dominance column formation by infusion of NT-4/5 or BDNF. *Science* 267, 1662–1666.
- Calcutt, N. A., Jorge, M. C., Yaksh, T. L. and Chaplan, S. R. (1996) Tactile allodynia and formalin hyperalgesia in streptozotocin-diabetic rats: effects of insulin, aldose reductase inhibition and lidocaine. *Pain* 68, 293–299.
- Calford, M. B. and Tweedale, R. (1990) Interhemispheric transfer of plasticity in the cerebral cortex. *Science* **249**, 805–807.
- Camborieux, L., Bertrand, N. and Swerts, J. P. (1998) Changes in expression and localization of hemopexin and its transcripts in injured nervous system: a comparison of central and peripheral tissues. *Neuroscience* 82, 1039–1052.
- Cameron, A. A., Pover, C. M., Willis, W. D. and Coggeshall, R. E. (1992) Evidence that fine primary afferent axons innervate a wider territory in the superficial dorsal horn following peripheral axotomy. *Brain Res.* 575, 151–154.
- Cameron, A. A., Cliffer, K. D., Dougherty, P. M., Garrison, C. J., Willis, W. D. and Carlton, S. M. (1997) Time course of degenerative and regenerative changes in the dorsal horn in a rat model of paripheral pourpositive *J. Comm. Neurol.* **270**, *423*, 442
- model of peripheral neuropathy. J. Comp. Neurol. 379, 428–442.
 Campbell, J. N., Meyer, R. A. and Raja, S. N. (1992) Is nociceptor activation by alpha-1 adrenoceptors the culprit in sympathetically maintained pain. Am. Pain Soc. J. 13, 344–350.
- Campbell, J. N., Raja, S. N., Selig, D. K., Belzberg, A. J. and Meyer, R. A. (1994) Diagnosis and management of sympathetically maintained pain. In: *Progress in Pain Research and Management*, pp. 200–220. Eds. H. L. Fields and J. C. Liebeskind. IASP Press, Seattle.
- Canossa, M., Griesbeck, O., Berninger, B., Campana, G., Kolbeck, R. and Thoenen, H. (1997) Neurotrophin release by neurotrophins: implications for activity-dependent neuronal plasticity. *Proc. natl Acad. Sci. U.S.A.* 94, 13279–13286.
- Carafoli, E., Nicotera, P. and Santella, L. (1997) Calcium signalling in the cell nucleus. *Cell Calcium* 22, 313–319.
- Cardell, L-O., Stjärne, P., Wagstaff, S. J., Agusti, C. and Nadel, J. A. (1997) PACAP-induced plasma extravasation in rat skin. *Regulatory Peptides* **71**, 67–71.
- Cardenas, C. G., Del-Mar, L. P. and Scroggs, R. S. (1997) Two parallel signaling pathways couple 5-HT_{1A} recepors to N- and L-type calcium channels in C-like rat dorsal root ganglion cells. *J. Neurophysiol.* 77, 3284–3296.

- Carli, G. and Zimmerman, M. (1996) Towards the neurobiology of chronic pain. Prog. Brain Res. 110, 278.
- Carlton, S. M. and Coggeshall, R. E. (1997) Immunohistochemical localization of 5-HT_{2A} receptors in peripheral sensory axons in rat glabrous skin. *Brain Res.* **763**, 271–275.
- Carlton, S. M., Hargett, G. L. and Coggeshall, R. E. (1995) Localization and activation of glutamate receptors in unmyelinated axons of rat glabrous skin. *Neurosci. Lett.* **197**, 25–28. Carlton, S. M., Zhou, S. and Coggeshall, R. E. (1996)
- Carlton, S. M., Zhou, S. and Coggeshall, R. E. (1996) Localization and activation of substance P receptors in unmyelinated axons of rat glabrous skin. *Brain Res.* 734, 103–108.
- Carlton, S. M., Rees, H., Gondesen, K. and Willis, W. D. (1997) Dextrorphan attenuates responses of spinothalamic tract cells in normal and nerve-injured monkeys. *Neurosci. Lett.* 229, 169–172.
- Carlton, S. M., Chung, K., Ding, Z. and Coggeshall, R. E. (1998a) Glutamate receptors on postganglionic sympathetic axons. *Neuroscience* 83, 601–605.
- Carlton, S. M., Hargett, L. and Coggeshall, R. E. (1998b) Plasticity in α-amino-3-hydroxy-5-methyl-4-isoxazolepropionic acid receptor subunits in the rat dorsal horn following deafferentation. *Neurosci. Lett.* **242**, 21–24.
- Carmignoto, G., Pizzorusso, T., Tia, S. and Vicini, S. (1997) Brain-derived neurotrophic factor and nerve growth factor potentiate excitatory synaptic transmission in the rat visual cortex. J. Physiol. 498, 153–164.
- Carstens, E. (1995) Neural mechanisms of hyperalgesia: peripheral or central sensitization? *News Physiol. Sci.* 10, 260–265.
- Carstens, E. (1997) Responses of rat spinal dorsal horn neurons to intracutaneous microinjection of histamine, capsaicin and other irritants. J. Neurophysiol. 77, 2499–2514.
- Casabona, G. (1997) Intracellular signal modulation: a pivotal role for protein kinase C. Prog. Neuro-Psychopharmacol. Biol. Psychiat. 21, 407–425.
- Casey, K. L. (1996) Match and mismatch: identifying the neuronal determinants of pain. Ann. Intern. Med. 124, 995–998.
- Casey, K. L. and Minoshima, S. (1995) The forebrain network for pain: an emerging image. In: *Forebrain Areas Involved in Pain Processing*, pp. 213–228. Eds. J-M. Besson, O. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Casey, K. L., Minoshima, S., Berger, K. L., Koeppe, R. A., Morrow, T. J. and Frey, K. A. (1994) Positron emission tomographic analysis of cerebral structures activated specifically by repetitive noxious heat stimuli. *J. Neurophysiol.* **71**, 802–807.
- Casey, K. L., Beydoun, A., Boivie, J., Sjolund, B., Holmgren, H., Leijon, G., Morrow, T. J. and Rosen, I. (1996a) Laser-evoked cerebral potentials and sensory function in patients with central pain. *Pain* 64, 485–491.
- Casey, K. L., Minoshima, S., Morrow, T. J. and Koeppe, R. A. (1996b) Comparison of human cerebral activation patterns during cutaneous warmth, heat pain and deep cold pain. J. Neurophysiol. 76, 571–581.
- Casey, K. L., Ledoux, J., Gabriel, M., Lenz, F. A., Gracely, R. H., Zirh, A. T., Romanoski, A. J., Staats, P. and Dougherty, P. M. (1997) The sensory-limbic model of pain memory: connections from thalamus to the limbic system mediate the learned component of the affective dimension of pain. *Pain Forum* 6, 22–43.
- Castrén, E., Zafra, F., Thoenen, H. and Lindholm, D. (1992) Light regulates expression of brain-derived neurotrophic factor mRNA in rat visual cortex. *Proc. natl Acad. Sci. U.S.A.* 89, 9444–9448.
- Castrén, E., Bergazhi, M. D. P., Lindholm, D. and Thoenen, H. (1993a) Differential effects of MK-801 on brain derived neurotrophic factor mRNA levels in different regions of the rat brain. *Exp. Neurol.* 122, 244–252.
- Castrén, E., Pitkanen, M., Sirvol, J., Parsadanian, A. and Linholm, D. (1993b) The induction of LTP increases BDNF and NGF mRNA but decreases NT-3 m RNA in the dentate gyrus. *NeuroReport* 4, 895–898.
- Castro, M. E., Pascual, J., Romon, T., Del Arco, C., Del Olmo, E. and Pazos, A. (1997) Differential distribution of [³H]sumatriptan binding sites (5-HT_{1B}, 5-HT_{1D} and 5-HT_{1F} receptors) in human brain: focus on brainstem and spinal cord. *Neuropharmacology* **36**, 535–542.
- Castro-Alamancos, M. A. (1997) Short-term plasticity in thalamocortical pathways: cellular mechanisms and functional roles. *Rev. Neurosci.* 8, 95–116.
- Castro-Alamancos, M. A. and Connors, B. W. (1996) Short-term synaptic enhancement and long-term potentiation in neocortex. *Proc. natl Acad. Sci. U.S.A.* 93, 1335–1339.
- Castro-Alamancos, M. A. and Connors, B. W. (1997) Thalamocortical synapses. Prog. Neurobiol. 51, 581–606.

- Castro-Lopes, J. M., Coimbra, A., Grant, G. and Arvidsson, J. (1990) Ultrastrutural changes of the central scalloped (C1) primary afferent endings of synaptic glomeruli in the substantia gelatinosa rolandi of the rat after peripheral neurotomy. J. Neurocytol. 19, 329-337.
- Castro-Lopes, J. M., Tavares, I., Tölle, T. R., Coito, A. and Coimbra, A. (1992) Increase in GABAergic cells and GABA levels in the spinal cord in unilateral inflammation of the hindlimb in the rat. Eur. J. Neurosci. 4, 296-301.
- Castro-Lopes, J. M., Tavares, I. and Coimbra, A. (1993) Gaba decreases in the spinal cord dorsal horn after peripheral neurectomy. Brain Res. 620, 287-291.
- Castro-Lopes, J. M., Tavares, I., Tölle, T. R. and Coimbra, A. (1994) Carrageenan-induced inflammation of the hind foot provokes a rise of GABA-immunoreactive cells in the rat spinal cord that is prevented by peripheral neurectomy or neonatal capsacin treatment. Pain 56, 193-201
- Castro-Lopes, J. M., Malcangio, M., Pan, B. and Bowery, N. G (1995) Complex changes of GABAA and GABAB receptor binding in the spinal cord dorsal horn following peripheral inflammation or neurectomy. Brain Res. 679, 289-297
- Caterina, M. J., Schumacher, M. A., Tominaga, M., Rosen, T. A., Levine, J. D. and Julius, D. (1997) The capsaicin receptor: a heat-activated ion channel in the pain pathway. Nature 389, 816-824.
- Catheline, G., Kayser, V. and Guilbaud, G. (1996) Further evidence for a peripheral component in the enhanced antinociceptive effect of systemic morphine in mononeuropathic rats: involvement of κ but not δ -opioid receptors. Eur. J. Pharmac. 315, 135-143.
- Cattaert, D., El Manira, A. and Clarac, F. (1994) Chloride conductance produces both presynaptic inhibition and antidromic spikes in primary afferents. Brain Res. 666, 109-112.
- Cavala, C. and Goldman-Rakic, P. S. (1989) Posterior parietal cortex in rhesus monkey. I. Parcellation of areas based on distinctive limbic and sensory corticocortical connections. J. Comp. Neurol. 287, 393-421.
- Cerne, R., Jiang, M. and Randic, M. (1992) Cyclic adenosine 3'5'monophosphate potentiates excitatory amino acid and synaptic responses of rat spinal dorsal horn neurons. Brain Res. 596, 111 - 123.
- Cervero, F. (1982) Noxious intensities of visceral stimulation are required to activate viscero-somatic multireceptive neurons in the thoracic spinal cord of the cat. Brain Res. 240, 350-352.
- Cervero, F. (1994) Sensory innervation of the viscera: peripheral basis of visceral pain. Physiol. Rev. 74, 95-138.
- Cervero, F. (1995a) Visceral pain: mechanisms of peripheral and central sensitization. Ann. Med. 27. 235-239
- Cervero, F. (1995b) What is a nociceptor-specific (class3) cell? Pain 62. 123-124.
- Cervero, F. and Laird, J. M. A. (1996a) Mechanisms of allodynia: interactions between sensitive mechanoreceptors and nociceptors. NeuroReport 7, 526-528.
- Cervero, F. and Laird, J. M. A. (1996b) Mechanisms of touchevoked pain (allodynia): a new model. Pain 68, 13-23.
- Cervero, F., Gilbert, R., Hammond, R. G. E. and Tanner, J. (1993) Development of secondary hyperalgesia following nonpainful thermal stimulation of the skin: a psychophysical study in man. Pain 54, 181-189.
- Cervero, F., Meyer, R. A. and Campbell, J. N. (1994) A psychophysical study of secondary hyperalgesia: evidence for increased pain due to input from nociceptors. Pain 58, 21-28.
- Cesare, P. and McNaughton, P. (1996) A novel heat-activated current in nociceptive neurons and its sensitization by bradykinin. Proc. natl Acad. Sci. U.S.A. 93, 15435-15439.
- Chahl, L. A. and Iggo, A. (1977) The effects of bradykinin and prostaglandin E1 on rat cutaneous afferent nerve activity. Br. J. Pharmac. 59, 343–347.
- Chalazonitis, A., Kessler, J. A., Twardzik, D. R. and Morrison, R. S. (1992) Transforming growth factor α , but not epidermal growth factor, promotes the survival of sensory neurons in vitro. J. Neurosci. 12, 583-594.
- Challacombe, J. F., Snow, D. and Letourneau, P. C. (1996) Role of the cytoskeleton in growth cone motility and axonal elongation. Seminars Neurosci. 8, 67-80.
- Chao, C. C., Hu, S. and Petterson, P. K. (1996) Glia: the not so
- innocent bystanders. J. Neurovirol. 2, 234–239. Chaplan, S. R., Pogrel, J. W. and Yaksh, T. L. (1995) Role of voltage-dependent calcium channel subtypes in experimental tactile allodynia. J. Pharmac. Exp. Ther. **269**, 1117–1123. Chaplan, S. R., Malmberg, A. B. and Yaksh, T. L. (1997)
- Efficacity of spinal NMDA receptor antagonism in formalin

hyperalgesia and nerve injury evoked allodynia in the rat. J. Pharmac. Exp. Ther. 280, 829-838.

- Chapman, V. and Besson, J.-M. (1997) Pharmacological studies of nociceptive systems using the c-fos immunohistochemical technique: an indicator of noxiously activated spinal neurones. In The Pharmacology of Pain. Handbook of Experimental Pharmacology, Vol. 130, pp. 235–280. Eds. A. Dickenson and Jof Experimental M. Besson. Springer-Verlag, Berlin.
- Chapman, V., Honoré, P., Buritova, J. and Besson, J.-M. (1995) The contribution of NMDA receptor activation to spinal c-fos expression in a model of inflammatory pain. Br. J. Pharmac. 116, 1628-1634.
- Chapman, V., Buritova, J., Honoré, P. and Besson, J.-M. (1996) Physiological contributions of neurokinin 1 receptor activation and interactions with NMDA receptors to inflammatory-evoked spinal C-fos expression. J. Neurophysiol. 76, 1817-1827.
- Chapman, V., Wildman, M. A. and Dickenson, A. H. (1997) Distinct electrophysiological effects of two spinally administered membrane stabilising drugs, bupivacaine and lamotrigine. Pain 71, 285–295.
- Chatrian, G., Canfield, R., Knauss, T. and Lettich, E. (1975) Cerebral responses to electrical tooth pulp stimulation in man. Neurology 25, 745-757.
- Cheema, S. S., Richards, L., Murphy, M. and Bartlett, P. F. (1994) Leukemia inhibitory factor prevents the death of axoto mised sensory neurons in the dorsal root ganglia of the neonatal rat. J. Neurosci. Res. 37, 213-218.
- Chen, L. and Huang, L. Y. M. (1992) Protein kinase C reduces Mg^{2+} block of NMDA-receptor channels as a mechanism of modulation. Nature 356, 521-523.
- Chen, Y., Michaelis, M., Janig, W. and Devor, M. (1996) Adrenoreceptor subtype mediating sympathetic-sensory coupling in injured sensory neurons. J. Neurophysiol. 76, 3721-3730.
- Chergui, K., Svenningsson, P., Nomikos, G. G., Gonon, F., Fredholm, B. B. and Svennson, T. H. (1997) Increased expression of NGFI-A mRNA in the rat striatum following burst stimulation of the medial forebrain bundle. Eur. J. Neurosci. 9, 2370 - 2382
- Chessel, I. P., Anderson, I. K., Rupniak, H. T. R. and Humphrey, P. P. A. (1997a) Pharmacological characteristics of microglial P_{2X7} purinoceptors. Br. J. Pharmac. 120, 127P.
- Chessel, I. P., Michel, A. D. and Humphrey, P. P. A. (1997b) Properties of the pore-forming P_{2X7} purinoceptor in mouse NTW8 microglial cells. *Br. J. Pharmac.* **121**, 1429–1437.
- Chiang, C. Y., Hu, J. W. and Sessle, B. J. (1997) NMDA receptor involvement in neuroplastic changes induced by neonatal capsaicin treatment in trigeminal nociceptive neurons. Neurophysiol. 78, 2799–2803.
- Ching, Y. P., Averill, S., Wilkin, G. P., Wotherspoon, G. and Priestley, J. V. (1994) Serotonergic terminals express a growth associated protein (GAP-43) in the adult rat spinal cord. Neurosci. Lett. 167, 67-72.
- Chizh, B. A., Cumberbatch, M. J., Birch, P. J. and Headley, P. M. (1995) Endogenous modulation of excitatory amino acid responsiveness by tachykinin NK1 and NK2 receptors in the rat spinal cord. Br. J. Pharmac. 115, 1013-1019.
- Chizh, B. A., Cumberbatch, M. J., Herrero, J. F., Stirk, G. C. and Headley, P. M. (1997) Stimulus intensity, cell excitation and the N-methyl-D-aspartate receptor component of sensory responses in the rat spinal cord in vivo. Neuroscience 80, 251-265.
- Chmielowska, J., Carvell, G. E. and Simons, D. J. (1989) Spatial organization of thalamocortical and corticothalamic projection systems in the rat: SmI barrel cortex. J. Comp. Neurol. 285, 325-338.
- Cho, H.-J., Kim, D-S., Lee, N-H., Kim, J-K., Lee, K-M., Han, K-S., Kang, Y-N. and Kim, K-J. (1997a) Changes in the $\alpha_2\text{-adre-}$ nergic receptor subtypes gene expression in rat dorsal root ganglion in an experimental model of neuropathic pain. *NeuroReport* **8**, 3119–3122.
- Cho, H.-J., Kim, J-K., Zhou, X-F. and Rush, R. A. (1997b) Increased brain-derived neurotrophic factor immunoreactivity in rat dorsal root ganglia and spinal cord following peripheral inflammation. Brain Res. 764, 269-272
- Cho, H.-J., Kim, S. Y., Park, M. J., Kim, D. S., Kim, J. K. and Chu, M. Y. (1997c) Expression of mRNA for brain-derived neurotrophic factor in the dorsal root ganglion following peripheral inflammation. Brain Res. 749, 358-362.
- Cho, H.-J., Park, E. H., Bae, M. A. and Kim, J-K. (1998) Expression of mRNAs for preprotachykinin and nerve growth factor receptors in the dorsal root-ganglion following peripheral inflammation. Brain Res. 716, 197-201.

- Choi, B. and Rowbotham, M. C. (1997) Effect of adrenergic receptor activation on post-herpetic neuralgia pain and sensory disturbances. *Pain* 69, 55–63.
- Choi, D. W. and Rothman, S. M. (1990) The role of glutamate neurotoxicity in hypoxic-ischemic neuronal death. A. Rev. Neurosci. 13, 171–182.
- Chong, M. S., Fitzgerald, M., Winter, J., Hu-Tsai, M., Emson, P. C., Wiese, U. and Woolf, C. J. (1992) GAP-43 mRNA in rat spinal cord and dorsal root ganglia neurons: developmental changes and re-expression following peripheral nerve injury. *Eur. J. Neurosci.* 4, 883–895.
- Chong, M. S., Reynolds, M. L., Irwin, N., Coggeshall, R. E., Emson, P. C., Benowitz, L. I. and Woolf, C. J. (1994) GAP-43 expression in primary sensory neurons following central axotomy. J. Neurosci. 14, 4375–4384.
- Chong, M. S., Woolf, C. J., Turmaine, M., Emson, P. C. and Anderson, P. N. (1996) Intrinsic versus extrinsic factors determining regeneration of the central processes of rat dorsal root ganglion neurons: the influence of a peripheral nerve graft. J. Comp. Neurol. 370, 97–104.
- Chulder, E. H. and Dong, W. K. (1995) The role of the basal ganglia in nociception and pain. *Pain* **60**, 3–38.
- Chung, K., McNeill, D. L., Hulsebosch, C. E. and Coggeshall, R. E. (1989) Changes in dorsal horn synaptic disc numbers following unilateral dorsal rhizotomy. *J. Comp. Neurol.* 283, 568–577. Chung, K., Lee, B. H., Yoon, Y. W. and Chung, J. M. (1996)
- Chung, K., Lee, B. H., Yoon, Y. W. and Chung, J. M. (1996) Sympathetic sprouting in the dorsal root ganglia of the injured peripheral nerve in a rat neuropathic pain model. J. Comp. Neurol. 376, 241–252.
- Chung, K., Yoon, Y. W. and Chung, J. M. (1997) Sprouting sympathetic fibers form synaptic varicosities in the dorsal root ganglion of the rat with neuropathic injury. *Brain Res.* 751, 275–280.
- Clatworthy, A. L., Illich, P. A., Castro, G. A. and Walters, E. T. (1995) Role of peri-axonal inflammation on the development of thermal hyperalgesia and guarding behavior in a rat model of neuropathic pain. *Neurosci. Lett.* 184, 5–8.
- Clayton, J. S., Gaskin, P. J. and Beattle, D. T. (1997) Attenuation of fos-like immunoreactivity in the trigeminal nucleus caualis following trigeminovascular activation in the anaesthetised guinea-pig. *Brain Res.* 775, 74–80.
- Clément, C. I., Keay, K. A., Owler, B. K. and Bandler, R. (1996) Common patterns of increased and decreased c-fos expression in midbrain and pons evoked by noxious deep somatic and noxious visceral manipulations in the rat. J. Comp. Neurol. 366, 495–515.
- Clementi, E. and Meldolesi, J. (1997) The cross-talk between nitric oxide and Ca²⁺: a story with a complex past and a promising future. *Trends Pharmac.* **18**, 266–269.
- Cliffer, K. D., Burstein, R. and Giesler, G. J., Jr (1991) Distibutions of spinothalamic, spinohypothalamic, and spinotelencephalic fibers revealed by anterograde transport of PHA-L in rats. J. Neurosci. 11, 852–868.
- Cline, M. A., Ochoa, J. and Torebjörk, H. E. (1989) Chronic hyperalgesia and skin warming caused by sensitized C nociceptors. *Brain* **112**, 621–647.
- Coderre, T. J. (1992) Contribution of protein kinase C to persistent nociception following tissue injury in rats. *Neurosci. Lett.* 140, 181–184.
- Coderre, T. J. (1993) The role of excitatory amino acid receptors and intracellular messengers in persistent nociception after tissue injury in rats. *Molecular Biol.* 7, 229–246.
- Coderre, T. J. and Melzack, R. (1991) Central neural mediators of secondary hyperalgesia following heat injury in rats: neuropeptides and excitatory amino acids. *Neurosci. Lett.* 131, 71–74.
- Coderre, T. J. and Melzack, R. (1992a) The contribution of excitatory amino acids to central sensitization and persistent nociception after formalin-induced tissue injury. J. Neurosci. 12, 3665–3670.
- Coderre, T. J. and Melzack, R. (1992b) The role of NMDA receptor-operated calcium channels in persistent nociception after formalin-induced tissue injury. *J. Neurosci.* 3671–3675.
- Coderre, T. J. and Yashpal, K. (1994) Intracellular messengers contributing to persistent nociception and hyperalgesia induced by L-glutamate and substance P in the rat formalin pain model. *Eur. J. Neurosci.* 6, 1328–1334.
- Coderre, T. J., Katz, J., Vaccarino, A. L. and Melzack, R. (1993) Contribution of central neuroplasticity to pathological pain: review of clinical and experimental widence. *Pain* **52**, 259, 285
- review of clinical and experimental evidence. *Pain* **52**, 259–285. Coffield, J. A., Bowen, K. K. and Miletic, V. (1992) Retrograde tracing of projections between the nucleus submedius, the ventrolateral orbital cortex and the midbrain in the rat. *J. Comp. Neurol.* **321**, 488–499.

- Coggeshall, R. E. (1994) A possible relationship between neuropathic pain and central sensory sprouting following peripheral nerve lesions. In: *Peripheral Neurons in Nociception: Physicopharmacological Aspects*, pp. 201–208, Eds. J.-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Coggeshall, R. E. and Carlton, S. M. (1997) Receptor localization in the mammalian dorsal horn and primary afferent neurons. *Brain Res. Rev.* 24, 28–66.
- Coggeshall, R. E. and Carlton, S. M. (1998) Ultrastructural analysis of NMDA, AMPA and kainate receptors on unmyelinated and myelinated axons in the periphery. J. Comp. Neurol. 391, 78–86.
- Coggeshall, R. E., Reynolds, M. L. and Woolf, C. J. (1991) Distribution of the growth associated protein GAP-43 in the central processes of axotomized primary afferents in the adult rat spinal cord; presence of growth cone-like structures. *Neurosci. Lett.* **131**, 37–41.
- Coggeshall, R. E., Zhou, S. and Carlton, S. M. (1997) Opioid receptors on peripheral sensory neurons. *Brain Res.* 764, 126– 132.
- Coggins, P. J. and Zwiers, H. (1991) B-50 (GAP-43): biochemistry and functional neurochemistry of a neuron-specific phosphoprotein. J. Neurochem. 56, 1095–1106.
- Coghill, R. C., Price, D. D., Hayes, R. L. and Mayer, D. J. (1991) Spatial distribution of nociceptive processing in the rat spinal cord. *J. Neurophysiol.* **65**, 133–140.
- Coghill, R. C., Talbot, J. D., Evans, A. C., Meyer, E., Gjedde, A., Busnell, M. C. and Duncan, G. H. (1994) Distributed processing of pain and vibration by the human brain. J. Neurosci. 14, 4095–4108.
- Cohen, R. H. and Perl, E. R. (1988) Chemical factors in the sensitization of cutaneous nociceptors. *Prog. Brain Res.* 74, 201–206.
- Cohen, Z. and Hamel, E. (1996) Expression of 5-HT_{1Dβ}, 5-HT_{1F}, 5-HT_{2B}, 5-HT_{2C} and 5-HT₇ receptor mRNAs in human intraparenchymal microvessels and related cells in cultures. *Am. Soc. Neurosci.* 22, 1787.
- Cohen, Z., Bonvento, G., Lacombe, P. and Hamel, E. (1996) Serotonin in the regulation of brain microcirculation. *Prog. Neurobiol.* 50, 335–362.
- Colbert, C. M. and Johnston, D. (1998) Protein kinase C activation decreases activity-dependent attenuation of dendritic Na⁺-current in hippocampal Ca1 pyramidal neurons. J. Neurophysiol. 79, 491–495.
- Colbert, R. A., Balbi, D., Jonhnson, A., Bailey, J. A. and Allen, J. M. (1994) Vasoactive intestinal peptide stimulates neuropeptide Y gene expression and causes neurite extension in PC12 cells throught independant mechanisms. J. Neurosci. 14, 7141–7147.
- Coleman, R. A., Smith, W. L. and Narumiya, S. (1994) VIII. International Union of Pharmacology classification of prostanoid receptors: properties, distribution, and structure of the receptors and their subtypes. *Pharmac. Rev.* 46, 205–229.
- Collingridge, G. L. and Singer, W. (1990) Excitatory amino acid receptors and synaptic plasticity. *Trends Pharmac.* 11, 290–296.
- Collins, D. R. and Davies, S. N. (1998) Arachidonic acid metabolites and the synaptic potentiation evoked by activation of metabotropic glutamate receptors. *Eur. J. Pharmac.* 342, 213– 216.
- Collo, G., North, R. A., Kawashima, E., Merlo-Pich, E., Neidhart, S., Surprenant, A. and Buell, G. (1996) Cloning of P_{2X5} and P_{2X6} receptors and the distribution and properties of an extended family of ATP-gated ion channels. *J. Neurosci.* 15, 2495–2507.
- Collo, G., Neidhart, S., Kawashima, E., Kosco-Vilbois, M., North, R. A. and Buell, G. (1997) Tissue distribution of the P_{2X7} receptor. *Neuropharmacology* **36**, 1277–1283.
- Colmers, W. F. and Bleackman, D. (1994) Effects of neuropeptide Y on the electrical properties of neurons. *Trends Neurosci.* 17, 373–379.
- Colvin, L. A., Mark, M. A. and Duggan, A. W. (1997) The effect of a peripheral mononeuropathy on immunoreactive (ir)-galanin release in the spinal cord of the rat. *Brain Res.* 766, 259–261.
- Conn, P. J. and Pin, J. P. (1997) Pharmacology and functions of metabotropic glutamate receptors. A. Rev. Pharmac. 37, 205– 237.
- Conner, J. M., Lauterborn, J. C., Yan, Q., Gall, C. M. and Varon, S. (1997) Distribution of brain-derived neurotrophic factor (BDNF) protein and mRNA in the normal adult rat CNS: evidence for anterograde axonal transport. *J. Neurosci.* 17, 2295–2313.
- Connor, H. E. and Beattie, D. T. (1996) 5-HT receptor subtypes and migraine. In: *Migraine: Pharmacology and Genetics*, pp. 18– 28, Eds. M. Sandler, M. Ferrari and S. Harnett. Chapman and Hall, London.

- Connor, M. and Henderson, G. (1997) Bradykinin inhibition of N- and L-calcium channel currents in NG108-15 cells. *Neuropharmacology* 36, 115–124.
- Connor, T. J. and Leonard, B. E. (1998) Depression, stress and immunological activation: the role of cytokines in depressive disorders. *Life Sci.* 62, 583–606.
- Conti, F. and Hicks, T. P. (1996) *Excitatory Amino Acids and the Cerebral Cortex*, p. 492. MIT Press, Boston.
- Conti, F., Minelli, A., De Biasi, S. and Melone, M. (1997) Neuronal and glial localization of NMDA receptors in the cerebral cortex. *Molec. Neurobiol.* 14, 1–18.
- Contreras, D., Destexhe, A. and Steriade, M. (1997) Intracellular and computional characterization of the intracortical inhibitory control of synchronized thalamic inputs in vivo. J. Neurophysiol. 78, 335–350.
- Coogan, A. and O'Connor, J. J. (1997) Inhibition of NMDA receptor-mediated synaptic transmission in the rat dentate gyrus in vitro by IL-1β. NeuroReport 8, 2107–2110.
- Cook, A. J., Woolf, C. J. and Wall, P. D. (1986) Prolonged Cfibre mediated facilitation of the flexion reflex in the rat is not due to changes in afferent terminal or motoneurone excitability. *Neurosci. Lett.* **70**, 91–96.
- Cook, A. J., Woolf, C. J., Wall, P. D. and McMahon, S. B. (1987) Dynamic receptive field plasticity in rat spinal cord dorsal horn following C-primary afferent input. *Nature* 325, 151–153.
- Cook, S. P., Vulchanova, L., Hargreaves, K. M., Elde, R. and McCleskey, E. W. (1997) Distinct ATP receptors on pain-sensing and stretch-sensing neurons. *Nature* 387, 505–508.
- Corness, J., Shi, T. J., Xu, Z. Q., Brulet, P. and Hokfelt, T. (1996) Influence of leukemia inhibitory factor on galanin/GMAP and neuropeptide Y expression in mouse primary sensory neurons after axotomy. *Exp. Brain Res.* **112**, 79–88.
- Coutinho, S. V., Meller, S. T. and Gebhart, G. F. (1996) Intracolonic zymosan produces visceral hyperalgesia in the rat that is mediated by spinal NMDA and NMDA receptors. *Brain Res.* **736**, 7–15.
- Cowen, T. and Gavazzi, I. (1998) Plasticity in adult and aging sympathetic neurons. Prog. Neurobiol. 54, 249–288.
- Coyle, D. E. (1996) Efficacy of animal models for neuropathic pain. *Pain Digest* 6, 7–20.
- Craig, A. D. (1994) Emotional aspects of pain. In: *Textbook of Pain*, 3rd edn, pp. 261–274. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Craig, A. D. (1995) Supraspinal projections of lamina I neurons. In: Forebrain Areas Involved in Pain Processing, pp. 13–26. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Craig, A. D. (1996) Pain, temperature and the sense of the body. In: Somesthesia and the Neurobiology of the Somatosensory Cortex, pp. 27–39. Eds. O. Franzen, R. Johannson and L. Terenius. Birkhäuser, Basel.
- Craig, A. D., Wiegand, S. J. and Price, J. L. (1982) The thalamocortical projection of the nucleus submedius in the cat. J. Comp. Neurol. 206, 28–48.
- Craig, A. D., Bushnell, M. C., Zhang, E. T. and Blomqvist, A. (1994) A thalamic nucleus specific for pain and temperature sensation. *Nature* **372**, 770–773.
- Craig, A. D., Reiman, E. M., Evans, A. and Bushnell, M. C. (1996) Functional imaging of an illusion of pain. *Nature* 384, 258–260.
- Crawford, J. H., Wooton, J. F., Seabrook, G. R. and Scott, R. H. (1997) Activation of Ca²⁺-dependent currents in dorsal root ganglion neurons by metabotropic glutamate receptors and cyclic ADP-ribose precursors. J. Neurophysiol. 77, 2573–2584.
- Crea, F., Pupita, G., Galassi, A. R., El-Tamimi, H., Kaski, J. C., Davies, G. and Maseri, A. (1990) Role of adenosine in pathogenesis of anginal pain. *Circulation* 81, 164–172.
- Crick, F. and Koch, C. (1998) Constraints on cortical and thalamic projections: the no-strong-loops hypothesis. *Nature* 391, 245–251.
- Cui, J.-G., O'Connor, W. T., Ungerstedt, U., Linderoth, B. and Meyerson, B. A. (1997a) Spinal cord stimulation attenuates augmented dorsal horn release of excitatory amino acids in mononeuropathy via a GABAergic mechanism. *Pain* 73, 87–95.
- Cui, J.-G., Sollevi, A., Linderoth, B. and Meyerson, B. A. (1997b) Adenosine receptor activation suppresses tactile hypersensitivity and potentiates spinal cord stimulation in mononeuropathic rats. *Neurosci. Lett.* 223, 173–176.
- Culberson, J. L., Haines, D. E., Kimmel, D. L. and Brown, P. B. (1979) Contralateral projection of primary afferent fibers to mammalian spinal cord. *Exp. Neurol.* 64, 83–97.

- Cumberbatch, M. J., Chizh, B. A. and Headley, P. M. (1994) AMPA receptors have an equal role in spinal nociceptive and non-nociceptive transmission. *NeuroReport* 5, 877–880.
- Cumberbatch, M. J., Hill, R. G. and Hargreaves, R. J. (1997) Rizatriptan has central antinociceptive effects against durally evoked responses. *Eur. J. Pharmac.* **328**, 37–40. Cummins, T. R. and Waxman, S. G. (1997) Downregulation of
- Cummins, T. R. and Waxman, S. G. (1997) Downregulation of tetrodotoxin-resistant sodium currents and upregulation of a rapidly repriming tetrodotoxin-sensitive sodium current in small spinal sensory neurons after nerve injury. J. Neurosci. 17, 3503– 35014.
- Cunha, F. Q., Lorenzetti, B., Poole, S. and Ferreira, S. H. (1991) Interleukin-8 as a mediator of sympathetic pain. *Br. J. Pharmac.* **104**, 765–767.
- Cunha, F. Q., Poole, S., Lorenzetti, B. and Ferreira, S. H. (1992) The pivotal role of tumour necrosis factor α in the development of inflammatory hyperalgesia. Br. J. Pharmac. 107, 660–664.
- of inflammatory hyperalgesia. Br. J. Pharmac. **107**, 660–664. Curtis, R., Adryan, K. M., Zhu, Y., Harkness, P. J., Lindsay, R. M. and Di Stefano, S. S. (1993) Retrograde axonal transport of ciliary neurotrophic factor is increased by peripheral nerve injury. Nature **365**, 253–255.
- Cutrer, F. M., Moussasaoui, S., Garret, C. and Moskowitz, M. A. (1995) The non-peptide neurokinin-1 antagonist, RPR 100,893, decreases c-fos expression in trigeminal nucleus caudalis following noxious chemical meningeal stimulation. *Neuroscience* 64, 741–750.
- Dahl, J. B., Brennum, J., Arendt-Nielsen, L., Jensen, T. S. and Kehket, H. (1993) The effect of pre-versus post-injury infiltration with lidocaine on thermal and mechanical hyperalgesia after heat injury to the skin. *Pain* 53, 43–51.
- Damaj, M. I., Glennon, R. A. and Martin, B. R. (1994) Involvement of the serotonergic system in the hypoactive and antinociceptive effects of nicotine in mice. *Brain Res. Bull.* 33, 199–203.
- Damas, J., Liégeois, J.-F. and Bourdon, V. (1997) Involvement of 5-hydroxytryptamine and bradykinin in the hyperalgesia induced in rats by collagenase from clostridium histolyticum. *Naunyn-Schmiedeberg's. Arch. Pharmac.* 355, 566–570.
- D'Arcangelo, G., Dodt, H. U. and Zieglgänsberger, W. (1997) Reduction of excitation by interleukin-1β in rat neocortical slices visualized using infrared-darkfield videomicroscopy. *NeuroReport* 8, 2079–2083.
- Darian-Smith, C. and Gilbert, C. D. (1995) Topographic reorganization in the striate cortex of the adult cat and monkey is cortically mediated. J. Neurosci. 15, 1631–1647.
- Davidson, E. M., Coggeshall, R. E. and Carlton, S. M. (1997) Peripheral NMDA and non-NMDA glutamate receptors contribute to nociceptive behaviors in the rat formalin test. *NeuroReport* 8, 941–946.
- Davies, A. M. (1997) Neurotrophin switching: where does it stand? *Curr. Opin. Neurobiol.* 7, 110–118.
- Davis, A. J. and Perkins, M. N. (1994) The involvement of bradykinin B₁ and B₂ receptor mechanisms in cytokine-induced mechanical hyperalgesia in the rat. Br. J. Pharmac. 113, 63–68.
- Davis, A. J. and Perkins, M. N. (1996) Substance P and capsaicininduced mechanical hyperalgesia in the rat knee joint: the involvement of bradykinin B₁ and B₂ receptors. *Br. J. Pharmac.* 118, 2206–2212.
- Davis, A. J. and Perkins, M. N. (1997) des Arg BK-induced mechanical hyperalgesia and analgesia in the rat: involvement of IL-1, prostaglandins and peripheral opioids. *Br. J. Pharmac.* 120, 74P.
- Davis, B. M., Lewin, G. R., Mendell, L. M., Jones, M. E. and Albers, K. M. (1993a) Altered expression of nerve growth factor in the skin of transgenic mice leads to changes in response to mechanical stimuli. *Neuroscience* 56, 789–792.
- Davis, B. M., Katz, D. M., Seroogy, K. B. and Albers, K. M. (1994) Overexpression of NGF in transgenic mice induces novel sympathetic projections to primary sensory neurons. *Am. Soc. Neurosci.* 20, 1090.
- Davis, B. M., Wang, H. S., Albers, K. M., Carlson, S. L., Goodness, T. P. and McKinnon, D. (1996) Effects of NGF overexpression on anatomical and physiological properties of sympathetic postganglionic neurons. *Brain Res.* 724, 47–54.
- Davis, B. M., Fundin, B. T., Albers, K. M., Goodness, T. P., Cronk, K. M. and Rice, F. L. (1997a) Overexpression of nerve growth factor in skin causes preferential increases among innervation to specific sensory targets. J. Comp. Neurol. 387, 489– 506.
- Davis, K. D., Treede, R. D., Raja, S. N., Meyer, R. A. and Campbell, J. N. (1991) Topical application of clonidine relieves hyperalgesia in patients with sympathetically maintained pain. *Pain* 47, 309–317.

- Davis, K. D., Meyer, R. A. and Campbell, J. N. (1993b) Chemosensitivity and sensitization of nociceptive afferents that innervate the hairy skin of monkey. J. Neurophysiol. 69, 1071– 1081.
- Davis, K. D., Taylor, S. J., Crawley, A. P., Wood, M. L. and Mikulis, D. J. (1997b) Functional MRI of pain and attention related activations in the human cingulate cortex. J. *Neurophysiol.* 77, 3370–3370.
- Davisson, R. L., Shaffer, R. A., Johnson, A. K. and Lewis, S. J. (1996) Stimulation of lumbar sympathetic nerves may produce hindlimb vasodilatation via the release of pre-formed stores of nitrosyl factors. *Neuroscience* 72, 881–887.
- Dawson, N. J., Schmid, H. and Pierau, F. K. (1992) Pre-spinal convergence between thoracic and visceral nerves of the rat. *Neurosci. Lett.* 138, 149–152.
- Dawson, T. M. and Snyder, S. H. (1994) Gases as biological messengers: nitric oxide and carbon monoxide in the brain. J. Neurosci. 14, 5147–5153.
- Dechant, G. and Barde, Y-A. (1997) Signalling through the neurotrophin receptor p75^{NTR}. Curr. Opin. Neurobiol. 7, 413–418.
- De Groat, W. C. (1986) Spinal cord projections and neuropeptides in visceral afferent neurons. *Prog. Brain Res.* 67, 111–132.
- De Groot, J. F., Coggeshall, R. E. and Carlton, S. M. (1997) μ-Opioid receptor plasticity in the superficial dorsal horn following sciatic nerve transaction. *Am. Soc. Neurosci.* **23**, 1015.
- Deisseroth, K., Heist, E. K. and Tsien, R. W. (1998) Translocation of calmodulin to the nucleus supports CREB phosphorylation in hippocampal neurons. *Nature* 392, 198–202.
- Dekker, L. V., De Graan, P. N. E., Versteeg, D. H. G., Oestreicher, A. B. and Gispen, W. H. (1989) Phosphorylation of B-50 (GAP 43) is correlated with neurotransmitter release in rat hippocampal slices. J. Neurochem. 52, 24–30.
- De Keyser, J., Vauquelin, G., De Backer, J-P., De Vos, H. and Wilczak, N. (1993) What intracranial tissues in humans contain sumatriptan-sensitive serotonin 5-HT_{1A}-type receptors? *Neurosci. Lett.* **164**, 63–66.
- De Koninck, Y. and Henry, J. L. (1991) Substance P-mediated slow excitatory postsynaptic potential elicited in dorsal horn neurons in vivo by noxious stimulation. *Proc. natl Acad. Sci.* U.S.A. 88, 11344–11348.
- De Koninck, Y., Ribeiro-da-Sylva, A., Henry, J. L. and Cuello, A. C. (1992) Spinal neurons exhibiting a specific nociceptive response receive abundant substance P containing synaptic contacts. *Proc. natl Acad. Sci. U.S.A.* 89, 5073–5077.
- De Leo, J. A., Colburn, R. W., Nichols, M. and Malkotra, A. (1996) Interleukin-6-mediated hyperalgesia/allodynia and increased spinal IL-6 expression in a rat mononeuropathy model. J. Interferon Cytokine Res. 16, 695–700.
- De Leo, J. A., Colburn, R. W. and Rickman, A. J. (1997) Cytokine and growth factor immunohistochemical spinal profiles in two animal models of mononeuropathy. *Brain Res.* **759**, 50–57.
- De Riemer, S. A., Strong, J. A., Albert, K. A., Greengard, P. and Kaczmarek, L. K. (1985) Enhancement of calcium current in Aplysia neurones by phorbol esters and protein kinase C. *Nature* **313**, 313–316.
- De Vries, P., Heiligers, J. P. C., Villalon, C. M. and Saxena, P. R. (1996) Blockade of porcine carotid vascular responses to sumatriptan by GR 127,935, a selective 5-HT_{1D} receptor antagonist. *Br. J. Pharmac.* **118**, 85–92.
- Delcroix, J-D., Tomlinson, D. R. and Fernyhough, P. (1997) Diabetes and axotomy-induced deficits in retrograde axonal transport of nerve growth factor correlate with decreased levels of $p75^{LNTR}$ protein in lumbar dorsal root ganglia. *Mol. Brain Res.* **51**, 82–90.
- Dellemijn, P. L. I. and Fields, H. L. (1994) Do benzodiazepines have a role in chronic pain management? *Pain* 57, 137–152.
- Deller, T. and Frotscher, M. (1997) Lesion-induced plasticity of central neurons: sprouting of single fibres in the rat hippocampus after unilateral entorhinal cortex lesion. *Prog. Neurobiol.* 53, 687–727.
- Derbyshire, S. W. G., Jones, A. K. P., Gyulai, F., Clark, S., Townsend, D. and Firestone, L. L. (1997) Pain processing during three levels of noxious stimulation produces differential patterns of central activity. *Pain* 73, 431–445.
- Dettbarn, C. and Palade, P. (1997) Ca²⁺ feedback on "quantal" Ca²⁺ release involving ryanodine receptors. *Mol. Pharmac.* **52**, 1124–1130.
- Deutsch, P. J. and Sun, Y. (1992) The 38-amino acid form of pituitary adenylate cyclase-activating polypeptide stimulates dual signaling cascades in PC12 cells and promotes neurite outgrowth. J. Biol. Chem. 267, 5108–5113.

- Devinsky, O. and Luciano, D. (1993) The contributions of cingulate cortex to human behavior. In: *Neurobiology of Cingulate Cortex and Limbic Thalamus: A Comprehensive Handbook*, pp. 112–127. Eds. B. A. Vogt and M. Gabriel. Birkhauser, Basel.
- Devinsky, O., Morrell, M. J. and Vogt, B. A. (1995) Contributions of anterior cingulate cortex to behaviour. *Brain* 118, 279–306.
- Devor, M. (1983) Potassium channels moderate ectopic excitability of nerve-end neuromas in rats. *Neurosci. Lett.* **40**, 181–186.
- Devor, M (1994) The pathophysiology of damaged peripheral nerves. In: *Textbook of Pain*, 3rd edn, pp. 79–100. Eds. R. Melzack and P. D. Wall. Churchill-Livingstone, Edinburgh.
- Devor, M. (1995) Peripheral and central nervous system mechanisms of sympathetic related pain. *Pain Clinic.* 8, 5–14.
- Devor, M. and Claman, D. (1980) Mapping and plasticity of acid phosphatase afferents in the rat dorsal horn. *Brain Res.* 190, 17– 28.
- Devor, M. and Dubner, R. (1988) Centrifugal activity in afferent C-fibers influences the spontaneous afferent barrage generated in nerve-end neuromas. *Brain Res.* **446**, 396–400.
- Devor, M., Wall, P. D. and Catelan, N. (1992a) Systemic lidocaine silences ectopic neuroma and DRG discharge without blocking nerve conduction. *Pain* 48, 261–268.
- Devor, M., White, D. M., Goetzl, E. J. and Levine, J. D. (1992b) Eicosanoids, but not tackykinins, excite C-fiber endings in rat sciatic nerve-end neuromas. *NeuroReport* 3, 21–24.
- Devor, M., Govrin-Lippmann, R. and Angelides, K. (1993) Na⁺channel immunolocalization in peripheral mammalian axons and changes following nerve injury and neuroma formation. *J. Neurosci.* **13**, 1976–1992.
- Devor, M., Jänig, W. and Michaelis, M. (1994) Modulation of activity in dorsal root ganglion neurons by sympathetic activation in nerve-injured rats. J. Neurophysiol. 71, 38–47.
- Di Biasi, S., Amadeo, A., Spreafico, R. and Rustioni, A. (1994) Enrichment of glutamate immunoreactivity in lemniscal terminals of the ventropostero-lateral thalamic nucleus of the rat: an immunogold and WGA-HRP study. *Anat. Record* 240, 131–140.
- Di Chiara, G. and Imperato, A. (1988) Drugs abused by humans preferentially increase synaptic dopamine concentrations. The mesolimbic system of freely-moving rats. *Proc. natl Acad. Sci.* U.S.A. 85, 5274–5278.
- Di Piero, V., Fiacco, F., Tombari, D. and Pantano, P. (1997) Tonic pain: a SPECT study in normal subjects and cluster headache patients. *Pain* 70, 185–191.
- Di Stefano, P. S., Friedman, B., Radziejewski, C., Alexander, C., Boland, P., Schick, C. M., Lindsay, R. M. and Wiegand, S. J. (1992) The neurotrophins BDNF, NT-3 and NGF display distinct patterns of retrograde axonal transport in peripheral and central neurons. *Neuron* 8, 983–993.
- Diamond, J. and Springer, T. A. (1994) The dynamic regulation of integrin adhesiveness. *Curr. Biol.* 4, 506–517.
- Diamond, J., Coughlin, M., MacIntyre, M., Holmes, M. and Vishau, B. (1987) Evidence that endogenous β nerve growth factor is responsible for the collateral sprouting, but not the regeneration, of nociceptive axons in adult rats. *Proc. natl Acad. Sci.* U.S.A. **84**, 6596–6600.
- Diamond, J., Coughlin, M. and Holmes, M. (1992a) Endogenous NGF and nerve impulses regulate the collateral sprouting of sensory axons in the skin of the adult rat. J. Neurosci. 12, 1454– 1466.
- Diamond, J., Foerster, A., Holmes, M. and Coughlin, M. (1992b) Sensory nerves regenerate and restore function to the skin independently of endogenous NGF. J. Neurosci. 12, 1467–1476.
- Diaz, A. and Dickenson, A. H. (1997) Blockade of spinal N- and P-type, but not L-type, calcium channels inhibits the excitatory of rat dorsal horn neurones produced by subcutaneous formalin inflammation. *Pain* 69, 93–100.
- Dib-Hajj, S., Black, J. A., Felts, P. A. and Waxman, S. G. (1996) Down-regulation of transcripts for Na⁺ channel α-SNS in spinal sensory neurones following axotomy. *Proc. natl Acad. Sci. U.S.A.* 93, 14950–14954.
- Dickenson, A. (1997) Mechanisms of central hypersensitivity: excitatory amino acid mechanisms and their control. In: *The Pharmacology of Pain. Handbook of Experimental Pharmacology*, Vol. 130, pp. 167–210. Eds. A. Dickenson and J-M. Besson. Springer-Verlag, Berlin.
- Dickenson, A. and Besson, J.M. (Eds.) (1997) The Pharmacology of Pain. Handbook of Experimental Pharmacology, Vol. 130. p. 479. Springer-Verlag, Berlin.
- p. 479. Springer-Verlag, Berlin.
 Diemel, L. T., Brewster, W. J., Fernyhough, P. and Tomlinson, D. R. (1994) Expression of neuropeptides in experimental diabetes: effects of treatment with nerve growth factor. *Mol. Brain Res.* 21, 171–175.

- Dimitriadou, V., Rouleau, A., Trung Tuong, M. D., Newlands, G. J. F., Miller, H. R. P., Luffau, G., Schwartz, J. C. and Garbarg, M. (1997) Functional relationships between sensory nerve fibers and mast cells of dura mater in normal and inflammatory conditions. *Neuroscience* 77, 829–839.
- Dirig, D. M. and Yaksh, T. L. (1995) Intrathecal baclofen and muscimol, but not midazolam, are antinociceptive using the ratformalin model. J. Pharmac. Exp. Ther. 275, 219–227.
- Dirig, D. M., Konin, G. P., Isakson, P. C. and Yaksh, T. L. (1997) Effect of spinal cyclooxygenase inhibitors in rat using the formalin test and in vivo prostaglandin E_2 release. *Eur. J. Pharmac.* **331**, 155–160.
- Diversé-Pierluissi, M., Remmers, A. E., Neubig, R. R. and Dunlap, K. (1997) Novel form of crosstalk between G protein and tyrosine kinase pathways. *Proc. natl Acad. Sci. U.S.A.* 94, 5417–5421.
- Dmitrieva, N. and McMahon, S. B. (1996) Sensitization of visceral afferents by nerve growth factor in the adult rat. *Pain* 66, 87– 97.
- Dmitrieva, N., Shelton, D., Rice, A. C. and McMahon, S. B. (1997) The role of nerve growth factor in a model of visceral in-flammation. *Neuroscience* **78**, 449–459.
- Do, K. Q., Binnis, K. E. and Salt, T. E. (1994) Release of the nitric oxide precursor, arginine, from thalamic neurons in vivo. *Neuroscience* 60, 581–586.
- Doak, G. J. and Sawynok, J. (1995) Complex role of peripheral adenosine in the genesis of the response to subcutaneous formalin in the rat. *Eur. J. Pharmac.* 281, 311–318.
 Doak, G. J. and Sawynok, J. (1997) Formalin-induced nociceptive
- Doak, G. J. and Sawynok, J. (1997) Formalin-induced nociceptive behavior and edema: involvement of multiple peripheral 5-hydroxytryptamine receptor subtypes. *Neuroscience* 80, 939–949.
- Docherty, R. J., Robertson, B. and Bevan, S. J. (1991) Capsaicin causes prolonged inhibition of voltage-dependant calcium currents in adult rat dorsal root ganglion neurones in culture. *Neuroscience* 40, 513–521.
 Docherty, R. J., Yeats, J. C., Bevan, S. and Boddeke, H. W.
- Docherty, R. J., Yeats, J. C., Bevan, S. and Boddeke, H. W. (1996) Inhibition of calcineurin inhibits the desensitization of capsacion-evoked currents in cultured dorsal root ganglion neurones from adult rats. *Pflüger's Arch.* 431, 828–837.
- Docherty, R. J., Yeats, J. C. and Piper, A. S. (1997) Capsazepine block of voltage-activated calcium channels in adult rat dorsal root ganglion neurones in culture. *Br. J. Pharmac.* **121**, 1461– 1467.
- Donaldson, L. F., Harmar, A. J., McQueen, D. S. and Seckl, J. R. (1992) Increased expression of preprotachykinin, calcitonin gene-related peptide, but not vasoactive intestinal peptide messenger RNA in dorsal root ganglia during the development of adjuvant monoarthritis in the rat. *Mol. Brain Res.* 16, 143– 149.
- Dong, W. K. and Chudler, E. H. (1995) Cortical nociceptive mechanisms. A review of neurophysiological and behavioral evidence in the primate. In: *Forebrain Areas Involved in Pain Processing*, pp. 183–196. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Dong, W. K., Salonen, L. D., Kawakami, Y. and Shiwaku, T. (1989) Nociceptive responses of trigeminal neurons in SII-7b cortex of awake monkeys. *Brain Res.* 484, 314–324.
 Dong, W. K., Chudler, E. H., Sugiyama, K., Roberts, V. and
- Dong, W. K., Chudler, E. H., Sugiyama, K., Roberts, V. and Hayashi, T. (1994) Somatosensory, multisensory and talk-related neurons in cortical area 7b (PF) of unanesthetized monkeys. J. Neurophysiol. 72, 1–23.
- Dong, W. K., Hayashi, T., Roberts, V. J., Fusco, B. M. and Chudler, E. H. (1996) Behavioral outcome of posterior parietal cortex injury in the monkey. *Pain* 64, 579–587.
- Donnerer, J. and Amann, R. (1993) The inhibition of neurogenic inflammation. Gen. Pharmac. 24, 519–529.
- Donnerer, J., Schuligoi, R. and Stein, C. (1992) Increased content and transport of substance P and calcitonin gene-related peptide in sensory nerves innervating inflamed tissue: evidence for a regulatory function of nerve growth factor in vivo. *Neuroscience* 49, 693–698.
- Donnerer, J., Schuligoi, R., Stein, C. and Amann, R. (1993) Upregulation, release and axonal transport of substance P and calcitonin gene-related peptide in adjuvant inflammation and regulatory function of nerve growth factor. *Regulatory Peptides* 46, 150-154.
- Donnerer, J., Amann, R., Schuligoi, R. and Skofitsch, G. (1996) Complete recovery by nerve growth factor of neuropeptide content and function in capsaicin-impaired sensory neurons. *Brain Res.* 741, 103–108.
- Dostrovsky, J. O. and Craig, A. D. (1996) Cooling-specific spinothalamic neurons in the monkey. J. Neurophysiol. 76, 3656-3665.

- Dostrovsky, J. O. and Guilbaud, G. (1990) Nociceptive responses in medial thalamus of the normal and arthritic rat. *Pain* 40, 93– 104.
- Dostrovsky, J. O., Hutchison, W. D., Davis, K. D. and Lozano, A. (1995) Potential role of orbital and cingulate cortices in nociception. In: *Forebrain Areas Involved in Pain Processing*, pp. 171–182. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Doubell, T. P. and Woolf, C. J. (1997) Growth-associated protein 43 immunoreactivity in the superficial dorsal horn of the rat spinal cord is localized in atrophic C-fiber, and not in sprouted A-fiber, central terminals after peripheral nerve injury. J. Comp. Neurol. 386, 111–118.
- Doubell, T. P., Mannon, R. J. and Woolf, C. J. (1997) Intact sciatic myelinated primary afferent terminals collaterally sprout in the adult rat dorsal horn following section of a neighbouring peripheral nerve. J. Comp. Neurol. 380, 95–104.
- Dougherty, P. M. and Willis, W. D. (1992) Enhanced responses of spinothalamic tract neurons to excitatory amino acids accompany capsaicin-induced sensitization in the monkey. J. Neurosci. 12, 883–894.
- Dougherty, P. M., Garrison, C. J. and Carlton, S. M. (1992a) Differential influence of local anesthetics upon two models of experimentally induced peripheral mononeuropathy in the rat. *Brain Res.* 570, 109–115.
- Dougherty, P. M., Palecek, J., Paleckova, V., Sorkin, L. and Willis, W. D. (1992b) The role of NMDA and non-NMDA excitatory amino-acid receptors in the excitation of primate spinothalamic tract neurons by mechanical, chemical, thermal and electrical stimuli. J. Neurosci. 12, 3025–3041.
- Dougherty, P. M., Sluka, K. A., Sorkin, L. S., Westlund, K. N. and Willis, W. D. (1992c) Neural changes in acute arthritis in monkeys I. Parallel enhancement of responses of spinothalamic tract neurons to mechanical stimulation and excitatory amino acids. *Brain Res. Rev.* 17, 1–13.
- Dougherty, P. M., Palecek, J., Paleckova, V. and Willis, W. D. (1995) Infusion of substance P or Neurokinin A by microdialysis alters responses of primate spinothalamic tract neurons to cutaneous stimuli and to iontophoretically released excitatory amino-acids. *Pain* **61**, 411-425.
- amino-acids. Pain 61, 411-425.
 Dougherty, P. M., Li, Y. L., Lenz, F. A., Rowland, L. and Mittman, S. (1996) Evidence that excitatory amino acids mediate afferent input to the primate somatosensory thalamus. Brain Res. 728, 267-273.
- Doyle, C. A. and Hunt, S. P. (1997) Reduced nuclear factor κB (p65) expression in rat primary sensory neurons after peripheral nerve injury. *NeuroReport* **8**, 2937–2942.
- nerve injury. *NeuroReport* **8**, 2937–2942. Draisci, G., Kaiander, K. C., Dubner, R., Bennett, G. J. and Iadarola, M. J. (1991) Up-regulation of opioid gene expression in spinal cord evoked by experimental nerve injuries and inflammation. *Brain Res.* **560**, 186–192.
- Dray, A. (1996) Chemosensitivity of cultured dorsal root ganglia cells. *News Physiol. Sci.* 11, 288–292.
- Dray, A. and Bevan, S. J. (1993) Inflammation and hyperalgesia: highlighting the team effort. *Trends Pharmac. Sci.* 14, 287–290.
- Dray, A. and Perkins, M. N. (1993) Bradykinin and inflammatory pain. Trends Neurosci. 16, 99–104.
- Dray, A. and Urban, L. (1996) New pharmacological strategies for pain relief. A. Rev. Pharmac. Toxicol. 36, 253–280.
- Dray, A., Urban, L. and Dickenson, A. (1994) Pharmacology of chronic pain. *Trends Pharmac.* 15, 190–197.
- Drummond, P. D. (1996) Independent effects of ischaemia and noradrenaline on thermal hyperalgesia in capsaicin-treated skin. *Pain* 67, 129–133.
- Drummond, P. D., Finch, P. M. and Smythe, G. A. (1991) Reflex sympathetic dystrophy: the significance of differing plasma catecholamine concentrations in affected and unaffected limbs. *Brain* 2025–2036.
- Drummond, P. D., Skipworth, S. and Finch, P. M. (1996) α_1 -Adrenoceptors in normal and hyperalgesic human skin. *Clin. Sci.* **91**, 73–77.
- Duarte, I. D. G., Lorenzetti, B. B. and Ferreira, S. H. (1990) Peripheral analgesia and activation of the nitric oxide-cyclic GMP pathway. *Eur. J. Pharmac.* 186, 289.
- Duarte, I. D. G., dos Santos, I. R., Lorenzetti, B. B. and Ferreira, S. H. (1992) Analgesia by direct antagonism of nociceptor sensitization involves the arginine-nitric oxide-cGMP pathway. *Eur. J. Pharmac.* 217, 225–227.
- Dubner, R. (1992) Hyperalgesia and expanded receptive fields. *Pain* **48**, 3–4.
- Dubner, R. and Bennett, G. J. (1983) Spinal and trigeminal mechanisms of nociception. A. Rev. Neurosci. 6, 381–418.

- Dubner, R. and Ruda, M. A. (1992) Activity-dependant neuronal plasticity following tissue injury and inflammation. *Trends Neurosci.* 15, 96–103.
- Dubner, R., Kenshalo, D. R., Maixner, W., Bushnell, M. C. and Oliveras, J. L. (1989) The correlation of monkey medullary dorsal horn neuronal activity and the perceived intensity of noxious heat stimuli. J. Neurophysiol. 62, 450–457.
- Duggan, A. W., Hope, P. J., Jarrot, B., Schaible, H. G. and Fleetwood-Walker, S. M. (1990) Release, spread and persistence of immunoreactive neurokinin A in the dorsal horn of the cat following noxious cutaneous stimulation: studies with antibody microprobes. *Neuroscience* 35, 195–202.
- Dumoulin, A., Alonso, G., Privat, A. and Feldblum, S. (1996) Biphasic response of spinal GABAergic neurons after a lumbar rhizotomy in the adult rat. *Eur. J. Neurosci.* 8, 2553–2563.
- Dumuis, A., Sebben, M., Haynes, L., Pin, J. P. and Bockaert, J. (1988) NMDA receptors activate the arachidonic acid cascade system in striatal neurons. *Nature* 336, 68–70.
- Dupouy, V. and Zajac, J-M. (1997) Neuropeptide FF receptors control morphine-induced analgesia in the parafascicular nucleus and the dorsal raphe nucleus. *Eur. J. Pharmac.* 330, 129–137.
- Durham, P. L., Sharma, R. V. and Russo, A. F. (1997) Repression of the calcitonin gene-related peptide promoter by 5-HT₁ receptor activation. J. Neurosci. 17, 9545–9553.
- Dykes, R. W., Landry, P., Metherate, R. and Hicks, T. P. (1984) Functional role of GABA in the cat primary somatosensory cortex: shaping receptive fields of cortical neurons. J. Neurophysiol. 52, 1066–1093.
- Eaton, S. A. and Salt, T. E. (1995) The role of excitatory aminoacid receptors in thalamic nociception. In: *Forebrain Areas Involved in Pain Processing*, pp. 131–142. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Eaton, S. A. and Salt, T. E. (1996) Role of N-methyl-D-aspartate and metabotropic glutamate receptors in corticothalamic excitatory postsynaptic potentials in vivo. *Neuroscience* 73, 1–5.
- Ebadi, M., Bashir, R. M., Heidrick, M. L., Hamada, F. M., El Refaey, H., Hamed, A., Helal, G., Baxi, M. D., Cerutis, D. R. and Lassi, N. K. (1997) Neurotrophins and their receptors in nerve injury and repair. *Neurochem. Int.* **30**, 347–374.
- nerve injury and repair. *Neurochem. Int.* **30**, 347–374. Ebersberger, A., Anton, F., Tölle, T. R. and Zieglgänsberger, W. (1995) Morphine, 5-HT₂ and 5-HT₃ receptor antagonists reduce c-fos expression in the trigeminal nuclear complex following noxious chemical stimulation of the rat nasal mucosa. *Brain Res.* **676**, 336–342.
- Eblen-Zajjur, A. A. and Sandkühler, J. (1997) Synchronicity of nociceptive and non-nociceptive adjacent neurons in the spinal dorsal horn of the rat: stimulus-induced plasticity. *Neuroscience* 76, 39–54.
- Eder, J. (1997) Tumour necrosis factor α and interleukin 1 signalling: do MAPKK kinases connect it all? *Trends Pharmac. Sci.* **18**, 319–322.
- Edvinsson, L. (1982) Sympathetic control of cerebral circulation. *Trends Neurosci.* **5**, 425–429.
- Edvinsson, L., Cantera, L., Jansen-Olensen, I. and Uddman, R. (1997) Expression of calcitonin gene-related peptide₁ receptor mRNA in human trigeminal ganglia and cerebral arteries. *Neurosci. Lett.* **229**, 209–211.
- Ehlers, M. D., Kaplan, D. R., Price, D. L. and Koliatsos, V. E. (1995) NGF-stimulated retrograde transport of TrkA in the mammalian nervous system *J. Cell. Biol.* **130**, 1–8
- mammalian nervous system. J. Cell. Biol. 130, 1–8.
 Elliott, J. R. (1997) Slow Na⁺ channel inactivation and bursting discharge in a simple model axon: implications for neuropathic pain. Brain Res. 754, 221–226.
- Elmér, E., Kokaia, Z., Kokaia, M., Carnahan, J., Nawa, H. and Lindvall, O. (1998) Dynamic changes of brain-derived neurotrophic factor protein levels in the rat forebrain after single and recurring kindling-induced seizures. *Neuroscience* 83, 351–362.
- Elmquist, J. K., Scammell, T. E. and Saper, C. B. (1997) Mechanisms of CNS response to systemic immune challenge: the febrile response. *Trends Neurosci.* 20, 565–570.
- England, S., Bevan, S. J. and Docherty, R. J. (1996) PGE₂ modulates the tetrodotoxin-resistant sodium current in neonatal rat dorsal root ganglion neurones via the cyclic AMP-protein kinase A cascade. J. Physiol. 495, 429–440.
- Ericsson, A.-C., Blomqvist, A., Craig, A. D., Ottersen, O. P. and Broman, J. (1995) Evidence for glutamate as a neurotransmitter in trigemino- and spinothalamic tract terminals in the nucleus submedius of cats. *Eur. J. Neurosci.* 7, 305–317.
- Eriksson-Mjöberg, M., Kristiansson, M., Carlström, K., Ölund, A. and Eklund, J. (1997) Infiltration of morphine into an abdominal wound: effects on pain relief and endocrine/immune response. *Pain* 73, 355–360.

- Ernfors, P., Wetmore, C., Olson, L. and Persson, H. (1990) Identification of cells in rat brain and peripheral tissues expressing mRNA for members of the nerve growth factor family. *Neuron* 5, 511–526.
- Eschalier, A., Mestre, C., Dubray, C. and Ardid, D. (1994) Why are antidepressants effective as pain relief? CNS Drugs 2, 261– 267.
- Ewald, D. A., Matthies, J. G., Perney, T. M., Walker, M. W. and Miller, R. J. (1988) The effect of down regulation of protein kinase C on the inhibition of dorsal root ganglion neuron Ca²⁺ currents by neuropeptide. J. Neurosci. 8, 2447–2451.
- Faggin, B. M., Tri Ngyen, K. and Nicolelis, M. A. L. (1997) Immediate and simultaneous sensory reorganization at cortical and subcortical levels of the somatosensory system. *Neurobiology* 94, 9428–9433.
- Fan, P., Visentin, S. and Weight, F. F. (1994) Effects of cocaine on the serotonin-induced inward current in rat nodose ganglion neurones. J. Pharmac. exp. Ther. 271, 262–266.
 Fang, F. and Proudfit, H. K. (1996) Spinal cholinergic and mono-
- Fang, F. and Proudfit, H. K. (1996) Spinal cholinergic and monoamine receptors mediate the antinociceptive effect of morphine microinjected in the periaqueductal gray on the rat tail, but not the feet. *Brain Res.* **722**, 95–108.
- Farkas-Szallasi, T., Lundberg, J. M., Wiesenfeld-Hallin, Z., Hökfelt, T. and Szallasi, A. (1995) Increased levels of GMAP, VIP and nitric oxide synthase, and their mRNAs, in lumbar dorsal root ganglia of the rat following systemic resiniferatoxin treatment. *NeuroReport* 6, 2230–2234.
- Farkas-Szallasi, T., Bennett, G. J., Blumberg, P. M., Hökfelt, T., Lundberg, J. M. and Szallasi, A. (1996) Vanilloid receptor loss is independent of the messenger plasticity that follows systemic resiniferatoxin administration. *Brain Res.* **719**, 213–218.
- Fawcett, J. W. (1992) Intrinsic neuronal determinants of regeneration. *Trends Neurosci.* 15, 5–8.
- Fawcett, J. W., Aloyz, R., McLean, J. H., Pareek, S., Miller, F. D., McPherson, P. S. and Murphy, R. A. (1997) Detection of brain-derived neurotrophic factor in a vesicular fraction of brain synaptosomes. J. Biol. Chem. 272, 8837–8840.
- Fazeli, S. and Walsh, F. S. (1996) The role of cell adhesion molecules during the development and regeneration of the neuro-muscular system. *Neuroscience* 8, 367–377.
 Felsby, S., Nielsen, J., Arendt-Nielsen, L. and Jensen, T. S. (1995)
- Felsby, S., Nielsen, J., Arendt-Nielsen, L. and Jensen, T. S. (1995) NMDA receptor blockade in chronic neuropathic pain: a comparison of ketamine and magnesium chloride. *Pain* 64, 283–291.
- Fergus, A. and Lee, K. S. (1997) Regulation of cerebral microvessels by glutamatergic mechanisms. *Brain Res.* 754, 35–45.
- sels by glutamatergic mechanisms. *Brain Res.* **754**, 35–45. Fernyhough, P., Diemel, L. T., Brewster, W. J. and Tomlinson, D. R. (1994) Deficits in sciatic nerve neuropeptide content coincide with a reduction in target tissue nerve growth factor mRNA in streptozotocin-diabetic rats: effects of insulin treatment. *Neuroscience* **62**, 337–344.
- Ferrari, M. D. and Saxena, P. R. (1993) On serotonin and migraine: a clinical and pharmacological review. *Cephalalgia* 13, 151–165.
- Ferreira, S. H. and Lorenzetti, B. B. (1994) Glutamate spinal retrograde sensitization of primary sensory neurons associated with nociception. *Neuropharmacology* 33, 1479–1485.
- Ferreira, S. H. and Lorenzetti, B. B. (1996) Intrathecal administration of prostaglandin E₂ causes sensitization of the primary afferent neuron via the spinal release of glutamate. *Inflamm. Res.* 45, 499–502.
- Ferreira, S. H., Cunha, F. Q., Lorenzetti, B. B., Michelin, M. A., Perretti, M., Flower, R. J. and Poole, S. (1997) Role of lipocortin-1 in the anti-hyperalgesic actions of dexamethasone. *Br. J. Pharmac.* 121, 883–888.
- Fiallos-Estrada, C. E., Kummer, W., Mayer, B., Bravo, R., Zimmerman, M. and Hergegen, T. (1993) Long-lasting increase of nitric oxide synthase immunoreactivity, NADPH-diaphorase reaction and cJUN co-expression in dorsal root ganglion neurones following sciatic nerve transection. *Neurosci. Lett.* **150**, 169–173.
- Field, M. J., Oles, R. J., Lewis, A. S., McCleary, S., Hughes, J. and Singh, L. (1997) Gabapentin (neurontin) and S-(+)-3-isobutylgaba represent a novel class of selective antihyperalgesic agents. *Br. J. Pharmac.* **121**, 1513–1522.
- Fields, H. L. and Basbaum, A. I. (1994) Central nervous system mechanisms of pain modulation. In: *Textbook of Pain*, 3rd edn, pp. 243–257. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Fields, H. L. and Rowbotham, M. C. (1994) Multiple mechanisms of neuropathic pain: a clinical perspective. In: *Proceedings of the 7th World Congress on Pain, Progress in Pain Research and Management*, Vol. 2, pp. 437–454. Eds. G. F. Gebhart, D. L. Hammond and T. S. Jensen. IASP Press, Seattle.

- Fields, H. L., Rowbotham, M. C. and Devor, M. (1997) Excitability blockers: anticonvulsants and low concentration local anesthetics in the treatment of chronic pain. In: *The Pharmacology of Pain. Handbook of Experimental Pharmacology*, Vol. 130, pp. 93–117. Eds. A. Dickenson and J-M. Besson. Springer-Verlag, Berlin.
- Fields, R. D. and Itoh-Kouichi (1996) Neural cell adhesion molecules in activity-dependent development and synaptic plasticity. *Trends Neurosci.* 19, 473–480.
- Figurov, A., Pozzo-Miller, L. D., Olafsson, P., Wang, T. and Lu, B. (1996) Regulation of synaptic responses to high frequency stimulation and LTP by neurotrophins in the hippocampus. *Nature* 381, 706–709.
- Finkbelner, S., Tavazoie, S. F., Maloratsky, A., Jacobs, K. M., Harris, K. M. and Greenberg, M. E. (1997) CREB: a major mediator of neuronal neurotrophin responses. *Neuron* 19, 1031– 1047.
- Fisher, K. and Coderre, T. J. (1996) Comparison of nociceptive effects produced by intrathecal administration of mGluR agonists. *NeuroReport* 7, 2743–2747.
- Fitzgerald, M. and Lynn, B. (1977) The sensitization of high threshold mechanoreceptors with myelinated axons by repeated heating. J. Physiol. 365, 549–563.
- Fitzgerald, M., Wall, P. D., Goedert, M. and Emson, P. C. (1985) Nerve growth factor counteracts the neurophysiological and neurochemical effects of chronic sciatic nerve section. *Brain Res.* 332, 131–141.
- Fitzgerald, M., Butcher, T. and Shortland, P. (1994) Developmental changes in the laminar termination of A fibre cutaneous sensory afferents in the rat spinal cord dorsal horn. *J. Comp. Neurol.* 348, 225–2133.
- Fleetwood-Walker, S. M., Mitchell, R., Hope, P. J., El Yassir, N., Molony, V. and Bladon, C. M. (1990) The involvement of neurokinin receptor subtypes in somatensory processing in the superficial dorsal horn of the cat. *Brain Res.* 519, 169–182.
 Fletcher, D. S., Widmer, W. R., Luell, S., Christen, A., Orevillo,
- Fletcher, D. S., Widmer, W. R., Luell, S., Christen, A., Orevillo, C., Shah, S. and Visco, D. (1998) Therapeutic administration of a selective inhibitor of nitric oxide synthase does not ameliorate the chronic inflammation and tissue damage associated with adjuvent-induced arthritis in rats. J. Pharmac. Exp. Ther. 284, 714–721.
- Fletcher, E. J. and Lodge, D. (1996) New developments in the molecular pharmacology of α-amino-3-hydroxy-5-methyl-4-isoxale propionate and kainate receptors. *Pharmac. Ther.* **70**, 65– 89.
- Florence, S. L., Garraghty, P. E., Carlston, M. and Kaas, J. H. (1993) Sprouting of peripheral nerve axons in the spinal cord of monkeys. *Brain Res.* 601, 343–348.
- Forti, M., Biella, G., Caccia, S. and de Curtis, M. (1997) Persistent excitability changes in the piriform cortex of the isolated guinea-pig brain after transient exposure to bicuculline. *Eur. J. Neurosci.* 9, 435–451.
- Fournier, A. E. and McKerracher, L. (1997) Expression of specific tubulin isotypes increase during regeneration of injured CNS neurons, but not after the application of brain derived neurotrophic factor (BDNF). J. Neurosci. 17, 4623–4632.
- Fournier, A. E., Beer, J., Arregui, C., Essagian, C., Aguayo, A. J. and McKerracher, L. (1997) Brain-derived neurotrophic factor modulates GAP-43 but not TNFα expression in injured retinal ganglion cells of adult rats. J. Neurosci. Res. 47, 561–572.
- Fozard, J. R. (1995) The 5-hydroxytryptamine nitric oxide connection: the key link in the initiation of migraine? Arch. Int. pharmacodyn. 329, 111–119.
- Fozard, J. R. and Kalkman, H. O. (1994) 5-Hydroxytryptamine (5-HT) and the initiation of migraine: new perspectives. *Naunyn-Schmiedeberg's Arch. Pharmac.* **350**, 225–229.
- Franklin, K. B. J. (1989) Analgesia and the neural substrate of reward. Neurosci. Biobehav. Rev. 35, 157–163.
- Fraser, H., Chapman, V. and Dickenson, A. H. (1992) Spinal local anaesthetic actions on afferent evoked responses and windup of nociceptive neurones in the rat spinal cord: combination with morphine produces marked potentiation of antinociception. *Pain* 49, 33-41.
- Fried, K., Govrin-Lippmann, R. and Devor, M. (1993) Close apposition among neighboring axonal endings in a neuroma. J. Neurocytol. 22, 663–681.
- Friedel, R. H., Schnürch, H., Stubbusch, J. and Barde, Y-A. (1997) Identification of genes differentially expressed by nerve growth factor- and neurotrophin-3-dependent sensory neurons. *Proc. natl Acad. Sci. U.S.A.* 94, 12670–12675.
- Frisén, J., Risling, M. and Fried, K. (1993) Distribution and axonal relations of macrophages in a neuroma. *Neuroscience* 55, 1003–1013.

- Frölich, J. C. (1997) A classification of NSAIDs according to the relative inhibition of cyclooxygenase isoenzymes. *Trends Pharmac.* 18, 30–34.
- Frost, S. A., Raja, S. N., Campbell, J. N., Meyer, R. A. and Khan, A. A. (1988) Does hyperalgesia to cooling stimuli characterize patients with sympathetically maintened pain (reflex sympathetic dystrophy)? In: *Proceedings of the Vth World Congress* on Pain, pp. 151–156. Eds. R. Dubner, G. F. Gebhart and M. R. Bond. Elsevier, Amsterdam.
- Fu, S. Y. and Gordon, T. (1997) The cellular and molecular basis of peripheral nerve regeneration. *Mol. Neurobiol.* 14, 67–116.
- Funakoshi, H., Frisen, J., Barnaby, G., Timmusk, T., Zachrisson, O., Verge, V. M. K. and Persson, H. (1993) Differential expression of mRNAs for neurotrophins and their receptors after axotomy of the sciatic nerve. J. Cell. Biol. 123, 455–465.
- Furukawa, S., Sugihara, Y., Iwasaki, F., Fukumitsu, H., Nitta, A., Nomoto, H. and Furukawa, Y. (1997) Brain-derived neurotrophic factor-like immunoreactivity in the adult rat central nervous system predominantly distributed in neurons with substantial amounts of brain-derived neurotrophic factor messenger RNA or responsiveness to brain-derived neurotrophic factor. *Neuroscience* 82, 653–670.
- Furuyama, T., Kiyama, H., Sato, K., Park, H. T., Maeno, H., Takagi, H. and Tohyama, M. (1993) Region-specific expression of subunits of ionotropic glutamate receptors (AMPA-type, KA-type and NMDA) in the rat spinal cord with special reference to nociception. *Mol. Brain Res.* 18, 141–151.
 Fuxe, K. and Agnati, L. F. (1991) Volume Transmission in the
- Fuxe, K. and Agnati, L. F. (1991) Volume Transmission in the Brain: Novel Mechanisms for Neural Transmission, p. 602. Raven Press, New York.
- Gadient, R. A. and Otten, U. H. (1997) Interleukin-6 (IL-6) a molecule with both beneficial and destructive potential. *Prog. Neurobiol.* 52, 379–390.
- Galer, B. S. (1995) Neuropathic pain of peripheral origin: advances in pharmacologic treatment. *Neurology* 45, S17–S25.
- Gallo, G., Lefort, F. B. and Letourneau, P. C. (1997) The TrkA receptor mediates growth cone turning toward a localized source of nerve growth factor. J. Neurophysiol. 17, 5445–5454.
- Gammon, C. M., Allen, A. C. and Morell, P. (1989) Bradykinin stimulates phosphoinositide hydrolysis and mobilisation of arachidonic acid in dorsal root ganglion neurons. J. Neurochem. 53, 95–101.
- Gamse, R. and Saria, A. (1987) Antidromic vasodilation in the rat hindpaw measured by laser Doppler flowmetry: pharmacological modulation. J. Auton. Nerv. Syst. 19, 105–111.
- Gao, K., Kim, Y-H. H. and Mason, P. (1997) Serotonergic pontomedullary neurons are not activated by antinociceptive stimulation in the periaqueductal gray. J. Neurosci. 17, 3285–3292.
- Gardano, N., Levi, A. and Alema, S. (1997) Modulation of nerve growth factor internalization by direct interaction between p75 and TrkA receptors. J. Neurosci. Res. 50, 1–12.
- Gardenas, C. G., Del Mar, L. P., Cooper, B. Y. and Scroggs, R. S. (1997) 5-HT₄ receptors couple positively to tetrodotoxin-insensitive sodium channels in a subpopulation of capsaicin-sensitive rat sensory neurons. *J. Neurosci.* 17, 7181-7189.
 Garland, A. M., Grady, E. F., Lovett, M., Vigna, S. R., Frucht,
- Garland, A. M., Grady, E. F., Lovett, M., Vigna, S. R., Frucht, M. M., Krause, J. E. and Bunnett, N. W. (1996) Mechanism of desensitization and resensitization of G protein-coupled neurokinin₁ and neurokinin₂ receptors. *Mol. Pharmac.* 49, 438–446.
- Garry, M. G., Abraham, E., Hargreaves, K. M. and Aanonsen, L. M. (1994a) Intrathecal injection of call-permeable analogs of cyclic 3',5'-guanosine monophosphate produces hyperalgesia in mice. *Eur. J. Pharmac.* 260, 129–131.
- Garry, M. G., Richardson, J. D. and Hargreaves, K. M. (1994b) Carrageenan-induced inflammation alters the content of icGMP and i-cAMP in the dorsal horn of the spinal cord. *Brain Res.* 646, 135–139.
- Garry, M. G., Richardson, J. D. and Hargreaves, K. M. (1994c) Sodium nitroprusside evokes the release of immunoreactive calcitonin gene-related peptide and substance P from dorsal horn slices via nitric oxide-dependent and nitric oxide-independent mechanisms. J. Neurosci. 14, 4329–4337.
- Garthwaite, J. and Boulton, C. L. (1995) Nitric oxide signaling in the central nervous system. A. Rev. Physiol. 57, 683–706.
- Gear, R. W. and Levine, J. D. (1995) Antinociception produced by an ascending spino-supraspinal pathway. J. Neurosci. 15, 3154–3161.
- Gebhart, G. F. (1995) Visceral Pain, p. 516. IASP Press, Seattle.
- Gee, N. S., Brown, J. P., Dissanayake, V. U. K., Offord, J., Thrulow, R. and Woodruff, G. N. (1996) The novel anticonvulsant drug, gabapentin (neurontin), binds to the $\alpha_2\delta$ subunit of a calcium channel. J. Biol. Chem. **271**, 5768.

- Gentry, C. T., West, K. J. and Camkpbell, E. A. (1997) The effects of sympathectomy on the nerve growth factor (NGF)—induced nociceptive responses in normal and neuropathic rats. *Br. J. Pharmac.* **120**, 231P.
- Geppetti, P., Del Bianco, E., Tramontana, M., Vigano, T., Folco, G. C., Maggi, C. A., Manzini, S. and Faciullacci, M. (1991) Arachidonic acid and bradykinin share a common pathway to release neuropeptides from capsaicin-sensitive sensory nerve fibers of the guinea pig heart. J. Pharmac. Exp. Ther. 259, 759– 765.
- Gerber, G., Kangrga, I., Ryu, P. D., Larew, J. S. A. and Randic, M. (1989) Multiple effects of phorbol esters in the rat spinal dorsal horn. J. Neurosci. 9, 3606–3617.
- Gereau, R. W. and Heinemann, S. F. (1998) Role of protein kinase C phosphorylation in rapid desensitization of metabotropic glutamate receptor 5. *Neuron* 20, 143–151.
- Ghosh, A. and Greenberg, M. E. (1995) Calcium signaling in neurons: molecular mechanisms and cellular consequences. *Science* 268, 239–247.
- Ghostine, S. Y., Comair, Y. G., Turner, D. M., Kassell, F. and Azar, C. G. (1984) Phenoxybenzamine in the treatment of causalgia. J. Neurosurg. 60, 1263–1268.
- Giesler, G. J. (1995a) Evidence of direct nociceptive projections from the spinal cord to the hypothalamus and telencephalon. *Neuroscience* **7**, 253–261.
- Giesler, G. J. (1995b) The spino-hypothalamic tract. In: Forebrain Areas Involved in Pain Processing, pp. 49–62. Eds. J.-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext. Paris
- G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris. Giesler, G. J., Katter, J. T. and Dado, R. J. (1994) Direct spinal pathways to the limbic system for nociceptive information. *Trends Neurosci.* **17**, 244–250.
- Gil, Z. and Amitai, Y. (1996a) Adult thalamocortical transmission involves both NMDA and non-NMDA receptors. J. Neurophysiol. 76, 2547–2554.
- Gil, Z. and Amitai, Y. (1996b) Properties of convergent thalamocortical and intracortical synaptic potentials in single neurons of neocortex. J. Neurosci. 16, 6567–6578.
- Gimeno, G., Carpentier, P. H., Desquand-Billiald, S., Hanf, R. and Finet, M. (1998) Histamine-induced biphasic macromolecular leakage in the microcirculation of the conscious Hamster: evidence for a delayed nitric oxide-dependent leakage. Br. J. Pharmac. 123, 943–951.
- Giordano, J. and Sacks, S. M. (1997) Sub-anesthetic doses of bupivacaine or lidocaine increase peripheral ICS-205,930induced analgesia against inflammatory pain in rats. *Eur. J. Pharmac.* 334, 39–41.
- Goadsby, P. J. and Hoskin, K. L. (1996) Inhibition of trigeminal neurons by intravenous administration of the serotonin (5-HT)1B/D receptor agonist zolmitriptan (311C90): are brain stem sites therapeutic targets in migraine? *Pain* 67, 355–359.
- Godefroy, F., Weil-Fugazza, J. and Besson, J. M. (1987) Complex temporal changes in 5-hydroxytryptamine synthesis in the central nervous system induced by experimental polyarthritis in the rat. *Pain* **28**, 223–238.
- Goff, J. R., Burkey, A. R., Goff, D. J. and Jasmin, L. (1998) reorganization of the spinal dorsal horn in models of chronic pain: correlation with behaviour. *Neuroscience* 82, 559–574.
 Gold, B. G. and Spencer, P. S. (1993) Neurotrophic function in
- Gold, B. G. and Spencer, P. S. (1993) Neurotrophic function in normal nerve and in peripheral neuropathies. In: *Neuroregeneration*, pp. 101–122. Ed. A. Gorio. Raven Press, New York.
- Gold, B. G., Mobley, W. C. and Mathesson, S. F. (1991) Regulation of axonal caliber, neurofilament content and nuclear localization in mature sensory neurons by nerve growth factor. *J. Neurosci.* 11, 943–955.
- Gold, B. G., Storm-Dickerson, T. and Austin, D. R. (1993) Regulation of transcription factor c-JUN by nerve growth factor in adult neurons. *Neurosci. Lett.* 154, 129–133.
- Gold, M. S., Reichling, D. B., Shuster, M. J. and Levine, J. D. (1996a) Hyperalgesic agents increase a tetrodotoxin-resistant Na⁺-current in nociceptors. *Proc. natl Acad. Sci. U.S.A.* 93, 1108–1112.
- Gold, M. S., Shuster, M. J. and Levine, J. D. (1996b) Characterization of six voltage-gated K^+ currents in adult rat sensory neurons. J. Neurophysiol. **75**, 2629–2646.
- Gold, M. S., Dastmalchi, S. and Levine, J. D. (1997) α₂-adrenergic receptor subtypes in rat dorsal root and superior cervical ganglion neurons. *Pain* 69, 179–190.
- Golding, J., Shewan, D. and Cohen, J. (1997) Maturation of the mammalian dorsal root entry zone—from entry to no entry. *Trends Neurosci.* 7, 303–308.
- Goodman, L. J., Valverde, J., Geschwind, M. D., Federoff, H. J., Geller, A. I. and Hefti, F. (1996) Regulated release and polar-

ized localization of brain-derived neurotrophic factor in hippocampal neurons. Mol. Cell. Neurosci. 7, 222-238.

- Goodness, T. P., Albers, K. M., Davis, F. E. and Davis, B. M. (1997) Overexpression of nerve growth factor in skin increases sensory neuron size and molulates Trk receptor expression. *Eur. J. Neurosci.* 9, 1574–1585.
- Gordon, B., Kinch, G., Kato, N., Keele, C., Lissman, T. and Fu, L. N. (1997a) Development of MK-801, kainate, AMPA and muscimol binding sites and the effect of dark rearing in rat visual cortex. J. Comp. Neurol. 383, 73–81.
- Gordon, S. M., Dionne, R. A., Brahim, J., Jabir, F. and Dubner, R. (1997b) Blockade of peripheral neuronal barrage reduces postoperative pain. *Pain* 70, 209–215.
- Gouardères, C., Kar, S. and Zajac, J. M. (1996) Presence of neuropeptide FF receptors on primary afferents of the rat spinal cord. *Neuroscience* 74, 21–27.
- Gouardères, C., Tafani, J. A. M., Mazarguil, H. and Zajac, J. M. (1997) Autoradiographic characterization of rat spinal neuropeptide FF receptors by using [¹²⁵I]-D-Tyr1 (Nme) Phe³ NPFF. *Brain Res. Bull.* 42, 231–238.
- Gracely, R. H., Lynch, S. A. and Bennett, G. J. (1992) Painful neuropathy: altered central processing maintained dynamically by peripheral input. *Pain* 51, 175–194.
- Graeff, F. G., Silveira, M. C. L., Nogueira, R. L., Audi, E. A. and de Oliviera, R. M. W. (1993) Role of the amygdala and periaqueductal gray in anxiety and panic. *Behav. Brain Res.* 58, 123–131.
- Grassi, F., Mileo, A. M., Monaco, L., Punturieri, A., Santoni, A. and Eusebi, F. (1994) TNF-α increases the frequency of spontaneous miniature synaptic currents in cultured rat hippocampal neurons. *Brain Res.* 659, 226–230.
- Graven-Nielsen, T., Arendt-Nielsen, L., Svensson, P. and Jensen, T. S. (1997) Stimulus-response functions in areas with experimentally induced referred muscle pain—a psychophysical study. *Brain Res.* 744, 121–128.
- Green, P. G., Jänig, W. and Levine, J. D. (1997) Negative feedback neuroendocrine control of inflammatory response in the rat is dependent on the sympathetic postganglionic neuron. J. Neurosci. 17, 3234–3238.
- Greene, L. A. and Kaplan, D. R. (1995) Early events in neurotrophin signalling via TRK and P⁷⁵ receptors. *Curr. Opin. Neurobiol.* 5, 579–587.
- Greenshaw, A. J. and Silverstone, P. H. (1997) The non-antiemetic uses of serotonin 5-HT₃ receptor antagonists. *Drugs* **53**, 20–39.
- Griffiths, C. D., Eldershaw, T. P. D., Geraghty, D. P., Hall, J. L. and Colquhoun, E. Q. (1996) Capsaicin-induced biphasic oxygen uptake in rat muscle: antagonism by capsazepine and ruthenium red provides further evidence for peripheral vanilloid receptor subtypes (VN₁/VN₂). *Life Sci.* **59**, 105–117.
- Grigg, P., Schaible, H. G. and Schmidt, R. F. (1986) Mechanical sensitivity of group III and IV afferents from posterior articular nerve in normal and inflamed cat knee. J. Neurophysiol. 55, 635–643.
- Grimes, M. L., Zhou, J., Beattie, E. C., Yuen, E. C., Hall, D. E., Valletta, J. S., Topp, K. S., LaVail, J. H., Bunnett, N. W. and Mobley, W. C. (1996) Endocytosis of activated TrkA: evidence that nerve growth factor induces formation of signaling endosomes. J. Neurosci. 16, 7950–7964.
- Gross, M. (1995) Current and future therapies for migraine. Opin. Invest. Drugs 4, 283–290.
- Grubb, B. D., McQueen, D. S., Iggo, A., Birrell, G. J. and Dutia, M. B. (1988) A study of 5-HT receptors associated with afferent nerves located in normal and inflamed rat ankle joints. *Agents Actions* 25, 216–218.
- Grubb, B. D., Riley, R. C., Hope, P. J., Pubols, L. and Duggan, A. W. (1996) The burst-like firing of spinal neurons in rats with peripheral inflammation is reduced by an antagonist of *N*methyl-D-aspartate. *Neuroscience* 74, 1077–1086.
- Grudt, T. J., Williams, J. T. and Travagli, R. A. (1995) Inhibition by 5-hydroxytryptamine and noradrenaline in substantia gelatinosa of guinea pig spinal trigeminal nucleus. J. Physiol. 485, 113–120.
- Gu, J. G. and MacDermott, A. B. (1997) Activation of ATP P_{2X} receptors elicits glutamate release from sensory neuron synapses. *Nature* **389**, 749–753.
- Gu, X. and Spitzer, N. C. (1995) Distinct aspects of neuronal differentiation encoded by frequency of spontaneous Ca²⁺ transients. *Nature* **375**, 784–787.
- Gu, Z. Z., Pan, Y.-C., Cui, J. K., Klebuc, M. J., Shenaq, S. and Liu, P. K. (1997) Gene expression and apoptosis in the spinal cord neurons after sciatic nerve injury. *Neurochem. Int.* **30**, 417– 426.

- Guerini, D. (1997) Calcineurin: not just a simple protein phosphatase. *Biochem. Biophys. Res. Commun* 235, 271–275.
- Guieu, R., Peragut, J. C., Roussel, P., Hassani, H., Sampieri, F. H., Bechis, G., Gola, R. and Rochat, H. (1996) Adenosine and neuropathic pain. *Pain* 68, 271–274.
- Guilbaud, G. (1988) Peripheral and central electrophysiological mechanisms of joint and muscle pain. In: *Proceedings of the Vth World Congress on pain*, pp. 201–215. Eds. R. Dubner, G. F. Gebhart and M. R. Bond. Elsevier, Amsterdam.
- Guilbaud, G. and Benoist, J. M. (1995) Thalamic and cortical processing in rat models of clinical pain. In: *Forebrain Areas Involved in Pain Processing*, pp. 79–92. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Guilbaud, G., Kayser, V., Benoist, J. M. and Gautron, M. (1986) Modifications in the responsiveness of rat ventrobasal thalamic neurons at different stages of carrageenin-produced inflammation. *Brain Res.* 385, 86–98.
- Guilbaud, G., Benoist, J. M., Eschalier, A., Gautron, M. and Kayser, V. (1989a) Evidence for peripheral serotonergic mechanisms in the early sensitization after carrageenin-induced inflammation: electrophysiological studies in the ventrobasal complex of the rat thalamus using a potent specific antagonist of peripheral 5-HT receptors. *Brain Res.* 502, 187–197.
- Guilbaud, G., Benoist, J. M., Eschalier, A., Kayser, V., Gautron, M. and Attal, N. (1989b) Evidence for central phenomena participating in the changes of responses of ventrobasal thalamic neurons in arthritic rats. *Brain Res.* 484, 383–388.
- Guilbaud, G., Benoist, J. M., Jazat, F. and Gautron, M. (1990) Neuronal responsiveness in the thalamic complex of rats with an experimental peripheral mononeuropathy. J. Neurophysiol. 64, 1537–1554.
- Guilbaud, G., Benoist, J. M., Levante, A., Gautron, M. and Willer, J. C. (1992) Primary somatosensory cortex in rats with pain-related behaviors due to a peripheral mononeuropathy after moderate ligation of one sciatic nerve: neuronal responsivity to somatic stimulation. *Exp. Brain Res.* 92, 227–245.
- Guilbaud, G., Bernard, J.-F. and Besson, J.-M. (1994) Brain areas involved in nociception and pain. In: *Textbook of Pain*, 3rd edn, pp. 113–128. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Gupta, P., Brown, D., Butler, P., Ellis, P., Grayson, K. L., Land, G. C., Macor, J. E., Robson, S. F., Wythes, M. J. and Shepperson, N. B. (1995) The in vivo pharmacological profile of a 5-HT₁ receptor agonist, CP-122,288, a selective inhibitor of neurogenic inflammation. Br. J. Pharmac. 116, 2385–2390.
- Guzman, M. G., Stühmer, W. and Soto, F. (1997) Molecular characterization and pharmacological properties of the human P_{2X3} purinoceptor. *Mol. Brain Res.* 47, 59–66.
- Gwag, B. J., Sessler, F. M., Watherhoue, B. D. and Springer, J. E. (1993) Regulation of nerve growth factor mRNA in the hippocampal formation: effects of N-methyl-D-aspartate receptor activation. *Exp. Neurol.* **121**, 160–171.
- Gybels, J. M. and Sweet, W. H. (1989) Neurosurgical treatment of persistent pain. Physiological and pathological mechanisms of human pain. In: *Pain and Headache*, Vol. 11, p. 442. Karger, Basel.
- Gybels, J. M., Handwerker, H. O. and Van Hees, J. (1979) A comparison between the discharges of human nociceptive nerve fibers and the subjects ratings of his sensations. J. Physiol. 292, 193–206.
- Häbler, H. J., Jänig, W. and Koltzenburg, M. (1988) Dichotomizing unmyelinated afferents supplying pelvic viscera and perineum are rare in the sacral segments of the cat. *Neurosci. Lett.* 94, 119–124.
- Hachisuka, K., Lais, A. C. and Dyck, P. J. (1989) Ultrastructural alterations of primary axons in the nucleus gracilis after peripheral nerve axotomy. J. Neuropathol. Exp. Neurol. 48, 413–424.
- Haley, J. E., Sullivan, A. F. and Dickenson, A. H. (1990) Evidence for spinal N-methyl-D-aspartate receptor involvement in prolonged chemical nociception in the rat. *Brain Res.* 518, 218–226.
- Haley, J. E., Dickenson, A. H. and Schachter, M. (1992) Electrophysiological evidence for a role of nitric oxide in prolonged chemical nociception in the rat. *Neuropharmacology* 31, 251–258.
- Hall, J. M. (1997) Bradykinin receptors. Gen. Pharmac. 28, 1-6.
- Hama, A. T., Unnerstall, J. R., Siegan, J. B. and Sagen, J. (1995) Modulation of NMDA receptor expression in the rat spinal cord by peripheral nerve injury and adrenal medullary grafting. *Brain Res.* 687, 103–115.
- Hamel, E., Fan, E., Linville, D., Ting, V., Villemure, J. G. and Chia, L. S. (1993) Expression of mRNA for the serotonin 5-

hydroxytryptamine_{1D β} receptor subtype in human and bovine cerebral arteries. *Mol. Pharmac.* **44**, 242–246.

- Hammond, D. L. (1997) Inhibitory neurotransmitters and nociception: role of GABA and glycine. In: *The Pharmacology of Pain. Handbook of Experimental Pharmacology*, Vol. 130, pp. 361– 384. Eds. A. Dickenson and J.-M. Besson. Springer-Verlag, Berlin.
- Hammond, D. L. and Gregory, S. A. (1996) Intrathecally administered cyclo-oxygenase-2 inhibitors attenuate carrageenaninduced thermal hyperalgesia in the rat. *Am. Soc. Neurosci.* 26, 712–717.
- Han, H. C., Na, H. S., Yoon, Y. W. and Chung, J. M. (1994) Ectopic discharges from injured afferent fibers in a rat model of neuropathic pain. Am. Soc. Neurosci. 20, 760.
- Handwerker, H. O. (1976) Influences of algogenic substances and prostaglandins on the discharges of unmyelinated cutaneous nerve fibers identified as nociceptors. In: Advances in Pain Research and Therapy, Vol. 1, pp. 41–45, Eds. J. J. Bonica and D. Albe-Fessard. Raven Press, New York.
- Handwerker, H. O. (1991) What peripheral mechanisms contribute to nociceptive transmission and hyperalgesia. In: *Towards a New Pharmacotherapy of Pain*, pp. 5–20. Eds. A. I. Basbaum and J.-M Besson. Wiley Press, New York.
- Handwerker, H. O. and Kobal, G. (1993) Psychophysiology of experimentally induced pain. *Physiol. Rev.* 73, 639–671.
- Handwerker, H. O., Kilo, S. and Reeh, P. W. (1991) Unresponsive afferent nerve fibres in the sural nerve of the rat. J. Physiol. (Lond.) 435, 229–242.
 Hantzopoulos, P., Suri, C., Glass, D. J., Goldfarb, M. P. and
- Hantzopoulos, P., Suri, C., Glass, D. J., Goldfarb, M. P. and Yan, D. (1994) The low affinity NGF receptor, p75, can collaborate with each of the Trks to potentiate functional responses to the neurotrophins. *Neuron* 13, 187–201.
- Hao, J. X., Xu, X. J., Aldskogius, H., Seiger, Å. and Wiesenfeld-Hallin, Z. (1992) Baclofen reverses the hypersensitivity of dorsal horn wide dynamic range neurons to mechanical stimulation after transient spinal cord ischaemia—implications for a tonic GABAergic control of myelinated fiber input. J. Neurophysiol. 68, 392–396.
- Hao, J. X., Yu, W., Xu, X. J. and Wiesenfeld-Hallin, Z. (1996a) Capsaicin-sensitive afferents mediate chronic cold, but not mechanical, allodynia-like behavior in spinally injured rats. *Brain Res.* **722**, 177–180.
- Hao, J. X., Yu, W., Xu, X.-J. and Wiesenfeld-Hallin, Z. (1996b) Effects of intrathecal vs systemic clonidine in treating chronic allodynia-like response in spinally injured rats. *Brain Res.* 736, 28–34.
- Hara, N., Minami, T., Okuda-Ashitaka, E., Sugimoto, T., Sakai, M., Onaka, M., Mori, H., Imanishi, T., Shingu, K. and Ito, S. (1997) Characterization of nociceptin hyperalgesia and allodynia in conscious mice. *Br. J. Pharmac.* **121**, 401–408.
- Harden, R. N., Duc, T. A., Williams, T. R., Coley, D., Cate, J. C. and Gracely, R. H. (1994) Norepinephrine and epinephrine levels in affected versus unaffected limbs in sympatheticallymaintained pain. J. Clin. Pain 19, 324–330.
- Hardingham, G. E., Chawla, S., Johnson, C. M. and Bading, H. (1997) Distinct functions of nuclear and cytoplasmic calcium in the control of gene expression. *Nature* 385, 260–265.
 Hardy, J. D., Wolff, H. G. and Goodell, H. (1950) Experimental
- Hardy, J. D., Wolff, H. G. and Goodell, H. (1950) Experimental evidence on the nature of cutaneous hyperalgesia. J. Clin. Invest. 29, 115–140.
- Hargett, G. L., Coggeshall, R. E. and Carlton, S. M. (1997) Upregulation of AMPA receptors in laminae III–V following dorsal rhizotomy: evidence for glutamate supersensitivity. Am. Soc. Neurosci. 23, 1015.
- Harris, J. A., Corsi, M., Quartaroli, M., Arban, R. and Bentivoglio, M. (1996) Upregulation of spinal glutamate receptors in chronic pain. *Neuroscience* 74, 7–12.
- Hay, C. and Belleroche, J. (1997) Carrageenan-induced hyperalgesia is associated with increased cyclo-oxygenase-2 expression in spinal cord. *NeuroReport* 8, 1249–1251.
- Hayashi, Y. and Maze, M. (1993) Alpha₂ adrenoceptor agonists and anaesthesia. *Br. J. Anaesthesia.* **71**, 108–118.
- Hayes, E. S. and Carlton, S. M. (1992) Primary afferent interactions: analysis of calcitonin gene-related peptide-immunoreactive terminals in contact with unlabeled and GABAimmunoreactive profiles in the monkey dorsal horn. *Neuroscience* 47, 873–896.
- Hayes, R. L., Mao, J., Price, D. D., Germano, A., D'Avella, D., Fiori, M. and Mayer, D. J. (1992) Pretreatment with gangliosides reduces abnormal nociceptive responses associated with a rodent mononeuropathy. *Pain* 48, 391–396.
 He, X., Schepelmann, K., Schaible, H. G. and Schmidt, R. F.
- He, X., Schepelmann, K., Schaible, H. G. and Schmidt, R. F. (1990) Capsaicin inhibits responses of fine afferents from the

knee joint of the cat to mechanical and chemical stimuli. *Brain Res.* **530**, 147–150.

- Heapy, C. G., Shaw, J. S. and Farmer, S. C. (1993) Differential sensitivity of antinociceptive assays to the bradykinin antagonist Hoe 140. Br. J. Pharmac. 108, 209–213.
- Heese, K., Fiebich, B. L., Bauer, J. and Otten, U. (1997) Nerve growth factor (NGF) expression in rat microglia is induced by adenosine A_{2a}-receptors. *Neuroscience* **231**, 83–86.
- Hegarty, J. L., Kay, A. R. and Green, S. H. (1997) Trophic support of cultured spinal ganglion neurons by depolarization exceeds and is additive with that by neurotrophins or cAMP and requires elevation of [Ca²⁺]; within a set range. J. Neurosci. 17, 1959–1970.
- Heinricher, M. M., Morgan, M., Tortorici, V. and Fields, H. L. (1994) Disinhibition of off-cells and antinociception produced by an opioid action within the rostral ventromedial medulla. *Neuroscience* 63, 279–288.
- Heinricher, M. M., McGaraughty, S. and Grandy, D. K. (1997) Circuitry underlying antiopioid actions of orphanin FG in the rostral ventromedial medulla. J. Neurophysiol. 78, 3351–3358.
- Helgren, M. E., Cliffer, K. D., Torrento, K., Cavnor, C., Curtis, R., DiStefano, P. S., Wiegand, S. J. and Lindsay, R. M. (1997) Neurotrophin-3 administration attenuates deficits of pyridoxineinduced large-fiber sensory neuropathy. J. Neurosci. 17, 372– 382.
- Helke, C. J., Capuano, S., Tran, N. and Zhuo, H. (1997) Immunocytochemical studies of the 5-HT_{1A} receptor in ventral medullary neurons that project to the intermediolateral cell column and contain serotonin or tyrosine hydroxylase immunoreactivity. J. Comp. Neurol. **379**, 261–270.
- Henrion, D., Dechaux, E., Dowell, F. J., Maclour, J., Samuel, J.-L., Lévy, B. I. and Michel, J.-B. (1997) Alteration of flowinduced dilation in mesenteric resistance arteries of L-NAME treated rats and its partial association with induction of cyclooxygenase-2. Br. J. Pharmac. 121, 83–90.Hens, J. J. H., Benfenati, F., Nielander, H. B., Valtorta, F.,
- Hens, J. J. H., Benfenati, F., Nielander, H. B., Valtorta, F., Gispen, W. H. and De Graan, P. N. E. (1993) B-50:GAP-43 binds to actin filaments without affecting actin polymerization and filament organization. *J. Neurochem.* 61, 1530–1533.
- Heppelmann, B. and Matthias, P. (1997) Inhibitory effect of somatostatin on the mechanosensitivity of articular afferents in normal and inflamed knee joints of the rat. *Pain* **73**, 377–382.
- Heppelmann, B. and Pawlak, M. (1997) Sensitisation of articular afferents in normal and inflamed knee joints by substance P in the rat. *Neurosci. Lett.* 223, 97–100.
- Heppenstall, P. A. and Fleetwood-Walker, S. M. (1997a) The glycine site of the NMDA receptor contributes to neurokinin₁ receptor agonist facilitation of NMDA receptor agonist-evoked activity in rat dorsal horn neurons. *Brain Res.* **744**, 235–245.
- Heppenstall, P. A. and Fleetwood-Walker, S. M. (1997b) Glycine receptor regulation of neurokinin₁ receptor function in rat dorsal horn neurones. *NeuroReport* 8, 3109–3112.
- Heppenstall, P. A., Fleetwood-Walker, S. M., Thompson, S. W. N. and Spiers, H. J. (1997) LIF mRNA distribution in lumbar spinal cord and its modification by peripheral inflammaton and nerve injury. J. Physiol. 505P, 42P.
- Herdegen, T., Kovary, K., Leah, J. and Bravo, R. (1991) Specific temporal and spatial distribution of JUN, FOS and KROX-24 proteins in spinal neurons following noxious trans-synaptic stimulation. J. Comp. Neurol. 313, 178–191.
- Herdegen, T., Fiallos-Estrada, C. E., Bravo, R. and Zimmermann, M. (1993) Colocalisation and covariation of c-JUN transcription factor with galanin in primary afferent neurons and with CGRP in spinal motoneurons following transection of rat sciatic nerve. *Mol. Brain Res.* 17, 147–154.
- Herdegen, T., Skene, P. and Bähr, M. (1997) The c-jun transcription factor—bipotential mediator of neuronal death, survival and regeneration. *Trends Neurosci.* 20, 227–231.
- Herrero, J. F. and Cervero, F. (1996) Supraspinal influences on the facilitation of rat nociceptive reflexes induced by carrageeenan monoarthritis. *Neurosci. Lett.* 209, 21–24.
- Herrero, J. F., de la Rubia, P. G. and Cervero, F. (1995) Wind-up of spinal cord neurones is influenced by inflammation, spinalization, endogenous opioids and type of stimulation. *Am. Soc. Neurosci.* 21, 1407.
- Herrero, J. F., Parrado, A. and Cervero, F. (1997) Central and peripheral actions of the NSAID ketoprofen on spinal cord nociceptive reflexes. *Neuropharmacology* 36, 1425–1431.
- Herrero, L., Miras-Portugal, M. T. and Sanchez-Prieto, J. (1992) Positive feedback of glutamate exocytosis by presynaptic receptor stimulation. *Nature* 360, 163–166.
- Herzberg, U., Eliav, E., Dorsey, J. M., Gracely, R. H. and Kopin, I. J. (1997) NGF involvement in pain induced by chronic con-

striction injury of the rat sciatic nerve. NeuroReport 8, 1613-1618.

- Heumann, R. (1994) Neurotrophin signalling. Curr. Opin. Neurobiol. 4, 668–679.
- Heumann, R., Korsching, S., Bandtlow, C. and Thoenen, H. (1987a) Changes of nerve growth factor synthesis in non-neuronal cells in response to sciatic nerve transection. J. Cell. Biol. 104, 1623–1631.
- Heumann, R., Lindholm, D., Bandtlow, C., Meyer, M., Radeke, M. J., Misiko, T. P., Schooter, E. and Thoenen, H. (1987b) Differential regulation of mRNA encoding nerve growth factor and its receptor in rat sciatic nerve during development, degeneration, and regeneration: role of macrophages. *Proc. natl* Acad. Sci. U.S.A. 84, 8735–8739.
- Heyser, C. J., Masliah, E., Samini, A., Campbell, I. L. and Gold, L. H. (1997) Progressive decline in avoidance learning paralleled by inflammatory neurodegeneration in transgenic mice expressing interleukin 6 in the brain. *Proc. natl Acad. Sci. U.S.A.* 94, 1500–1505.
- Hicks, T. P., Metherate, R., Landry, B. and Dykes, R. W. (1986) Bicuculline-induced alterations of response properties in functionally identified ventroposterior thalamic neurones. *Exp. Brain. Res.* 63, 248–264.
- Higashida, H., Streaty, R. A., Klee, W. and Nirenberg, M. (1986) Bradykinin-activated transmembrane signals are coupled via NG or Gi to production of inositol 1, 4, 5-triphosphate, a second messenger in NG 108-15 neuroblastoma-glioma hybrid cells. *Proc. natl Acad. Sci. U.S.A.* 83, 942–946.
- Hikawa, N. and Takenaka, T. (1996) Myelin-stimulated macrophages release neurotrophic factors for adult dorsal root ganglion neurons in culture. *Cell. Mol. Neurobiol.* 16, 517–528.
- Hill, S. J., Ganellin, C. R., Timmerman, H., Schwartz, J. C., Shankley, N. P., Young, J. M., Schunack, W., Levi, R. and Haas, H. L. (1997) International union of pharmacology. XIII. Classification of histamine receptors. *Pharmac. Rev.* 49, 253– 278.
- Hingtgen, C. M. and Vasko, M. R. (1994) Prostacyclin enhances the evoked-release of substance P and calcitonin gene-related peptide from rat sensory neurons. *Brain Res.* 655, 51–60.
- Hingtgen, C. M., Waite, K. J. and Vasko, M. R. (1995) Prostaglandins facilitate peptide release fron rat sensory neurons by activating the adenosine 3',5'-cyclic monophosphate tranduction cascade. J. Neurosci. 15, 5411–5419.
- Hirshberg, R. M., Al-Chaer, E. D., Lawand, N. B., Westlund, K. N. and Willis, W. D. (1996) Is there a pathway in the posterior funiculus that signals visceral pain? *Pain* 67, 291–305.
- Ho, J., Mannes, A. J., Dubner, R. and Candle, R. M. (1997) Putative kappa-2 opioid agonists are antihyperalgesic in a rat model of inflammation. J. Pharmac. Exp. Ther. 281, 136–141.
- Höglund, A. U. and Baghdoyan, H. A. (1997) M₂, M₃, M₄, but not M₁, muscarinic receptor subtypes are present in rat spinal cord. J. Pharmac. Exp. Ther. 281, 470–477.
- Hoheisel, U. and Mense, S. (1989) Long term changes in discharge behaviour of cat dorsal horn neurons following noxious stimulation of deep tissues. *Pain* 36, 239–247.
- Hoheisel, U. and Mense, S. (1990) Response behaviour of cat dorsal horn neurones receiving input from skeletal muscle and other deep somatic tissues. J. Physiol. 426, 265–280.
- Hoheisel, U., Sander, B. and Mense, S. (1995) Blockade of nitric oxide synthase differentially influences background activity and electrical excitability in rat dorsal horn neurones. *Neurosci. Lett.* 188, 143–146.
- Hoheisel, U., Sander, B. and Mense, S. (1997) Myositis-induced functional reorganisation of the rat dorsal horn: effects of spinal superfusion with antagonists to neurokinin and glutamate receptors. *Pain* 69, 219–230.
- Hökfelt, T., Zhang, X. and Wiesenfeld-Hallin, Z. (1994) Messenger plasticity in primary sensory neurons following axotomy and its functional implications. *Trends Neurosci.* 17, 22–30.
- Hollmann, M. and Heinemann, S. (1994) Cloned glutamate receptors. A. Rev. Neurosci. 17, 31–108.
- Holm, N. R., Christophersen, P., Olesen, S. P. and Gammeltoft, S. (1997) Activation of calcium-dependent potassium channels in rat brain neurons by neurotrophin-3 and nerve growth factor. *Proc. natl Acad. Sci. U.S.A.* 94, 1002–1006.
- Hölscher, C. (1997) Nitric oxide, the enigmatic neuronal messenger: its role in synaptic plasticity. *Trends Neurosci.* 20, 298–303. Holstege, J. C., Van Dijken, H., Buijs, R. M., Goedknegt, H.,
- Holstege, J. C., Van Dyken, H., Buys, R. M., Goedknegt, H., Gosens, T. and Bongers, C. M. H. (1996) Distribution of dopamine immunoreactivity in the rat, cat, and monkey spinal cord. *J. Comp. Neurol.* **376**, 631–652.

134

- Holthusen, H. (1997) Involvement of the NO/cyclic GMP pathway in bradykinin-evoked pain from veins in humans. Pain 69, 87-
- Holthusen, H. and Arndt, J. O. (1994) Nitric oxide evokes pain in humans on intracutaneous injection. Neurosci. Lett. 165, 71-74
- Holthusen, H. and Ding, Z. (1997) Nitric oxide is not involved in vascular nociception of noxious physical stimuli in humans. Neurosci. Lett. 227, 111–114.
- Holthusen, H., Kindgen-Milles, D. and Ding, Z. P. (1997) Substance P is not involved in vascular nociception in humans. Neuropeptides 31, 445-448.
- Holzer, P. (1988) Local effector functions of capsaicin-sensitive sensory nerve endings: involvement of tachykinins, calcitonin gene-related peptide and other neuropeptides. Neuroscience 24, 739–768.
- Hong, Y. and Abbott, F. V. (1996) Contribution of peripheral $\alpha_{1A}\text{-}adrenoreceptors$ to pain induced by formalin or by α methyl-5-hydroxytryptamine plus noradrenaline. Eur. J Pharmac. 301, 41-48.
- Horie, H., Iwasaki, T., Kurahashi, Y. and Nishino, T. (1997a) Calcium-dependent inactivation of neuronal nitric oxide synthase: evidence for the existence of stabilization/activation
- factor. Biochem. Biophys. Res. Commun. 234, 476–480. Horie, H., Sakai, I., Akahori, Y. and Kadoya, T. (1997b) IL-1β enhances neurite regereration from transected-nerve terminals of adult rat DRG. NeuroReport 8, 1955-1959.
- Horigome, K., Pryor, J. C., Bullock, E. D. and Johnson, E. M. (1993) Mediator release from mast cells by nerve growth factor: neurotrophin specificity and receptor mediation. J. Biol. Chem. 268, 14881-14887.
- Hoskin, S. K. L., Kaube, H. and Goadsby, P. J. (1996) Central activation of the trigeminovascular pathway in the cat is inhibited by dihydroergotamine: a c-fos and electrophysiology study. Brain 119, 249-256.
- Hou, W. Y., Shyu, B. C., Chen, T. M., Lee, J. W., Shieh, J. Y. and Sun, W. Z. (1997) Intrathecally administered c-fos antisense oligodeoxynucleotide decreases formalin-induced nociceptive
- behavior in adult rats. *Eur. J. Pharmac.* **329**, 17–26. Houghton, A. K., Wang, C. C., Willis, W. D. and Westlund, K. N. (1997) The effect of dorsal column (DC) lesions upon the central actions of pancreatic afferent fibres in the rat. J. Physiol. 505P, 39P.
- Hsieh, J. C., Belfrage, M., Stone-Elander, S., Hansson, P. and Ingvar, M. (1995) Central representation of chronic ongoing neuropathic pain studied by positron emission tomography. *Pain* **63**, 225–236.
- Hsu, M.-M. and Shyu, B-C. (1997) Electrophysiological study of the connection between medial thalamus and anterior cingulate cortex in the rat. NeuroReport 8, 2701-2707.
- Hu, H. Z. and Li, Z. W. (1996) Substance P potentiates ATP-activated currents in rat primary sensory neurons. Brain Res. 739, 163 - 168
- Hu, H. Z. and Li, Z. W. (1997) Modulation by adenosine of GABA-activated current in rat dorsal root ganglion neurons. J. Physiol. 501, 67-75.
- Hu, H. Z., Li, Z. W. and Si, J. Q. (1997) Evidence for the existence of substance P autoreceptor in the membrane of rat dorsal root ganglion neurons. *Neuroscience* **77**, 535–541.
- Hu, S. and Zhu, J. (1989) Sympathetic facilitation of sustained discharges of polymodal nociceptors. Pain 38, 85-90.
- Hua, X. Y., Chen, P., Fox, A. and Myers, R. R. (1996) Involvement of cytokines in lipopolysaccaride-induced facilitation of CGRP release from capsaicin sensitive nerves in the trachea: studies with $IL-1\beta$ and $TNF-\alpha$. J. Neurophysiol. 16, 4742-4748.
- Hua, X-Y., Calcutt, N. A. and Malmberg, A. B. (1997) Neonatal capsaicin treatment abolishes formalin-induced spinal PGE2 release. NeuroReport 8, 2325–2329. Huber, K. M., Sawtell, N. B. and Bear, M. F. (1998) Effects of
- the metabotropic glutamate receptor antagonist MCPG on phosphoinositide turnover and synaptic plasticity in visual cortex. J. Neurosci. 18, 1-9.
- Hubscher, C. H. and Berkley, K. J. (1995) Spinal and vagal influences on the responses of rat solitary nucleus neurons to stimulation of uterus, cervix and vagina. Brain Res. 702, 251-254.
- Huerta, P. T. and Lisman, J. E. (1995) Bidirectional synaptic plasticity induced by a single burst during cholinergic theta oscillation in CA1 in vitro. Neuron 15, 1053-1063.
- Huettner, J. E. (1990) Glutamate receptor channels in rat DRG neurons: activation by kainate and quisqualate and blockade of desensitization by Con A. Neuron 5, 255-266. Hughes, P. and Dragunow, M. (1995) Induction of immediate-
- early genes and the control of neurotransmitter-regulated gene

expression within the nervous system. Pharmac. Rev. 47, 133-178.

- Hughes, P., Beilharz, E., Gluckman, P. and Dragunow, M. (1993) Brain-derived neurotrophic factor is induced as an immediate early gene following *N*-methyl-D-aspartate receptor activation. *Neuroscience* **57**, 319–328.
- Hunt, S. P. and Rossi, J. (1985) Peptide- and non-peptide-containing unmyelinated primary afferents: the parallel processing of nociceptive information. Phil. Trans. Roy. Soc. Lond. B 308, 283 - 289
- Hunt, S. P., Pini, A. and Evan, A. (1987) Induction of c-fos like protein in spinal cord neurons following sensory stimulation. Nature 328, 632-634.
- Hunter, J. C. and Singh, L. (1994) Role of excitatory amino acid receptors in the mediation of the nociceptive response to formalin in the rat. Neurosci. Lett. 174, 217-221.
- Hunter, J. C., Woodburn, V. L., Durieux, C., Pettersson, E. K. E., Poat, J. A. and Hughes, J. (1995) C-fos antisense oligodeoxynucleotide increases formalin-induced nociception and regulates preprodynorphin expression. Neuroscience 65, 485-492.
- Hunter, J. C., Lewis, R., Eglen, R. M. and Fontana, D. J. (1996) The role of α and β adrenoceptors in a rodent model of neuropathic pain. Br. J. Pharmac. 117, 237P.
- Hunter, J. C., Fontana, D. J., Hedley, L. R., Jasper, J. R., Lewis, R., Link, R. E., Secchi, R., Sutton, J. and Eglen, R. M. (1997a) Assessment of the role of α_2 -adrenoceptor subtypes in the antinociceptive, sedative and hypothermic action of dexmedetomi-
- dine in transgenic mice. Br. J. Pharmac. 122, 1339-1344. Hunter, J. C., Gogas, K. R., Hedley, L. R., Jacobson, L. O., Kassotakis, L., Thompson, J. and Fontana, D. J. (1997b) The effect of novel anti-epileptic drugs in rat experimental models of acute and chronic pain. Eur. J. Pharmac. 324, 153-160.
- Huskisson, E. C., Berry, H., Gishen, P., Jubb, R. W. and Whitehead, J. (1995) Effects of anti-inflammatory drugs on the progression of osteoarthritis of the knee. J. Rheumatol. 22, 1941-1946.
- Hwang, J. H. and Yaksh, T. L. (1997) The effect of spinal GABA receptor agonists on tactile allodynia in a surgically-induced neuropathic pain model in the rat. Pain 70, 15-22.
- Hylden, J. L. K., Nahin, R. L., Traub, R. J. and Dubner, R. (1989) Expansion of receptive fields of spinal lamina I projection neurons in rats with unilateral adjuvant-induced inflammation: the contribution of dorsal horn mechanisms. Pain 37, 239-243
- Hylden, J. L. K., Nahin, R. L., Traub, R. J. and Dubner, R. (1991) Effects of kappa-opioid receptor agonists on the responsiveness of nociceptive superficial dorsal horn neurones. Pain 44, 187-193.
- Iadecola, C., Zhang, F., Casey, R., Clark, H. B. and Ross, M. E. (1996) Inducible nitric oxide synthase gene expression in vascular cells following transient focal cerebral ischemia. Stroke 27, 1373-1380.
- Ibuki, T., Hama, A. T., Wang, X. T., Pappas, G. D. and Sagens, J. (1997) Loss of gaba-immunoreactivity in the spinal dorsal horn of rats with peripheral nerve injury and promotion of
- recovery by adrenal medullary grafts. *Neuroscience* **76**, 845–858. Ichitani, Y., Shi, T., Haeggstrom, J. Z., Samuelsson, B. and Hökfelt, T. (1997) Increased levels of cyclooxygenase-2 mRNA in the rat spinal cord after peripheral inflammation: an in situ hybridization study. NeuroReport 8, 2949-2952.
- Idänpään-Heikkilä, J. J., Perrot, S., Guilbaud, G. and Kayser, V. (1997) In mononeuropathic rats, the enhancement of morphine antinociception by L-365,260, a selective CCK_B receptor antagonist, depends on the dose of systemic morphine and stimulus characteristics. Eur. J. Pharmac. 325, 155-164.
- Ide, Y., Maehara, Y., Tsukahara, S., Kitahata, L. M. and Collins, J. G. (1997) The effects of an intrathecal NMDA antagonist (AP5) on the behavioral changes induced by colorectal inflammation with turpentine in rats. Life Sci. 60, 1359-1363.
- Igwe, O. J. and Ning, L. (1994) Regulation of the second-messenger systems in the rat spinal cord during prolonged peripheral inflammation. Pain 58, 63-75.
- Iljima, K., Sato, M., Kojima, N. and Ohtomo, K. (1992) Immunocytochemical and in situ hybridization evidence for the coexistence of Gaba and tyrosine hydroxylase in the rat locus ceruleus. Anat. Record 234, 593-604.
- Inoue, A., Hashimoto, T., Hide, I., Nishio, H. and Nakata, Y (1997) 5-Hydroxytryptamine-facilitated release of substance P from rat spinal cord slices is mediated by nitric oxide and cyclic GMP. J. Neurochem. 68, 128–133.
- Inoue, K., Nakazawa, K., Inoue, K. and Fujimori, K. (1995) Nonselective cation channels coupled with tachykinin receptors in rat sensory neurons. J. Neurophysiol. 73, 736-742.

- Isaac, J. T., Crair, M. C., Nicoll, R. A. and Malenka, R. C. (1997) Silent synapses during development of thalamocortical inputs. *Neuron* 18, 269–280.
- Itazawa, S. I., Isa, T. and Ozawa, S. (1997) Inwardly rectifying and Ca²⁺-permeable AMPA-type glutamate receptor channels in rat neocortical neurons. *J. Neurophysiol.* **78**, 2592–2601.
- Jackson, D. L., Graff, C. B., Richardson, J. D. and Hargreaves, K. M. (1995) Glutamate participates in the peripheral modulation of thermal hyperalgesia in rats. *Eur. J. Pharmac.* 284, 321–325.
- Jahr, C. E. and Jessel, T. M. (1983) ATP excites a sub-population of dorsal horn neurons. *Nature* **304**, 730–733.
 Jalonen, T. O., Margraf, R. R., Wielt, D. B., Charniga, C. J.,
- Jalonen, T. O., Margraf, R. R., Wielt, D. B., Charniga, C. J., Linne, M.-L. and Kimelberg, H. K. (1997) Serotonin induces inward potassium and calcium currents in rat cortical astrocytes. *Brain Res.* **758**, 69–82.
- Jänig, W. (1993) Spinal visceral afferents, sympathetic nervous system and referred pain. In: New Trends in Referred Pain and Hyperalgesia, pp. 83–98, Eds. L. Vecchiet, D. Abbe-Fessard, U. Lindblom and M. A. Giamberardino. Elsevier, Amsterdam.
- Janscó, G. and Lawson, S. N. (1990) Transganglionic degeneration of capsaicin-sensitive C-fiber primary afferent terminals. *Neuroscience* 39, 501–511.
- Janssen-Olesen, I., Ottosson, A., Cantera, L., Strunk, S., Lassen, L. H., Olesen, J., Mortensen, A., Engel, U. and Edvinsson, L. (1997) Role of endothelium and nitric oxide in histamineinduced responses in human cranial arteries and detection of mRNA encoding H₁- and H₂-receptors by RT-PCR. Br. J. Pharmac. 121, 41–48.
- Jarvis, C. R., Xiong, Z. G., Plant, J. R., Churchill, D., Lu, W.-Y., MacVicar, B. A. and MacDonald, J. F. (1997) Neurotrophin modulation of NMDA receptors in cultured murine and isolated rat neurons. J. Neurophysiol. 78, 2363–2371.
- Jasmin, L., Burkey, A. R., Card, J. P. and Basbaum, A. I. (1997) Transneuronal labelling of a nociceptive pathway, the spino-(trigemino-)parabrachio-amygdaloid, in the rat. J. Neurosci. 17, 3751–3765.
- Jeftinija, S., Jeftinija, K., Liu, F., Skilling, S. R., Smullin, D. H. and Larson, A. A. (1991) Excitatory amino acids are released from rat primary afferent neurons in vitro. *Neurosci. Lett.* **125**, 191.
- Jenkins, R., McMahon, S. B., Bond, A. B. and Hunt, S. P. (1993) Expression of c-jun as a response to dorsal root and peripheral nerve section in damaged and adjacent intact primary sensory neurons in the rat. *Eur. J. Neurosci.* 5, 751–759.
- neurons in the rat. Eur. J. Neurosci. 5, 751–759. Jensen, T. S. and Yaksh, T. L. (1992) Brainstem excitatory amino acid receptors in nociception: microinjection mapping and pharmacological characterization of glutamate-sensitive sites in the brainstem associated with algogenic behavior. Neuroscience 46, 535–547.
- Ji, R. R., Zhang, X. and Wiesenfeld Hallin, Z. (1994) Expression of neuropeptide Y and neuropeptide Y (Y₁) receptor mRNA in rat spinal cord and dorsal root ganglia following peripheral tissue inflammation. J. Neurosci. 14, 6423–6434.
- Ji, R. R., Zhang, X., Zhang, Q., Dagerlind, Å., Nilsson, S., Wiesenfeld Hallin, Z. and Hökfelt, T. (1995) Central and peripheral expression of galanin in response to inflammation. *Neuroscience* 68, 563–576.
- Jia, Y.-P. and Seybold, V. S. (1997) Spinal NK₂ receptors contribute to the increased excitability of the nociceptive flexor reflex during persistent peripheral inflammation. *Brain Res.* **751**, 169– 174
- Jiang, H., St. Ulme, D., Dickens, G., Chabuk, A., Lavarreda, M., Lazarovici, P. and Guroff, G. (1997) Both p140^{trk} and p75^{NGFR} nerve growth factor receptors mediate nerve growth factor stimulated calcium uptake. *J. Biol. Chem.* **272**, 6835–6837.
- Jing, S., Yu, Y., Fang, M., Hu, Z., Holst, P. L., Boone, T., Delaney, J., Schultz, H., Zhou, R. and Fox, G. M. (1997) GFRα-2 and GFRα-3 are two new receptors for ligands of the GDNF family. J. Biol. Chem. 272, 33111–33117.
- Joe, E. and Angelides, K. (1992) Clustering of voltage-dependent sodium channels on axons depends on Schwann cell contact. *Nature* 356, 333–335.
- Johnson, K. W., Schaus, J. M., Durkin, M. M., Audia, J. E., Kaldor, S. W., Flaugh, M. E., Adham, N., Zgombick, J. M., Cohen, M. L., Branchek, T. A. and Phebus, L. A. (1997) 5-HT_{1F} receptor agonists inhibit neurogenic dural inflammation in guinea pigs. *NeuroReport* 8, 2237–2240.
- Johnson, S. W., Seutin, V. and North, R. A. (1992) Burst firing in dopamine neurons induced by N-methyl-D-aspartate: role of electrogenic sodium pump. Science 258, 665–667.
- Jonakait, G. M. (1993) Neural-immune interactions in sympathetic ganglia. *Trends Neurosci.* 10, 419–423.

- Jones, A. K. P., Brown, W. D., Friston, K. J., Ly, Q. and Frackowiak, R. S. J. (1991a) Cortical and subcortical localization of response to pain in man using positron emisssion tomography. Proc. R. Soc. Lond. (Biol). 244, 39–44.
- Jones, A. K. P. and Derbyshire, S. W. G. (1995) Cortical and thalamic imaging in normal volunteers and patients with chronic pain. In: *Forebrain Areas Involved in Pain Processing*, pp. 229– 238. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Jones, A. K. P. and Derbyshire, S. W. G. (1996) Cerebral mechanisms operating in the presence and absence of inflammatory pain. Ann. Rheum. Dis. 5, 411–420.
- Jones, B. E., Holmes, C. J., Rodriguez-Veiga, E. and Mainville, L. (1991b) GABA-synthesizing neurones in the medulla: their relationship to serotonin-containing and spinally projecting neurones in the rat. J. Comp. Neurol. 313, 349–367.
- Jones, E. G. (1985) *The Thalamus*, p. 512. Plenum Press, New York.
- Jones, E. G. and Powell, T. (1969) Connections of the somatic sensory cortex of the rhesus monkey. II. Contralateral cortical connections. *Brain* 92, 717–730.
- Jones, J. F. X., Martin, G. and Ramage, A. G. (1995) Evidence that 5-HT_{1D} receptors mediate inhibition of sympathetic ganglionic transmission in anaesthetized cats. *Br. J. Pharmac.* 116, 1715–1717.
- Jones, K. R., Farinas, I., Backus, C. and Reichardt, L. F. (1994) Targeted disruption of the BDNF gene perturbs brain and sensory neuron development but not motor neuron development. *Cell* **76**, 989–999.
- Jones, S. L. (1992) Noradrenergic modulation of noxious heatevoked fos-like immunoreactivity in the dorsal horn of the rat sacral spinal cord. J. Comp. Neurol. 325, 435–445.
- Jordan, L. M., Kenshalo, D. R., Martin, R. F., Haber, L. H. and Willis, W. D. (1979) Two populations of spinothalamic tract neurones with opposite responses to 5-hydroxytryptamine. *Brain Res.* 164, 342–346.
- Kaas, J. H. (1991) Plasticity of sensory and motor maps in adult mammals. A. Rev. Neurosci. 14, 137–167.
- Kajander, K. C. and Bennett, G. J. (1992) Onset of a painful peripheral neuropathy in rat: a partial and differential deafferentation and spontaneous discharge in $A\beta$ and $A\gamma$ primary afferent neurons. J. Neurophys. **68**, 734–744.
- Kajander, K. C., Sahara, Y., Iadarola, M. J. and Bennett, G. J. (1990) Dynorphin increases in the dorsal spinal cord in rats with a painful peripheral neuropathy. *Peptides* 11, 719–728.
- Kajander, K. C., Wakisaka, S. and Bennett, G. J. (1992) Spontaneous discharge originates in the dorsal root ganglion at the onset of a painful peripheral neuropathy in the rat. *Neurosci. Lett.* **138**, 225–228.
- Kalaria, R. N., Stockmeier, C. A. and Harik, S. I. (1989) Brain microvessels are innervated by locus coeruleus noradrenergic neurons. *Neurosci. Lett.* 97, 203–208.
- Kalkman, H. O. (1994) Is migraine prophylactic activity caused by 5-HT_{2B} or 5-HT_{2C} receptor blockade? *Life Sci.* **54**, 641–644.
- Kalso, E., Tramèr, M. R., Caroll, D., McQuay, H. J. and Moore, R. A. (1997) Pain relief from intra-articular morphine after knee surgery: a qualitative systematic review. *Pain* **71**, 127–134.
- Kane, C. J. M., Sims, T. J. and Gilmore, S. A. (1997) Astrocytes in the aged rat spinal cord fail to increase GFAP mRNA following sciatic nerve axotomy. *Brain Res.* 759, 163–165.
- Kaneko, M. and Hammond, D. L. (1997) Role of spinal γ-aminobutyric acid_A receptors in formalin-induced nociception in the rat. J. Pharmac. Exp. Ther. 282, 928–938.
- Kang, H. and Schuman, E. M. (1995) Long-lasting neurotrophininduced enhancement of synaptic transmission in the adult hippocampus. *Science* 267, 1658–1662.
- Kangrga, I. and Randic, M. (1990) Tachykinins and calcitonin gene-related peptide enhance release of endogenous glutamate and aspartate from the rat spinal dorsal horn slice. *J. Neurosci.* **10**, 2026–2038.
- Kao, C. Q. and Coulter, D. A. (1997) Physiology and pharmacology of corticothalamic stimulation-evoked responses in rat somatosensory thalamic neurons *in vitro*. J. Neurophysiol. 77, 2661–2676.
- Kapadia, S. E. and LaMotte, C. C. (1987) Deafferentation-induced alterations in the rat dorsal horn: I. Comparison of peripheral; nerve injury vs rhizotomy effects on presynaptic, postsynaptic and glial processes. J. Comp. Neurol. 266, 183–197.
- Kaplan, D. R. and Stephens, R. M. (1994) Neurotrophin signal transduction by TRK receptors. J. Neurobiol. 25, 1404–1417.
- Kaplan, M. R., Meyer-Franke, A., Lambert, S., Bennett, V., Duncan, I. D., Levinson, S. R. and Barres, B. A. (1997)

Induction of sodium channel clustering by oligodendrocytes. *Nature* **386**, 724–728.

- Kapur, A., Lytton, W. W., Ketchum, K. L. and Haberly, L. B. (1997) Regulation of the NMDA component of EPSPs by different components of postsynaptic GABAergic inhibition: computer simulation analysis in piriform cortex. J. Neurophysiol. 78, 2546–2559.
- Kashiba, H., Senba, E., Kawai, Y., Ueda, Y. and Tohyama, M. (1992) Axonal blockade induces the expression of vasoactive intestinal polypeptide and galanin in rat dorsal root ganglion neurons. *Brain Res.* 577, 19–28.
- Kashiba, H., Noguchi, K., Ueda, Y. and Senba, E. (1995) Coexpression of trk family members and low-affinity neurotrophin receptors in rat dorsal root ganglion neurons. *Mol. Brain Res.* 30, 158–164.
- Kashiba, H., Ueda, Y. and Senba, E. (1996) Coexpression of preprotackykinin-A (PPTA), *α*-calcitonin gene-related peptide (*α*-CGRP), somatostatin (SOM) and neurotrophin receptor family mRNAs in rat dorsal root ganglion neurons. *Neuroscience* 70, 179–189.
- Kashiba, H., Ueda, Y. and Senba, E. (1997) Systemic capsaicin in the adult rat differentially affects gene expression for neuropeptides and neurotrophin receptors in primary sensory neurons. *Neuroscience* 76, 299–312.
- Kask, K., Berthold, M. and Bartfai, T. (1997) Galanin receptors: involvement in feeding, pain, depression and alzheimer's disease. *Life Sci.* **60**, 1523–1533.
- Kastrup, J., Petersen, P., Dejgard, A., Angelo, H. and Hilstead, J. (1987) Intravenous lidocaine infusion: a new treatment of chronic painful diabetic neuropathy? *Pain* 28, 69–75.
- Katter, J. T., Dado, R. J., Kostarczyk, E. and Giesler, G. J. (1996a) Spinothalamic and spinohypothalamic tract neurons in the sacral spinal cord of rats. 1. Locations of antidromically identified axons in the cervical cord and diencephalon. J. Neurophysiol. 75, 2581–2605.Katter, J. T., Dado, R. J., Kostarczyk, E. and Giesler, G. J.
- Katter, J. T., Dado, R. J., Kostarczyk, E. and Giesler, G. J. (1996b) Spinothalamic and spinohypothalamic tract neurons in the sacral spinal cord of rats. 2. Responses to cutaneous and visceral stimuli. J. Neurophysiol. 75, 2606–2628.
- Katz, J., Redahan, C., Clairoux, M., Levytam, S., Jackson, M., Roger, S., Nierenberg, H., Kavanagh, B. P. and Sandler, A. N. (1996) High dose alfentanil pre-empts pain after abdominal hysterectomy. *Pain* 67, 212.
- Kaufmann, W. E., Worley, P. F., Pegg, J., Bremer, M. and Isakson, P. (1996) COX-2, a synaptically induced enzyme, is expressed by excitatory neurons at postsynaptic sites in rat cerebral cortex. *Proc. natl Acad. Sci. U.S.A.* 93, 2317–2321.
- Kapumann, K., Huggel, K., Heid, J., Flor, P. J., Bischoff, S., Mickel, S. J., McMaster, G., Angst, C., Bittiger, H., Froesti, W. and Bettker, B. (1997) Expression cloning of GABA_B receptors uncovers similarity to metabotropic glutamate receptors. *Nature* 386, 239–246.
- Kawabata, A., Manabe, S., Manabe, Y. and Takagi, H. (1994) Effect of topical administration of L-arginine on formalininduced nociception in the mouse: a dual role of peripherallyformed NO in pain modulation. Br. J. Pharmace 112, 547–550
- formed NO in pain modulation. Br. J. Pharmac. **112**, 547–550. Kawaja, M. D. and Crutcher, A. K. (1997) Sympathetic axons invade the brains of mice overexpressing nerve growth factor. J. Comp. Neurol. **383**, 60–72.
- Kawaja, M. D., Walsh, G. S., Petruccelli, K. and Coome, G. E. A. (1997) Sensory nociceptive axons invade the cerebellum of transgenic mice overexpressing nerve growth factor. *Brain Res.* 774, 77–86.
- Kawakami, M., Weinstein, J. N., Spratt, K. F., Chatani, K., Taub, R. J., Meller, S. T. and Gebhart, G. F. (1994) Experimental lumbar radiculopathy: immunohistochemical and quantitative demonstrations of pain induced by lumbar nerve root irritation of the rat. *Spine* **19**, 1780–1794.
- Kawamata, M. and Omote, K. (1996) Involvement of increased excitatory amino acids and intracellular Ca²⁺ concentration in the spinal dorsal horn in an animal model of neuropathic pain. *Pain* 68, 85–96.
- Kawamura, T., Akira, T., Watanabe, W. and Kagitani, Y. (1997a) Prostaglandin E₁ prevents apoptotic cell death in superficial dorsal horn of rat spinal cord. *Neuropharmacology* 36, 1023– 1030.
- Kawamura, T., Yamauchi, T., Koyama, M., Maruyama, T., Akira, T. and Nakamura, N. (1997b) Expression of prostaglandin EP₂ receptor mRNA in the rat spinal cord. *Life Sci.* 61, 2111–2116.
- Kawano, S., Okajima, S., Mizoguchi, A., Tamai, K., Hirasawa, Y. and Ide, C. (1997) Immunocytochemical distribution of Ca²⁺ independent protein kinase C subtypes (δ , ϵ , and ζ) in regenerat-

ing axonal growth cones of rat peripheral nerve. *Neuroscience* **81**, 263–273.

- Kayser, V., Peschanski, M. and Guilbaud, G. (1985) Neuronal loss in the ventrobasal complex of the rat thalamus alters behavioral responses to noxious stimulation. *Adv. Pain Res. Ther.* 9, 277–284.
- Keay, K. A., Feil, K., Gordon, B. D., Herbert, H. and Bandler, R. (1997) Spinal afferents to functionally distinct periaqueductal gray columns in the rat: an anterograde and retrograde tracing study. J. Comp. Neurol. 358, 207–229.
- Keele, C. A. and Armstrong, D. (1964) Substances Producing Pain and Itch, p. 271. Arnold Press, London.
- Kelly, D. C., McQueen, D. S. and Perkins, M. N. (1997) Indomethacin prevents the IL-1β-induced increase in neural discharge and enhancement of kinin-mediated activation of C-fibre mechanociceptors innervating the rat knee joint. Br. J. Pharmac. 122, 357P.
- Kelly, J. S., Zehava, G. and Schon, F. (1973) Reduction in GAD I activity from the dorsal lateral region of deafferented rat spinal cord. *Brain Res.* 26, 73–87.
- Kelso, S. R., Nelson, T. E. and Leonard, J. P. (1992) Protein kinase C-mediated enhancement of NMDA currents by metabotropic glutamate receptors in Xenopus oocytes. J. Physiol. 449, 705–718.
- Kemp, T., Spike, R. C., Watt, C. and Todd, A. J. (1996) The μopioid receptor (MOR1) is mainly restricted to neurons that do not contain GABA or glycine in the superficial dorsal horn of the rat spinal cord. *Neuroscience* **75**, 1231–1238.
- Kenney, A. M. and Kocsis, J. D. (1997) Temporal variability of jun family transcription factor levels in peripherally or centrally transected adult rat dorsal root ganglia. *Mol. Brain Res.* 52, 53– 61.
- Kenshalo, D. R., Jr and Isensee, O. (1983) Responses of primate SI cortical neurons to noxious stimuli. J. Neurophysiol. 50, 1479–1496.
- Kenshalo, D. R. Jr and Willis, W. D. (1991) The role of the cerebral cortex in pain sensation. In: *Cerebral Cortex*, pp. 153–212. Ed. A. Peters. Plenum Press, New York.
- Kenshalo, D. R., Jr, Leonard, R. B., Chung, J. M. and Willis, W. D. (1979) Responses of primate spinothalamic neurons to graded and to repeated noxious heat stimuli. *J. Neurophysiol.* 42, 1370–1389.
- Kenshalo, D. R., Jr, Giesler, G. J., Jr, Leonard, R. B. and Willis, W. D. (1980) Responses of neurons in primate ventral posterior lateral nucleus to noxious stimuli. J. Neurophysiol. 43, 1594– 1614.
- Kenshalo, D. R., Jr, Leonard, R. B., Chung, J. M. and Willis, W. D. (1982) Facilitation of the response of primate spinothalamic cells to cold and to tactile stimuli by noxious heating of the skin. *Pain* 12, 141–152.
- Kerekes, N., Landry, M., Rydh-Rinder, M. and Hökfelt, T. (1997) The effect of NGF, BDNF and bFGF on expression of galanin in cultured rat dorsal root ganglia. *Brain Res.* 754, 131–141.
- Kessey, K. and Mogul, D. J. (1997) NMDA-independent LTP by adenosine A₂ receptor-mediated postsynaptic AMPA potentiation in hippocampus. J. Neurophysiol. 78, 1965–1972.
- Khakh, B. S., Humphrey, P. P. A. and Henderson, G. (1997) ATP-gated cation channels (P_{2x} purinoceptors) in trigeminal mesencephalic nucleus neurones of the rat. J. Physiol. 498, 709– 715.
- Khan, A. A., Raja, S. N., Manning, D. C., Campbell, J. N. and Meyer, R. A. (1992) The effects of bradykinin and sequence-related analogs on the response properties of cutaneous nociceptors in monkey. *Somatosens Motor Res.* 9, 97–106.
- Khandwala, H., Hodge, E. and Loomis, C. W. (1997) Comparable dose-dependent inhibition of AP-7 sensitive strychnine-induced allodynia and paw pinch-induced nociception by mexiletine in the rat. *Pain* 72, 299–308.
- Kharazia, V. N. and Weinberg, R. J. (1997) Tangential synaptic distribution of NMDA and AMPA receptors in rat neocortex. *Neurosci. Lett.* 238, 41–44.
- Khasar, S. G., Ouseph, A. K., Choub, B., Ho, T., Green, P. G. and Levine, J. D. (1995) Is there more than one prostaglandin E receptor subtype mediating hyperalgesia in the rat hindpaw? *Neuroscience* 64, 1161–1165.
- Kiang, J. G. and Wei, E. T. (1985) CRF: an inhibitor of neurogenic plasma extravasation produced by saphenous nerve stimulation. *Eur. J. Pharmac.* 114, 111–112.
- Kikuchi, K., Ishida, H. and Kimura, T. (1997) Molecular cloning of a novel member of semaphorin family genes, semaphorin Z^{1,2}. Mol. Brain Res. 51, 229–237.

- Kilo, S., Schmelz, M., Koltzenburg, M. and Handwerker, H. O. (1994) Different patterns of hyperalgesia induced by experimental inflammations in human skin. *Brain* 117, 385–396.
- Kilo, S., Harding-Rose, C., Hargreaves, K. M. and Flores, C. M. (1997) Peripheral CGRP release as a marker for neurogenic inflammation: a model system for the study of neuropeptide secretion in rat paw skin. *Pain* 73, 201–207.
- Kim, H. G., Wang, T., Olafsson, P. and Lu, B. (1994) Neurotrophin 3 potentiates neuronal activity and inhibits gamma-aminobutyrergic synaptic transmission in cortical neurons. *Proc. natl Acad. Sci. U.S.A.* 91, 12341–12345.
- Kim, S. H. and Chung, J. M. (1991) Sympathectomy alleviates mechanical allodynia in an experimental animal model for neuropathy in the rat. *Neurosci. Lett.* **134**, 131–134.
- Kim, S. H. and Chung, J. M. (1992) An experimental model for peripheral neuropathy produced by segmental spinal nerve ligation in the rat. *Pain* 50, 355–363.
- Kim, U., Sanchez-Vives, M. V. and McCormick, D. A. (1997a) Functional dynamics of GABAergic inhibition in the thalamus. *Science* 278, 130–134.
- Kim, Y. I., Na, H. S., Han, J. S. and Hong, S. K. (1995) Critical role of the capsaicin-sensitive nerve fibers in the development of the causalgic symptoms produced by transecting some but not all of the nerves innervating the rat tail. J. Neurosci. 15, 4133– 4139.
- Kim, Y. I., Na, H. S., Yoon, Y. W., Han, H. C., Ko, K. H. and Hong, S. K. (1997b) NMDA receptors are important for both mechanical and thermal allodynia from peripheral nerve injury in rats. *NeuroReport* 8, 2149–2153.
- Kimelberg, H. K. (1995) Receptors on astrocytes: what possible functions? *Neurochem. Int.* 26, 27–40.
- King, A. E. and Thompson, S. W. N. (1995) Brief and prolonged changes in spinal excitability following peripheral injury. *Semin. Neurosci.* 7, 233–243.
- King, A. E., Slack, J. R., Lopez-Garcia, J. A. and Ackley, M. A. (1997a) Tachykinin actions on deep dorsal horn neurones in vitro: an electrophysiological and morphological study in the immature rat. *Eur. J. Neurosci.* 9, 1037–1046.
- King, B. F., Wildman, S. S., Ziganshina, L. E., Pintor, J. and Burnstock, G. (1997b) Effects of extracellular pH on agonism and antagonism at the recombinant P_{2X2} receptor. Br. J. Pharmac. 121, 1445–1453.
- Kingery, W. S. (1997) A critical review of controlled clinical trials for peripheral neuropathic pain and complex regional pain syndromes. *Pain* 73, 123–139.
- Kinnman, E. and Levine, J. D. (1995) Involvement of the sympathetic postganglionic neuron in capsaicin-induced secondary hyperalgesia in the rat. *Neuroscience* 65, 283–291.
- Kinnman, E., Nygårds, E. V. and Hansson, P. (1997) Peripheral α adrenoceptors are involved in the development of capsaicininduced ongoing and stimulus evoked pain in humans. *Pain* **69**, 79–85.
- Kirschstein, T., Büsselberg, D. and Treede, R-D. (1997) Coexpression of heat-evoked and capsaicin-evoked inward currents in acutely dissociated rat dorsal root ganglion neurons. *Neurosci. Lett.* 231, 33–36.
- Kitchener, P. D., Wilson, P. and Snow, P. J. (1994) Sciatic axotomy compromises axonal transport of tranganglionic tracer BSI-B₄ from the soma to the central terminals of C fibre afferents. *Neurosci. Lett.* 166, 121–125.
- Kitto, K. F., Haley, J. E. and Wilcox, G. L. (1992) Involvement of nitric oxide in spinally mediated hyperalgesia in the mouse. *Neurosci. Lett.* 148, 1–5.
- Kitzukawa, T., Shinizy, M., Sanbo, M., Hirata, T., Taniguchi, M., Bekku, Y., Yagi, T. and Fujisawa, H. (1997) Neurophilinsemaphorin III/D-mediated chemorepulsive signals play a crucial role in peripheral nerve projection in mice. *Neuron* 19, 995– 1005.
- Kjørsvik, A., Størkson, R., Tjølsen, A. and Hole, K. (1997) Differential effects of activation of lumbar and thoracic 5-HT_{2A} and 5-HT_{2C} receptors on nociception in rats. *Pharmac. Biochem. Behav.* **56**, 523–527.
- Klement, W. and Arndt, J. O. (1992) The role of nociceptors of cutaneous veins in the mediation of cold pain in man. J. *Physiol.* 449, 73–83.
- Knipper, M., Leung, L. S., Zhao, D. and Rylett, R. J. (1994) Short-term modulation of glutamatergic synapses in adult rat hippocampus by NGF. *NeuroReport* 5, 2433–2436.
 Von Knorring, L. (1989) Serotonin metabolites in the CSF of
- Von Knorring, L. (1989) Serotonin metabolites in the CSF of chronic pain patients. In: Serotonin and Pain. Excerpta Medica, pp. 85–303. Ed. J. M. Besson. Elsevier, Amsterdam.
- Knyihar-Csillik, E. and Török, A. (1989) Cytochemical restoration in the upper dorsal horn after transganglionic degenerative atro-

phy: temporospatial and fine structural correlates. *Neuroscience* **33**, 75–91.

- Knyihar-Csillik, E., Rakic, P. and Csillik, B. (1987) Transganglionic degeneration atrophy in the substantia gelatinosa of the spinal cord after peripheral nerve transection in rhesus monkeys. *Cell Tissue Res.* 247, 599–604.
 Knvihar-Csillik, E., Rakic, P. and Csillik, B. (1989) Transneuronal
- Knyihar-Csillik, E., Rakic, P. and Csillik, B. (1989) Transneuronal degeneration in the Rolando substance of the primate spinal cord evoked by axotomy-induced transganglionic degenerative atrophy of central primary sensory terminals. *Cells Tissue Res.* 258, 515–525.
- Knyihar-Csillik, E., Török, A. and Csillik, B. (1990) Primary afferent origin of substance P-containing axons in the superficial dorsal horn of the rat spinal cord: depletion, regeneration and replenishment of presumed nociceptive central terminals. J. Comp. Neurol. 297, 594–6120.
- Knyihar-Csillik, E., Kreutzberg, G. W., Raivich, G. and Csillik, B. (1991) Vasoactive intestinal polypeptide in dorsal root terminals of the rat spinal cord is regulated by axoplasmic transport in the peripheral nerve. *Neurosci. Lett.* **131**, 83–87.
- Knyihar-Csillik, E., Csillik, B. and Oestreicher, A. B. (1992) Light and electron microscopic localisation of B-50 (GAP-43) in the rat spinal cord during transganglionic degenerative atrophy and regeneration. J. Neurosci. Res. 32, 93–109.
- Kobayashi, H., Koppel, A. M., Luo, Y. and Raper, J. A. (1997a) A role for collapsin-1 in olfactory and cranial sensory axon guidance. J. Neurosci. 17, 8339–8352.
- Kobayashi, J. I., Ohta, M. and Terada, Y. (1996) Tetrodotoxin-resistant Na⁺ spikes of C fibers have at least two subtypes in the isolated bullfrog sciatic nerve. *Neurosci. Lett.* **221**, 9–12.
- Kobayashi, N. R., Fan, D-P., Giehl, K. M., Bedard, A. M., Wiegand, S. J. and Tetzlaff, W. (1997b) BDNF and NT-4/5 prevent atrophy of rat rubrospinal neurons after cervical axotomy, stimulate GAP-43 and Tα1-tubulin mRNA expression, and promote axonal regeneration. J. Neurosci. 17, 9583–9595.
- Koerber, H. R. (1996) Plasticity of dorsal horn cell receptive fields after peripheral nerve regeneration. J. Neurophysiol. 75, 2255– 2267.
- Kokaia, A., Gidö, G., Ringstedt, T., Bengzon, J., Kokaia, M., Siesjö, B. K., Persson, H. and Lindvall, O. (1993) Rapid increase of BDNF mRNA levels in cortical neurons following spreading depression: regulation by glutaminergic mechanisms independent of seizure activity. *Molec. Brain Res.* 19, 277–286.
- Kolhekar, R. and Gebhart, G. F. (1996) Modulation of spinal visceral nociceptive transmission by NMDA receptor activation in rat. J. Neurophysiol. 75, 2344–2353.
- Kolhekar, R., Murphy, S. and Gebhart, G. F. (1997) Thalamic NMDA receptors modulate inflammation-produced hyperalgesia in the rat. *Pain* **71**, 31–40.
- Köller, H., Siebler, M. and Hartung, H. P. (1997) Immunologically induced electrophysiological dysfunction: implications for inflammation diseases of the CNS and PNS. *Prog. Neurobiol.* 52, 1–56.
- Kolodkin, A. L. (1996) Growth cones and the cues that repel them. *Trends Neurosci.* 19, 507–513.
- Kolodkin, A. L., Levengood, D. V., Rowe, E. G., Tai, Y-T., Giger, R. J. and Ginty, D. D. (1997) Neurophilin is a semaphorin III receptor. *Cell* **90**, 753–762.
- Koltzenburg, M. (1995) Stability and plasticity of nociceptor function and their relationship to provoked and ongoing pain. *Semin. Neurosci.* 7, 199–210.
- Koltzenburg, M., Lundberg, L. E. R. and Torebjork, H. E. (1992) Dynamic and static components of mechanical hyperalgesia in human hairy skin. *Pain* 51, 207–220.
- Koltzenburg, M., Torebjork, H. E. and Wahren, L. K. (1994) Nociceptor modulated central plasticity causes mechanical hyperalgesia in chronic neuropathic and acute chemogenic pain. *Brain* 117, 579–591.
- Koltzenburg, M., Lewin, G. R., Toyka, K. V., Thoenen, H. and Carrol, P. (1995) Electrophysiological analysis of cutaneous sensory neurons in neonatal wildtype mice and transgenic animals lacking BNDF. Am. Soc. Neurosci. 21, 1054.
- Koltzenburg, M., Stucky, C. L. and Lewin, G. R. (1997) Receptive properties of mouse sensory neurons innervating hairy skin. J. Neurophysiol. 78, 1841–1850.
- Komuro, H. and Rakic, P. (1993) Modulation of neuronal migration by NMDA receptors. *Science* 260, 95–97.
 Komuro, H. and Rakic, P. (1996) Intracellular Ca²⁺ fluctuations
- Komuro, H. and Rakic, P. (1996) Intracellular Ca²⁺ fluctuations modulate the rate of neuronal migration. *Neuron* 17, 275–285.
- Kontinen, V. K., Aarnisalo, A. A., Idänpään-Heikkilä, J. J., Panula, P. and Kalso, E. (1997) Neuropeptide FF in the rat spinal cord during carrageenan inflammation. *Peptides* 18, 287– 292.

- Koplas, P. A., Rosenberg, R. L. and Oxford, G. S. (1997) The role of calcium in the desensitization of capsaicin responses in rat dorsal root ganglion neurons. J. Neurosci. 10, 3525-3537
- Korte, M., Carroll, P., Wolf, E., Brem, G., Thoenen, H. and Bonhoeffer, T. (1995) Hippocampal long-term potentiation is impaired in mice lacking brain-derived neurotrophic factor. Proc. natl Acad. Sci. U.S.A. 92, 8856-8860.
- Kostarczyk, E., Zhang, X. J. and Giesler, G. J. (1997) Spinohypothalamic tract neurons in the cervical enlargement of rats: locations of antidromically identified ascending axons and their collateral branches in the contralateral brain. J. Neurophysiol. 77, 435-451.
- Kotzbauer, P. T., Lampe, P. A., Heuckeroth, R. D., Golden, J. P., Creedon, D. J., Johnson, E. M. and Milbrandt, J. (1996) Neurturin, a relative of glial cell-line-derived neurotrophic factor. Nature 384, 467-471.
- Kozlova, E. N., Strömberg, I., Bygdeman, M. and Aldskogius, H. (1995) Peripherally grafted human foetal dorsal root ganglion cells extend axons into the spinal cord of adult host rats by circumventing dorsal root entry zone astrocytes. NeuroReport 6, 269-276.
- Kramis, R. C., Roberts, W. J. and Gillette, R. G. (1996) Post-sympathectomy neuralgia: hypotheses on peripheral and central neuronal mechanisms. Pain 64, 1-9.
- Krause, J. E., Dimaggio, D. A. and McCarson, K. E. (1995) Alterations in neurokinin 1 receptor gene expression in models of pain and inflammation. Can. J. Physiol. Pharmac. 73, 854-859
- Kress, M. and Reeh, P. W. (1996) More sensory competence for nociceptive neurons in culture. Proc. natl Acad. Sci. U.S.A. 14995-14997
- Kress, M., Koltzenburg, M., Reeh, P. W. and Handwerker, H. O. (1992) Responsiveness and functional attributes of electrically localized terminals of cutaneous C-fibers in vivo and in vitro. J Neurophysiol. 68, 581-595.
- Kress, M., Riedl, B., Rödl, J. and Reeh, P. W. (1994) Effects of nitric oxide, methylene blue and cyclic nucleotides on nocicentive nerve endings in rat skin in vitro. In: The Biology of Nitric Oxide: Physiology and Clinical Aspects, Vol. 3, pp. 319-323 Eds. S. Moncada, M. Feelisch, R. Busse and E. A. Higgs Portlands Press, London.
- Kress, M., Rödl, J. and Reeh, P. W. (1996) Stable analogues of cyclic AMP but not cyclic GMP sensitize unmyelinated primary afferents in rat skin to heat stimulation but not inflammatory mediators, in vitro. Neuroscience 74, 609-617.
- Kress, M., Reeh, P. W. and Vyklichy, L. (1997) An interaction of inflammatory mediators and protons in small diameter dorsal root ganglion neurons of the rat. Neurosci. Lett. 224, 37-40.
- Krishek, B. J., Amato, A., Connolly, C. N., Moss, S. J. and Smart, T. G. (1996) Proton sensitivity of the GABA_A receptor is associated with the receptor subunit composition. J. Physiol. 492, 431-443
- Kuhn, T. B., Williams, C. V., Dou, P. and Kater, S. B. (1998) Laminin directs growth cone navigation via two temporally and functionally distinct calcium signals. J. Neurosci. 18, 184-194.
- Kullmann, D. M. and Asztely, F. (1998) Extrasynaptic glutamate spillover in the hippocampus: evidence and implications. Trends Neurosci. 21, 8–14.
- Kumazawa, T., Mizumura, K., Minagawa, M. and Tsuiji, Y (1991) Sensitizing effects of bradykinin on the heat responses of the visceral nociceptor. J. Neurophysiol. 66, 1819-1824.
- Kuraischi, Y., Nanayama, T., Ohno, H., Fuji, N., Otaka, A., Yajima, H. and Satoh, M. (1989) Calcitonin gene-related peptide increases in the dorsal root ganglia of adjuvant arthritic rat. *Peptides* **10**, 447–452.
- Kurvers, H. A. J. M., Tangelder, G. J., De Mey, J. G. R., Slaaf, D. W., Van Den Wildenberg, F. A. J. M., Kitslaar, P. J. E. H. M., Reneman, R. S., Rouwet, E. V. and Jacobs, J. H. M. (1996) Skin blood flow disturbances in the contralateral limb in a peripheral mononeuropathy in the rat. *Neuroscience* **74**, 935-943.
- Kyrozis, A., Albuquerque, C., Gu, J. and MacDermott, A. B. (1996) Ca²⁺-dependent inactivation of NMDA receptors: fast kinetics and high Ca2+ sensitivity in rat dorsal horn neurons. J. Physiol. 495, 449-463.
- Ladiwala, U., Lachance, C., Simoneau, S. J. J., Bhakar, A. Barker, P. A. and Antel, J. P. (1998) p75 Neurotrophin receptor expression on adult human oligodendrocytes: signalling without cell death in response to NGF. J. Neurosci. 18, 1297-1304.
- Laem, J. W., Willis, W. D., Weller, S. C. and Chung, J. M. (1993) Differential activation and classification of cutaneous afferents in the rat. J. Neurophysiol. 70, 2411-2424.

- Laird, J. M. A. and Cervero, F. (1989) A comparative study of the changes in receptive-field properties of multireceptive and nociceptive rat dorsal horn neurons following noxious mechanical stimulation. J. Neurophysiol. 62, 854-863.
- Laird, J. M. A. and Bennett, G. J. (1992) Dorsal root potentials and afferent input to the spinal cord in rats with an experimental peripheral neuropathy. Brain Res. 584, 181-190.
- Lam, H. H. D., Hanley, D. F., Trapp, B. D., Saito, S., Raja, S., Dawson, T. M. and Yamaguchi, H. (1996) Induction of spinal cord neuronal nitric oxide synthase (NOS) after formalin injection in the rat hind paw. Neurosci. Lett. 210, 201-204.
- LaMotte, C. C. and Kapadia, S. E. (1993) Deafferentation-induced terminal field expansion of myelinated saphenous afferents in the adult rat dorsal horn following pronase injection of the sciatic nerve. J. Comp. Neurol. 330, 83-94.
- LaMotte, C. C., Kapadia, S. E. and Kocol, C. M. (1989) Deafferentation-induced expansion of sciatic terminal field labelling in the adult rat dorsal horn following pronase injection of the sciatic nerve. J. Comp. Neurol. 228, 311–325.
- LaMotte, R. H. and Thalhammer, J. G. (1982) Response properties of high-threshold cutaneous cold receptors in the primate. Brain Res. 244, 279-287.
- LaMotte, R. H., Thalhammer, J. G., Torebjörk, H. E. and Robinson, C. J. (1982) Peripheral neural mechanisms of cutaneous hyperalgesia following mild injury by heat. J. Neurosci. 2, 765-781.
- LaMotte, R. H., Simone, D. A., Ngeow, J. Y. F., White-House, J., Becerra-Cabal, L. and Putterman, G. J. (1987) The magnitude and duration of itch produced by intracutaneous injections of histamine. Somatosens. Res. 5, 81–92. LaMotte, R. H., Shain, C. N., Simone, D. A. and Tsai, E. F.
- (1991)Neurogenic hyperalgesia: psychophysical studies of underlying mecanisms. J. Neurophysiol. 66, 190-211
- Lamour, Y., Guilbaud, G. and Willer, J. C. (1983) Altered properties and laminar distribution of neuronal responses to peripheral stimulation in the Sm I. Cortex of the arthritic rat. Brain Res. 273, 183-187.
- Lance, J. W. (1993) Mechanism and Management of Headache, 5th edn, p. 512. Butterworths, London.
- Lang, E., Novak, A., Reeh, P. W. and Handwerker, H. O. (1990) Chemosensitivity of fine afferents from rat skin in vitro. J. Neurophysiol. 63, 887–901.
- Langenbach, R., Morham, S. G. and Tiano, H. F. (1995) Prostaglandin synthase 1 gene disruption in mice reduces arachidonic acid-induced inflammation and indomethacin-induced gastric ulceration. Cell 83, 483-492.
- Langlois, A., Pascaud, X., Junien, J.-L., Dahl, S. G. and Rivière, P. J. M. (1996) Response heterogeneity of 5-HT₃ receptor antagonist in a rat visceral hypersensitivity model. Eur. J. Pharmac. 318, 141-144.
- Lanteri-Minet, M., Weil-Fugazza, J., de Pommery, J. and Menétrey, D. (1994) Hindbrain structures involved in pain processing as revealed by the expression of c-fos and other immediate early gene proteins. Neuroscience 58, 287-298.
- Larson, A. A. and Kitto, K. F. (1997) Mutual antagonism between nerve growth factor and substance P N-terminal activity on nociceptive activity in mice. J. Pharmac. Exp. Ther. 282, 1345-1350.
- Laszlo, F., Whittle, B. J. R., Evans, S. M. and Moncada, S. (1995) Association of microvascular leakage with induction of nitric oxide synthase: effects of nitric oxide synthase inhibitors in var-
- ious organs. Eur. J. Pharmac. 283, 47–53.
 Laughlin, T. M., Vanderah, T. W., Lashbrook, J., Nichols, M. L., Ossipov, M., Porreca, F. and Wilcox, J. L. (1997) Spinally administered dynorphin A produces long-lasting allodynia: involvement of NMDA but not opioid receptors. Pain 72, 253-260.
- Lawand, N. B., Willis, W. D. and Westlund, K. N. (1997a) Blockade of joint inflammation and secondary hyperalgesia by L-NAME, a nitric oxide synthase inhibitor. NeuroReport 8, 895-
- Lawand, N. B., Willis, W. D. and Westlund, K. N. (1997b) Excitatory amino acid receptor involvement in peripheral nociceptive transmission in rats. Eur. J. Pharmac. 324, 169-177.
- Lawson, S. N. (1992) Morphological and biochemical cell types of sensory neurons. In: Sensory Neurons, Diversity, Development and Plasticity, pp. 27-59. Ed. P. Scott. Oxford University Press, New York
- Lawson, S. N., Crepps, B. A. and Perl, E. R. (1997) Relationship of substance P to afferent characteristics of dorsal root ganglion neurones in guinea-pig. J. Physiol. 505, 177–191. Le, K. T., Villeneuve, P., Ramjaun, A. R., McPherson, P. S.
- Beaudet, A. and Séguéla, P. (1998) Sensory presynaptic and

widespread somatodendritic immunolocalization of central ionotropic P_{2X} ATP receptors. *Neuroscience* **83**, 177–190.

- Leah, J. D., Sandkuhler, J., Herdegen, T., Murashov, A. and Zimmermann, M. (1992) Potentiated expression of fos protein in the rat spinal cord following bilateral noxious cutaneous stimulation. *Neuroscience* 48, 525–532.
- Le Bars, D. (1988) Serotonin and pain, In: *Neuronal Serotonin*, pp. 171–230. Eds. N. N. Osborne and M. Hamon. Wiley and Sons, London.
- Leclere, P., Ekström, P., Edström, A., Priestley, J., Averill, S. and Tonge, D. A. (1997) Effects of glial cell line-derived neurotrophic factor on axonal growth and apoptosis in adult mammalian sensory neurons in vitro. *Neuroscience* 82, 545–558.
- Ledent, C., Vaugeois, J-M., Schiffmann, S. N., Pedrazzini, T., El Yacoubi, M., Vanderhaeghen, J-J., Costentin, J., Heath, J. K., Vassart, G. and Parmentier, M. (1997) Aggressiveness, hypoalgesia and high blood pressure in mice lacking the adenosine A_{2A} receptor. *Nature* 388, 674–678.Lee, D. H., Chung, K. and Chung, J-M. (1997a) Strain differences
- Lee, D. H., Chung, K. and Chung, J-M. (1997a) Strain differences in adrenergic sensitivity of neuropathic pain behaviors in an experimental rat model. *NeuroReport* 8, 3453–3456.
- Lee, M. R., Li, L. and Kitazawa, T. (1997b) Cyclic GMP causes Ca^{2+} desensitization in vascular smooth muscle by activating the myosin light chain phosphatase. *J. Biol. Chem.* **272**, 5063–5068.
- Lee, S. M. and Ebner, F. F. (1992) Induction of high frequency activity in the somatosensory thalamus of rats in vivo results in long-term potentiation of responses in the SI cortex. *Exp. Brain Res.* **90**, 253–261.
- Lee, Y. W. and Yaksh, T. L. (1996) Pharmacology of the spinal adenosine receptor which mediates the antiallodynic action of intrathecal adenosine agonists. J. Pharmac. Exp. Ther. 277, 1642–1648.
- Leem, J. W., Gwak, Y. S., Nam, T. S. and Paik, K. S. (1997) Involvement of α_2 -adrenoceptors in mediating sympathetic excitation of injured dorsal root ganglion neurons in rats with spinal nerve ligation. *Neurosci. Lett.* **234**, 39–42.
- Leeson, P. D. and Iversen, L. L. (1994) The glycine site on the NMDA receptor. Structure-activity relationships and therapeutic potential. J. Med. Chem. 37, 4054–4067.
 Leidenheimer, N. J. (1996) Effect of PKG activation on recombi-
- Leidenheimer, N. J. (1996) Effect of PKG activation on recombinant GABA_A receptors. *Mol. Brain Res.* **42**, 131–134.
 Leidenheimer, N. J., McQuilkin, S. J., Hahner, L. D., Whiting, P.
- Leidenheimer, N. J., McQuilkin, S. J., Hahner, L. D., Whiting, P. and Harris, R. A. (1992) Activation of protein kinase C selectively inhibits the *y*-aminobutyric acid_A receptor: role of desensitization. *Mol. Pharmac.* **41**, 1116–1123.
- Lejeune, F., Newman-Tancredi, A., Audinot, V. and Millan, M. J. (1997) Interactions of (+)- and (-)-8- and 7-hydroxy-2-(di-npropylamino)tetralin at human (h)D₃, hD₂ and h serotonin_{1A} receptors and their modulation of the activity of serotoninergic and dopaminergic neurones in the rat. *J. Pharmac. Exp. Ther.* **280**, 1241–1249.
- Lekan, H. A., Carlton, S. M. and Coggeshall, R. E. (1996) Sprouting of B-HRP fixed fibers of the L5 spinal nerve following L5 and L6 spinal ligation in the rat. *Neurosci. Lett.* 208, 147–150.
- Lekan, H. A., Chung, K., Yoon, Y. W., Chung, J. M. and Coggeshall, R. E. (1997) Loss of dorsal root ganglion cells concomitant with dorsal root axon sprouting following segmental nerve lesions. *Neuroscience* 81, 527–534.
- Lenz, F. A. (1995) The region posterior inferior to the human thalamic principal sensory nucleus (Vc) may contribute to the affective dimension of pain through thalamo-corticolimbic connections. In: *Forebrain Areas Involved in Pain Processing*, pp. 119–130. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Lenz, F. A., Kwan, H. C., Dostrovsky, J. O. and Tasker, R. R. (1989) Characteristics of the bursting pattern of action that occurs in the thalamus of patients with central pain. *Brain Res.* 496, 357–360.
- Lenz, F. A., Gracely, R. H., Zirh, A. T., Romanoski, A. J. and Dougherty, P. M. (1997) The sensory-limbic model of pain memory. *Pain Forum* 6, 22–31.
- Leon, A., Buriani, A., Dal Toso, R., Fabris, M., Romanello, S., Aloe, L. and Levi-Montalcini, R. (1994) Mast cells synthetize, store and release nerve growth factor. *Proc. natl Acad. Sci.* U.S.A. 91, 3739–3743.
- Lerea, L. S. (1997) Glutamate receptors and gene induction: signaling from receptor to nucleus. *Cell. Signalling*. 9, 219–226.
- Lerea, L. S., Carlson, N. G., Simonato, M., Morrow, J. D., Roberts, J. L. and McNamara, J. O. (1997) Prostaglandin $F2\alpha$ is required for NMDA receptor-mediated induction of c-fos mRNA in dentate gyrus neurons. J. Neurosci. **17**, 117–124.

- Leslie, T. A., Emson, P. C., Dowd, P. M. and Woolf, C. J. (1995) Nerve growth factor contributes to the upregulation of GAP-43 and preprotachykinin A mRNA in primary sensory neurons following peripheral inflammation. *Neuroscience* 67, 753–761.
- Levi-Montalcini, R., Skaper, S. D., Dal Toso, R., Petrelli, L. and Leon, A. (1996) Nerve growth factor: from neurotrophin to neurokine. *Trends Neurosci.* 19, 514–520.
- Levine, E. S., Dreyfus, C. F., Black, I. B. and Plummer, M. R. (1995) Brain-derived neurotrophic factor rapidly enhances synaptic transmission in hipppocampal neurons via postsynaptic tyrosine kinase receptors. *Proc. natl Acad. Sci. U.S.A.* 92, 8074– 8077.
- Levine, J. D. and Taiwo, Y. (1994) Inflammatory pain. In: *Textbook of Pain*, 3rd edn, 1524, pp. 45–56. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Levine, J. D., Lau, W., Kwait, G. and Goetzl, E. J. (1984) Leukotriene B4 produces hyperalgesia that is dependent on polymorphonuclear leucocytes. *Science* 225, 743–745.
- Levine, J. D., Lam, D., Taiwo, Y. O., Donatoni, P. and Goetzl, E. J. (1986a) Hyperalgesic properties of 15-lipoxygenase products of arachidonic acid. *Proc. natl Acad. Sci. U.S.A.* 83, 5331–5334.
- Levine, J. D., Taiwo, Y. O., Collins, S. D. and Tam, J. K. (1986b) Noradrenaline hyperalgesia is mediated through interaction with sympathetic postganglionic neurone terminals rather than activation of primary afferent nociceptors. *Nature* 323, 158–160.
- Levine, J. D., Fields, H. L. and Basbaum, A. I. (1993) Peptides and the primary afferent nociceptor. J. Neurosci. 13, 2273–2286.
- Lewin, G. R. (1995) Neurotrophic factors and pain. Semin. Neurosci. 7, 227–232.
 Lewin, G. R. and Barde, Y.-A. (1996) Physiology of neurotro-
- Lewin, G. R. and Barde, Y.-A. (1996) Physiology of neurotrophins. A. Rev. Neurosci. 19, 289–317.
- Lewin, G. R. and McMahon, S. B. (1991) Physiological properties of primary sensory neurons appropriately and inappropriately innervating skeletal muscle in adult rats. J. Neurophysiol. 66, 1218–1231.
- Lewin, G. R. and Mendell, L. M. (1993) Nerve growth factor and nociception. *Trends Neurosci.* 16, 353–358.
- Lewin, G. R. and Mendell, L. M. (1994) Regulation of cutaneous C-fiber heat nociceptors by nerve growth factor in the developing rat. J. Neurophysiol. 71, 941–949.
- Lewin, G. R., Ritter, A. M. and Mendell, L. M. (1993) Nerve growth factor-induced hyperalgesia in the neonatal and adult rat. J. Neurosci. 13, 2136–2148.
- Lewin, G. R., McKintosh, E. and McMahon, S. B. (1994a) NMDA receptors and activity-dependant tuning of the receptive fields of spinal cord neurons. *Nature* 369, 482–485.
- Lewin, G. R., Rueff, A. and Mendell, L. M. (1994b) Peripheral and central mechanisms of NGF-induced hyperalgesia. *Eur. J. Neurosci.* 6, 1903–1912.
- Lewis, C., Neldhart, S., Holy, C., North, R. A., Buell, G. and Surprenant, A. (1995) Coexpression of P_{2X2} and P_{2X3} receptor subunits can account for ATP-gated currents in sensory neurons. *Nature* 377, 432–435.
- Lewis, T. (1942) Pain, p. 183. Macmillan Press, New York
- Lexin, G. R. and Barde, Y. A. (1996) Physiology of the neurotrophins. A. Rev. Neurosci. 19, 289–317.
- Li, H., Ohishi, H., Kinoshita, A., Shigemoto, R., Nomura, S. and Mizuno, N. (1997a) Localization of a metabotropic glutamate receptor, mGluR7, in axon terminals of presumed nociceptive, primary afferent fibers in the superficial layers of the spinal dorsal horn: an electron microscope study in the rat. *Neurosci. Lett.* 223, 153–156.
- Li, J. and Perl, E. R. (1995) ATP modulation of synaptic transmission in the spinal substantia gelatinosa. J. Neurosci. 15, 3357–3365.
- Li, J. L., Ding, Y. Q., Shigemoto, R. and Mizuno, N. (1996) Distribution of trigeminothalamic and spinothalamic tract neurons showing substance P receptor-like immunoreactivity in the rat. *Brain Res.* **719**, 207–212.
- Li, J. L., Kaneko, T., Shigemoto, R. and Mizuno, N. (1997b) Distribution of trigeminohypothalamic and spinohypothalamic tract neurons displaying substance P receptor-like immunoreactivity in the rat. J. Comp. Neurol. **378**, 508–521.
- Li, J. L. and Mizuno, N. (1997) Parabrachial nucleus neurons providing axons to both the thalamus and the spinal cord in the rat. *Brain Res.* **745**, 321–327.
- Li, Y., Erzurumlu, R. S., Chen, C., Jhaveri, S. and Tonegawa, S. (1994) Whisker-related neuronal patterns fail to develop in the trigeminal brainstem nuclei of NMDAR1 knockout mice. *Cell* 76, 1–20.
- Li, Y., Hou, X. E. and Dahlstrom, A. (1995) GAP-43 and its relation to autonomic and sensory neurons in sciatic nerve and

gastrocnemius muscle in the rat. J. Auton. Nerv. Syst. 50, 299-309.

- Lichtenberger, L. M., Wang, Z.-M., Romero, J. J., Ulloa, C., Perez, J. C., Giraud, N. M. and Barreto, J. C. (1995) Non-steroidal anti-inflammatory drugs (NSAIDs) associate with zwitterionic phospholipids: insight into the mechanism and reversal of NSAID-induced gastrointestinal injury. *Nature Med.* 1, 154– 158.
- Lieberman, A. R. (1971) The axon reaction: a review of the principal features of perikaryal responses to axon injury. *Int. Rev. Neurobiol.* 14, 49–124.
- Lieberman, D. N. and Mody, I. (1994) Regulation of NMDA channel function by endogenous Ca²⁺-dependent phosphatase. *Nature* **369**, 235–239.
- Liebl, D. J., Tessarollo, L., Palko, M. E. and Parada, L. F. (1997) Absence of sensory neurons before target innervation in brainderived neurotrophic factor-, neurotrophin 3-, and TrkC-deficient embryonic mice. J. Neurosci. 17, 9113–9121.
- Liedtke, W. and Edelmann, W. (1996) GFAP is necessary for the integrity of CNS white matter architecture and long-term maintenance of myelination. *Neuron* 17, 607–615.
- Light, A. R. and Kavookjian, A. M. (1988) Morphology and ultrastructure of physiologically identified substantia gelatinosa (lamina II) neurons with axons that terminate in deeper dorsal horn laminae (III-V). J. Comp. Neurol. 267, 172–189.
- Light, A. R. and Perl, E. R. (1979) Reexamination of the dorsal root projection to the spinal dorsal horn including observations on the differential termination of coarse and fine fibres. J. Comp. Neurol. 186, 117–132.
- Lima, D. and Coimbra, A. (1988) The spinothalamic system of the rat: structural types of retrogradely labelled neurons in the marginal zone (lamina I). *Neuroscience* 27, 215–230.
- Lin, Q., Peng, Y. B. and Willis, W. D. (1994) Glycine and GABA_A antagonists reduce the inhibition of primate spinothalamic tract neurons produced by stimulation in periaqueductal gray. *Brain Res.* 654, 286–302.
- Lin, Q., Peng, Y. B. and Willis, W. D. (1996a) Inhibition of primate spinothalamic tract neurons by spinal glycine and GABA is reduced during central sensitization. J. Neurophysiol. 76, 1005–1014.
- Lin, Q., Peng, Y. B. and Willis, W. D. (1996b) Possible role of protein kinase C in the sensitization of primate spinothalamic tract neurons. J. Neurosci. 16, 3026–3034.
- Lin, Q., Peng, Y. B. and Willis, W. D. (1996c) Role of GABA receptor subtypes in inhibition of primate spinothalamic tract neurons: difference between spinal and periaqueductal gray inhibition. J. Neurophysiol. 75, 109–123.
- Lin, Q., Peng, Y. B. and Willis, W. D. (1996d) The effects of 8bromo-cGMP on the inhibition of primate spinothalamic neurons produced by stimulation in periaqueductal gray and GABA receptors. In: 8th World Congress on Pain, International Association for the Study of Pain, pp. 473–474.
 Lin, Q., Peng, Y. B., Wu, J. and Willis, W. D. (1997) Involvement
- Lin, Q., Peng, Y. B., Wu, J. and Willis, W. D. (1997) Involvement of cGMP in nociceptive processing by and sensitization of spinothalamic neurons in primates. J. Neurosci. 17, 3293–3302.
- Lin, S.-Z., Chiou, A.-L. and Wang, Y. (1996) Ketamine antagonizes nitric oxide release from cerebral cortex after middle cerebral artery ligation in rat. *Stroke* 27, 747–752.
- Linden, D. J. and Connor, J. A. (1996) Long-term synaptic depression. A. Rev. Neurosci. 18, 319–357.
- Lindblom, U. (1985) Assessment of abnormal evoked pain in neurological pain patients and its relationship to spontaneous pain: a descriptive and conceptual model with some analytical results. In: Advances in Pain Resarch and Therapy, Vol. 9, pp. 409–424. Eds. H. L. Fields, R. Dubner, F. Cervero and L. E. Jones. Raven Press, New York.
- Lindholm, D., Heumann, R., Meyer, M. and Thoenen, H. (1987) Interleukin-1 regulates synthesis of nerve growth factor in nonneuronal cells of rat sciatic nerve. *Nature* 330, 658–659.
- Lindsay, R. M. and Harmar, A. J. (1989) Nerve growth factor regulates expression of neuropeptide genes in adult sensory neurons. *Nature* 337, 362–364.
- Lindsay, R. M., Wiegand, S. J., Altar, C. A. and Di Stefano, P. S. (1994) Neurotrophic factors: from molecule to man. *Trends Neurosci.* 17, 182–190.
- Lindvall, O., Érnfors, P., Bengzon, J., Kokaia, Z., Smit, M.-L., Siesjö, B. K. and Persson, H. (1992) Differential regulation of mRNAs for nerve growth factor, brain derived neurotrophic factor, and neurotrophin 3 in the adult rat brain following cerebral ischaemia and hypoglycemic coma. *Proc. natl Acad. Sci.* U.S.A. 89, 648–652.
- Lingueglia, E., de Weille, J. R., Bassilana, F., Heurteaux, C., Sakait, H., Waldmann, R. and Lazdunski, M. (1997) A modula-

tory subunit of acid sensing ion channels in brain and dorsal root ganglion cells. J. Biol. Chem. 272, 29778–29783.

- Lipchick, G. L., Holroyd, K. A., France, C. R., Kvaal, S. A., Segal, D., Cordingley, G. E., Rokicki, L. A. and McCool, H. R. (1996) Central and peripheral mechanisms in chronic tension-type headache. *Pain* 64, 467–475.
- Lisman, J. (1994) The CaM kinase II hypothesis for the storage of synaptic memory. *Trends Neurosci.* 17, 406–413.
 Lisman, J. E. (1997) Bursts as a unit of neural information: mak-
- Lisman, J. E. (1997) Bursts as a unit of neural information: making unreliable synapses reliable. *Trends Neurosci.* 20, 38–43.
- Lisman, J. E., Malenka, R. C., Nicoll, R. A. and Malinow, R. (1997) Learning mechanisms: the case for CaM-KII. *Science* **276**, 2001–2002.
- Lisney, S. J. W. (1983) Changes in the somatotopic organization of the cat lumbar spinal cord following peripheral nerve transection and regeneration. *Brain Res.* 259, 31–39.
- Lisney, S. J. W. and Devor, M. (1987) Afterdischarge and interactions among fibers in damaged peripheral nerves in the rat. *Brain Res.* 415, 122–136.
- Littlewood, N. K., Todd, A. J., Spike, R. C., Watt, C. and Shehab, S. A. S. (1995) The types of neuron in spinal dorsal horn which possess neurokinin-1 receptors. *Neuroscience* 66, 597–608.
- Liu, H., Brown, J. L., Jasmin, L., Maggio, J. E., Vigna, S. R., Manthyh, P. W. and Basbaum, A. I. (1994a) Synaptic relationship between substance P and the substance P receptor: Light and electron miscroscopic characterisation of the mismatch between neuropeptides and their receptors. *Proc. natl Acad. Sci.* U.S.A. 91, 1009–1013.
- Liu, H., Wang, H., Sheng, M., Jan, L. Y., Jan, Y. N. and Basbaum, A. I. (1994b) Evidence for presynaptic N-methyl-Daspartate autoreceptors in the spinal cord dorsal horn. Proc. natl Acad. Sci. U.S.A. 91, 8383–8387.
- Liu, H., Force, T. and Bloch, K. D. (1997a) Nerve growth factor decreases soluble guanylate cyclase in rat pheochromocytoma PC12 cells. *J. Biol. Chem.* 272, 6038–6043.
 Liu, H., Mantyh, P. W. and Basbaum, A. I. (1997b) NMDA-
- Liu, H., Mantyh, P. W. and Basbaum, A. I. (1997b) NMDAreceptor regulation of substance P release from primary afferent nociceptors. *Nature* 386, 721–724.
- Liu, L. and Simon, S. A. (1994) A rapid capsaicin-activated current in rat trigeminal ganglion neurons. *Proc. natl Acad. Sci.* U.S.A. 91, 738–741.
- Liu, L., Lo, Y.-C., Chen, I.-J. and Simon, S. A. (1997c) The responses of rat trigeminal ganglion neurons to capsaicin and two nonpungent vanilloid receptor agonists, olvanil and glyceril nonamide. J. Neurosci. 17, 4101–4111.
- Liu, M., Max, M. B., Parada, S., Rowan, J. S. and Bennett, G. J. (1996) The sympathetic nervous system contributes to capsaicinevoked mechanical allodynia but not pinprick hyperalgesia in humans. J. Neurosci. 15, 7331–7335.
- Liu, P. K., Salminen, A., He, Y. Y., Jiang, M. H., Xue, J. J., Liu, J. S. and Hsu, C. Y. (1994c) Suppression of ischemia-induced Fos expression and AP-1 activity by an antisense oligodeoxynucleotide to c-fos mRNA. Ann. Neurol. 36, 566–576.
- Liu, X. G. and Sandkühler, J. (1997) Characterization of longterm potentiation of C-fiber-evoked potentials in spinal dorsal horn of adult rat: essential role of neurokinin 1 and neurokinin 2 receptors. J. Neurophysiol. **78**, 1982–1983.
- Liu, Y., Souverbie, F., Sundström, E. and von Euler, G. (1997d) Unique properties of [³H]-MK801 binding in membranes from the rat spinal cord. *Brain Res.* **757**, 167–175.
- Liu, Z., Vergnes, M., Depaulis, A. and Marescaux, C. (1991) Evidence for a critical role of GABAergic transmission within the thalamus in the genesis and control of absence seizures in the rat. *Brain Res.* 545, 1–7.
- Livesey, F. J., O'Brien, J. A., Li, M., Smith, A. G., Murphy, L. J. and Hunt, S. P. (1997) A schwann cell mitogen accompanying regeneration of motor neurons. *Nature* 390, 614–618.
- Lo, D. C. (1995) Neurotrophic factors and synaptic plasticity. *Neuron* 15, 979–981.
- Löfgren, O., Yu, L-C., Theodorsson, E., Hansson, P. and Lundeberg, T. (1997) Intrathecal CGRP₈₋₃₇ results in a bilateral increase in hindpaw withdrawal latency in rats with a unilateral thermal injury. *Neuropeptides* **31**, 601–607.
- Lombard, M.-C. and Larabi, Y. (1983) Electrophysiological study of the cervical dorsal horn cells in partially deafferented rats. In: *Advances in Pain Research and Therapy*, Vol. 5, p. 147. Eds. J. J. Bonica, U. Lindblom and A. Iggo. Raven Press, New York.
- Lopez-Garcia, J. A. and King, A. E. (1996) A novel methodology for simultaneous assessment of the effects of 5-hydroxytryptamine on primary afferent polarisation and synaptic transmission in rat dorsal horn neurones in vitro. J. Neurosci. Meth. 68, 1–6.

- Lopshire, J. C. and Nicol, G. D. (1997) Activation and recovery of the PGE₂-mediated sensitization of the capsaicin response in rat sensory neurons. *J. Neurophysiol.* **78**, 3154–3164.
- Lou, Y-P. and Lundberg, J. M. (1992) Inhibition of low pH evoked activation of airway sensory nerves by capsazepine, a novel capsaicin-receptor antagonist. *Biochem. Biophys. Res. Commun.* 189, 537–544.
- Lovick, T. A. (1993) Integrated activity of cardiovascular and pain regulatory systems: role in adaptive behavioural responses. *Prog. Neurobiol.* 40, 631–644.
- Lovinger, D. M. and Weight, F. F. (1988) Glutamate induces depolarization of adult rat dorsal root ganglion neurons that is mediated predominantly by NMDA receptors. *Neurosci. Lett.* 94, 314.
- Lu, B. and Figurov, A. (1997) Role of neurotrophins in synapse development and plasticity. *Rev. Neurosci.* 8, 1–12.
- Lu, B., Yokoyama, M., Dreyfus, C. F. and Black, I. B. (1991) Depolarizing stimuli regulate nerve growth factor gene expression in cultured hippocampal neurons. *Proc. natl Acad. Sci.* U.S.A. 88, 6289–6292.
- Lucaites, V. L., Krushinski, J., Schaus, J. M., Audia, J. E., Gehlert, D. R. and Nelson, D. L. (1996) Autoradiographic localization of the serotonin_{1F} (5-HT_{1F}) receptor in rat brain using [³H]LY133,470, a selective 5-HT_{1F} receptor radioligand. *Am. Soc. Neurosci.* 22, 1331.
- Luckenbill-Edds, L. (1997) Laminin and the mechanism of neuronal outgrowth. Brain Res. 23, 1–27.
- Ludlam, W. H., Chandross, K. J. and Kessler, J. A. (1995) LIFand IL-1 beta mediated increases in substance P receptor mRNA in axotomized, explanted or dissociated sympathetic ganglia. *Brain Res.* 685, 12–20.
- Lumer, E. D., Edelman, G. M. and Tononi, G. (1997) Neural dynamics in a model of the thalamocortical system. 2: the role of neural synchrony tested through pertubations of spike timing. *Cerebral Cortex.* 7, 228–236.
- Luo, L. and Wiesenfeld-Hallin, Z. (1995) The effects of pretreatment with tachykinin antagonists and galanin on the development of spinal cord hyperexcitability following sciatic nerve section in the rat. *Neuropeptides* 28, 161–166.
- Lynn, B. (1990) Capsaicin actions on nociceptive C-fibres and therapeutic potential. *Pain* **41**, 61–69.
- Lynn, B. (1997) Substance P and nociceptive afferent neurones. J. Physiol. 505, 1.
- Lynn, B. and Carpenter, S. E. (1982) Primary afferent units from the hairy skin of the rat hind limb. *Brain Res.* 238, 29-43.
- Lynn, B., Schütterle, S. and Pierau, F. K. (1996) The vasodilator component of neurogenic inflammation is caused by a special subclass of heat-sensitive nociceptors in the skin of the pig. J. Physiol. 494, 587–593.
- Lyons, P. D. and Blalock, J. E. (1997) Pro-opiomelanocortin gene expression and protein processing in rat mononuclear leukocytes. J. Neuroimmunol. 78, 47–56.
- Lytton, W. W., Contreras, D., Destexhe, A. and Steriade, M. (1997) Dynamic interactions determine partial thalamic quiescence in a computer network model of spike-and-wave seizures. J. Neurophysiol. 77, 1679–1696.
- Ma, Q.-P. and Woolf, C. J. (1996a) Basal and touch-evoked foslike immunoreactivity during experimental inflammation in the rat. *Pain* 67, 307–316.
- Ma, Q.-P. and Woolf, C. J. (1996b) Progressive tactile hypersensitivity: an inflammation-induced incremental increase in the excitability of the spinal cord. *Pain* 67, 97–106.
- Ma, Q.-P. and Woolf, C. J. (1997) Tachykinin NK₁ receptor antagonist RP 67,580 attenuates progressive hypersensitivity of flexor reflex during experimental inflammation in rats. *Eur. J. Pharmac.* 322, 165–171.
- Ma, Q.-P., Zhou, J. S. and Han, J. S. (1993) Noxious stimulation accelerated the expression of c-fos protooncogene in cholecystokininergic and dopaminergic neurons in the ventral tegmental area. *Peptides* 14, 561–566.
- Ma, W. and Bisby, M. A. (1997) Differential expression of galanin immunoreactivities in the primary sensory neurons following partial and complete sciatic nerve injuries. *Neuroscience* 79, 1183–1195.
- Ma, Y. L., Wang, H. L., Wu, H. C., Wei, C. L. and Lee, E. H. Y. (1998) Brain-derived neurotrophic factor antisense oligonucleotide impairs memory retention and inhibits long-term potentation in rats. *Neuroscience* 82, 957–967.
- MacGhee, D. S., Goy, M. F. and Oxford, G. S. (1992) Involvement of the nitric oxide-cyclic GMP pathway in the desensitization of bradykinin responses of cultured rat sensory neurons. *Neuron* 9, 315–324.

- Mackawa, K., Minami, M., Masuda, T. and Satoh, M. (1995) Expression of μ - and k-, but not δ -, opoid receptor mRNAs is enhanced in the spinal dorsal horn of the arthritic rats. *Pain* **64**, 365–371.
- MacPhee, I. J. and Barker, P. A. (1997) Brain-derived neurotrophic factor binding to the p75 neurotrophin receptor reduces TrkA signaling while increasing serine phosphorylation in the TrkA intracellular domain. J. Biol. Chem. 272, 23547–23551.
- Madison, D. V., Malenka, R. C. and Nicoll, R. A. (1986) Phorbol esters block a voltage-sensitive chloride current in hippocampal pyramidal cells. *Nature* **321**, 695–697.
- Maeda, K., Fernyhough, P. and Tomlinson, D. R. (1996) Regenerating sensory neurones of diabetic rats express reduced levels of mRNA for GAP-43, γ-preprotachykinin and the nerve growth factor receptors, trkA and p75^{NGFR}. *Mol. Brain Res.* 37, 166–174.
- Maggi, C. A. (1991a) Capsaicin and primary afferent neurons: from basic science to human therapy? J. Auton. Nerv. Syst. 33, 1–14.
- Maggi, C. A. (1991b) The pharmacology of the efferent function of sensory nerves. J. Auton. Pharmac. 11, 173–208.
- Maggi, C. A. and Schwartz, T. W. (1997) The dual nature of the tachykinin NK₁ receptor. *Trends Pharmac. Sci.* 18, 351–355.
- Maier, S. F., Wiertelak, E. P., Martin, D. and Watkins, L. R. (1993) Interleukin-1 mediates the behavioral hyperalgesia produced by lithium chloride and endotoxin. *Brain Res.* 623, 321– 325.
- Mailiani, A., Pagani, M. and Lombardi, F. (1984) Visceral versus somatic mechanisms. In: *Textbook of Pain*, pp. 100–121. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Mailis, A., Amani, N., Umana, M., Basur, R. and Roe, S. (1997) Effect of intravenous sodium amytal on cutaneous sensory abnormalities, spontaneous pain and algiometric pain pressure thresholds in neuropathic pain patients: a placebo-controlled study II. *Pain* 70, 69–81.
- Malcangio, M. and Bowery, N. G. (1994) Spinal cord SP release and hyperalgesia in monoarthritic rats: involvement of the GABA_B receptor system. *Br. J. Pharmac.* 113, 1561–1566.
- Malcangio, M. and Bowery, N. G. (1996a) Calcitonin gene-related peptide content, basal outflow and electrically-evoked release from monoarthritic rat spinal cord in vitro. *Pain* 66, 351–358.
- Malcangio, M. and Bowery, N. G. (1996b) GABA and its receptors in the spinal cord. *Trends Pharmac. Sci.* 17, 457–462.
- Malcangio, M., Bowery, N. G., Flower, R. J. and Perretti, M. (1996) Effect of interleukin-1β on the release of substance P from rat isolated spinal cord. *Eur. J. Pharmac.* 299, 113–118.
- Malcangio, M., Garrett, N. E., Cruwys, S. and Tomlinson, D. R. (1997a) Nerve growth factor- and neurotrophin-3-induced changes in nociceptive threshold and the release of substance P from the rat isolated spinal cord. J. Neurosci. 17, 8459–8467.
- Malcangio, M., Garrett, N. E. and Tomlinson, D. R. (1997b) Nerve growth factor treatment increases stimulus-evoked release of sensory neuropeptides in the rat spinal cord. *Eur. J. Neurosci.* 9, 1101–1104.
- Malek-Ahmadi, P. (1996) Neuropsychiatric aspects of cytokine research: an overview. *Neurosci. Biobehav. Rev.* 20, 359–365.
- Malen, P. L. and Chapman, P. F. (1997) Nitric oxide facilitates long-term potentiation, but not long-term depression. J. Neurosci. 17, 2645–2651.
- Malenka, R. C. and Nicoll, R. A. (1997) Silent synapses speak up. *Neuron* 19, 473–476.
- Malmberg, A. B. and Yaksh, T. L. (1992) Hyperalgesia mediated by spinal glutamate or substance P receptor blocked by spinal cyclooxygenase inhibition. *Science* 257, 1276–1279.
- Malmberg, A. B. and Yaksh, T. L. (1994) Voltage-sensitive calcium channels in spinal nociceptive processing: blockade of Nand P-type channels inhibits formalin-induced nociception. J. Neurosci. 14, 4885–4890.
- Malmberg, A. B. and Yaksh, T. L. (1995) Cyclooxygenase inhibition and the spinal release of prostaglandin E₂ and amino acids evoked by paw formalin injection: a microdialysis study in unanesthetized rats. J. Neurosci. 15, 2768–2776.
- Malmberg, A. B., Brandon, E. P., Idzerda, R. L., Liu, H., Mcknight, G. S. and Basbaum, A. I. (1997a) Diminished inflammation and nociceptive pain with preservation of neuropathic pain in mice with a targeted mutation of the type I regulatory subunit of cAMP-dependent protein kinase. J. Neurosci. 17, 7462–7470.
- Malmberg, A. B., Chen, C., Tonegawa, S. and Basbaum, A. I. (1997b) Preserved acute pain and reduced neuropathic pain in mice lacking PKC_γ. *Science* 278, 279–283.

- Manning, D. C., Raja, S. N., Meyer, R. A. and Campbell, J. N. (1991) Pain and hyperalgesia after intradermal injection of bradykinin in humans. *Clin. Pharmac. Ther.* **50**, 721–729.
- Mannion, R. J., Doubell, T. P., Coggeshall, R. E. and Woolf, C. J. (1996) Collateral sprouting of injured primary afferent A-fibers into the superficial dorsal horn of the adult rat spinal cord after topical capsaicin tratment to the sciatic nerve. J. Neurosci. 16, 5189–5195.
- Mansikka, H. and Pertovaara, A. (1997a) Supraspinal influence on hindlimb withdrawal thresholds and mustard oil-induced secondary allodynia in rats. *Brain Res. Bull.* 42, 359–365.
- Mansikka, H. and Pertovaara, A. (1997b) Submodality-selective hyperalgesia adjacent to partially injured sciatic nerve in the rat is dependent on capsaicin-sensitive afferent fibers and independent of collateral sprouting or a dorsal root reflex. *Brain Res. Bull.* 44, 237–245.
- Mansour, A., Fox, C. A., Akil, H. and Watson, S. J. (1995) Opioid receptor mRNA expression in the rat CNS: anatomical and functional implications. *Trends Neurosci.* 18, 22–29.
- Mantyh, P. W., Catton, M. D., Boehmer, C. G., Welton, M. L., Passaro, E. P., Maggio, J. E. and Vigna, S. R. (1989) Receptors for sensory neuropeptides in human inflammatory diseases: implications for the effector role of sensory neurons. *Peptides* 10, 627–645.
- Mantyh, P. W., Allen, C. J., Rogers, S., DeMaster, E., Ghilard, J. R., Mosconi, T., Krugger, L., Mannon, P. J., Taylor, I. L. and Vigna, S. R. (1994) Some sensory neurones express neuropeptide Y receptors: potential paracrine inhibition of primary afferent nociceptors following peripheral nerve injury. *J. Neurosci.* 14, 3958–3968.
- Mantyh, P. W., Allen, C. J., Ghilardi, J. R., Rogers, S. D., Mantyh, C. R., Liu, H., Basbaum, A. I., Vigna, S. R. and Maggio, J. E. (1995) Rapid endocytosis of a G protein-coupled receptor: Substance P-evoked internalization of its receptor in the rat striatum in vivo. *Proc. natl Acad. Sci. U.S.A.* 92, 2622– 2626.
- Mantyh, P. W., Rogers, S. D., Honore, P., Allen, B. J., Ghilardi, J. R., Li, J., Daughters, R. S., Lappi, D. A., Wiley, R. G. and Simone, D. A. (1997) Inhibition of hyperalgesia by ablation of lamina I spinal neurons expressing the substance P receptor. *Science* 278, 275–279.
- Manzoni, O., Prezeau, L., Marin, P., Deshager, S., Bochaert, J. and Fagni, L. (1992) Nitric oxide induced blockade of NMDA receptors. *Neuron* 8, 653–662.
- Mao, J., Hayes, R. L., Price, D. D., Coghill, R. C., Lu, J. and Mayer, D. J. (1992a) Post-injury treatment with GM1 ganglioside reduces nociceptive behaviors and spinal cord metabolic activity in rats with experimental peripheral mononeuropathy. *Brain Res.* 584, 18–27.
- Mao, J., Price, D. D., Coghill, R. C., Mayer, D. J. and Hayes, R. L. (1992b) Spatial patterns of spinal cord-[¹⁴C]2-deoxyglucose metabolic activity in a rat model of painful peripheral mono-neuropathy. *Pain* 50, 89–100.
- Mao, J., Price, D. D., Hayes, R. L., Lu, J. and Mayer, D. J. (1992c) Differential roles of NMDA and non-NMDA receptor activation in induction and maintenance of thermal hyperalgesia in rats with painful peripheral mononeuropathy. *Brain Res.* 598, 271–278.
- Mao, J., Price, D. D., Mayer, D. J. and Hayes, R. L. (1992d) Pain-related increases in spinal cord membrane-bound protein kinase C following peripheral nerve injury. *Brain Res.* 588, 144– 149.
- Mao, J., Mayer, D. J., Hayes, R. L. and Price, D. D. (1993a) Spatial patterns of increased spinal cord membrane-bound protein kinase C and their relation to increases in [¹⁴C]2-deoxyglucose metabolic activity in rats with painful peripheral mononeuropathy. J. Neurophysiol. **70**, 470–481.
- Mao, J., Price, D. D., Hayes, R. L., Lu, J., Mayer, D. J. and Frenck, H. (1993b) Intrathecal treatment with dextrorphan or ketamine potently reduces pain-related behaviors in a rat model of peripheral mononeuropathy. *Brain Res.* 605, 164–168.
- Mao, J., Price, D. D., Zhu, J., Lu, J. and Mayer, D. J. (1997) The inhibition of nitric oxide-activated poly (ADP-ribose) synthetase attenuates transynaptic alteration of spinal cord dorsal horn neurons and neuropathic pain in the rat. *Pain* 72, 355–366.
- Marceau, F. (1995) Kinin B₁ receptors: a review. *Immunology* **30**, 1–26.
- Marchand, J. E., Wurm, W. H., Kato, T. and Kream, R. M. (1994) Altered tachykinin expression by dorsal root ganglion neurons in a rat model of neuropathic pain. *Pain* 58, 219–231.
- Marchettini, P., Simone, D. A., Caputi, G. and Ochoa, J. L. (1996) Pain from excitation of identified muscle nociceptors in humans. *Brain Res.* 740, 109–116.

- Marcoux, F. W. (1994) Pharmacologic therapeutic window in acute stroke. *Drug News Persp.* 7, 228–235.
- Mark, M. A., Colvin, L. A. and Duggan, A. W. (1998) Spontaneous release of immunoreactive neuropeptide Y from the central terminals of large diameter primary afferents of rats with peripheral nerve injury. *Neuroscience* 83, 571–589.
- Marlier, L., Poulat, P., Rajaofetra, N., Sandillon, F. and Privat, A. (1992) Plasticity of serotonergic innervation of the dorsal horn of the rat spinal cord following neonatal capsaicin treatment. J. Neurosci. Res. 31, 346–358.
- Marshall, G. E., Shehab, S. A. S., Spike, R. C. and Todd, A. J. (1996) Neurokinin-1 receptors on lumbar spinothalamic neurons in the rat. *Neuroscience* 72, 255–263.
- Martin, H. A. (1990) Leukotriene B₄ induced decrease in mechanical and thermal thresholds of C-fiber mechanociceptors in rat hairy skin. *Brain Res.* 509, 273–279.
- Martin, H. A., Basbaum, A. I., Goetzl, E. J. and Levine, J. D. (1988) Leukotriene B_4 decreases the mechanical and thermal thresholds of C-fiber nociceptors in the hairy skin of the rat. *J. Neurophys.* **60**, 438–445.
- Martin, S. and Morris, R. G. M. (1997) (R,S)-α-Methyl-4-carboxyphenylglycine (MCPG) fails to block long-term potentiation under urethane anaesthesia in vivo. *Neuropharmacology* 36, 1339–1354.
- Martini, R. (1994) Expression and functional roles of neural cell adhesion molecules and extracellular matrix components during development and regeneration of peripheral nerves. J. Neurocytol. 23, 1–28.
- Martiniak, J., Saganova, K. and Chavko, M. (1991) Free and peptide-bound amino acids as indicators of ischemic damage of the rabbit spinal cord. J. Neuropathology Exp. Neurol. 50, 73–81.
- Marty, S., Weinitz, J.-M. and Peschanski, M. (1994) Target dependence of adult neurons: pattern of terminal arborisations. J. Neurosci. 14, 5257–5266.
- Marty, S., Berzaghi, M. P. and Berninger, B. (1997) Neurotrophins and activity-dependent plasticity of cortical interneurons. *Trends Neurosci.* 20, 198–202.
- Maruki, C., Spatz, M., Ueki, Y., Nagatsu, I. and Bembry, J. (1984) Cerebrovascular endothelial cell culture: metabolism and synthesis of 5-hydroxytryptamine. J. Neurochem. 43, 316–319.
- Marvizon, J. C. G., Martinez, V., Grady, E. F., Bunnett, N. W. and Mayer, E. A. (1997) Neurokinin 1 receptor internalization in spinal cord slices induced by dorsal root stimulation is mediated by NMDA receptors. J. Neurosci. 17, 8129–8136.
- Masterman, D. L. and Cummings, J. L. (1997) Frontal-subcortical circuits: the anatomic basis of executive, social and motivated behaviors. J. Psychopharmac. 11, 107–114.
- Matheson, C. R., Carnahan, J., Urich, J. L., Bocangel, D., Zhang, T. J. and Yan, Q. (1997) Glial cell line-derived neurotrophic factor (GDNF) is a neurotrophic factor for sensory neurons comparison with the effects of the neurotrophins. J. Neurobiol. 32, 22–32.
- Mathiau, P., Bakalara, N. and Aubineau, P. (1994) Tryptophan hydroxylase can be present in mast cells and nerve fibers of the rat dura mater but only mast cells contain serotonin. *Neurosci. Lett.* 182, 133–137.
- Matsubara, T., Moskowitz, M. A. and Huang, Z. (1992) UK 14,304, R(-)α-methylhistamine and SMS 201,995 block plasma protein leakage within dura mater by prejunctional mechanisms. *Eur. J. Pharmac.* 224, 145–150.
- Matsumoto, N., Bester, H., Menendez, L., Besson, J.-M. and Bernard, J. F. (1996) Changes in the responsiveness of parabrachial neurons in the arthritic rat: an electrophysiological study. *J. Neurophysiol.* 76, 4113–4126.
- Matsumura, K., Watanabe, Y., Imai-Matsumara, K., Connolly, Y., Koyama, Y. O., Onoe, H. and Watanabe, Y. (1992) Mapping of prostaglandin E₂ binding sites in rat brain using quantitative autoradiography. *Brain Res.* 581, 292–298.
- Matsumura, K., Watanabe, Y., Onoe, H. and Watanabe, Y. (1995) Prostacyclin receptors in the brain and central terminals of primary sensory neurons: an autoradiographic study using a stable prostacyclin analogue [³H]-iloprost. *Neuroscience* 65, 493– 503.
- Matsuoka, I., Nakane, A. and Kurihara, K. (1997) Induction of LIF-mRNA by IGF-β1 in Schwann cells. *Brain Res.* 776, 170– 180.
- Mattson, M. P. and Mark, R. J. (1996) Excitotoxicity and excitoprotection in vitro. Adv. Neurol. 71, 1–30.
- Matzner, O. and Devor, M. (1987) Contrasting thermal sensitivity of spontaneously active A- and C-fibers in experimental nerveend neuromas. *Pain* 30, 373–384.

- Matzner, O. and Devor, M. (1994) Hyperexcitability at sites of nerve injury depends on voltage-sensitive Na⁺ channels. J. Neurosci. **72**, 349–359.
- Maves, T. J., Pechman, P. S., Gebhart, G. F. and Meller, S. T. (1993) Possible chemical contribution from chromic gut sutures produces disorders of pain sensation like those seen in man. *Pain* **54**, 57–69.
- Maxwell, D. J. and Rethelyi, M. (1987) Ultrastructure and synaptic connections of cutaneous afferents fibres in the spinal cord. *Neuroscience* 10, 117–123.
- Maxwell, L., Maxwell, D. J., Neilson, M. and Kerr, R. (1996) A confocal microscopic survey of serotoninergic axons in the lumbar spinal cord of the rat: co-localization with glutamate decarboxylase and neuropeptides. *Neuroscience* **75**, 471–480.
- May, A., Kaube, H., Büchel, C., Eichten, C., Tijntjes, M., Jüptner, M., Weiller, C. and Deiner, H. C. (1998) Experimental cranial pain elicited by capsaicin: a PET study. *Pain* 74, 61–66.
- Mayer, C., Wächtler, J., Kamleiter, M. and Grafe, P. (1997) Intracellular calcium transients mediated by P₂ receptors in the paranodal schwann cell region of myelinated rat spinal root axons. *Neurosci. Lett.* 224, 49–52.
- Mayer, M. L. and Miller, R. J. (1990) Excitatory amino acid receptors, second messengers and regulation of intracellular Ca²⁺ in mammalian neurons. *Trends Pharmac.* **11**, 254–260.
- McCall, R. B. and Clement, M. E. (1994) Role of serotonin_{1A} and serotonin₂ receptors in the central regulation of the cardiovascular system. *Pharmac. Rev.* **46**, 231–243.
- McCarson, K. E. and Krause, J. E. (1994) NK-1 and NK-3 type tachykinin receptor mRNA expression in the rat spinal cord dorsal horn is increased during adjuvant or formalin-induced nociception. *J. Neurosci.* 14, 712–720.
- McCarson, K. E. and Krause, J. E. (1996) The neurokinin-1 receptor antagonist LY 306,740 blocks nociception-induced increases in dorsal horn neurokinin-1 receptor gene expression. *Mol. Pharmac.* 50, 1189–1199.
- McCormack, K. (1994) Non-steroidal anti-inflammatory drugs and spinal nociceptive processing. *Pain* **59**, 9–43.
- McCormick, D. A. (1992) Neurotransmitter actions in the thalamus and cerebral cortex and their role in neuromodulation of thalamocortical activity. *Prog. Neurobiol.* **39**, 337–388.
- McEachern, A. E., Shelton, E. R., Bhakta, S., Obernolte, R., Bach, C., Zuppan, P., Fujisaki, J., Aldrich, R. W. and Jarnagin, K. (1991) Expression cloning of a rat B₂ bradykinin receptor. *Proc. natl Acad. Sci. U.S.A* 88, 7724–7728.
- McEachern, J. C. and Shaw, C. A. (1996) An alternative to the LTP orthodoxy: a plastic-pathology continuous model. *Brain Res. Rev.* 22, 51–92.
- McGaraughty, S. and Henry, J. L. (1997) Relationship between mechano-receptive fields of dorsal horn convergent neurons and the response to noxious immersion of the ipsilateral hindpaw in rats. *Pain* **70**, 133–140.
- McGiven, J. G., Koch, B. D., Dietrich, P. S., Sangameswaran, L. and Hunter, J. C. (1997) Electrophysiological characterisation of the novel tetrodotoxin-sensitive sodium channel, rPN4. *Br. J. Pharmac.* 122, 366P.
- McGuirk, S. M. and Dolphin, A. C. (1992) G-Protein mediation in nociceptive signal transduction and investigation into the excitatory action of bradykinin in a subpopulation of cultured rat sensory neurons. *Neuroscience* 49, 117–128.
- McKenna, J. E. and Melzack, R. (1994) Bilateral VB thalamotomy does not reduce pain in the formalin test. Am. Soc. Neurosci. 20, 1572.
- McLachlan, E. M., Jänig, W., Devor, M. and Michaelis, M. (1993) Peripheral nerve injury triggers noradrenergic sprouting within dorsal root ganglia. *Nature* 363, 543–546.
- McMahon, S. B. (1994) Mechanisms of cutaneous, deep and visceral pain. In: *Textbook of Pain*, 3rd edn, pp. 129–153. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- McMahon, S. B. (1996) NGF as a mediator of inflammatory pain. *Philos. Trans. R. Soc. Lond. [Biol.]* 351, 431–440.
- McMahon, S. B. and Bennett, D. L. H. (1997) Growth factors and pain. In: *The Pharmacology of Pain. Handbook of Experimental Pharmacology*, Vol. 130, pp. 135–166. Eds. A. Dickenson and J-M. Besson. Springer-Verlag, Berlin.
- McMahon, S. B. and Gibson, S. (1987) Peptide expression is altered when afferent nerves reinnervate inappropriate tissue. *Neurosci. Lett.* **73**, 9–15.
- McMahon, S. B. and Kett-White, R. (1991) Sprouting of peripherally regenerating primary sensory neurones in the adult central nervous system. J. Comp. Neurol. 304, 307–315.
- McMahon, S. B. and Koltzenburg, M. (1990a) Novel classes of nociceptors: beyond Sherrington. *Trends Neurosci.* 13, 199–201.

- McMahon, S. B. and Koltzenburg, M. (1990b) The changing role of primary afferent neurones in pain. *Pain* 43, 269–272.
- McMahon, S. B. and Priestley, J. V. (1995) Peripheral neuropathies and neurotrophic factors: animal models and clinical perspectives. *Curr. Opin. Neurobiol.* 5, 616–624.
- McMahon, S. B. and Wall, P. D. (1984) Receptive fields of rat lamina 1 projection cells move to incorporate a nearby region of injury. *Pain* 19, 235–247.
 McMahon, S. B., Armanini, M. P., Ling, L. H. and Phillips, H. S.
- McMahon, S. B., Armanini, M. P., Ling, L. H. and Phillips, H. S. (1994) Expression and coexpression of Trk receptors in subpopulations of adult primary sensory neurons projecting to identified peripheral targets. *Neuron* 12, 1161–1171.
- McMahon, S. B., Bennett, D. L., Priestley, J. V. and Shelton, D. L. (1995) The biological effects of endogenous nerve growth factor on adult sensory neurons revealed by a TrkA-IgG fusion molecule. *Nature Med.* 1, 740–741.
- McNeill, D. L. and Hulsebosch, C. E. (1987) Intraspinal sprouting of rat primary afferents after deafferentation. *Neurosci. Lett.* 81, 57–62.
- McNeill, D. L., Carlton, S. M., Coggeshall, R. E. and Hulsebosch, C. E. (1990) Denervation-induced intraspinal synaptogenesis of calcitonin gene-related peptide containing primary afferent terminals. J. Comp. Neurol. 296, 263–268.
- Mearow, K. M. and Kril, Y. (1995) Anti-NGF treatment blocks the upregulation of NGF receptor mRNA expression associated with collateral sprouting of rat dorsal root ganglion neurons. *Neurosci. Lett.* 184, 55–58.
- Mearow, K. M., Kril, Y., Gloster, A. S. and Diamond, J. (1994) Expression of NGF receptor and GAP-43 mRNA in DRG neurons during collateral sprouting and regeneration of dorsal cutaneous nerves. J. Neurobiol. 25, 127–142.
- Meiri, K. F. and Burdick, D. (1991) Nerve growth factor stimulation of GAP-43 phosphorylation in intact isolated growth cones. J. Neurosci. 11, 3155–3164.
- Melino, G. and Bernassola, F. (1997) S-nitrosylation regulates apoptosis. *Nature* 388, 432–433.
- Meller, S. T. and Gebhart, G. F. (1992) A critical review of the afferent pathways and the potential chemical mediators involved in cardiac pain. *Neuroscience* 48, 501–524.
- Meller, S. T. and Gebhart, G. F. (1993) Nitric oxide (NO) and nociceptive processing in the spinal cord. *Pain* **52**, 127–136.
- Meller, S. T., Dykstra, C. and Gebhart, G. F. (1992a) Production of endogenous nitric oxide and activation of soluble guanylate cyclase are required for *N*-methyl-D-aspartate-produced facilitation of the nociceptive tail flick reflex. *Eur. J. Pharmac.* 214, 93–96.
- Meller, S. T., Gebhart, G. F. and Maves, T. J. (1992b) Neonatal capsaicin treatment prevents the development of the thermal hyperalgesia produced in a model of neuropathic pain in the rat. *Pain* 51, 317–321.
- Meller, S. T., Pechman, P. S., Gebhart, G. F. and Maves, T. J. (1992c) Nitric oxide mediates the thermal hyperalgesia produced in a model of neuropathic pain in the rat. *Neuroscience* 50, 7– 10.
- Meller, S. T., Dykstra, C. L. and Gebhart, G. F. (1993) Acute mechanical hyperalgesia is produced by coactivation of AMPA and metabotropic glutamate receptors. *NeuroReport* 4, 879–882.
- Meller, S. T., Dykstra, C. L., Grzybycki, D., Murphy, S. and Gebhart, G. F. (1994) The possible role of glia in nociceptive processing and hyperalgesia in the spinal cord of the rat. *Neuropharmacology* 33, 1471–1478.
- Melzack, R. and Wall, P. D. (1965) Pain mechanisms: a new theory. Science 150, 971–979.
- Menendez, L., Bester, H., Besson, J.-M. and Bernard, J.-F. (1996) Parabrachial area—electrophysiological evidence for an involvement in cold nociception. J. Neurophysiol. 75, 2099–2116.
- Menétrey, D., Gannon, A., Levine, J. D. and Basbaum, A. I. (1989) Expression of c-fos in interneurons and projection neurons of the rat spinal cord in response to noxious somatic, articular, and visceral stimulation. J. Comp. Neurol. 285, 177– 195.
- Mense, S. (1986) Slowly conducting afferent fibers from deep tissues: neurobiological properties and central nervous actions. *Prog. Sensory Physiol.* 6, 139–220.
- Mense, S. (1993) Nociception from skeletal muscle in relation to clinical muscle pain. *Pain* 54, 241–289.
- Mense, S. and Meyer, H. (1985) Different types of slowly conducting afferent units in cat skeletal muscle and tendon. J. Physiol. 363, 403–417.
- Mense, S. and Meyer, H. (1988) Bradykinin-induced modulation of the response behaviour of different types of feline group III and IV muscle receptors. J. Physiol. 398, 49–63.
- Mense, S. and Stahnke, M. (1983) Responses in muscle afferent fibres of slow conduction velocity to contractions and ischaemia in the cat. J. Physiol. 342, 383–397.
- Mense, S., Light, A. R. and Perl, E. R. (1981) Spinal terminations of subcutaneous high-threshold mechanoreceptors. In: *Spinal Cord Sensation*, pp. 111–123. Eds. A. G. Brown and M. Rethelvi. Scottish Academic Press, Edinburgh.
- Merhi, M., Dusting, G. J. and Khalil, Z. (1998) CGRP and nitric oxide of neuronal origin and their involvement in neurogenic vasodilatation in rat skin microvasculature. *Br. J. Pharmac.* 123, 863–868.
- Merighi, A., Polak, J. M. and Theodosis, D. T. (1991) Ultrastructural visualisation of glutamate and aspartate immunoreactivities in the rat dorsal horn, with special reference to the co-localisation of glutamate, substance P and calcitoningene related peptide. *Neuroscience* 40, 67–80.
- Merrill, J. E. and Benveniste, E. N. (1996) Cytokines in inflammatory brain lesions: helpful and harmful. *Trends Neurosci.* 19, 331–338.
- Merskey, H. (1986) Pain terms—a current list with definitions and notes on usage. *Pain* **Suppl. 3** (Part II), S217.
- Merskey, H. and Bogduk, N. (1994) Classification of Chronic Pain: Descriptions of Chronic Pain Syndromes and Definition of Pain Terms, 2nd edn, p. 287. IASP Press, Seattle.
- Mertz, P. M., DeWitt, D. L., Stetler-Stevenson, W. G. and Wahl, L. M. (1994) Interleukin-10 suppression of monocyte prostaglandin H synthase—2. J. Biol. Chem. 269, 21322–21329.
- Messaouidi, E. (1998) Acute intrahippocampal infusion of BDNF induces lasting potentiation of synaptic transmission in the rat dentate gyrus. J. Neurophysiol. 79, 496–499.
- Messersmith, E. K., Leonardo, E. D., Shatz, C. J., Tessier-Lavigne, M., Goodman, C. S. and Kolodkin, A. L. (1995) Semaphorin III can function as a selective chemorepellent to pattern sensory projections in the spinal cord. *Neuron* 14, 949– 959.
- Meunier, J-C. (1997) Nociceptin/orphanin FQ and the opioid receptor-like ORL1 receptor. *Eur. J. Pharmac.* 340, 1–15.
- Meyer, R. A. and Campbell, J. N. (1981) Myelinated nociceptive afferents account for the hyperalgesia that follows a burn to the hand. *Science* 213, 1527–1529.
- Meyer, R. A., Raja, S. N. and Campbell, J. N. (1985) Coupling of action potential activity between unmyelinated fibers in the peripheral nerve of monkey. *Science* 227, 184–187.
- Meyer, R. A., Davis, K. D., Cohen, R. H., Treede, R. D. and Campbell, J. N. (1991) Mechanically insensitive afferents (MIAs) in cutaneous nerves of monkey. *Brain Res.* 561, 252– 261.
- Meyer, R. A., Campbell, J. N. and Raja, S. N. (1994) Peripheral neural mechanisms of nociception In: *Textbook of Pain*, pp. 13– 44. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Miao, F. J., Jänig, W. and Levine, J. D. (1996) Role of sympathetic postganglionic neurons in synovial plasma extravasation induced by bradykinin. J. Neurophysiol. 75, 715–724.
- Michael, G. J. and Priestley, J. V. (1995) Expression of GAP-43 mRNA in preganglionic sympathetic neurones of the adult rat spinal cord. *NeuroReport* 7, 338–342.
 Michael, G. J., Averill, S., Nitkuman, A., Rattray, M., Bennett,
- Michael, G. J., Averill, S., Nitkuman, A., Rattray, M., Bennett, D. L. H., Yan, Q. and Priestley, J. V. (1997a) Nerve growth factor treatment increases brain-derived neurotrophic factor selectively in TrkA-expressing dorsal root ganglion cells and in their central terminations within the spinal cord. J. Neurophysiol. 17, 8476–8490.
- Michael, G. J., Kaya, E., Averill, S., Rattray, M., Clary, D. O. and Priestley, J. V. (1997b) TrkA immunoreactive neurones in the rat spinal cord. J. Comp. Neurol. 385, 441–455.
- Michaelis, M., Vogel, C., Blenk, K-H. and Jänig, W. (1997) Algesics excite axotomised afferent nerve fibres within the first hours following nerve transection in rats. *Pain* 72, 347–354.
- Miettinen, S., Fusco, F. R., Yrjjänheikki, J., Keinänen, R., Hirvonen, T., Roivainen, R., Närhi, M., Hökfelt, T. and Koistinaho, J. (1997) Spreading depression and focal brain ischemia induce cyclooxygenase-2 in cortical neurons through *N*-methyl-D-aspartic acid receptors and phospholipase A₂. *Neurobiology* 94, 6500–6505.
- Miljanich, G. P. and Ramachandran, J. (1995) Antagonists of neuronal calcium channels: structure, function and therapeutic implications. A. Rev. Pharmac. Toxicol. 35, 707–734.
- Miki, K., Fukuoka, T., Tokunaga, A. and Noguchi, K. (1998) Calcitonin gene-related peptide increase in the rat spinal dorsal horn and dorsal column nucleus following peripheral nerve injury: up-regulation in a subpopulation of primary afferent sensory neurons. *Neuroscience* 82, 1243–1252.

- Miletic, V. and Tan, H. (1988) Iontophoretic application of calcitonin gene-related peptide produces a slow and prolonged excitation of neurons in the cat lumbar dorsal horn. *Brain Res.* 446, 169–172.
- Millan, M. J. (1986) Multiple opioid systems and pain: a review. *Pain* **26**, 303–349.
- Millan, M. J. (1990) Kappa-opioid receptors and analgesia. Trends. Pharmac. Sci. 11, 70–76.
- Millan, M. J. (1993) Multiple Opioid Systems and Chronic Pain. In: Opioids II. Handbook of Experimental Pharmacology, Vol. 104, pp. 127-162. Ed. A. Herz. Springer-Verlag, Berlin.
- Millan, M. J. (1995) Serotonin (5-HT) and pain: a reappraisal of its role in the light of receptor multiplicity. *Semin. Neurosci.* 7, 409–419.
- Millan, M. J. (1997) The role of descending noradrenergic and serotonergic pathways in the modulation of nociception: focus on receptor multiplicity. In: *The Pharmacology of Pain. Handbook* of Experimental Pharmacology, Vol. 130, pp. 385-446. Eds. A. Dickenson and J-M. Besson. Springer-Verlag, Berlin.
- Millan, M. J. (1998) α_2 -adrenergic mechanisms of analgesia. Strategies for improving their therapeutic window and identification of the novel, potent α_{2A} -adrenergic receptor agonist, S 18616. *Adv. Pharmac.* **42**, 575–579.
- Millan, M. J. and Colpaert, F. C. (1991) Opioid systems in the response to inflammatory pain: sustained blockade of k- but not μ -opioid receptors suggests a role in modulation of nociception, behaviour and pathology. *Neuroscience* **42**, 541–553.
- Millan, M. J. and Gobert, A. (1997) Autoreceptor and heteroceptor control of 5-HT, dopamine (DA) and noradrenaline (NAD) release in frontal cortex: strategies for improved antidepressant agents. *Br. J. Pharmac.* **122**, 468P.
- Millan, M. J., Bervoets, K., Rivet, J.-M., Widdowson, P., Renouard, A., Le Marouille-Girardon, S. and Gobert, A. (1994) Multiple alpha₂-adrenergic receptor subtypes. II. Evidence for a role of rat Ra_{2A}-ARs in the control of nociception, motor behaviour and hippocampal synthesis of noradrenaline. J. Pharmac. Exp. Ther. 270, 958–972.
- Millan, M. J., Seguin, L., Honoré, P., Girardon, S. and Bervoets, K. (1996) Pro- and antinociceptive actions of serotonin (5-HT)_{1A} agonists and antagonists in rodents: relationship to algesiometric paradigm. *Behav. Brain Res.* **73**, 69–77.
- Miller, F. D., Speelman, A., Mathew, T. C., Fabian, J., Chang, E., Pozniak, C. and Toma, J. G. (1994) Nerve growth factor derived from terminals selectively increases the ratio of p75 to trkA NGF receptors on mature sympathetic neurons. *Dev. Biol.* 161, 206–217.
- Miller, K. J., Mariano, C. L. and Cruz, W. R. (1997) Serotonin 5-HT_{2A} receptor activation inhibits inducible nitric oxide synthase activity in C6 glioma cells. *Life Sci.* 61, 1819–1827.
- Miller, L. G., Galpern, W. R., Dunlap, K., Dinarello, C. A. and Turner, T. J. (1991) Interleukin-1 augments gamma-aminobyturic acid_A receptor function in brain. *Mol. Pharmac.* 39, 105–108.
- Milne, B., Duggan, S., Jhamandas, K. and Loomis, C. (1996) Innocuous hair deflection evokes a nociceptive-like activation of catechol oxidation in the rat locus coeruleus following intrathecal strychnine: a biochemical index of allodynia using in vivo voltammetry. *Brain Res.* **718**, 198–202.
- Minami, T., Nishihara, I., Uda, R., Ito, S., Hyodo, M. and Hayaishlo, O. (1994) Characterization of EP receptor subtypes involved in allodynia and hyperalgesia induced by intrathecal administration of prostaglandin E_2 to mice. *Br. J. Pharmac.* **112**, 740–765.
- Minami, T., Okuda-Ashitaka, E., Nishizawa, M., Mori, H. and Ito, S. (1997a) Inhibition of nociceptin-induced allodynia in conscious mice by prostaglandin D₂. Br. J. Pharmac. **122**, 605– 610.
- Minami, T., Sugatani, J., Sakimura, K., Abe, M., Mishina, M. and Ito, S. (1997b) Absence of prostglandin E₂-induced hyperalgesia in NMDA receptor *e* subunit knockout mice. Br. J. Pharmac. **120**, 1522–1526.
- Ming, G. L., Song, H. J., Berninger, B., Holt, C. E., Tessier-Lavigne, M. and Poo, M. M. (1997) cAMP-dependent growth cone guidance by netrin-1. *Neuron* 19, 1225–1235.
- Minghetti, L. and Levi, G. (1998) Microglia as effector cells in brain damage and repair: focus on prostanoids and nitric oxide. *Progr. Neurobiol.* 54, 99–125.
- Minghetti, L., Polazzi, E., Nicolini, A., Créminon, C. and Levi, G. (1997) Up-regulation of cyclooxygenase-2 expression in cultured microglia by Prostaglandin E₂, cyclic AMP and non-steroidal anti-inflammatory drugs. *Eur. J. Neurosci.* 9, 934–940.
- Miranda, H. F., Bustamante, D., Kramer, V., Pelisser, T., Saavedra, H., Paeile, C., Fernadez, E. and Pinardini, G. (1992)

Antinociceptive effects of Ca^{2+} channel blockers. *Eur. J. Pharmac.* **217**, 137–141.

- Mistry, R., Golding, N. and Challiss, R. A. J. (1998) Regulation of phosphoinositide turnover in neonatal rat cerebral cortex by group I- and II-selective metabotropic glutamate receptor agonists. *Br. J. Pharmac.* **123**, 581–589.
 Mitchell, J. A., Saunders, M., Barnes, P. J., Newton, R. and
- Mitchell, J. A., Saunders, M., Barnes, P. J., Newton, R. and Belvisi, M. G. (1997) Sodium salicylate inhibits cyclo-oxygenase-2 activity independently of transcription factor (nuclear factor κB) activation: role of arachidonic acid. *Mol. Pharmac.* **51**, 907–912.
- Miyamoto, A., Laufs, U., Pardo, C. and Liao, J. K. (1997) Modulation of bradykinin receptor ligand binding affinity and its coupled G-proteins by nitric oxide. J. Biol. Chem. 272, 19601–19608.
- Miyasaka, N. and Hirata, Y. (1997) Nitric oxide and inflammatory arthritides. *Life Sci.* 61, 2073–2081.
- Mjellem-Joly, N., Lund, A., Berge, O-G. and Hole, K. (1991) Potentiation of behavioural response in mice by spinal coadministration of substance P and excitatory amino acid agonists. *Neurosci. Lett.* 133, 121–124.
- Mnich, S. J., Veenhuizen, A. W., Monahan, J. B., Sheehan, K. C. F., Lynch, K. R., Isakson, P. C. and Portanova, J. P. (1995) Characterization of a monoclonal antibody that neutralizes the activity of prostagandin E₂. J. Immun. 155, 4437–4444.
 Mohammadian, P., Hummel, T., Loetsch, J. and Kobal, G. (1997)
- Mohammadian, P., Hummel, T., Loetsch, J. and Kobal, G. (1997)
 Bilateral hyperalgesia to chemical stimulation of the nasal mucosa following unilateral inflammation. *Pain* 73, 407–412.
 Mohiuddin, L., Fernandez, K., Tomlinson, D. R. and
- Mohiuddin, L., Fernandez, K., Tomlinson, D. R. and Fernyhough, P. (1995) Nerve growth factor and neurotrophin-3 enhance neurite outgrowth and up-regulate the levels of messenger RNA for growth-associated protein GAP-43 and α_1 -Tubulin in cultured adult rat sensory neurones. *Neurosci. Lett.* **185**, 20– 23
- Molander, C. and Grant, G. (1986) Laminar distribution and somatotopic organization of primary afferent fibres from hindlimb nerves in the dorsal horn: a study by transganglionic transport of horseradish peroxidase in the rat. *Neuroscience* **19**, 297– 312.
- Molander, C., Hongpaisan, J. and Grant, G. (1992) Changing pattern of c-fos expression in spinal cord neurons after electrical stimulation of chronically injured sciatic nerve in the rat. *Neuroscience* 50, 223–236.
- Molander, C., Hongpaisan, J., Svensson, M. and Aldskogius, H. (1997) Glial cell reactions in the spinal cord after sensory nerve stimulation are associated with axonal injury. *Brain Res.* 747, 122–129.
- Molderings, G. J., Frölich, D., Likungu, J. and Göthert, M. (1996) Inhibition of noradrenaline release via presynaptic 5-HT_{1Dz} receptors in human atrium. *Naunyn-Schmiedeberg's Arch Pharmac.* **353**, 272–280.
- Molinari, M., Bentivoglio, M., Minciacchi, D., Granato, A. and Macchi, G. (1986) Spinal afferents and cortical afferents of the anterior intralaminar nuclei: an anterograde-retrograde tracing study. *Neurosci. Lett.* **72**, 258–264.
- Mollenholt, P., Post, C., Paulsson, I. and Rawal, N. (1990) Intrathecal and epidural somatostatin in rats: can antinociception, motor effects and neurotoxicity be separated? *Pain* 43, 363–370.
- Moller, K., Reimer, M., Ekblad, E., Hannibal, J., Fahrenkrug, J., Kanje, M. and Sundler, F. (1997a) The effects of axotomy and preganglionic denervation on the expression of pituitary adenylate cyclase activating peptide (PACAP), galanin and PACAP type 1 receptors in the rat superior cervical ganglion. *Brain Res.* 775, 166–182.
- Moller, K., Reimer, M., Hannibal, J., Fahrenkrug, J., Sundler, F. and Kanje, M. (1997b) Pituitary adenylate cyclase-activating peptide (PACAP) and PACAP type 1 receptor expression in regenerating adult mouse and rat superior cervical ganglia in vitro. Brain Res. 775, 156–165.
- Molliver, D. C., Radeke, M. J., Feinstein, S. C. and Snider, W. D. (1995) Presence or absence of TrkA protein distinguishes subsets of small sensory neurons with unique cytochemical characteristics and dorsal horn projections. J. Comp. Neurol. 361, 404– 416.
- Molliver, D. C., Wright, D. E., Leitner, M. L., Parsadanian, A. S., Doster, K., Wen, D., Yan, Q. and Snider, W. D. (1997) IB-4 binding DRG neurons switch from NGF to GDNF dependance in early postnatal life. *Neuron* 19, 849–861.
- Momiyama, A., Feldmeyer, D. and Cull Candy, S. G. (1996) Identification of a low-conductance NMDA channel with reduced sensitivity to Mg^{2+} in rat central neurones. *J Physiol.* **494**, 479–492.

- Monhemius, R., Li, H. S. and Roberts, M. H. T. (1997) Descending influences differentially modulate superficial lamina I and deep multireceptive dorsal horn neurones in the rat. *J. Physiol.* **505**, 43P.
- Montieh, G. R. and Roufogalis, B. D. (1995) The plasma membrane calcium pump: a physiological perspective on its regulation. *Cell Calcium* 18, 459–470.
- Morgan, M. M. (1998) Direct comparison of heat-evoked activity of nociceptive neurons in the dorsal horn with the hindpaw withdrawal reflex in the rat. J. Neurophysiol. 79, 174–180.
- Morgan, M. M., Heinricher, M. M. and Fields, H. L. (1994) Inhibition and facilitation of different nocifensor reflexes by spatially remote noxious stimuli. J. Neurophysiol. 72, 1152– 1160.
- Morgan, M. M., Grisel, J. E., Robbins, C. S. and Grandy, D. K. (1997) Antinociception mediated by the periaqueductal gray is attenuated by orphanin FQ. *NeuroReport* 8, 3431–3434.
- Morisset, V. and Nagy, F. (1996) Modulation of regenerative membrane properties by stimulation of metabotropic glutamate receptors in rat dorsal horn neurons. J. Neurophysiol. 76, 2794– 2798.
- Morris, B. J. (1997) Stabilization of dendritic mRNAs by nitric oxide allows localized, activity-dependent enhancement of hippocampal protein synthesis. *Eur. J. Neurosci.* 9, 2334–2339.
- Moskowitz, M. A. (1992) Neurogenic versus vascular mechanisms of sumatriptan and ergot alkaloids in migraine. *Trends Pharmac.* 13, 307–311.
- Moufid-Bellancourt, S., Razafimanalina, R. and Velley, L. (1996) Interaction between mu and kappa receptors located in the parabrachial area in the opioid control of preference threshold for saccharin: modulatory role of lateral hypothalamic neurones. *Behav. Pharmac.* 7, 798–809.
- Mulder, H., Zhang, Y., Danielson, N. and Sundler, F. (1997) Islet amyloid polypeptide and calcitonin gene-related peptide expression are uprelated in lumbar dorsal root ganglia after unilateral adjuvant-induced inflammation in the rat paw. *Mol. Brain Res.* 50, 127–135.
- Mulderry, P. K. and Lindsay, R. M. (1990) Rat dorsal root ganglion neurons in culture express vasoactive intestinal peptide (VIP) independently of nerve growth factor. *Neurosci. Lett.* 108, 314–320.
- Munglani, R., Hudspith, M. J. and Hunt, S. P. (1996) The therapeutic potential of neuropeptide Y. Drugs 52, 371–389.
- Munro, F. E., Fleetwood-Walker, S. M. and Mitchell, R. (1994) Evidence for a role of protein kinase C in the sustained activation of rat dorsal horn neurons evoked by cutaneous mustard oil application. *Neurosci. Lett.* **170**, 199–202.
- Murase, K. and Randic, M. (1984) Actions of substance P on rat spinal dorsal horn neurones. J. Physiol. 346, 203–217.
- Murata, T., Ushikubi, F., Matsuoka, T., Irata, M., Yamasaki, A., Sugimoto, Y., Ichikawa, A., Aze, Y., Tanaka, T., Yoshida, N., Ueno, A., Oh-Ishi, S. and Narumlya, S. (1997) Altered pain perception and inflammatory response in mice lacking prostacyclin receptor. *Nature* 388, 678–682.
- Murphy, P. G., Grondin, J., Altares, M. and Richardson, P. M. (1995) Induction of interleukin-6 in axotomized sensory neurons. J. Neurosci. 15, 5130–5138.
- Murphy, S. (1993) Astrocytes: Pharmacology and Function, p. 412. Academic Press, San Diego.
- Myers, R. R., Heckmann, H. M. and Rodriguez, M. (1996) Reduced hyperalgesia in nerve-injured WLD mice: relationship to nerve fiber phagocytosis, axonal degeneration and regeneration in normal mice. *Exp. Neurol.* 141, 94–101.
- Na, H. S., Leem, J. W. and Chung, J. M. (1993) Abnormalities of mechanoreceptors in a rat model of neuropathic pain: possible involvement in mediating mechanical allodynia. *J. Neurophysiol.* **70**, 522–528.
- Na, H. S., Yoon, Y. W. and Chung, J. M. (1996) Both motor and sensory abnormalities contribute to changes in foot posture in an experimental rat neuropathic model. *Pain* 67, 173–178.
- Nachemson, A. K. and Bennett, G. J. (1993) Does pain damage spinal cord neurons? Transsynaptic degeneration in rats following a surgical incision. *Neurosci. Lett.* **162**, 78–80.
- Naciff, J. M., Kaetzel, M. A., Behbehani, M. M. and Dedman, J. R. (1996) Differential expression of annexins I-VI in the rat dorsal root ganglia and spinal cord. J. Comp. Neurol. 368, 356–370.
- Nadeson, R., Guo, Z., Porter, V., Gent, J. P. and Goodchild, C. S. (1996) γ-Aminobutyric acid_A receptors and spinally mediated antipociception in rate. *L. Pharmac. Exp. Ther.* **278**, 620–626
- antinociception in rats. J. Pharmac. Exp. Ther. **278**, 620–626. Nagashima, T., Tanabe, H., Fujioka, Y., Nakashima, K. and Hidaka, H. (1991) Immunohistochemical localization of protein kinase C isozymes in human cerebellum, hippocampus and spinal cord. Acta Histochem. Cytochem. **24**, 441–446.

146

- Nagy, I., Maggi, C. A., Dray, A., Woolf, C. J. and Urban, L. (1993) The role of neurokinin and *N*-methy-D-aspartate receptors in synaptic transmission from capsaicin sensitive primary afferents in the rat spinal cord in vitro. *Neuroscience* 52, 1029– 1037.
- Nagy, I., Miller, B. A. and Woolf, C. J. (1994) NK₁ and NK₂ receptors contribute to C-fibre evoked slow potentials in the rat spinal cord. *NeuroReport* 5, 2105–2108.
- Nahin, R. L. and Hylden, K. K. (1991) Peripheral inflammation is associated with increased glutamic acid decarboxylase immunoreactivity in the rat spinal cord. *Neurosci. Lett.* **128**, 226–230.
- Nahin, R. L., Ren, K., DeLeon, M. and Ruda, M. (1994) Primary sensory neurons exhibit altered gene expression in a rat model of neuropathic pain. *Pain* 58, 95–108.
- Naim, M., Spike, R. C., Watt, C., Shehab, S. A. S. and Todd, A. J. (1997) Cells in laminae III and IV of the rat spinal cord that possess the neurokinin-1 receptor and have dorsally directed dendrites receive a major synaptic input from tachykinin-containing primary afferents. J. Neurosci. 17, 5536–5548.
- Nakamura, F. and Strittmatter, S. M. (1996) P_{2Y1} purinergic receptors in sensory neurons: contribution to touch-induced impulse generation. J. Neurobiol. 93, 10465–10470.
- Nakamura, M. and Ferreira, S. H. (1988) Peripheral analgesic action of clonidine: mediation by release of endogenous enkephalin-like substances. *Eur. J. Pharmac.* 146, 223–228.
- Namgung, U., Matzuyama, S. and Routtenberg, A. (1997) Longterm potentiation activates the GAP-43 promoter: selective participation of hippocampal mossy cells. *Proc. natl Acad. Sci.* U.S.A. 94, 11675–11680.
- Naranjo, J. R., Mellstrom, B., Achaval, M. and Sassone-Corsi, P. (1991) Molecular pathways of pain: Fos/Jun-mediated activation of a noncanonical AP-1 site in the prodynorphin gene. *Neuron* 6, 607–617.
- Nasström, J., Karlsson, U. and Post, C. (1992) Antinociceptive actions of different classes of excitatory amino acid receptor antagonists in mice. *Eur. J. Pharmac.* 212, 21–29.
- Nauta, H. J. W., Hewitt, E., Westlund, K. N. and Willis, W. D. (1997) Surgical interruption of a midline dorsal column visceral pain pathway: case report and review of the literature. J. Neurosurg. 86, 538-542.
- Neurosurg, **86**, 538–542. Naveilhan, P., Eishamy, W. M. and Ernfors, P. (1997) Differential regulation of mRNAs for GDNF and its receptors Ret and GDNFR α after sciatic nerve lesion in the mouse. *Eur. J. Neurosci.* **9**, 1450–1460.
- Neal, J. W., Pearson, R. C. A. and Powell, T. P. S. (1990) The ipsilateral cortico-cortical connections of the area 7b PF in the parietal and temporal lobes of the monkey. *Brain Res.* 524, 119–132.
- Nebe, J., Vanegas, H., Neugebauer, V. and Schaible, H.-G. (1997) ω -Agatoxin IVA, a P-type calcium channel antagonist, reduces nociceptive processing in spinal cord neurons with input from the inflamed but not from the normal knee joint—an electrophysiological study in the rat in vivo. *Eur. J. Neurosci.* **9**, 2193– 2201.
- Nedergaard, M. (1994) Direct signalling from astrocytes to neurons in cultures of mammalian brain cells. *Science* 263, 1768– 1771.
- Nedergaard, M., Cooper, A. J. L. and Goldman, S. A. (1995) Gap junctions are required for the propagation of spreading depression. J. Neurobiol. 28, 433–444.
- Ness, T. J. and Gebhart, G. F. (1990) Visceral pain: a review of experimental studies. *Pain* **41**, 167–234.
- Neugebauer, V. and Schaible, H. G. (1990) Evidence for a central component in the sensitization of spinal neurons with joint input during development of acute arthritis in cat's knee. J. Neurophysiol. 64, 299–311.
- Neugebauer, V., Schaible, H. G. and Schmidt, R. F. (1989) Senzitization of articular afferents to mechanical stimuli by bradykinin. *Pflüger's Arch.* **415**, 330–335.
- Neugebauer, V., Lucke, T. and Schaible, H. G. (1993) N-Methyl-D-aspartate (NMDA) and non-NMDA receptor antagonists block the hyperexcitability of dorsal horn neurons during development of acute arthritis in rat's knee joint. J. Neurophysiol. 70, 1365–1377.
- Neugebauer, V., Lucke, T. and Schaible, H-G. (1994) Requirement of metabotropic glutamate receptor for the generation of inflammation-evoked hyperexitability in rat spinal cord. *Eur. J. Neurosci.* 6, 1179–1189.
- Neugebauer, V., Weiretter, F. and Schaible, H. G. (1995) Involvement of substance P and neurokinin-1 receptors in the hyperexcitability of dorsal horn neurons during development of acute arthritis in rat's knee joint. J. Neurophysiol. **73**, 1574– 1583.

- Neugebauer, V., Vanegas, H., Nebe, J., Rumenapp, P. and Schaible, H. G. (1996) Effects of N- and L-type calcium channel antagonists on the responses of nociceptive spinal cord neurones to mechanical stimulation of the normal and the inflamed knee joint. J. Neurophysiol. **76**, 3740–3749.
- Neumann, S., Doubell, T. P., Leslie, T. and Woolf, C. J. (1996) Inflammatory pain hypersensitivity mediated by phenotypic switch in myelinated primary sensory neurons. *Nature* 384, 360– 364.
- Newberry, N. R. and Gilbert, M. J. (1989) 5-Hydroxytryptamine evokes three distinct responses in the rat superior cervical ganglion in vitro. *Eur. J. Pharmac.* 162, 197–205.
- Newberry, N. R., Watkins, C. J., Volenec, A. and Flanigan, T. P. (1996) 5-HT_{2B} receptor mRNA in guinea pig superior cervical ganglion. *NeuroReport* 7, 2909–2911.
- Newman, H. M., Stevens, R. T. and Apkarian, A. V. (1996) Direct spinal projections to limbic and striatal areas: anterograde transport studies from the upper cervical spinal cord and the cervical enlargement in squirrel monkey and rat. J. Comp. Neurol. 365, 640–658.
- Newton, A. C. (1997) Regulation of protein kinase C. Curr. Opin. Cell. Biol. 9, 161–167.
- Newton, B., Kuitert, L. M. E., Bergmann, M., Adcock, I. M. and Barnes, P. J. (1997) Evidence for involvement of NF-_KB in the transcriptional control of COX-2 gene expression by IL-1β. *Biochem. Biophys. Res. Commun.* 237, 28–32.
- Nicholas, A. P., Pieribone, V. A., Arvidsson, U. and Hökfelt, T. (1992) Serotonin-, substance P- and glutamate/aspartate immunoreactivities in medullo-spinal pathways in rat and primate. *Neuroscience* 48, 545–559.
- Nicholas, A. P., Hökfelt, T. and Pieribone, V. A. (1996) The distribution and significance of CNS adrenoceptors examined with in situ hybridization. *Trends Pharmac.* 17, 245–254.
- Nicholls, D. and Attwell, D. (1990) The release and uptake of excitatory amino acids. *Trends Pharmac.* 11, 462–468.
- Nichols, M. L., Bian, D., Ossipov, M. H., Lai, J. and Porreca, F. (1995) Regulation of opioid antiallodynic efficacy by cholecystokinin in a model of neuropathic pain in rats. *J. Pharmac. Exp. Ther.* 275, 1339–1345.
- Nichols, M. L., Bian, D., Ossipov, M. H., Malan Jr, T. P. and Porreca, F. (1996) Antiallodynic effects of a CCK_B antagonist in rats with nerve ligation injury: role of endogenous enkephalins. *Neurosci. Lett.* **215**, 161–164.
- Nichols, M. L., Lopez, Y., Ossipov, M. H., Bian, D. and Porreca, F. (1997) Enhancement of the antiallodynic and antinociceptive efficacy of spinal morphine by antisera to dynorphin A (1-13) or MK-801 in a nerve-ligation model of peripheral neuropathy. *Pain* 69, 317–322.
- Nicol, G. D., Klingberg, D. K. and Vasko, M. R. (1992) Prostaglandin E₂ increases calcium conductance and stimulates release of substance P in avian sensory neurons. *J. Neurosci.* 12, 1917–1927.
- Nicol, G. D., Vasko, M. R. and Evans, A. R. (1997) Prostaglandins suppress an outward potassium current in embryonic rat sensory neurons. J. Neurophysiol. 77, 167–176.
- embryonic rat sensory neurons. J. Neurophysiol. 77, 167–176.
 Nicolelis, M. A. L., Lin, R. C. S., Woodward, D. J. and Chapin, J. K. (1993) Induction of immediate spatiotemporal changes in thalamic networks by peripheral block of ascending cutaneous information. Nature 361, 533–536.
- Niedbala, B., Sanchez, A. and Feria, M. (1995) Nitric oxide mediates neuropathic pain behavior in peripheral denervated rats. *Neurosci. Lett.* 188, 57–60.
- Nieto, A. (1996) Molecular biology of axon guidance. *Neuron* 17, 1039–1048.
- Nishihara, I., Minami, T., Watanabe, Y., Ito, S. and Hayashi, O. (1995) Prostaglandin E2 stimulates glutamate release from synaptosomes of rat spinal cord. *Neurosci. Lett.* **196**, 57–60.
- Nishio, E. and Watanabe, Y. (1998) NO induced apoptosis accompanying the change of oncoprotein expression and the activation of CPP32 protease. *Life Sci.* 62, 239–245.
- Nishiyama, K. and Sakuta, M. (1995) Mexiletine for painful alcoholic neuropathy. Int. Med. 34, 577–579.
- Nishiyama, K., Brighton, B. W., Bossut, D. F. and Perl, E. R. (1993) Peripheral nerve injury enhances α_2 -adrenergic receptor expression by some DRG neurons. *Am. Soc. Neurosci.* **19**, 499.
- Noel, M. B. and Gratton, A. (1995) Electrochemical evidence of increased dopamine transmission in prefrontal cortex and nucleus accumbens elicited by ventral tegmental μ-opioid receptor activation in freely behaving rats. Synapse 21, 110–122.
- Nogawa, S., Zhang, F., Ross, M. E. and Iadecola, C. (1997) Cyclo-oxygenase-2 gene expression in neurons contributes to ischemic brain damage. J. Neurosci. 17, 2746–2755.

- Noguchi, K. and Ruda, M. A. (1992) Gene regulation in an ascending nociceptive pathway: inflammation-induced increase in preprotachykinin mRNA in rat lamina I spinal projection neurons. J. Neurosci. 12, 2563–2572.
- Noguchi, K., De Leon, M., Nahin, R. L., Senba, E. and Ruda, M. A. (1993) Quantification of axotomy-induced alteration of neuropeptide mRNAs in dorsal root ganglion neurons with special reference to neuropeptide Y mRNA and the effects of neonatal capsaicin treament. J. Neurosci. Res. 33, 54–66.
- Noguchi, K., Dubner, R., De Leon, M., Senba, E. and Ruda, M. A. (1994) Axotomy induced preprotachykinin gene expression in a subpopulation of dorsal root ganglion neurons. J. Neurosci. Res. 37, 596–603.
- Nordlind, K., Johansson, O., Lidén, S. and Hökfelt, T. (1993) Glutamate- and aspartate-like immunoreactivities in human normal and infimed skin. *Virchows Arch. B. Cell. Path.* 64, 75– 82.
- Nörenberg, W., Cordes, A., Blöhbaum, G., Fröhlich, R. and Illes, P. (1997) Coexistence of purino- and pyrimidinoceptors on activated rat microglial cells. *Br. J. Pharmac.* **121**, 1087–1098.
- North, R. A. (1996) P_{2X} plethora. Semin. Neurosci. 8, 187–194.
- North, R. A. and Barnard, E. A. (1997) Nucleotide receptors. *Curr. Opin. Neurobiol.* **7**, 346–357.
- Nothias, F., Peschanski, M. and Besson, J.-M. (1988) Somatotropic reciprocal connections between the somatosensory cortex and the thalamic PO nucleus in the rat. *Brain Res.* 447, 169–174.
- Nowak, L. M. and MacDonald, R. L. (1982) Substance P: ionic basis for depolarizing responses of mouse spinal cord neurons in cell culture. J. Neurosci. 2, 1119–1128.
- Oaklander, A. L. and Belzberg, A. J. (1997) Unilateral nerve injury down-regulates mRNA for Na⁺ channel SCN10A bilaterally in rat dorsal root ganglia. *Mol. Brain Res.* **52**, 162–165.
- Ogden, D. (1996) Intracellular calcium release in central neurones. *Neuroscience* **8**, 281–291.
- Oh, U., Hwang, S. W. and Kim, D. (1996) Capsaicin activates a nonselective cation channel in cultured neonatal rat dorsal root ganglion neurons. J. Neurosci. 16, 1659–1667.
- O'Halloran, K. D. and Perl, E. R. (1997) Effects of partial nerve injury on the responses of C-fiber polymodal nociceptors to adrenergic agonists. *Brain Res.* **759**, 233–240.
- Oka, T., Oka, K., Hosoi, M. and Hori, T. (1995) Intracerebroventricular injection of interleukin-6 induces thermal hyperalgesia in rats. *Brain Res.* 692, 123–128.
- Oka, T., Hori, T., Hosoi, M., Oka, K., Abe, M. and Kubo, C. (1997) Biphasic modulation in the trigeminal nociceptive neuronal responses by the intracerebroventricular prostaglandin E₂ may be mediated through different EP receptors subtypes in rats. Brain Res. 7711, 278–284.
- Okajima, S., Mizoguchi, A., Tamai, K., Hirasawa, Y. and Ide, C. (1995) Distribution of protein kinase C (α,β,γ subtypes) in normal nerve fibers and in regenerating growth cones of the rat peripheral nervous system. *Neuroscience* **66**, 645–654.
- Oke, A. F., Carver, L. A., Gouvion, C. M. and Adams, R. N. (1997) Three-dimensional mapping of norepinephrine and serotonin in human thalamus. *Brain Res.* **763**, 69–78.
- Oku, R., Nanayama, T. and Satoh, M. (1988) Calcitonin gene-related peptide modulates calcium mobilization in synaptosomes of rat spinal dorsal horn. *Brain Res.* 475, 356–360.
- Olesen, J. and Janssen-Olesen, I. (1997) Novel targets for drug development in migraine. In: *Headache Treatment: Trial Methodology and New Drugs* Vol. 6, pp. 313–320, Eds. J. Olesen and P. Tfelt-Hansen. Lippincott-Raven Publishers, Philadelphia.
- Olesen, J., Iversen, H. K. and Thomsen, L. L. (1993) Nitric oxide supersensitivity: a possible molecular mechanism of migraine pain. *NeuroReport* 4, 1027–1030.
- Oliveira, A. L. R., Risling, M., Deckner, M., Lindholm, T., Langone, F. and Cullheim, S. (1997) Neonatal sciatic nerve transection induces TUNEL labeling of neurons in the rat spinal cord and DRG. *NeuroReport* 8, 2837–2840.
- Olivera, B. M., Miljanich, G. P., Ramachandran, J. and Adams, M. E. (1994) Calcium channel diversity and neurotransmitter release: the ω-conotoxins and ω-agatoxins. A. Rev. Biochem. 63, 823–867.
- Oliveras, J. L. and Montagne-Clavel, J. (1994) The GABA_A receptor antagonist picrotoxin induces a "pain-like" behavior when administered into the thalamic reticular nucleus of the behaving rat: a possible model for "central" pain? *Neurosci. Lett.* **179**, 21–24.
- Oliveras, J. L. and Montagne-Clavel, J. (1996) Picrotoxin produces a "central" pain-like syndrome when microinjected into the somato-motor cortex of the rat. *Physiol. Behav.* 60, 1425–1434.

- Olson, L. (1997) Regeneration in the adult central nervous system: experimental repair strategies. *Nature Med.* **12**, 1329–1335.
- Omana-Zapata, I., Khabbaz, M. A., Hunter, J. C. and Bley, K. R. (1997a) QX-314 inhibits ectopic nerve activity associated with neurpathic pain. *Brain Res.* 771, 228–237.
- Omana-Zapata, I., Khabbaz, M. A., Hunter, J. C., Clarke, D. E. and Bley, K. R. (1997b) Tetrodotoxin inhibits neuropathic ectopic activity in neuromas, dorsal root ganglia and dorsal horn neurons. *Pain* 72, 41–49.
- O'Neill, L. A. and Kaltschmidt, C. (1997) NF k-B: a crucial transcription factor for glial and neuronal cell function. *Trends Neurosci.* 20, 252–258.
- Ossipov, M. H., Lopez, Y., Nichols, M. L., Bian, D. and Porreca, F. (1995) The loss of antinociceptive efficacy of spinal morphine in rats with nerve ligation injury is prevented by reducing spinal afferent drive. *Neurosci. Lett.* **199**, 87–90.
- Ossipov, M. H., Kovelowski, C. J., Wheeker-Aceto, H., Cowan, A., Hunter, J. C., Lai, J., Malan, T. P. and Porreca, F. (1996) Opioid antagonists and antisera to endogenous opioids increase the nociceptive response to formalin: demonstration of an opioid kappa and delta inhibitory tone. J. Pharmac. Exp. Ther. 277, 784–788.
- Ossipov, M. H., Malan, T. P., Lai, J. and Porreca, F. (1997) Opioid pharmacology of acute and chronic pain. In: *The Pharmacology of Pain. Handbook of Experimental Pharmacology*, Vol. 130, pp. 305–334. Eds. A. Dickenson and J-M. Besson. Springer-Verlag, Berlin.
- Otsuka, M. and Yoshioka, K. (1993) Neurotransmitter functions of mammalian tachykinins. *Phys. Rev.* 73, 229–308.
- Ottersen, O. P. and Landsend, A. S. (1997) Organization of glutamate receptors at the synapse. *Eur. J. Neurosci.* 9, 2219–2224.
- Ouseph, A. K. and Levine, J. D. (1995) α_1 -Adrenoceptor-mediated sympathetically dependent mechanical hyperalgesia in the rat. *Eur. J. Pharmac.* **273**, 107–112.
- Owens, C. M., Zhang, D. and Willis, W. D. (1992) Changes in the response states of primate spinothalamic tract cells caused by mechanical damage of the skin or activation of descending controls. J. Neurophysiol. 67, 1509–1527.
- Oyelese, A. A., Rizzo, M. A., Waxman, S. G. and Kocsis, J. D. (1997) Differential effects of NGF and BDNF on axotomyinduced changes in GABA_A-receptor mediated conductance and sodium currents in cutaneous afferent neurons. J. Neurophysiol. 78, 31–42.
- Ozawa, S. (1996) Permeation of calcium through glutamate receptor channels. *Semin. Neurosci.* 8, 261–269.
- Pairet, M. and Engelhardt, G. (1996) Distinct isoforms (COX-1 and COX-2) of cyclooxygenase: possible physiological and therapeutic implications. *Fundam. Clin. Pharmac.* 10, 1–15.
- Palecek, J., Paleckova, V., Doucherty, P. M. and Willis, W. D. (1994) The effect of phorbol esters on the responses of primate spinothalamic neurons to mechanical and thermal stimuli. J. Neurophysiol. 71, 529–537.
- Paleckova, V., Palecek, J., McAdoo, D. J. and Willis, W. D. (1992) The non-NMDA antagonist CNQX prevents release of amino acids into the rat spinal cord dorsal horn evoked by sciatic nerve stimulation. *Neurosci. Lett.* 148, 19–22.
- Pan, W., Zadina, J. E., Harlan, R. E., Weber, J. T., Banks, W. A. and Kastin, A. J. (1997) Tumor necrosis factor-α: a neuromodulator in the CNS. *Neurosci. Biobehav. Rev.* 21, 603–613.
- Pandita, R. K., Persson, K. and Andersson, K.-E. (1997) Capsaicin-induced bladder overactivity and nociceptive behaviour in conscious rats: involvement of spinal nitric oxide. J. Autonomic Nervous System 67, 184–191.
- Paré, D., Dossi, D. C. and Steriade, M. (1991) Three types of inhibitory postsynaptic potentials generated by interneurons in the anterior thalamic complex of cat. J. Neurophysiol. 66, 1190– 1204.
- Parker, E. M., Izzarelli, D. G., Nowak, H. P., Mahle, C. D., Iben, L. G., Wang, J. and Goldstein, M. E. (1995) Cloning and characterization of the rat GALR1 galanin receptor from Rin14B insulinoma cells. *Mol. Brain Res.* 34, 179–189.
- Parker, J. L., Wood, M. L. and Dostrovsky, J. O. (1998) A focal zone of thalamic plasticity. J. Neurosci. 18, 548–558.
- Parnham, M. J. (1996) Inflammation mechanisms and therapeutics. Drugs New Perspectives 9, 631–639.
- Parnham, M. J. (1997) Selective COX-2 inhibitors. Drugs News Perspectives 10, 182–187.
- Parsons, A. M., Honda, C. N., Jia, Y. P., Budai, D., Xu, X. J., Wiesenfeld-Hallin, Z. and Seybold, V. S. (1996) Spinal NK₁ receptors contribute to the increased excitability of the nociceptive flexor reflex during persistent peripheral inflammation. *Brain Res.* **739**, 263–275.

148

- Pasti, L., Volterra, A., Pozzan, T. and Carmignoto, G. (1997) Intracellular calcium oscillations in astrocytes: a highly plastic bidirectional form of communication between neurons and astrocyte in situ. J. Neurosci. 17, 7817-7830.
- Pastoriza, L. N., Morrow, T. J. and Casey, K. L. (1996) Medial frontal cortex lesions selectively attenuate the hot plate response: possible nocifensive apraxia in the rat. Pain 64, 11-17.
- Patterson, P. H. and Nawa, H. (1993) Neuronal differentiation factors/cytokines and synaptic plasticity. Cell 72, 123-137.
- S. L., Grover, L. M., Schwartzkroin, P. A. Patterson, Bothwell, M. (1992) Neurotrophin expression in rat hippocampal slices: a stimulus paradigm inducing LTP in CA1 evokes increases in BDNF and NT-3 mRNAs. Neuron 9, 1081-1088.
- Patterson, S. L., Abel, T., Deuel, T. A. S., Martin, K. C., Rose, J C. and Kandel, E. R. (1996) Recombinant BDNF rescues deficits in basal synaptic transmission and hippocampal LTP in BDNF knockout mice. *Neuron* **16**, 1137–1145. Paul, W. and Page, C. P. (1997) Effect of α_2 -adrenoceptor agonists
- on bradykinin-induced plasma protein extravasation in guineapig skin. Br. J. Pharmac. 120, 172P.
- Payá, M., Pastor, P. G., Coloma, J. and Alcaraz, M. J. (1997) Nitric oxide synthase and cyclo-oxygenase pathways in the inflammatory response induced by zymosan in the rat air pouch. Br. J. Pharmac. 120, 1445-1452
- Pedersen, J. L., Rung, G. W. and Kehlet, H. (1997) Effects of sympathetic nerve block on acute inflammatory pain and hyperalgesia. Anesthesiology 86, 293-301.
- Pedersen-Bjergaard, U., Nielsen, L. B., Jensen, K., Edvinsson, L. Jansen, I. and Olesen, J. (1989) Algesia and local responses induced by neurokinin A and substance P in human skin and temporal muscle. Peptides 10, 1147-1152.
- Pedrosa-Ribeiro, C. M. and Putney, J. W., Jr (1996) Differential effects of protein kinase C activation on calcium storage and capacitative calcium entry in NIH 3T3 cells. J. Biol. Chem. 271, 21522-21528.
- Pehl, U. and Schmid, H. A. (1997) Electrophysiological responses of neurons in the rat spinal cord to nitric oxide. Neuroscience 77, 563-573.
- Peinado, A., Yuste, R. and Katz, L. C. (1993) Gap junctional communication and the development of local circuits in neo-cortex. Cerebral Cortex. **3**, 488–498.
- Pellegrini-Giampietro, D. E., Fan, S., Ault, B., Miller, B. E. and Zukin, R. S. (1994) Glutamate receptor gene expression in spinal cord of arthritic rats. J. Neurosci. 14, 1576-1583
- Pellegrini-Giampietro, D. E., Gorter, J. A., Bennett, M. V. L. and Zukin, R. S. (1997) The GluR2 (GluR-B) hypothesis: Ca²⁺-permeable AMPA receptors in neurological disorders. Trends Neurosci. 20, 464-470.
- Peng, Y. B., Lin, Q. and Willis, W. D. (1996) Effects of GABA and glycine receptor antagonists on the activity and PAGinduced inhibition of rat dorsal horn neurons. Brain Res. 736, 189 - 201.
- Perkins, D. J. and Kniss, D. A. (1997) Tumor necrosis factor-a promotes sustained cyclooxygenase-2 expression: attenuation by dexamethasone and NSAIDs. Prostaglandins 54, 727-743.
- Perkins, M. N. and Dray, A. (1996) Novel pharmacological strat-
- egies for analgesia. Ann. Rheumatic Dis. 55, 715–722. Perkins, M. N. and Kelly, D. (1994) Interleukin-1 beta-induced desArg⁹bradykinin-mediated hyperalgesia in rat. Neuropharmacology 33, 657-660.
- Perl, E. R. (1994) A re-evaluation of mechanisms leading to sympathetically related pain. In: Pharmacological Approaches to the Treatment of Chronic Pain: New Concepts and Critical Issues pp. 129-150. Eds. H. L. Fields and J. C. Liebeskind. IASP Press, Seattle.
- Perry, V. H., Brown, M. C. and Andersson, P. B. (1993) Macrophage response to central and peripheral nerve injury. Adv. Neurol. 5, 309-314.
- Pertovaara, A., Wei, H. and Hämäläinen, M. M. (1996) Lidocaine in the rostroventromedial medulla and the periaqueductal gray attenuates allodynia in neuropathic rats. Neurosci. Lett. 218, 127 - 130.
- Pertovaara, A., Hämäläinen, M. M., Kauppila, T. and Panula, F (1998) Carrageenan-induced changes in spinal nociception and its modulation by the brain stem. *NeuroReport* **9**, 351–355.
- Petersen, M., Zhang, J., Zhang, J. M. and LaMotte, R. H. (1996) Abnormal spontaneous activity and responses to norepinephrine in dissociated dorsal root ganglion cells after chronic nerve con striction. Pain 67, 391-397.
- Petersen, M., Erkert, A. S., von Blanchet, G. S., Heppelmann, B., Klusch, A. and Kniffki, K. D. (1998a) Plasticity in the expression of bradykinin binding sites in sensory neurons after mechanical nerve injury. Neuroscience 83, 949-959.

- Petersen, M., von Banchet, G. S., Heppelmann, B. and Koltzenburg, M. (1998b) Nerve growth factor regulates the expression of bradykinin binding sites on adult sensory neurons via the neurotrophin receptor p75. Neuroscience 83, 161-168.
- Pettit, M. J. and Schwark, H. D. (1996) Capsaicin-induced rapid receptive field reorganization in cuneate neurons. *Neurophysiol.* **75**, 1117–1125.
- Petty, B. J., Cornblath, D. R., Adornato, B. T., Chaudhry, V., Flexner, C., Wachsman, M., Sinicropi, D., Burton, L. E. and Peroutka, S. J. (1994) The effect of systematically administered recombinant human nerve growth factor in healthy human subjects. Ann. Neurol. 36, 244-246.
- Pfrieger, F. W. and Barres, B. A. (1997) Synaptic efficacy enhanced by glial cells in vitro. Science 277, 1684-1687.
- Phebus, L. A., Johnson, K. W., Audia, J. E., Cohen, M. L., Dressman, B. A., Fritz, J. E., Kaldor, S. W., Krushinski, J. H., Schenck, K. W., Zgombick, J. M., Branchek, T. A., Adham, N. and Schaus, J. M. (1996) Characterization of LY 334,370, a potent and selective 5-HT_{1F} receptor agonist, in the neurogenic dural inflammation model of migraine pain. Am. Soc. Neurosci. 22. 1331
- Philippi, M., Vyklicky, L., Kuffler, D. P. and Orkand, R. K. (1995) Serotonin- and proton-induced and modified ionic currents in frog sensory neurones. J. Neurosci. Res. 40, 387-395.
- Piani, D., Frei, K., Do, K. Q., Cuenod, M. and Fontana, A. (1991) Murine brain macrophages induce NMDA receptor mediated neurotoxicity in vitro by secreting glutamate. Neurosci. Lett. 133, 159.
- Picard, P. R., Tramèr, M. R., McQuay, H. J. and Moore, R. A. (1997) Analgesic efficacy of peripheral opioids (all except intraarticular): a qualitative systematic review of randomised controlled trials. Pain 72, 309-318.
- Pierce, P. A., Xie, G. X., Levine, J. D. and Peroutka, S. J. (1996) 5-hydroxytryptamine receptor subtype messenger RNAs in rat peripheral sensory and sympathetic ganglia: a polymerase chain reaction study. Neuroscience 70, 553-559.
- Pierce, P. A., Xie, G-X., Meuser, T. and Peroutka, S. J. (1997) 5-Hydroxytryptamine receptor subtype messenger RNAs in human dorsal root ganglia: a polymerase chain reaction study. Neuroscience 81, 813-819.
- Pieribone, V. A., Nicholas, A. P., Dagerlind, A. and Hokfelt, T. (1994) Distribution of alpha 1 adrenoceptors in rat brain revealed by in situ hybridization experiments utilizing subtype specific probes. J. Neurosci. 14, 4252-4268.
- Pin, J. P. and Duvoisin, R. (1995) Neurotransmitter receptors. I. The metabotropic glutamate receptors: structure and functions. Neuropharmacology 34, 1-26.
- Planas, B., Kolb, P. E., Raskind, M. A. and Miller, M. A. (1997) Nerve growth factor induces galanin gene expression in the rat basal forebrain: implications for the treatment of cholinergic dysfunction. J. Comp. Neurol. 379, 563-570.
- Plenz, D. and Kitai, S. T. (1996) Generation of high-frequency oscillations in local circuits of rat somatosensory cortex cultures. J. Neurophysiol. 76, 4180–4184.
- Polistina, D. C., Murray, M. and Goldberger, M. E. (1990) Plasticity of dorsal root and descending serotoninergic projec tions after partial deafferentation of the adult rat spinal cord. J. Comp. Neurol. 299, 349-363.
- Pollock, R., Kerr, R. and Maxwell, D. J. (1997) An immunocytochemical investigation of the relationship between substance P and the neurokinin-1 receptor in the lateral horn of the rat thoracic spinal cord. Brain Res. 777, 22-30.
- Pons, T. P., Garraghty, P. E., Ommaya, A. K., Kaas, J. H., Taub, E. and Mishkin, M. (1991) Massive cortical reorganization after sensory deafferentation in adult macaques. Science 252, 1857-1860.
- Poole, S., Cunha, F. Q., Selkirk, S., Lorenzetti, B. B. and Ferreira, S. H. (1995) Cytokine mediated inflammatory hyperalgesia lim-
- b. II. (1993) Cytokin Incuracy Impartment Internation in protogena Impartment in the protocol of the protection of the state of the sta
- Portenov, R. K., Folev, K. M. and Inturrisi, C. (1990) The nature of opioid responsiveness and its implications for neuropathic pain: new hypotheses derived from studies of opioid infusions. Pain 43, 273-286.
- Prado, G. N., Taylor, L. and Polgar, P. (1997) Effects of intracellular tyrosine residue mutation and carboxyl terminus truncation on signal transduction and internalization of the rat bradykinin B2 receptor. J. Biol. Chem. 272, 14638-14642.
- Presley, R. W., Menétrey, D., Levine, J. D. and Basbaum, A. I. (1990) Systemic morphine suppresses noxious stimulus-evoked

fos protein-like immunoreactivity in the rat spinal cord. J. Neurophysiol. 10, 323–335.

- Price, D. D., Hayes, R. L., Ruda, M. A. and Dubner, R. (1978) Spatial and temporal tranformations of input to spinothalamic tract neurons and their relation to somatic sensation. J. Neurophysiol. 41, 933–946.
- Price, D. D., Mao, J. and Mayer, D. J. (1995) Brain neural images of persistent pain states in rats using 2-DG autoradiography. In: *Forebrain Areas Involved in Pain Processing*, pp. 143–154. Eds. J. M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Procacci, P. and Zoppi, M. (1983) Pathophysiolology and clinical aspects of visceral and referred pain. In: *Proceedings of the Third World Congress on Pain*, pp. 643–658. Eds. J. J. Bonica, U. Lindblom and A. Iggo. Raven Press, New York.
- Procacci, P., Zoppi, M. and Maresca, M. (1994) Heart and vascular pain. In: *Textbook of Pain*, 3rd edn, pp. 541–554. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Pubols, L. M. and Bowen, D. C. (1988) Lack of central sprouting of primary afferent fibres after ricin deafferentation. J. Comp. Neurol. 275, 282–287.
- Püschel, A. W. (1996) The semaphorins: a family of axonal guidance molecules? *Eur. J. Neurosci.* 8, 1317–1321.
- Püschel, A. W., Adams, R. H. and Betz, H. (1996) The sensory innervation of the mouse spinal cord may be patterned by differential expression of and differential responsiveness to semaphorins. *Mol. Cell. Neurosci.* 7, 419–431.
- Puttick, R. M. (1992) Excitatory action of prostaglandin E₂ on rat neonatal cultured dorsal root ganglion cells. *Br. J. Pharmac.* 105, 133.
- Puttick, R. M., Pinnock, R. D. and Woodruff, G. N. (1994) Galanin-induced membrane depolarization of neonatal rat cultured dorsal root gangalion cells. *Eur. J. Pharmac.* 254, 303– 306.
- Qian, J., Brown, S. D. and Carlton, S. M. (1996a) Systemic ketamine attenuates nociceptive behaviors in a rat model of peripheral neuropathy. *Brain Res.* 715, 51–62.
- Qian, Y., Chao, D. S., Santillano, D. R., Cornwell, T. L., Nairn, A. C., Greengard, P., Lincoln, T. M. and Bredt, D. S. (1996b) cGMP-dependent protein kinase in dorsal root ganglion: relationship with nitric oxide synthase and nociceptive neurons. J. Neurosci. 16, 3130–3138.
- Quartara, L. and Maggi, C. A. (1997) The tachykinin NK₁ receptor. Part I: ligands and mechanisms of cellular activation. *Neuropeptides* **31**, 537–563.
- Radhakrishnan, V. and Henry, J. L. (1997) Electrophysiological evidence that neurokinin A acts via NK-1 receptors in the cat dorsal horn. *Eur. J. Pharmac.* 9, 1977–1985.
- Radhakrishnan, V., Yashpal, K., Hui-Chan, C. W. and Henry, J. L. (1995) Implications of a nitric oxide synthase mechanism in the action of substance P. L-NAME blocks thermal hyperalgesia induced by endogenous and exogenous subtance P in the rat. *Eur. J. Pharmac.* 7, 1920–1925.
- Raffa, R. B., Schupsky, J. J., Lee, D. K. H. and Jacoby, H. I. (1996) Characterization of endothelin-induced nociception in mice: evidence for a mechanistically distinct analgesic model. J. Pharmac. Exp. Ther. 278, 1–7.
- Rainville, P., Duncan, G. H., Price, D. D., Carrier, B. and Bushnell, M. C. (1997) Pain affect encoded in human anterior cingulate but not somatosensory cortex. *Science* 277, 968–971.
- Raivich, G., Hellweg, R. and Kreusberg, G. W. (1991) NGF receptor-mediated reduction in axonal NGF uptake and retrograde transport following sciatic nerve injury and during regeneration. *Neuron* 7, 151–164.
 Raja, S. N. (1995) Role of the sympathetic nervous system in
- Raja, S. N. (1995) Role of the sympathetic nervous system in acute pain and inflammation. Ann. Med. 27, 241–246.
- Raja, S. N., Campbell, J. N. and Meyer, R. A. (1984) Evidence for different mechanisms of primary and secondary hyperalgesia following heat injury to the glabrous skin. *Brain* 107, 1179– 1188.
- Raja, S. N., Meyer, R. A. and Campbell, J. N. (1988) Peripheral mechanisms of somatic pain. *Anesthesiology* 68, 571–590.
- Raja, S. N., Treede, R. D., Davis, K. D. and Campbell, J. N. (1991) Systemic alpha-adrenergic blockade with phentolamine: a diagnostic test for sympathetically maintained pain. *Anesthesiology* 74, 691–698.
- Raja, S. N., Turnquist, J. L., Melecka, S. and Campbell, J. N. (1996) Monitoring adequacy of α-adrenoceptor blockade following systemic phentolamine administration. *Pain* 64, 197–204.
- Ralston, H. J. and Ralston, D. D. (1992) The primate dorsal spinothalamic tract: evidence for a specific termination in the posterior nuclei (Po/SG) of the thalamus. *Pain* 48, 107–118.

- Ralston, H. J., Ohara, P. T. Meng, X. W. and Ralston, D. D. (1995) The organization of spinothalamic tract circuitry in the macaque and the role of GABA information processing. In: *Forebrain Areas Involved in Pain Processing*, pp. 1–13. Eds. J-M. Besson, G. Guilbaud and H. Ollat. John Libbey Eurotext, Paris.
- Ramakers, G. M. J., Passinelli, P., Hens, J. J. H., Gispen, W. H. and De Graan, P. N. E. (1997) Protein kinase C in synaptic plasticity: changes in the in situ phosphorylation state of identified pre-and postsynaptic substrates. *Prog. Neuro-Psychopharmacol. Biol. Psychiat.* 21, 455–486.
- Ramer, M. S. and Bisby, M. A. (1997a) Rapid sprouting of sympathetic axons in dorsal root ganglia of rats with a chronic constriction injury. *Pain* 70, 237–244.
- Ramer, M. S. and Bisby, M. A. (1997b) Reduced sympathetic sprouting occurs in dorsal root ganglia after axotomy in mice lacking low-affinity neurotrophin receptor. *Neurosci. Lett.* 228, 9–12.
- Ramer, M. S., French, G. D. and Bisby, M. A. (1997) Wallerian degeneration is required for both neuropathic pain and sympathetic sprouting into the DRG. *Pain* 72, 71–78.
- Randic, M., Jiang, M. C. and Cerne, R. (1993) Long-term potentiation and long-term depression of primary afferent neurotransmission in the rat spinal cord. J. Neurosci. 13, 5228–5241.
- Rang, H. P., Bevan, S. J. and Dray, A. (1991) Chemical activation of nociceptive peripheral neurones. Br. Med. Bull. 47, 534–548.
- Rao, M. S., Sun, Y., Escary, J. L., Perreau, J., Tresser, S., Patterson, P. H., Zigmond, R. E., Brulet, P. and Landis, S. C. (1993) Leukemia inhibitory factor mediates an injury response not a target-directed developmental transmitter switch in sympathetic neurons. *Neuron* 11, 1175–1185.
- Rasband, M. N., Trimmer, J. S., Schwarz, T. L., Levinson, S. R., Ellisman, M. H., Schachner, M. and Shrager, P. (1998) Potassium channel distribution, clustering and function in remeylinating rat axons. J. Neurosci. 18, 36–47.
- Rash, J. E., Dillman, R. K., Bilhartz, B. L., Duffy, H. S., Whalen, L. R. and Yasumura, T. (1996) Mixed synapses discovered and mapped throughout mammalian spinal cord. *Proc. natl Acad. Sci. U.S.A.* 93, 4235–4239.
- Rausell, E., Bae, C. S., Viñuela, A., Huntley, G. W. and Jones, E. G. (1992a) Calbindin and parvalbumin cells in monkey VPL thalamic nucleus: distribution, laminar cortical projections and relations to spinothalamic terminations. *J. Neurophysiol.* 12, 4088–4111.
- Rawlings, S. R. and Hezareh, M. (1996) Pituitary adenylate cyclase-activating polypeptide (PACAP) and PACAP/vasoactive intestinal polypeptide receptors: actions on the anterior pituitary gland. *Endocrine Rev.* 17, 4–29.
- Raymond, P., Décarie, A., Lantin, F., Raut, R., Morais, R., Couture, R. and Adam, A. (1996) A role for B₁ and B₂ kinin receptors in the modulation of T-kininogen during the acute phase response of inflammation. *Peptides* 17, 1163–1170.
- Razzaque, Z., Shaw, D., Smith, D., Hopkins, R., Sirinathsinghji, D., Maskell, L., Stanton, J. A., Beer, M. S., Hill, R. G., Pickard, J. and Longmore, J. (1997) Pharmacological analysis of 5-HT receptor-mediated vasoconstriction of human middle meningeal arteries: determining the contribution of 5-HT_{1B} and 5-HT_{1F} receptor activation. *Br. J. Pharmac.* **120**, 211P.
- Rebeck, G. W., Maynard, K. I., Hyman, B. T. and Moskowitz, M. A. (1994) Selective 5-HT_{1Dz} serotonin receptor gene expression in trigeminal ganglia: implications for antimigraine drug development. *Proc. natl Acad. Sci. U.S.A.* 91, 3666–3669.
- drug development. Proc. natl Acad. Sci. U.S.A. 91, 3666–3669. Redburn, J. L. and Leah, J. D. (1997) Accelerated breakdown and enhanced expression of c-Fos in the rat brain after noxious stimulation. Neurosci. Lett. 237, 97–100.
- Redford, E. J., Hall, S. M. and Smith, K. J. (1995) Vascular changes and demyelination induced by the intraneural injection of tumour necrosis factor. *Brain* 118, 869–878.
- Reeh, P. W. and Kress, M. (1995) Effects of classical algogens. Semin. Neurosci. 7, 221–226.
- Reeh, P. W., Kocher, L. and Jung, S. (1986) Does neurogenic inflammation alter the sensitivity of unmyelinated nociceptors in the rat? *Brain Res.* 384, 42–50.
- Reeh, P. W., Bayer, J., Kocher, L. and Handwerker, H. O. (1987) Sensitization of nociceptive cutaneous nerve fibers from the rat tail by noxious mechanical stimulation. *Exp. Brain Res.* 65, 505–512.
- Rees, H. and Roberts, M. H. T. (1993) The anterior pretectal nucleus: a proposed role in sensory processing. *Pain* 53, 121–135.
- Rees, H., Sluka, K. A., Westlund, K. N. and Willis, W. D. (1995a) The role of glutamate and GABA receptors in the gen-

eration of dorsal root reflexes by acute arthritis in the anaesthetised rat. J. Physiol. **484**, 437–445.

- Rees, H., Terenzi, M. G. and Roberts, M. H. T. (1995b) Anterior pretectal nucleus facilitation of superficial dorsal horn neurones and modulation of deafferentation pain in the rat. J. Physiol. 489, 159–169.
- Rees, H., Sluka, K. A., Lu, Y., Westlund, K. N. and Willis, W. D. (1996) Dorsal root reflexes in articular afferents occur bilaterally in a chronic model of arthritis in rats. J. Neurophysiol. 76, 4190–4193.
- Reeve, A. J. and Dickenson, A. H. (1995) The roles of spinal adenosine receptors in the control of acute and more persistent nociceptive responses of dorsal horn neurones in the anesthetized rat. Br. J. Pharmac. 116, 2221–2228.
- Reichling, D. B. and Levine, J. D. (1997) Heat tranduction in rat sensory neurons by calcium-dependent activation of a cation channel. *Proc. natl Acad. Sci. U.S.A.* 94, 7006–7011.
- Reichling, D. B. and MacDermott, A. B. (1996) NMDA receptormediated calcium entry in the absence of AMPA receptor activation in rat dorsal horn neurons. *Neurosci. Lett.* 204, 17–20.
- Reichling, D. B., Barratt, L. and Levine, J. D. (1997) Heatinduced cobalt entry: an assay for heat transduction in cultured rat dorsal root ganglion neurons. *Neuroscience* 77, 291–294.
- Reinhardt, R. R., Chin, E., Zhang, B., Roth, R. A. and Bondy, C. A. (1994) Selective coexpression of insulin receptor—related receptor (IRR) and TRK in NGF-sensitive neurons. J. Neurosci. 14, 4674–4683.
- Ren, K. and Dubner, R. (1996) Enhanced descending modulation of nociception in rats with persistent hindpaw inflammation. J. *Neurophysiol.* 76, 3025–3037.
- Ren, K., Thomas, D. A. and Dubner, R. (1995) Nerve growth factor alleviates a painful peripheral neuropathy in rats. *Brain Res.* 699, 286–292.
- Ren, K., Iadarola, M. J. and Dubner, R. (1996) An isobolographic analysis of the effects of *N*-methyl-D-aspartate and NK₁ tachykinin receptor antagonists on inflammatory hyperalgesia in the rat. *Br. J. Pharmac.* **117**, 196–202.
- Renz, H., Gong, J. H., Schmidt, A., Nain, M. and Gemsa, D. (1995) Release of tumor necrosis factor- α from macrophages: enhancement and suppression are dose-dependently regulated by prostaglandin E₂ and cyclic nucleotides. *J. Immun.* **141**, 2388–2393.
- Rexed, B. (1952) The cytoarchitectonic organization of the spinal cord in the rat. J. Comp. Neurol. 96, 415–466.
- Riaz, S. S. and Tomlinson, D. R. (1997) Stimulation of nerve growth factor levels in control and diabetic rats by treatment with clenbuterol. Br. J. Pharmac. 122, 291P.
- Ribeiro-Da-Sylva, S. and Cuello, A. C. (1990) Choline acetyltransferase-immunoreactive profiles are presynaptic to primary sensory fibers in the rat superficial dorsal horn. J. Comp. Neurol. 295, 370–384.
- Riccio, A., Pierchala, B. A., Ciarallo, C. L. and Ginty, D. D. (1997) An NGF-TrkA-mediated retrograde signal to transcription factor CREB in sympathetic neurons. *Science* 277, 1097– 1100
- Rice, A. S. C. (1995a) The relative effects of analgesic therapy administered before, during or after surgical injury. *Semin. Neurosci.* 7, 245–251.
- Rice, A. S. C. (1995b) Topical spinal administration of a nitric oxide synthase inhibitor prevents the hyper-reflexia associated with a rat model of persistent pain. *Neurosci. Lett.* 187, 111– 114.
- Rich, K. M., Luszczynski, J. R., Osborne, P. A. and Johnson Jr, E. M. (1987) Nerve growth factor protects adult sensory neurons from cell death and atrophy caused by nerve injury. J. Neurocytol. 16, 261–268.
- Richardson, J. D., Aanonsen, L. and Hargreaves, K. M. (1998) Hypoactivity of the spinal cannabinoid system results in NMDA-dependent hyperalgesia. J. Neurosci. 18, 451–457.
- Ridet, J. L., Rajaofetra, N., Teilhac, J. R., Geffard, M. and Privat, A. (1993) Evidence for nonsynaptic serotonergic and noradrenergic innervation of the rat dorsal horn and possible involvement of neuron–glia interactions. *Neuroscience* 52, 143– 157.
- Ridet, J. L., Malhotra, S. K., Privat, A. and Gage, F. H. (1997) Reactive astrocytes: cellular and molecular cues to biological function. *Trends Neurosci.* 20, 570–577.
- Riley, R. C., Zhao, Z. Q. and Duggan, A. W. (1996) Spinal release of immunoreactive dynorphin $A_{(1-8)}$ with the development of peripheral inflammation in the rat. *Brain Res.* **710**, 131–142.
- Rizzo, M. A., Kocsis, J. S. D. and Waxman, S. G. (1994) Slow sodium conductances of dorsal root ganglion neurons: intra-

neuronal homogeneity and interneuronal heterogeneity. J. Neurophysiol. 72, 2796–2815.

- Ro, L. S. and Jacobs, J. M. (1993) The role of the saphenous nerve in experimental sciatic nerve mononeuropathy produced by loose ligatures: a behavioural study. *Pain* 52, 359–369.
- Ro, L. S., Chen, S. T., Tang, L. M. and Chang, H. S. (1996) Local application of anti-NGF blocks the collateral sprouting in rats following chronic constriction injury of the sciatic nerve. *Neuroscience* 218, 87–90.
- Robello, M., Amico, C. and Cupello, A. (1993) Regulation of GABA_A receptor in cerebellar granule cells in culture: differential involvement of kinase activities. *Neuroscience* 53, 131–138.
- Robello, M., Amico, C., Bucossi, G., Cupello, A., Rapallino, M. V. and Thellung, S. (1996) Nitric oxide and GABA_A receptor function in the rat cerebral cortex and cerebellar granule cells. *Neuroscience* 74, 99–105.
- Robello, M., Amico, C. and Cupello, A. (1997) A dual mechanism of impairment of GABA_A receptor activity by NMDA receptor activation in rat cerebellum granule cells. *Eur. Biophys. J.* 25, 181–187.
- Roberts, W. A., Eaton, S. A. and Salt, T. E. (1992) Widely distributed GABA-mediated afferent inhibition processes within the ventrobasal thalamus of rat and their possible relevance to pathological pain states and somatotopic plasticity. *Exp. Brain Res.* 89, 363–372.
- Roberts, W. J. and Elardo, S. M. (1985) Sympathetic activation of A-delta nociceptors. *Somatosens. Res.* **3**, 33–44.
- Robertson, S. J., Rae, M. G., Rowan, E. G. and Kennedy, C. (1996) Characterization of a P_{2x} purinoceptor in cultured neurones of the rat dorsal root ganglia. *Br. J. Pharmac.* **118**, 951–957.
- Robitaille, R. (1995) Purinergic receptors and their activation by endogenous purines at perisynaptic glial cells of the frog neuromuscular junction. J. Neurosci. 15, 7121–7131.
- Rocamora, N., Welker, E., Pascual, M. and Soriano, E. (1996) Upregulation of BDNF mRNA expression in the barrel cortex of adult mice after sensory stimulation. J. Neurosci. 16, 4411– 4419.
- Roche, A. K., Cook, M., Wilcox, G. L. and Kajander, K. C. (1996) A nitric oxide synthesis inhibitor (L-NAME) reduces licking behavior and Fos-labeling in the spinal cord of rats during formalin-induced inflammation. *Pain* 66, 331–341.
- Rodriguez-Alvarez, J., Lafon-Cazal, M., Blanco, I. and Bockaert, J. (1997) Different routes of Ca²⁺ influx in NMDA-mediated generation of nitric oxide and arachidonic acid. *Eur. J. Neurosci.* 9, 867–870.
- Romano, J. M. and Turner, J. A. (1985) Chronic pain and depression: does the evidence support a relationship. *Psychol. Bull.* 97, 18–34.
- Romita, V. V. and Henry, J. L. (1996) NMDA receptor involvement in spinal inhibitory controls of nociception in the rat. *NeuroReport* 7, 1705–1708.
- Rondé, P. and Nichols, R. A. (1997) 5-HT₃ receptors induce rises in cytosolic and nuclear calcium in NG108-15 cells via calciuminduced calcium release. *Cell Calcium* 22, 265–357.
- Rostas, J. A. P., Brent, V. A., Voss, K., Errington, M. L., Bliss, T. V. P. and Gurd, J. W. (1996) Enhanced tyrosine phosphorylation of the 2B subunit of the *N*-methyl-D-aspartate receptor in long-term potentiation. *Proc. natl Acad. Sci. U.S.A.* 93, 10452– 10456.
- Rothwell, N. J. (1997) Neuroimmune interactions: the role of cytokines. Br. J. Pharmac. 121, 841–847.
- Rozenburt, E. (1985) Convergent signaling in the action of integrins, neuropeptides, growth factors and oncogenes. *Cancer Surveys* 24, 81–96.
- Rubin, G., Kaspi, T., Rappaport, Z. H., Cohen, S., Ravikovitch, M., Lomazov, P. and Devor, M. (1997) Adrenosensitivity of injured afferent neurons does not require the presence of postganglionic sympathetic terminals. *Pain* 72, 183–191.
- Rubio, N. (1997) Mouse astrocytes store and deliver brain-derived neurotrophic factor using the non-catalytic gp95^{trkb} receptor. *Eur. J. Neurosci.* 9, 1847–1853.
- Ruda, M. A., Bennett, G. J. and Dubner, R. (1986) Neurochemistry and neural circuitry in the dorsal horn. *Prog. Brain Res.* 66, 219–268.
- Rudomin, P. (1990) Presynaptic inhibition of muscle spindle and tendon organ afferents in the mammalian spinal cord. *Trends Neurosci.* 13, 499–505.
- Rueff, A. and Dray, A. (1992) 5-Hydroxytryptamine-induced sensitization and activation of peripheral fibres in the neonatal rat are mediated via different 5-hydroxytryptamine receptors. *Neuroscience* 50, 899–905.

- Rueff, A. and Mendell, L. M. (1996) Nerve growth factor and NT-5 induce increased thermal sensitivity of cutaneous nociceptors in vitro. J. Neurophysiol. 76, 3593–3596.
- Rueff, A., Dawson, A. J. L. R. and Mendell, L. M. (1996) Characteristics of nerve growth factor induced hyperalgesia in adult rats: dependence on enhanced bradykinin-1 receptor activity but not neurokinin-1 receptor activation. *Pain* 66, 359– 372.
- Ruoff, G. E. (1996) Depression in the patient with chronic pain. J. Fam. Pract. 43, S25–S34.
- Rupniak, N. M. J., Carlson, E., Boyce, S., Webb, J. K. and Hill, R. G. (1996) Enantioselective inhibition of the formalin paw late phase by the NK₁ receptor antagonist L-733 060 in gerbils. *Pain* 67, 189–195.
- Rupniak, N. M. J., Boyce, S., Webb, J. K., Williams, A. R., Carlson, E. J., Hill, R. G., Borkowski, J. A. and Hess, J. F. (1997) Effects of bradykinin B_1 receptor antagonist des-Arg⁹[Leu⁸]bradykinin and genetic disruption of the B_2 receptor on nociception in rats and mice. *Pain* **71**, 89–97.
- Rush, A. M. and Elliott, J. R. (1997) Phenytoin and carbamazepine: differential inhibition of sodium currents in small cells from adult rat dorsal root ganglia. *Neurosci. Lett.* 226, 95–98.
- Rusin, K. I., Ryu, P. D. and Randic, M. (1992) Modulation of excitatory amino acid responses in rat dorsal horn neurons by tachykinins. J. Neurophysiol. 68, 265–286.
- Rusin, K. I., Bleakman, D., Chard, P. S., Randic, M. and Miller, R. J. (1993) Tachykinins potentiate N-methyl-D-aspartate responses in acutely isolated neurons from the dorsal horn. J. Neurochem. 60, 952–960.
- Rutherford, L. C., DeWan, A., Lauer, H. M. and Turrigiano, G. G. (1997) Brain-derived neurotrophic factor mediates the activity-dependent regulation of inhibition in neocortical cultures. J. Neurosci. 17, 4527–4535.
- Rydh-Rinder, M., Holmberg, K., Elfvin, L. G., Wiesenfeld-Hallin, Z. and Hökfelt, T. (1996) Effects of peripheral axotomy on neuropeptides and nitric oxide synthase in dorsal root ganglia and spinal cord of the guinea pig: an immunohistochemical study. *Brain Res.* 707, 180–188.
- Sabol, S. L. and Higuchi, H. (1990) Transcriptional regulation of the neuropeptide Y gene by nerve growth factor: antagonism by adenosine 3',5'-monophosphate and phorbol esters. *Mol. Endocrinol.* 4, 384–392.
- Safieh-Garabedian, B., Poole, S., Allchorne, A., Winter, J. and Woolf, C. J. (1995) Contribution of interleukin-1β to the inflammation-induced increase in nerve growth factor levels and inflammatory hyperalgesia. *Br. J. Pharmac.* **115**, 1265–1275.
- Safieh-Garabedian, B., Poole, S., Allchorne, A., Kanaan, S., Saade, N. and Woolf, C. J. (1996) Zinc reduces the hyperalgesia and upregulation of NGF and IL-1β produced by peripheral inflammation in the rat. *Neuropharmacol.* 35, 599–603.
- Safieh-Garabedian, B., Kanaan, S. A., Haddad, J. J., Abou Jaoude, P., Jabbur, S. J. and Saadé, N. E. (1997) Involvement of interleukin-1 β , nerve growth factor and prostaglandin E₂ in endotoxin-induced localized inflammatory hyperalgesia. *Br. J. Pharmac.* **121**, 1619–1626.
- Sakai, N., Yamada, M., Numakawa, T., Ogura, A. and Hatanaka, H. (1997) BDNF potentiates spontaneous Ca²⁺ oscillations in cultured hippocampal neurons. *Brain Res.* **778**, 318–328.
- Sakai, Y., Nishijima, Y., Mikuni, N. and Iwata, N. (1979) An experimental model of hyper-irritability in the trigeminal skin field of the rat. *Pain* 7, 147–157.
- Sakatani, K., Chesler, M., Hassan, A. Z., Lee, M. and Young, W. (1993) Non-synaptic modulation of dorsal column conduction by endogenous GABA in neonatal rat spinal cord. *Brain Res.* 622, 43–50.
- Salamone, S., Morel, N. and Godfraind, T. (1997) Role of nitric oxide in the contractile response to 5-hydroxytryptamine of the basilar artery from Wistar Kyoto and stroke-prone rats. Br. J. Pharmac. 121, 1051–1058.
- Saleh, T. M. (1997) Visceral afferent stimulation-evoked changes in the release of peptides into the parabrachial nucleus in vivo. *Brain Res.* 778, 56–63.
- Salt, T. E. and Eaton, S. A. (1995) Modulation of sensory neurone excitatory and inhibitory responses in the ventrobasal thalamus by activation of metabotropic excitatory amino acid receptors. *Neuropharmacology* 34, 1043–1051.
- Salt, T. E. and Eaton, S. A. (1996) Functions of ionotropic and metabotropic glutamate receptors in sensory transmission in the mammalian thalamus. *Progr. Neurobiol.* 48, 55–72.
- Salt, T. E. and Hill, R. G. (1983) Neurotransmitter candidates of somatosensory primary afferent fibres. *Neuroscience* 10, 1083– 1103.

- Salt, T. E., Meier, C. L., Seno, N., Krucker, T. and Herrling, P. L. (1995) Thalamocortical and corticocortical excitatory postsynaptic potentials mediated by excitatory amino acid receptors in the cat motor cortex in vivo. *Neuroscience* 64, 433–442.
- Salter, M. W. and Henry, J. L. (1987) Evidence that adenosine mediates the depression of spinal dorsal horn neurones induced by peripheral vibration in the cat. *Neuroscience* 22, 631–650.
- Salter, M. W., De Koninck, Y. and Henry, J. L. (1993) Physiological roles for adenosine and ATP in synaptic transmission in the spinal dorsal horn. *Prog. Neurobiol.* 41, 125–156.
- Salvemini, D., Settle, S. L., Masferrer, J. L., Seibert, K., Currie, M. G. and Needleman, P. (1995) Regulation of prostaglandin production by nitric oxide: an in vivo analysis. *Br. J. Pharmac.* 114, 1171–1178.
- Salvemini, D., Seibert, K. and Marino, M. H. (1996) New concepts in inflammation and therapy. *Drug News Persp.* 4, 204– 219.
- Samdani, A. F., Newcamp, C., Resink, A., Facchinetti, F., Hoffman, B. E., Dawson, V. L. and Dawson, T. M. (1997) Differential susceptibility to neurotoxicity mediated by neurotrophins and neuronal nitric oxide synthase. J. Neurosci. 17, 4633– 4641.
- Sánchez-Prieto, J., Budd, D. C., Herrero, I., Vásquez, E. and Nicholls, D. G. (1996) Presynaptic receptors and the control of glutamate exocytosis. *Trends Neurosci.* 19, 235–239.
- Sandkühler, J. (1996) The organization and function of endogenous antinociceptive systems. Prog. Neurobiol. 50, 49–81.
- Sandkühler, J., Chen, J. G., Cheng, G. and Randic, M. (1997) Low-frequency stimulation of afferent $A\delta$ -fibers induces longterm depression at primary afferent synapses with substantia gelatinosa neurons in the rat. J. Neurosci. **17**, 6483–6491.
- Sang, C. N., Gracely, R. H., Max, M. B. and Bennett, G. J. (1996) Capsaicin-evoked mechanical allodynia and hyperalgesia cross nerve territories. Evidence for a central mechanism. *Anesthesiology* 85, 491–496.
- Sang, C. N., Hostetter, M., Gracely, R., Chappell, A., Iyengar, S., Lee, G., Caruso, R., Whitcup, S. and Max, B. (1997) Analgesic effects of AMPA/kainate antagonist LY 293558 in human: reduction of capsaicin evoked allodynia and hyperalgesia but not sensations in normal skin. *Am. Soc. Neurosci.* 23, 1015.Sangameswaran, L., Delgado, S. G., Fish, L. M., Koch, B. D.,
- Sangameswaran, L., Delgado, S. G., Fish, L. M., Koch, B. D., Jakeman, L. B., Stewart, G. R., Sze, P., Hunter, J. C., Eglen, R. M. and Herman, R. C. (1996) Structure and function of a novel voltage-gated, tetrodotoxin-resistant sodium channel specific to sensory neurons. J. Biol. Chem. 271, 5953–5956.
- Sangameswaran, L., Fish, L. M., Koch, B. D., Rabert, D. K., Delgado, S. G., Ilnicka, M., Jakeman, L. B., Novakovic, S., Wong, K., Sze, P., Tzoumaka, E., Stewart, G. R., Herman, R. C., Chan, H., Eglen, R. M. and Hunter, J. C. (1997) A novel tetrodotoxin-sensitive, voltage-gated sodium channel expressed in rat and human dorsal root ganglia. J. Biol. Chem. 272, 14805–14809.
- Sann, H., Dux, M., Schemann, M. and Jancso, G. (1996) Neurogenic inflammation in the gastrointestinal tract of the rat. *Neurosci. Lett.* 219, 147–150.
- Sansum, A. J., Chessell, I. P. and Humphrey, P. P. A. (1997) Characterisation of purinoceptors mediating responses to ATP on rat dorsal roots. *Br. J. Pharmac.* **120**, 128P.
- Saria, A., Javorsky, F., Humpel, C. and Gamse, R. (1990) 5-HT receptor antagonists inhibit sensory neuropeptide release from the rat spinal cord. *NeuroReport* 1, 104–106.
- Sasaki, T., Yasuda, H., Maeda, K. and Kikkawa, R. (1998) Hyperalgesia and decreased neuronal nitric oxide synthase in diabetic rats. *NeuroReport* 9, 243–247.
- Sato, J. and Perl, E. R. (1991) Adrenergic excitation of cutaneous pain receptors induced by peripheral nerve injury. *Science* 251, 1608–1610.
- Sato, J., Suzuki, S., Iseki, T. and Kumazawa, T. (1993a) Adrenergic excitation of cutaneous nociceptors in chronically inflamed rats. *Neurosci. Lett.* 164, 225–228.
- Sato, J., Suzuki, S., Tamura, R. and Kumazawa, T. (1994) Norepinephrine excitation of cutaneous nociceptors in adjuvantinduced inflamed rats does not depend on sympathetic neurons. *Neurosci. Lett.* **177**, 135–138.
- Neurosci. Lett. 177, 135–138. Sato, K., Kiyama, H., Park, H. P. and Tohyama, M. (1993b) AMPA, KA and NMDA receptor are expressed in the rat DRG neurons. NeuroReport 4, 1263–1265.
- Sato, T., Shigematsu, S. and Arita, M. (1995) Mexiletine-induced shortening of the action potential duration of ventricular muscle by activation of ATP-sensitive K⁺ channels. Br. J. Pharmac. 115, 381–382.
- Satoh, M., Kuraishi, Y. and Kawamura, M. (1992) Effects of intrathecal antibodies to substance-P, calcitonin gene-related

peptide and galanin on repeated cold stess-induced hyperalgesia: comparison with carrageenan-induced hyperalgesia. *Pain* **49**, 273–278.

- Satoh, O. and Omote, K. (1996) Roles of monoaminergic, glycinergic and GABAergic inhibitory systems in the spinal cord in rats with peripheral mononeuropathy. *Brain Res.* 728, 27–36.Sawai, H., Clarke, D. B., Kittlerova, P., Bray, G. M. and Aguayo,
- Sawai, H., Clarke, D. B., Kitterova, P., Bray, G. M. and Aguayo, A. J. (1996) Brain-derived neurotrophic factor and NT-4/5 stimulate the growth of axonal branches from regenerating retinal ganglion cells. J. Neurosci. 16, 3887–3894.
- Sawynok, J. and Reid, A. (1996) Neurotoxin-induced lesions to central serotonergic, noradrenergic and dopaminergic systems modify caffeine-induced antinociception in the formalin test and locomotor stimulation in rats. J. Pharmac. Exp. Ther. 277, 646– 653.
- Sawynok, J. and Reid, A. (1997) Peripheral adenosine 5'-triphosphate enhances nociception in the formalin test via activation of a purinergic P_{2x} receptor. *Eur. J. Pharmac.* 330, 115–121.
 Sawynok, J. and Sweeney, M. I. (1989) The role of purines in
- Sawynok, J. and Sweeney, M. I. (1989) The role of purines in nociception. *Neuroscience* 32, 557–569.
- Sawynok, J., Zarrindast, M-R., Reid, A. R. and Doak, G. J. (1997) Adenosine A₃ receptor activation produces nociceptive behaviour and edema by release of histamine and 5-hydroxytryptamine. *Eur. J. Pharmac.* 333, 1–7.Saxena, P. R. and Ferrari, M. D. (1989) 5-HT₁-like receptor ago-
- Saxena, P. R. and Ferrari, M. D. (1989) 5-HT₁-like receptor agonists and the pathophysiology of migraine. *Trends Pharmac.* 10, 200–204.
- Schaden, H., Stürmer, C. A. O. and Bähr, L. (1994) GAP-43 immunoreactivity and axon regeneration in retinal ganglion cells of the rat. J. Neurobiol. 25, 1570–1578.
- Schäfer, M., Carter, L. and Stein, C. (1994) Interleukin 1β and corticotropin-releasing factor inhibit pain by releasing opioids from immune cells in inflamed tissue. *Proc. natl Acad. Sci.* U.S.A. 91, 4219–4223.
- Schäfer, M., Mousa, S. A. and Stein, C. (1997) Corticotropinreleasing factor in antinociception and inflammation. *Eur. J. Pharmac.* 323, 1–10.
- Schäfer, M., Zhou, L. and Stein, C. (1998) Cholecystokinin inhibits peripheral opioid analgesia in inflamed tissue. *Neuroscience* 82, 603–611.
- Schaffar, N., Rao, H., Kessler, J-P. and Jean, A. (1997) Immunohistochemical detection of glutamate in rat vagal sensory neurons. *Brain Res.* 778, 302–308.
- Schaible, H-G. and Grubb, B. D. (1993) Afferent and spinal mechanism of joint pain. *Pain* 55, 5–54.
 Schaible, H-G. and Schmidt, R. F. (1988) Time course of mechan-
- Schaible, H-G. and Schmidt, R. F. (1988) Time course of mechanosensitivity changes in articular afferents during a developing experimental arthritis. J. Neurophysiol. 60, 2180–2195.
- Schaible, H-G., Schmidt, R. F. and Willis, W. D. (1987) Convergent inputs from articular, cutaneous and muscle receptors onto ascending tract cells in the cat spinal cord. *Exp. Brain Res.* 66, 479–488.
- Schaible, H-G., Neugebauer, V., Cervero, F. and Schmidt, R. F. (1991) Changes in tonic descending inhibition of spinal neurons with articular input during the development of acute arthritis in the cat. J. Neurophysiol. 66, 1021–1032.
- Schaible, H. G., Hope, P. J., Lang, C. W. and Duggan, A. W. (1992) Calcitonin gene-related peptide causes intraspinal spreading of substance P released by peripheral stimulation. *Eur. J. Neurosci.* 4, 750–757.
- Schaller, K. L., Krzemien, D. M., Yarowsky, P. J., Krueger, B. K. and Caldwell, J. H. (1995) A novel, abundant sodium channel expressed in neurons and glia. J. Neurosci. 15, 3231–3242.
- Schepelmann, K., Meblinger, K., Schaible, H. G. and Schmidt, R. F. (1992) Inflammatory mediators and nociception in the joint: excitation and sensitization of slowly conducting afferent fibers of cat's knee by prostaglandin I₂. *Neuroscience* **50**, 237–247.
- Scheuren, N., Neupert, W., Ionac, M., Neuhuber, W., Brune, K. and Geisslinger, G. (1997) Peripheral noxious stimulation releases spinal PGE₂ during the first phase in the formalin assay of the rat. *Life Sci.* **60**, 295–300.
- Schild, J. H. and Kunze, D. L. (1997) Experimental and modeling study of Na⁺ current heterogeneity in rat nodose neurons and its impact on neuronal discharge. J. Neurophysiol. 78, 3198– 3209.
- Schmahmann, J. D. and Pandya, D. N. (1990) Anatomical investigation of projections from the posterior parietal cortex in the rhesus monkey: a WGA-HRP and fluorescent tracer study. J. Comp. Neurol. 295, 299–326.
- Schmelz, M., Luz, O., Averbeck, B. and Bickel, A. (1997a) Plasma extravasation and neuropeptide release in human skin as measured by intradermal microdialysis. *Neurosci. Lett.* 230, 117–120.

- Schmelz, M., Schmidt, R., Bickel, A., Handwerker, H. O. and Torebjörk, H. E. (1997b) Specific C-receptors for itch in human skin. J. Neurosci. 17, 8003–8008.
- Schmidt, R., Schmelz, M., Forster, C., Ringkamp, M., Torebjörk, E. and Handwerker, H. (1995) Novel classes of responsive and unresponsive C nociceptors in human skin. J. Neurosci. 15, 333–341.
- Schmidt, R., Schmelz, M., Ringkamp, M., Handwerker, H. O. and Torebjork, H. E. (1997) Innervation territories of mechanically activated C nociceptor units in human skin. J. Neurophysiol. 78, 2641–26248.
- Schmidt, R. F. (1971) Pre-synaptic inhibition in the vertebrate nervous systems. *Rev. Physiol. Biochem. Pharmac.* 63, 21–101.Schmidt-Kastner, R., Wetmore, C. and Olson, L. (1996)
- Schmidt-Kastner, R., Wetmore, C. and Olson, L. (1996) Comparative study of brain-derived neurotrophic factor messenger RNA and protein at the cellular level suggests multiple roles in hippocampus, striatum and cortex. *Neuroscience* 74, 161–183.
- Schott, G. D. (1995) An unsympathetic view of pain. Lancet 345, 634-636.
- Schreyer, D. J. and Skene, J. H. P. (1991) Fate of GAP-43 in ascending spinal axons of DRG neurons after peripheral nerve injury: delayed accumulation and correlation with regenerative potential. J. Neurosci. 11, 3738–3751.
- Schulz, P. E. and Fitzgibbons, J. C. (1997) Differing mechanisms of expression for short- and long-term potentiation. J. Neurophysiol. 78, 321–334.
- Schwab, M. E. (1990) Myelin-associated inhibitors of neurite growth and regeneration in the CNS. *Trends Neurosci.* 13, 452– 456.
- Schwab, M. E. (1996) Molecules inhibiting neurite growth: a minireview. Neurochem. Res. 21, 755–761.
- Seabrook, G. R., Bowery, B. J., Heavens, R., Brown, N., Ford, H., Sirinathsinghi, D. J. S., Borkowski, J. A., Hess, J. F., Strader, C. D. and Hill, R. G. (1997) Expression of B₁ and B₂ bradykinin receptor mRNA and their funtional roles in sympathetic ganglia and sensory dorsal root ganglia neurones from wild-type and B₂ receptor knockout mice. *Neuropharmacology* **36**, 1009–1017.
- Sebert, M. E. and Shooter, E. M. (1993) Expression of messenger RNA for neurotrophic factors and their receptors in the rat dorsal root ganglion after sciatic nerve injury. *J. Neurosci. Res.* 36, 357–367.
- Seguin, L. and Millan, M. J. (1994) The glycine B receptor partial agonist, (+)-HA 966, enhances induction of antinociception by RP 67580 and CP-99,994. *Eur. J. Pharmac.* 253, R1–R3.
- Seguin, L., Le Marouille-Girardon, S. and Millan, M. J. (1995) Antinociceptive profiles of non-peptidergic neurokinin₁ and neurokinin₂ receptor antagonists: a comparison to other classes of antinociceptive agent. *Pain* **61**, 325–343.
- Seibert, K., Zhinag, Y., Leahy, K., Hauser, S. and Masferrer, J. (1994) Pharmacological and biochemical demonstration of the role of cyclooxygenase 2 in inflammation and pain. *Proc. natl Acad. Sci. U.S.A.* 91, 12013–12017.
- Seltzer, Z. (1995) The relevance of animal neuropathy models for chronic pain in humans. Semin. Neurosci. 7, 211–219.
- Seltzer, Z. and Devor, M. (1979) Ephatic transmission in chronically damaged peripheral nerves. *Neurology* **29**, 1061–1064.
- Seltzer, Z., Cihin, S., Ginzburg, R. and Beilin, B. (1991) Modulation of neuropathic pain behavior in rats by spinal disinhibition and NMDA receptor blockade of injury discharge. *Pain* 45, 69–75.
- Selve, N., Englberger, W., Friderichs, E., Hennies, H-H., Reimann, W. and Wilffert, B. (1996) Galanin receptor antagonists attenuate spinal antinociceptive effects of DAMGO, tramadol and non-opioid drugs in rats. *Brain Res.* **735**, 177–187.
- Sengupta, J. N. and Gebhart, G. F. (1994) Characterization of mechanosensitive pelvic nerve afferent fibres innervating the colon of the rat. J. Neurophysiol. 71, 2046–2060.
- Sharma, G. and Stevens, C. F. (1996) A mutation that alters magnesium block of *N*-methyl-D-aspartate receptor channels. *Proc. natl Acad. Sci. U.S.A.* 93, 9259–9263.
- Shaw, P. J. and Salt, T. E. (1997) Modulation of sensory and excitatory amino acid responses by nitric oxide donors and glutathione in the ventrobasal thalamus of the rat. *Eur. J. Neurosci.* 9, 1507–1513.
- Sheardown, M. J., Shannon, H. E., Swedbergn, M. D. B., Suzdak, P. D., Bymaster, F. P., Olesen, P. H., Mith, C. H., Ward, J. S. and Sauerberg, P. (1997) M₁ receptor agonist activity is not a requirement for muscarinic antinociception. *J. Pharmac. Exp. Ther.* 281, 868–875.
- Sheen, K. and Chung, J. M. (1993) Signs of neuropathic pain depend on signals from injured nerve fibers in a rat model. *Brain Res.* **610**, 62–68.

- Shehab, S. A. and Atkinson, M. E. (1986a) Vasoactive intestinal peptide increases in areas of the dorsal horn of the spinal cord in which other neuropeptides are depleted following peripheral axotomy. *Exp. Brain Res.* 62, 422–430.
- Shehab, S. A. and Atkinson, M. E. (1986b) Vasoactive intestinal polypeptide (VIP) increases in the spinal cord after peripheral axotomy of the sciatic nerve originate from primary sensory neurons. *Brain Res.* 372, 37-44.
- Shen, K. Z. and North, R. A. (1992) Substance P opens cation channels and closes potassium channels in rat locus coerulus neurons. *Neuroscience* 50, 345–353.
- Shen, Y., Li, R. and Shiosaki, K. (1997) Inhibition of P^{75} tumor necrosis factor receptor by antisense oligonucleotides increases hypoxic injury and β -amyloid toxicity in human neuronal cell line. *J. Biol. Chem.* **272**, 3550–3553.
- Shepherd, S. L., Williamson, D. J., Beer, M. S., Hill, R. G. and Hargreaves, R. J. (1997) Differential Effects of 5-HT_{1B/1D} receptor agonists on neurogenic dural plasma extravasation and vasodilation in anaesthetized rats. *Neuropharmacology* 36, 525– 533.
- Sherman, S. E. and Loomis, C. W. (1996) Strychnine-sensitive modulation is selective for non-noxious somatosensory input in the spinal cord. *Pain* 66, 321–330.
- Sherman, S. E., Luo, L. and Dostrovsky, J. O. (1997a) Altered receptive fields and sensory modalities of rat VPL thalamic neurons during spinal strychnine-induced allodynia. J. Neurophysiol. 78, 2296–2308.
- Sherman, S. E., Luo, L. and Dostrovsky, J. O. (1997b) Spinal strychnine alters response properties of nociceptive-specific neurons in rat medial thalamus. J. Neurophysiol. 78, 628–637.
- Sherman, S. M. and Guillery, R. W. (1996) Functional organization of thalamocortical relays. J. Neurophysiol. 76, 1367–1395.
- Sherwood, N. T., Lesser, S. S. and Lo, D. C. (1997) Neurotrophin regulation of ionic currents and cell size depends on cell context. *Proc. natl Acad. Sci. U.S.A.* 94, 5917–5922.
- Shi, H-C., Won, C-K., Jung, S-C., Oh, S., Park, S. and Sohn, J-H. (1997) Interhemispheric modulation of sensory transmission in the primary somatosensory cortex of rats. *Neurosci. Lett.* 230, 137–139.
- Shigeri, Y. and Fujimoto, M. (1993) Neuropeptide Y stimulates DNA synthesis in vascular smooth muscle cells. *Neurosci. Lett.* 149, 19–22.
- Shimoyama, N., Shimoyama, M., Davis, A. M., Inturrisi, C. E. and Elliott, K. J. (1997) Spinal gabapentin is antinociceptive in the rat formalin test. *Neurosci. Lett.* 221, 65–67.
- Shin, H-C., Oh, S. J., Jung, S-C., Park, J. and Won, C-K. (1997) Differential modulation of short and long latency sensory responses in the SI cortex by IL-6. *NeuroReport* 8, 2841–2844.
- Shinder, V. and Devor, M. (1994) Structural basis of neuron-toneuron cross-excitation in dorsal root ganglia. J. Neurocytol. 23, 15–531.
- Shir, Y. and Seltzer, Z. (1990) A-fibers mediate mechanical hyperesthesia and allodynia and C-fibers mediate thermal hyperalgesia in a new model of causalgiform pain disorders in rats. *Neurosci. Lett.* 115, 62–67.
- Shortland, P. and Fitzgerald, M. (1994) Neonatal sciatic nerve section results in a rearrangement of the central terminals of saphenous and axotomized sciatic nerve afferents in the dorsal horn of the spinal cord of the adult rat. *Eur. J. Neurosci.* 6, 75–86.
- Shortland, P. and Woolf, C. J. (1993) Chronic peripheral nerve section results in a rearrangement of the central axonal arborizations of axotomized $A\beta$ primary afferent neurons in the rat spinal cord. J. Comp. Neurol. **330**, 65–82.
- Shortland, P., Molander, M., Fitzgerald, M. and Woolf, C. J. (1990) Neonatal capsaicin treatment induces invasion of the substantia gelatinosa by the arborizations of HFA's in the rat dorsal horn. J. Comp. Neurol. 296, 23–31.
- Silver, R., Silverman, A. J., Vitkovic, L. and Lederhendler, I. I. (1996) Mast cells in the brain: evidence and functional significance. *Trends Neurosci.* 19, 25–31.
- Simmons, D. R., Spike, R. C. and Todd, A. J. (1995) Galanin is contained in GABAergic neurons in the rat spinal dorsal horn. *Neurosci. Lett.* 187, 119–122.
- Simon, J., Kidd, E. J., Smith, F. M., Chessell, I. P., Murrell-Lagnado, R., Humphrey, P. P. A. and Barnard, E. A. (1997) Localization and functional expression of splice variants of the P2X₂ receptor. *Mol. Pharmac.* **52**, 237–248.
- Simone, D. A. and Kajander, K. C. (1997) Responses of cutaneous A-fiber nociceptors to noxious cold. J. Neurophysiol. 77, 2049– 2060.
- Simone, D. A., Baumann, T. K., Collins, J. G. and LaMotte, R. H. (1989a) Sensitization of cat dorsal horn neurons to innocu-

ous mechanical stimulation after intradermal injection of capsaicin. Brain Res. 486, 185–189.

- Simone, D. A., Baumann, T. K. and LaMotte, R. H. (1989b) Dose-dependant pain and mechanical hyperalgesia in humans after intradermal injection of capsaicin. *Pain* 38, 99–107.
- Simone, D. A., Sorkin, L. S., Oh, U., Chung, J. M., Owens, C., LaMotte, R. H. and Willis, W. D. (1991) Neurogenic hyperalgesia: central neural correlates in responses of spinothalamic tract neurons. J. Neurophysiol. 66, 228–246.
- Simpson, P. B., Challiss, R. A. J. and Nahorski, S. R. (1995) Neuronal Ca²⁺ stores: activation and function. *Trends Neurosci.* 18, 299–306.
- Siuciak, J. A., Altar, C. A., Wiegand, S. J. and Lindsay, R. M. (1994) Antinociceptive effect of brain-derived neurotrophic factor and neurotrophin-3. *Brain Res.* 633, 326–330.
- Siuciak, J. A., Wong, V., Pearsall, D., Wiegand, S. J. and Lindsay, R. M. (1995) BDNF produces analgesia in the formalin test and modifies neuropeptide levels in rat brain and spinal cord areas associated with nociception. *Eur. J. Neurosci.* 7, 663–670.
- Sivilotti, L. and Woolf, C. J. (1994) The contribution of GABA and glycine receptors to central sensitization: disinhibition and touch-evoked allodynia in the spinal cord. J. Neurophysiol. 72, 179.
- Sjaastad, M. D. and Nelson, W. J. (1997) Integrin-mediated calcium signaling and regulation of cell adhesion by intracellular calcium. *BioEssays* 19, 47–55.
- Sjölund, K. F., Sollevi, A., Segerdahl, M., Hansson, P. and Lundeberg, T. (1996) Intrathecal and systemic R-phenylisopropyladenosine reduces scratching behaviour in a rat mononeuropathy model. *NeuroReport* 7, 1856–1860.
- Sjölund, K. F., von Heijne, M., Hao, J. X., Xu, X. J., Sollevi, A. and Wiesenfeld-Hallin, Z. (1998) Intrathecal administration of the adenosine A₁ receptor agonist R-phenylisopropyl adenosine reduces presumed pain behaviour in a rat model of central pain. *Neurosci. Lett.* 243, 89–92.
- Skilling, S. R., Smullin, D. H. and Larson, A. A. H. (1990) Differential effects of C- and N terminal substance P metabolites on the release of amino acid neurotransmitters from the spinal cord: potential role in nociception. *Pain* 42, 93–101.
- Skinner, K., Basbaum, A. I. and Fields, H. L. (1997) Cholecystokinin and enkephalin in brain stem pain modulating circuits. *NeuroReport* 8, 2995–2998.
- Skofitsch, G., Zamir, N., Helke, C. J., Savitt, J. M. and Jacobowitz, D. M. (1985) Corticotropin-releasing factor-like immunoreactivity in sensory ganglia and capsaicin-sensitive neurons of the rat central nervous system: colocalization with other neuropeptides. *Peptides* 6, 307.
- Sloan, S. I., Stucky, C. L., Jansen, E. M., Galeazza, M. T., Seybold, V. S. and Aanonsen, L. (1991) Changes in PCP binding sites in rat spinal cord in an experimental model of acute, peripheral inflammation. *Am. Soc. Neurosci.* 17, 435.
- Sluka, K. A. (1997) Blockade of calcium channels can prevent the onset of secondary hyperalgesia and allodynia induced by intradermal injection of capsaicin in rats. *Pain* **71**, 157–164.
- Sluka, K. A. and Westlund, K. N. (1993) Behavioral and immunohistochemical changes in an experimental arthritis model in rats. *Pain* 55, 367–377.
- Sluka, K. A. and Willis, W. D. (1997) The effects of G-protein and protein kinase inhibitors on the behavioral responses of rats to intradermal injection of capsaicin. *Pain* **71**, 165–178.
- Sluka, K. A., Willis, W. D. and Westlund, K. N. (1993) Joint inflammation and hyperalgesia are reduced by spinal bicuculline. *NeuroReport* 5, 109–112.
- Sluka, K. A., Willis, W. D. and Westlund, K. N. (1994a) Central control of peripheral joint inflammation and heat hyperalgesia. In: Proceedings of the 7th World Congress on Pain. Progress in Pain Research and Management, 2nd edn, pp. 169–186. Eds. G. F. Gebhart, D. L. Hammond and T. S. Jensen. IASP Press, Seattle.
- Sluka, K. A., Willis, W. D. and Westlund, K. N. (1994b) Inflammation-induced release of excitatory amino acids is prevented by spinal administration of a GABA_A but not GABA_B receptor antagonist in rats. J. Pharmac. Exp. Ther. 271, 76–82.
- Sluka, K. A., Rees, H., Westlund, K. N. and Willis, W. D. (1995) Fiber types contributing to dorsal root reflexes induced by joint inflammation in cats and monkeys. J. Neurophysiol. 74, 981– 989.
- Sluka, K. A., Milton, M. A., Willis, W. D. and Westlund, K. N. (1997a) Differential roles of neurokinin 1 and neurokinin 2 receptors in the development and maintenance of heat hyperalgesia induced by acute inflammation. *Br. J. Pharmac.* 120, 1263–1273.

- Sluka, K. A., Rees, H., Chen, P-S., Tsuruoka, M. and Willis, W. D. (1997b) Capsaicin-induced sensitization of primate spinothalamic tract cells is prevented by a protein kinase C inhibitor. *Brain Res.* 772, 82–86.
- Smeyne, R., Klein, R., Schnapp, A., Long, L. K., Bryant, S., Lewin, A., Lira, S. A. and Barbacid, M. (1994) Severe sensory and sympathetic neuropathies in mice carrying a disrupted Trk/ NGF receptor gene. *Nature* 368, 246–249.
- Smith, C. H., Barker, J., Morris, R. W., MacDonald, D. M. and Lee, T. H. (1993) Neuropeptides induce rapid expression of endothelial cell adhesion molecules and elicit infiltration in human skin. J. Immun. 151, 3274–3282.
- Smith, D. J., Hawranko, A. A., Monroe, P. J., Gully, D., Urban, M. O., Craig, C. R., Smith, J. P. and Smith, D. L. (1997a) Dose-dependent pain-facilitory and -inhibitory actions of neurotensin are revealed by SR 48692, a nonpeptide neurotensin antagonist: influence on the antinociceptive effect of morphine. J. Pharmac. Exp. Ther. 282, 899–908.
- Smith, G. D., Harrison, S. M., Birch, P. J., Elliott, P. J., Malcangio, M. and Bowery, N. G. (1994a) Increased sensibility to the antinociceptive activity of (±) baclofen in an animal model of chronic neuropathic, but not chronic inflammatory hyperalgesia. *Neuropharmacology* 33, 1103–1108.
- Smith, G. D., Wiseman, J., Harrison, S. M., Elliott, P. J. and Birch, P. J. (1994b) Pre-treatment with MK-801, a non-competitive NMDA antagonist, prevents development of mechanical hyperalgesia in a rat model of chronic neuropathy, but not in a model of chronic inflammation. *Neurosci. Lett.* 165, 79–83.
- Smith, G. M., Berry, R. L., Yang, J. and Tanelian, D. (1997b) Electrophysiological analysis of dorsal root ganglion neurons pre- and post-coexpression of green fluorescent protein and functional 5-HT₃ receptor. J. Neurophysiol. 77, 3115–3121.
- Smith, M. A., Zhang, L-X., Lyons, W. E. and Mamounas, L. A. (1997c) Anterograde transport of endogenous brain-derived neurotrophic factor in hippocampal mossy fibers. *NeuroReport* 8, 1829–1834.
- Smith, S. J. (1994) Neuromodulatory astrocytes. Curr. Biol. 4, 807–810.
- Smith, W. L. and De Witt, D. L. (1996) Prostaglandin endoperoxide H synthases-1 and -2. Adv. Immun. 62, 167–215.
- Smith, W. L., Garavito, R. M. and De Witt, D. L. (1996) Prostaglandin endoperoxide H synthases (cyclooxygenases)-1 and -2. J. Biol. Chem. 271, 33157–33160.
- Smullin, D. H., Skilling, R. and Larson, A. A. (1990) Interactions between substance P, calcitonin gene-related peptide, taurine and excitatory amino acids in the spinal cord. *Pain* 42, 93–101.
- Snow, P. J. and Wilson, P. (1991) Plasticity in the somatosensory system of developing and mature mammals. The effects of injury to the central and peripheral nervous system. *Prog. Sensory Physiol.* 11, 118–132.
- Soares, J. C. and Mann, J. J. (1997) The anatomy of mood disorders. Review of structural neuroimaging studies. *Biol. Psychiat.* 41, 86–106.
- Sollevi, A., Belfrage, M., Lundeberg, T., Segerdahl, M. and Hansson, P. (1993) Systemic adenosine infusion: a new treatment modality to alleviate neuropathic pain. *Pain* 61, 155–158.
- Solum, D., Hughes, D., Major, M. S. and Parks, T. N. (1997) Prevention of normally occurring and deafferentation-induced neuronal death in chick brainstem auditory neurons by periodic blockade of AMPA/kainate receptors. J. Neurosci. 17, 4744– 4751.
- Sommervaille, T., Reynolds, M. L. and Woolf, C. J. (1991) Timedependent differences in the increase in GAP-43 expression in dorsal root ganglion cells after peripheral axotomy. *Neuroscience* 45, 213–220.
- Song, H-J., Ming, G-L. and Poo, M-M. (1997) cAMP-induced switching in turning direction of nerve growth cones. *Nature* 388, 275–279.
- Sontheimer, H., Black, J. A. and Waxman, S. G. (1996) Voltagegated Na⁺ channels in glia: properties and possible functions. *Trends Neurosci.* 19, 325–331.
- Sorkin, L. S. (1993) NMDA evokes an L-NAME sensitive spinal release of glutamate and citrulline. *NeuroReport* **4**, 479–482.
- Sorkin, L. S., McAdoo, D. J. and Willis, W. D. (1993) Raphe magnus stimulation-induced antinociception in the cat is associated with release of amino acids as well as serotonin in the lumbar dorsal horn. *Brain Res.* **618**, 95–108.
- Sorkin, L. S. and Puig, S. (1996) Neuronal model of tactile allodynia produced by spinal strychnine: effects of excitatory amino acid receptor antagonists and a μ-opiate receptor agonist. *Pain* 68, 283–292.

- Sorkin, L. S., Xiao, W-H., Wagner, R. and Myers, R. R. (1997) Tumour necrosis factor α induces ectopic activity in nociceptive primary afferent fibres. *Neuroscience* 81, 255–262.
- Sotgiu, M. L. and Biella, G. (1997) Role of input from saphenous afferents in altered spinal processing of noxious signals that follows sciatic nerve constriction in rat. *Neurosci. Lett.* 223, 101– 104.
- Sotgiu, M. L., Biella, G. and Lacerenza, M. (1996) Injured nerve block alters adjacent nerves spinal interaction in neuropathic rats. *NeuroReport* 7, 1385–1388.
- Southall, A., Vernallis, A. B., Heath, J. K. and Thompson, S. W. N. (1996) LIF induces GAL but not VIP-immunoreactivity in small diameter rat sensory neurons in vivo. *Am. Soc. Neurosci.* 22, 769–776.
- Spanswick, D., Pickering, A. E., Gibson, I. C. and Logan, S. D. (1995) Excitation of sympathetic preganglionic neurons via metabotropic excitatory amino acid receptors. *Neuroscience* 68, 1247–1261.
- Stamford, J. A. (1995) Descending control of pain. Br. J. Anaesthesia. 75, 217–227.
- Stanfa, L. C., Dickenson, A., Xu, X-J. and Weisenfeld-Hallin, Z. (1994) Cholecystokinin and morphine analgesia. *Trends Pharmac.* 15, 65–66.
- Stanfa, L. C., Misra, C. and Dickenson, A. H. (1996) Amplification of spinal nociceptive transmission depends on the generation of nitric oxide in normal and carrageenan rats. *Brain Res.* **737**, 92–98.
- Stanfa, L. C., Singh, L., Williams, R. G. and Dickenson, A. (1997) Gabapentin, ineffective in normal rats, markedly reduces C-fibre evoked responses after inflammation. *NeuroReport* 8, 587–590.
- Stanley, E. F. (1997) The calcium channel and the organization of the presynaptic transmitter release face. *Trends Neurosci.* 20, 404-409.
- Stanley, M., Berger, R. J., Zuccarello, M. and Keller, J. T. (1993) Serotonin (5-HT) fibers of the rat dura mater: 5-HT-positive, but not authentic serotoninergic, tryptophan hydroxylase-like fibers. *Neurosci Lett.* **162**, 89–92.
- Steen, K. H. and Reeh, P. W. (1993) Sustained graded pain and hyperalgesia from harmless experimental tissue acidosis in human skin. *Neurosci. Lett.* **154**, 113–116.
- Steen, K. H., Reeh, P. W., Anton, F. and Handwerker, H. O. (1992) Protons selectively induce lasting excitation and sensitization to mechanical stimulation of nociceptors in rat skin, *in vitro. J. Neurosci.* 12, 86–99.
- Steen, K. H., Steen, A. E. and Reeh, P. W. (1995) A dominant role of acid pH in inflammatory excitation and sensitization of nociceptors in rat skin, in vitro. J. Neurosci. 15, 3982–3989.
- Steen, K. H., Steen, A. E., Kreysel, H-W. and Reeh, P. W. (1996) Inflammatory mediators potentiate pain induced by experimental tissue acidosis. *Pain* 66, 163–170.
- Stefani, A., Pisani, A., Mercuri, N. B. and Calabresi, P. (1996) The modulation of calcium currents by the activation of mGluRs. Functional implications. *Mol. Neurobiol.* 13, 81–95.
- Stein, C. (1993) Peripheral mechanisms of opioid analgesia. Anesth. Analg. 76, 182–191.
- Stein, C. and Yassouridis, A. (1997) Peripheral morphine analgesia. Pain 71, 119–121.
- Stein, C., Millan, M. J., Shippenberg, T. S., Peter, K. and Herz, A. (1989) Peripheral opioid receptors mediating antinociception: evidence for involvement of mu-, delta- and kappa- receptors. J. Pharmac. Exp. Ther. 248, 1269–1275.
- Pharmac. Exp. Ther. 248, 1269–1275.
 Steinhäuser, C. and Gallo, V. (1996) News on glutamate receptors in glials cells. *Trends Neurosci.* 19, 339–345.
- Stensaas, L. J., Partlow, L. M., Burgess, P. R. and Horch, K. W. (1987) Inhibition of regeneration: the ultrastructure of reactive astrocytes and abortive axon terminals in the transitional zone of the dorsal root. *Prog. Brain Res.* **71**, 457–467.
- Steriade, M. (1998) Dynamic properties of corticothalamic neurons and local cortical interneurons generating fast rhythmic (30-40 Hz) spike bursts. J. Neurophysiol. 79, 483–490.
- Sterne, G. D., Brown, R. A., Green, C. J. and Terenghi, G. (1997) Neurotrophin-3 delivered locally via fibronectin mats enhances peripheral nerve regeneration. *Eur. J. Neurosci.* 9, 1388–1396.
- Stevens, C. F., Tonegawa, S. and Wang, Y. (1994) The role of calcium-calmodulin kinase II in three forms of synaptic plasticity. *Curr. Biol.* 4, 687–693.
- Stiller, R. U., Grubb, B. D. and Schaible, H. G. (1993) Neurophysiological evidence for increased kappa opioidergic control of spinal cord neurons in rats with unilateral inflammation at the ankle. *Eur. J. Neurosci.* 5, 1520–1527.
- Strack, S., Wadzinski, B. E. and Ebner, F. F. (1996) Localization of the calcium/calmodulin-dependent protein phosphatase, calci-

neurin, in the hindbrain and spinal cord of the rat. J. Comp. Neurol. 375, 66–76.

- Strittmatter, S. M., Fankhauser, C., Huang, P. L., Mashimo, H. and Fishman, M. C. (1995) Neuronal pathfinding is abnormal in mice lacking the neuronal growth cone protein GAP-43. *Cell* 80, 445–452.
- Stucky, C. L., Thayer, S. A. and Seybold, V. S. (1996) Prostaglandin E_2 increases the proportion of neonatal rat dorsal root ganglion neurons that respond to bradykinin. *Neuroscience* **74**, 1111–1123.
- Study, R. E. and Kral, M. G. (1996) Spontaneous action potential activity in isolated dorsal root ganglion neurons from rats with a painful neuropathy. *Pain* 65, 235–242.
 Suarez-Roca, H. and Maixner, W. (1993) Activation of Kappa
- Suarez-Roca, H. and Maixner, W. (1993) Activation of Kappa opioid receptors by U 50488H and morphine enhances the release of substance P from rat trigeminal nucleus slices. J. Pharmac. Exp. Ther. 264, 648–653.
- Sucher, N. J., Awobuluyi, M., Choi, Y. B. and Lipton, S. A. (1996) NMDA receptors: from genes to channels. *Trends Pharmac.* 17, 348–355.
- Suen, P-C., Wu, K., Levine, E. S., Mount, H. T. J., Xu, J-L., Lin, S-Y. and Black, I. B. (1997) Brain-derived neurotrophic factor rapidly enhances phosphorylation of the postsynaptic N-methyl-D-aspartate receptor subunit 1. Proc. natl Acad. Sci. U.S.A. 94, 8191–8195.
- Sugimoto, T., Takemura, M., Sakai, A. and Ishimaru, M. (1987) Rapid transneuronal destruction following peripheral nerve translation in the modullary dorsal horn is enhanced by strychnine, picrotoxin and bicuculline. *Pain* **30**, 385–393.
- Sugimoto, T., Bennett, G. J. and Kajander, K. C. (1989) Strychnine-enhanced trans-synaptic degeneration of dorsal horn neurons in rats with an experimental painful peripheral neuropathy. *Neurosci. Lett.* 98, 139–143.
- Sugimoto, T., Bennett, G. J. and Kajander, K. C. (1990) Transsynaptic degeneration in the superficial dorsal horn after sciatic nerve injury: effects of a chronic constriction injury, transection and strychnine. *Pain* 42, 205–213.
- Sugimoto, T., Fujiyoshi, Y., Xiao, C., He, Y. F. and Ichikawa, H. (1997) Central projection of calcitonin gene-related peptide (CGRP)-and substance P (SP)-immunoreactive trigeminal primary neurons in the rat. J. Comp. Neurol. 378, 425–442.
- Sugimoto, Y., Shigemoto, R., Namba, T., Negishi, M., Mizuno, N., Narumiya, S. and Ichikawa, A. (1994) Distribution of the messenger RNA for the prostaglandin E receptor subtype EP₃ in the mouse nervous system. *Neuroscience* 62, 919–928.
- Sugiura, Y., Lee, C. L. and Perl, E. R. (1986) Central projections of identified unmyelinated (C) afferent fibers innervating mammalian skin. *Science* 234, 359–361.
- Sun, L., Philipson, L. H. and Miller, R. J. (1998) Regulation of K^+ and Ca^{2+} channels by a family of neuropeptide Y receptors. *J. Pharmac. Exp. Ther.* **284**, 625–632.
- Sundblom, D. M., Kalso, E., Tigerstedt, I., Wahlbeck, K., Panulas, P. and Fyhrquist, F. (1997) Neuropeptide FF-like immunoreactivity in human cerebrospinal fluid of chronic pain patients and healthy controls. *Peptides* 18, 923–927.
- Sutton, J. L., Hesson, D. P., Goldman, J. M., King, G. W. and Kajander, K. C. (1997) Kainate receptor antagonist SYM 2081 attenuates mechanical allodynia and thermal hyperalgesia in a rat. Am. Soc. Neurosci. 23, 1015.
- Suzuki, T., Kishimoto, Y. and Misawa, M. (1996) Formalin- and carrageenan-induced inflammation attenuates place preferences produced by morphine, methamphetamine and cocaine. *Life Sci.* 59, 1667–1674.
- Svendsen, F., Tjølsen, A. and Hole, K. (1997) LTP of spinal $A\beta$ and C-fibre evoked responses after electrical sciatic nerve stimulation. *NeuroReport* **8**, 3427–3430.
- Svensson, P., Minoshima, S., Beydoun, A., Morrow, T. J. and Casey, K. L. (1997) Cerebral processing of acute skin and muscle pain in humans. J. Neurophysiol. 78, 450–460.
- Svichar, N., Shmigol, A., Verkhratsky, A. and Kostyuk, P. (1997a) ATP induces Ca²⁺ release from IP₃-sensitive Ca²⁺ stores exclusively in large DRG neurones. *NeuroReport* 8, 1555– 1559.
- Svichar, N., Shmigol, A., Verkhratsky, A. and Kostyuk, P. (1997b) InP₃-induced Ca²⁺ release in dorsal root ganglion neurones. *Neurosci. Lett.* **227**, 107–110.
- Swett, J. E. and Woolf, C. J. (1985) Somatotopic organization of primary afferent terminals in the superficial dorsal horn of the rat spinal cord. J. Comp. Neurol. 231, 66–71.
 Swierkosz, T. A., Mitchell, J. A., Warner, T. D., Bottingj, R. M.
- Swierkosz, T. A., Mitchell, J. A., Warner, T. D., Bottingj, R. M. and Vane, J. R. (1995) Co-induction of nitric oxide synthase and cyclo-oxygenase: interactions between nitric oxide and prostanoids. *Br. J. Pharmac.* **114**, 1335–1342.

- Swope, S. L., Moss, J. S., Blackstone, C. D. and Huganir, R. L. (1992) Phosphorylation of ligand-gated ion channels: a possible mode of synaptic plasticity. *FASEB J.* 6, 2514–2523.
- Szallasi, A. (1997) Perspectives on vanilloids in clinical practice. Drugs News Perspectives 10, 522–527.
- Szallasi, A. and Blumberg, P. (1996) Vanilloid receptors: new insights enhance potential as a therapeutic target. *Pain* 68, 195– 208.
- Szallasi, A., Nilsson, S., Frakas-Szallasi, T., Blumberg, P. M., Hökfelt, T. and Lundberg, J. M. (1995) Vanilloid (capsaicin) receptors in the rat: distribution in the brain, regional differences in the spinal cord, axonal transport to the periphery and depletion by systemic vanilloid treatment. *Brain Res.* **703**, 175– 183.
- Székely, J. I., Kedves, R., Maté, I., Török, K. and Tarnawa, I. (1997) Apparent antinociceptive and anti-inflammatory effects of GYKI 52466. *Eur. J. Pharmac.* 336, 143–154.
- Tachibana, M., Wenthold, R. J., Morioka, H. and Petralia, R. S. (1994) Light and electron microscopic immunocytochemical localization of AMPA-selective glutamate receptors in the rat spinal cord. J. Comp. Neurol. 344, 431–454.
- Tafreschi, A. P., Zhou, X-F. and Rush, R. A. (1998) Endogenous nerve growth factor and neurotrophin-3 act simultaneously to ensure the survival of postnatal sympathetic neurons in vivo. *Neuroscience* 83, 373–380.
- Taiwo, Y. O. and Levine, J. D. (1986) Indomethacin blocks central nociceptive effects of PGF_{2x}. *Brain Res.* 373, 81–84.
 Taiwo, Y. O. and Levine, J. D. (1988) Characterization of the ara-
- Taiwo, Y. O. and Levine, J. D. (1988) Characterization of the arachidonic acid metabolites mediating bradykinin and noradrenaline hyperalgesia. *Brain Res.* 458, 402–406.
- Taiwo, Y. O. and Levine, J. D. (1991) Further confirmation of the role of adenyl-cyclase and of cAMP-dependent protein kinase in primary afferent hyperalgesia. *Neuroscience* 44, 131–135.
- Taiwo, Y. O. and Levine, J. D. (1992) Serotonin is a directly-acting hyperalgesic agent in the rat. *Neuroscience* 48, 485–490.
 Taiwo, Y. O., Bjerknes, L. K., Goetzl, E. J. and Levine, J. D.
- Taiwo, Y. O., Bjerknes, L. K., Goetzl, E. J. and Levine, J. D. (1989) Mediation of primary afferent peripheral hyperalgesia by the cAMP second messenger system. *Neuroscience* 32, 577–580.
- Takahashi, Y., Nagajima, Y., Sakamoto, T., Moriya, H. and Takahashi, K. (1993) Capsaicin applied to rat lumbar intervertebral disc causes extravasation in the groin skin: a possible mechanism of referred pain of the intervetebral disc. *Neurosci. Lett.* 161, 1–3.
- Takaishi, K., Eisele, J. H. and Carstens, E. (1996) Behavioral and electrophysiological assessment of hyperalgesia and changes in dorsal horn responses following partial sciatic nerve ligation in rats. *Pain* 66, 297–306.
- Tal, M. and Bennett, G. T. (1994) Extra-territorial pain in rats with a peripheral mononeuropathy: mechano-hyperalgesia and mechano-allodynia in the territory of an uninjured nerve. *Pain* 57, 175–382.
- Tal, M. and Liberman, R. (1997) Local injection of nerve growth factor (NGF) triggers degranulation of mast cells in rat paw. *Neurosci. Lett.* 221, 129–132.
- Talley, N. J. (1992) 5-Hydroxytryptamine agonists and antagonists in the modulation of gastrointestinal motility and sensation: clinical implications. *Aliment. Pharmac. Ther.* 6, 273–289.
- clinical implications. Aliment. Pharmac. Ther. 6, 273–289.
 Tanaka, J., Koshimura, K., Murakami, Y., Sohmiya, M., Yanaihara, N. and Kato, Y. (1997a) Neuronal protection from apoptosis by pituitary adenylate cyclase-activating polypeptide. *Regulatory Peptides* 72, 1–8.
- Tanaka, T., Saito, H. and Matsuki, N. (1997b) Inhibition of GABA_A synaptic responses by brain-derived neurotrophic factor (BDNF) in rat hippocampus. J. Neurosci. 17, 2959–2966.
- Tancredi, V., Zona, C., Velotti, F., Eusebi, F. and Santoni, A. (1990) Interleukin-2 suppresses established long-term potentiation and inhibits its induction in the rat hippocampus. *Brain. Res.* 525, 149–151.
- Tanelian, D. L. and MacIver, M. B. (1991) Analgesic concentrations of lidocaine suppress tonic A-delta and C-fiber discharges produced by acute injury. *Anesthesiology* 74, 934–936.
- Tanelian, D. L., Barry, M. A., Johnston, S. A., Le, T. and Smith, G. M. (1997) Semaphorin III can repulse and inhibit adult sensory afferents in vivo. *Nature Med.* 12, 1398–1402.
- sory afferents in vivo. *Nature Med.* **12**, 1398–1402. Tani, E., Shiosaka, S., Sato, M., Ishikawa, T. and Tohyama, M. (1990) Histamine acts directly on calcitonin gene-related peptide and substance P-containing trigeminal ganglion neurons as assessed by calcium influx and immunocytochemistry. *Neurosci. Lett.* **115**, 171–176.
- Taniguchi, M., Yusa, S., Fujisawa, H., Naruse, I., Saga, S., Mishina, M. and Yagi, T. (1997) Disruption of semaphorin III/ D gene causes severe abnormality in peripheral nerve projection. *Neuron* 19, 519–530.

- Tao, Y. X., Wei, F. and Zhao, Z. Q. (1997) A contribution of neurokinin-1 receptors to formalin-induced c-fos expression in the rat spinal dorsal horn. *Neurosci. Lett.* 221, 105–108.
- Tasker, R. R. (1990) Pain resulting from central nervous system pathology (central pain). In: *The Management of Pain*, Vol. 1, 2nd edn, pp. 266–283. Ed. J. J. Bonica. Lea and Febiger Press, Philadelphia.
- Tavares, I., Lima, D. and Coimbra, A. (1993) Neurons in the superficial dorsal horn of the rat spinal cord projecting to the medullary ventrolateral reticular formation express c-fos after noxious stimulation of the skin. *Brain Res.* 623, 278–286.
- Taylor, B., Peterson, M. A. and Basbaum, A. I. (1995) Persistent cardiovascular and behavioral responses to subcutaneous formalin require peripheral nerve input. J. Neurosci. 15, 7575–7584.
- Taylor, D. C. M., Pierau, Fr.-K. and Szolcsanyi, J. (1985) Capsaicin-induced inhibition of axoplasmic transport is prevented by nerve growth factor. *Cell. Tiss. Res.* 240, 569–573.
- Telford, S., Wang, S. M. and Redgrave, P. (1996) Analysis of nociceptive neurones in the rat superior colliculus using c-fos immunohistochemistry. J. Comp. Neurol. 375, 601–617.
- Teoh, H., Malcangio, M. and Bowery, N. G. (1996) GABA, glutamate and substance P-like immunoreactivity release: effects of novel GABA_B antagonists. Br. J. Pharmac. 118, 1153–1160.
- Terenghi, G., Mann, D., Kopelman, P. G. and Anand, P. (1997) TrkA and trkC expression is increased in human diabetic skin. *Neurosci. Lett.* 228, 33–36.
- Terron, J. (1996) GR 127,935 is a potent antagonist of the 5-HT₁like receptor mediating contraction in the canine coronary artery. *Eur. J. Pharmac.* **300**, 109–112.
- Tessier-Lavigne, M. and Goodman, C. (1996) The molecular biology of axon guidance. *Science* 274, 1123-1133.
- Tessler, A., Himes, B. T., Houle, J. and Reier, P. J. (1988) Regeneration of adult dorsal root axons into transplants of embryonic spinal cord. J. Comp. Neurol. 270, 537–548.
- embryonic spinal cord. J. Comp. Neurol. **270**, 537–548. Tetzlaff, W., Alexander, S. W., Miller, F. D. and Bisby, M. A. (1991) Response of facial and rubrospinal neurons to axotomy: changes in mRNA expression for cytoskeletal proteins and GAP-43. J. Neurosci. **11**, 2528–2544.
- Thalhammer, J. G. and LaMotte, R. H. (1982) Spatial properties of nociceptor sensitization following heat injury of the skin. *Brain Res.* 231, 257–265.
- Theoharides, T. C., Singh, L. K., Boucher, W., Pang, X., Letourneau, R., Webster, E. and Chrousos, G. (1998) Cortocotropin-releasing hormone induces skin mast cell degranulation and increased vascular permeability, a possible explanation for its proinflammatory effects. *Endocrinology* 139, 403– 413.
- Thippeswamy, T. and Morris, R. (1997) Nerve growth factor inhibits the expression of nitric oxide synthase in neurones in dissociated cultures of rat dorsal root ganglia. *Neurosci. Lett.* 230, 9–12.
- Thoenen, H. (1995) Neurotrophins and neuronal plasticity. *Science* **270**, 593–598.
- Thomas, D., Ren, K. and Dubner, R. (1993) Nerve growth factor alleviates a painful peripheral neuropathy in rats. Am. Soc. Neurosci. 19, 969.
- Thomazzi, S. M., Ribeiro, R. A., Campos, D. I., Cunha, F. Q. and Ferreira, S. H. (1997) Tumour necrosis factor, interleukin-1 and interleukin-8 mediate the nociceptive activity of the supernatant of LPS-stimulated macrophages. *Mediators Inflam.* 6, 195–200.
- Thompson, S. W. N. and Wall, P. D. (1996) The effect of GABA and 5-HT receptor antagonists on rat dorsal root potentials. *Neurosci. Lett.* 217, 153–156.
- Thompson, S. W. N. and Majithia, A. (1997) LIF and NGF induce noradrenergic fibre sprouting in the intact dorsal root ganglion of adult rats in vivo. J. Physiol. 505.P, 42P.
- Thompson, S. W. N., Urban, L. and Dray, A. (1993a) Contribution of NK_1 and NK_2 receptor activation to high threshold afferent fibre evoked ventral root responses in the rat spinal cord in vitro. *Brain Res.* **625**, 100–108.
- Thompson, S. W. N., Woolf, C. J. and Sivilotti, L. G. (1993b) Small caliber afferent inputs produce a heterosynaptic facilitation of the synaptic responses evoked by primary afferent Afibers in the neonatal rat spinal cord in vitro. J. Neurophysiol. 69, 2116–2128.
- Thompson, S. W. N., Dray, A. and Urban, L. (1994) Injuryinduced plasticity of spinal reflex activity: NK₁ receptor activation and enhanced A and C-fiber mediated responses in the rat spinal cord in vitro. J. Neurosci. 14, 3672–3687.
- Thompson, S. W. N., Dray, A., McCarson, K. E., Krause, J. E. and Urban, L. (1995) Nerve growth factor induces mechanical allodynia associated with novel A-fibre-evoked spinal reflex ac-

tivity and enhanced neurokinin-1 receptor activation in the rat. *Pain* **62**, 219–231.

- Thompson, S. W. N., Dray, A. and Urban, L. (1996) Leukemia inhibitory factor induces mechanical allodynia but not thermal hyperalgesia in the juvenile rat. *Neuroscience* 71, 1091–1094.
- Thompson, S. W. N., Vernallis, A. B., Heath, J. K. and Priestley, J. V. (1997) Leukaemia inhibitory factor is retrogradely transported by a distinct population of adult rat sensory neurons: co-localization with TrkA and other neurochemical markers. *Eur. J. Pharmac.* 9, 1244–1251.
- Thorne, P. and Housley, G. (1996) Purinergic signalling in sensory systems. *Semin. Neurosci.* 8, 233–246.
- Thurston, C. L. and Helton, E. S. (1996) Effects of intravenous phenylephrine on blood pressure, nociception, and neural activity in the rostral ventral medulla in rats. *Brain Res.* 717, 81– 90.
- Todd, A. J. and Spike, R. C. (1993) The localization of classical transmitters and neuropeptides within neurons in laminae I–III of the mammalian spinal dorsal horn. *Prog. Neurobiol.* 41, 609– 646.
- Todd, A. J., Spike, R. C., Brodbelt, A. R., Price, R. F. and Shehab, S. A. S. (1994a) Some inhibitory neurons in the spinal cord develop c-fos immunoreactivity after noxious stimulation. *Neuroscience* 63, 805–813.
- Todd, A. J., Spike, R. C., Price, R. F. and Neilson, M. (1994b) Immunocytochemical evidence that neurotensin is present in glutamatergic neurons in the superficial dorsal horn of the rat. J. Neurosci. 14, 774–784.
- Todd, A. J., Watt, C., Spike, R. C. and Sieghart, W. (1996) Colocalization of GABA, glycine, and their receptors at synapses in the rat spinal cord. J. Neurosci. 16, 974–982.
- Todorovic, S. M. and Lingle, C. J. (1998) Pharmacological properties of T-type Ca²⁺ current in adult rat sensory neurons: effects of anticonvulsant and anesthetic agents. J. Neurophysiol. 79, 240–252.
- Todorovic, S. M., Scroggs, R. S. and Anderson, E. G. (1997) Cationic modulation of 5-HT₂ and 5-HT₃ receptors in rat sensory neurons: the role of K⁺, Ca²⁺ and Mg²⁺. *Brain Res.* **765**, 291–300.
- Toledo-Aral, J. J., Brehm, P., Halegoua, S. and Mandel, G. (1995) A single pulse of nerve growth factor triggers long-term neuronal excitation through sodium channel gene induction. *Neuron* 14, 607–611.
- Toledo-Aral, J. J., Moss, B. L., He, Z. J., Koszowski, A. G., Whisenand, T., Levinson, S. R., Wolf, J. J., Silos-Santiago, I., Halegoua, S. and Mandel, G. (1997) Identification of PN1, a predominant voltage-dependent sodium channel expressed principally in peripheral neurons. *Proc. natl Acad. Sci. U.S.A.* 94, 1527–1532.
- Tomiyama, M., Rodriguez-Puertas, R., Cortès, R., Christnacher, A., Sommer, B., Pazos, A., Palacios, J. M. and Mengod, G. (1996) Differential regional distribution of AMPA receptor subunit messenger RNAs in the human spinal cord as visualized by in situ hybridization. *Neuroscience* **75**, 901–915.
- Tommerdahl, M., Delemos, K. A., Vierck, C. J., Favorov, O. V. and Whitsel, B. L. (1996) Anterior parietal cortical response to tactile and skin-heating stimuli applied to the same skin site. J. Neurophysiol. 75, 2662–2670.
- Toms, N. J., Roberts, P. J., Salt, T. E. and Staton, P. C. (1996) Latest eruptions in metabotropic glutamate receptors. *Trends Pharmac.* 17, 429–435.
- Tonussi, C. R. and Ferreira, S. H. (1997) Bradykinin-induced knee joint incapacitation involves bradykinin B₂ receptor mediated hyperalgesia and bradykinin B₁ receptor-mediated nociception. *Eur. J. Pharmac.* 326, 61–65.
- Torebjörk, H. E. and Hallin, R. G. (1973) Perceptual changes accompanying controlled preferential blocking of A and C fibre responses in intact human skin nerves. *Exp. Brain Res.* 16, 321– 332.
- Torebjörk, H. E., LaMotte, R. H. and Robinson, C. J. (1984) Peripheral neural correlation of magnitude of cutaneous pain and hyperalgesia: simultaneous recordings in humans of sensory judgements of pain and evoked response in nociceptors with Cfibres. J. Neurophysiol. 51, 325.
- Torebjörk, H. E., Lundberg, L. E. R. and LaMotte, R. H. (1992) Central changes in processing of mechanoreceptive input in capsaicin-induced secondary hyperalgesia in humans. J. Physiol. 448, 765–780.
- Torebjörk, H. E., Wahren, L., Wallin, G., Hallin, R. and Koltzenburg, M. (1995) Noradrenaline-evoked pain in neuralgia. *Pain* 63, 11–20.
- Towers, S., Billinton, A., Bittiger, H., Urban, L. and Bowery, N. G. (1997) $GABA_B$ receptors in spinal cord and dorsal root

ganglia of sciatic nerve axotomised and control rats: localization of receptor protein by autoradiography using [³H]-CGP62349. *Br. J. Pharmac.* **122**, 345P.

- Tracey, D. J. and Walker, J. S. (1995) Pain due to nerve damage: are inflammatory mediators involved? *Inflamm. Res.* 44, 407– 411.
- Tracey, D. J., Cunningham, J. E. and Romm, M. A. (1995) Peripheral hyperalgesia in experimental neuropathy: mediation by α_2 adrenoreceptors on postganglionic sympathetic terminals. *Pain* **60**, 317–327.
- Traub, R. J. (1996) The spinal contribution of substance P to the generation and maintenance of inflammatory hyperalgesia in the rat. Pain 67, 151–161.
- Traub, R. J. (1997) The spinal contribution of the induction of central sensitization. *Brain Res.* **778**, 34–42.
- Travagli, R. A. (1996) Muscarine receptor activation in the substantia gelatinosa of the spinal trigeminal nucleus of the guinea pig. J. Neurophysiol. 76, 3817–3822.
- Travagli, R. A. and Williams, J. T. (1996) Endogenous monoamines inhibit glutamate transmission in the spinal trigeminal nucleus of the guinea pig. J. Physiol. 491, 177–185.
- Treede, R. D. and Cole, J. D. (1993) Dissociated secondary hyperalgesia in a subject with large fibre sensory neuropathy. *Pain* 53, 169–174.
- Treede, R. D. and Magerl, W. (1995) Modern concepts of pain and hyperalgesia: beyond the polymodal C-nociceptor. *News Physiol. Sci.* 10, 216–228.
- Treede, R. D., Meyer, R. A. and Campbell, J. N. (1990a) Comparison of heat and mechanical receptive fields of cutaneous C-fiber nociceptors in monkey. J. Neurophysiol. 64, 1502–1513.
- Treede, R. D., Raja, S. N., Davis, K. D., Meyer, R. A. and Campbell, J. N. (1990b) Evidence that alpha-adrenergic receptors mediate sympathetically maintained pain. *Pain* Suppl. 1990, Abstr. S487.
- Treede, R. D., Davis, K. D., Campbell, J. N. and Raja, S. N. (1992a) The plasticity of cutaneous hyperalgesia during sympathetic ganglion blockade in patients with neuropathic pain. *Brain* 115, 607–621.
- Treede, R. D., Meyer, R. A., Raja, S. N. and Campbell, J. N. (1992b) Peripheral and central mechanisms of cutaneous hyperalgesia. *Prog. Neurobiol.* 38, 397–421.
- Trezise, D. J., John, V. H. and Xie, X. (1997) Electrophysiological studies on the effects of lamotrigine on rat dorsal root ganglion neurones *in vitro*. Br. J. Pharmac. 122, 31P.
- Tseng, G. F. (1996) Structural and functional alterations in rat corticospinal neurons after axotomy. J. Neurophysiol. 75, 248– 267.
- Tsumoto, T. and Yasuda, H. (1996) A switching role of postsynaptic calcium in the induction of long-term potentiation or long-term depression in visual cortex. *Neuroscience* 8, 311–319.
- Turrigiano, G. G., Leslie, K. R., Desai, N. S., Rutherford, L. C. and Nelson, S. B. (1998) Activity dependent scaling of quantal amplitude in neocortical neurons. *Nature* **391**, 892–896.
- Ulrich, D. and Huguenard, J. R. (1997) GABA (A)-receptor mediated rebound burst firing and burst shunting in thalamus. J. Neurophysiol. 78, 1748–1751.
- Urban, M. O. and Gebhart, G. F. (1997) Characterization of biphasic modulation of spinal nociceptive transmission by neurotensin in the rat rostral ventromedial medulla. J. Neurophysiol. 78, 1550–1562.
- Urban, L., Naeem, S., Patel, I. A. and Dray, A. (1994a) Tachykinin induced regulation of excitatory amino acid responses in the rat spinal cord in vitro. *Neurosci. Lett.* **168**, 185– 188.
- Urban, L., Thompson, S. W. N. and Dray, A. (1994b) Modulation of spinal excitability: co-operation between neurokinin and excitatory amino acid neurotransmitters. *Trends Neurosci.* 17, 432–438.
- Urban, M. O., Jiang, M. C. and Gebhart, G. F. (1996a) Participation of central descending nociceptive facilitatory systems in secondary hyperalgesia produced by mustard oil. *Brain Res.* **737**, 83–91.
- Urban, M. O., Smith, D. J. and Gebhart, G. F. (1996b) Involvement of spinal cholecystokinin_B receptors in mediating neurotensin hyperalgesia from the medullary nucleus raphe magnus in the rat. J. Pharmac. Exp. Ther. 278, 90–96.
- Usachev, Y. M. and Thayer, S. A. (1997) All-or-none Ca²⁺ release from intracellular stores triggered by Ca²⁺ influx through voltage-gated Ca²⁺ channels in rat sensory neurons. *J. Neurosci.* **17**, 7404–7414.

- Utzschneider, D., Kocsis, J. and Devor, M. (1992) Mutual excitation among dorsal root ganglion neurons in the rat. *Neurosci. Lett.* 146, 53–56.
- Vaccarino, A. L. and Melzack, R. (1989) Analgesia produced by injection of lidocaine into the anterior cingulum bundle of the rat. *Pain* 39, 213–219.
- Vaccarino, A. L. and Melzack, R. (1992) Temporal processes of formalin pain: differential role of the cingulum bundle, fornix pathway and medial bulboreticular formation. *Pain* 49, 257– 271.
- Vaccarino, F., Guidotti, A. and Costa, E. (1987) Ganglioside inhibition of glutamate-mediated-protein kinase C translocation in primary cultures of cerebellar neurons. *Proc. natl Acad. Sci. U.S.A.* 84, 8707–8711.
- Vaccarino, A. L., Clemmons, H. R., Mader, G. J., Jr and Magnusson, J. E. (1997) A role of periaqueductal grey NMDA receptors in mediating formalin-induced pain in the rat. *Neurosci. Lett.* 236, 117–119.
- Valentin, J. P., Bonnafous, R. and Gareth, W. J. (1996) Influence of the endothelium and nitric oxide on the contractile responses evoked by 5-HT_{1D} receptor agonists in the rabbit isolated saphenous vein. Br. J. Pharmac. 119, 35–42.
- Valerio, A., Paterlini, M., Boifava, M., Memo, M. and Spano, P. F. (1997) Metabotropic glutamate receptor mRNA expression in rat spinal cord. *NeuroReport* 8, 2695–2699.
- Valeyev, A. Y., Hackman, J. C., Wood, P. M. and Davidoff, R. A. (1996) Pharmacologically novel GABA receptors in human dorsal root ganglion neurons. J. Neurophysiol. 76, 3555–3558.
- Valtschanoff, J. G., Weinberg, R. J. and Rustioni, A. (1992) NADPH diaphorase in the spinal cord of rats. J. Comp. Neurol. 321, 209–222.
- Valtschanoff, J. G., Weinberg, R. J. and Rustioni, A. (1995) Central release of tracer after noxious stimulation of the skin suggests non-synaptic signaling by unmyelinated fibers. *Neuroscience* 64, 851–854.
- Van Coppenolle, F., Ahidouch, A., Guilbault, P. and Houadid, H. (1997) Regulation of endogenous Ca²⁺ channels by cyclic AMP and cyclic GMP-dependent protein kinases in pleurodeles oocytes. *Mol. Cell Biochem.* **168**, 155–161.
- Van der Zee, C. E. E. M., Nielander, H. B., Vos, J. P., Lopes da Silva, S. L., Verhaagen, J., Oestreicher, A. B., Schrama, L. H., Schotman, P. and Gispen, W. H. (1989) Expression of growthassociated protein B-50 (GAP 43) in dorsal root ganglia and sciatic nerve during regenerative sprouting. *J. Neurosci.* 9, 3505– 3512.
- Vanderah, T. W., Bernstein, R. N., Yamamura, H. I., Hruby, V. J. and Porreca, F. (1996a) Enhancement of morphine antinociception by a CCK_B antagonist in mice is mediated via opioid delta receptors. *J. Pharmac. Exp. Ther.* 278, 212–219.
- Vanderah, T. W., Laughlin, T., Lashbrook, J. M., Nichols, M. L., Wilcox, G. L., Ossipov, M. H., Malan, T. P. and Porreca, F. (1996b) Single intrathecal injections of dynorphin A or des-Tyrdynorphins produce long-lasting allodynia in rats: blockade by MK-801 but not naloxone. *Pain* 68, 275–281.
- Vanegas, H., Tortorici, V., Eblen-Zajjur, A. and Vasquez, E. (1997) PAG-microinjected dipyrone (metamizol) inhibits responses of spinal dorsal horn neurons to natural noxious stimulation in rats. *Brain Res.* **759**, 171–174.
- Vanhoesen, G. W. (1995) Anatomy of the medial temporal lobe. Magnetic Resonance Imaging 13, 1047–1055.
- Vaudano, E., Campbell, G., Anderson, P. N., Davies, A. P., Woolhead, C., Schreyer, D. J. and Lieberman, A. R. (1995) The effects of a lesion or a peripheral nerve graft on GAP-43 upregulation in the adult rat brain: an in situ hybridization and immunocytochemical study. J. Neurosci. 15, 3594–3611.
- Vecchiet, L., Dragani, L., de Bigontina, P., Obletter, G. and Giamberardino, M. A. (1993) Experimental referred pain and hyperalgesia from muscles in humans. In: *New Trends in Referred Pain in Hyperalgesia*, pp. 222–232. Eds. L. Vecchiet, D. Albe-Fessard, U. Lindblom and M. A. Giamberardino. Elsevier, Amsterdam.
- Venstrom, B. G. and Reichardt, L. (1993) Extracellular matrix 2. Role of extracellular matrix molecules and their receptors in the nervous system. *FASEB J.* 7, 996–10003.
- Verdugo, R. J. and Ochoa, J. L. (1994) Sympathetically maintained pain. I. Phentolamine block questions the concept. *Neurology* 44, 1003–1010.
- Verdugo, R. J., Campero, M. and Ochoa, J. L. (1994) Phentolamine sympathetic block in painful polyneuropathies. II. Further questioning of the concept of "sympathetically maintained pain". *Neurology* 44, 1010–1014.

- Vergé, V. M. K., Tetzlaff, W., Richardson, P. M. and Bisby, M. A. (1990) Correlation between GAP 43 and nerve growth factor receptors in rat sensory neurons. J. Neurosci. 10, 926–934.
- Vergé, V. M. K., Merlio, J. P., Grondin, J., Ernfors, P., Persson, H., Riopelle, R. J., Hökfelt, T. and Richardson, P. M. (1992a) Colocalisation of NGF binding sites, trk mRNA, and low-affinity NGF receptor mRNA in primary sensory neurons: responses to injury and infusion of NGF. J. Neurosci. 12, 4011– 4022.
- Vergé, V. M. K., Xu, Z., Xu, X. J., Wiensenfeld-Hallin, Z. and Hökfelt, T. (1992b) Marked increase in nitric oxide synthase mRNA levels in rat dorsal root ganglion neurons after peripheral axotomy. *Proc. natl Acad. Sci. U.S.A.* 89, 11617–11621.
- Vergé, V. M. K., Gratto, L. A., Karchewski, L. A. and Richardson, P. M. (1995a) Neurotrophins and nerve injury in the adult. *Philos. Trans. R. Soc. Lond. B* 351, 423–430.
- Vergé, V. M. K., Richardson, P. M., Wiesenfeld-Hallin, Z. and Hökfelt, T. (1995b) Differential influence of nerve growth factor on neuropeptide expression in vivo: a novel role in peptide suppression in adult sensory neurons. J. Neurosci. 15, 2081–2096.
- Verheggen, R., Freudenthaler, S., Meyer-Dulheuer, F. and Kaumann, A. J. (1996) Participation of 5-HT₁-like and 5-HT_{2A} receptors in the contraction of human temporal artery by 5-hydroxytryptamine and related drugs. *Br. J. Pharmac.* **117**, 283– 289.
- Verheggen, R., Schindler, M. and Kaumann, A. J. (1997) A comparison of 5-HT receptor mRNA in human temporal artery and brain. *Naunyn-Schmiedberg's Arch. Pharmac.* 355, R30.
 Verkade, P., Schrama, L. H., Verkleij, A. J., Gispen, W.-H. and
- Verkade, P., Schrama, L. H., Verkleij, A. J., Gispen, W.-H. and Oestreicher, A. B. (1997) Ultrastructural co-localization of calmodulin and B-50/growth-associated protein-43 at the plasma membrane of proximal unmyelinated axon shafts studied in the model of the regenerating rat sciatic nerve. *Neuroscience* 79, 1207–1218.
- Verkhratsky, A. and Kettenmann, P. (1996) Calcium signalling in glial cells. *Trends Neurosci.* 19, 346–352.
- Verma, A. and Kulkarni, S. K. (1993) Modulatory role of D_1 and D_2 dopamine receptor subtypes in nociception in mice. *J. Psychopharmac.* **7**, 270–275.
- Vernadakis, A. (1996) Glia-neuron intercommunications and synaptic plasticity. Prog. Neurobiol. 49, 185–214.
- Vertongen, P., Schiffmann, S. N., Gourlet, P. and Robberecht, P. (1997) Autoradiographic visualization of the receptor subclasses for vasoactive intestinal polypeptide. *Peptides* 18, 1547–1554.
- Vickery, R. M., Morris, S. H. and Bindman, L. J. (1997) Metabotropic glutamate receptors are involved in long-term potentiation in isolated slices of rat medial frontal cortex. J. Neurophysiol. 78, 3039–3046.
- Vierck, C. J., Greenspan, J. D., Ritz, L. A. and Yeomans, D. C. (1986) The spinal pathways contributing to the ascending conduction and the descending modulation of pain sensations and reactions. In: *Spinal Afferent Processing*, pp. 275–329, Ed. T. L. Yaksh. Plenum Press, New York.
- Vierck, C. J., Cannon, R. L., Fry, G., Maixner, W. and Whitsel, B. L. (1997) Characteristics of temporal summation of second pain sensations elicited by brief contact of glabrous skin by a preheated thermode. J. Neurophysiol. 78, 992–1002.
- Villar, M. J., Cortés, R., Theodorsson, E., Wiesenfeld-Hallin, Z., Schalling, M., Fahrenkrug, J., Emson, P. C. and Hökfelt, T. (1989) Neuropeptide expression in rat dorsal root ganglion cells and spinal cord after peripheral nerve injury with special reference to galanin. *Neuroscience* 33, 587–604.
- Vizzard, M. A., Erdman, S. L. and de Groat, W. C. (1995) Increased expression of nitric oxide synthase in dorsal root ganglia neurones after systemic capsaicin administration. *Neuroscience* 67, 1–5.
- Vo, P. A. and Tomlinson, D. R. (1997) Effects of adrenergic stimulation on sciatic nerve blood flow in diabetic and control rats. *Eur. J. Pharmac.* 329, 147–152.
- Vogt, B. A. and Gabriel, M. (1993) Neurobiology of Cingulate Cortex and Limbic Thalamus: A Comprehensive Handbook, p. 694. Birkhauser, Boston.
- Vogt, B. A., Sikes, R. W. and Vogt, L. J. (1993) Anterior cingulate cortex and the medial pain system. In: *Neurobiology of Cingulate Cortex and Limbic Thalamus: A Comprehensive Handbook*, pp. 512–536. Eds. B. A. Vogt and M. Gabriel. Birkhauser, Boston.
- Vogt, B. A., Derbyshire, S. and Jones, A. K. P. (1996) Pain processing in four regions of the human cingulate cortex localised and coregistered with PET and MR imaging. *Eur. J. Neurosci.* 8, 1461–1473.

- Von Bartheld, C. S., Byers, M. R., Williams, R. and Bothwell, M. (1996) Anterograde transport of neurotrophins and axodendritic transfer in the developing visual system. *Nature* 379, 830–833.
- Vulchanova, L., Ardvidsson, U., Roedl, M., Wang, J., Buell, G., Surprenant, A., North, R. A. and Elde, R. (1996) Differential distribution of ATP-gated ion channels (P_{2X} receptors) determined by immunohistochemistry. *Proc. natl Acad. Sci. U.S.A.* 93, 8063–8067.
- Vulchanova, L., Riedl, M. S., Shuster, S. J., Buell, G., Surprenant, A., North, R. A. and Elde, R. (1997) Immunohistochemical study of the P_{2X2} and P_{2X3} receptor subunits in rat and monkey sensory neurons and their central terminals. *Neuropharmacology* **36**, 1229–1242.
- Wagner, R. and Myers, R. R. (1996a) Endoneurial injection of TNF α produces neuropathic pain behaviors. *NeuroReport* 7, 2897–2901.
- Wagner, R. and Myers, R. R. (1996b) Schwann cells produce tumor necrosis factor α expression in injured and non-injured nerves. *Neuroscience* **73**, 625–629.
- Wagner, R., Janjigian, M. and Myers, R. R. (1998) Anti-inflammatory interleukin-10 therapy in CCi neuropathy decreases thermal hyperalgesia, macrophage recruitment, and endoneurial TNF-α expression. *Pain* 74, 35–42.
- Wahren, L. K., Torebjörk, H. E. and Jorum, E. (1989) Central suppression of cold-induced C-fibre pain by myelinated fibre input. *Pain* 38, 313–319.
- Wahren, L. K., Gordh, T. and Torebjörk, H. E. (1995) Effects of regional intravenous guanethidine in patients with neuralgia in the hand: a follow-up study over a decade. *Pain* 62, 379–385.Wakisaka, S., Kajander, K. C. and Bennett, G. J. (1991) Increased
- Wakisaka, S., Kajander, K. C. and Bennett, G. J. (1991) Increased neuropeptide Y (NPY)-like immunoreactivity in rat sensory neurons following peripheral axotomy. *Neurosci. Lett.* **124**, 200– 203.
- Waldmann, R., Bassilanat, F., de Weillet, J., Champigny, G., Heurteaux, C. and Lazdunski, M. (1997a) Molecular cloning of a non-inactivating proton-gated Na⁺ channel specific for sensory neurons. J. Biol. Chem. 272, 20975–20978.
- Waldmann, R. G., Champigny, G., Bassilana, F., Heurteaux, C. and Lazdunski, M. (1997b) A proton-gated cation channel involved in acid-sensing. *Nature* 386, 173–177.
- Walker, K., Perkins, M. and Dray, A. (1995) Kinins and kinin receptors in the nervous system. *Neurochem. Int.* 26, 1–16.
- Wall, P. D. (1995) Do nerve impulses penetrate terminal arborizations? A presynaptic control mechanism. *Trends Neurosci.* 18, 99–103.
- Wall, P. D. and Bennett, D. L. H. (1994) Postsynaptic effects of long-range afferents in distant segments caudal to their entry point in rat spinal cord under the influence of picrotoxin or strychnine. J. Neurophysiol. 72, 2703–2713.
- Wall, P. D. and Devor, M. (1981) The effect of peripheral nerve injury on dorsal root potentials and on transmission of afferent signals into the spinal cord. *Brain Res.* 209, 95–111.
- Wall, P. D. and Lidierth, M. (1997) Five sources of a dorsal root potential: their interactions and origins in the superficial dorsal horn. J. Neurophysiol. 78, 860–871.
- Wall, P. D. and Melzack, R. (1994) *Textbook of Pain*, 3rd edn, p. 1524. Churchill-Livingstone, Edinburgh.
- Wall, P. D. and Woolf, C. J. (1984) Muscle but not cutaneous Cafferent input produces prolonged increases in the excitability of the flexion reflex in the rat. J. Physiol. 356, 443–458.
- Wallin, B. G., Torebjörk, H. E. and Hallin, R. G. (1976) Preliminary observations on the pathophysiology of hyperalgesia in the causalgic pain syndrome. In: Sensory Functions of the Skin of Primates with Special Reference to Man, pp. 489–499. Ed. Y. Zotterman. Pergamon, Oxford.
- Wang, B. Y. C., Alderman, C., Peh, K. H. and Assem, E. S. K. (1997a) The mode of action of bradykinin and other basic secretagogues on histamine release from rat peritoneal mast cells. *Br. J. Pharmac.* 122, 106P.
- Wang, H., Liu, R. J., Zhang, R. X. and Qioa, J. T. (1997b) Peripheral NMDA receptors contribute to activation of nociceptors: a c-fos expression study in rats. *Neurosci. Lett.* 221, 101– 104.
- Wang, J., Ren, M. and Han, J. (1992) Mobilization of calcium from intracellular stores as one of the mechanisms underlying the antiopioid effects of cholecytokinin octapeptide. *Peptides* 13, 947–951.
- Wang, J. F., Khazar, S. G., Ahlgren, S. C. and Levine, J. D. (1996a) Sensitization of C-fibres by prostaglandin E₂ in the rat is inhibited by guanosine 5'-O-(2-thiodiphosphate), 2',5'-dideoxyadenosine and walsh inhibitor peptide. *Neuroscience* 71, 259– 263.

- Wang, J. H. and Kelly, P. T. (1997) Postsynaptic calcineurin activity downregulates synaptic transmision by weakening intrasignaling mechanisms in hippocampal CA1 cellular Ca neurons. J. Neurosci. 17, 4600-4611.
- Wang, L., Millecchai, P. and Brown, P. B. (1997c) Correlation of peripheral innervation density and dorsal horn map scale. J. Neurophysiol. 78, 689-702.
- Wang, S., Yan, L., Wesley, R. A. and Danner, R. L. (1997d) Nitric oxide increases tumor necrosis factor production in differentiated U 937 cells by decreasing cyclic AMP. J. Biol. Chem. 272, 5959-5965.
- Wang, S. D., Goldberger, M. E. and Murray, M. (1991) Plasticity of spinal systems after unilateral lumbosacral dorsal rhizotomy in the adult rat. J. Comp. Neurol. 304, 555-568.
- Wang, X. and Robinson, P. (1997) Cyclic GMP-dependent protein kinase and cellular signaling in the nervous system. J. Neurochem. 68, 443-456.
- Wang, Y., Wu, J., Rowan, M. J. and Anwyl, R. (1996b) Ryanodine produces a low frequency stimulation-induced NMDA receptor-independent long-term potentiation in the rat dentate gyrus in vitro. J. Physiol. (Lond). 495, 755-767.
- Wang, Y., Li, L., Liu, J., Wang, Z., Jiang, C. and Liu, X. (1997e) The analgesic domain of IL-2. Biochem. Biophys Res. Commun. 230, 542-545.
- Wang, Y., Lin, S.-Z., Chiou, A.-L., Williams, L. R. and Hoffer, B. J. (1997f) Glial cell line-derived neurotrophic factor protects against ischemia-induced injury in the cerebral cortex. J. Neurosci. 17, 4341–4348.
- Wang, Y., Rowan, M. J. and Anwyl, R. (1997g) Induction of LTD in the dentate gyrus in vitro is N-methyl-D-aspartate receptor independent, but dependent on Ca^{2+} influ voltage activated Ca^{2+} channels and release of Ca^{2+} influx via low from intracellular stores. J. Neurophysiol. 77, 812-825.
- Wang, Y. T. and Salter, M. W. (1994) Regulation of NMDA receptors by tyrosine kinases and phosphatases. Nature 369, 233-235.
- Wang, Z., Van Den Berg, R. J. and Ypey, D. L. (1997h) Hyperpolarization-activated currents in the growth cone and soma of nonatal rat dorsal root ganglion neurons in culture. J. Neurophysiol. 78, 177-186
- Warncke, T., Jørum, E. and Stubhaug, A. (1997a) Local treatment with the N-methyl-D-aspartate receptor antagonist, ketamine, inhibits development of secondary hyperalgesia. Neurosci. Lett. 227. 1-4.
- Warncke, T., Stubhaug, A. and Jorum, E. (1997b) Ketamine, an NMDA receptor antagonist, suppresses spatial and temporal properties of burn-induced secondary hyperalgesia in man: a double-blind, cross-over comparison with morphine and placebo. Pain 72, 99-106.
- Washburn, M. S., Numberger, M., Zhang, S. and Dingledine, R. (1997) Differential dependence on GluR2 expression of three characteristic features of AMPA receptors. J. Neurosci. 17, 9393-9406.
- Watkins, L. R., Wiertelak, E. P., Goehler, L. Mooneyheiberger, K., Martinez, J., Furness, L., Smith, K. P. and Maier, S. F. (1994) Neurocircuitry of illness-induced hyperalgesia. Brain Res. 639, 283-299.
- Watkins, L. R., Goehler, L. E., Relton, J., Brewer, M. T. and Maier, S. F. (1995) Mechanisms of tumor necrosis factor-alpha (TNF-alpha) hyperalgesia. Brain Res. 692, 244-250.
- Watkins, L. R., Martin, D., Ulrich, P., Tracey, K. J. and Maier, S. F. (1997a) Evidence for the involvement of spinal cord glia in subcutaneous formalin induced hyperalgesia in the rat. Pain 71, 225-235.
- Watkins, L. R., McGorry, M., Schwartz, B., Sisk, D., Wiertelak, E. P. and Maier, S. F. (1997b) Reversal of spinal cord non-opiate analgesia by conditioned anti-analgesia in the rat. Pain 71, 237-247
- Watson, A., Ensor, E., Symes, A., Winter, J., Kendall, G. and Latchman, D. (1995) A minimal CGRP gene promoter is inducible by nerve growth factor in adult rat dorsal root ganglion neurons but not in PC12 phaeochromocytoma cells. Eur. J. Neurosci. 7, 394-400.
- Waxman, S. G., Kocsis, J. D. and Black, J. A. (1994) Type III sodium channel mRNA is expressed in embryonic but not adult spinal sensory neurons, and is re-expressed following axotomy. J. Neurophysiol. 72, 466-470.
- Wegert, S., Ossipov, M. H., Nichols, M. L., Bian, D., Vanderah, T. W., Malan Jr, T. P. and Porreca, F. (1997) Differential activities of intrathecal MK-801 or morphine to alter responses to thermal and mechanical stimuli in normal or nerve-injured rat. Pain 71, 57-64.

- Wegner, H. (1996) Diltiazem blocks the pH-induced excitation of rat nociceptors together with their mechanical and electrical excitability in vitro. J. Neurophysiol. 75, 1-10.
- Weil-Fugazza, J. (1989) Central metabolism and release of serotonin in pain and analgesia. In: Serotonin and Pain, pp. 221-238. Ed. J. M. Besson. Elsevier, Amsterdam.
- Weinreich, D. and Wonderlin, W. F. (1987) Inhibition of calciumdependant spike after-hyperpolarization increases excitability of rabbit visceral sensory neurons. J. Physiol. 394, 415-427.
- Weinreich, D., Moore, K. A. and Taylor, G. E. (1997) Allergic in-Molet, D., Molet, K. A. and Taylor, G. E. (1977) Antrige in-flammation in isolated vagal sensory ganglia unmasks silent NK-2 tachykinin receptors. J. Neurosci. 17, 7683–7693.
 Weiss, S., Dunne, C., Hewson, J., Wohl, C., Wheatley, M., Peterson, A. C. and Reynolds, B. A. (1996) Multipotent CNS
- stem cells are present in the adult mammalian spinal cord and ventricular neuroaxis. J. Neurosci. 16, 7599-7609.
- Weiya, M. A., Ribeiro-Da-Silva, A., De Koninck, Y., Radhakrishnan, V., Henry, J. L. and Cuello, A. C. (1996) Quantitative analysis of substance P-immunoreactive boutons on physiologically characterized dorsal horn neurons on the cat lumbar spinal cord. J. Comp. Neurol. 376, 45-64.
- Wells, M. R., Racis, S. P. and Vaidya, U. (1992) Changes in plasma cytokines associated with peripheral nerve injury. J. Neuroimmun. 39, 261-268.
- Wells, M. R., Vaidya, U. and Schwartz, J. P. (1994) Bilateral phasic increases in dorsal root ganglia nerve growth factor synthesis after unilateral sciatic nerve crush. Exp. Brain Res. 101, 53-58.
- Werner, P., Seward, E. P., Buell, G. N. and North, R. A. (1996) Domains of P2X receptors involved in desensitization. Proc. natl Acad. Sci. U.S.A. 93, 15485-15490.
- Westlund, K. N., Sun, Y. C., Sluka, K. A., Dougherty, P. M., Sorkin, L. S. and Willis, W. D. (1992) Neuronal changes in acute arthritis in monkeys. II. Increased glutamate immunoreactivity in the medial articular nerve. Brain Res. Rev. 17, 15.
- Whalley, E. T., Clegg, S., Stewart, J. M. and Vavrek, R. J. (1987) The effect of kinin agonists and antagonists on the pain response of the human blister base. Naunyn-Schmiedberg's Arch. Pharmac. 336, 652-655.
- White, D. M. (1996) Mechanism of prostaglandin E2-induced substance P release from cultured sensory neurons. Neuroscience 70. 561-565.
- White, D. M. and Mansfield, K. (1996) Vasoactive intestinal polypeptide and neuropeptide Y act indirectly to increase neurite outgrowth of dissociated dorsal root ganglion cells Neuroscience 73, 881-887.
- White, D. M., Basbaum, A. I., Goetzl, E. J. and Levine, J. D. (1990) The 15-lipoxygenase product, 8R, 15S-diHETE, stereo-specifically sensitizes C-fiber mechanoheat nociceptors in hairy skin of rat. J. Neurophysiol. 63, 966-970.
- White, J. C. and Sweet, W. H. (1969) Pain and the Neurosurgeon, p. 412. Thomas Press, Springfield.
- Wiertelak, E. P., Maier, S. F. and Watkins, L. R. (1992a) Cholecystokinin antianalgesia: safety cues abolish morphine analgesia. Science 256, 830-833.
- Wiertelak, E. P., Watkins, L. R. and Maier, S. F. (1992b) Conditioned inhibition of analgesia. Anim. Learn. Behav. 20, 339-349
- Wiertelak, E. P., Furness, L. E., Horan, R., Martinez, J., Maier, S. F. and Watkins, L. R. (1994a) Subcutaneous formalin produces centrifugal hyperalgesia at a non-injected site via the NMDA-nitric oxide cascade. Brain Res. 649, 19-26
- Wiertelak, E. P., Furness, L. E., Watkins, L. R. and Maier, S. F. (1994b) Illness-induced hyperalgesia is mediated by a spinal NMDA-nitric oxide cascade. Brain Res. 664. 9-16.
- Wiertelak, E. P., Smith, K. P., Furness, L., Mooney-Heiberger, K., Mayr, T., Maier, S. F. and Watkins, L. R. (1994c) Acute and conditioned hyperalgesic responses to illness. Pain 56, 227-234
- Wiertelak, E. P., Yang, H.-Y. Y., Mooney-Heiberger, K., Maier, S. F. and Watkins, L. R. (1994d) The nature of conditioned anti-analgesia: spinal cord opiate and anti-opiate neurochemistry. Brain Res. 634, 214-226.
- Wiesenfeld-Hallin, Z. (1998) Combined opioid-NMDA antagonist therapies. Drugs 56, 1-4. Wiesenfeld-Hallin, Z., Villar, M. J. and Hökfelt, T. (1989) The
- effects of intrathecal galanin and C-fiber stimulation on the flexor reflex in the rat. Brain Res. 486, 205-213.
- Wiesenfeld-Hallin, Z., Xu, X. J., Håkanson, R., Feng, D. M. and Folkers, K. (1990) Plasticity of the peptidergic mediation of spinal reflex facilitation after peripheral nerve section in the rat. Neurosci. Lett. 116, 293-298.
- Wiesenfeld-Hallin, Z., Xu, X-J., Langel, U., Bedecs, K., Hökfelt, T. and Bartfai, T. (1992) Galanin-mediated control of pain:

enhanced role after nerve injury. Proc. natl Acad. Sci. U.S.A. 89, 3334–3337.

- Wiesenfeld-Hallin, Z., Hao, J-X., Xu, X-J. and Hokfelft, T. (1993) Nitric oxide mediates ongoing discharges in dorsal root ganglion cells after peripheral nerve injury. J. Neurophysiol. 70, 2350–2353.
- Wilcox, G. L. (1991) Excitatory neurotransmitters and pain. In: *Pain Research and Clinical Management. Proc VIth World Congress on Pain*, Vol. 4, pp. 97–117. Eds. M. R. Bond, J. E. Charlton and C. J. Woolf. Elsevier, Amsterdam.
- Wilcox, G. L. (1993a) Sources and cellular targets for peripherally acting analgesics mediated by G protein-coupled receptors. J. Am. Pain Soc. 2, 60–65.
- Wilcox, G. L. (1993b) Spinal modulators of nociceptive neurotransmission and hyperalgesia: relationships among synaptic plasticity, analgesic tolerance and blood flow. J. Am. Pain Soc. 2, 265–275.
- Wildering, W. C., Lodder, J. C., Kits, K. S. and Bulloch, A. G. M. (1995) Nerve growth factor (NGF) enhances high-voltageactivated calcium currents in molluscan neurons. *J. Neurophysiol.* 74, 2778–2781.
- Wildman, S. S., King, B. F. and Burnstock, G. (1997) Potentiation of ATP-responses at recombinant P_{2X2} receptor by neurotransmitters and related substances. Br. J. Pharmac. 120, 221–224.
- Wilk-Blaszczak, M. A., Stein, B., Xu, S., Barbosa, M. S., Cobb, M. H. and Belardetti, F. (1998) The mitogen-activated protein kinase p38-2 is necessary for the inhibition of N-type calcium current by bradykinin. J. Neurosci. 18, 112–118.
- Wilkinson, G. A., Fariñas, I., Backus, C., Yoshida, C. K. and Reichardt, L. F. (1996) Neurotrophin-3 is a survival factor in vivo for early mouse trigeminal neurons. J. Neurosci. 16, 7661– 7669.
- Williamson, D. J., Shepheard, S. L., Hill, R. G. and Hargreaves, R. J. (1997) The novel anti-migraine agent rizatriptan inhibits neurogenic dural vasodilatation and extravasation. *Eur. J. Pharmac.* **328**, 61–64.
- Willingale, H. L. and Grubb, B. D. (1996) Evidence that prostaglandins may modulate wind-up in a spinal nociceptive reflex in anaesthetised rats. *Br. J. Pharmac.* **118**, 49.
- Willingale, H. L., Gardiner, N. J., McLymont, N., Giblett, S. and Grubb, B. D. (1997) Prostanoids synthesized by cyclo-oxygenase isoforms in rat spinal cord and their contribution to the development of neuronal hyperexcitability. *Br. J. Pharmac.* 122, 1593–1604.
- Willis, W. D. (1985) The Pain System: The Neural Basis of Nociceptire Transmission in the Mammalian Nervous System, p. 346. Karger, Basel.
- Willis, W. D. (1988) Anatomy and physiology of descending control of nociceptive responses of dorsal horn neurons: a comprehensive review. *Prog. Brain Res.* 77, 1–29.
 Willis, W. D. (1989) The origin and destination of pathways
- Willis, W. D. (1989) The origin and destination of pathways involved in pain transmission. In: *Textbook of Pain*, 2nd edn, pp. 112–127. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Willis, W. D. (1994) Central plastic responses to pain. In: Proceeding of the 7th World Congress on Pain, Vol. 2, pp. 301– 324. Eds. G. F. Gebhart, D. L. Hammond and T. S. Jensen. IASP Press, Seattle.
- Willis, W. D. (1997) Is central sensitization of nociceptive transmission in the spinal cord a variety of long-term potentiation? *NeuroReport* 8, iii.
- Willis, W. D. and Coggeshall, R. E. (1991) Sensory Mechanisms of the Spinal Cord, 2nd edn. p. 595. Plenum Press, New York.
- Wilson, J. L., Nayanar, V. and Walker, J. S. (1996a) The site of anti-arthritic action of the κ -opioid, U-50 488H, in adjuvant arthritis: importance of local administration. *Br. J. Pharmac.* **118**, 1754–1760.
- Wilson, M. A. and Tonegawa, S. (1997) Synaptic plasticity, place cells and spatial memory: studies with second generation knockouts. *Trends Neurosci.* 20, 102–106.
- Wilson, P. and Kitchener, P. D. (1996) Plasticity of cutaneous primary afferent projections to the spinal dorsal horn. *Prog. Neurobiol.* 48, 105–129.
- Wilson, P., Kitchener, P. D. and Snow, P. J. (1996b) Intraaxonal injection of neurobiotin reveals the long-ranging projections of $A\beta$ -hair follicle afferent fibers to the cat dorsal horn. J. Neurophysiol. 76, 242–254.
- Wimalawansa, S. J. (1996) Calcitonin gene-related peptide and its receptors: molecular genetics, physiology, pathophysiology and therapeutic potential. *Endocrine Rev.* 17, 533–585.
 Wimalawansa, S. J. and El Kholy, A. A. (1993) Comparative
- Wimalawansa, S. J. and El Kholy, A. A. (1993) Comparative study of distribution and biochemical characterization of brain

calcitonin gene-related peptide receptors in five different species. *Neuroscience* 54, 513-519.

- Winder, D. G. and Conn, P. J. (1995) Metabotropic glutamate receptor (mGlu receptor)-mediated potentiation of cyclic AMP responses does not require phosphoinositide hydrolysis: mediation by a group II-like mGlu receptor. J. Neurochem. 64, 592–599.
- Winder, D. G. and Conn, P. J. (1996) Roles of metabotropic glutamate receptors in glial function and glial-neuronal communication. J. Neurosci. Res. 46, 131–137.
- Winder, D. G., Mansuy, I. M., Osman, M., Moallem, T. M. and Kandel, E. R. (1998) Genetic and pharmacological evidence for a novel, intermediate phase of long-term potentiation suppressed by calcineurin. *Cell* **92**, 25–37.
- Winter, J. (1998) Brain derived neurotrophic factor, but not nerve growth factor, regulates capsaicin sensitivity of rat vagal ganglion neurones. *Neurosci. Lett.* 241, 21–24.
- Winter, J., Evison, C. J., O'Brien, C., Benowitz, L., Lindsay, R. M., Mulderry, P. and Woolf, C. (1992) Neurotoxic damage evokes regenerative responses from adul rats sensory neurones. *Neurosci. Lett.* 146, 48–52.
- Winter, J., Walpole, C. S. J., Bevan, S. and James, I. F. (1993) Characterization of resiniferatoxin binding sites on sensory neurons: co-regulation of resinferatoxin binding and capsaicinsensitivity in adult rat dorsal root ganglia. *Neuroscience* 57, 747–757.
- Winter, J., Bevan, S. and Campbell, E. A. (1995) Capsaicin and pain mechanisms. Br. J. Anaesthesia 75, 157–168.
- Wisden, W., Errington, M. L., Williams, S., Dunnett, S. B., Waters, C., Hitchcock, D., Evan, G., Bliss, T. V. P. and Hunt, S. P. (1990) Differential expression of immediate early genes in the hippocampus and spinal cord. *Neuron* 4, 603–614.
- Witte, O. W., Buchkremer-Ratzmann, I., Schiene, K., Neumann-Haefelin, T., Hagemann, G., Kraemer, M., Zilles, K. and Freund, H. J. (1997) Lesion-induced network plasticity in remote brain areas. *Trends Neurosci.* 20, 348–349.
- Wolfe, F., Russell, I. J., Vipraio, G., Ross, K. and Andersson, J. (1997) Serotonin levels, pain threshold, and fibromyalgia symptoms in the general population. J. Rheumatol. 24, 555.
- Won, C-K., Oh, S-J., Jung, S-C., Choi, Y-R., Kim, Y-I., Leem, J. W. and Shin, H-C. (1997) Activity-dependent conduction velocity changes of Aδ fibers in a rat model of neuropathy. *NeuroReport* 8, 3201–3205.
- Wong, J. and Oblinger, M. M. (1990) A comparison of peripheral and central axotomy effects on neurofilament and tubulin gene expression in rat dorsal root ganglion neurons. J. Neurosci. 10, 2215–2222.
- Wong, J. and Oblinger, M. M. (1991) NGF rescues substance P expression but not neurofilament or tubulin gene expression in axotomized sensory neurons. J. Neurosci. 11, 543–552.
- Wood, J. N. and Docherty, R. J. (1997) Chemical activators of sensory neurons. A. Rev. Physiol. 59, 457–482.
- Woolf, C. J. (1983) Evidence for a central component of postinjury pain hypersensitivy. *Nature* 306, 686–688.
- Woolf, C. J. (1994) A new strategy for the treatment of inflammatory pain. Prevention or elimination of central sensitization. *Drugs* 47, 1–9.
- Woolf, C. J. (1996) Phenotypic modification of primary sensory neurons: the role of nerve growth factor in the production of persistent pain. *Philos. Trans. R. Soc. Lond. [Biol.]* 351, 441– 448.
- Woolf, C. J. and Chong, M. S. (1993) Preemptive analgesia: treating postoperative pain by preventing the establishment of central sensitization. *Anesth. Analg.* **77**, 362–379.
 Woolf, C. J. and Doubell, T. P. (1994) The pathophysiology of
- Woolf, C. J. and Doubell, T. P. (1994) The pathophysiology of chronic pain increases sensitivity to low threshold $A\beta$ -fibre inputs. *Curr. Opin. Neurobiol.* **4**, 525–534.
- Woolf, C. J. and King, A. E. (1989) Subthreshold components of the cutaneous mechanoreceptive fields of dorsal horn neurons in the rat lumbar spinal cord. J. Neurophysiol. 62, 907–916.
- Woolf, C. J. and King, A. E. (1990) Dynamic alterations in the cutaneous mechanoreceptive fields of dorsal horn neurons in the rat spinal cord. J. Neurosci. 10, 2717–2726.
- Woolf, C. J. and McMahon, S. B. (1985) Injury-induced plasticity of the flexor reflex in chronic decerebrate rats. *Neuroscience* 16, 395–404.
- Woolf, C. J. and Thompson, S. W. N. (1991) The induction and maintenance of central sensitization is dependant on N-methyl-D-aspartate receptor activation: implications for the treatment of post-injury pain hypersensitivity states. *Pain* 44, 293–299.
- Woolf, C. J. and Wall, P. D. (1986) Relative effectiveness of C primary afferent fibers of different origins in evoking a prolonged

facilitation of the flexor reflex in the rat. J. Neurosci. 6, 1433–1442.

- Woolf, C. J. and Wiesenfeld-Hallin, Z. (1986) Substance P and calcitonin gene-related peptide synergistically modulate the gain of the nociceptive flexor withdrawal reflex in the rat. *Neurosci. Lett.* 66, 226–230.
- Woolf, C. J., Reynolds, M. L., Molander, C., O'Brien, C., Lindsay, R. M. and Benowitz, L. I. (1990) The growth-associated protein GAP-43 appears in dorsal root ganglion cells and in the dorsal horn of the rat spinal cord following peripheral nerve injury. *Neuroscience* 34, 465–478.
- Woolf, C. J., Shortland, P. and Coggeshall, R. E. (1992) Peripheral nerve injury triggers central sprouting of myelinated afferents. *Nature* 355, 75–78.
- Woolf, C. J., Safieh-Garabedian, B., Ma, Q-P., Crilly, P. and Winter, J. (1994) Nerve growth factor contributes to the generation of inflammatory sensory hypersensitivity. *Neuroscience* 62, 327–331.
- Woolf, C. J., Ma, Q-P., Allchorne, A. and Poole, S. (1996) Peripheral cell types contributing to the hyperalgesic action of nerve growth factor in inflammation. J. Neurosci. 16, 2716– 2723.
- Woolf, C. J., Allchorne, A., Safieh-Garabedian, B. and Poole, S. (1997) Cytokines, nerve growth factor and inflammatory hyperalgesia: the contribution of tumour necrosis factor α. Br. J. Pharmac. 121, 417–424.
- Wright, C. E. and Angus, J. A. (1996) Effects of N-, P- and Qtype neuronal calcium channel antagonists on mammalian peripheral neurotransmission. Br J. Pharmac. 119, 49–56.
- Wright, D. E. and Snider, W. D. (1995) Neurotrophin receptor expression defines distinct subpopulations of neurons in rat dorsal root ganglia. J. Comp. Neurol. 351, 329–338.
- Wright, D. E., White, F. A., Gerfen, R. W., Silos-Santiago, I. and Snider, W. D. (1995) The guidance molecule semaphorin III is expressed in regions of spinal cord and periphery avoided by growing sensory axons. *J. Comp. Neurol.* 361, 321–333.
 Wu, W. and Wessendorf, M. W. (1992) Organization of the sero-
- Wu, W. and Wessendorf, M. W. (1992) Organization of the serotonergic innervation of spinal neurons in rats. I Neuropeptide coexistence in varicosities innervating some spinothalamic tract neurons but not in those innervating postsynaptic dorsal column neurons. *Neuroscience* **50**, 855–898.
- Xi, Z.-X., Fuller, S. A. and Stein, E. A. (1998) Dopamine release in the nucleus accumbens during heroin self-administration is modulated by *kappa* opioid receptors: an in vivo fast-cyclic voltammetry study. J. Pharmac. Exp. Ther. 284, 151–161.
- Xiao, W. H. and Bennett, G. J. (1995) Calcium channel blockade with synthetic omega-conopeptides applied to the site of nerve injury suppresses neuropathic pain in rats. J. Pharmac. Exp. Ther. 274, 666–672.
- Xie, J., Yoon, Y. W., Yom, S. S. and Chung, J. M. (1995a) Norepinephrine rekindles mechanical allodynia in sympathectomized neuropathic rats. *Analgesia* 12, 107–113.
- Xie, Y., Zhang, J., Petersen, M. and LaMotte, R. H. (1995b) Functional changes in dorsal root ganglion cells after chronic nerve constriction in the rat. J. Neurophysiol. 73, 1811–1820.
- Xu, L., Anwyl, R. and Rowan, M. J. (1997a) Behavioural stress facilitates the induction of long-term depression in the hippocampus. *Nature* 387, 497–500.
- Xu, T. L., Nabekura, J. and Akaike, N. (1996a) Protein kinase Cmediated enhancement of glycine responses in rat sacral dorsal commissural neurones by serotonin. J. Physiol. 496, 491–501.
- Xu, X., Fukuyama, H., Yazawa, S., Mima, T., Hanakawa, T., Magata, Y., Kanda, M., Fujiwara, N., Shindo, K., Nagamine, T. and Shibasaki, H. (1997b) Functional localization of pain perception in the human brain studied by PET. *NeuroReport* 8, 555-559.
- Xu, X. J. and Wiesenfeld-Hallin, Z. (1996) Intrathecal pituitary adenylate cyclase activating polypeptide facilitates the spinal nociceptive flexor reflex in the rat. *Neuroscience* 72, 801–804.
- Xu, X. J. and Wiesenfeld-Hallin, Z. (1997) Novel modulators in nociception. In: *The Pharmacology of Pain. Handbook of Experimental Pharmacology*, Vol. 130, pp. 211–234. Eds. A. Dickenson and J-M. Besson. Springer-Verlag, Berlin.
- Xu, X. J., Wiesenfeld-Hallin, Z., Villar, M. J., Fahrenkrug, P. and Hökfelt, T. (1990) On the role of galanin, substance P and other neuropeptides in primary sensory neurons of the rat: studies on spinal reflex excitability and peripheral axotomy. *Eur. J. Neurosci.* 2, 733–743.
- Xu, X. J., Dalsgaard, C-J. and Wiesenfeld-Hallin, Z. (1992) Spinal substance P and N-methyl-D-aspartate receptors are coactivated in the induction of central sensitization of the nociceptive flexor reflex. Neuroscience 51, 641–648.

- Xu, X. J., Hao, J. X., Seiger, A. and Wiensenfeld-Hallin, Z. (1993a) Systemic excitatory amino acid receptor antagonists of the α-amino-3-hydroxy-5-methyl-4-isoxazolepropionic acid (AMPA) receptor and of the N-methyl-D-aspartate (NMDA) receptor relieve mechanical hypersensitivity after transient spinal cord ischemia in rats. J. Pharmac. Exp. Ther. 267, 140–144. Xu, X. J., Puke, M. J. C., Vergé, V. M. K., Wiesenfeld-Hallin, Z.,
- Xu, X. J., Puke, M. J. C., Verge, V. M. K., Wiesenfeld-Hallin, Z., Hughes, J. and Hökfelt, T. (1993b) Up-regulation of cholecystokinin in primary sensory neurons is associated with morphine insensitivity in experimental neuropathic pain in the rat. *Neurosci. Lett.* **152**, 129–132.
- Xu, X. J., Hökfelt, T., Hughes, J. and Wiesenfeld-Hallin, Z. (1994) The CCK-B antagonist CI 988 enhances the reflex depressive effect of morphine in axotomized rats. *NeuroReport* 5, 718–720.
- Xu, X. J., Andell, S., Zhang, X., Wiesenfeld-Hallin, Z., Langel, U., Bedecs, K., Hökfelt, T. and Bartfai, T. (1995a) Peripheral axotomy increases the expression of galanin message-associated peptide (GMAP) in dorsal root ganglion cells and alters the effects of intrathecal GMAP on the flexor reflex in the rat. *Neuropeptides* 28, 299–307.
- Xu, X. J., Zhang, X., Hökfelt, T. and Wiesenfeld-Hallin, Z. (1995b) Plasticity in spinal nociception after peripheral nerve section: reduced effectiveness of the NMDA receptor antagonist MK-801 in blocking wind-up and central sensitization of the flexor reflex. *Brain Res.* 670, 342–346.
- Xu, X. J., Farkas-Szallasi, T., Lundberg, J. M., Hökfelt, T., Wiesenfeld-Hallin, Z. and Szallasi, A. (1997c) Effects of capsaicin analogue resiniferatoxin on spinal nociceptive mechanisms in the rat: behavioral, electrophysiological and in situ hybridization studies. *Brain Res.* **752**, 52–60.
- Xu, X. M., Guénard, V., Kleitman, N., Aebischer, P. and Bartlett Bunge, M. (1995c) A combination of BDNF and NT-3 promotes supraspinal axonal regeneration into Schwann cell grafts in adult rat thoracic spinal cord. *Exp. Neurol.* 134, 261–272.
- Xu, Z. Q., Shi, T. J., Landry, M. and Hökfelt, T. (1996b) Evidence for galanin receptors in primary sensory neurones and effect of axotomy and inflammation. *NeuroReport* 8, 237–242.
- Xu, Z. Q., Zhang, X., Grillner, S. and Hökfelt, T. (1997d) Electrophysiological studies on rat dorsal root ganglion neurons after peripheral axotomy: changes in responses to neuropeptides. *Proc. natl Acad. Sci. U.S.A.* 94, 13262–13266.
- Yabuuchi, K., Maruta, E., Yamamoto, J., Nishiyori, A., Takami, S., Minami, M. and Satoh, M. (1997) Intracerebroventricular injection of isoproterenol produces its analgesic effect through interleukin-1β production. *Eur. J. Pharmac.* 334, 133–140. Yajiri, Y., Yoshimura, M., Okamoto, M., Takahashi, H. and
- Yajiri, Y., Yoshimura, M., Okamoto, M., Takahashi, H. and Higashi, H. (1997) A novel slow excitatory postsynaptic current in substantia gelatinosa neurons of the rat spinal cord in vitro. *Neuroscience* **76**, 673–688.
- Yaksh, T. L. (1989) Behavioral and autonomic correlates of the tactile evoked allodynia produced by spinal glycine inhibition: effects of modulatory receptor systems and excitatory amino acid antagonists. *Pain* 37, 111–123.
- Yaksh, T. L. and Malmberg, A. B. (1994) Central pharmacology of nociceptive transmission. In: *Textbook of Pain*, 3rd edn, pp. 165–200. Eds. P. D. Wall and R. Melzack. Churchill-Livingstone, Edinburgh.
- Yaksh, T. L., Pogrel, J. W., Lee, Y. W. and Chaplan, S. R. (1995) Reversal of nerve ligation-induced allodynia by spinal alpha₂ adrenoceptor agonists. J. Pharmac. Exp. Ther. 272, 207–214.
- Yamagata, K., Andreasson, K. I., Kaufmann, W. E., Barnes, C. A. and Worley, P. F. (1993) Expression of a mitogen-inducible cyclooxygenase in brain neurons: regulation by synaptic activity and glucocorticoids. *Neuron* 11, 371–386.
- Yamamoto, T. and Nozaki-Taguchi, N. (1996) Analysis of the effects of cyclooxygenase (COX)-1 and COX-2 in spinal nociceptive transmission using indomethacin, a non-selective COX inhibitor, and NS-398, a COX-2 selective inhibitor. *Brain Res.* 739, 101–110.
- Yamamoto, T. and Nozaki-Taguchi, N. (1997) Role of spinal cyclooxygenase (COX)-2 on thermal hyperalgesia evoked by carageenan injection in the rat. *NeuroReport* 8, 2179–2182.
- Yamamoto, T. and Tatsuno, I. (1995) Antinociceptive effect of intrathecally administered pituitary adenylate cyclase activating polypeptide (PACAP) on the rat formalin test. *Neurosci. Lett.* 184, 32–35.
- Yamamoto, T. and Yaksh, T. L. (1993) Effects of intrathecal strychnine and bicuculline on nerve compression-induced thermal hyperalgesia and selective antagonism by MK-801. *Pain* 54, 79–84.
- Yan, Q., Radeke, M. J., Matheson, C. R., Talvenheimo, J., Weltcher, A. A. and Feinstein, S. C. (1997) Immunocytochemical localization of trkB in the central nervous system of the adult rat. J. Comp. Neurol. 378, 135–157.

162

- Yang, L., Marsala, M. and Yaksh, T. L. (1996) Characterization of time course of spinal amino acids, citrulline, and PGE₂ release after carrageenan/kaolin-induced knee joint inflammation: a chronic microdialysis study. *Pain* 67, 345–354.
- Yang, S-W., Zhang, Z-H., Chen, J-Y., Xie, Y-F., Qiao, J. T. and Dafny, N. (1994) Morphine and norepinephrine induced antinociception at the spinal level is mediated by adenosine. *NeuroReport* 5, 1441–1444.
- Yang, S-W., Guo, Y. Q., Kang, Y. M., Qiao, J. T., Laufman, L. E. and Dafny, N. (1998) Different GABA-receptor types are involved in the 5-HT-induced antinociception at the spinal level: a behavioral study. *Life Sci.* 62, 143–148.
- Yao, G. L., Tohyama, M. and Senba, E. (1992) Histamine-caused itch induces Fos like immunoreactivity in dorsal horn neurons: effect of morphine pretreatment. *Brain Res.* 599, 333–337.
- Yao, G. L., Kato, H., Khalil, M., Kiryi, S. and Kiyama, H. (1997) Selective upregulation of cytokine receptor subchain and their intracellular signalling molecules after peripheral nerve injury. *Eur. J. Neurosci.* 9, 1047–1054.
- Yashpal, K., Pitcher, G. M., Parent, A., Quirion, R. and Coderre, T. J. (1995) Noxious thermal and chemical stimulation induce increases in ³H-phorbol 12,13-dibutyrate binding in spinal cord dorsal horn as well as persistent pain and hyperalgesia which is reduced by inhibition of protein kinase C. J. Neurosci. 15, 3263–3272.
- Yasui, Y., Saper, C. B. and Cechetto, D. F. (1989) Calcitonin gene-related peptide immunoreactivity in the visceral sensory cortex, thalamus and related pathways in the rat. J. Comp. Neurol. 290, 487–501.
- Ye, Z. and Westlund, K. N. (1996) Ultrastructural localization of glutamate receptor subunits (NMDAR1, AMPA GluR1 and GluR2/3) on spinothalamic tract cells. *NeuroReport* 7, 2581– 2585.
- Yeomans, D. C. and Proudfit, H. K. (1996) Nociceptive responses to high and low rates of noxious cutaneous heating are mediated by different nociceptors in the rat: electrophysiological evidence. *Pain* 68, 141–150.
- Yezierski, R. P., Gerhart, K. D., Schrock, B. J. and Willis, W. D. (1983) A further examination of effects of cortical stimulation on primate spinothalamic tract cells. J. Neurophysiol. 49, 424–441.
- Yin, J. C. P. and Tully, T. (1996) CREB and the formation of long-term memory. *Curr Opin. Neurobiol.* 6, 264–268.
- Yoder, E. J., Tamir, H. and Ellisman, M. H. (1997) Serotonin receptors expressed by myelinating Schwann cells in rat sciatic nerve. *Brain Res.* **753**, 299–308.
- Yonehara, N., Takemura, M. and Shigenaga, Y. (1997) Involvement of nitric oxide in re-innervation of rat molar tooth pulp following transection of the inferior alveolar nerve. *Brain Res.* **757**, 31–36.
- Yoon, Y. W., Na, H. S. and Chung, J. M. (1996) Contributions of injured and intact afferents to neuropathic pain in an experimental rat model. *Pain* 64, 27–36.
- Yoshida, A., Chen, K., Moritani, M., Yabuta, N. H., Nagase, Y., Takemura, M. and Shigenaga, Y. (1997) Organization of the descending projections from the parabrachial nuclear complex and spinal dorsal horn in the rat. J. Comp. Neurol. 383, 94–111.
- Yoshimura, M. and Jessel, T. (1990) Amino acid-mediated EPSPs at primary afferent synapses with substantia gelatinosa neurones in the rat spinal cord. J. Physiol. 430, 315–335.
- Yoshimura, M. and Nishi, S. (1992) Excitatory amino acid receptors involved in primary afferent-evoked polysynaptic EPSPs of substantia gelatinosa neurons in the adult rat spinal cord slice. *Neurosci. Lett.* 143, 131–134.
- Yoshimura, M. and Nishi, S. (1995) Primary afferent-evoked glycine and GABA-mediated IPSPs in substantia gelatinosa neurones in the rat spinal cord in vitro. J. Physiol. 482, 29–38.
- Youn, S. H., Sakuda, M., Kurisu, K. and Wakisaka, S. (1997) Regeneration of periodontal primary afferents of the rat incisor following injury of the inferior alveolar nerve with special reference to neuropeptide Y-like immunoreactive primary afferents. *Brain Res.* **752**, 161–169.
- Young, K. W., Pinnock, R. D., Gibson, W. J. and Young, J. M. (1998) Dual effects of histamine and substance P on intracellular calcium levels in human U373 MG astrocytoma cells: role of protein kinase C. Br. J. Pharmac. 123, 545–557.
- Young, M. R., Fleetwood-Walker, S. M., Mitchell, R. and Munro, F. E. (1994) Evidence for a role of metabotropic glutamate receptors in sustained nociceptive inputs to rat dorsal horn neurons. *Neuropharmacology* 33, 141–144.
- Young, M. R., Fleetwood-Walker, S. M., Mitchell, R. and Dickinson, T. (1995) The involvement of metabotropic glutamate receptors and their intracellular signalling pathways in sus-

tained nociceptive transmission in rat dorsal horn neurons. *Neuropharmacology* **34**, 1033–1041.

- Young, M. R., Fleetwood-Walker, S. M., Dickinson, T., Blackburn-Munro, G., Sparrow, H., Birch, P. J. and Bountra, C. (1997) Behavioural and electrophysiological evidence supporting a role for group I metabotropic glutamate receptors in the mediation of nociceptive inputs to the rat spinal cord. *Brain Res.* 777, 161–169.
- Yu, L. C., Hansson, P., Brodda-Jansen, G., Theodorsson, E. and Lundeberg, T. (1996) Intrathecal CGRP₈₋₃₇-induced bilateral increase in hindpaw withdrawal latency in rats with unilateral inflammation. Br. J. Pharmac. 117, 43–50.
- Yu, X.-J., Cutrer, F. M., Moskowitz, M. A. and Waeber, C. (1997) The 5-HT_{1D} receptor antagonist GR-127 935 prevents inhibitory effects of sumatriptan but not CP-122 288 and 5-CT on neurogenic plasma extravasion within guinea pig dura mater. *Neuropharmacology* 36, 83–91.
- Yuan, Y., Huang, Q. and Wu, H. M. (1997) Myosin light chain phosphorylation: modulation of basal and agonist stimulated venula permeability. *Am. J. Physiol.* 272, H1437.
- Zachariou, V., Goldstein, B. D. and Yeomans, D. C. (1997) Low but not high rate noxious radiant skin heating evokes a capsaicin-sensitive increase in spinal cord dorsal release of substance P. Brain Res. 752, 143–150.
- Zadina, J. E., Hackler, L., Ge, L. J. and Kastin, A. J. (1997) A potent and selective endogenous agonist for the μ-opiate receptor. *Nature* 386, 499–502.
- Zafra, F., Castrén, E., Thoenen, H. and Lindholm, D. (1991) Interplay btween glutamate and y-aminobutyric acid transmitter systems in the physiological regulation of brain derived neurotrophic factor and nerve growth factor synthesis in hippocampal neurons. *Proc. natl Acad. Sci. U.S.A.* 88, 10037–10041.
- von Zahn, J., Möller, T., Kettenmann, H. and Nolte, C. (1997) Microglial phagocytosis is modulated by pro- and anti-inflammatory cytokines. *NeuroReport* 8, 3851–3856. Zarri, I., Bucossi, G., Cupello, A., Rapallino, M. W. and Robello,
- Zarri, I., Bucossi, G., Cupello, A., Rapallino, M. W. and Robello, M. (1994) Modulation by nitric oxide of rat brain GABA_A receptors. *Neurosci. Lett.* 180, 1895–1905.
- Zeilhofer, H. U., Swandulla, D., Reeh, P. W. and Kress, M. (1996) Ca²⁺-permeability of the sustained proton-induced cation current in adult rat dorsal ganglion neurons. J. Neurophysiol. 76, 2834–2840.
- Zeilhofer, H. U., Kress, M. and Swandulla, D. (1997) Fractional Ca²⁺ currents through capsaicin- and proton-activated ion channels in rat dorsal root ganglia neurones. J. Physiol. 503, 67–78.
- Zemlan, F. P., Kow, L. and Pfaff, D. W. (1983) Spinal serotonin (5-HT) receptor subtypes and nociception. J. Pharmac. Exp. Ther. 226, 477–485.
- Zhang, A. L., Hao, J. X., Seiger, A., Xu, X. J., Wiesenfeld-Hallin, Z., Grant, G. and Aldskogius, H. (1994a) Decreased GABA immunoreactivity in spinal cord dorsal horn neurons after transient spinal cord ischemia in the rat. *Brain Res.* 656, 187–190.
- Zhang, E. T. and Craig, A. D. (1997) Morphology and distribution of spinothalamic lamina I neurons in the monkey. J. Neurosci. 17, 3274–3284.
- Zhang, H. Q., Murray, G. M., Turman, A. B., Mackie, P. D., Coleman, G. T. and Rowe, M. J. (1996a) Parallel processing in cerebral cortex of the marmoset monkey: effect of reversible SI inactivation on tactile responses in S II. J. Neurophysiol. 76, 3633–3655.
- Zhang, J. and Snyder, S. H. (1995) Nitric oxide in the nervous system. A. Rev. Pharmac. Toxicol. 35, 213–233.
- Zhang, J. M., Kitahata, L. M. and LaMotte, R. H. (1994b) Verapamil inhibits the spontaneous activities originating in dorsal root ganglion cells after chronic nerve constriction in rats. *Am. Soc. Neurosci.* 20, 760.
- Zhang, J-M., Donnelly, D-F., Song, X-J. and Lamotte, R. H. (1997a) Axotomy increases the excitability of dorsal root ganglion cells with unmyelinated axons. J. Neurophysiol. 78, 2790–2794.
- Zhang, J-M., Song, X-J. and LaMotte, R. H. (1997b) An *in vitro* study of ectopic discharge generation and adrenergic sensitivity in the intact, nerve-injured rat dorsal root ganglion. *Pain* 72, 51–57.
- Zhang, L., Peoples, R. W., Oz, M., Harvey-White, J., Weight, F. F. and Brauneis, U. (1997c) Potentiation of N-methyl-D-aspartate receptor-mediated responses by dynorphin at low extracellular glvcine concentrations. J. Neurophysiol. **78**, 582–590.
- cellular glycine concentrations. J. Neurophysiol. 78, 582–590.
 Zhang, L. J., Wang, X. J. and Hans, J. S. (1990) Phorbol ester suppression of opioid analgesia in rats. Life Sci. 47, 1775–1782.
- Zhang, R. X., Ruda, M. A. and Qiao, J. T. (1998a) Pre-emptive intrathecal MK-801, a non-competitive N-methyl-D-aspartate

receptor antagonist, inhibits the up-regulation of spinal dynorphin mRNA and hyperalgesia in a rat model of chronic inflammation. *Neurosci. Lett.* **241**, 57–60.

- Zhang, S. P. and Rowe, M. J. (1997) Quantitative analysis of cuneate neurone responsiveness in the cat in association with reversible, partial deafferentation. J. Physiol. 505, 769–783.
- reversible, partial deafferentation. J. Physiol. 505, 769–783.
 Zhang, X., Dagerlind, A., Elde, R. P., Castel, M. N., Broberger, C., Wiesenfeld-Hallin, Z. and Hökfelt, T. (1993a) Marked increase in cholecystokinin B receptor messenger RNA levels in rat dorsal root ganglia after peripheral axotomy. Neuroscience 57, 227–233.
- Zhang, X., Vergé, V., Wiensenfeld-Hallin, Z., Ju, G., Bredt, D., Snyder, S. H. and Hökfelt, T. (1993b) Nitric oxide synthaselike immunoreactivity in lumbar dorsal root ganglia and spinal cord of rat and monkey and effect of peripheral axotomy. J. Comp. Neurol. 335, 563–575.
- Zhang, X., Wiesenfeld-Hallin, Z. and Hökfelt, T. (1994c) Effect of peripheral axotomy on expression of neuropeptide Y receptor mRNA in rat lumbar dorsal root ganglia. *Eur. J. Neurosci.* 6, 43–57.
- Zhang, X., Bean, A. J., Wiesenfeld-Hallin, Z., Xu, X.-J. and Hökfelt, T. (1995) Ultrastructural studies on peptides in the dorsal horn of the rat spinal cord—III. Effects of peripheral axotomy with special reference to galanin. *Neuroscience* 64, 893–915.
- Zhang, X., La Salle, G., Ridoux, V., Yu, P. H. and Ju, G. (1997d) Prevention of kainic acid-induced limbic seizures and fos expression by the GABA-A receptor agonist muscimol. *Eur. J. Neurosci.* 9, 29–40.
- Zhang, X., Shi, T., Holmberg, K., Landry, M., Huang, W., Xiao, H., Ju, G. and Hökfelt, T. (1997e) Expression and regulation of the neuropeptide Y (Y2) receptor in sensory and autonomic ganglia. *Proc. natl Acad. Sci. U.S.A.* 94, 729–734.
- Zhang, X., Bao, L., Shi, T-J., Ju, G., Elde, R. and Hökfelt, T. (1998b) Down-regulation of μ-opioid receptors in rat and monkey dorsal root ganglion neurons and spinal cord after peripheral axotomy. *Neuroscience* 82, 223–240.
- Zhang, Y-Q., Tang, J-S., Yuan, B. and Jia, H. (1997f) Inhibitory effects of electrically evoked activation of ventrolateral orbital cortex on the tail-flick reflex are mediated by periaqueductal gray in rats. *Pain* 72, 127–135.
- Zhang, Y. Z., Sjölund, B., Moller, K., Håkanson, R. and Sundler, F. (1993c) Pituitary adenylate cyclase activating peptide produces a marked and long-lasting depression of a C-fibre-evoked flexion reflex. *Neuroscience* 57, 733–737.
- Zhang, Y. Z., Hannibal, J., Zhao, Q., Moller, K., Danielsen, N., Fahrenkrug, J. and Sundler, F. (1996b) Pituitary adenylate cyclase activating peptide expression in the rat dorsal root ganglia: up-regulation after peripheral nerve injury. *Neuroscience* 74, 1099–1110.
- Zhang, Y. Z., Malmberg, A. B., Yaksh, T. L., Sjölund, B., Sundler, F. and Råkanson, R. (1997g) Capsaicin-evoked release

of pituitary adenylate cyclase activating peptide (PACAP) and calcitonin gene-related peptide (CGRP) from rat spinal cord in vivo. *Regulatory Peptides* **69**, 83–87.

- Zheng, X., Zhang, L., Wang, A. P., Bennett, M. V. L. and Zukin, R. S. (1997) Ca²⁺ influx amplifies protein kinase C potentiation of recombinant NMDA receptors. *J. Neurosci.* 17, 8676–8686.
- Zhou, F.-M. and Hablitz, J-J. (1997) Metabotropic glutamate receptor enhancement of spontaneous IPSCs in neocortical interneurons. J. Neurophysiol. 78, 2287–2295.
- Zhou, L.-M., Gu, Z. Q., Costa, A. M., Yamada, K. A., Mansson, P. E., Giordano, T., Skolnick, P. and Jones, K. A. (1997a) (2S, 4R)-4-Methylglutamic acid (SYM 2081): a selective, high-affinity ligand for kainate receptors. J. Pharmac. Exp. Ther. 280, 422-427.
- Zhou, Q., Godwin, D. W., Omalley, D. M. and Adams, P. R. (1997b) Visualization of calcium influx through channels that shape size burst and tonic firing modes of thalamic relay cells. *J. Neurophysiol.* 77, 2816–2825.
- Zhou, S., Bonasera, L. and Carlton, S. M. (1996a) Peripheral administration of NMDA, AMPA or KA results in pain behaviors in rats. *NeuroReport* 7, 895–900.
- Zhou, X. F. and Rush, R. A. (1994) Localization of neurotrophin-3-like immunoreactivity in the rat central nervous system. *Brain Res.* 643, 162–172.
- Zhou, X. F. and Rush, R. A. (1996) Endogenous brain-derived neurotrophic factor is anterogradely transported in primary sensory neurons. *Neuroscience* 74, 945–951.
- Zhou, X. F., Parada, L. F., Soppet, D. and Rush, R. A. (1993) Distribution of *trkB* tyrosine kinase immunoreactivity in the rat central nervous system. *Brain Res.* 622, 63–70.
- Zhou, X. F., Rush, R. A. and McLachlan, E. M. (1996b) Differential expression of the p75 nerve growth factor receptor in glia and neurons of the rat dorsal root ganglia after peripheral nerve section. J. Neurosci. 16, 2901–2911.
- Zhuo, M. and Gebhart, G. F. (1990) Spinal cholinergic and monoaminergic receptors mediate descending inhibition from the nuclei reticularis gigantocellularis and gigantocellularis pars alpha in the rat. *Brain Res.* 535, 67–78.
- Zhuo, M. and Gebhart, G. F. (1992) Characterization of descending facilitation and inhibition of spinal nociceptive transmission from the nuclei reticularis gigantocellularis and gigantocellularis pars alpha in the rat. J. Neurophysiol. 67, 1599–1614.
- Zhuo, M. and Gebhart, G. F. (1997) Biphasic modulation of spinal nociceptive transmission from the medullary raphe nuclei in the rat. J. Neurophysiol. 78, 746–758.
- Zochodne, D. W., Misra, M., Cheng, C. and Sun, H. (1997) Inhibition of nitric oxide synthase enhances peripheral nerve regeneration in mice. *Neurosci. Lett.* 228, 71–74.
- Zoli, M. and Agnati, L. F. (1996) Wiring and volume transmission in the central nervous system: the concept of closed and open synapses. *Prog. Neurobiol.* 49, 363–380.